







780 N. Commercial Street P.O. Box 330 Manchester, NH 03101-0330

Jessica Chiavara Senior Counsel, Eversource Energy

Phone: 603-634-2972 jessica.chiavara@eversource.com

December 16, 2022

Via Electronic Mail Daniel Goldner, Chair New Hampshire Public Utilities Commission 21 S. Fruit Street, Suite 10 Concord, NH 03301-2429

Re: IR 22-042 Investigation of Energy Efficiency Planning, Programming, and Evaluation

Joint responses to Commission inquiries by the NH Utilities

Chair Goldner:

In response to the inquiries issued by the New Hampshire Public Utilities Commission in this docket on November 1 and 4, 2022, and consistent with the procedural order issued on November 23, 2022, extending the deadline for certain responses, Liberty Utilities (Granite State Electric) Corp. d/b/a Liberty; New Hampshire Electric Cooperative, Inc.; Public Service Company of New Hampshire d/b/a Eversource Energy; Unitil Energy Systems, Inc.; Liberty Utilities (EnergyNorth Natural Gas) Corp d/b/a Liberty; and Northern Utilities, Inc. (collectively, the "NH Utilities") submit the attached supplemental responses, adding to those provided by the NH Utilities on November 30.

Consistent with current Commission policy, this filing will be made electronically only; paper copies will not follow. Please contact me if there are any questions or further information is needed. Thank you.

Regards

Jessica A. Chiavara Senior Counsel, Eversource Energy o/b/o the NH Utilities

Attachments cc: IR 22-042 Service List

Date Request Received: November 01, 2022 Data Request No. RR 1-001 Date of Response: December 16, 2022 Page 1 of 12

Request from: New Hampshire Public Utilities Commission

Request:

<u>The Granite State Test, the Total Resource Cost Test, and Discount rates:</u> The Commission wants to better understand the assumptions and parameters used toestimate the energy efficiency savings generated by each program of the Energy Efficiency Program. The Commission requests that the Joint Utilities submit a joint response to each of the following questions:

- 1. Please create a document that lists the assumptions used to estimate the energy efficiency gains generated by each program. For each assumption, please cite its source; please include any notes or comments that would be helpful for the commission to understand why these sources were selected and how the assumptions have changed over the program's life.
- 2. Please report on assumptions used by the other New England States where the Joint Utilities operate or have visibility in developing their energy efficiency programs. Please explain the rationale for any differences between the assumptions used in the other New England States and those used in New Hampshire.
- 3. Have the Joint Utilities and their consultants conducted, or are they aware of, any analyses that estimate the real-world energy savings of energy efficiency investments? These analyses might include academic studies that use experimental or observational techniques to estimate the impacts that specific energy efficiency investments have on household, commercial and industrial energy consumption.
- 4. If the Energy Efficiency Program were to directly or indirectly provide funding for smart meters, how might these investments impact the collection of customer data on energy efficiency savings? How would smart meters impact demand-side energy management generally?
- 5. How do the Joint Utilities determine when free ridership has a material impact on savings figures? How do the Joint Utilities quantify this impact? Please provide answers with respect to the impacts on both the GST and TRCT.
- 6. Do the Joint Utilities adjust the energy savings calculations to account for the incremental energy demand associated with the projected positive impact on the NH

Date Request Received: November 01, 2022 Data Request No. RR 1-001

Date of Response: December 16, 2022 Page 2 of 12

economy generated by the Energy Efficiency Program? Please identify the economic impacts of the Energy Efficiency Program and explain how these are considered by the JointUtilities.

- 7. The Commission wants to understand how other regulatory entities determine the discount rate used in evaluating the cost-effectiveness of their energy efficiency programs. Please report on the following:
 - i. Please identify the discount rates used by each of the other New England states in 2020 and 2021;
 - ii. To the extent the Joint Utilities operate or have visibility on the discount rates identified in I-7-i above, please describe the methodologies used to derive these discount rates; and
 - iii. Please identify the discount rate used by the U.S. Government to determine the cost-effectiveness of investments for federal energy management and planning programs.
- 8. Is using either an average of the rate-regulated utilities' Return on Equity (ROE) or using an average of the rate-regulated utilities' Weighted Average Cost of Capital (WACC) an approach that could be used to determine the discount rate used in evaluating the cost-effectiveness of energy efficiency investments? Why or why not?
- 9. When was the discount rate used to evaluate the cost effectiveness of NHEE Investments first developed? Has the discount-rate setting approach been evaluated or updated since?
- 10. What would the real discount rate be if the current methodology for calculating the discount rate was applied today? Please cite the day(s) that the prime rate and inflation rate data were collected in making this calculation.
- 11. Please create a single summary sheet that compares the cost-effectiveness ratios of the GST and TRCT for each program using each of the following discount rates:
 - i. An average of the Joint Utilities'ROEs;
 - ii. An average of the Joint Utilities' WACC; and
 - iii. Real discounts rates of 3%, 5.5%, 8%, and 10%.

Date Request Received: November 01, 2022 Data Request No. RR 1-001 Date of Response: December 16, 2022 Page 3 of 12

Response:

 Please reference Attachment A to the 2022-2023 Plan, submitted in docket DE 20-092. The New Hampshire Technical Reference Manual ("TRM") is a compendium of energy savings calculation and input assumptions for each measure offered by the NHSaves programs, including references to third party evaluations, ENERGY STAR[®] calculators, federal and state equipment standards and building codes that serve to inform those assumptions. The NH EM&V Working Group conducts an annual review of the TRM and updates the savings assumptions as needed based on relevant evaluation findings and changes to codes and standards. Each TRM contains a record of changes and updates made since the previous version. An electronic version of the TRM can be found at https://etrm.anbetrack.com/#/workarea/home?token=294a255e6c37343b615e.

Because each two or three-year term of the NHSaves Programs is considered in a separate docket and stands on its own, the Utilities have not maintained a record of how the assumptions have changed over time prior to the development of the TRM. However, for both planning and reporting, the Utilities utilize a Benefit Cost model that includes assumptions than can be compared year over year, as well as attachments to the Plan that include the primary assumptions for each measure for the current year and previous two years (e.g., for the 2022-2023 Plan, see Attachments E2, F2, G2, H2, I2, and J2 for the program summaries, which include each of the Utilities' measure lists and associated assumptions).

2. Please refer to the below links for the most recent documents:

Massachusetts Technical Reference Manual: <u>https://fileservice.eea.comacloud.net/FileService.Api/file/FileRoom/14154670</u> Electronic version: <u>https://www.masssavedata.com/Public/TechnicalReferenceLibrary</u>

Connecticut Program Savings Document:

https://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/7059ba bc24eec078852588ee00496229/\$FILE/Final%202023%20PSD%20(11-1-22).pdf

Rhode Island: https://ripuc.ri.gov/sites/g/files/xkgbur841/files/2022-10/2233-RIE-TechnicalReferenceManual2023%2010-20-22.pdf

Date Request Received: November 01, 2022 Data Request No. RR 1-001

Date of Response: December 16, 2022 Page 4 of 12

Maine:

https://www.efficiencymaine.com/docs/EMT-TRM_Commercial_Industrial_Multifamily_v2023_1.pdf, and https://www.efficiencymaine.com/docs/EMT-TRM_Retail_Residential_v2023_1.pdf

Vermont:

https://puc.vermont.gov/sites/psbnew/files/doc_library/Vermont%20TRM%20Savings%20V erification%202018%20Version_FINAL.pdf

Because each New England state's energy efficiency program is operated independently and based on each state's applicable regulations and statutes, and legislative and regulatory priorities and schedules, each state's technical reference manual differs. In developing both its programs and its assumptions, the Joint Utilities and third-party contractors operating in New Hampshire refer to research from other jurisdictions, particularly Massachusetts and Connecticut, when leveraged, throughout the technical reference manual. Differences in assumptions related to the savings of energy efficiency measures are explained by a myriad of factors including market saturation and penetration of each measure, climate conditions (which in turn impacts full load operating hours, heating and cooling degree days and energy savings), building codes and appliance standards, the inclusion or exclusion of non-energy impacts, implementation practices, and participant demographic/firmographic characteristics, among other variables.

The NH TRM includes insight regarding the assumptions that will help inform the Commission on this issue. Please refer to the response to Question RR 1-001 part 1 for a link to access the NH TRM. Regarding differences in assumptions used in other states versus those used in New Hampshire, the EM&V Working Group adopted the following language, which is contained in the Introduction to the TRM, on page 12:

Inputs and assumptions are based on New Hampshire-specific evaluations or data where available. Other factors being equal, New Hampshire jurisdiction-specific results will be favored over results from other jurisdictions in order to account for differences in climate, hours of use, program design and delivery, market conditions, and evaluation frameworks. However, when relevant results exist both from New Hampshire and from other states, it may be necessary to balance the desirable attributes of state-specificity and data reliability. When considering whether to apply results from a study originating in another jurisdiction to New Hampshire programs, the EM&V Working Group (with support from independent evaluation firms as needed), will make the determination based

Date Request Received: November 01, 2022 Data Request No. RR 1-001

Date of Response: December 16, 2022 Page 5 of 12

on (1) the similarity of evaluated program/measures to those offered in NH; (2) the similarity of relevant markets and customers base; (3) the recency of the study relative to the recency of any applicable NH results; and (4) the quality of the study's methodology and sample size. In addition to third-party evaluations, inputs may also be based on sources including manufacturer and industry data, data from government agencies such as the U.S. Department of Energy or Environmental Protection Agency, or credible and realistic factors developed using engineering judgment.

3. Yes, the TRM documents and provides hyperlinks to the extensive research performed by third-party evaluation firms experienced in researching the real-world energy savings resulting from energy efficiency programs. The studies utilize industry standards and proven EM&V approaches that follow guidance from International Performance Measurement and Verification Protocol (IPMVP) and DOE Uniform Methods Project. These studies typically employ various types of research methods based on quasi- and experimental designs, indepth interviews with customers, vendors, distributors and other market actors, billing analysis and on-site metering to estimate or verify savings. Completed evaluations for New Hampshire, Massachusetts, and Connecticut are posted at the following websites.

New Hampshire: https://www.puc.nh.gov/electric/Monitoring Evaluation Report List.htm

Massachusetts: https://ma-eeac.org/studies/

Connecticut: https://energizect.com/eeb-evaluation-reports-and-studies

4. "Smart meters" do not produce energy efficiency savings and therefore would not be costeffective under the Granite State Test. Smart meters provide customers and utilities with data on the aggregate electric usage behind the meter at periodic intervals throughout the day. However, for energy efficiency purposes, the meter would not provide data at a granular enough level of detail to determine how much energy any given piece of equipment was using, nor would it be able to answer the question of how much a different, more (or less) efficient piece of equipment *would have* used.

Outside of the energy efficiency program context, the additional data generated by smart meters potentially could allow for connection with home area network (HAN) devices, which can provide customers with real-time consumption data. This configuration could allow a customer to better understand how their own electricity-using appliances and devices as well as usage behaviors impact their behind-the-meter consumption over time.

Date Request Received: November 01, 2022 Data Request No. RR 1-001

Date of Response: December 16, 2022 Page 6 of 12

https://www.cpuc.ca.gov/industries-and-topics/electrical-energy/infrastructure/the-benefitsof-smart-meters#:~:text=Allows customers to make informed,consumption and reduce their billshttps://www.esource.com/601211hiym/6-dsm-program-types-benefit-ami

- 5. The EM&V Working Group, which consists of consultants with subject matter expertise, collectively determine when freeridership may have a material impact on savings from individual measures or delivery pathways. The TRM is reviewed and updated each year along with the Benefit Cost models. Research and/or other adjustments to freeridership are undertaken when the group determines that current assumptions are not accurately reflecting the impact of freeridership. For example, the EM&V Working Group regularly reviews research on freeridership assumptions in other jurisdictions and may update values in the TRM based on that research. The benefits resulting from energy savings used to calculate the GST and TRCT (and by extension performance incentive) are net of freeridership and are referred to as "net savings".
- 6. No, the NHSaves program savings do not reflect economic impacts to the New Hampshire economy beyond those accounted for in the Benefit Cost ("BC") models. The benefits estimated by the BC models, and included in the primary and secondary cost tests, relate specifically to the net present value of energy (electricity, natural gas, oil, propane and kerosene) savings and certain non-energy impacts (indirect environmental benefits, water savings, and participant impacts) as described in the BC Working Group report and prior discovery and testimony. The impacts of the programs yield direct and indirect economic benefits for consumers and the state.

Whether to explicitly identify and quantify economic impacts in the benefit-cost framework was considered at length by the BC Working Group, as detailed in the New Hampshire Cost-Effectiveness Review report, completed on October 14, 2019, and submitted to the Commission by the Working Group on October 31, 2019, in Docket No. DE 17-136. That report was developed in accordance with the <u>National Standards Practice Manual</u>, (see Attachment RR 1-001A for the 2017 version of the manual in force at the time of the Working Group's report, and Attachment RR 1-001B for the most recent version, published in 2020), which is a publication of the <u>National Efficiency Screening Project</u> (<u>https://www.nationalenergyscreeningproject.org/</u>), which provides a consistent framework for energy efficiency programs across the country to develop jurisdiction-specific cost-effectiveness tests. This framework is centered on a review of priorities reflected in state policy, law, and regulation, and does not consider "incremental energy demand associated with the projected positive impact on the NH economy", nor are the Utilities aware of any

Date Request Received: November 01, 2022 Data Request No. RR 1-001

Date of Response: December 16, 2022 Page 7 of 12

jurisdiction that considers incremental demand for energy resulting from energy efficiency programs. The final report developed for New Hampshire explains the benefit cost framework, summarizes the discussion of the working group members, contains a comprehensive list of all laws and statutes considered by the Working Group, and makes recommendations for changes to the NH cost-effectiveness test based on the results of the review. These recommendations were subsequently summarized and presented to the Commission by the Working Group, and the Commission reviewed and approved the recommendations in Order 26,322. The new test was first applied to program year 2020.

Appendix B of the Cost-Effectiveness Review discusses the Economic Development Impacts of energy efficiency programs beyond those that can be readily quantified and integrated into the cost-effectiveness testing. The authors recommended "that economic development benefits be considered separately from the benefit-cost analysis results. In other words, the economic development benefits should be considered alongside the other energy efficiency benefits but should not be added to them." Because the recommendation was to not include these economic development benefits in the primary or secondary benefit cost-tests (i.e., the Granite State Test, TRCT or UCT), the Utilities did not prioritize evaluation of the economic development benefits associated with the programs.

However, in response to the directive to the Joint Utilities in Commission Order No. 26,621 in Docket DE 20-092, and the interest of providing additional information about the economic impacts of the programs on the New Hampshire economy, the NH Utilities have recently contracted with an external consultant with expertise in this area to model the economic impacts of the programs on New Hampshire. Consistent with the description of how to estimate economic impacts contained in Appendix B of the Cost-Effectiveness Review, the consultant will be utilizing the input-output model IMPLAN.

The current timeline is for this analysis to be completed by March 31, 2023, consistent with the Commission's Order No. 26,642, issued on June 21, 2022, in Docket No. DE 20-092.

7. I. Please see below for the requested information for New Hampshire and the other New England states.

New Hampshire: 3.50% Real Discount Rate (5.50% Nominal Discount Rate, 1.94% Inflation) for 2020, <u>https://www.puc.nh.gov/Regulatory/Docketbk/2017/17-136/LETTERS-MEMOS-TARIFFS/17-136_2019-11-</u> 01_EVERSOURCE_UPDATED_EE_PLAN_REV.PDF; 1.41% Real Discount Rate (3.25%

Date Request Received: November 01, 2022 Data Request No. RR 1-001

Date of Response: December 16, 2022 Page 8 of 12

Nominal Discount Rate, 1.81% Inflation) for 2021, https://www.puc.nh.gov/Regulatory/Docketbk/2020/20-092/INITIAL%20FILING%20-%20PETITION/20-092 2020-09-01 NHUTILITIES EE PLAN.PDF

Massachusetts: 0.46% Real Discount Rate (2.33% Nominal Discount Rate, 1.86% Inflation) for 2019-2021, <u>https://ma-eeac.org/wp-content/uploads/Exhibit-5-2019-2021-BCR-Model-2-19-19-Eversource-Electric.xlsx</u>, <u>https://www.mass.gov/doc/dpu-08-50-a-march-16-2009/download</u> (MA DPU cost-effectiveness and benefits guidelines)

Connecticut: 0.98% Effective Real Discount Rate (3.00% Nominal Discount Rate, 2.00% inflation) for 2010-2021, <u>https://portal.ct.gov/-/media/DEEP/energy/ConserLoadMgmt/Final-2020-Plan-Update-Text-11-1-19.pdf</u>, <u>https://portal.ct.gov/-/media/DEEP/energy/ConserLoadMgmt/FINAL-2021-Plan-Update-Filed-10302020.pdf</u>, <u>https://portal.ct.gov/-/media/DEEP/energy/ConserLoadMgmt/AttachmentBDEEPPolicyMemoRationaleforDiscountRateAppliedinCLMPlans121918pdf.pdf</u> (Rationale for Discount Rate to be Applied in Connecticut's Conservation and Load Management Plans)

According to the Database of State Efficiency Screening Practices (DSESP) (https://www.nationalenergyscreeningproject.org/state-database-dsp/database-of-stateefficiency-screening-practices/), which is a resource of the National Efficiency Screening Project (NESP), Maine uses a real discount rate of 2.8%. This is based on the current yield of 10-year US Treasury securities, plus 200 basis points, adjusted for inflation. The reference policy for the discount rate is the <u>Code of Maine Rules 95-648</u>: Efficiency <u>Maine Trust</u> (chapter 3, section 4) (<u>https://www.maine.gov/sos/cec/rules/90/chaps90-.htm</u>). The DSESP collected the discount rate value from the <u>Efficiency Maine 2020-22 Triennial Plan IV</u> (appendix L) (<u>https://www.efficiencymaine.com/triennial-plan-iv/</u>).

According to DSESP, Rhode Island uses a real discount rate of 0.27%. The nominal rate is 1.84% and inflation is 1.56%. The reference policy for the discount rate is the Rhode Island Public Utilities Commission Least Cost Procurement Standards (2018) (https://rieermc.ri.gov/least-cost-procurement-standards-2018-revisions/), and the DSESP used the value from National Grid's Annual Energy Efficiency Plan for 2018 Settlement of the Parties (http://rieermc.ri.gov/wp-content/uploads/2017/11/4755-ngrid-eepp2018_11-1-17.pdf)

Date Request Received: November 01, 2022 Data Request No. RR 1-001

Date of Response: December 16, 2022 Page 9 of 12

According to DSESP, Vermont uses a real discount rate of 3%. The DSESP used the reference policy and value from the legacy Vermont PUC case <u>Order Re: EEU Avoided</u> <u>Costs, Externality Adjustments, and Other Screening Components for 2017-2018 Time</u> <u>Period</u> (Case No. EEU-2015-04).

II. Please refer to the files provided in response to part I. of this question.

III. The US Office of Management and Budget annually revises the forecasted real and nominal interest rates on treasury notes and bonds in the section on Discount Rates for Cost-Effectiveness Analysis of Federal Programs,

https://www.federalregister.gov/documents/2022/05/24/2022-11085/discount-rates-for-costeffectiveness-analysis-of-federal-programs, published in the Federal Register. For 2022, the 30-year nominal interest rate is 2.6% and the 30-year real interest rate is 0.5%.

The Office of Management and Budget also publishes Circular A-94 annually including discount rates that "are to be used for cost-effectiveness analysis, including lease-purchase analysis, as specified in the revised Circular. These rates do not apply to regulatory analysis." Those rates can be found through the following link for calendar year 2022: https://www.whitehouse.gov/wp-content/uploads/2022/05/Appendix-C.pdf.

The Federal Energy Management Program's (FEMP) 2021 discount rates, which are aligned with the guidance in Circular A-94, can be located at the following link: https://www.energy.gov/sites/default/files/2021-04/2021discountrates.pdf,

The NH Utilities are unaware of exactly how these rates are applied and have no subject matter expertise regarding the development or application of discount rates by the federal government.

8. Please refer to the Utilities response to Reporting Requirement 1.2 from Order No. 26,621, filed on August 1, 2022, in this docket, where the Utilities explained why the New Hampshire EE discount rate is appropriate for use within the NHSaves programs.

The WACC and ROE, either for each individual company or in the aggregate, are not appropriate for use as discount rates to evaluate the cost-effectiveness of energy efficiency programs for several reasons. First, in New Hampshire the ROE is typically heavily negotiated/litigated in the context of a base distribution rate case and therefore the ultimate figure may be the result of substantial compromise of the parties to achieve objectives

Date Request Received: November 01, 2022 Data Request No. RR 1-001

Date of Response: December 16, 2022 Page 10 of 12

specific to the particulars of a given rate case. Second, energy efficiency investments should not have the same discount rates as capital investments because they do not have the same characteristics, risks, or collection period of long-lived distribution assets. Third, energy efficiency investments are driven by public policy, and accordingly, the Granite State Test evaluates benefits in a social policy context. Thus, a (lower) societal discount rate properly reflects this objective. Synapse Energy Economics, Inc. supports this principle in its review of the New Hampshire cost-effectiveness test and recommended that Utilities should continue the practice of using a low-risk discount rate, stating (emphasis added):

The low-risk discount rate gives more weight to long-term impacts, reflects the regulatory perspective in the Granite State Test, and is consistent with the objectives of cost effectiveness analyses. We find this is an appropriate approach to valuing energy efficiency benefits, *and there is no rationale or new policy for utilities to alter their current practices*. This discount rate should be applied to the Granite State Test as well as any secondary tests.

Synapse Energy Economics, Inc., New Hampshire Cost-Effectiveness Review at 44 (Oct. 14, 2019), filed on October 31, 2019, in Docket No. DE 17-136 and also available at https://www.puc.nh.gov/EESE%20Board/EERS Working Groups.html.

- 9. The calculation of a specific discount rate has been developed on a plan-specific basis for each plan since the inception of the energy efficiency programs. The formula for identifying a discount rate for use in cost-effectiveness testing was reviewed as part of the Benefit Cost Working Group (see response to Question RR 1-001 part 6). The report recommended that the "Utilities should continue the current practice of using a low-risk discount rate," which the Utilities consider to be an essential component of the Granite State Test and secondary test, and which was consistently codified in HB 549, which directs the NH Utilities and the Commission to "use the Granite State test as the primary test..." Synapse Energy Economics, Inc., New Hampshire Cost-Effectiveness Review at 10 (Oct. 14, 2019), filed on October 31, 2019 in Docket No. DE 17-136; and also available at https://www.puc.nh.gov/EESE%20Board/EERS_Working_Groups.html; N.H. Rev. Stat. § 374-F:3, VI.
- 10. Per the methodology outlined in the Lookups tab of the Benefit-Cost Models, the Nominal Discount Rate, Inflation Rate, and calculated Real Discount Rate are determined as follows.

Nominal Discount Rate: Based on the prime rate (generally regarded as the federal funds rate + 3) in effect as of the most recent June preceding the Plan filing. This is done in accordance

Date Request Received: November 01, 2022 Data Request No. RR 1-001

Date of Response: December 16, 2022 Page 11 of 12

with the Final Energy Efficiency Group Report, dated July 6, 1999, in DR 96-150. Since the rates for mortgages, small business loans, and personal loans are based on the prime rate, most banks and financial institutions post this rate on their websites. Current and historical prime rates can be found here: <u>https://www.jpmorganchase.com/about/our-business/historical-prime-rate</u>

<u>Inflation Rate</u>: Based on the inflation rate from the year prior Q1 and most recent Q1 readings preceding the Plan filing. The calculated annual rate uses the GDP: Implicit Price Deflator from the U.S. Bureau of Economic Analysis, which can be found here: <u>https://fred.stlouisfed.org/data/GDPDEF.txt</u>

<u>Real Discount Rate</u>: This value is calculated and follows the methodology outlined in the AESC 2021 study:

Real Discount Rate = [(1 + Nominal Discount Rate) / (1 + Inflation Rate)] - 1

To calculate the real discount rate as of today, using the current methodology and timelines for accessing the rates, the Utilities would use the prime rate in effect as of June 1, 2022 (4.00%) as the nominal discount rate, and the inflation rate between Q1 2021 and Q1 2022 (6.94%). The calculated real discount rate would be: -2.75% = [(1 + 0.0400) / (1 + 0.0694)] - 1

Alternatively, to calculate the real discount rate as of today, using the current methodology, but updating the timelines to employ the most recent available rates, the Utilities would use the prime rate in effect as of November 30, 2022 (7.00%) as the nominal discount rate, and the inflation rate between Q3 2021 and Q3 2022 (7.07%). The calculated real discount rate would be:

-0.07% = [(1 + 0.0700) / (1 + 0.0707)] - 1

11.11/30/2022 Response:The Utilities were granted an extension for the response to this question.

12/16/2022 Response:

Date Request Received: November 01, 2022 Data Request No. RR 1-001

Date of Response: December 16, 2022 Page 12 of 12

(i) and (ii). For the reasons discussed in the Joint Utilities' response to RR 1-001 part 8, it is not appropriate to use each individual utility's WACC or ROE as the discount rate in the Granite State Test to evaluate the cost-effectiveness of energy efficiency investments.¹

Despite the inappropriateness of using WACC or ROE as the discount rate (either for each individual company or in the aggregate), in the interest of being responsive to this request, the Joint Utilities have provided (for illustrative purposes only) computations of the cost-effectiveness ratios for each individual utility using the ROE and WACC specific to each utility in Excel Attachment RR 1-001C. However, the computations provided by the Joint Utilities should not be construed as adopting or otherwise endorsing this approach for the reasons stated in this response and the response to RR 1-001 part 8.

(iii) In the interest of being responsive to this request, the Joint Utilities have provided (for illustrative purposes only) computations of the cost-effectiveness ratios for each utility using the discount rates set forth in this request. However, the computations provided by the Utilities should not be construed as adopting or otherwise endorsing this approach.

Altering the discount rate methodology, as suggested by this record request, would limit the measures that could be offered cost-effectively within the NHSaves programs. Altering the discount rate from a "low discount rate," as developed and approved through the BC Working Group's efforts establishing the Granite State Test (GST), would also arguably result in non-conformance with HB 549, which enshrined the GST as the primary cost-effectiveness test.

¹ Moreover, the directive in this Information Request to average the WACCs and ROEs of the different companies to derive a combined, average discount rate is flawed for several reasons. First, ROE and WACC should reflect the risks faced by the specific company and the opportunity to earn a fair and reasonable return consistent with other businesses of similar risk. Second, the ROE and WACCs for each Company are set in the context of base distribution rate cases, which can be several years apart from each other. Thus, the average, whether a simple average or weighted average of the Joint Utilities ROE or WACC is not necessarily indicative of current market/economic conditions. Third, each company has its own risk profile based on characteristics and circumstances specific to each company, including company size, business risks, peer authorized returns, and credit ratings. Thus, blending the ROE or the WACC of disparate entities is not analytically sound.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001A Page 1 of 146

National Standard Practice Manual

for Assessing Cost-Effectiveness of Energy Efficiency Resources

EDITION 1 Spring 2017





NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001A Page 2 of 146

National Standard Practice Manual for Assessing Cost-Effectiveness of Energy Efficiency Resources

May 18, 2017

Prepared by The National Efficiency Screening Project



NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001A Page 3 of 146

Copyright 2017, National Efficiency Screening Project (NESP). All rights reserved.

The National Efficiency Screening Project

The National Efficiency Screening Project (NESP) is a group of organizations and individuals working to update and improve the way that utility customer-funded electricity and natural gas energy efficiency resources are assessed for cost-effectiveness and compared to other resource investments. This National Standard Practice Manual (NSPM) is a publication of the NESP.

The NSPM builds on the 2014 NESP publication <u>The Resource Value Framework –</u> <u>Reforming Energy Efficiency Cost Effectiveness Screening</u>, a foundational document that presented a first version of the Resource Value Framework. The NESP and 2014 publication, through the emergence of this NSPM, was managed and supported by the Home Performance Coalition, and is currently coordinated by E4TheFuture. We acknowledge the generous funding support over the years that made this project and report possible: the MacArthur Foundation, the United States Department of Energy, and E4TheFuture.

The NSPM, and related materials from the NESP, are available at: <u>https://nationalefficiencyscreening.org</u>

Report Authors

The NSPM was prepared by Tim Woolf (Synapse Energy Economics), Chris Neme (Energy Futures Group), Marty Kushler (American Council for an Energy-Efficient Economy), Steven R. Schiller (Schiller Associates), and Tom Eckman (Consultant, formerly with Northwest Power & Conservation Council). Coordination of final document: Julie Michals (E4TheFuture).

Review Committee

The project team would like to thank the following individuals for offering their insights and perspectives on this report and/or participating in Review Committee webinars. The individuals and their affiliations are listed for identification purposes only. Participation on the Review Committee does not indicate support for this document in whole or in part.

Roger Baker, Commonwealth Edison Will Baker, Midwest Energy Efficiency Alliance Eric Belliveau, Optimal Energy Carmen Best, California Public Utilities Commission Rob Bevill, South-central Partnership for Energy Efficiency as a Resource Michael Brandt, Commonwealth Edison Todd Bianco, Rhode Island Public Utility Commission Joe Bryson, US Environmental Protection Agency Brian Buckley, Northeast Energy Efficiency Partnership Mohit Chhabra, Natural Resources Defense Council Greg Dierkers, US Department of Energy Niko Dietsch, Environmental Protection Agency Philippe Dunsky, Dunsky Energy Consulting David Farnsworth, Regulatory Assistance Project Julia Friedman, Midwest Energy Efficiency Alliance Don Gilligan, National Association of Energy Service Companies Harrison Grubbs, CLEAResult

Ely Jacobsohn, US Department of Energy Elliott Jacobson, Low-Income Energy Affordability Network Val Jenson, Commonwealth Edison Miles Keogh, National Association of Regulated Utility Commissioners Sami Khawaja, The Cadmus Group Benjamin King, US Department of Energy Jack Laverty Columbia Gas Dan Lauf, National Association of Regulated Utility Commissioners Robin LeBaron, Home Performance Coalition Doug Lewin, CLEAResult Carl Linville, Regulatory Assistance Project Craig Massey, Consortium for Energy Efficiency Janine Migden-Ostrander, Regulatory Assistance Project Deborah Miller, Home Performance Coalition Joy Morgenstern, California Public Utility Commission

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001A Page 5 of 146

Jeremy Newberger, National Grid

Luke Nickerman, Pacific Gas & Electric

Jerry Oppenheim, Low-Income Energy Affordability Network

Sonny Popowsky, Former PA Consumer Advocate Deborah Reynolds, Washington Utilities and

Transportation Commission

Kara Saul-Rinaldi, Home Performance Coalition

Rich Sedano, Regulatory Assistance Project

- Dick Spellman, GDS Associates
- Tom Stanton, National Regulatory Research Institute

Frank Stern, Navigant Susan Stratton, Northwest Energy Efficiency Alliance Elizabeth Titus, Northeast Energy Efficiency Partnerships Michelle Vigen, California Efficiency and Demand Management Council Dan Violette, Navigant Liz Weiner, CLEAResult Carol White, National Grid Johanna Zetterberg, US Department of Energy

Abstract

This National Standard Practice Manual (NSPM) is intended to provide a comprehensive framework for assessing the cost-effectiveness of energy efficiency resources. The manual is directly applicable to all types of electric and gas utilities and jurisdictions where energy efficiency resources are funded by and implemented on behalf of electric or gas utility customers. The intended audience are those involved in assessing the cost-effectiveness of energy resources, including regulators, utilities, program administrators, energy resource planners, consumer advocates, and other stakeholders.

The NSPM provides guidance that incorporates lessons learned over the past 20 years, responds to current needs, and addresses and takes into account the relevant policies and goals of each jurisdiction undertaking efficiency investments.

The NSPM presents an objective and neutral Resource Value Framework that can be used to define a jurisdiction's *primary* cost-effectiveness test, which is referred to as a Resource Value Test. The Resource Value Framework is based on six principles that encompass the perspective of a jurisdiction's applicable policy objectives, and it includes and assigns value to all relevant impacts (costs and benefits) related to those objectives.

The NSPM also provides information, guidance, and templates that support the selection of components of a jurisdiction's Resource Value Test (e.g., the range of costs and benefits to consider and appropriate discount rates), the application of such tests (e.g., defining of analysis periods), and the documentation of the relevant policies as well as quantification of relevant costs and benefits. The NSPM also addresses the use of secondary tests in addition to a primary Resource Value Test.

Contents

Ab	stract.	
Ex	ecutive	Summaryvi
	ES.1	Universal Principlesvi
	ES.2	Resource Value Frameworkvii
	ES.3	Resource Value Testiv
	ES.4	Applicability to Other Types of Resourcesxii
	ES.5	Foundational Information Covered in the NSPMxii
IN'	TRODU	CTION: Purpose, Scope and Format1
PA	RT I.	DEVELOPING COST-EFFECTIVENESS TESTS USING THE RESOURCE VALUE FRAMEWORK8
1.	Princip	les of Cost-Effectiveness Analyses
2.	The Re	source Value Framework and Primary Test15
	2.1	Summary of Key Points15
	2.2	The Resource Value Framework15
	2.3	The Resource Value Test as the Primary Test16
	2.4	The RVT as a Dynamic Test
3.	Develo	ping the Resource Value Test1٤
	3.1	STEP 1: Identify and Articulate Applicable Policy Goals
	3.2	STEP 2: Include Utility System Costs and Benefits
	3.3	STEP 3: Decide Which Non-Utility Costs and Benefits to Include23
	3.4	STEP 4: Ensure the Test Is Symmetrical
	3.5	STEP 5: Ensure the Analysis Is Forward-Looking and Incremental
	3.6	STEP 6: Develop Methodologies to Account for All Relevant Impacts
	3.7	STEP 7: Ensure Transparency
4.	Relatio	nship to Traditional Tests
	4.1	Summary of Key Points
	4.2	Resource Value Test Examples
	4.3	Conceptual Differences between the RVT and Traditional Tests41
5.	Second	lary Cost-Effectiveness Tests43
	5.1	Summary of Key Points43
	5.2	Potential Reasons for Analysis with Multiple Tests43
	5.3	Secondary Tests to Consider
PA	RT II.	DEVELOPING INPUTS FOR COST-EFFECTIVENESS TESTS47
6.	Energy	Efficiency Costs and Benefits48
	6.1	Summary of Efficiency Resource Impacts48

		NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042		
		11-01-2022 IR F Attachment RE	Requests	
	<u> </u>	Page	8 of 146	
	6.2		50	
	6.3	Non-Utility System Impacts	54	
7.	Method	dologies to Account for Relevant Impacts	59	
	7.1	Summary of Key Points	59	
	7.2	Jurisdiction-Specific Studies	59	
	7.3	Studies from Other Jurisdictions	60	
	7.4	Proxies	60	
	7.5	Quantitative and Qualitative Information	62	
	7.6	Alternative Thresholds	64	
	7.7	Sensitivity Analyses	64	
	7.8	Reliability of Data	65	
8.	Particip	pant Impacts	66	
	8.1	Summary of Key Points	66	
	8.2	Policy and Symmetry	66	
	8.3	Account for the Impacts on All Customers Combined	67	
	8.4	Account for the Total Cost of the Resource	67	
	8.5	Protect Program Participants	69	
	8.6	Account for Low-Income Program Participant Benefits	70	
	8.7	Account for Other Fuel and Water Impacts	70	
	8.8	Quality of the Information	71	
9.	Discou	nt Rates	72	
	9.1	Summary of Key Points	72	
	9.2	The Purpose of Discount Rates	73	
	9.3	Commonly Used Discount Rates	75	
	9.4	The Regulatory Perspective	76	
	9.5	The Investor-Owned Utility Perspective	77	
	9.6	The Publicly Owned Utility Perspective	79	
	9.7	The Utility Customer Perspective	79	
	9.8	Risk Considerations	80	
	9.9	Determining the Discount Rate	81	
10.	Asse	ssment Level	85	

9.8	Risk Considerations	80
9.9	Determining the Discount Rate	81
10. Ass	essment Level	85
10.1	Summary of Key Points	85
10.2	Assessment Level Options	85
10.3	Properly Accounting for Fixed and Variable Costs	88
11. Ana	lysis Period and End Effects	91
11.1	Summary of Key Points	91
11.2	Analysis Period	91
11.3	End-Effects Problems	92
11.4	Remedies for End-Effects Problems	

12. Analy	vsis of Early Replacement	94		
12.1	Summary of Key Points	94		
12.2	Overview	94		
12.3	Accounting for Changes in the Timing of Costs	95		
12.4	Accounting for Multiple Baselines for Both Costs and Savings	96		
13. Free-	Riders and Spillover	99		
13.1	Summary of Key Points			
13.2	Applicability and Definitions	100		
13.3	Economic Treatment of Free-Rider Impacts	100		
13.4	Economic Treatment of Spillover Effects	102		
Reference	es	104		
Appendix	A. Traditional Cost-Effectiveness Tests	109		
A.1 A.2 A.3 A.4 A.5 A.6	Overview Utility Cost Test Total Resource Cost Test Societal Cost Test Participant Cost Test Rate Impact Measure Test			
Appendix	B. Costs and Benefits of Other Types of DERs	116		
Appendix	C. Accounting for Rate and Bill Impacts	121		
C.1 C.2 C.3 C.4 C.5	Multiple Factors Affecting Rate Impacts Limitations of the Rate Impact Measure Test Rate Impacts and Customer Equity A Better Approach for Analyzing Rate Impacts Relationship to the Cost-Effectiveness Analysis			
Appendix	Appendix D. Glossary of Terms127			

Table of Tables

Table 1. Overview of the National Standard Practice Manual	6
Table 2. Examples of Energy-Related and Other Applicable Policy Goals	11
Table 3. Example Summary of a Jurisdiction's Applicable Policy Goals	20
Table 4. Example Electric Utility System Impacts to Include in Cost-Effectiveness Tests	22
Table 5. Example Gas Utility System Impacts to Include in Cost-Effectiveness Tests	23
Table 6. Examples of Commonly Considered Non-Utility Impacts	24
Table 7. Program Participant Costs and Benefits	25
Table 8. Points and Counter-Points Regarding Whether to Include Participant Impacts	26
Table 9. Non-Utility Low-Income Costs and Benefits	27
Table 10. Examples of Other Fuel Impacts in Efficiency Programs	28
Table 11. Examples of Environmental Impacts of EE Resources	29
Table 12. Different Approaches to Account for All Relevant Impacts	34
Table 13. Efficiency Cost-Effectiveness Reporting Template	35
Table 14. Mix of Policy Objectives Leading to Different Jurisdictional RVTs	39
Table 15. Comparing the RVT and the Traditional Tests	41
Table 16. Using Secondary Test to Address Program Selection or Design Questions	46
Table 17. Summary of Efficiency Resource Impacts	49
Table 17. Summary of Efficiency Resource Impacts Table 18. Participant Non-Resource Benefits	49 55
Table 17. Summary of Efficiency Resource Impacts Table 18. Participant Non-Resource Benefits Table 19. Discount Rate Options for Cost-Effectiveness Analyses	49 55 75
Table 17. Summary of Efficiency Resource Impacts Table 18. Participant Non-Resource Benefits Table 19. Discount Rate Options for Cost-Effectiveness Analyses Table 20. Proper Analysis with 25 Percent Fixed Portfolio Costs Included at Portfolio-Level Analysis	49 55 75 89
Table 17. Summary of Efficiency Resource Impacts Table 18. Participant Non-Resource Benefits Table 19. Discount Rate Options for Cost-Effectiveness Analyses Table 20. Proper Analysis with 25 Percent Fixed Portfolio Costs Included at Portfolio-Level Analysis Table 21. Improper Analysis with 25 Percent Fixed Portfolio Costs Allocated to Individual Programs.	49 55 75 89 90
Table 17. Summary of Efficiency Resource Impacts	49 55 75 89 90 92
Table 17. Summary of Efficiency Resource Impacts	49 55 75 89 90 92 93
 Table 17. Summary of Efficiency Resource Impacts Table 18. Participant Non-Resource Benefits	49 55 75 89 90 92 93 96
 Table 17. Summary of Efficiency Resource Impacts	49 55 75 89 90 92 93 96 98
 Table 17. Summary of Efficiency Resource Impacts	49 55 75 90 92 93 96 98 99
 Table 17. Summary of Efficiency Resource Impacts	49 55 75 90 90 93 93 96 98 99 99 102
 Table 17. Summary of Efficiency Resource Impacts	49 55 75 89 90 90 93 93 96 98 99 93 91

Table 31. Summary of the CaSPM Cost-effectiveness Tests	115
Table 32. Relevant Costs and Benefits of Distributed Energy Resources	117
Table 33. Utility System Costs and Benefits of DERs	118
Table 34. Non-Utility System Costs and Benefits of DERs	120

Figures

Figure 1. The Foundation to Developing a Jurisdiction's Primary Test	16
Figure 2. The Resource Value Framework Steps	18
Figure 3. The Implications of Participant Benefits on Residential Efficiency Programs	32
Figure 4. Mix of Policy Objectives that Lead to a Jurisdictional RVT Identical to a Traditional Test	40
Figure 5. Implications of Discount Rates (annual present value dollars)	74
Figure 6. Implications of Discount Rates (cumulative present value dollars)	74

Executive Summary

Assessing the cost-effectiveness of energy resources such as efficiency involves comparing the costs and benefits of such resources with other resources that meet energy and other applicable objectives. Historically, energy efficiency (EE) has been assessed through integrated resource planning processes or via standard tests defined in the California Standard Practice Manual (CaSPM). These assessments entail comparing the cost of EE resources to forecasts of avoided supply-side resources and other relevant costs and benefits. This National Standard Practice Manual (NSPM) builds and expands upon the decades old CaSPM, providing current experience and best practices with the following additions:

- Guidance on how to develop a jurisdiction's primary cost-effectiveness test that meets the applicable policy goals of the jurisdiction.¹ The guidance also addresses the difficulties jurisdictions have had in consistently implementing concepts presented in the CaSPM.
- Information on the inputs and considerations associated with selecting the appropriate costs and benefits to include in a cost-

The NSPM presents:

- Universal Principles for developing and applying costeffectiveness assessments.
- A step-by-step Resource Value Framework for jurisdictions to use to develop their primary cost-effectiveness test: the Resource Value Test (RVT), which addresses all of the traditional components of cost-effectiveness testing – but with explicit consideration of the specific policy framework for the particular jurisdiction.
- Neutral, objective guidance and foundational information for selecting and quantifying the components of a jurisdiction's test(s), and for applying and documenting the policies and data that were used to define the test, building on lessons learned over the past 20 years and responding to current needs.

effectiveness test and accounting for applicable hard-to-monetize costs and benefits, with guidance on a wide range of fundamental aspects of cost-effectiveness analyses.

The NSPM is relevant to all types of electric and gas utilities, including: investor-owned utilities, publicly owned utilities, federal power authorities, and cooperatives, as well as to any jurisdiction where EE resources are funded and implemented on behalf of electric or gas utility customers.

While this NSPM focuses on the assessment of utility EE resources, the core concepts—including the principles described in Chapter 1 and the Resource Value Framework ('the Framework') described in Chapter 2—can generally be used to assess the cost-effectiveness of supply-side resources or distributed energy resources (DERs).

ES.1 Universal Principles

A unique attribute of the NSPM, and embedded in the Resource Value Framework, is a set of universal principles to follow when developing an RVT for any particular jurisdiction. These principles, provided in Table ES-1, represent sound economic and

¹ The NSPM uses the term "jurisdiction" broadly to encompass states, provinces, federal power authorities, municipalities, cooperatives, etc.

regulatory practices, and are consistent with the input received from a broad range of stakeholders during the development of this manual.

Efficiency as a Resource	EE is one of many resources that can be deployed to meet customers' needs, and therefore should be compared with other energy resources (both supply-side and demand-side) in a consistent and comprehensive manner.
Policy Goals	A jurisdiction's primary cost-effectiveness test should account for its energy and other applicable policy goals and objectives. These goals and objectives may be articulated in legislation, commission orders, regulations, advisory board decisions, guidelines, etc., and are often dynamic and evolving.
Hard-to-Quantify Impacts	Cost-effectiveness practices should account for all relevant, substantive impacts (as identified based on policy goals,) even those that are difficult to quantify and monetize. Using best-available information, proxies, alternative thresholds, or qualitative considerations to approximate hard-to-monetize impacts is preferable to assuming those costs and benefits do not exist or have no value.
Symmetry	Cost-effectiveness practices should be symmetrical, where both costs and benefits are included for each relevant type of impact.
Forward-Looking Analysis	Analysis of the impacts of resource investments should be forward- looking, capturing the difference between costs and benefits that would occur over the life of the subject resources as compared to the costs and benefits that would occur absent the resource investments.
Transparency	Cost-effectiveness practices should be completely transparent, and should fully document all relevant inputs, assumptions, methodologies, and results.

Table ES-1. Universal Principles

ES.2 Resource Value Framework

The Resource Value Framework is used to construct a jurisdiction's primary costeffectiveness test, the RVT, using a series of seven steps that define the framework. In some cases, the steps align directly with one of the universal principles.

The Framework encompasses the perspective of a jurisdiction's applicable policy objectives, and it includes and assigns value to all relevant impacts (costs and benefits) related to those objectives. The NSPM refers to this as the 'regulatory' perspective, which is intended to reflect the important responsibilities of institutions, agents, or other decision-makers authorized to determine utility resource cost-effectiveness and funding priorities. This perspective flows from the notion that determining whether a resource has benefits that exceed its costs requires clarity about the purpose of the resource investment decision.

The NSPM further provides information, templates,

Regulators/decision-makers

refers to institutions, agents, or other decision-makers that are authorized to determine utility resource cost-effectiveness and funding priorities. Such institutions or agents include public utility commissions, legislatures, boards of publicly owned utilities, the governing bodies for municipal utilities and cooperative utilities, municipal aggregator governing boards, and more.

and examples that can support a jurisdiction in applying the universal principles, and also in constructing appropriate tests in a structured, logical, and documented manner

that meets the specific interests and needs (as defined by policies) of the jurisdiction. The seven steps of the Framework are summarized in Figure ES-1 below.

Figure ES-1. Resource Value Framework Steps

STEP 1	Identify and articulate the jurisdiction's applicable policy goals.
STEP 2	Include all the utility system costs and benefits.
STEP 3	Decide which non-utility impacts to include in the test, based on applicable policy goals.
STEP 4	Ensure that the test is symmetrical in considering both costs and benefits.
STEP 5	Ensure the analysis is forward looking and incremental.
STEP 6	Develop methodologies to account for all relevant impacts, including hard to quantify impacts.
STEP 7	Ensure transparency in presenting the inputs and results of the cost-effectiveness test.

ES.3 Resource Value Test

The RVT is the primary cost-effectiveness test designed to represent a regulatory perspective, which reflects the objective of providing customers with safe, reliable, low-cost energy services, while meeting a jurisdiction's other applicable policy goals and objectives. As described in detail within the NSPM, each jurisdiction can develop its own RVT using the Resource Value Framework.

The RVT focus on the regulatory perspective differs from the three most common CaSPM traditional tests—the Utility Cost Test (UCT), Total Resource Cost (TRC) test and Societal Cost Test (SCT). These tests provide the perspective of the utility, the utility and participants, and society as a whole, respectively.

The RVT and Secondary Tests

The RVT serves as a primary test which assesses cost-effectiveness of efficiency resources relative to a jurisdiction's applicable policy goals that are under the purview of the jurisdiction's regulators or other decision-makers. However, there can be value in assessing cost-effectiveness of efficiency resources from perspectives represented by other tests. Among the potential purposes of using additional tests are:

- To inform decisions regarding how much utility customer money could or should be invested to acquire cost-effective savings;
- To inform decisions regarding which efficiency programs to prioritize if not all cost-effective resources will be acquired;
- To inform efficiency program design; and/or
- To inform public debate regarding efficiency resource acquisition.

Depending on a jurisdiction's energy and other applicable policy goals, the resulting RVT may or may not be different from the traditional cost-effectiveness tests. Put another way, it is possible for a jurisdiction's applicable policy goals to align with one of the traditional CaSPM tests, in which case its RVT will be identical to one of those tests. However, it is also possible—and indeed likely in many cases—that a jurisdiction's energy and other policy goals will not align well with goals implicit in any of the traditional tests. In such cases, the RVT will be different than all the traditional tests.

Furthermore, each jurisdiction's RVT can be unique, where the categories of impacts included in the RVT can vary across jurisdictions and/or over time. This is because the impacts are based on each jurisdiction's policy concerns, which can and do vary. *In contrast, the traditional UCT, TRC, and SCT tests are conceptually static; they do not change geographically or over time if applied in their purest conceptual form.* Table ES-2 compares the RVT with the CaSPM tests.

Test	Perspective	Key Question Answered	Categories of Costs and Benefits Included
Utility Cost Test	The utility system	Will utility system costs be reduced?	Includes the costs and benefits experienced by the utility system
Total Resource Cost Test	The utility system plus participating customers	Will utility system costs plus program participants' costs be reduced?	Includes the costs and benefits experienced by the utility system, plus costs and benefits to program participants
Societal Cost	Society as a whole	Will total costs to society be reduced?	Includes the costs and benefits experienced by society as a whole
Resource Value Test	Regulator/decisio n makers	Will utility system costs be reduced, while achieving applicable policy goals?	Includes the utility system costs and benefits, plus those costs and benefits associated with achieving relevant applicable policy goals

Table ES-2. Comparison of RVT with the Traditional CaSPM Tests

In those cases where a jurisdiction's policy goals align with one of the other tests, the RVT will be the same as that other test. This is discussed in Chapter 4.

Figure ES-1 compares the traditional cost-effectiveness tests to one that is developed using the Resource Value Framework. The gold circle in the center represents the utility system impacts, which should be included in any cost-effectiveness test. The sections around the circles represent non-utility system impacts that jurisdictions can choose to include in their primary test. Three of the circles indicate the impacts that would be included using the traditional cost-effectiveness tests. The fourth circle indicates a different set of impacts that would be included by a jurisdiction whose policies suggest accounting for other fuel impacts, low-income impacts, public health impacts, jobs and economic development, and energy security.

Figure ES-1. Examples of Primary Tests that Jurisdictions Could Develop Using the Resource Value Framework



To support the core principle to transparently document cost-effectiveness practices, this NSPM presents an RVT template, shown in Table ES-3, to assist jurisdictions in documenting assumptions and results of their analysis. More detail with examples is provided in Part I of the NSPM.

Program/Sector/Portfolio Name:		Date:	
A. Monetized Utility System Costs		B. Monetized Utility System Benefits	
Measure Costs (utility portion)		Avoided Energy Costs	
Other Financial or Technical Support Costs		Avoided Generating Capacity Costs	
Program Administration Costs		Avoided T&D Capacity Costs	
Evaluation, Measurement, & Verification		Avoided T&D Line Losses	
Shareholder Incentive Costs		Energy Price Suppression Effects	
		Avoided Costs of Complying with RPS	
		Avoided Environmental Compliance Costs	
		Avoided Bad Debt, Arrearages, etc.	
		Reduced Risk	
Sub-Total Utility System Costs		Sub-Total Utility System Benefits	
C. Monetized Non-Utility Costs		D. Monetized Non-Utility Benefits	
Participant Costs		Participant Benefits	
Low-Income Customer Costs		Low-Income Customer Benefits	
Other Fuel Costs	Include to the	Other Fuel Benefits	Include to the
Water and Other Resource Costs	extent these	Water and Other Resource Benefits	extent these
Environmental Costs	impacts are	Environmental Benefits	impacts are
Public Health Costs	RVT.	Public Health Benefits	RVT.
Economic Development and Job Costs		Economic Development and Job Benefits	
Energy Security Costs		Energy Security Benefits	
Sub-Total Non-Utility Costs		Sub-Total Non-Utility Benefits	
E. Total Monetized Costs and Benefit	s		
Total Costs (PV\$)		Total Benefits (PV\$)	
Benefit-Cost Ratio		Net Benefits (PV\$)	
F. Non-Monetized Considerations			
Economic Development and Job Impacts	Quantitative information, and discussion of how considered		
Market Transformation Impacts Qualitative cor		nsiderations, and discussion of how considered	
Other Non-Monetized Impacts	Quantitative information, qualitative considerations, and how considered		
Determination:	Do Efficiency	Resource Benefits Exceed Costs? [Y	es / No]

Table ES-3: Efficiency Cost-Effectiveness Reporting Template

ES.4 Applicability to Other Types of Resources

While this NSPM focuses on the assessment of EE resources, the core concepts can be applied to other types of resources as well. The cost-effectiveness principles described in Chapter 1, and the Resource Value Framework described in Chapter 2, can be used to assess the cost-effectiveness of supply-side resources or distributed energy resources (DERs)—including EE, demand response, distributed generation, distributed storage, electric vehicles, and strategic electrification technologies.

With regard to supply-side resources, the cost-effectiveness principles can be used in the context of integrated resource planning or when conducting any sort of economic analyses of specific generation, transmission, or distribution infrastructure investments. The Resource Value Framework can be used to identify the primary test for assessing these supply-side investments, or to identify the criteria that would be used to select the preferred resource plan in the context of an IRP. This approach would not only ensure sound practices for analyzing supply-side resources, it would also ensure that EE resources are analyzed comparably and consistently with supply-side resources.

With regard to DERs, the cost-effectiveness principles and the Resource Value Framework can be used as the foundation for assessing their cost-effectiveness. There are, however, ways in which other types of DERs might need to be treated differently from EE resources. These important DER-specific issues are beyond the scope of this NSPM, but should be addressed by each jurisdiction as they develop cost-effectiveness practices for DERs.

ES.5 Foundational Information Covered in the NSPM

Supporting the implementation of the Resource Value Framework for developing an RVT requires understanding of a wide range of costeffectiveness related topics. These include identifying, quantifying, and documenting relevant policies, costs, and benefits—in addition to the analysis of related foundational considerations of costeffectiveness tests. Thus, the NSPM not only presents the universal principles, the Framework, and associated RVT concepts and examples, but also provides information

Questions the RVT Does and Does Not Answer

The primary RVT can be used to answer the fundamental question of *which resources have benefits that exceed their costs*, where the benefits and costs are defined by the applicable policy goals of a jurisdiction and developed via Framework 7-step process. With this Framework, the resource investment decision question is addressed in a comprehensive and transparently documented manner.

Regulators and decision-makers typically need to answer a second critical question: *how much utility customer funding should be spent on EE resources?* The primary cost-effectiveness test is necessary but may not be sufficient for answering this second question, which requires consideration of jurisdictionspecific factors through a process such as integrated resource planning or rate proceedings.

on related foundational topics that can be particularly valuable to those responsible for developing the RVT and its inputs. The NSPM can also be helpful for those seeking to understand the range of options and outcomes that can result from different RVTs.

The foundational topics covered in the NSPM, found in Parts I, II, or in the appendices, are as follows:

• Ensuring transparency of the assumptions, analysis and results (Chapter 3)

- Use of primary vs secondary cost-effectiveness tests (Chapter 5)
- Identifying relevant impacts (costs and benefits) to include in a Resource Value Test (Chapter 6)
- Methods that can be used to determine or account for all relevant impacts (Chapter 7)
- Considerations for including Participant Impacts (Chapter 8)
- Identifying appropriate discount rates (Chapter 9)
- Selecting an assessment level (Chapter 10)
- Selection of an analysis period (Chapter 11)
- Treatment of Early Replacement (Chapter 12)
- Treatment of Free Riders and Spillover (Chapter 13)
- Traditional Cost-Effectiveness Tests (Appendix A)
- DER Costs and Benefits (Appendix B)
- Accounting for Rate and Bill Impacts (Appendix C)

INTRODUCTION: Purpose, Scope and Format

Purpose

The purpose of this National Standard Practice Manual (NSPM) is to help guide the development of a cost-effectiveness test for regulators, utilities, program administrators, efficiency planners, consumer advocates, and other efficiency stakeholders. In its

simplest form, assessing the costeffectiveness of energy resources involves comparing the costs and benefits of such resources with other resources. The manual describes the principles, concepts, and methodologies for sound, comprehensive, balanced assessment of the costeffectiveness of EE resources, and can help involved parties identify the full range of

This manual is intended to serve as an objective, neutral guidance document that does not prescribe any one type of cost-effectiveness test *per se*.

efficiency resources whose benefits exceed their costs. Utility resource decision-makers can then use this information to decide which resources to acquire to meet their specific EE objectives, standards, or targets.

This manual is intended to serve as an objective, neutral guidance document that does not prescribe any one type of cost-effectiveness test *per se*. Rather it sets forth a framework that includes key principles and steps to use within a jurisdiction to develop a primary cost-effectiveness test, and also to inform use of secondary tests.

The goal of this manual is to provide guidance that: (1) builds from the lessons learned over the past decades, (2) responds to current needs, (3) addresses the specific goals of each jurisdiction, and (4) can eventually be fully expanded to address all types of distributed energy resources (DER).

Why the Need for this NSPM?

Since the 1980s, the prevailing cost-effectiveness guidance document for EE resources has been the *California Standard Practice Manual* (CaSPM), which sets forth several 'traditional tests' commonly referred to as the Utility Cost Test (UCT), the Total Resource Cost (TRC) test, and the Societal Cost Test (SCT).² Last updated in 2002, the CaSPM presents important limitations with which jurisdictions have increasingly struggled over the years. This has led to the inconsistent application of the traditional tests. These limitations are generally characterized as follows:

a) The CaSPM does not provide guidance on how to develop a cost-effectiveness *framework*, and associated primary test, that reflects a jurisdiction's energy and

² See Appendix A for a summary of the Traditional Tests. The CA SPM's chapters are organized around 4-5 tests: the Participant Test; the RIM test; the TRC test; the SCT (characterized as a variant of the TRC); and the Program Administrator Costs test, also referred to as the Utility Cost Test (UCT). This manual focuses on the most commonly used cost-effectiveness tests in practice today: the TRC test, UCT, and SCT.

other applicable policy goals. Such goals should be directly relevant to identifying the range of costs and benefits to include in a jurisdiction's cost-effectiveness analyses.

- b) The three commonly used traditional tests (UCT, TRC, and SCT) are typically defined as having a specific set of costs and benefits depending on the perspective of either the utility, the utility and program participants, or society as a whole.³ A jurisdiction's energy policies, however, seldom align precisely with any one of these types of perspectives. Moreover, these three tests do not account for a critical perspective: the perspective of reducing total utility costs to customers (relative to other resources) while also explicitly taking into account the jurisdiction's applicable policy goals. That broader perspective is intended to be reflective of the important responsibilities of a utility regulator. Hence the NSPM introduces this concept as the *regulatory perspective*.
- c) Jurisdictions have struggled with ongoing debates about what costs and benefits should be included in their analyses, and whether and/or how to account for certain impacts. This is especially the case for hard-to-quantify non-energy impacts. These issues have been particularly challenging for the TRC test, the predominantly used screening test. Research has shown that most jurisdictions that use the TRC test treat costs and benefits asymmetrically by accounting for participant costs but not benefits (ACEEE 2012). The CaSPM lacks key principles and guidance that can help jurisdictions determine which impacts to consider. It further lacks options for how to account for such impacts, including those that are difficult to quantify.

Over time, implementation across the states has led to inconsistent application of the traditional tests. The result has been a myriad of variations of the tests, in particular the TRC test. For example, a TRC test in one state can look more like an SCT (e.g., due to the inclusion of environmental impacts), and TRC test results from one state to another often vary considerable due to different treatment of non-energy benefits where many states do not include benefits that are hard to quantify, thus resulting in asymmetrical treatment of costs and benefits. As a result, the benefit-cost ratios of similar programs using the TRC test are not comparable across jurisdictions—and the test itself is no longer the TRC test in its pure and intended definition.

More broadly, as the electricity industry evolves to increasingly plan for and implement DERs, there is a need for a comprehensive cost-effectiveness framework that jurisdictions can use to apply to all DERs. The core principles and concepts in this NSPM can be used as the foundation for developing cost-effectiveness practices for all types of DERs.

Scope of this Manual

This NSPM focuses on the assessment of EE resources whose acquisition is funded by, and implemented on behalf of, electricity and gas utility customers, and where the value of efficiency resources is assessed using estimates of avoided utility system costs and other relevant impacts. The manual is intended as a tool to inform decision-making regarding which particular EE program (or set of programs) should be implemented using customer funding.

³ While most jurisdictions have historically used the CaSPM as the foundation for their cost-effectiveness tests, in practice many jurisdictions have deviated from those tests.

Note that the cost-effectiveness practices described in this manual are similar to integrated resource planning (IRP) practices, but different in some important respects.

The concepts in this NSPM can also apply to the assessment of other types of efficiency resources, such as building codes and appliance standards, governmentfunded efficiency resources, tax incentives for efficiency improvements, and more. However, this manual is focused on the assessment of ratepayer-funded EE programs because these programs have different types of costs and benefits and typically require more regulatory review and oversight.

Applicability to Other Types of Utility Resources

While this NSPM focuses on the assessment of utility EE resources, the core concepts can be applied to other types of utility resources as well. The costeffectiveness principles described in Chapter 1 and the Resource Value Framework described in Chapter 2 can be used to assess the cost-effectiveness of supply-side or distributed energy resources—including EE, demand response, distributed generation, distributed storage, electric vehicles, and strategic electrification technologies.

With regard to supply-side resources, the cost-effectiveness principles can be used in the context of integrated resource planning or when conducting any sort of economic analyses of specific generation, transmission, or distribution infrastructure investments. The Resource Value Framework can be used to identify the primary test for assessing these supply-side

Integrated Resource Planning (IRP) – the Other Way to Assess Cost Effectiveness

Some jurisdictions use long-term, IRP to help identify the portfolio of resources (supply-side and demand-side) that is least-cost and meets energy policy goals. Such IRP processes typically involve optimizing the costs, performance, and other attributes of all resource options in a dynamic fashion using optimization models, scenario analyses, and sensitivity analyses.

The cost-effectiveness practices described in this manual are similar to IRP practices, but different in some important respects. Both practices compare the long-run, marginal costs of different scenarios of resources to identify those with benefits that exceed costs, and both should use similar inputs regarding the future costs of EE, demand-side, and supply-side resources.

However, IRP and cost-effectiveness testing differ in that IRP typically allows for more sophisticated analyses of the impacts of EE impacts on utility system costs (e.g., modeling of EE loadshape impacts on power plant dispatch over time), and provides more flexibility for conducting scenario analyses and sensitivity analyses. On the other hand, though perhaps less dynamic, costeffectiveness analyses using fixed avoided cost assumptions is commonly used to assess EE at a more granular level. It allows for assessment of a range of different types of programs, program designs, and even efficiency measures.

investments, or to identify the criteria that would be used to select the preferred resource plan in the context of an IRP. This approach would not only ensure sound practices for analyzing supply-side resources, it would also ensure that EE resources are analyzed comparably and consistently with supply-side resources.

With regard to DERs, the principles and Resource Value Framework can be used as the foundation for assessing their cost-effectiveness.⁴ However, there are important ways in

⁴ Most recent studies of DER cost-effectiveness use the CaSPM as a starting point. See for example (IREC 2013), (NYSERDA 2015), and (Consumers Union 2016).

which other types of DERs might need to be treated differently from EE resources. For example:

- Some costs and benefits of EE might not be applicable to other types of DER, and vice versa. Some of the costs and benefits of EE might have different magnitudes relative to other types of DERs, including time-varying differences and locational differences.⁵
- The approach for addressing rate, bill, and participant impacts might be different for different types of DERs.
- In some jurisdictions, the policy goals supporting other types of DERs might be different from those supporting EE.

These important DER-specific issues are beyond the scope of this NSPM, but should be addressed by each jurisdiction as they develop cost-effectiveness practices for DERs. In the future, this EE manual could be expanded to address these other types of DER specific issues.

How this Manual Differs from the California Standard Practice Manual

This Manual builds upon the concepts and techniques of the CaSPM by addressing limitations and applying lessons learned over the years in the use of the CaSPM "traditional" tests. The NSPM expands on the CaSPM in various ways:

- 1. It provides a set of *universal principles* that should be used to guide the development of cost-effectiveness tests.
- 2. It includes the foundational principle that a jurisdiction should consider *applicable policy goals* when developing its primary cost-effectiveness test; it thereby introduces the perspective of the regulator/agent relative to the relevant policy goals, which may differ from the perspectives provided in the CaSPM.
- 3. Rather than specify a set of pre-defined tests, it provides a *framework and a process* for a jurisdiction to develop *its own specific primary test* (or tests).
- 4. It provides more information on the different types of EE resource costs and benefits, and how they should be treated when developing a cost-effectiveness test.
- 5. It provides guidance on how to account for applicable hard-to-monetize costs and benefits, as well as guidance on how to apply qualitative considerations.
- 6. It provides guidance on how to develop inputs for cost-effectiveness tests, such as discount rates, early replacement of measures, free-riders, and spillover.

⁵ Appendix B provides a comparison of costs and benefits of EE relative to other types of DERs.
Format of this Manual

Guidance on the Resource Value Framework and associated RVT is organized as follows:

Part I provides guidance on *how to develop* cost-effectiveness tests using the Resource Value Framework. It sets forth the set of universal principles that can be applied to any cost-effectiveness assessment, and provides a step-by-step process for jurisdictions to use to develop their primary RVT. Examples are provided, along with guidance on the use of secondary tests.

Part II provides more detailed information to assist jurisdictions in *developing inputs* for their RVTs, with guidance on what to include or not in the test by applying the Resource Value Framework process, and determining values for the inputs used in their primary test.

Appendices provide further detail on topics which may be relevant for some jurisdictions.

The intended audience for Part I is for regulators and other decision makers, policymakers, program administrators, EE and other DER stakeholders, evaluators, and other EE practitioners. Part II provides detailed guidance on key topics for those interested in delving into more details.

Table 1 shows the layout of the NSPM, with descriptions of the topics covered in each chapter.

Part/Chapter	Торіс	Description			
Part I	Developing Cost-Effectiveness Tests Using the Resource Value Framework				
Chapter 1	Principles	Describes the key principles that should be applied in any resource cost-effectiveness assessment			
Chapter 2	The Resource Value Framework	Provides an overview of the Framework and embodied principles, describes the dynamic nature of the RVT and its relevance to traditional cost-effectiveness tests			
Chapter 3	Developing the Resource Value Test (RVT)	Sets forth the multi-step process for developing a primary test based on principles and framework set forth in Chapters 1-2; provides templates to document applicable policies, inputs, and results using a standard format			
Chapter 4	RVT Relationship to Traditional Tests	Provides examples of hypothetical RVTs, and describes how a jurisdiction's RVT could compare to the traditional tests: UCT, TRC and SCT			
Chapter 5	Secondary Cost- Effectiveness Tests	This chapter provides information about the potential role of secondary tests, their benefits and limits, and selecting and constructing such tests			
Part II	Developing Inputs f	or Cost-Effectiveness Tests			
Chapter 6	Energy Efficiency Costs and Benefits	Describes the range of EE costs and benefits, both utility system and non-utility system, and information for selecting impacts to include in tests			
Chapter 7	Methods to Account for Relevant Impacts	Provides guidance on options for accounting for relevant cost and benefits, including hard-to-quantify impacts as well as approaches for qualitatively including non-monetary impacts			
Chapter 8	Participant Impacts	Expands upon guidance in Chapter 3 regarding how to determine whether to include participant impacts in the RVT			
Chapter 9	Discount Rates	Describes ways to determine discount rates that are consistent with the jurisdiction's applicable policy goals			
Chapter 10	Assessment Level	Describes the advantages and disadvantages of assessing EE at measure, program, or portfolio levels, and assessment level for fixed costs			
Chapter 11	Analysis Period and End Effects	Describes the time period over which cost-effectiveness analysis should be conducted, and how to address any potential "end effects" problems			
Chapter 12	Early Replacement	Describes how to analyze the costs and benefits of replacing operating equipment before the end of its useful life			
Chapter 13	Free-Riders and Spillover	Describes how to address free-riders and spillover effects in cost- effectiveness analyses for jurisdictions that use net savings			
Appendices					
Appendix A	Traditional Cost- Effectiveness Tests	Summarizes the commonly used traditional cost-effectiveness tests from the California Standard Practice Manual			
Appendix B	DER Costs and Benefits	Summarizes similarities and differences in costs and benefits across different types of DERs			
Appendix C	Rate and Bill Impacts	Describes key factors affecting rates and bills, and an approach for assessing related trade-offs			
Appendix D	Glossary of Terms	Provides definitions for commonly used terms throughout the manual			

Table 1. Overview of the National Standard Practice Manual

Key Terminology Used in this Manual

Terms with specific meaning in the context of the concepts offered in this NSPM are provided below, with additional terms in Appendix D.

- <u>Avoided costs</u>, refers to the costs of those electricity and gas resources that are deferred or avoided by the EE resources being evaluated for cost-effectiveness. The avoided costs are what make up the utility system benefits of EE resources.
- <u>Distributed energy resources</u> (DERs), refers to electricity and gas resources that are installed on customers' premises (behind the meter), to improve customer consumption patterns and reduce customer costs. These include EE, demand response, distributed generation, storage, plug-in electric vehicles, strategic electrification technologies, and more.
- <u>Energy efficiency resource</u>, refers to EE technologies, services, measures, or programs funded by, and promoted on behalf of, electricity and gas utility customers.
- <u>Impacts</u>, refers to both the costs and the benefits of a supply-side or demandside resource.
- <u>Jurisdiction</u>, refers to states, provinces, utilities, municipalities, or other regions for which EE resources are planned and implemented.
- <u>Primary cost-effectiveness test</u>, refers to the cost-effectiveness framework that a jurisdiction most relies upon when choosing the efficiency resources in which to invest ratepayer money.
- <u>Regulators and Other Decision Makers</u>, refers to institutions, agents, or other decision-makers that are authorized to determine utility resource costeffectiveness and funding priorities. Such institutions or agents include public utility commissions, legislatures, boards of publicly owned utilities, the governing bodies for municipal utilities and cooperative utilities, municipal aggregator governing boards, and more.
- <u>Regulatory perspective</u>, refers to the perspective of regulators or other agents that oversee efficiency resource investment choices. This perspective is guided by the jurisdiction's energy and other applicable policy goals—whether in laws, regulations, organizational policies, or other codified forms—under which they operate.
- <u>Resource Value Framework</u>, refers to a series of seven steps that can guide any jurisdiction to develop its primary test for assessing EE (and other DERs) cost-effectiveness. The Resource Value Framework embodies the key principles of cost-effectiveness analyses described in Chapter 1.
- <u>Resource Value Test (RVT)</u>, refers to the primary cost-effectiveness test that a
 jurisdiction has developed using the Resource Value Framework. It embodies all
 of the key principles of cost-effectiveness analyses and accounts for that
 jurisdiction's applicable policy goals.
- <u>Utility system</u>, refers to all elements of the electricity or gas system necessary to deliver services to the utility's customers. For electric utilities, this includes generation, transmission, distribution, and utility operations. For gas utilities, this includes transportation, delivery, fuel, and utility operations. This term refers to any type of utility ownership or management, including investor-owned utilities, publicly owned utilities, municipal utility systems, cooperatives, etc.

PART I. Developing Cost-Effectiveness Tests Using the Resource Value Framework

1. Principles of Cost-Effectiveness Analyses

This chapter presents the six core principles that are embodied in the Resource Value Framework and are fundamental to helping guide jurisdictions in the development of their primary cost-effectiveness test. These principles represent sound economic and regulatory practices and are consistent with the input received from a wide range of stakeholders during the development of this manual.

The following principles should be applied when developing and applying a jurisdiction's primary EE cost-effectiveness test:

- 1. Efficiency as a Resource. EE is one of many resources that can be deployed to meet customers' needs, and therefore should be compared with other energy resources (both supply-side and demand-side) in a consistent and comprehensive manner.
- 2. Applicable Policy Goals. A jurisdiction's primary cost-effectiveness test should account for its energy and other applicable policy goals. These goals may be articulated in legislation, commission orders, regulations, advisory board decisions, guidelines, etc., and are often dynamic and evolving.
- **3.** Hard-to-Quantify Impacts. Cost-effectiveness practices should account for all relevant, substantive impacts (as identified based on policy goals,) even those that are difficult to quantify and monetize. Using best-available information, proxies, alternative thresholds, or qualitative considerations to approximate hard- to- monetize impacts is preferable to assuming those costs and benefits do not exist or have no value.
- **4. Symmetry.** Efficiency assessment practices should be symmetrical, for example by including both costs and benefits for each relevant type of impact.
- **5.** Forward Looking. Analysis of the impacts of efficiency investments should be forward-looking, capturing the difference between costs and benefits that would occur over the life of efficiency measures and those that would occur absent the efficiency investments.⁶
- 6. **Transparency.** Efficiency assessment practices should be completely transparent and should fully document all relevant inputs, assumptions, methodologies, and results.

These principles are relevant to cost-effectiveness analyses of any resource, supply or demand, and are embodied within the Resource Value Framework provided in this manual. The key issues associated with their application to such analyses will differ

⁶ As further discussed in this chapter, sunk costs and benefits are not relevant to a cost-effectiveness analysis.

somewhat from resource to resource, depending on the unique characteristics of each resource.

Principle #1: Efficiency as a Resource

EE is a resource that can be used to defer or avoid spending on other electricity or gas resources. Consequently, an EE cost-effectiveness assessment should enable a full and fair assessment of the benefits and costs of the efficiency resource relative to other types of resources. The assessment should include comparisons to both supply-side resources and other demand resources to ensure accurate results. This principle necessitates that utility system costs and benefits always be included in cost-effectiveness analyses (see more detailed discussion in Chapter 3).

Principle #2: Applicable Policy Goals

A jurisdiction's EE cost-effectiveness framework should account for the energy and other applicable policy goals and objectives that apply to that jurisdiction. The choice between an investment in EE or investments in other demand and/or supply resources—i.e., what

Each jurisdiction's primary costeffectiveness should recognize the full "resource value" of EE. happens if efficiency investments are not made—can materially affect the costs, timeframe, and even ability to achieve such other policy goals. Cost-effectiveness analyses should guide or inform resource choices in that context.

Thus, each jurisdiction's primary cost-

effectiveness test should include all categories of relevant impacts (costs and benefits) consistent with its applicable policy goals. In other words, each jurisdiction's primary cost-effectiveness should recognize the full "resource value" of EE.

A jurisdiction's applicable policy goals are formally stated policy objectives that provide the overall policy context within which regulators and other agents make decisions regarding utility resource investments. These goals can be articulated in several different ways, including: legislation; executive orders; regulations; commission or board guidelines, standards or orders; and other pronouncements from a relevant governing agency. Importantly, identifying applicable policies for a jurisdiction is not a static process, but likely to evolve. For example, some jurisdictions may not have explicit statutes or regulations that address certain impacts that have been identified as important by stakeholders. In these instances, stakeholder input and due process often inform such policy development.

Table 2 below provides examples of policy goals. Some of these goals may overlap with each other, as is the case with reducing

'Regulators/decision-makers' refers to all types of entities that oversee EE investments such as: utility regulators; boards or management teams of unregulated municipal or cooperative utilities; or federal, regional, or state power planning agencies.

Energy and other applicable policy goals often evolve over time in response to changes in the energy industries, changing perspectives from the legislature and regulators, and the evolving interests of and input from industry stakeholders. As such, identifying applicable policies for a jurisdiction is not a static process, but likely to evolve (e.g., as part of regulatory processes and stakeholder discussions.) The jurisdiction's costeffectiveness test(s) may need to periodically evolve as well.

system risk and promoting resource diversity. Others may sometimes conflict with each other, as with reducing utility system costs and improving reliability, promoting customer

equity, and/or reducing environmental impacts. Such trade-offs can only be systematically assessed and EE investment decisions can only be optimized if costeffectiveness analyses account for all categories of impacts relevant to the jurisdiction's goals. Importantly, the constellation of applicable policy goals in any one jurisdiction is likely to differ in some ways from that of other jurisdictions.

Table 2. Examples of Energy-Related and Other Applicable Policy Goals⁷

Common Overarching Goals: Provide safe, reliable, low-cost electricity and gas services; protect low-income and vulnerable customers; maintain or improve customer equity.

Efficiency Resource Goals: Reduce electricity and gas system costs; develop least-cost energy resources; promote customer equity; improve system reliability and resiliency; reduce system risk; promote resource diversity; increase energy independence (and reduce dollar drain from the jurisdiction); reduce price volatility.

Other Applicable Goals: Support fair and equitable economic returns for utilities; provide reasonable energy costs for consumers; ensure stable energy markets; reduce energy burden on low-income customers; reduce environmental impact of energy consumption; promote jobs and local economic development; improve health associated with reduced air emissions and better indoor air quality.

Finally, this principle serves as a fundamental first step in developing a jurisdiction's primary costeffectiveness—the RVT, as discussed in Chapters 2 and 3. The primary test thus reflects a mix of various perspectives impacted by the jurisdiction's applicable policies, otherwise referred to within this NSPM as the 'regulatory' perspective.

Principle #3: Hard-to-Quantify Impacts

Ideally, all costs and benefits of EE resources that are relevant to a jurisdiction's applicable policy goals should be estimated in monetary terms, so that they can be directly compared. Fundamental to Principle #2 is the concept of the 'regulatory' perspective, which includes consideration of the full scope of issues for which regulators/decision-makers are responsible: (1) overall objective of requiring electricity/gas utilities to provide safe, reliable, low-cost services to customers; and (2) meeting their jurisdiction's other applicable policy goals.

Some impacts are challenging to quantify and put into monetary terms. Data may not be readily available, studies may require a considerable amount of time and/or resources to implement, and such studies might still result in significant uncertainty. That can be the case for impacts that are common to assessment of any type of resource. Examples include some utility system impacts (e.g., forecasts of resource needs and costs, impacts of future government regulations, and the magnitude and value of risk mitigation) as well as impacts that can be relevant to other jurisdictional policy objectives (e.g., value of reduced environmental impacts). It can also be the case for some impacts that may be unique to efficiency resources (e.g., benefits of improved comfort or business productivity).

Nevertheless, efficiency costs and benefits that are relevant to a jurisdiction's applicable policy goals and that can reasonably be assumed to be real and substantial should not be excluded or ignored because they are difficult to quantify and monetize. There are a

⁷ This list is not intended to be exhaustive, nor is it intended to imply a recommendation of any policies for any jurisdiction. It is intended to illustrate the types of policies that jurisdictions typically establish.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001A Page 31 of 146

variety of ways to develop estimates of impacts that are reasonable enough to inform investment decisions (see discussion in Chapter 7). Using "best available" information to approximate hard-to- quantify impacts is preferable to assuming that those costs and benefits do not exist or have no value. In a worst-case scenario, excluding substantive impacts from efficiency resource assessment will lead to results that are inaccurate and misleading.

Using "best available" information to approximate hard-to-quantify impacts is preferable to assuming that those costs and benefits do not exist or have no value.

Principle #4: Symmetry

For each type of impact included in a cost-effectiveness test, it is important that both the costs and the benefits be included in a symmetrical way. Otherwise, the test may be skewed and provide misleading results.

For starters, this means that all utility system costs (i.e., costs of running efficiency programs) and all utility system benefits (see Chapter 6 for a more detailed discussion of the range of utility system benefits) should be included in cost-effectiveness analyses.

It is important that both the costs and the benefits be included in a symmetrical way. Otherwise, the test may be skewed and provide misleading results. In addition, if a jurisdiction's applicable policy goals dictate that impacts on efficiency program participants be included in its costeffectiveness test, then both costs borne by those participants and benefits received by those participants should be included. On the cost side, this would most commonly be a portion of the efficiency measure costs (e.g., if the incremental cost of an efficiency

measure is \$1,000 and the utility program is providing a rebate of \$300, then the participants are incurring the remaining \$700 cost).⁸ On the benefits side, depending on the measures or program, there may be a variety of non-energy benefits that are part of the reason a customer invested in the measure (e.g., improved comfort, improved building durability, improved business productivity, etc.). If the participant costs are included in the cost-effectiveness test, then such benefits would need to be included as well.

Similarly, if a jurisdiction's applicable policies dictate that other categories of impacts should be included in its cost-effectiveness test—whether other fuel, water, low income, environmental, public health, economic development, and/or other impacts—then all incremental⁹ negative (cost) and positive (benefit) impacts should be captured in the test.

⁸ In this example, the \$300 rebate would already be included in the cost-effectiveness analysis as a utility system cost.

⁹ Some of these impacts may already be partially captured in utility system impacts. For example, some environmental impacts may be captured in estimates of avoided costs that capture the impact of current and/or projected future environmental regulations. Thus, to avoid double-counting, only additional "incremental" impacts should be included.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001A Page 32 of 146

Principle #5: Forward-Looking Analyses

Analysis of the impacts of efficiency investments should be forward-looking, capturing the difference between costs and benefits that would occur over the life of efficiency measures and those that would occur absent the efficiency investments.

This principle embodies three inter-related concepts. First, cost-effectiveness analyses should only consider forward-looking impacts. Historical (or "sunk") costs should not be included when estimating the impacts of future investment decisions — they cannot be

changed and will remain in place under any future scenario. Therefore, they are not relevant when comparing future investment scenarios.¹⁰

Second, cost-effectiveness analyses should include long-run costs and benefits. Electric and gas resources, including many efficiency resources, can last decades. As a result, often the resource decisions made today will affect customers far into the future. Utilities Historical (or "sunk") costs should not be included when estimating the impacts of future investment decisions —they cannot be changed and will remain in place under any future scenario.

have a responsibility to meet customer needs in a safe, reliable, and low-cost way over the long term. Regulators have a responsibility to protect customers over both the short term and the long term. Over-emphasis on short-term costs could unduly increase longterm costs for customers (see Chapter 11) for related discussion of analysis periods and Chapter 9 for discussion of discount rates used to analytically balance trade-offs between short-term and long-term impacts).

Third, cost-effectiveness analyses should consider only marginal impacts. These are defined as the incremental changes that will occur because of the EE resource, relative to a scenario where the resource is not in place.

Principle #6: Transparency

EE cost-effectiveness analyses require many detailed assumptions and methodologies, and they typically produce many detailed results. For regulators, other decision-makers, and other stakeholders to properly assess and understand cost-effectiveness analyses—and therefore to ultimately ensure that cost-effectiveness conclusions are reasonable and robust—key inputs, assumptions, methodologies, and results should be clearly documented in sufficient detail to enable independent reproduction of cost-

Results should be clearly documented in sufficient detail to enable independent reproduction of costeffectiveness screening results. effectiveness screening results. This should include all aspects of the resource assessment, including: all costs and benefits included (including all hard-to-monetize impacts); modeling parameters such as study period, treatment of risk, and discount rates; and approaches to account for additional

¹⁰ Historical costs do have important implications for rate impacts and potential cost shifting between customers. These costs should be considered in a separate rate impact analysis, as discussed in more detail in Appendix C.

considerations.¹¹ Such documentation should also be sufficient to replicate calculated cost-effectiveness values.

The purpose of the *Transparency Principle* is to support clear and accessible information regarding (1) the underlying jurisdiction's policies used to identify relevant impacts for inclusion in the primary test; and (2) reporting of key assumptions, results, and references from the cost-effectiveness analyses. This principle also serves as the final step in the Framework process. In Chapter 3, template tables are provided to support jurisdictions in applying this principle.

¹¹ Because the cost-effectiveness of EE is measured relative to the avoided costs of other resources, the assessment of those avoidable costs should be similarly transparent.

2. The Resource Value Framework and Primary Test

This chapter introduces the Resource Value Framework as a multi-step process to develop a jurisdiction's primary cost-effectiveness test – the RVT. The chapter includes an overview of the purpose of a primary test, the dynamic nature of the RVT, and its relevance to traditional cost-effectiveness tests.

2.1 Summary of Key Points

- Jurisdictions typically require a primary test to identify cost-effective efficiency resources. The Resource Value Framework is a 7-step process for jurisdictions to develop their primary cost-effectiveness test: the Resource Value Test (RVT).
- The Framework embodies the universal principles presented in Chapter 1, and in some cases discrete steps in multi-step process reflect application of a specific principle.
- While the RVT serves as a primary cost-effectiveness test, there can be value in assessing cost-effectiveness of efficiency resources from perspectives represented by other, secondary tests.
- The RVT is based upon a dynamic concept, where categories of impacts included in the test can vary across jurisdictions and/or over time because it is based on each jurisdiction's applicable policy concerns, which can vary.

2.2 The Resource Value Framework

The Framework is a series of seven steps, as shown below, that can guide any jurisdiction to develop its primary EE cost-effectiveness test. The Framework embodies the key principles described in Chapter 1, some of which represent a specific step in the framework process. Chapter 3 provides details on each of these steps.

Step 1: Identify and articulate the jurisdiction's applicable policy goals.

Step 2: Include all the utility system costs and benefits.

Step 3: Decide which non-utility impacts to include in the test, based on applicable policy goals.

Step 4: Ensure that the test is symmetrical in considering both costs and benefits.

Step 5: Ensure the analysis is forward looking and incremental.

Step 6: Develop methodologies to account for all relevant impacts, including hard to quantify impacts.

Step 7: Ensure transparency in presenting the inputs and results of the costeffectiveness test.

The relationship between the Framework, the underlying principles, and development of a primary RVT is provided in Figure 1 below and summarized further in this chapter.

Figure 1. The Foundation to Developing a Jurisdiction's Primary Test



2.3 The Resource Value Test as the Primary Test

Jurisdictions typically rely upon a primary test to identify cost-effective efficiency resources. Developing a single, primary test can be useful when comparing many different types and scenarios of efficiency resources, and it is often necessary when an efficiency resource passes one type of test, but not others.

The primary test should answer the fundamental question: *Which efficiency resources have benefits that exceed costs, where these impacts are defined by the jurisdiction's applicable policy goals*? The Resource Value Framework's underlying principles and multi-step process can support a jurisdiction's effort to answer this question, resulting in a comprehensive and transparent process that can help inform decisions on efficiency policies and practices in the jurisdiction.

The RVT serves as a primary test which assesses cost-effectiveness of efficiency resources relative to a jurisdiction's applicable policy goals that are under the purview of the jurisdiction's regulators and/or other decision-makers. However, there can be value in assessing cost-effectiveness of efficiency resources from perspectives represented by

other, secondary tests.¹² Among the potential purposes of using secondary tests are:

- To inform decisions regarding how much utility customer money could or should be invested to acquire costeffective savings;
- To inform decisions regarding which efficiency programs to prioritize if not all cost-effective resources will be acquired;

The primary test should answer the fundamental question: Which efficiency resources have benefits that exceed costs, where these impacts are defined by the jurisdiction's applicable policy goals?

- To inform efficiency program design; and
- To inform public debate regarding efficiency resource acquisition.

For example, the primary cost-effectiveness test is necessary but may not be sufficient for answering a second critical question: *How much utility customer funding should be*

¹² Chapter 5 provides more detail on the use of multiple cost-effectiveness tests.

spent on EE resources? This question will need to be answered by considering multiple factors such as:

- The results of the primary cost-effectiveness test;
- The results of secondary cost-effectiveness tests;
- Statutory or other requirements to implement all cost-effective EE;
- Statutory or other budget caps or constraints on efficiency resources;
- Statutory or other EE resource standards or other targets;
- Goals related to customer equity, or to providing access to all customer classes and customer types;
- Goals related to minimizing lost opportunities, or to addressing all electricity and gas end-use markets; and
- Rate, bill, and participation impacts of efficiency resources.¹³

2.4 The RVT as a Dynamic Test

The RVT reflects the impacts for which regulators/other decision-makers are responsible, including utility system impacts plus the impacts related to applicable policy goals. As such, different jurisdictions have different policy goals, and therefore they may develop different RVTs. While the RVT is conceptually a single test, in practice it might be different across jurisdictions because jurisdictions typically have a different mix of applicable policies that inform the inclusion of costs and benefits to the cost-effectiveness assessment.

The RVT is, therefore, based upon a *dynamic concept*, where categories of impacts included in the test can vary across jurisdictions and/or over time because jurisdictions' policy objectives can vary. This differs from the most common traditional tests—the UCT, TRC, and SCT—which are by associated perspectives (utility, utility plus participant, and society as a whole) *conceptually static*. The RVT can be tailored to a jurisdiction's specific interests and goals, while adhering to sound economic and public policy principles. The RVT thus provides a jurisdiction with flexibility to align with its energy and other applicable policies goals, and not be limited to the traditional tests.

A jurisdiction's application of the Framework may result in developing a primary RVT that is the same as one of the traditional tests (UCT, TRC or SCT.) This could happen if the jurisdiction's applicable policy goals are conceptually aligned with one of those traditional tests. See Chapter 4 for examples and more details.

The dynamic nature of the RVT means that for any jurisdiction, depending on its applicable policy goals, the regulatory perspective (as described in Chapter 1) may be the same as or broader than the utility perspective. Or, it may be the same as or narrower than the societal perspective, if indeed a jurisdiction's policies reflect taking into consideration the range of all costs and benefits to society. Regulators/other decision-makers in some jurisdictions might have a relatively broad scope of responsibilities, based on their specific policy goals, while others may have a relatively narrow scope.

Chapter 3 provides detailed information on how jurisdictions can use the Framework to develop an RVT using the 7-step process. Chapter 4 provides examples of RVTs, including how they compare to common traditional cost-effectiveness tests.

¹³ Appendix C provides a discussion of techniques for accounting for rate and bill impacts.

3. Developing the Resource Value Test

This chapter sets forth the detailed step-by-step process for developing a jurisdiction's primary cost-effectiveness test. The chapter ties in the principles introduced in Chapter 1, and provides template tables jurisdictions can use to support transparency in documenting cost-effectiveness analyses assumptions and results.

The Resource Value Framework's multi-step process, outlined in Figure 2 below, can be used to develop a jurisdiction's RVT as the primary cost-effectiveness test. This chapter provides guidance on each of these steps, and references relevant chapters and appendices where more detailed information is provided.

Figure 2. The Resource Value Framework Steps

STEP 1	Identify and articulate the jurisdiction's applicable policy goals.
STEP 2	Include all the utility system costs and benefits.
STEP 3	Decide which non-utility impacts to include in the test, based on applicable policy goals.
STEP 4	Ensure that the test is symmetrical in considering both costs and benefits.
STEP 5	Ensure the analysis is forward looking and incremental.
STEP 6	Develop methodologies to account for all relevant impacts, including hard to quantify impacts.
STEP 7	Ensure transparency in presenting the inputs and results of the cost-effectiveness test.

The first step is to identify and articulate the applicable policy goals of the jurisdiction. Articulating these goals at the outset of developing a framework, using a transparent process, will help ensure that the cost-effectiveness test is designed to properly account for them.

The second step is to recognize that EE is a resource that can be used to defer or avoid other energy resources, which requires that EE costs and benefits be evaluated consistently with the costs and benefits of other energy resources. As such, a cost-effectiveness test should begin by including all utility system impacts.

The *Key Principles* from Chapter 1 are embodied in the 7-step process, and in some cases, represent a discrete step.

The third step is to ensure that non-utility system impacts—both costs and benefits associated with the jurisdiction's applicable policy goals are accounted for.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001A Page 38 of 146

Once these first three steps are taken, then it is critical to ensure symmetry in the inclusion of the relevant impacts; to ensure the analysis is forward-looking and incremental; and to develop methods to account for all the relevant impacts. The final step is to provide transparency in presenting the inputs and results from the cost-effectiveness analysis.

3.1 STEP 1: Identify and Articulate Applicable Policy Goals

3.1.1 The Importance of Policy Goals

The first step is for a jurisdiction to identify and

articulate its applicable policy goals, consistent with the *Policy Goals Principle* from Chapter 1. Documenting applicable goals at the outset of developing a test is necessary to ensure that the cost-effectiveness test explicitly and properly accounts for such goals.

Most regulators/decision-makers have broad statutory authority to: set rates that are fair,

just, and reasonable; ensure that utilities and comparable entities provide customers with safe, reliable, and low-cost services; and generally guide utility actions that are in the public interest. This authority is typically defined in statutes and related regulations or other governing body decisions.

Most regulators/decision-makers also operate in the context of other relevant policies that affect their jurisdiction, many of

which are applicable to the investment of customer funds in EE resources. Table 2 (in Chapter 1) provides examples of such policies.

These goals are established in many ways, typically by statutes, regulations, orders, state energy plans, and other government directives. As emphasized earlier, these policy goals evolve over time to reflect changing conditions and governmental and public priorities.

Importantly, this first step of the Framework establishes a regulatory perspective, which reflects a mix of the various perspectives impacted by the jurisdiction's applicable policies.

3.1.2 Documenting Applicable Policy Goals

Transparency of a jurisdiction's applicable policy goals is key to helping identify the relevant costs and benefits to include a primary cost-effectiveness test. Table 3 illustrates a simplified version of how a jurisdiction could articulate its applicable policy goals. It shows how a jurisdiction's laws, regulations, orders, etc. could be documented to identify the relevance of certain policy goals to efficiency cost-effectiveness assessment. This exercise would help to provide a clear platform from which interested parties can inform and confirm priorities, gaps, or missing needs, and identify appropriate costs and benefits.

Review all 7 steps on page 18.

STEP

This first step of the Framework

establishes a regulatory perspective,

which reflects a mix of the various

jurisdiction's applicable policies.

perspectives impacted by the

	Policy Impacts Reflected in Laws, Regulations, Orders, etc.						
Applicable Laws, Regulations, Orders, etc.	Least- Cost	Fuel Diversity	Risk	Reliability	Low- Income	Environ -mental	Economic Develop- ment
PSC statutory authority	Х			Х			
Low-income protection	Х		Х	Х	Х		
EE or DER law or rules	Х	Х	Х	Х	Х		Х
State energy plan	Х	Х	Х	Х	Х	Х	Х
Integrated resource planning	x	х	х	х	х	х	Х
Renewable portfolio standard		х				х	Х
Climate change		Х	Х			Х	
Environmental protection		Х	Х			Х	

Table 3. Example Summary of a Jurisdiction's Applicable Policy Goals

This table is presented for illustrative purposes only, does not represent the policies of any particular jurisdiction, and is not meant to be an exhaustive list of applicable policy goals.

A more comprehensive version of the table above would ideally also:

- document the specific applicable policies;
- include a description of the relevant applicable policies;
- identify areas where policies are evolving or may evolve and should be considered; and
- identify the specific costs and benefits that should be accounted for in the test.

3.1.3 **Process and Stakeholder Input**

Some jurisdictions may have little experience or precedent for evaluating their applicable policy goals that are applicable to utility resource cost-effectiveness analyses. Other jurisdictions may have a long history of statutes, regulations, commission orders, and other directives that provide guidance on specific applicable policy goals. Either way, when developing a primary EE cost-effectiveness test, it is important to start with a clear articulation of all applicable policy goals.

Ideally, applicable policy goals should be assessed and articulated with a process that is transparent and open to all relevant stakeholders such as consumer advocates, low-income representatives, state agencies, efficiency representatives, environmental advocates, and others. Key stakeholders can provide important viewpoints regarding the value of EE in the context of the jurisdiction's policy goals.

This stakeholder input can be achieved through a rulemaking process, a generic jurisdiction-wide docket, commission orders on specific EE plans, working groups, technical sessions, or other approaches appropriate for the jurisdiction. The process should address objectives based on current jurisdiction policies, and should also be flexible to address new or modified polices that are adopted over time.

Some jurisdictions may wish to incorporate input from government agencies or representatives that do not typically make decisions regarding EE cost-effectiveness, but would nonetheless have insights on the jurisdiction's applicable policy goals. For example, a state's public utility commission may wish to incorporate input from that

state's department of environmental protection or department of health and human services (Regulatory Assistance Project 2013a).¹⁴

3.2 STEP 2: Include Utility System Costs and Benefits

The second step in developing an RVT is to include the utility system impacts that will be affected by the efficiency resource. The term utility system is used



Review all 7 steps on page 18.

here to represent the entire utility system used to provide service to retail customers. In the case of electric utilities, this includes the generation, transmission, and distribution of electricity services. In the case of gas utilities, this includes the transportation, storage, and distribution of gas services. This term refers to any type of utility ownership or management, including investor-owned utilities, publicly owned utilities, municipal utility systems, cooperatives, etc.

The utility system costs and benefits should provide the foundation for every costeffectiveness test. This ensures that the test will, at a minimum, indicate the extent to which total utility system costs will be reduced (or increased) by the efficiency resource over a specified period. It will also indicate the extent to which average customer bills will be reduced (or increased) by the efficiency resource, because total utility system costs determine average customer bills.¹⁵

It is essential to ensure that avoided cost estimates are comprehensive, up-to-date, informed by stakeholders, and ultimately reviewed and approved by regulators.

Further, every cost-effectiveness test should include relevant utility system costs and benefits. In terms of costs, this should include the portion of the efficiency measure paid by the utility, other financial or technical support provided to participants, and any other utility-system costs associated with program administration and management. Regarding benefits, this should include all the utility system costs that will be avoided or deferred by implementing the EE resource.¹⁶

Utility system avoided costs are one of the most important inputs to any costeffectiveness analyses of EE resources, and will significantly affect the results of the analyses. Therefore, it is essential to ensure that avoided cost estimates are

¹⁴ A recent statute in Michigan requires the commission to request an advisory opinion from the department of environmental quality regarding whether any potential decrease in emissions of sulfur dioxide, oxides of nitrogen, mercury, and particulate matter would reasonably be expected to result if the integrated resource plan proposed by the electric utility was approved (State of Michigan 2016).

¹⁵ Note that the three traditional cost-effectiveness tests, the UCT, the TRC, and the SCT, all include utility system impacts, at a minimum.

¹⁶ For the purposes of cost-effectiveness evaluation, the value of avoided utility system costs establishes the maximum amount that the utility system can contribute to a measure's costs, in order to be considered cost-effective without taking into consideration other participant and/or societal benefits and costs.

comprehensive, up-to-date, informed by stakeholders, and ultimately reviewed and approved by regulators.¹⁷

Including all utility system costs and benefits in any efficiency cost-effectiveness test is consistent with the *Efficiency as a Resource Principle* described in Chapter 1: that EE is a resource that should be compared with both supply-side and other demand-side energy resources in a consistent and comprehensive manner. Further, in a jurisdiction with competitive wholesale markets and distribution-only electricity utilities, it is important to account for the impacts on generation, transmission, and distribution because all these resources will be affected by the efficiency resource—even if distribution customers provide the funding of the efficiency resource.

Table 4 and Table 5 provide illustrations of the utility system costs and benefits that should be included in every cost-effectiveness test. Chapter 6 provides more detail on these utility system impacts, and Chapter 7 provides guidance on methods to develop values for these impacts.

Table 4. Example Electric Utility System Impacts to Include in Cost Effectiveness Tests

Scope	Costs	Benefits
Utility System	Measure Costs (utility portion) Other Financial or Technical Support Program Administration Marketing and Outreach Evaluation, Measurement, and Verification Utility Performance Incentives	Avoided Energy Costs Avoided Generating Capacity Costs Avoided T&D Costs Avoided T&D Line Losses Avoided Ancillary Services Wholesale Price Suppression Effects Avoided Costs of Complying with RPS Avoided Costs of Complying with RPS Avoided Environmental Compliance Costs Avoided Credit and Collection Costs Reduced Risk Increased Reliability

This table is presented for illustrative purposes, and is not meant to be an exhaustive list.

¹⁷ For good examples of this approach, see the New England Avoided Energy Supply Cost studies (AESC Study Group 2015); and the California Public Utility Commission cost-effectiveness calculator that embeds the state's official avoided costs in a model to calculate cost-effectiveness (CPUC 2016)

Table 5. Example Gas Utility System Impacts to Include in Cost-Effectiveness Tests

Scope	Costs	Benefits
Utility System	Measure Costs (utility portion) Other Financial or Technical Support Program Administration Marketing and Outreach Evaluation, Measurement, and Verification Utility Performance Incentives	Avoided Gas Costs Avoided Gas Pipeline Costs Avoided Gas Distribution Costs Avoided Gas Line Losses Wholesale Price Suppression Effects Avoided Environmental Compliance Costs Avoided Credit and Collection Costs Reduced Risk Increased Reliability

This table is presented for illustrative purposes, and is not meant to be an exhaustive list.

3.3 STEP 3: Decide Which Non-Utility Costs and Benefits to Include

The decision of which non-utility system costs and benefits to include in the RVT should build on Steps 1 and 2 of the Framework. Specifically, once a jurisdiction's applicable policies have been identified



Review all 7 steps on page 18.

and articulated in Step 1, and utility system costs and benefits are identified to account for overarching goal to reduce electricity/gas costs and customer bills, Step 3 then involves deciding which non-utility costs and benefits to include in the test, based on applicable policy goals.

In some cases, the decision to include an impact might be straightforward. For instance, legislation establishing an EE resource standard might explicitly state that one of the goals of the standard is to promote economic development. In other cases, the decision might be less clear. For example, whether to include participant costs and benefits in the primary EE cost-effectiveness test might not be articulated anywhere (as discussed in Section 3.3). In these cases, the policy decision will need to be made by regulators and other decision-makers with appropriate input from relevant stakeholders.

Table 6 below presents a summary of commonly considered non-utility impacts that could be included in a primary test to the extent they are relevant to a jurisdiction. The table also indicates the relevant section in this chapter where each of the impacts is summarized, with more detail provided in Chapter 6 on the considerations for selecting EE costs and benefits. In applying Step 3, regulators/ decision-makers, with input from stakeholders, can crossreference the broad range of non-utility costs and benefits addressed in this section, and further in Chapter 6. Jurisdictions can also build on the Table 3 template (from Step 1) by adding the specific costs and benefits that apply based on the identified applicable policy goals.

Table 6. Examples of Commonly Considered Non-Utility Impacts

Non-Utility Impact	Subsection	Description
Participant impacts	3.3.1	Impacts on program participants, includes participant portion of measure cost, other fuel savings, water savings, and participant non-energy costs and benefits
Impacts on low-income customers	3.3.2	Impacts on low-income program participants that are different from or incremental to non-low-income participant impacts. Includes reduced foreclosures, reduced mobility, and poverty alleviation
Other fuel impacts	3.3.3	Impacts on fuels that are not provided by the funding utility, for example, electricity (for a gas utility), gas (for an electric utility), oil, propane, and wood
Water impacts	3.3.4	Impacts on water consumption and related wastewater treatment
Environmental impacts	3.3.5	Impacts associated with CO ₂ emissions, criteria pollutant emissions, land use, etc. Includes only those impacts that are not included in the utility cost of compliance with environmental regulations
Public health impacts	3.3.6	Impacts on public health; includes health impacts that are not included in participant impacts or environmental impacts, and includes benefits in terms of reduced healthcare costs
Economic development and jobs	3.3.7	Impacts on economic development and jobs
Energy security	3.3.8	Reduced reliance on fuel imports from outside the state, region, or country

This table is presented for illustrative purposes, and is not meant to be an exhaustive list.

See also Step 6 in this chapter, and supporting Chapter 6, which provides information and guidance on methods for accounting for relevant costs and benefits.

3.3.1 Ensuring that Utility Customer Payments Are Justified by Customer Benefits

Regulators/decision-makers are sometimes concerned that including non-utility system impacts in the cost-effectiveness analysis could unduly burden utility customers, particularly customers who do not participate in EE programs. Regulators and consumer advocates sometimes ask: Why should electricity customers pay for participant gas or oil savings? Why should gas customers pay for participant electricity or oil savings? Why should utility customers pay for environmental, jobs, or other societal benefits?

The answer to these questions is that utility customers should pay for these benefits if called for by applicable policies in statutes, regulations, and orders, as consistent with *Policy Principle*. Presumably, the advantages of these policy benefits will outweigh the disadvantages. In many cases, such as with reliability, reduced risk, fuel diversity, economic development, energy security, and environmental benefits, all utility customers will collectively share in the non-utility system benefits.

3.3.2 Consider Participant Impacts

Efficiency program participants experience several types of costs and benefits. Program participant impacts are summarized in Table 7, and discussed in more detail in Chapters 6 and 8.

Affected Party	Costs	Benefits
Efficiency Program Participant	Measure Costs (customer portion) Financial Costs (customer portion) Transaction Costs Increased O&M Costs Increased Other Fuel Consumption Increased Water Consumption	Reduced Bills (typically reflected as avoided utility system costs) Reduced O&M Costs Increased Comfort Increased Health & Safety Increased Productivity Improved Aesthetics Property Improvements Reduced Other Fuel Consumption Reduced Water Consumption Additional Benefits for Low-Income Customers

Table 7. Program Participant Costs and Benefits

This table is presented for illustrative purposes and is not meant to be an exhaustive list. Note that some of these impacts are energy related with others are not. Those that are not energy related are conventionally referred to as non-energy costs or non-energy benefits.

When considering whether to include participant impacts in the cost-effectiveness tests, it is important to consider two overarching points:

- 1. The decision of whether to include participant impacts in the primary costeffectiveness test is a policy decision. Regulators may choose to include participant impacts in the primary cost-effectiveness test if that would achieve the jurisdiction's policy goals.
- 2. If regulators decide to include participant costs in any cost-effectiveness test, the test must also include participant benefits, and *vice versa*. This is necessary to ensure symmetrical treatment of participant impacts, consistent with *Symmetry Principle* set forth in Chapter 1.

With regard to the first point above, some jurisdictions may not have an explicit policy goal regarding whether to include program participant impacts when assessing EE resources. Legislators and other decision-makers may not have addressed this question when promulgating legislation or regulations related to EE resources. In these cases, regulators and other decision-makers should decide whether to include participant impacts based upon the policy context that does exist in the jurisdiction and with appropriate input from relevant stakeholders.

Rationale for Including Participant Impacts

Several key issues should be addressed when deciding whether to account for participant impacts in the primary cost-effectiveness test. Regulators and other decision-makers should determine whether there is a policy justification for including participant impacts in the primary test. They should also consider the rationale and advantages of including participant impacts in the primary test.

Table 8 provides a summary of the reasons to include participant impacts in their primary cost-effectiveness test, as well counter-points to these reasons. These points and counter-points are discussed in more detail in Chapter 8.

Table 8. Points and Counter-Points Regarding Whether to Include Participant Impacts

Reasons for Including Participant Impacts	Counter-Points
Including participant impacts accounts for the costs on all utility customers: participants and non-participants.	Participant impacts fall outside the scope of utility system impacts. If EE is treated purely as a utility system resource, then participant impacts are less relevant.
Including participant impacts accounts for the total cost of the resource. If the cost of a resource is split between two entities, then it might appear to be cost-effective when it is not.	If regulators prefer to account for the total cost of a resource in order to address concerns about costs being split between two entities, it is necessary to also account for the total benefits. This objective essentially requires the use of the SCT. If this objective is important enough, jurisdictions could use an SCT as a pre-screening test and an RVT as the primary test.
Including participant impacts will help protect program participants. Excluding such costs might result in participants paying "too much" for efficiency.	Including participant impacts will not accurately capture the benefits of program participants, because in practice the primary participant benefit is typically represented in terms of avoided utility costs, not reduced customer bills. The Participant Cost test is one way to protect participants. ¹⁸ In addition, program design is the best way to protect program participants, and sound program design will result in participants being better off.
Excluding participant impacts would exclude low-income participant benefits from the analysis	Low-income participant impacts can be included in the RVT, without including all participant impacts, if justified by policy goals. Well-defined low-income programs do not require participant costs, which eliminates the typical rationale for including participant impacts.
Excluding participant impacts would exclude other fuel and water impacts from the analysis.	Other fuel and water impacts can be included in the primary test, without including all participant impacts, if justified by policy goals.

Implications for Non-Participants

Including participant impacts in the cost-effectiveness test sometimes raises concerns about how this will affect non-participants. Should all utility customers pay for nonenergy benefits that are enjoyed by only participants? Will including participant impacts unduly increase the cost of EE for all

customers?

For those jurisdictions that choose to include participant impacts in the RVT, these concerns can be addressed through program design. The incentives offered to the EE program participant could be capped at a level equal to the utility system avoided costs. This would prevent non-participants Including participant impacts in the cost-effectiveness test sometimes raises concerns about how this will affect non-participants.

from paying more than the benefits they receive from the EE resource. This point is also discussed in Section 3.3.1.

In addition, recall that participant non-energy benefits should be included in the RVT if participant costs are included, and vice versa—consistent with the *Symmetry Principle*.

¹⁸ The Participant Cost Test is described in Appendix A. As noted there, the Participant Cost Test is not well-suited for the purpose of assessing the value of EE resources. Nonetheless, it could be used as a secondary test for the purpose of protecting participants.

Those jurisdictions that do not want to support EE programs as a result of benefits that accrue only to participants could decide to exclude participant costs and benefits in the primary cost-effectiveness test.

3.3.3 Consider Low-Income Impacts

It is widely acknowledged that efficiency programs serving low-income customers and low-income communities provide important benefits beyond utility system impacts. Table 9 presents a summary of the types of low-income impacts beyond utility system impacts.

Affected Party	Costs	Benefits
Efficiency Program Participant	Typically, none. Well-designed low- income programs cover all costs and remove all barriers to low-income customers.	Reduced energy burden Reduced O&M costs Increased comfort Increased health & safety/reduced medical costs Increased productivity Improved aesthetics Property improvements Reduced home foreclosures Reduced need to move/relocate due to unpaid bills
Society	Typically, none.	Alleviating poverty Improving low-income community strength and resiliency Reduced home foreclosures

Table 9. Non-Utility Low-Income Costs and Benefits

This table is presented for illustrative purposes, and is not meant to be an exhaustive list.

Many of the benefits to low-income participants accrue to non-low-income efficiency program participants as well. However, the magnitude of some of these benefits can be greater in low-income homes, because (a) the pre-program condition of low-income housing can be worse than that of non-low-income housing, and (b) because the financial condition of low-income customers often more significantly constrains how they manage and live in their homes.

As indicated in Table 9 some low-income benefits affect low-income program participants while some affect society in general. Other low-income benefits, such as reduced foreclosures, could be characterized as accruing to both the participant and society.

Jurisdictions that have policy goals requiring or encouraging the protection of low-income customers should include low-income impacts in their RVT. It is not The Colorado PUC requires Public Service Company of Colorado to account for low-income benefits by increasing avoided costs with a 25% proxy multiplier (Skumatz 2014).

necessary to include all participant impacts in the RVT in order to include low-income impacts.

Regulators and other decision-makers who choose to include low-income benefits in the RVT do not need to distinguish between benefits to the participant versus those to society. In both cases, the low-income benefits fall outside the scope of utility system impacts, and in both cases these benefits can be included in the primary test, as identified by the jurisdiction's applicable policies.

As noted earlier, some jurisdictions may not have explicit statutes or regulations that address whether low-income impacts should be included in EE cost-effectiveness analyses. In these instances, regulators should develop a policy on how to address low-income impacts; ideally with stakeholder input and due process.

3.3.4 Consider Other Fuel Impacts

Some efficiency resources can either reduce or increase the consumption of "other fuels," which includes fuels beyond those provided by the utility funding the efficiency resource. Other fuels can include savings or increased use of gas (for an electric utility funding the efficiency resource), electricity (for a gas utility funding the efficiency resources), oil, propane, biomass, or other fuels used in a home or business. Table 10 presents several examples of where other fuel impacts can occur in efficiency programs. Further detail on Other Fuels is provided Chapter 6.

Program Option	Description
Multi-fuel measures	When efficiency measures for one type of fuel result in savings of another type; for example, when insulation is installed in buildings that are cooled with electric air conditioning but heated with other types of fuels. Multi-fuel efficiency measures are frequently used in building retrofit programs and in new construction programs.
Fuel-optimization measures	When customers can choose from multiple fuel types to optimize the efficiency of an end-use. For example, customers may be given the option to switch from an inefficient oil heating system to a high-efficiency gas heating system.
Fuel-neutral programs	When regulators and efficiency planners choose to offer whole-building efficiency programs that address all fuel types with a single program provided by a single program administrator. This results in more efficient program delivery, fewer transaction costs, greater efficiency measure adoption, and better customer service in general.
Combined heat and power programs	When technologies are used to generate electricity efficiently, but require increased consumption in other fuels such as natural gas or biomass.
Strategic electrification options	When programs are designed to promote switching from non-electric to electric fuel for policy reasons. For example, an electric utility may wish to promote electric vehicles to achieve environmental and transportation policy goals.

Table 10. Examples of Other Fuel Impacts in Efficiency Programs

Some efficiency programs might include more than one of the program options listed above. For example, fuel-neutral programs typically include multi-fuel measures and can include fuel-optimization measures.

Jurisdictions that have policy goals promoting the efficient use of other fuels should include other fuel impacts in their RVT. This would be appropriate for jurisdictions with goals relating to multi-fuel measures, fuel-optimization measures, fuel-neutral programs, combined heat and power programs, or strategic electrification programs.

As described in Appendix C, it is not necessary to include participant impacts in the RVT in order to include other fuel impacts. Whenever other fuel impacts are included in a cost-effectiveness test it is important to ensure that the test properly accounts for both reductions and increases in the other fuels.

Illinois law requires that electric EE cost effectiveness testing account for quantifiable societal benefits, including avoided natural gas utility costs, and that natural gas EE cost-effectiveness considers other quantifiable societal benefits, including avoided electric utility costs (Illinois 2009).

3.3.5 Consider Water Impacts

Some efficiency measures affect the consumption of water resources, where efficiency can reduce water consumption and wastewater costs by making certain end-uses, such as water heaters, dish washers, or clothes washers, more efficient. EE measures can also reduce water consumption and wastewater costs by reducing the need for electricity generation from power plants that consume water (Regulatory Assistance Project 2013c). Further detail on water impacts is provided in Chapter 6.

Jurisdictions whose applicable policy goals require or encourage the reduction in water and wastewater resources should include these impacts in their RVT. It is not necessary to include participant impacts in the RVT in order to include water impacts. Either way, care should be taken to ensure there is no overlap in participant, utility, or societal water savings. Whenever these resources are included in a cost-effectiveness test it is important to ensure that both reductions and increases in water and wastewater resources are accounted for properly.

The Oregon Commission has determined that efficiency costeffectiveness analyses should include total costs and total benefits, including quantifiable non-energy benefits, which should encompass water savings (Oregon 1994).

3.3.6 Consider Environmental Impacts

Efficiency resources can provide a variety of benefits by reducing the environmental impacts of the energy resources that are avoided or deferred. Table 11 summarizes some of these key environmental benefits. In some cases, efficiency programs might cause environmental costs, which must be accounted for along with environmental benefits. Further detail on environmental impacts is provided in Chapter 6.

Table 11. Examples of Environmental Impacts of EE Resources

Types of Environmental Impacts

- Reduced carbon emissions
- Reduced emissions of criteria and other air pollutants
- Reduced liquid and solid waste (nuclear, coal ash, etc.)
- Reduced water for cooling electric generating stations, extracting natural gas (e.g., "fracking"), and other purposes
- Reduced adverse impacts on the land that must be developed for new generating facilities
- Reduced adverse impacts on land, air, and water from fuel mining or extraction

This table is presented for illustrative purposes and is not meant to be an exhaustive list. These environmental impacts can be in the form of costs or benefits. For each type of environmental impact included in the RVT, both costs and benefits, should be included.

The costs of complying with current and future environmental regulations should be included in the utility system costs. Only additional environmental impacts that might occur despite compliance with environmental regulations (i.e., residual impacts), should be considered a non-utility system impact. Regulators and efficiency planners should treat these two types of environmental impacts separately, to avoid double-counting.

Jurisdictions that have applicable policy goals requiring or encouraging the reduction of environmental impacts should include environmental impacts in their RVT.

3.3.7 Consider Public Health Impacts

One of the results of some of the environmental emission and waste reductions discussed above is a reduction in the frequency and/or severity of health problems of populations impacted by fuel extraction and combustion. Such reductions can reduce the level of societal investment required in medical facility infrastructure, as well as in the health, well-being, and economic productivity of the populace. District of Colombia law requires that in "supervising and regulating utility or energy companies, the Commission shall consider the *public safety,* the economy of the District, the conservation of natural resources, and the preservation of environmental quality" (District of Columbia 2008).

Public health benefits can take the form of direct benefits in health of the populace caused by reduced air emissions from power plant generation due to EE investments. Health issues typically considered here include those associated with poor air quality due to ozone or smog, such as respiratory problems and asthma. Public health benefits can also take the form of indirect benefits from reduced healthcare costs for customers.

In addition to improved *outdoor* air quality and associated public health impacts, EE investments in buildings can improve the health of occupants by addressing and improving *indoor* air quality (IAQ), largely through improved building envelope and ventilation measures. While direct health impacts to home occupants, especially related to reduced asthma incidences, are relevant to participant impacts (as addressed above), there are also important broader public health impacts associated with reduced emergency room visits, and associated medical costs.

Jurisdictions whose applicable policy goals include improving public health should include public health impacts in their RVT. Jurisdictions that choose to include

participant, environmental, and public health impacts should ensure that there is no double-counting across these three types of impacts.

3.3.8 Consider Economic Development and Job Impacts

All types of utility resource investments will have economic development and job impacts. EE resources will typically increase jobs and economic development, relative to investments in supply-side Rhode Island law establishes state greenhouse gas reduction goals, and articulates that "consideration of the impacts of climate change shall be deemed to be within the powers and duties of all state departments, agencies, commissions, councils, and instrumentalities..." (Rhode Island 2014).

resources. The types of jobs associated with EE generally fall into three categories:

- Jobs associated with managing, delivering, and evaluating the efficiency programs.
- Jobs associated with additional work and revenue that EE programs funnel to the supply chains associated with efficiency measures being installed in homes and

businesses; this includes contractors, builders/developers, equipment vendors, product retailers, distributors, manufacturers, and others (E4TheFuture 2016b, 4).

• Indirect impacts, where customers with reduced energy bills will have more disposable income that may be spent in the local community (or beyond), which helps create jobs and spur economic development.

Jurisdictions whose applicable policy goals include promoting jobs and economic development should include these impacts in their RVT. When this is done, it is necessary to also account for jobs lost or reduced economic development. In other words, the cost-effectiveness analysis should include *net* economic and job impacts from the efficiency program.

Delaware's Energy Conservation and Efficiency Act states that the benefits of cost-effective EE include new economic development opportunities (Delaware 2009).

3.3.9 Consider Energy Security

EE can reduce the consumption of fuels and resources that are imported from outside the relevant jurisdiction. This can include fossil fuels that are imported from other regions, electricity that is imported by transmission lines, and natural gas that is imported through pipelines. It can also include fossil fuels that are imported from other parts of the world, including countries that are politically or economically unstable. Over-reliance upon imported fuels can increase price volatility and increase risks associated with energy supply and reliability.

Jurisdictions whose applicable policy goals include promoting energy security should include these impacts in their RVT. When this is done, it is necessary to ensure that there is no double-counting of this impact in other impacts, such as utility-system risk impacts and jobs and economic development impacts.

3.4 STEP 4: Ensure the Test Is Symmetrical

Once it has been determined what categories of impacts to include in a jurisdiction's RVT in Step 3, Step 4 is to ensure that the test includes all costs and all benefits associated with each category of impacts. If some costs are excluded, the framework will be



A Washington statute states

conservation and the use of

appropriately sited renewable

energy facilities will promote energy independence in the

that "increasing energy

state and the Pacific

(Washington 2006).

Northwest region

Review all 7 steps on page 18.

inappropriately biased in favor of efficiency; if some benefits are excluded, the framework will be inappropriately biased against efficiency. If the test results in a bias

If the test results in a bias either in favor of or against EE resources, the result will be a misallocation of resources, with higher than necessary costs incurred by utility customers. either in favor of or against EE resources, the result will be a misallocation of resources, with higher than necessary costs incurred by utility customers. Hence the importance of applying the *Symmetry Principle* as a discrete step in the Framework process.

One example of where this is especially important is regarding program participant costs and benefits. Where states have used

the TRC test, which should include participant costs, most states do not in reality include

participant *benefits* (ACEEE 2012).¹⁹ This leads to a cost-effectiveness test that is skewed against EE. The results will understate the benefits of efficiency resources, and lead to higher utility costs than necessary (Regulatory Assistance Project 2012).

Figure 3 presents the percent of total benefits that are created by different types of benefits, including participant NEBs, using the results of cost-effectiveness analyses for actual efficiency programs operated by a Massachusetts electric utility (Eversource 2017). As indicated, participant NEBs can represent a large portion of total benefits, and will significantly affect the cost-effectiveness results.



Figure 3. Implications of Participant Benefits on Residential Efficiency Programs

Finally, applying the principle of symmetry sometimes requires estimating "net" impacts for certain types of benefits. For example, if economic development gains from EE resources are included in the cost-effectiveness framework, it is important to also include economic development losses associated with not implementing the avoided resources in the counter-factual scenario. This is frequently achieved by estimating net economic development gains from efficiency resources.

3.5 STEP 5: Ensure the Analysis Is Forward-Looking and Incremental

Step 5 applies the *Forward-Looking Principle,* which requires that cost-effectiveness analyses should be forward-looking and incremental. This requires accounting for future, long-run, marginal costs and benefits, which embodies three inter-related concepts.



Review all 7 steps on page 18.

¹⁹ Throughout this discussion, the term "participant benefits" refers to all of the benefits other than the reduction in the participant's utility bill.

- Cost-effectiveness analyses should only consider forward-looking impacts. Historical (or "sunk") costs should not be included when estimating the impacts of future investment decisions. Historical costs cannot be changed, and will remain in place under any future scenario, and therefore are not relevant when comparing future investment scenarios.²⁰
- 2) Cost-effectiveness analyses should include long-run costs and benefits. Electric and gas resources can last for forty or even sixty years. Thus, the resource decisions made today will affect customers for decades in the future. Utilities have a responsibility to meet customer needs in a safe, reliable, and low-cost way over the long term. Regulators have a responsibility to protect customers over both the short term and the long term. Over-emphasis on short-term costs could unduly increase long-term costs for customers.²¹
- 3) Cost-effectiveness analyses should consider only marginal impacts. These are defined as the incremental changes that will occur because of the EE resource, relative to a scenario where the resource is not in place.

3.6 STEP 6: Develop Methodologies to Account for All Relevant Impacts

Step 6 applies the *All Relevant Impacts Principle*. This requires that all relevant impacts of EE resources that a jurisdiction has chosen to assess via



Review all 7 steps on page 18.

its cost-effectiveness test should ideally be estimated in monetary terms. In this way, they can be readily compiled and compared directly. However, some EE impacts are difficult to quantify in monetary terms, either due to the nature of the impact or the lack of information available about the impacts.

Substantive EE resource costs and benefits should not be excluded or ignored because they are difficult to quantify and monetize. Approximating hard-to- quantify impacts is preferable to assuming that those substantive costs and benefits do not exist or have no value.

Table 12 summarizes five different approaches that can be used to account for all impacts of EE resources that a jurisdiction has chosen to include in its cost-effectiveness test. The approaches are listed in order of technical rigor and preference.

²⁰ Historical costs do have important implications for rate impacts and potential cost-shifting between customers. These costs should be considered in a separate rate impact analysis, as discussed in more detail in Appendix C.

²¹ Discount rates are used to enable the regulators to properly balance short-term and long-term impacts on customers. This topic is addressed in Chapter 9.

Approach	Description
Jurisdiction-specific studies	Jurisdiction-specific studies on EE costs and avoided cost offer the best approach for estimating and monetizing relevant impacts.
Studies from other jurisdictions	If jurisdiction-specific studies are not available; studies from other jurisdictions or regions, as well as national studies, can be used for estimating and monetizing relevant impacts.
Proxies	If monetized impacts are not available; well-informed and well- designed proxies can be used as a simple substitute.
Quantitative and qualitative information	Relevant quantitative and qualitative information can be used to consider impacts that cannot or should not be monetized.
Alternative thresholds	Pre-determined thresholds that are different from one (1.0) can be used as a simplistic way to account for relevant impacts that are not otherwise accounted for.

Table 12. Different Approaches to Account for All Relevant Impacts

3.7 STEP 7: Ensure Transparency

The *Transparency Principle* provided in Chapter 1 constitutes a discrete and final step in the Resource Value Framework process. Transparency is critical to supporting a successful RVT. EE cost-effectiveness analyses require many detailed assumptions and methodologies, and they typically produce many detailed results.



Review all 7 steps on page 18.

There are two key junctures where transparency is addressed in this NSPM. The first is addressed as part of Step 1 earlier in Chapter 3.1, which includes a template format (Table 3) for how a jurisdiction could articulate its energy and other applicable policy

goals. This exercise can help to provide a clear platform from which interested parties can confirm priorities, gaps or missing needs, and identify appropriate costs and benefits.

The second juncture for providing transparency is with regard to documenting the inputs, assumptions, and results of the cost-effectiveness analyses. A reporting template can be used to provide clear and consistent information for all interested parties. If used across jurisdictions, this template can provide comparability across cost-effectiveness assumptions and results to support sharing of data, where appropriate, and identification of possible opportunities for improvements in program design. Why Transparency? In order for regulators and other stakeholders to properly assess and understand cost-effectiveness analyses—and therefore to ultimately ensure that costeffectiveness conclusions are reasonable and robust—key inputs, assumptions, methodologies and results should be clearly documented in sufficient detail to enable independent reproduction of costeffectiveness screening results.

3.7.1 Template Reporting Table

As a jurisdiction applies the Resource Value Framework to develop its cost-effectiveness test, transparent documentation of all key inputs, assumptions, methodologies, and results will help ensure that the approach to cost-effectiveness analysis is consistent with fundamental economic principles. It will also help to support stakeholder discussions and input to regulatory and other policymaker considerations and decisions.

The use of a standard template will help to provide a comprehensive, consistent, and easily accessible structure for such documentation. The template should present both the monetized and non-monetized findings of the assessment. It should include references for all key assumptions and methodologies used. The scope of reporting can be at the program, sector, or portfolio level. The sample template is provided in Table 13 below.

Program/Sector/Portfolio Name:		Date:		
A. Monetized Utility System Costs		B. Monetized Utility System Benefits		
Measure Costs (utility portion)		Avoided Energy Costs		
Other Financial or Technical Support Costs		Avoided Generating Capacity Costs		
Program Administration Costs		Avoided T&D Capacity Costs		
Evaluation, Measurement, & Verification		Avoided T&D Line Losses		
Shareholder Incentive Costs		Energy Price Suppression Effects		
		Avoided Costs of Complying with RPS		
		Avoided Environmental Compliance Costs		
		Avoided Bad Debt, Arrearages, etc.		
		Reduced Risk		
Sub-Total Utility System Costs		Sub-Total Utility System Benefits		
C. Monetized Non-Utility Costs	1	D. Monetized Non-Utility Benefits	1	
Participant Costs		Participant Benefits	_	
Low-Income Customer Costs		Low-Income Customer Benefits	Include to the extent these impacts are	
Other Fuel Costs	Include to the	Other Fuel Benefits		
Water and Other Resource Costs	extent these	Water and Other Resource Benefits		
Environmental Costs	impacts are	Environmental Benefits		
Public Health Costs	RVT.	Public Health Benefits	RVT.	
Economic Development and Job Costs		Economic Development and Job Benefits		
Energy Security Costs		Energy Security Benefits		
Sub-Total Non-Utility Costs		Sub-Total Non-Utility Benefits		
E. Total Monetized Costs and Benefit	ts			
Total Costs (PV\$)		Total Benefits (PV\$)		
Benefit-Cost Ratio		Net Benefits (PV\$)		
F. Non-Monetized Considerations				
Economic Development and Job Impacts	Quantitative information, and discussion of how considered			
Market Transformation Impacts	Qualitative considerations, and discussion of how considered			
Other Non-Monetized Impacts	Quantitative information, qualitative considerations, and how considered			
Determination:	Do Efficiency	Resource Benefits Exceed Costs? [Y	es / No]	

Table 13. Efficiency Cost-Effectiveness Reporting Template

Note that the most useful and appropriate way to present the results of analyses of monetized efficiency costs and benefits is in present value (PV\$) terms. Present value is defined as the value today (or a given year) of a certain amount of money in the future, where the future value is converted to PV\$ using a discount rate. (See Chapter 9 for discussion of discount rates).

In addition, the PV\$ values should cover the full life of the resource being analyzed (see Chapter 11 for discussion of analysis periods), or what is sometimes referred to as the cumulative present value or the present value of lifecycle costs and benefits. A cumulative or lifecycle present value is the discounted sum of a stream of current and future annual costs and benefits.

3.7.2 Reporting Categories and Descriptions

The key reporting categories in Table 13, and supporting descriptions, are as follows:

- **Monetized Utility System Costs and Benefits.** Sections A-B of the template table report on the utility system impacts, the foundation of any cost-effectiveness analysis, consistent with the *Efficiency as a Resource Principle*. More detailed information on the sub-categories of utility system costs and benefits can be found in Chapter 6 of this manual.
- Monetized Non-Utility Costs and Benefits. Sections C-D of the template table report on the non-utility impacts, as identified and informed by the Framework Steps 1-6. Consistent with the *Symmetry Principle* for treatment of costs and benefits, for any category of costs included on the left side of the template in Table 13 (Section C) there should also be corresponding benefits included on the right side of the table (Section D)—and vice versa. More detailed information on the sub-categories of non-utility system costs and benefits can be found in Chapter 6 of this manual. A discussion of methodologies for monetizing impacts can be found in Chapter 7.
- **Benefit-Cost Ratios and Net Benefits.** Section E of the template table includes several reporting parameters that provide critical information regarding cost-effectiveness test results:
- Total Costs (PV\$) and Total Benefits (PV\$) are simply the sum of all monetized utility system and non-utility costs and benefits.
- Benefit-Cost Ratio is equal to the ratio of the cumulative present value of benefits to the cumulative present value of costs. This metric is especially useful as a simple benchmark for determining cost-effectiveness: if an efficiency resource's BCR exceeds 1.0, it means that benefits exceed costs. That criterion is typically used to indicate that something is cost-effective.

The BCR metric can be useful for comparing efficiency resources with each other (i.e., a higher BCR indicates one resource is "more cost-effective" than another), because it effectively normalizes the results for programs of different sizes. This metric is also useful for comparing efficiency resources across utilities and

jurisdictions of different sizes, again because it effectively normalizes the results for any differences in size.²²

The BCR metric provides an important element of information that is not provided by a net benefits metric. It does this by indicating the relative effectiveness of the money spent on the resource. i.e., how many dollars of benefits are received per dollar spent. For example, a net benefit of \$10 million in PV\$ does not indicate how much money was needed to generate those net benefits. It could have cost \$90 million, with benefits of \$100 million and a BCR of 1.11. Or it could have cost \$4 million, with benefits of \$14 million and a BCR of 3.50.²³

• Net Benefits (PV\$) is equal to the difference between the cumulative present value of benefits and the cumulative present value of costs. This metric is useful as a benchmark for determining cost-effectiveness: if an efficiency resource's net benefits are greater than zero, it should be deemed to be cost-effective.

The net benefits metric provides an important element of information that is not provided by the BCR metric, by indicating the absolute magnitude of the benefits to be gained by the efficiency resource. For example, a BCR of 2.2 does not indicate how much money will be saved by the resource. It might save \$1 million, \$10 million, or \$100 million.

The net benefits of efficiency resources cannot easily be used to compare efficiency resources across different utilities and jurisdictions. A large utility would naturally expect to have higher net benefits than a small utility for a comparable type of program.

• Non-Monetized Considerations. Section F of the template shown in Table 13 is where discussion of the non-monetized impacts should be summarized. See Chapter 7 for discussion of techniques for consideration of non-monetized impacts.

²² However, in making such comparisons it is important to recognize that different utilities and jurisdictions might have different avoided costs, i.e., different benefits for the same amount of savings. Different jurisdictions might also include different impacts in their resource assessment test.

²³ On the other hand, trying to maximize the BCR by including only measures/programs with the highest BCRs can result in excluding resources that are still cost-effective and would contribute to greater net benefits. This is sometimes referred to as "cream-skimming."

4. Relationship to Traditional Tests

This chapter provides examples of the RVT for a hypothetical set of jurisdictions, emphasizes the variable nature of the RVT, and discusses its relationship with the cost-effectiveness tests that have traditionally been most commonly used (the UCT, TRC and SCT).

4.1 Summary of Key Points

- Because the RVT is based on each jurisdiction's policy objectives, and those objectives can vary across jurisdictions, it can—indeed, it should—take a variety of different forms across different jurisdictions.
- Among the forms the RVT can potentially take are the conceptual forms of the three traditional used tests: the UCT, TRC test, or the SCT. The RVT will align with one of those tests only if the jurisdiction's policy objectives are (1) limited to just minimizing utility system costs (UCT); (2) concerned with minimizing the combination of utility system costs, other fuel costs, and efficiency program participant costs—but with no other impacts (TRC); or (3) concerned with all potential societal impacts (SCT).
- However, in most jurisdictions, the mix of relevant policy objectives will lead to an RVT that is different in at least some respects from the conceptual construct of each of the traditional tests.
- Many jurisdictions that have been nominally using one of the traditional tests have actually modified the tests—adding or subtracting categories of impacts—to the point where they are fundamentally different from the conceptual construct of such tests. In effect, those jurisdictions have attempted to do what the Resource Value Framework is designed to do: develop a test that aligns with their policy objectives. However, because such efforts are not always as systematic, transparent, or grounded in key principles of cost-effectiveness as they could be, the resulting tests can be less effective in addressing jurisdictional policy objectives than if an RVT was developed using the framework put forward in this manual.

4.2 Resource Value Test Examples

As explained in Chapters 1–3, using the Framework process leads a jurisdiction to develop a primary RVT that is specific to each jurisdiction, based on its applicable policy objectives. Thus, RVTs can and should take a variety of different forms across different

jurisdictions. Among the forms an RVT could potentially take are the conceptual forms of the traditional tests: the UCT, the TRC test, and the SCT.

Alternatively, a jurisdiction's RVT can take and probably often will take—a form that is different from the conceptual construct of the traditional tests. The extent to which a jurisdiction's RVT diverges from or aligns Alternatively, a jurisdiction's RVT can take—and probably often will take—a form that is different from the conceptual construct of the traditional tests.

with the traditional tests will be a function of the jurisdiction's relevant policy objectives.

This is shown for six hypothetical jurisdictions described in the bullets below and summarized in Table 14. For illustrative purposes, the six jurisdictions are split into two groups. First, in hypothetical jurisdictions 1 through 3, the application of the Resource Value Framework leads to development of an RVT that differs from the traditional cost-effectiveness tests. Second, in hypothetical jurisdictions 4 through 6, the application of Framework leads the jurisdiction to the development of an RVT or primary test where the impacts included are consistent with what should be included in the traditional cost-effectiveness tests, in their conceptual form.

	Jurisdiction					
Impacts	1	2	3	4	5	6
	RVTs Differ	from Any Trac	litional Test	RVT = UCT	RVT = TRC	RVT = SCT
Utility System	✓	✓	✓	✓	✓	✓
Other Fuels	\checkmark	✓	✓		✓	✓
Water	✓		✓		✓	✓
Participants			✓		✓	✓
Low-Income Participants		1	~		√	✓
Low-Income Societal		~				✓
Environmental	✓		✓			✓
Public Health		✓				✓
Economic Development		~				✓
Energy Security		\checkmark				\checkmark

Table 14. Mix of Policy Objectives Leading to Different Jurisdictional RVTs

- Jurisdiction #1 is interested in not just minimizing utility system costs, but also with minimizing total energy costs (i.e., across all fuels), minimizing water costs, and minimizing environmental costs. Because it is concerned with more than utility system costs, its RVT is not the same as the UCT. Because it is not concerned with participant costs but is concerned with environmental costs, its RVT is not the same as the UCT. Because it is not concerned with participant costs but is concerned with environmental costs, its RVT is not the same as the TRC. And because it is not concerned with either participant costs or a range of other impacts (other than the environment), its RVT is not the same as the SCT.
- Jurisdiction #2 represents a jurisdiction that is interested in utility system impacts, other fuel impacts, low-income impacts, public health impacts, economic development impacts, and energy security impacts. Again, that mix of concerns is not the same as the mix represented by either the UCT, TRC, or SCT.
- Jurisdiction #3 is interested in utility system, other fuel, water, participant, lowincome participant, and environmental impacts. That mix of concerns is clearly much more than those captured by the UCT or TRC and less than those captured by a strict application of the SCT. In short, it is somewhere "between" the TRC and SCT.
- **Jurisdiction #4** determines that its only policy interest related to efficiency investments is in minimizing costs to the funding utility system, producing an RVT that is conceptually identical to the UCT.

- **Jurisdiction #5** determines that its policy interests are limited to impacts on the utility system plus impacts on other fuels, water, and EE program participants (low-income and non-low-income). Therefore, its RVT is conceptually consistent with the TRC.²⁴
- **Jurisdiction #6** determines that its policy interest extends to all utility, other fuel, water, participant, low-income, environmental, public health, economic development, energy security, and any another relevant non-utility impacts, producing an RVT that is conceptually identical to the SCT.²⁵

These six scenarios are also illustrated graphically in Figure 4. The graphics for Jurisdictions 1, 2 and 3 show that the applicable policies for these jurisdictions would lead these jurisdictions to an RVT that differs from any one of the traditional cost-effectiveness tests. While for Jurisdictions 4, 5 and 6, the applicable policies would lead these jurisdictions to developing a primary test that aligns with the traditional UCT, TRC, and SCT, respectively.





- ²⁴ The phrase "conceptually consistent with the TRC" is used because the concept underlying the TRC is consideration of utility system plus participant impacts. As discussed further in Appendix A, the application of the TRC in most jurisdictions has historically often not been consistent with that concept because most jurisdictions that use the TRC include all participant costs but only a portion of or even no participant non-energy benefits, violating the symmetry principle described in Chapter 1 of this manual.
- ²⁵ The phrase "conceptually identical to the SCT" is used because the concept underlying the SCT is consideration of all utility, other resource, participant, and societal impacts. As discussed further in Appendix A, the application of the SCT in most jurisdictions is not consistent with that concept because most jurisdictions that use the SCT (1) include all participant costs but only a portion of or even no participant non-energy benefits and (2) do not fully account for all societal impacts.
Note: The size of the "pie pieces" in these graphs is not intended to convey any sense of relative magnitude or importance of the different categories of benefits.

4.3 Conceptual Differences between the RVT and Traditional Tests

Conceptually, each of the three traditional tests represents a different perspective on cost-effectiveness: the perspective of the utility system (UCT), the combined perspective of the utility system plus efficiency program participants (TRC), and the societal perspective (SCT). Thus, each addresses a fundamentally different cost-effectiveness question and includes a different set of costs and benefits. A more detailed discussion of these tests is included in Appendix A.

The new test put forward in this manual—the RVT—represents a different perspective: minimizing costs in the context of a jurisdiction's applicable policy goals. As Table 15 shows, analysis from that perspective answers a conceptually different cost-effectiveness question than any of the three questions answered by the traditional tests: will utility system costs be reduced while achieving relevant policy goals? As discussed in Section 4.2, depending on the energy policies of a jurisdiction, that may or may not lead to inclusion of different categories of impacts (costs and benefits) in the test. The conceptual differences between the RVT and the three traditional tests are summarized in Table 15.

Test	Perspective	Key Question Answered	Categories of Costs and Benefits Included
Utility Cost Test	The utility system	Will utility system costs be reduced?	Includes the costs and benefits experienced by the utility system
Total Resource Cost Test	The utility system plus participating customers	Will utility system costs plus program participants' costs be reduced?	Includes the costs and benefits experienced by the utility system, plus costs and benefits to program participants
Societal Cost	Society as a whole	Will total costs to society be reduced?	Includes the costs and benefits experienced by society as a whole
Resource Value Test	Regulators or decision-makers	Will utility system costs be reduced, while achieving applicable policy goals?	Includes the utility system costs and benefits, plus those costs and benefits associated with achieving energy policy goals

Table 15. Comparing the RVT and the Traditional Tests

In those cases where a jurisdiction's policy goals align with one of the other tests, the RVT will be the same as that other test.

Importantly, the RVT is conceptually dynamic rather than static, i.e., it can include different types of impacts in different jurisdictions because policy objectives can vary across jurisdictions. And within any given jurisdiction, the components of the RVT can evolve over time as policies change. In contrast, the categories of impacts included in the traditional tests—UCT, TRC, and SCT—are conceptually fixed. They would not change (either across jurisdictions or over time) if the tests were applied in their purest conceptual form (as shown in Figure 5 for example.)

That said, in reality many jurisdictions have used and/or are currently using tests that go by the name of one of the traditional tests, but are fundamentally different from the conceptual construct of those tests. Examples include:

- States that nominally use the TRC, but exclude other fuel impacts and/or exclude participant nonenergy benefits even though such impacts would need to be included to represent the conceptual construct of the TRC—i.e., cost-effectiveness from the combined perspective of the utility system and efficiency program participants;
- States that nominally use the TRC, but include environmental or other impacts that are beyond the conceptual scope of the TRC; and

The Regulatory

Perspective flows from the notion that it cannot be determined whether a resource has benefits that exceed its costs without first being clear about what goals the resource investment decisions should accomplish.

• States that nominally use the SCT, but do not include any societal impacts other than environmental impacts—i.e., falling short of a true societal perspective.

In effect, some jurisdictions appear to have been doing or trying to do what the RVT is explicitly designed to do: developing a test that aligns with their policy objectives. However, rather than systematically building such a test from the ground up using the Framework described in this manual, decision-makers started with one of the traditional tests and then added categories of impacts that were construed to be important to add, and/or subtracted categories of impacts that were not considered important enough to include. Such a process could potentially lead to the very same test that the application of the Resource Value Framework would produce.²⁶

However, such a piecemeal approach suffers from several drawbacks. First, the process

is not likely to be initially grounded in the key principles of cost-effectiveness analysis enunciated in this manual. Second, it begins with a traditional test, which may not be the best starting point and whose economic implications may not be fully understood. Third, the consideration of policy objectives may not be systematic or sufficiently thorough. As a result, such a process can lead to a test that does not fully align with the jurisdiction's policy objectives or other cost-

In reality, many jurisdictions have used and/or are currently using tests that go by the name of one of the traditional tests, but are fundamentally different from the conceptual construct of those tests.

effectiveness fundamentals. Finally, the process for arriving at the test may not be transparent enough to enable an adequate level of understanding and informed input by stakeholders. For these multiple reasons, the use of the Framework to develop a jurisdiction-specific cost-effectiveness test is the recommended approach.

²⁶ The California Public Utility Commission Staff recently proposed a new cost-effectiveness test for DERs that is generally consistent with an RVT (California Public Utility Commission Staff 2017). The Staff proposes to use a test that includes utility system impacts, participant impacts, and specific environmental impacts. California has been using the TRC test for many years, and the environmental impacts were added based on legislative directives. While the Staff proposal refers to its new test as an SCT, it does not include all societal impacts. Rather, the California test accounts for the state's applicable policies — and thus is consistent with an RVT.

5. Secondary Cost-Effectiveness Tests

This chapter provides information about the potential role of secondary tests, their benefits and limits, and selecting and constructing such tests.

5.1 Summary of Key Points

- The purpose of the primary RVT is to address the threshold question of whether a resource has benefits that exceed its costs and therefore merits acquisition.
- Secondary tests can help address other important questions such as how much utility customers should be expected to pay for a resource that is cost-effective under the RVT, which programs to prioritize if it is not possible to pursue all cost-effective efficiency and/or if there should be constraints on key program design features (e.g., financial incentive levels).
- Secondary tests can also help clarify sensitivities to and/or inform decisions regarding which categories of impacts to include in the RVT.
- There is a wide range of potential secondary tests to consider. Decisions on which secondary tests to use should be a function of the primary purpose(s) for using them and the policy priorities of the jurisdiction.

5.2 Potential Reasons for Using Multiple Tests

As covered in Chapter 3, the RVT is designed to answer for jurisdictions the most fundamental question in assessing efficiency resources: *what is the universe of resources whose benefits exceed their costs and therefore merit acquisition (in lieu of acquiring other supply or demand-side resources)?* However, there can also be value in assessing cost-effectiveness of efficiency resources from other perspectives represented by other tests. Among the potential purposes of using additional tests are:

- To inform decisions regarding which categories of impacts to include in the primary RVT. In many cases, the decision as to whether a jurisdiction's applicable policies would support inclusion of a category of impacts in the RVT will be very clear. However, in some cases it may not be quite so obvious or straight-forward. In those cases, there may be value to assessing efficiency resources through two or more potential variations of the RVT to fully understand the sensitivity of results to and therefore the implications of the inclusion or exclusion of one or more categories of impacts in the primary RVT.
- To inform decisions regarding how much utility customer money could or should be invested to acquire cost-effective savings. As noted above, the RVT is designed to answer the threshold cost-effectiveness question of which efficiency resources have benefits that exceed costs and therefore merit acquisition. Depending on the policies of a jurisdiction, it may or may not necessarily answer (or fully answer) questions of how those resources should be acquired or who should pay for their acquisition, including how much the utility system (i.e., utility customers through their utility bills) should be prepared to pay to acquire them. Secondary cost-effectiveness test results can be used to help inform answers to such questions.

- To inform decisions regarding which efficiency programs to prioritize if not all cost-effective resources will be acquired. As noted above, the RVT is designed to answer the threshold question of which resources are cost-effective. In a policy environment in which all cost-effective resources must be acquired, the RVT may be all that is needed to inform decisions on which efficiency programs to fund. However, jurisdictions that do not attempt to acquire all cost-effective efficiency—for example because of statutorily-set funding constraints— may need to make choices *between* cost-effective resources to decide which cost-effective efficiency programs to fund. Jurisdictions may choose to prioritize programs based on RVT net benefits (i.e., which programs have the greatest economic net benefits under their primary test). Alternatively, they may decide to also consider the results of other cost-effectiveness tests to inform such decisions.
- **To inform efficiency program design**. Related to the two points above, there can be important efficiency program design implications associated with decisions to limit how much utility customers should pay for efficiency resources. If secondary cost-effectiveness tests are used to inform decisions on utility customer spending limits, they can also be used to inform related program design decisions (e.g., rebate levels for efficiency measures).
- To inform public debate regarding efficiency resource acquisition. Decisions on which categories of impacts to include in a jurisdiction's RVT may be controversial. Thus, by looking at cost-effectiveness through different perspectives that may be favored by different stakeholders, analysis with multiple tests can provide information useful to ongoing dialogue regarding the merits of different levels or types of efficiency resource acquisition.

5.3 Secondary Tests to Consider

There is a wide range of options jurisdictions can consider for secondary tests. At one end of the spectrum is the UCT, which includes only benefits and costs to the utility system funding efficiency resource acquisition. At the other end of the spectrum is the SCT, which includes the full universe of impacts resulting from efficiency resource acquisition. There are numerous additional options in between. Decisions on which of these options to use as secondary tests should be driven by the primary purpose(s) of the secondary analyses.

5.3.1 Understanding Implications of Impacts Included in the RVT

One appropriate purpose of using multiple tests would be to understand the implications of including or excluding certain categories of impacts in a jurisdiction's RVT (primary test.) In particular, this would allow for the examination of categories of impacts about which there may have been some uncertainty, or even controversy, regarding their inclusion (or exclusion) in the RVT. For example, if there was some uncertainty regarding whether either participant impacts or public health impacts should be included in the RVT, with the ultimate decision being to include both, it may be useful to supplement RVT cost-effectiveness analysis with three sensitivity analyses: (1) removing participant impacts from the RVT; (2) removing public health impacts from the RVT; and (3) removing both participant and public health impacts from the RVT.

5.3.2 Informing Efficiency Program Selection, Spending, and/or Design Decisions

Another purpose of secondary tests could be to inform decisions regarding how much utility customers should pay for efficiency resources, which would have implications for which programs should be prioritized over others and/or program design (particularly participant rebates or other forms of financial incentives). In such a case, the secondary test or tests should be those that best represent the perspective of regulators or other decision-makers regarding such decisions.

For example, if the jurisdiction decides that utility customers (i.e., the utility system) should not pay more for an efficiency resource than they receive back in benefits (i.e., reduced utility system costs), then the UCT would be the secondary test to use. Consider, for example, the following hypothetical scenario for a jurisdiction using the UCT for this purpose:

- a jurisdiction whose RVT included utility system impacts, other fuel impacts, participant impacts, low-income impacts, and environmental impacts;
- a non-low-income efficiency program which provides rebates for efficiency measures equal to 80 percent of the measure costs and has administration, marketing, and other non-incentive costs equal to 20 percent of the total program budget; and
- as illustrated in Table 16, an RVT benefit-cost ratio of 1.67, but with only 40
 percent of the benefits being utility system benefits and the other 60 percent
 being other fuel, participant, and environmental benefits such that the UCT
 benefit-cost ratio is 0.80.

In this example, the RVT suggests that the efficiency program is cost-effective so that the efficiency resource merits acquisition. However, because the jurisdiction does not want utility customers to pay more for efficiency resources than the value to the utility system (i.e., it does not want utility customers to be paying for other fuel savings, improved participant comfort, or other non-utility benefits), it may choose not to run the program—or at least not run it as initially designed. Another option would be to reduce the rebate level enough so that the utility program does pass the UCT—in this case to something less than 60 percent of the measure cost.

Table 16. Using Secondary Test to Address Program Selection orDesign Questions

	RVT			UCT		
Impact	Question: Is resource worth acquiring?		Question: How much is it appropriate for utility customers to pay for it?			
Category	Benefits	Costs	Net Bens	Benefits	Costs	Net Bens
Utility System	\$8	Rebate: \$8	-\$2	\$8	Rebate: \$8	-\$2
		Admin: \$2			Admin: \$2	
		Total: \$10			Total: \$10	
Participant	\$7	\$2	\$5			
Low Income	\$0	\$0	\$0			
Other Fuels	\$3		\$3			
Environmental	\$2	\$0	\$2			
Total	\$20	\$12	\$8	\$8	\$10	-\$2
Ben-Cost Ratio			1.67 to 1			0.80 to 1

Alternatively, the policy framework for a jurisdiction may allow a determination that it is acceptable for utility customers to pay for certain types of non-utility benefits. For example, regulators may decide, based on a jurisdiction's existing policies, that they are willing to allow utility customers to pay for benefits from saving other fuels and benefits to low-income customers, but not non-low income participants' benefits, environmental benefits, public health benefits, etc. In this example, the secondary test of interest would be a test that includes utility system impacts, other fuel impacts, and low-income impacts. Under that secondary test, the program in the hypothetical example described above would pass cost-effectiveness screening because the sum of the utility system benefits, other fuel benefits, and low-income benefits (i.e., \$11 in aggregate) would exceed the program cost (\$10).

5.3.3 Informing Public Debate

If secondary tests are to be conducted to inform public debate, it may make sense to consider a range of secondary tests. This range could include both ends of the cost-effectiveness perspective continuum—the UCT and the SCT—as well as any others that represent perspectives that are held by important stakeholders within the jurisdiction. This process could be useful for assisting in the development of the ultimate primary RVT for a jurisdiction.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001A Page 66 of 146

PART II. Developing Inputs for Cost-Effectiveness Tests

6. Energy Efficiency Costs and Benefits

This chapter describes the range of EE costs and benefits (i.e., impacts), both utility system and non-utility system impacts, and information for selecting cost and benefits to include in cost-effectiveness assessments.

6.1 Summary of Efficiency Resource Impacts

In Part I of this NSPM, Chapter 3 set forth the key Framework Steps 2–3 to consider both utility-system and non-utility system impacts. These steps relate to the underlying principles that (a) a jurisdiction's energy and other relevant policies are central to the decision of which impacts to apply, (b) utility system impacts are the foundation of any cost-effectiveness test, and (c) every cost should be matched with its associated benefit, and vice versa, to ensure symmetry.

This chapter builds on Chapter 3 by providing more detail on the wide range of EE costs and benefits that could be considered in cost-effectiveness testing. Information on the range of impacts includes a description of the cost, benefit, and/or net impact, along with any necessary context or key considerations. Where helpful, additional resources are provided for even further guidance.

Examples of different types of EE resource impacts are summarized in Table 17.

	Type of Impact	Description	
Utility System	Costs incurred or saved by the utility that funds the efficiency resource	Includes costs to utility of acquiring efficiency resources. Savings can include reductions in costs to the utility system associated with both avoided capital investments (e.g., for new generating facilities, environmental compliance and T&D) and avoided variable operating costs (e.g., energy/fuel costs).	
Non- Utility System	Participant measure costs	Participant measure costs accrue when the financial incentives provided by efficiency programs cover only a portion of the cost of an efficiency measure. Program participants bear the balance of the measure cost.	
	Participant non- resource impacts	Impacts on program participants that are not related to resource (fuel or water) savings. Including asset value, productivity, economic well-being, comfort, health and safety, and customer satisfaction.	
	Incremental low-income participant impacts	Impacts on low-income program participants that are different from or incremental to non-low-income participant impacts. Includes reduced foreclosures, reduced transiency, and povert alleviation.	
	Other fuel impacts	Impacts on end-use fuels that are not provided by the funding utility, for example, electricity (for a gas utility), gas (for an electric utility), oil, propane, and wood.	
	Water impacts	Impacts on participant water consumption and related wastewater treatment.	
	Environment	Impacts associated with CO ₂ emissions, criteria pollutant emissions, land use, etc. Includes only those impacts that are n included in the utility cost of compliance with environmental regulations.	
	Public health	Impacts on public health. Includes health impacts that do not overlap with participant impacts or environmental impacts, and includes benefits in terms of reduced health care costs.	
	Economic development and jobs	Impacts on economic development and jobs.	
	Energy security	Reduced reliance on fuel imports from outside the state, region, or country.	

Table 17. Summary of Efficiency Resource Impacts

This table is presented for illustrative purposes and is not meant to be an exhaustive list. The non-utility impacts presented here can be either a cost or a benefit, or can have a net impact that accounts for both costs and benefits. For a comprehensive discussion of EE resource impacts, see Regulatory Assistance Project 2013c.

The balance of this chapter provides additional detail on the impacts referenced in Table 17. Appendix B provides more information about how the costs and benefits relate to other DERs.

6.2 Utility System Impacts

There are a variety of relevant utility system costs and benefits which should be included in any primary cost-effectiveness test.

6.2.1 Utility System Costs

EE Measure Costs

The utility portion of measure costs can take a variety of forms. Among the most common are rebates provided to program participants, whether end-use customers or other market actors such as retailers, contractors, distributors, and manufacturers. Also common are buy-downs of interest rates for financing investments in efficiency measures.

Other Efficiency Financial Incentives

Other incentives include payments to support trade ally reporting on sales of efficient products, and/or funding or co-funding of marketing of efficient products by trade allies. "Spiffs" are another common incentive. These are sales bonuses provided to retail or contractor sales staff for selling efficient products.

Other Efficiency Program and Administrative Costs

These additional costs support utility outreach to trade allies, technical training, other forms of technical support, marketing, and administration and management of efficiency programs and/or portfolios of programs.

Evaluation, Measurement, and Verification (EM&V)

EM&V costs entail either the analysis of markets for efficiency products and services to inform the design of efficiency programs or the retrospective assessment of the effectiveness of efficiency programs.

Performance Incentives

In regulated utility systems, utilities often receive payments for meeting specific performance metrics related to the success of efficiency programs.

6.2.2 Utility System Benefits

Avoided Energy Costs

These are the values of avoiding the generation or the purchase of electric energy (i.e., kilowatt-hours, or kWh)²⁷ and/or natural gas resulting from investments in efficiency. The marginal cost of avoided energy can vary considerably by both season and time of day. The load shapes of different efficiency resources—i.e., the portion of energy savings that occur during different seasons and different times of day—can also vary substantially. The value of avoided energy costs should account for such differences to the extent possible and practical.

²⁷ Typically valued at either forecast wholesale market prices in jurisdictions with competitive wholesale markets or forecast marginal costs of generation for jurisdictions that regulate vertically integrated utilities.

Avoided Generating Capacity Costs

Some portion of the savings of efficiency resources will occur at times that are coincident with system peak demands. Thus, efficiency resources will reduce the amount of money that must be invested in electric generating capacity.²⁸ The magnitude and type of that reduction will vary considerably from measure to measure, depending on the portion of energy savings that occur during times of system peak demand. Over the long term, efficiency programs can also defer or avoid the need for construction of baseload generation.

Avoided Reserves

Electric utilities and/or electric system operators always plan to have at their disposal reserve capacity that can be deployed when a generator shuts down or there is some other form of disruption to the supply of generating capacity. Typically ranging from 7 percent to 25 percent, reserve requirements vary depending on the size of the system and its principle sources of generating capacity (Regulatory Assistance Project 2011). When efficiency resources reduce the amount of generating capacity required for a system, they can also reduce the

Understanding T&D Line Losses

When estimating the magnitude of avoided line losses, it is important to recognize that line losses grow exponentially with load. As a result, the marginal loss rate associated with the last increment of load added to—or removed from—the T&D system (i.e. incremental losses divided by incremental load) is greater than the average loss rate for all load (i.e. total losses divided by total load). Thus, the magnitude of line loss reductions associated with efficiency savings should be based on estimates of marginal not average—line loss rates (Regulatory Assistance Project 2011).

Further, there should be separate average marginal line loss rates for energy savings and peak savings. By definition, marginal line loss rates at the time of peak will be considerably higher than the weighted average of marginal line loss rates across all hours of the year when energy is saved. Two studies suggest that weighted average marginal loss rates over the course of a year are typically on the order of about 150 percent of average annual loss rates and that marginal loss rates at the hours of system peak (i.e. related to avoided generating capacity) might be twice as great, or on the order of 300 percent of average annual loss rates (Regulatory Assistance Project 2011). (Illinois Commerce Commission 2014).

amount of reserves needed. The value of avoided reserves should either be included in estimates of avoided capacity costs or included separately.

Avoided T&D Costs

Efficiency resources reduce loads on the T&D system. To the extent that at least some portion of those load reductions occur during T&D peaks, they can defer or eliminate the need for investments that would otherwise be required to address localized T&D capacity constraints.

Such deferrals can be passive, meaning they result from system-wide efficiency programs implemented for broad-based economic or other reasons not related to the

²⁸ There are some exceptions. For example, some heating efficiency measures installed in electric service territories that are summer peaking (and vice versa) will not avoid generating capacity costs. Alternatively, jurisdictions that are forecast to have excess generating capacity well into the future—i.e. beyond the life of the efficiency savings being analyzed—may have no avoidable capacity costs.

intent to defer specific T&D projects. In such cases, the value of avoided T&D costs in some parts of the system are spread across total system T&D peak savings.²⁹

They can also be active, such as when geographically targeted efficiency investments are intentionally designed to defer specific T&D projects. The value of active deferrals per peak kW saved will typically be considerably higher than the value per kW for passive deferrals.

In recent years, there has been an increased interest in the value of avoiding distribution costs with DERs. The value of avoided distribution costs can vary significantly depending upon the specific location on the electricity grid. As EE resources become increasingly used, along with other types of DERs, to avoid distribution costs it will be important to develop more sophisticated estimates of the locational values of avoided distribution costs (Analysis Group 2016; ICF International 2016; SEPA 2016; National Grid 2015).

Avoided T&D Line Losses

A portion of all electricity produced at electric generating facilities is lost as it travels from the generating facilities to the homes and businesses that ultimately use the power.³⁰ Thus, every kWh of efficiency savings realized at the customer's side of the meter equates to more than one kWh of savings at the electric generator. Similarly, every peak kW of savings by end-use customers equates to more than one peak kW of generating capacity. Another key characteristic of line losses is that they expand exponentially as the system experiences higher volumes. For this reason, it is important that calculations account for marginal loss rates for energy savings and peak savings.

Avoided Ancillary Services

Ancillary services are those services required to maintain electric grid stability and security. They include frequency regulation, voltage regulation, spinning reserves, and operating reserves. Efficiency resources may reduce the need for these services by reducing loads on the T&D system. To the extent that these reduced loads lead to lower ancillary services costs, those avoided costs should be included as a benefit.

Energy and/or Capacity Price Suppression Effects

In jurisdictions with competitive wholesale energy and/or capacity markets, prices will be a function primarily of the magnitude of demand. Thus, increased investment in efficiency resources is likely to benefit all consumers through reduced market clearing prices (at least to some extent and for some period of time).

It should be noted that price suppression effects from efficiency resources acquired in a given utility service territory will typically extend beyond the borders of that service territory. This is due to the regional nature of most wholesale markets, which tend to

²⁹ Estimates of avoided T&D costs can be very utility-specific. For example, 2015 values for New England electric utilities varied between \$33/kW-year for Connecticut Light and Power to \$200/kW-year for National Grid Rhode Island, with the unweighted average of reported values being \$113/kW-year (AESC Study Group 2015). Another benchmarking study found that the avoided distribution cost assumptions across 25 utilities ranged from \$0 to \$171/kW-year, with an average of just over \$48; it also found average avoided transmission cost assumptions to range from \$0 to \$89/kW-year, with an average of about \$20 (Mendota Group 2014).

³⁰ There are analogous "pipe losses" on gas T&D systems, though they tend to be much smaller in magnitude (in percentage terms) than electric losses.

encompass multiple utility service territories. Thus, regulators that include price suppression effects in cost-effectiveness analyses also need to decide whether to include only the value of price reductions to customers in the utility service territory in question, in the entire jurisdiction under the regulator's purview, or in the entire region.

Another consideration is the ongoing debate regarding whether price suppression effects should be considered a benefit or whether there is no net benefit because consumer price decreases are counter-balanced by reductions in generators' profits. This is particularly relevant in jurisdictions that adopt a broader, more "societal" view of impacts on cost-effectiveness analyses.

Avoided Costs of Compliance with RPS Requirements

In jurisdictions that have adopted a Renewable Portfolio Standard (RPS) expressed as a percentage of electric generation, new efficiency resources will by definition reduce the absolute amount of renewable resources that must be purchased. When those required renewable resources are forecast to cost more than other sources of electric generation, their avoided purchase represents avoided RPS compliance costs. Thus the efficiency resources provide an additional utility system benefit, provided the avoided costs are not already reflected in the avoided energy, capacity, and T&D costs discussed above.

Avoided Environmental Compliance Costs

By reducing the amount of electricity that needs to be generated, efficiency resources can lower future costs of complying with environmental regulations. In estimating the value of such savings, it is important to account both for all regulations that have already been promulgated and those that have a significant probability of being promulgated in the future (Regulatory Assistance Project 2012).

Avoided Credit and Collection Costs

All utilities incur some costs associated with customers who are not keeping up with their energy bill payments. Those costs can take a variety of forms, including costs of notices and support provided to customers in arrears, costs associated with shutting off service and turning it back on, carrying costs associated with arrears, and costs of writing off bad debt.

Because efficiency programs lower customers' energy use and energy bills, they can reduce the probability of customers falling behind or defaulting on bill payment obligations. That can be a particularly important benefit of efficiency programs targeted to low-income customers. Since these benefits are costs avoided by the utility and they accrue directly to all utility customers, they are classified here as a utility system benefit.

Reduced Risk

Efficiency resources can reduce utility system risk in several ways. Key among them are: creating a more diverse portfolio of resources that can meet customers' energy needs (all other things being equal, diversity reduces risk); reducing uncertainty in forecasts of future loads and related capital investment needs; and reducing exposure to potential future fuel price volatility associated with other resource types (particularly natural gas, oil, and/or coal-fired generation) (Ceres 2012). Also, as a resource that can be implemented in many relatively small increments, efficiency resources provide more optionality than large central generation facilities.

There are different ways to value risk reduction. For example, the most recent New England regional avoided cost study estimated a "risk premium" of nine percent. This

was added to avoided energy costs to account for one aspect of efficiency's risk mitigating effects: uncertainty in the range of future wholesale energy prices (AESC Study Group 2015). Similarly, another screening tool approach is to report cost-effectiveness for several scenarios; e.g., a "best estimate" of future avoided costs, versus a probability-weighted average of future avoided costs.³¹ The difference between the two essentially represents a "risk premium" associated with future price volatility. Alternatively, Vermont's regulators have mandated since 1992 that efficiency resource costs be reduced by 10 percent to reflect efficiency's "comparative risk and flexibility advantages" relative to supply resources (VT PSB 1990).

Increased Reliability

By lowering loads on the grid, efficiency can reduce the probability and/or likely duration of customer service interruptions. The magnitude of the value of this benefit will vary, with less value to systems that are projected to be in a good state of reliability for years into the future and more value to systems that are not. There could be some overlap between this benefit and the benefits of reduced risk, avoided capacity costs and/or avoided T&D costs. Thus, any assessment of the value of increased reliability would need to ensure that there is no "double-counting" of overlap with such other benefits.

6.3 Non-Utility System Impacts

This section describes the different types of non-utility system impacts. Many of these impacts can be experienced in the form of costs or benefits, or both. For example, some efficiency measures might increase or decrease the use of other fuels. For each type of impact included in a cost-effectiveness test, both costs and benefits should be included in order to be consistent with the *Principle of Symmetry*.

6.3.1 Participant Impacts

Efficiency program participants typically incur costs and realize benefits beyond those associated with utility system impacts. A more detailed discussion of these costs and benefits is provided below.

Efficiency Measure Costs

Participant measure costs accrue when the financial incentives provided by efficiency programs cover only a portion of the cost of an efficiency measure. Program participants bear the balance of the measure cost. Participant measure costs should include only the participant's portion of the incremental measure costs, i.e., the extent to which the EE measure cost exceeds the baseline measure cost.

Participant Non-Resource Costs and Benefits

Non-resource participant costs and benefits can be divided into residential and business impacts. Residential efficiency measures can provide a wide variety of other non-resource benefits to customers. Some notable examples include improved comfort such as from sealing and insulating leaky homes, improved building durability such as

³¹ One tool for example is Integral Analytics' DSMore cost-effectiveness screening tool. Other approaches include Value-at-Risk, a common approach used to examine risk in probabilistic scenario analyses.

eliminating creation of "ice dams" through sealing and insulating attics, improved health and safety (E4TheFuture 2016a), and improved aesthetics.

For businesses, non-resource benefits can come in a variety of forms, but are commonly distilled down to improved productivity (ACEEE 2015). Such benefits can apply to many types of commercial and industrial customers, including private business, schools, hospitals, government agencies, and more.

Table 18 provides a summary of the different types of participant non-resource benefits.

Category	Examples
Accetivatio	 Equipment functionality/performance improvement Equipment life extension
Asset value	Increased building value Increased eace of colling building
	Increased ease of setting building Reduced labor costs
	Improved labor productivity
Productivity	Reduced waste streams
	Reduced spoilage/defects
	Impact of improved aesthetics, comfort, etc. on product sales
	Fewer bill-related calls to utility
	 Fewer utility intrusions & related transactions costs (e.g., shut- offs, reconnects)
Economic well-being	Reduced foreclosures
	• Fewer moves
	Sense of greater "control" over economic situation
	Other manifestations of improved economic stability Thermal comfort
Comfort	Noise reduction
Connort	Improved light quality
	Improved "well-being" due to reduced incidence of illness—
	hyperthermia)
	 Reduced medical costs (emergency room visits, drug
Health & safety	prescriptions)
	Fewer sick days (work and school)
	Reduced deaths
	Reduced insurance costs (e.g., for reduced fire, other risks)
Satisfaction/pride	Improved sense of self-sufficiency
Calisiaciion/pride	 Contribution to addressing environmental/other societal concerns

Table 18. Participant Non-Resource Benefits³²

In some cases, participating customers might experience non-resource costs. For example, some EE measures might increase labor costs or result in increased noise.

Low-Income Participant Costs and Benefits

Low-income participants can incur the same types of costs as non-low-income participants. However, in recognition of the reality that low-income consumers usually cannot afford to pay even a fraction of the cost of efficiency measures, their portion of

³²See Synapse 2014 and Skumatz 2014 for more detail.

measure costs are often lower by design than the portion borne by non-low-income customers.

Low-income benefits can come in two forms:

- Benefits include the same *types* of participant benefits as realized by non-lowincome residential participants—O&M savings, other fuel savings, water savings, and non-resource benefits described above—though the *magnitude* of some of these benefits are often greater for low-income customers than for non-lowincome customers. This is because the condition of the low-income housing stock is often worse and/or because the economic stress under which lowincome customers live can result in greater sacrifice of amenity (e.g., comfort) absent efficiency investments.
- 2. Some participant non-resource benefits—particularly those related to economic well-being—are unique, or largely unique, to this subset of residential customers. Examples include reduced home foreclosures and reduced need to move residence as a result of unpaid bills.

The value of low-income benefits can be substantial, potentially greater than the value of utility system and other energy benefits (SERA 2014).

Operation and Maintenance (O&M) Costs and Benefits

Efficiency measures have the potential to either increase or reduce O&M costs for participants. For example, when an efficient heat pump is installed to displace much less

efficient electric resistance heating, there is a modest ongoing annual cost associated with maintaining or servicing the heat pump (compared to no significant maintenance costs for electric resistance baseboard heat). In other cases, efficient technologies provide O&M benefits. Commonly cited examples include efficient lighting technologies such as compact fluorescent lamps (CFLs) and/or Light Emitting Diode (LED) lamps that last longer than their baseline alternatives. They therefore

Literature on Non-Energy Impacts

There is a wealth of literature available on the non-energy impacts of EE resources. The following references may be useful for those seeking further information on this topic: ACEEE 2006, Lawrence Berkeley National Laboratory 2014, International Energy Agency 2014, NMR 2011, Oak Ridge National Laboratory 2014, SERA 2006, SERA 2010, SERA 2014, SERA 2016, Tetratech 2012.

eliminate both the need to purchase and the time and labor required to install several replacement products in the future.

Other Fuels Costs and Benefits

Many efficiency measures reduce consumption of both electricity and non-electric energy sources such as natural gas, fuel oil, propane, and wood. The reduction of these fuels provides a benefit that is outside the utility system. Among the most common examples are: building envelope measures such as insulation and air sealing; HVAC distribution system measures such as duct sealing and insulation; and control measures in buildings that are cooled electrically and heated by gas, oil, or propane. In such cases, there is economic value associated with reductions in fuels not supplied by the funding utility.

Conversely, some electric efficiency measures increase consumption of other fuels. For instance, electric efficiency resources can reduce the "waste heat" from inefficient lighting, refrigeration, or air flow components, thereby increasing the need for other fuels

used for building space heating. In such cases, the economic benefit of electricity efficiency can be offset—at least in part—by the economic cost of increased consumption of other fuels. Similarly, the economic benefit of reduced consumption of one fuel resulting from fuel-switching measures can be offset—at least in part—by the cost of increasing consumption of other fuels.

Water and Wastewater Costs and Benefits

A number of EE measures also reduce water use. Indeed, in many cases, energy is saved precisely because less water is needed. Examples include low-flow devices (e.g., showerheads, faucet aerators, spray-rinse valves for commercial dish-washing, clothes washers, and improved agriculture techniques). In such cases, there can be economic value associated with both reduced water consumption and reduced wastewater treatment.

6.3.2 Societal Impacts

Environmental Impacts

Efficiency resources can provide a wide range of environmental benefits. These can include reductions in air emissions associated with fossil fuel combustion; the disposal costs of waste from various energy sources (nuclear, coal ash, etc.); the amount of water needed for cooling electric generating stations, extracting natural gas (e.g., "fracking") and other purposes; the amount of land that must be cleared and/or developed for new generating facilities; and adverse impacts on land, air, and water from fossil fuel mining or extraction. Examples of negative environmental impacts include additional waste streams and/or emissions from the production, use, and disposal of efficient products.

It is important to avoid overlap between impact categories. Some positive impacts may be accounted for in calculations of utility system costs under the utility cost of compliance with environmental regulations. Similarly, only those negative impacts that are incremental to impacts from standard or inefficient products should be included.

Public Health Impacts

Some of the environmental emission and waste reductions discussed in the point above result in a reduction in the frequency and/or severity of health problems of populations impacted by fuel extraction and combustion. Such reductions can have positive implications for the level of societal investment required in medical facility infrastructure, as well as in the health, well-being, and economic productivity of the populace.

There could be some overlap between public health benefits and either participant benefits or environmental benefits. Thus, any quantification of public health benefits should ensure that any such overlap is not double-counted.

Economic Development and Jobs

Investment in efficiency resources will result in additional jobs and economic development in several ways.

- First, there are jobs associated with managing and delivering the efficiency programs.
- Second, there are jobs and economic development effects associated with additional work and revenue that such programs funnel to the supply chains

associated with efficiency measures being installed in homes and businesses. These supply chains include: contractors, builders/developers, equipment vendors, product retailers, distributors, manufacturers, and other elements.

• Third, to the extent that the efficiency resources are less expensive than the energy they save, consumers will have more disposable income. When that additional disposable income is spent in the local community (or beyond), it helps to create jobs and spurs economic development.

Conversely, by reducing or avoiding supply-side resources, efficiency resources will reduce the number of job and related local economic development benefits of supply-side investments. Jurisdictions that include economic development and/or job impacts in their primary cost-effectiveness test should account for both positive and negative impacts.

Net economic development and/or job gains are often expressed in terms of increased gross domestic product (GDP) or gross state product (GSP) and/or job-years. It is not clear how these metrics can be translated into monetary terms suitable for inclusion in efficiency benefit-cost analyses, particularly since the drivers of these benefits (efficiency program spending and reduced utility system costs) are already included in the cost-effectiveness analyses. At a minimum, such benefits can be considered without using monetary values. This point is discussed in Chapter 7.

Societal Low-Income Impacts

In some cases there may be low-income community or societal impacts that go beyond those realized by program participants. Examples include poverty alleviation, improving low-income community strength and resiliency, and reduced home foreclosures (any societal impacts from reduced foreclosures must be incremental to the participant impacts related to foreclosures).

Energy Security Impacts

Some jurisdictions have policies designed to increase energy independence and/or energy security. EE investments that reduce imports of various forms of energy inherently advance such goals. There could be some overlap between (a) the benefit of improved energy independence and security and (b) either local jobs and economic development or risk reductions. Thus, any assessment of the magnitude or value of improved energy independence would need to ensure that there is no double-counting of overlap with such other benefits.

Other Impacts

There may be other impacts not included in the list above. These would need to be assessed to ensure they do not overlap with the impacts already defined.

Several of the non-utility system impacts described above, notably the impacts on environment, public health, and economic development, will likely accrue within a broader territory. They can accrue: within the utility service territory in which an efficiency program is run; outside of that service territory but within the jurisdiction of regulators overseeing the program (e.g., within a state); and outside of the jurisdiction governed by the regulators. Thus, in jurisdictions for which energy policies dictate that such impacts be considered, regulators will need to consider the geographic boundary of the impacts.

7. Methodologies to Account for Relevant Impacts

This chapter provides guidance on options for accounting for relevant cost and benefits, including hard-to-quantify impacts as well as approaches for qualitatively including non-monetary impacts.

7.1 Summary of Key Points

All impacts of EE resources that a jurisdiction has chosen to assess via its costeffectiveness test should ideally be estimated in monetary terms so that they can be compiled readily and compared directly. However, some EE impacts are difficult to quantify in monetary terms, either due to the nature of the impact or the lack of information available about the impacts.

The third key principle described in Chapter 1 is that cost-effectiveness practices should account for all relevant, important impacts, even those that are difficult to quantify and monetize. Approximating hard-to-monetize or hard-to- quantify impacts is preferable to assuming that substantive costs and benefits do not exist or have no value.

Table 12 from Chapter 3.6 summarizes five different approaches that can be used to account for all impacts of EE resources that a jurisdiction has chosen to include in its cost-effectiveness test. The approaches are listed in order of technical rigor and preference.

Preferably, any impacts included in a cost-effectiveness test would be based on monetary values that are rigorously estimated and transparently documented. The first two subsections below discuss using studies from within or outside of a jurisdiction to develop monetary values. The next three sub-sections discuss approaches for addressing hard-to-monetize impacts.

7.2 Jurisdiction-Specific Studies

Jurisdiction-specific studies that quantify costs and monetize relevant benefits as possible are the most rigorous and reliable way to estimate the benefits of EE programs. These studies should use local information to the greatest extent possible, by utility, by

state, by province, or by the relevant Regional Transmission

Organization/Independent System Operator. These studies should be derived from, or at least be consistent with, the most recent integrated resource planning studies available, wherever they exist.

Jurisdiction-specific avoided cost studies should be comprehensive, transparent, use

Jurisdiction-specific avoided cost studies should be comprehensive, transparent, use best practices, and use all relevant information available at the time.

best practices, and use all relevant information available at the time. These avoided cost studies should be updated periodically, to reflect the most recently available information.

Ideally, these avoided cost studies should be prepared by independent third parties, guided by stakeholders, and ultimately reviewed and approved by regulators. For a good example of this approach, see the New England Avoided Energy Supply Cost studies (AESC Study Group 2015). Another example is the California Public Utility Commission cost-effectiveness calculator that embeds the state's official avoided costs in a model to calculate cost-effectiveness (CPUC 2016).

Many jurisdictions have developed technical reference manuals (TRM) to document the costs and operating characteristics of EE resources. TRMs are critical for jurisdictions to support the cost inputs of a jurisdiction's EE cost-effectiveness tests. TRMs should use information that is as up-to-date as possible, and should account for jurisdiction-specific costs as much as possible (Beitel et al. 2016).

7.3 Studies from Other Jurisdictions

In some cases, for some impacts, a jurisdiction-specific study might not provide all the information needed for a cost-effectiveness test. In these cases, it may be appropriate to use results from other jurisdictions. This could include studies prepared for other utilities, other states, other jurisdictions, other regions. It could also include regional or national studies that do not necessarily focus on any one jurisdiction or region.

However, efficiency planners must take care to ensure that the value of a particular cost or benefit in another jurisdiction is equal to, or sufficiently comparable to, the value in the jurisdiction of interest. If not, it may be necessary to adjust values from other jurisdictions before using them. For example, labor costs in one part of the country might be significantly different from other parts of the country. These differences can be accounted for by adjusting costs accordingly.

7.4 Proxies

For the purpose of EE cost-effectiveness analyses, a proxy is a simple, quantitative value that can be used as a substitute for a value that is not monetized by conventional means. Proxies can be applied to any type of cost or benefit that is hard to monetize and

Proxies should be developed by making informed approximations based upon the best information currently available regarding the relevant impact. is expected to be of significant magnitude (NEEP 2014).

Proxy values are typically based on professional judgment; but they should not be developed or perceived as arbitrary values. Proxies should be developed by making informed approximations based upon the best information currently available

regarding the relevant impact. This should include a review of relevant literature on the specific impact, as much quantification of the impact that is both feasible and reasonable, a review of proxy values used by other jurisdictions, and consideration of conditions specific to the relevant jurisdiction.

To date, proxies have most frequently been used to account for efficiency resource benefits such as low-income benefits, participant non-energy benefits, or risk benefits (NEEP 2014). However, proxies can also be used to account for other hard-to-monetize efficiency costs and benefits. Proxies could be used, for example, to account for the degradation of energy savings over time, i.e., to account for a "rebound" effect where customers increase energy consumption as a result of reduced energy costs.

Level of Application

Proxy values can be developed for different levels of application, ranging from a single proxy value that applies to an entire portfolio of efficiency resources to different proxy values for each efficiency impact.

When choosing the level of detail to apply to a proxy, there may be a tradeoff between accuracy and feasibility. Proxies that are more detailed are likely to more accurately represent the magnitude of the specific impact in question. However, proxies that are more detailed are also likely to require more information and greater costs to develop.

One advantage of more detailed proxies is that they are more transferrable across programs, across utilities, and over time. For example, an impact-level proxy such as improved health and safety, applied to residential retrofit efficiency programs, is likely to be generally applicable to other residential retrofit programs and remain relatively constant over time. Conversely, a sector-level proxy to account for all participant non-energy benefits for the residential sector should, in theory, be different for different programs and could change over time as the mix of efficiency measures changes over time.

Type of Proxy

Several different types of proxies can be used to account for EE program impacts.

- Percentage Adder: A percentage adder approximates the value of nonmonetized impacts by scaling up impacts that are monetized. This type of proxy is the simplest and easiest to apply.
- Electricity Savings Multiplier (\$/MWh): An electricity savings multiplier approximates the value of non-monetized benefits or costs relative to the quantity of electricity saved by an efficiency resource.
- Gas Savings Multiplier (\$/therm): This is the same as an electricity multiplier, but can be applied to programs that primarily, or exclusively, provide gas efficiency improvements. It offers the same advantages and disadvantages of electricity multipliers.
- Fuel Savings Multiplier (\$/MMBtu): A fuel multiplier approximates the value of non-monetized benefits or costs relative to the total quantity of fuel saved by an efficiency resource, regardless of the type of fuel saved (e.g., electricity, gas, oil, propane).
- Customer Adder (\$/customer): A customer adder (or subtraction) approximates the value of non-monetized benefits relative to the number of customers served by an efficiency program.
- Measure Multiplier (\$/measure): A measure multiplier (positive or negative) approximates the value of non-monetized benefits or costs relative to the number of measures installed by an efficiency program.

As with the choice of level of application for a proxy, the choice of which type can result in a tradeoff between accuracy and feasibility. Proxies that are more focused (e.g., by measure, by customer, or by fuel) are more likely to accurately represent the magnitude of the specific impact in question. However, proxies that are more focused are also likely more difficult and expensive to develop.

7.5 Quantitative and Qualitative Information

Some impacts might be difficult to put into monetary terms or to address through proxies. Other impacts may not even be appropriate to put into monetary terms.³³ In these cases, other types of quantitative and qualitative information can be used to inform the cost-effectiveness decision.

Once all efforts to monetize EE costs or benefits have been considered and exhausted, the following steps can be used to consider additional quantitative and qualitative information.

Step A: Provide as much quantitative evidence as possible

For those impacts that remain non-monetized, it may be possible to put them into quantitative terms. Quantitative values generally provide more concrete information for decision-makers to consider, relative to qualitative values or no values at all. Quantitative values of efficiency impacts should be documented in detail, along with justification for why and how the values are relevant to the cost-effectiveness analysis.

For example, jurisdictions that choose to include job impacts might want to present this impact in terms of the number of job-years, rather than a monetized value for jobs. Regulators and efficiency planners could then compare different energy resources according to how many job-years are created by each one.

Step B: Provide as much qualitative evidence as possible

Those impacts that are not monetized or quantified should be addressed qualitatively. Qualitative information can provide some information for decision-makers to consider, relative to no information at all. For those efficiency impacts that are addressed qualitatively, efficiency planners should develop and present as much qualitative evidence as possible regarding those impacts. This evidence should also include a justification for why the considerations are relevant to the cost-effectiveness analysis.

For example, a jurisdiction might choose to consider incremental market transformation benefits without quantifying or monetizing such benefits. In this case, regulators or efficiency planners would consider the incremental market transformation benefits, without necessarily estimating what those benefits are either in terms of energy savings or dollar savings.

Step C: Present quantitative and qualitative evidence alongside monetary results

The monetary impacts of EE resources should be the core of the cost-effectiveness analysis, and ideally should include the vast majority of the impacts being considered. These monetary results should be presented in a transparent, detailed, easily-reviewable way, as described in Section 3.7.

³³ For example, it may not be appropriate to directly compare the monetary values of economic development and job impacts to the other monetary values in a cost-effectiveness analysis. This issue is addressed in Section 6.3.2.

Any non-monetized impacts of efficiency resources should be presented along-side the monetary impacts.³⁴ This allows the regulators and other decision-makers to directly compare the monetized, quantitative, and qualitative factors.

Step D: Decide upon the implications of the guantitative and gualitative evidence

Regulators and other decision-makers should then use the monetary, quantitative, and qualitative evidence to decide whether an efficiency resource is cost-effective. In some cases, the monetary results alone might be sufficient to make this decision, e.g., if the monetary benefits exceed the monetary costs, and all the non-monetary evidence indicates there will be additional benefits. The cost-effectiveness decision might also be easy if the monetary benefits are slightly less than the monetary costs, but the non-monetary benefits are clearly significant enough to make up the difference.

In other cases, the decision might not be so clear. For example, if the monetary benefits do not exceed the costs, but the nonmonetary benefits are not necessarily significant enough to make up the difference. In these cases, regulators and other decision-makers should make a costeffectiveness determination, based on all the evidence presented, and with input from relevant stakeholders.

Step E: Document and justify the decision

Finally, the cost-effectiveness decision should be fully documented and justified. This is necessary to provide transparency regarding the decision for the resource in question, and to provide guidance on how

Example of Using Qualitative Information

The Oregon PUC has two orders (UM551 and UM 590) that set forth a specific set of qualitative conditions under which violation of strict cost-effectiveness limits could be justified to account for non-monetary impacts.

Measures that are not cost effective could be included in utility programs if the following can be demonstrated:

- The measure produces significant nonquantifiable non-energy benefits. In this case, the incentive payment should be set at no greater than the cost-effective limit (defined as present value of avoided costs plus 10 percent) less the perceived value of bill savings, e.g. two years of bill savings.
- 2. Inclusion of the measure will increase market acceptance and is expected to lead to reduced cost of the measure.
- 3. The measure is included for consistency with other DSM programs in the region.
- 4. Inclusion of the measure helps to increase participation in a cost-effective program.
- 5. The package of measures cannot be changed frequently and the measure will be cost effective during the period the program is offered.
- 6. The measure or package of measures is included in a pilot or research project intended to be offered to a limited number of customers.
- 7. The measure is required by law or is consistent with Commission policy and/or direction.

The conditions above apply both to measures and programs with the exception of Item D (OR PUC, 2014).

similar decisions will be made in future cost-effectiveness analyses.

³⁴ Section 3.7 presents an example template for how the monetized, quantified, and qualitative information could be presented.

7.6 Alternative Thresholds

Alternative thresholds are another approach for addressing hard-to-monetize impacts. Such thresholds allow efficiency resources to be considered cost-effective at predetermined benefit-cost ratios that are different from one (1.0). Regulators can apply a benefit-cost ratio of greater than one (1.0) to account for efficiency resource costs that have not been monetized, or a benefit-cost ratio of less than one (1.0) to account for non-monetized benefits. Regulators can apply alternative thresholds to account for hardto-monetize impacts at the program, sector, or portfolio level.

Alternative thresholds are, by design, a simplistic way of recognizing that the hard-tomonetize impacts are significant enough to influence the cost-effectiveness analysis. The primary advantage of this approach is that it does not require the development of specific monetary or proxy values. Instead, it is more of a general reflection of the regulators' willingness to be flexible in accounting for certain impacts.

Note that using alternative benchmarks can essentially have the same effect as applying a proxy value if the proxy is applied at the same level of the cost-effectiveness screening (e.g., measure or portfolio). For example, an alternative portfolio level benefit-cost ratio benchmark of 0.9 is equivalent to a portfolio level benefit multiplier of 11 percent; and an alternative benefit-cost ratio benchmark of 0.8 is equivalent to a benefit multiplier of 25 percent.

Regulators should ensure that alternative thresholds are as transparent as possible and are established prior to the cost-effectiveness analysis. Regulators should articulate which resources the alternative thresholds can be applied to, what the threshold is, and the basis for the threshold chosen.

7.7 Sensitivity Analyses

Sensitivity analyses can be used to test the implications of input assumptions that are hard to monetize or whose monetary values are especially uncertain. The cost-

effectiveness test can be applied with high, medium, and low estimates of certain inputs to see how the range of estimates will affect the results.

Sensitivity analyses of hard-to-monetize inputs offer two advantages. First, they indicate the extent to which these costs or benefits will affect the cost-effectiveness results. Those costs or benefits with a minor Sensitivity analyses can be used to test the implications of input assumptions that are hard to monetize or whose monetary values are especially uncertain.

impact on the results, regardless of whether a high or low value is used may not require much additional attention. Conversely, those with a major impact on the results might warrant additional research and analysis to improve the estimates of their magnitudes.

Second, sensitivity analyses indicate the extent to which the accuracy of the input will affect the cost-effectiveness results. If an efficiency resource is clearly cost-effective, or clearly not cost-effective, regardless of whether the high or low input assumption is used, then there may be little need or value in improving the accuracy of that input. Conversely, if the input has a notable impact on the cost-effectiveness results depending upon whether the high or low value is used, then it may be necessary to take some additional steps to improve the accuracy of the input or account for it in other ways.

Sensitivity analyses can be used regardless of whether the estimate is monetized, is a proxy, or is somehow addressed with quantitative or qualitative information. However, for administrative ease jurisdictions may want to limit sensitivity analyses to cost-effectiveness inputs that are relatively uncertain and are likely to have a significant impact on the results.

7.8 Reliability of Data

All future costs and benefits of electricity and gas utility resources need to be estimated, and thus there is uncertainty in analyzing the cost-effectiveness of any type of energy resource—demand or supply. Including hard-to-monetize impacts does not change a cost-effectiveness calculation from an absolute to an estimated range of values. It may appear that accounting for hard-to-monetize impacts will reduce the accuracy and precision of the decision, but in fact the results will be more reliable than simply ignoring the hard-to-monetize impacts altogether.

The line between a rigorously established, monetary value and one that is less rigorously established can be subjective, because some level of professional judgement and estimation is typically involved in the development of all cost-effectiveness inputs. For example, the projected values for avoided costs or the effective useful life of an efficiency measure cannot be directly measured in advance.

All substantive impacts should be included in a jurisdiction's analyses, with documentation of the assumptions and analyses. It should account for them in decision-making, recognizing the limits of the reliability of the overall cost-effectiveness analyses. To not include all substantive impacts increases the risk of making an error of omission (not including efficiency resources that are more cost-effective than other resources), as well as an error of commission, including efficiency resources that are not as cost-effective as other resources.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001A Page 85 of 146

8. Participant Impacts

This chapter expands upon guidance in Subsection 3.3.2 regarding how to determine whether to include participant impacts in the RVT. It explains the policy objectives that might suggest including participant impacts, as well as key considerations regarding those objectives.

8.1 Summary of Key Points

Efficiency program participants experience several types of costs and benefits. Program participant impacts are summarized in Table 7 (Chapter 3) and discussed in more detail in Chapter 6.3.1Appendix C.

When considering whether to include participant impacts in the RVT, it is important to recognize two overarching points. First, the decision of whether to include participant impacts in the primary cost-effectiveness test is a policy decision. Second, if regulators decide to include participant costs in any cost-effectiveness test, the test must also include participant benefits, and *vice versa*.

Table 8 in Chapter 3 provides a summary of the reasons to include participant impacts in the primary cost-effectiveness test, as well counter-points to these reasons. These points and counter-points are discussed in more detail below.

8.2 Policy and Symmetry

When considering whether to include participant impacts in the cost-effectiveness tests, it is important to recognize two overarching points:

- 1. The decision of whether to include participant impacts in the primary costeffectiveness test is a policy decision. Regulators may choose to include participant impacts in the primary cost-effectiveness test if that would achieve the jurisdiction's policy goals.
- 2. If regulators decide to include participant costs in any cost-effectiveness test, the test must also include participant benefits, and *vice versa*. This is necessary to ensure symmetrical treatment of participant impacts, consistent with *Symmetry Principle* set forth in Chapter 1.

With regard to the first point above, some jurisdictions may not have an explicit policy goal regarding whether to include program participant impacts when assessing EE resources. Legislators and other decision-makers may not have addressed this question when promulgating legislation or regulations related to EE resources. In these cases, regulators and other decision-makers should decide whether to include participant impacts based upon the policy context that does exist in the jurisdiction and with appropriate input from relevant stakeholders.

In making this decision, it is important to consider the rationale and implications of including participant impacts in the primary test. These are discussed in the following sections.

8.3 Account for the Impacts on All Customers Combined

One of the reasons for including participant impacts in the primary cost-effectiveness test is to account for the impacts on all utility customers, both program participants and

non-participants, regardless of who experiences the impacts. This allows for a broader accounting of impacts than what is included as utility system costs alone.

However, it is important to recognize that participant impacts fall outside the scope of utility system impacts, and that this It is important to recognize that participant impacts fall outside the scope of utility system impacts.

distinction is important when assessing efficiency resource cost-effectiveness. Some of the participant impacts are energy-related while others are not. For example, a customer might use an efficient lighting rebate to install high-end lighting measures that offer aesthetic benefits as well as efficiency improvements. In this case, the customer incurs non-energy costs (higher costs than the low-end efficiency measure), and enjoys nonenergy benefits (in terms of improved aesthetics). The presence of non-energy costs and non-energy benefits is an important consideration when deciding whether to include participant impacts in the primary efficiency screening test.

8.4 Account for the Total Cost of the Resource

Another reason sometimes mentioned for including participant impacts in the primary cost-effectiveness analysis is to account for the total cost of the resource. This reason is predicated on the concern that *not* accounting for the total cost of a resource might result in a decision that appears cost-effective but is not. In other words, if the cost of a resource is divided up between two entities (the utility and the participant), then there is a risk that the total cost of the resource exceeds the total benefit, but neither the utility nor the participant would recognize this because each entity is concerned with only its own costs. This could be considered an uneconomic outcome, because the total (utility plus participant) costs might exceed the total benefits. This point is explained in the example in the text box.

If the goal of the cost-effectiveness analysis is to assess the total cost of a resource, then it is necessary to include the total benefits of the resource as well. And the total benefits must include utility system, participant, and societal benefits. In this example,

there may be non-utility system benefits (participant or societal) that are not considered. One example is environmental benefits. Continuing the text box example, assume that the resource in question has environmental benefits that are equal to 2 cents/kWh. This would mean that the total benefit of the utility system plus the environmental benefits would be 12 cents/kWh, which is higher than the total costs of 11 cents/kWh. This would mean that the resource is in fact cost-effective when this additional benefit is accounted for.

This example illustrates why, if regulators are interested in the total costs of a resource to avoid uneconomic outcomes, they must also account for the total benefits of the resource. In theoretical terms, this naturally leads to the conclusion that the only way to avoid this type of uneconomic outcome is to apply an SCT that accounts for all the costs and benefits of the resource. Using a test that includes all the participant impacts, without other impacts, will not answer this key question.

However, this conclusion does not mean that regulators must necessarily use an SCT as the primary test for assessing EE cost-effectiveness. If regulators are interested in the total cost of a resource

An Incomplete Picture of Costs and Benefits

Assume that an electricity utility has an avoided cost (including all utility system benefits) of 10 cents/kWh, with retail rates equal to 14 cents/kWh, and that an efficiency resource has a total (incremental) cost of 11 cents/kWh. This efficiency resource would be considered to be not cost effective if the total cost were accounted for (because 11 cents is greater than 10 cents).

Now assume that the utility offers a customer rebate of 5 cents/kWh to adopt this measure, which requires the customer to pay 6 cents/kWh for the remainder of the cost. If the total cost were split between the utility and the participating customer in this way, then the UCT would indicate the resource is costeffective (because 5 cents is less than 10 cents/kWh), and the customer would conclude that the resource is cost-effective (because 6 cents is less than 14 cents).

In this example, if the total cost were not considered as part of the cost-effectiveness analysis, then *it appears as though* an uneconomic resource would be deemed to be cost-effective from purely a total costs perspective.

However, this conclusion does not account for all the benefits of the resource, and thus provides an incomplete picture of costs and benefits.

solely to avoid potentially uneconomic outcomes, an SCT could be used as a preliminary, pre-screening test to ensure that all efficiency resources being considered will not result in the uneconomic outcome described above. Then the RVT could be applied as the primary test for determining whether the *relevant* benefits exceed the *relevant* costs.

Finally, if regulators and others are concerned about utility customers paying "too much" for an efficiency resource because the total costs have not been compared to the total benefits, then regulators can require that utility incentives to the participant for EE resources be capped at a level equal to the utility system avoided costs. Continuing the example above, the customer incentive would be capped at 10 cents/kWh, which means that utility customers would never be required to pay more than what the resource is worth to them. This concept is discussed in Subsection 3.3.1 as well.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001A Page 88 of 146

8.5 Protect Program Participants

Another reason to include participant impacts in the primary cost-effectiveness test would be to protect program participants. This reason is based on the presumption that including participant impacts in the test will ensure that participants' benefits will exceed costs.

There are several considerations regarding the extent to which including participant impacts in the cost-effectiveness test will protect program participants. First, the conventional method of including participant impacts in a cost-effectiveness test does not provide a clear indication of the impact on participants. The benefits to participating

The conventional method of including participant impacts in a costeffectiveness test does not provide a clear indication of the impact on participants.

customers will be in the form of reduced bills, which will be driven by the energy savings times the retail prices they pay for energy. However, the benefits that are included in the cost-effectiveness test used to account for participant impacts (the TRC test) are in the form of avoided utility costs, not reduced bills. In short, the difference between retail energy prices and utility avoided costs will typically distort the overall impacts on efficiency program participants.

Second, the Participant Cost test is a much more accurate means of protecting efficiency program participants, because this test uses reduced bills as the primary benefit to participants. Also, the Participant Cost test does not dilute the impacts on participants by combining them with the utility system impacts. The Participant Cost test is discussed in more detail in Appendix A.

Finally, the best way to ensure that program participants are protected is through efficiency program design. Successful and effective efficiency programs should be designed to entice customers to participate. This naturally leads to program designs that ensure that participants' benefits exceed their costs. If a program design results in participants' benefits not exceeding costs, then the program is not likely to be successful and should be redesigned. The Participant Cost test can, and often is, used as a way to ensure that programs are designed in a way that will entice customers by providing them with net benefits.

California's Methodology for Treating Non-Energy Costs and Benefits

The California efficiency program administrators have used the TRC test as their primary efficiency cost-effectiveness test, and they have applied an atypical methodology for addressing the challenges associated with the participant impacts. The California program administrators do not include either the participants' non-energy costs or non-energy benefits. In this way, the California TRC test includes only energy-related impacts—the utility system impacts plus the participants' energy-related impacts.

- The participant costs are determined by first estimating the total participant cost, and then subtracting estimated participant non-energy costs from those.
- The participant benefits are defined as only those related to energy impacts. Therefore, all participant non-energy impacts (comfort, health, safety, aesthetics, productivity, etc.) are excluded from the cost-effectiveness analysis.

8.6 Account for Low-Income Program Participant Benefits

Another reason to include participant impacts in the primary cost-effectiveness test would be to allow for the inclusion of low-income participant benefits. Efficiency programs can provide significant benefits to low-income customers, including reduced energy burden, improved health and safety, improved comfort, and more. If program participant impacts are included, then it follows that low-income participant benefits must be included as well.

There are two important considerations when deciding whether participant benefits should be included in the primary test to ensure that low-income benefits are included. First, if a jurisdiction has a policy goal of providing efficiency programs for the benefit of low-income participants, this does not mean that the primary cost-effectiveness test must account for the participant benefits of all customers to do so.

While it is true that if program participant costs are included in a test, then low-income customer benefits should be included as well, the inverse is not necessarily true. A jurisdiction might have a clear policy goal to account for low-income participant benefits, but not a comparable goal to account for all customer participant impacts. In fact, some states already do this. For example, Connecticut and Michigan use the UCT as the primary cost-effectiveness test, but do not require low-income efficiency programs to pass a cost-effectiveness test because of their participant benefits.

The second, and related, consideration is that well-designed low-income programs typically do not include any participant costs. By their very nature, low-income customers are unable or unlikely to participate in efficiency programs if there is any kind of participant cost, or even any significant participant transaction costs. This makes low-income efficiency programs fundamentally different from other efficiency programs. Some of the reasons that might support the inclusion of participant impacts in the primary cost-effectiveness test, such as considering all costs and protecting participants, are not relevant if there are no participant costs.

8.7 Account for Other Fuel and Water Impacts

Similarly, another reason to include participant impacts in the primary cost-effectiveness test would be to allow for the inclusion of other fuel and water impacts. Some efficiency programs can save a significant amount of other fuels, such as electricity (for a gas utility), gas (for an electric utility), oil, propane, or wood. These other fuel savings can sometimes represent a large portion of the savings from efficiency measures, particularly for certain programs such as home retrofit or new construction programs. They can also allow for a fuel-neutral, whole building approach to EE program delivery. If program participant costs are included in the primary cost-effectiveness test, then it follows that participant benefits must be included as well.

While it is true that if program participant costs are included in the primary costeffectiveness test, then participant other fuel and water impacts must be included as well, the inverse is not necessarily true. A jurisdiction might have a clear policy goal to account for other fuel and water savings, but not a comparable goal to account for all customer participant impacts. This could happen, for example, if a jurisdiction has policy goals supporting fuel-neutral, whole building approaches to efficiency program delivery, but not a comparable goal to account for all participant impacts. A jurisdiction might also have a policy goal of considering all potential fuel savings in order to assess strategic electrification opportunities, but not a comparable goal to account for all participant impacts. This issue is also addressed in Subsection 3.3.4.

8.8 Quality of the Information

Some participant costs and benefits can be difficult to quantify and monetize, for three reasons.

- <u>Total incremental costs</u>.³⁵ When designing and implementing efficiency programs, the cost to the utility system, i.e., the financial incentive provided to the participant, is known with great certainty. The amount that the participant pays is known with less certainty, and in some cases, can be very difficult to estimate. This is particularly true for efficiency measures where a wide range of customer options and costs are available.
- <u>Non-energy costs</u>. For some efficiency measures, a portion of the incremental costs are a result of product features that are not related to efficiency savings. These non-energy costs often result in a wide range of total incremental costs for efficiency measures, creating a challenge for efficiency planners who typically require one cost estimate for cost-effectiveness analysis.
- <u>Non-energy benefits</u>. The nature of some of these impacts, such as improved productivity, increased health and safety, and improved aesthetics, makes them uncertain, variable by customer and by program. They require different types of analyses to identify them (SERA 2014).

The fact that there are challenges with estimating participant costs and benefits does not, in and of itself, mean that they should be ignored in cost-effectiveness analyses.³⁶ It does mean that regulators and other decision-makers should consider these challenges, along with the other factors described above, when deciding whether to include participant impacts in the primary cost-effectiveness test.

³⁵ The term "incremental cost" is used to refer to the portion of cost associated with the improved efficiency of the measure, which is equal to the difference between the cost of the efficiency measure and a baseline measure.

³⁶ As described in Chapter 1, one of the key principles of cost-effectiveness analyses is that all relevant impacts should be accounted for, even the hard-to-quantify and hard-to-monetize benefits. In addition, Chapter 7 provides methodologies and techniques for accounting for all relevant costs and benefits, including those that are hard to monetize.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001A Page 91 of 146

9. Discount Rates

This chapter provides guidance on how to determine a discount rate for the RVT that is consistent with the objective of the cost-effectiveness analysis and the jurisdiction's applicable policy goals. The concepts described in this chapter can also be used to determine discount rates for other cost-effectiveness tests, including tests used for DERs and supply-side resources.

9.1 Summary of Key Points

The discount rate reflects a particular pattern of "time preference," which is the relative importance of short- versus long-term impacts. A higher discount rate gives more weight to short-term impacts, while a lower discount rate gives more weight to long-term impacts.

The choice of discount rate is a policy decision that should be informed by the jurisdiction's energy and other applicable policies—and thus should reflect the regulatory perspective, as described earlier in the manual. This perspective recognizes that the objective of efficiency cost-effectiveness analysis is to identify those utility resources that will best serve customers over the long term, while also achieving applicable policy goals of the jurisdiction.

The following steps can assist jurisdictions in determining the discount rate for the RVT:

<u>Step A: Articulate the jurisdiction's applicable policy goals</u>. These should be the same goals used in developing the RFT and should serve as the basis of the jurisdiction's regulatory perspective.

<u>Step B: Consider the relevance of a utility's weighted average cost of capital</u> (<u>WACC</u>). Is the utility investor time preference consistent with the jurisdiction's applicable policy goals?

<u>Step C: Consider the relevance of the average customer discount rate.</u> Should the discount rate be based on the average utility customer time preference? Does this time preference adequately address applicable policy goals and future utility customers?

<u>Step D: Consider the relevance of a societal discount rate.</u> Is a societal time preference and use of a societal discount rate consistent with the jurisdiction's policy goals and associated regulatory perspective?

<u>Step E: Consider an alternative discount rate</u>. Given that the regulatory perspective may be different from the utility, customer, and societal perspective, the discount rate does not need to be tied to any one of these three perspectives. For example, regulators/decision-makers could decide to use a discount rate that is lower than the utility WACC and the customer discount rate, but higher than the societal discount rate.

<u>Step F: Consider risk implications</u>. Consider using a low-risk discount rate for EE cost-effectiveness if the net risk benefits of EE resources are not somehow accounted for elsewhere in the cost-effectiveness analysis.

9.2 The Purpose of Discount Rates

Discount rates are an essential aspect for assessing any multi-year project or investment. They allow analysts to compare costs and benefits that occur over different time periods.

Some utility costs, such as power plant siting, licensing, and construction, occur in the short term. Other utility costs such as fuel and O&M stretch into the long-term future. A power plant takes a few years to build, and then generates electricity for decades. Many efficiency resources can be implemented

within a year or two, and then save energy for many years thereafter.

The key point here is that dollars at different times in the future are not directly comparable values; they are apples and oranges. Applying discount rates turns costs and benefits in different years into comparable values. The discount rate essentially reflects a particular pattern of "time preference," which is the relative importance of short- versus long-term costs and benefits.

The discount rate essentially reflects a particular pattern of "time preference," which is the relative importance of short- versus long-term costs and benefits. A higher discount rate gives more weight to short-term costs and benefits than to long-term costs and benefits, while a lower discount rate weighs short-term and long-term impacts more equally. Different economic actors may have differing discount rates, based on their own time preferences.

The choice of discount rates is a critical element of any long-term cost-effectiveness analysis because it has large impacts on the results. This is especially true when the analysis involves long-lived efficiency resources such as building retrofit programs and new construction programs.

Figure 5. Implications of Discount Rates (annual present value dollars) illustrates how EE benefits (e.g., avoided generating fuel costs) can be affected by different discount rates. This example starts with an annual fuel costs savings of \$10 per year over the course of a 20-year period. The top, blue line indicates the magnitude of the future avoided costs assuming no discount rate. The other lines present the annual present value of the avoided fuel benefit, depending upon the discount rate used. As indicated, higher discount rates will dramatically reduce the value of avoided fuel savings benefits in Year 20, while lower discount rates have a much smaller impact.



Figure 5. Implications of Discount Rates (annual present value dollars)

These benefits are presented as real dollars (i.e., excluding inflation), and the discount rates are real discount rates.

Figure 6 presents the same information using cumulative present values. Without discounting, a stream of \$10 over 20 years would equal \$200. The cumulative present value of this stream would be considerably lower. A real discount rate of 8 percent would result in a cumulative present value that is half the cumulative value of the original stream.



Figure 6. Implications of Discount Rates (cumulative present value dollars)

9.3 Commonly Used Discount Rates

Different Perspectives and Time Preferences

Table 19 summarizes several types of discount rates that could be used for energy resource cost-effectiveness assessment. For each type of discount rate, it indicates the time preference represented by that rate, a range of typical values, some brief notes, and sources.

Table 19. Discount Rate O	ptions for Cost-Effectiveness Analys	es
---------------------------	--------------------------------------	----

Type of Discount Rate	Potential Indicator of Time Preference	Typical Values (in real terms)	Notes and Sources
Societal	Societal cost of capital, adjusted to consider intergenerati onal equity or other societal values	<0% to 3%	In addition to low-risk financing, government agencies have a responsibility to consider intergenerational equity, which suggests a lower discount rate (US OMB 2003). Society's values regarding environmental impacts might warrant the use of a negative discount rate (Dasgupta, Maler, and Barrett 2000).
Low-Risk	Interest rate on 10-year U.S. Treasury Bonds	-1.0% to 3%	Over the past decade the real interest rate on 10-year U.S. Treasury Bonds ranged between -0.6% and 3.0% percent. As of the publication of this document, the real interest rate on 10-year U.S. Treasury Bonds was 0.4 percent (multpl.com 2017).
Utility Customers on Average	Customers' opportunity cost of money	varies	Customers' opportunity costs can be represented by either the cost of borrowing or the opportunity costs of alternative investments (Pindyck and Rubinfeld 2001, 550). The real rate on long-term government debt may provide a fair approximation of a discount rates for private consumption (US OMB 2003).
Publicly Owned Utility	Publicly owned utility's cost of borrowing	3% to 5%	Publicly owned utility costs of capital are available from the Federal Energy Regulatory Commission Form 1, Securities Exchange Commission 10k reports, and utility Annual Reports.
Investor- Owned Utility	Investor- owned utility's weighted average cost of capital	5% to 8%	Investor-owned utility costs of capital are available from the Federal Energy Regulatory Commission Form 1, Securities Exchange Commission 10k reports, and utility Annual Reports.

Typical values of discount rates are in real terms, as opposed to nominal. Real discount rates should always be applied to real cash flows, and nominal discount rates should always be applied to nominal cash flows. The utility cost of capital should be after-tax.

The typical values presented in Table 19 are provided for illustrative purposes only; other values outside these ranges are also possible. Other points to consider include: that these values can change over time according to changing economic conditions; that there are multiple options for determining a low-risk discount rate; and that different utility customers will have different time preferences, which can be determined in multiple ways. It is also worth noting that the value to use for the societal discount rate is

subject to much debate. Further discussion on the range of values for discount rates is beyond the scope of this manual.

EE planners and other stakeholders often recommend that the choice of discount rate for efficiency analysis should reflect the perspective represented by the costeffectiveness test in use. For example, the U.S. Department of Energy and Environmental Protection Agency's National Action Plan on Energy Efficiency (NAPEE, 2007, 5-4) states that:

- The societal discount rate should be applied when using the SCT.
- The utility weighted average cost of capital should be applied when using the UCT, the TRC test, or the Rate Impact Measure (RIM) test.
- A customer discount rate should be used when applying the Participant Cost test.

While there is some logic to the concept of matching the discount rate to the perspective of the test used, this logic must be applied carefully. First, it is important to recognize the role of the applicable policies in developing the cost-effectiveness test and in determining the appropriate time preference. Second, it is important to be clear on whose perspective is actually represented in particular discount rates. These issues are discussed in the following sections.

The Role of the Cost of Capital

In general, the cost of capital is a key factor in determining discount rates. It indicates the time value of money (or the opportunity cost for alternative investments) for the relevant entity. However, cost of capital is not the only factor that dictates the appropriate discount rate to use for utility investments.

As described above, the primary objective of a utility cost-effectives analysis is to identify those utility resources that will best serve customers over the long term, while also achieving applicable policy goals of the jurisdiction. In light of this objective, the time preference for cost-effectiveness analysis should account for more than just the cost of capital; it should also account for the value of utility service over the long term and applicable policy goals. In other words, important utility services (such as providing safe and reliable power) and important policy goals (such as protecting low-income customers or promoting economic development) are all factors that affect the time preference relevant to the cost-effectiveness analysis.

This point is widely accepted in the application of the societal discount rate. That rate, which is used in multiple applications, reflects more than simply the cost of capital to society. It also reflects societal values and priorities, such as long-term benefits to society, achieving societal goals, addressing the needs and interests of multiple entities across society, and more. In a similar way, the discount rate used for cost-effectiveness analysis could reflect more than just the cost of capital.

9.4 The Regulatory Perspective

The regulatory perspective is an important concept for determining a jurisdiction's primary cost-effectiveness test (as described in Chapters 1 and 2), and associated discount rate. This perspective is typically not recognized or accounted for in the traditional cost-effectiveness tests, yet it is critical for identifying the costs, benefits, and priorities most relevant for any one jurisdiction.
NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001A Page 96 of 146

The regulatory perspective is the most relevant perspective for determining a discount rate for the primary costeffectiveness test. The regulatory perspective includes the full scope of issues for which regulators and other relevant decision-makers are responsible. It is typically based upon statutes, regulations, executive orders, commission orders, and ongoing policy discussions.

Chapters 1 and 2 address why the regulatory

perspective should be used to develop the primary RVT for a jurisdiction, and Chapter 3 provides more detailed guidance. By the same logic, the regulatory perspective is the most relevant perspective for determining a discount rate for the primary cost-effectiveness test.

9.5 The Investor-Owned Utility Perspective

When deciding which discount rate is most appropriate to use for cost-effectiveness analyses, regulators and other decision-makers should carefully consider the relevance of the "utility perspective." The investor-owned utility perspective is discussed in this section, and the publicly owned utility perspective is discussed in the next section.

The Investor-Owned Utility Perspective

The utility WACC is typically used to indicate the time preference for investor-owned utilities (i.e., reflects the time preference of the utility investors, which is the after-tax cost of equity and the cost of debt). The key goal of utility investors is to maximize the returns on their investments. Therefore, the time preference of utility investors is not necessarily the same as the time preference of utility customers, or the regulatory time preference.

Regulators/decision-makers should recognize this important distinction when considering whether to use the utility WACC as a discount rate. The primary objective of the cost-effectiveness analysis is to identify those utility resources that will best serve customers with safe, reliable, low-cost energy services over the long term. This objective is fundamentally different from the objective of maximizing utility investors' returns. These different objectives dictate different time preferences.

Another objective of the cost-effectiveness analysis is to meet the jurisdiction's applicable policy goals, which might include, for example, reducing the energy burden for low-income customers, reducing price volatility, reducing reliance upon fossil fuels, and reducing carbon emissions. Again, this objective of meeting applicable policy goals is fundamentally different from the objective of maximizing utility investors' returns; and these different objectives dictate different time preferences. These longer-term, broader objectives suggest that utility cost-effectiveness analyses should place a higher value on future impacts than utility investors would.

The Cost of Capital of Different Utility Resources

The goal of cost-effectiveness analysis is to compare the relative economics of investing in different resource options. The cost of capital used for resource acquisition varies across resource types. Therefore, even from a utility perspective, the discount rate used for such comparisons should reflect the cost of capital across the resource options under consideration.

A subset of resource costs, such as avoided capacity for generation, transmission, and distribution facilities, are financed by utility debt and equity. In contrast, it is often the case that EE resources and some supply-side resource costs have a much lower cost of capital than the WACC. The utility system costs of acquiring efficiency resources are typically recovered promptly through reconciling charges, and therefore involve no debt or equity costs. Similarly, some supply-side resource costs, such as fuel and purchased power costs are recovered promptly through reconciling charges, and therefore have little to no cost of capital.

In sum, when considering all of the resources used in the cost-effectiveness analyses (EE, avoided energy, avoided purchased power, avoided capacity) the actual WACC is considerably lower than the utility WACC, given the amount of resources that are not financed with debt or equity. This suggests that the utility WACC may be too high for the purposes of comparing the cost-effectiveness of different resources in utility resource planning.

Collection of Revenues to Pay for Debt and Equity

It is sometimes argued that the utility WACC should be used as a discount rate because investor-owned utilities need to collect sufficient revenues to pay dividends and interest to their investors. However, this rationale is not valid because the choice of the discount rate has no impact on the ability of the utility to recover its cost of capital.

The recovery of any debt and equity costs associated with resource acquisition should be included in the calculation of each resource's costs and benefits in the costeffectiveness analysis. For example, the avoided capital cost of a new power plant should be calculated in terms of annual revenue requirements, which should include depreciation plus the recovery of debt, equity, and taxes over the book life of the asset. Given that the recovery of debt and equity costs should be included in all of the relevant costs and benefits of the resources, there is no need to tie the utility cost of capital to the discount rate.

Unregulated Companies Versus Regulated Utilities

It is also important to consider whether the concept of using the investor-owned utility WACC for a discount rate is appropriate for regulated utilities. While this concept is standard practice for unregulated companies, there are several important differences between unregulated businesses and regulated utilities.

The differences between unregulated businesses and regulated utilities are similar to those described above regarding the utility investor perspective. In fact, the utility investor perspective is essentially the same as the perspective of unregulated businesses, where the primary objective is to maximize profits. Regulated utilities have broader and longer-term objectives, which suggests that regulated utilities should place a higher value on future impacts than unregulated businesses do.

This point is particularly important given that using utility WACC for discount rates is so deeply embedded in utility industry practices. Much of the reason for this is likely due to the conventional practices used in other industries. Before continuing the use of conventional practices for unregulated businesses, regulators/decision-makers should carefully consider whether those conventional practices apply to regulated utilities.

9.6 The Publicly Owned Utility Perspective

Publicly owned utilities, such as public power authorities, municipal utilities, and cooperatives, likely have a different time preference than investor-owned utilities. First, the cost of capital for publicly owned utilities is typically based solely on debt, and therefore is much lower than the WACC of investor-owned utilities.

Second, publicly owned utilities are different from investor-owned utilities by design. One of the reasons for creating publicly owned utilities is to shift the focus of the utility management away from utility investors and toward the needs and interests of customers. Therefore, the time preference of publicly owned utilities is likely to be more aligned with the time preference of utility customers as a whole.

Many publicly owned utilities are overseen and managed by public or customer representatives. For example, municipal utilities are typically overseen by municipal selectmen, councilmen, or boards of customer representatives, and cooperative utilities are typically managed directly by boards of customers or customer representatives.

The boards and agencies that manage publicly owned utilities (i.e., the ultimate decisionmakers on resource assessment) essentially act as both the "regulators" and the utility management. Consequently, for publicly owned utilities the utility perspective is naturally more aligned with the "regulatory" perspective. This suggests that publicly owned utilities should naturally place a higher value on long-term costs and benefits than investorowned utility investors would.

9.7 The Utility Customer Perspective

As described above, the primary objective of utility cost-effectiveness analysis is to identify those utility resources that will best serve customers over the long term, while also achieving applicable policy goals of the jurisdiction. Given that a key objective of the analysis is to serve customers, the utility customer time preference is an important consideration in determining the appropriate discount rate for the analysis.

Regulators/decision-makers should consider several issues when assessing customer time preference. The customers' cost of capital is only one factor that will influence the customers' time preference. Customers are interested in several aspects of utility services beyond just the costs. For example, they may also be interested in reliability of services, price volatility, power quality, etc. These additional aspects of utility service mean that customers might place a different time preference on dollars spent on utility services relative to dollars spent on other products or other investments.

In addition, the customer cost of capital varies considerably across customer classes, and also across customers within classes. Any one cost-effectiveness test, however, can use only one discount rate. Therefore, to the extent that

In some ways, the time preference from a regulatory perspective is aligned with utility customers' time preference. In both cases, time preference should be consistent with the objective of identifying those resources that will best serve customers. The time preference from the regulatory perspective, however, captures two additional considerations. First regulators/other decision-makers have a responsibility to ensure that utility resources will meet applicable policy goals. Second, regulators have a responsibility to consider both current and future customer interests. For both of these reasons, the regulatory perspective should place a higher value on long-term costs and benefits than the utility customer perspective.

the customer cost of capital is used to inform the determination of a discount rate, it should be an average cost of capital that represents the broad range of utility customers.

9.8 Risk Considerations

Accounting for Risk in Determining the Discount Rate

Risk is often cited as an important factor to consider when determining a discount rate, because risk can affect the value that one might place on long-term versus short-term impacts. However, risk can be represented in different ways in a cost-effectiveness analysis, and it is important to be careful that any treatment of risk in the discount rate recognizes how risk is addressed in the rest of the analysis to ensure that there is no double-counting or under-counting of risk.

Risks can vary considerably across different types of utility resources. For example, EE resources tend to create relatively low risk; generators create different amounts of capital cost, siting, and construction risks; fossil-fueled generators create price escalation and volatility risks; and transmission and distribution facilities impose their own kinds of risks (Ceres 2012).

In general, it is preferable to account for such resource-specific risks separately and explicitly for each resource type, rather than embed it in a discount rate. Discount rates are applied to all resources in a cost-effectiveness analysis. Applying a single discount rate to all resources to reflect risks associated with any one of those resources, could conflate the treatment of resource-specific risk with the overall choice of time preference. Instead, resource-specific risk should be accounted for in developing the cost and benefit inputs to the cost-effectiveness analysis.

Addressing Resource-Specific Risk

There are at least three techniques for addressing resource-specific risk. First, resourcespecific risk should be accounted for in the financing costs of the resources themselves. The cost of capital used to determine the cost of each resource should reflect the capital and construction risks associated with that resource. For example, a large new nuclear plant could be assumed to have a high, risk-adjusted, cost of capital to reflect the relevant nuclear capital and construction risks. In contrast, the cost of acquiring EE resources are typically recovered promptly through reconciling charges, and therefore no financing costs are included in their costs.

Once the financial risk of each resource has been accounted for in the financing costs, any other resource-specific risk considerations should be explicitly applied to the costs of those resources. For example, for efficiency resources that avoid potential fuel price volatility or escalating carbon emissions costs (i.e., risk benefits that are not captured in the avoided costs themselves) this risk benefit can be accounted for by either reducing the cost of the efficiency resources or increasing the magnitude of avoided costs (VT PSB 1990).

Finally, the analysis used to develop avoided costs should employ risk assessment techniques to account for the risks associated with the portfolio of resources that define avoided costs (Ceres 2012). There are multiple techniques for portfolio risk assessment, including scenario analyses and probabilistic analyses.

Energy Efficiency Risk

There may be situations where the costs or benefits used in the EE cost-effectiveness analysis do not properly reflect resource-specific risks. For example, the full set of risks associated with avoided costs (e.g., risks associated with avoided fuel costs) may not be fully captured in the avoided costs that are input to the cost-effectiveness analysis.

In such situations, regulators/decision-makers may choose to apply a low-risk discount rate to reflect the net risk benefits of EE resources, because those benefits are not otherwise accounted for in the inputs to the analysis. There are multiple options for determining a low-risk discount rate; the interest rate on 10-year U.S. Treasury bonds is frequently used for this purpose. Several states currently use this low-risk indicator for determining the discount rate their EE cost-effectiveness analyses (NEEP 2014, 43).

9.9 Determining the Discount Rate

9.9.1 Discount Rate for the Resource Value Test

Ultimately, the choice of discount rate is a policy decision—a decision regarding how much weight to give to long-term versus short-term costs and benefits. When determining the discount rate for the RVT, this policy decision should be guided by the regulatory perspective, the same perspective that is used to define that test.

The regulatory perspective may differ from one jurisdiction to another. Therefore, each jurisdiction should determine a discount rate for the RVT based on its own policies and goals. Regulators/decision-makers can take the following steps to make this determination.

The regulatory perspective may differ from one jurisdiction to another. Therefore, each jurisdiction should determine a discount rate for the RVT based on its own policies and goals.

Step A: Articulate Policy Goals

Section 3.1 describes how regulators should identify and articulate policy goals as the first step in the Resource Value Framework. Those same policy goals should be articulated and applied when determining the discount rate for the RVT.

Step B: Consider the Utility Investor Perspective

Regulators should consider whether the utility WACC represents the regulatory time preference, based on the considerations outlined above. Is the utility investor time preference consistent with the jurisdiction's regulatory perspective and policy goals? Is the utility investor time preference the appropriate time preference for resource planning? Does the utility WACC accurately reflect the cost of capital of efficiency and the other resources being assessed?

- If the answer to these questions is "yes," then the utility WACC could be used as the discount rate.
- If the answer to these questions is "no," then a discount rate that is lower than the utility WACC could be used. A lower discount rate would be warranted if either (a) the actual cost of capital across all resources is lower than the utility's WACC; or (b) the regulatory perspective places a greater value on long-term impacts than utility investors.

Step C: Consider the Average Customer Discount Rate

Regulators should consider whether the average customer discount rate represents the regulatory time preference, based on the considerations outlined above. Should the discount rate be based on the average utility customer cost of capital? Does this time preference adequately address applicable policy goals and future utility customer?

- If the answer to these questions is "yes," then the average customer discount rate as the discount rate could be used.
- If the answer to these questions is "no," then a discount rate that is lower than the average customer discount rate could be used. A lower discount rate would be warranted if the customer discount rate does not adequately account for policy goals and long-term customer impacts.

Step D: Consider the Societal Discount Rate

Regulators should also consider whether a societal discount rate is appropriate for the primary cost-effectiveness test, based on the considerations outlined above. Is a societal time preference consistent with the jurisdiction's applicable policy goals?

- If the answer to this question is "yes," then a societal discount rate could be used.
- If the answer to these questions is "no," then a discount rate that is higher than the societal discount rate could be used. A higher discount rate would be warranted if the jurisdiction's places less value on long-term impacts than society would.

Step E: Consider an Alternative Discount Rate

Regulators/decision makers should also consider whether to use a discount rate that is not tied to any one of the three perspectives described above. The regulatory perspective may be different from the perspective of utility investors, customers, and society; thus, the regulatory time preference and discount rate could be different as well.

- Does the jurisdiction's regulatory perspective suggest a greater value on longterm impacts than that of utility investors?
 - If so, then use a discount rate that is lower than the utility WACC. If not, then use a discount rate that is higher than the utility WACC.
- Does the jurisdiction's regulatory perspective suggest a greater value on longterm impacts than that of customers?
 - If so, then use a discount rate that is lower than that of customers. If not, then use a discount rate that is higher than that of customers.
- Does the jurisdiction's regulatory perspective suggest a greater value on longterm impacts than that of society?
 - If so, then use a discount rate that is lower than that of society. If not, then use a discount rate that is higher than that of society.

Step F: Consider Risk Implications

Resource-specific risk issues are best accounted for in estimating the costs of each resource, for example in the resource-specific cost of capital, as adjustments to a resources costs or benefits, and/or in the avoided cost portfolio modeling process.

Nonetheless, there may be situations where the EE costs or benefits do not properly reflect resource-specific risks. For example, the full set of risks associated with avoided costs (e.g., risks associated with avoided fuel costs, risks associated with construction costs) are often not captured in the cost-effectiveness inputs. In such situations, regulators and other decision-makers may choose to apply a low-risk discount rate to reflect the net risk benefits of EE resources, because those benefits are not otherwise accounted for in the inputs to the analysis.

9.9.2 Discount Rates for Different Cost-Effectiveness Tests

The discount rate concepts and considerations described in this chapter are not only relevant to the RVT; they are also relevant to other tests.

The Utility Cost Test

For all the reasons discussed above in Section 9.5, regulators and other decisionmakers should be circumspect about using the utility WACC as the discount rate for the UCT. The utility WACC represents the perspective of utility investors, which is fundamentally different from the customer or regulatory perspectives.

This distinction between the customer or regulatory perspectives and utility investor perspectives is relevant regardless of which test is used for EE cost-effectiveness. In all cost-effectiveness analyses, the purpose is to identify resources that best serve

customers, and the regulators are in the best position to define what is in the long-term interest of customers. Therefore, the discount rate to use for the RVT should be used for the UCT as well.

Note that the UCT does not represent the perspective of the "utility" *per se* (i.e., in terms of the interests of utility investors or utility management). This test includes all the costs and benefits within the scope of the

In all cost-effectiveness analyses, the purpose is to identify resources that best serve customers, and the regulators are in the best position to define what is in the long-term interest of customers.

"utility system" that is used to serve customers, as described in Section 3.3 and Section 6.2.

This distinction between the "utility" (i.e., investors) and the "utility system" (i.e., customers) is important when considering whether the utility WACC is relevant for the UCT. The purpose of the UCT is to identify those resources that will best serve customers, including all costs that customers pay to the utility, and all benefits that customers receive from the utility. This is different from the goal of maximizing value for utility investors.

Total Resource Cost Test

The choice of a discount rate for the TRC test should be based on the same considerations as the choice for the UCT. Adding participant impacts in the test does not change the fact that the purpose of the cost-effectiveness analysis is to provide the best services to customers, and not to maximize shareholder value.

The Societal Cost Test

It is widely accepted that the societal discount rate should be used for the SCT. This is consistent with the notion of aligning the discount rate with the relevant perspective of

the test. It is also consistent with the concepts and considerations described above regarding a societal preference for achieving policy objectives and placing greater weight on long-term resource impacts.

The Participant Cost Test

It is widely accepted that a customer-based discount rate should be used in the Participant Cost test. Since the objective of this test it to determine the impacts on program participants, and is not to compare efficiency resources with other resources, a customer-based discount rate is appropriate for this test.

9.9.3 Discount Rates for Analyzing Different Resource Types

The overarching purpose of cost-effectiveness analyses for any type of utility resource is to identify those resources that will best serve customers over the long term. Therefore, one of the central concepts of this chapter—that the discount rate should be based on the regulatory perspective, which may be different from the utility investor perspective—is applicable to all types of utility resources.

Regulators and other decision-makers should use the steps described in subsection 9.9.1 to determine the discount rate for analyzing the cost-effectiveness of any type of utility resource. This includes all types of DERs (EE, demand response, distributed generation, and storage), as well as all types of supply-side resources (generation, transmission, and distribution).

The rationale for determining the discount rate for the RVT is relevant across all of these resources. Further, using the same discount rate across all utility resource cost-effectiveness analyses will make the results of those analyses comparable. It will also allow for a more direct comparison across all resource types.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001A Page 104 of 146

10. Assessment Level

The cost-effectiveness of efficiency resources can be assessed at several levels of aggregation. Assessments can focus on individual measures, individual customer-specific projects, individual programs combining multiple measures and/or projects, sectors (e.g. all residential or all business programs), or portfolios of programs (across all sectors). This chapter discusses the advantages and disadvantages of conducting cost-effectiveness analyses at each of those levels. It also discusses the level at which fixed costs should be included in analyses.

10.1 Summary of Key Points

- Cost-effectiveness assessment at all levels—measure, project, program, sector, and portfolio—can provide valuable insight into program design and implementation. Efficiency planners and other stakeholders may want to analyze efficiency resources at several, if not all, of these levels.
- When applying the primary cost-effectiveness test, or otherwise determining which efficiency resources merit funding, regulators and efficiency planners should rely upon program-level, sector-level or portfolio-level cost-effectiveness results.
- When applying the primary cost-effectiveness test, or otherwise determining which efficiency resources merit funding, regulators and efficiency planners should not rely upon measure-level or project-level cost-effectiveness results. Any advantages of measure-level and/or project-level application are typically outweighed by the disadvantages.
- Consistent with the principle that cost-effectiveness analyses should be forward-looking and focused only on marginal impacts (see discussion in Chapter 1), efficiency program costs should be included in cost-effectiveness analyses only at the level at which they become variable. For example, fixed program costs should not be allocated to measures for the purpose of assessing the cost-effectiveness of individual measures and fixed portfolio-level costs should not be allocated to programs for the purpose of assessing the cost-effectiveness of individual measures of assessing the cost-effectiveness of individual programs.

10.2 Assessment Level Options

10.2.1 Measure-Level Assessment

Resource assessment at the measure level means that each individual measure promoted by an efficiency program must be cost-effective on its own. Screening at the measure level is the most restrictive application of the cost-effectiveness tests.

Measure-level application of cost-effectiveness requirements will essentially guarantee that every measure included in an efficiency program will be cost-effectiveness on its own. However, application of cost-effectiveness requirements at that level can have

perverse implications. In some cases, it could reduce the overall net economic benefits of efficiency investments. That can occur for any of the following reasons:

- A customer's interest in a non-cost-effective measure may be key to persuading the customer to install a package of measures that are cost-effective in aggregate. In such cases, the flexibility to promote the non-cost-effective measure as part of a package will lead to greater overall net benefits.
- A customer's interest in a non-cost-effective measure may be key to the development of a relationship with the customer that can lead to installation of cost-effective measures in the future. In that sense, promotion of the non-cost-effective measure can be analogous to a marketing investment.
- Installation of a non-cost-effective measure may be necessary in order to technically or safely enable the installation of other cost-effective measures. An example of this would be the installation of non-cost-effective mechanical ventilation in order to make indoor air quality acceptable when tightening up a building.

Another disadvantage of requiring all measures to be cost-effective is that it can be difficult to account for non-energy impacts, hard-to-monetize impacts, or additional considerations at the measure level. Some non-energy impacts, such as improved health and safety, are obtained through a package of multiple measures, and it is impractical to apply such impacts on each measure.

10.2.2 Project-Level Assessment

Resource assessment at the project level means that the combination of measures implemented together in a package for an individual customer must be cost-effective on its own. Project-level assessments are typically conducted only for projects undertaken by larger business customers for which the transaction cost of a site-specific assessment can be justified.

Project-level application of cost-effectiveness requirements will essentially guarantee that every project included in an efficiency program will be cost-effectiveness on its own. However, application of cost-effectiveness requirements at that level can have some (though fewer) of the perverse implications of measure-level cost-effectiveness requirements. Specifically, supporting the implementation of a non-cost-effective package of measures in which a customer is interested can facilitate development of a relationship with customer that can produce a more cost-effective project later. Also, depending on whether and how participant non-energy benefits are included in cost-effectiveness assessments, the full value of non-energy benefits of a project may not be captured in project-level cost-effectiveness assessments.³⁷

³⁷ The focus of this discussion is solely on the use of cost-effectiveness analysis to determine which investments merit acquisition from either utility system or broader perspectives. Efficiency programs targeted to large business customers often present costs and benefits to individual customers from the customer's perspective (i.e. using retail energy prices rather than avoided system costs, as well as considering customer non-energy benefits that may or may not be part of a jurisdiction's cost-effectiveness test). Similarly, some low-income programs base the determination of which measures to install on the savings-to-investment ratio (i.e., benefit-to-cost ratio) derived using the customer's retail rate. The merits of such customer-focused analyses are fundamentally different from those discussed here regarding utility system resource analyses.

10.2.3 Program-Level Assessment

Resource assessment at the program level means that the measures and/or projects within a program must be cost-effective collectively. Some individual measures and/or projects may not be cost-effective on their own, but could still be included in the program if the overall program were cost-effective.

The primary advantage of this approach is that it best represents the costs and benefits of initiatives that combine a set of actions (e.g., marketing, education, technical support, financial support, etc.) into a single package offered to customers. In addition, resource assessment at the program level avoids the problems noted above regarding missing the interrelationships between measures. These include technical connections and the ability to engage customers in ways that can lead to increasing net economic benefits, as well as the ability to properly capture customer non-energy benefits where warranted.

A disadvantage of this approach is that a program might include one or more measures that are not individually cost-effective and are not needed to account for the concerns addressed above. This has the effect of decreasing to some extent the overall costeffectiveness of the program. However, this concern can be addressed with sound program design. Efficiency program planners and designers should include only those efficiency measures that effectively contribute to achieving the specific goals of the program.

One other potential concern with program-level screening is that it might preclude certain special programs that address important objectives at the sector or portfolio level. For example, pilot programs to test new and unproven program designs might not appear cost-effective, but might provide future sector or portfolio benefits that cannot be identified in the present. For that reason, jurisdictions that apply program-level screening may want to allow these types of programs to be considered in a sector-level assessment.

10.2.4 Sector-Level Assessment

Resource assessment at the sector level means that the programs within a sector (e.g., low-income, residential, commercial and industrial)³⁸ must be cost-effective collectively. Some programs may not be cost-effective on their own, but could still be implemented if the combined impact of all of the programs targeted to a given sector were cost-effective.

The primary advantage of this approach is that it indicates the costs and benefits of initiatives to provide a package of efficiency services to an entire sector. This may allow for non-cost-effective programs to be provided to a sector for the purpose of providing a complete set of efficiency services to that sector—an objective often driven by concerns

³⁸ Some jurisdictions treat low-income programs as their own "sector," because of the special consideration often given to such customers in program design and delivery. Others treat low-income programs as part of the residential sector. Alternatively, though commercial and industrial customers could be considered to be different "sectors," most efficiency programs targeted to business customers are do not differentiate between those two groups of customers, creating what are called business, non-residential, or commercial & industrial (C&I) sector programs. For the purpose of this manual, we call out low-income, residential, and C&I as three sectors of interest for illustrative purposes only. The conceptual discussion in this section applies regardless of whether low income is treated as its own sector or as part of the residential sector and regardless of whether commercial and industrial are treated as their own sectors or combined.

about equitable access to efficiency programs across a large range and number of customers.

The primary disadvantage of this approach is that it could result in the inclusion of efficiency measures or programs that are not individually cost-effective, thereby decreasing the economic value of the suite of programs for that sector.

10.2.5 Portfolio-Level Assessment

Evaluation at the portfolio level means that the programs within a portfolio (i.e., combining all programs together) must be cost-effective collectively. Some programs may not be cost-effective on their own, but could still be pursued if the combined impact of all of the programs was cost-effective.

The primary advantage of this approach is that it indicates the costs and benefits of the entire suite of EE programs.

The primary disadvantage of this approach is that it could result in implementing efficiency measures or programs that are not cost-effective, thereby decreasing the economic value of the overall portfolio.

10.3 Properly Accounting for Fixed and Variable Costs

A variety of costs are incurred in the acquisition of efficiency resources. It is important that those costs be included at the proper analytical level—e.g., measure, program, sector and/or portfolio—when analyzing the economics of efficiency resources. In a nutshell, only costs that are variable at a given analytical level should be included in

cost-effectiveness analysis for that level because they are the only costs that can be avoided as a result of the analysis. Costs that are largely fixed at a particular analytical level should not be "allocated" or otherwise included *at that level;* doing so could lead to rejection of investments whose marginal benefits exceed their marginal costs, thereby lowering net economic benefits. That does

Only costs that are variable at a given analytical level should be included in cost-effectiveness analysis for that level.

not mean that costs that are fixed at a given analytical level should be omitted or ignored altogether. Instead, they can and should be included at higher level analyses at which they are variable and therefore are avoidable.

For example, when assessing the economics of efficiency measures, one should include only costs that largely increase or decrease in proportion to the number of measures installed. That will obviously include the cost of the measures themselves, and could also include some program costs that are largely variable. Examples would include rebate processing costs, if the program administrator is paying a vendor a price for every rebate processed, and inspection costs if the program is committed to inspecting a certain percentage of all projects.³⁹ However, other program costs that are either largely

³⁹ Alternatively, if the program is committed to inspecting enough projects to get a statistically valid sample, such that the number of inspections would not change significantly or at all between a level of 2000 and

fixed or do not change in proportion to program participation levels, such as the costs of marketing⁴⁰ or managing and evaluating the program, should not be included in the economic analysis of individual measures. Rather, they should be included only at program-level cost-effectiveness assessment.

Similarly, portfolio costs that are either largely fixed or do not change in proportion to the number of programs or participation levels in those programs should not be allocated to programs for the purpose of analyzing the economics of individual programs. Rather, they should only be included at portfolio-level cost-effectiveness analysis. Such costs can include portfolio-level marketing, management, and evaluation costs.

The tables below illustrate the importance of accounting for largely fixed costs at the proper analytical level. Table 20 shows that for each of five programs analyzed, the benefits exceed the variable costs of the programs. When largely fixed portfolio costs (equal to about 25 percent of the sum of the five program costs) are added to the sum of the variable impacts of the five programs, the portfolio itself is shown to be cost-effective, providing total net benefits of \$800,000.

	Benefits (\$000)	Costs (\$000)	Net benefits (\$000)	Positive net benefits?
Program 1	\$500	\$250	\$250	Yes
Program 2	\$300	\$200	\$100	Yes
Program 3	\$1000	\$400	\$600	Yes
Program 4	\$500	\$300	\$200	Yes
Program 5	\$1000	\$850	\$150	Yes
Sum of all programs	\$3300	\$2000	\$1300	Yes
Portfolio-level costs	\$0	\$500	-\$500	
Total portfolio impacts	\$3300	\$2500	\$800	Yes

Table 20. Proper Analysis with 25 Percent Fixed Portfolio CostsIncluded at Portfolio-Level Analysis

Table 21 shows that when the fixed portfolio-level costs are improperly allocated as 25 percent "adders" to each of the programs, the fifth program is no longer seen as cost-effective. If that program is then removed from the portfolio, but with portfolio costs remaining unchanged, the portfolio net benefits decline by \$150,000 (i.e., the marginal impact of the fifth program on the portfolio) to \$650,000.⁴¹ In short, including fixed costs

^{10,000} participants, then such inspection costs should be treated as largely fixed and captured at the program level rather than at the measure level.

⁴⁰ Marketing costs can be somewhat variable in the sense that more marketing should lead to more participation. However, that relationship is rarely linear with the number of measures installed. In addition, and perhaps more importantly, program marketing budgets are often treated as largely fixed. That is, while marketing can play an important role in driving program participation, the costs of marketing do not go up and down as the number of participants goes up and down.

⁴¹ Removing the fifth program would require a reallocation of the fixed portfolio cost to the remaining four programs (i.e. each of the remaining four programs would now be allocated a larger portion of the fixed portfolio costs). In this example, the four remaining programs would still all be cost-effective even after absorbing this larger allocation. However, under a different set of example programs, it is possible that the resulting larger allocation of fixed costs would render another program cost-ineffective.

at the improper level can reduce the economic benefits of efficiency resource acquisition.

	Benefits (\$000)	Costs (\$000)	Net benefits (\$000)	Positive net benefits?
Program 1	\$500	\$313	\$188	yes
Program 2	\$300	\$250	\$50	yes
Program 3	\$1000	\$500	\$500	yes
Program 4	\$500	\$375	\$125	yes
Program 5	\$1000	\$1063	-\$63	no
Sum of all programs	\$3300	\$2500	\$800	yes
Portfolio-level costs	Incl	uded as ad	dder for each p	rogram
Total portfolio if non-cost- effective programs excluded	\$2300	\$1650	\$650	yes

Table 21. Improper Analysis with 25 Percent Fixed Portfolio CostsAllocated to Individual Programs

11. Analysis Period and End Effects

Analysis period refers to the number of years over which the costs and benefits of a resource investment are forecast and compared. This chapter describes the time period over which cost-effectiveness analysis should be conducted, and how to address any potential 'end effects.'

11.1 Summary of Key Points

- The analysis period should be long enough to capture the full stream of costs and benefits associated with the efficiency resources being analyzed.
- Since most efficiency resource costs are incurred immediately while benefits are spread out over time, failing to use an analysis period that covers the full life of the resource creates an "end effects" problem that biases cost-effectiveness assessments against efficiency resources.
- If it is not possible or is impractical to extend the analysis period to the full life of the efficiency resources being analyzed, then a second best alternative is to amortize costs of the efficiency resource over the full life of the benefits and then compute the net present value (NPV) of both costs and benefits for the same number of years. This better aligns the portion of the costs being considered with the portion of the benefits being considered.

11.2 Analysis Period

Analysis period refers to the number of years over which the costs and benefits of a resource investment are estimated and compared when assessing the resource's cost-effectiveness. The analysis period should be long enough to capture the full stream of costs and benefits associated with the resources under analysis.

For example, an assessment of three years of implementation of an efficiency program which includes measures that last 30 years (a common assumption for some building envelope measures such as insulation upgrades) should have at least a 32-year analysis period—i.e., long enough to assign value to benefits (and costs) for each of the 30 years of life of a measure installed in the third of the three program years analyzed.

If any of the programs are projected to have longer-term market effects, the analysis period should be extended to account for the life of the savings from the post-program period increases in measure installations. For example, if a three-year program promoting building envelop efficiency measures is expected to affect market penetrations of such measures for five years after the three-year program period ends (i.e., in Years 4 through 8), then the analysis period should be extended to 37 years. This is long enough to assign value to the benefits and costs for each of the 30 years of life of a measure installed in the eighth (and last) year of the forecast, post-program period market effects.

11.3 End-Effects Problems

If the cost-effectiveness analysis does not fully capture all of the impacts, there may be what is commonly called an "end effects" problem in which the analysis captures the full cost of an efficiency resource, but not all of the benefits. This occurs because costs are usually incurred at the time of installation of an efficiency measure and therefore are entirely within the analysis period, while benefits are typically spread out over the life of the measure, with some of the benefits occurring after the end of the analysis period. The asymmetrical treatment of costs and benefits results in an analytical bias against efficiency.⁴²

This is illustrated in Table 22, which compares the results of using (A) a proper analysis period for an efficiency resource with a 20-year life, and (B) a truncated analysis period of 15 years that creates an end-effects problem. In this hypothetical example, an analysis of the full lifetime benefits of the efficiency resource suggests the resource is cost-effective, with a benefit-cost ratio of 1.15. In contrast, when only 15 of the 20 years of benefits are counted because the analysis period is shorter than the resource life, one would reach the inaccurate conclusion that the resource is not cost-effective, with a benefit-cost ratio of 0.96.

Table 22. How Truncated Analysis Period Leads to End-Effects Problems

Resource Cost	\$1,000
Annual Benefit	\$80
Resource Life	20
Real Discount	
Rate	3%

A. Full Analysis Period (20 Years)—No End-Effects Problem

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	NPV
Cost	\$1000	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$1000
Benefit	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$1226
Net Be	nefit																				\$226
Benefit	-Cost	Ratic)																		1.23

B. Truncated Analysis Period (15 Years)—End-Effects Problem

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	NPV
Cost	\$1000	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0						\$1000
Benefit	\$80	\$80	\$8	\$8	\$8	\$8	\$8	\$8	\$80	\$80	\$80	\$80	\$80	\$80	\$80						\$984
			0	0	0	0	0	0													
Net Be	nefit																				(\$16)
Benefit	t-Cost	Ratio)																		0.98

11.4 Remedies for End-Effects Problems

The preferred remedy to an end-effects problem is to extend the analysis period to cover the full life of the efficiency resource whose installation is influenced by an efficiency program. However, if that is determined to be impractical, then a "second best"

⁴² Note that there can also be some O&M costs or cost savings that occur over the life of an efficiency resource. Use of a proper analysis period is important to accurately reflect the economic value of such O&M changes as well.

alternative is to account for only a portion of the costs of the measure (comparable to the portion of the benefits captured). A simple way to accomplish this is to amortize the costs over the life of the efficiency measure and then calculate the NPV of the resulting annualized costs. This is done over the same period that the NPV of the benefits of the measure are computed.

Table 23 illustrates the result of this approach, using the same assumptions as in the example in Table 22. Part A shows that amortizing costs produces the same NPV result as not amortizing costs when analyzing the full 20-year life of the resource. Part B shows that amortizing the cost in this way produces the same benefit-cost *ratio* under a truncated analysis period as under an analysis period long enough to capture impacts over the full life of the resource. However, the net benefits under this approach (\$181 in this example) are lower than under an analysis period that captures impacts over the full life of the resource (\$226 in this example). Thus, though this approach is clearly preferable to a truncated analysis that captures all of the resource costs and only some of the resource benefits, it is still better to extend the analysis period to cover the full life of the resource benefits, when possible.⁴³

Table 23. How Amortizing Costs to Align with Resource Life AmelioratesEnd-Effects Problems

A. Full Analysis Period (20 Years) with Cost Amortized over Resource Life

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	NPV
Cost	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$1000
Benefit	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$1226
Net Bene	əfit																				\$226
Benefit-0	Cost R	atio																			1.23

B. Truncated Analysis Period (15 Years) with Cost Amortized over Resource Life

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	NPV
Cost	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65	\$65						\$802
Benefit	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80	\$80						\$984
Net Bene	əfit																				\$181
Benefit-0	Cost R	atio																			1.23

⁴³ The difference in net benefits can be important if they are necessary to cover fixed program costs to make a program cost-effective (or to cover fixed portfolio costs to make a portfolio of programs cost-effective). For example, if the \$1,000 cost assumption in Table 22 and Table 23 was only a per unit efficiency measure cost, and if a program could lead to installation of 10,000 measures, the net benefits from the measures alone would be \$1.81 million under the "truncated analysis with costs amortized" approach (i.e., \$181 in net benefits per measure from Part B of Table 23 multiplied by 10,000). Thus, if fixed program costs were \$2.00 million, the program would appear to not be cost-effective under the "truncated analysis with costs amortized approach." However, it would be cost-effective if the analysis period covered the full life of the efficiency measures for which the net benefits of the measures would be more accurately calculated at \$2.26 million (the \$226 per measure from Table 22 Part A multiplied by 10,000 measures).

12. Analysis of Early Replacement

Early replacement occurs when a functioning piece of equipment is replaced with a more efficient model before it normally would have been replaced. This chapter provides guidance on how to analyze the costs and benefits of such early replacement efficiency measures.

12.1 Summary of Key Points

- Under cost-effectiveness tests that do not include participant impacts, the early replacement measure cost is simply the cost the utility incurs to promote the installation of the measure.
- Under cost-effectiveness tests that include participant impacts, the initial cost of an early replacement measure is partially offset by the benefit of deferring the replacement cost that would otherwise have been incurred several years later (i.e., by pushing the date on which the next replacement piece of equipment will have to be purchased much farther out into the future).
- The benefits of early replacement measures are partially a function of the efficiency of the equipment that would have been installed later in the baseline scenario. If the future baseline replacement efficiency is the same as that of the early replacement measure, there is simply one stream of benefits for just the duration of the early replacement period. In other instances, the early replacement measure is more efficient than the new equipment that would otherwise have been purchased in several years (the future baseline replacement efficiency). If this is the case, cost-effectiveness analysis should account for two different streams of impacts: one for the duration of the early replacement measure.

12.2 Overview

This section addresses why cost-effectiveness analysis of early replacement measures and programs requires special attention, as compared to other common measure categories.

Efficiency measures typically fall into one of four categories:

New Construction: in which a building is going to be constructed, and an efficiency program prompts developers, builders, or contractors to install more efficient products or use more efficient construction practices than they otherwise would have.

Time-of-Sale/Natural Replacement: in which a product is going to be sold and purchased, such as when an appliance breaks down and needs to be replaced, and an efficiency program is designed to persuade a vendor to sell and/or a customer to purchase a more efficient product than they otherwise would have.

Retrofit: in which efficiency programs incentivize customers to install new efficiency measures in an existing space, such as an un-insulated attic.

Early Replacement: in which an existing inefficient product is functioning and would not otherwise be replaced until a future year, and an efficiency program prompts a customer to replace it with a more efficient product sooner than he or she otherwise would have.

For the first three of those efficiency measure classifications, the cost impacts are commonly felt only in the first year (i.e., the incremental cost of an efficiency upgrade over a standard measure that would otherwise have been purchased or the full cost of a retrofit measure). The savings are thus simply the difference between the baseline efficiency and the new efficiency that will recur annually for the life of the measure.

Characterization of both the costs and savings of early replacement measures can be more complicated for two reasons:

- Early replacement changes the timing of costs relative to when they could be incurred in the baseline scenario (i.e., absent the early replacement)—at least in cases where a jurisdiction chooses to include participant costs and benefits; and
- That change in timing can lead to the need to account for multiple baseline assumptions (assumptions that change over time) for both costs and savings.

This section provides guidance on how to account for changes in the timing of costs, and accounting for multiple baselines for both costs and savings/benefits.

12.3 Accounting for Changes in the Timing of Costs

Under an early replacement scenario, there is the initial full cost of the replacement product. However, there are also potential cost savings from not having to buy the new product that would otherwise have been purchased several years into the future (depending on which categories of impacts are included in the cost-effectiveness test selected per guidance in Chapter 3).

Consider, for example, the following hypothetical early replacement scenario:

- The customer has a 10-year-old and still functioning heating system with a 70 percent efficiency rating, and the heating system is normally assumed to last 15 years;
- Absent an efficiency program influence, the customer is expected to replace its 10-year-old heating system in five years with a new 90 percent efficient model that will cost \$5,000;
- *With the efficiency program influence*, the customer decides to scrap its existing inefficient heating system and replace it today with a new 90 percent efficient model that costs \$5,000.

In this case, there would be only five years of savings from the early replacement. If the cost-effectiveness test includes participant impacts, the net cost of the efficiency resource is equal to the \$5000 initial cost of the early replacement *minus the NPV of the benefit of deferring a new purchase from the beginning of Year 6 to the beginning of Year 16.*⁴⁴ It is critically important that the reduction in cost associated with deferring the next new purchase be incorporated into cost-effectiveness analyses. To not account for

⁴⁴ Year 6 is when the customer would otherwise have had to buy a new replacement heating system; Year16 is when the customer will have to replace the new heating system that was just installed.

it would result in markedly overstating the costs of early replacement measures and programs.⁴⁵

Calculating the value of that deferral requires a cost amortization approach identical to that of minimizing the end-effects problems outlined in Chapter 11. This serves to align the mismatched timing of costs under the baseline condition and the early replacement condition, as illustrated in Table 24.

In short, the amortizing or annualizing of the different purchase times under the baseline and early replacement scenarios has the effect of lining up costs so that the only difference is five years of annualized costs under the early replacement scenario. (The annualized cost under the baseline and early replacement scenarios are the same in Years 6 through 20, cancelling each other out.) Importantly, that also aligns the cost analysis with the benefits analysis (i.e., both costs and benefits occur only in Years 1 through 5).

Table 24. Amortization to Address Mismatched Timing of Baseline andEarly Replacement Costs

Costs	\$5000	Savings	90%
Standard New Product Cost	\$5000	Standard New Product Efficiency	90%
Resource Life	15	Existing Efficiency	70%
Existing Product	5	Savings Annual Value (Years 1-5)	\$000
Remaining Life Real Discount Rate	5 3%	Savings Annual Value (Years 6 and Beyond)	\$0

A. Mismatched Timing of Costs Incurred under Baseline and Early Replacement Program Scenarios

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20
Baseline	-	-	-	-	-	\$5000	-	-	-	-	-	-	-	-	-	-	-	-	-	\$5000
Early Replace	\$5000	-	-	-	-	-	-	-	-	-	-	-	-	-	-	\$5000	-	-	-	\$5000

B. Net Costs and Benefits of Early Retirement Calculated through Cost Amortization

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	NPV
Costs																					
Baseline	-	-	-	-	-	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$4313
Early Replace	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$6231
Net	\$407	\$407	\$407	\$407	\$407	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	\$1918
Benefits	\$600	\$600	\$600	\$600	\$600	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	\$2830
Net Benefits	\$141	\$141	\$141	\$141	\$141	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	\$912
Benefit-Co	ost Ra	itio																		1.	.48

12.4 Accounting for Multiple Baselines for Both Costs and Savings

Unlike in the more straightforward example above, there can also be differences between the cost and efficiency of the early replacement measure that is installed today

⁴⁵ Again, this is only an issue if the cost-effectiveness test includes participant impacts. If it does not, the change in timing of costs associated with future equipment purchases is not relevant.

and the standard new product that would have otherwise been installed five years from now. For example, consider the following modifications to the hypothetical scenario outlined above:

- The customer has a 10-year-old and still functioning heating system with a 70 percent efficiency rating;
- This class of products is normally assumed to last 15 years, so absent an efficiency program influence, the customer is expected to replace its 10-year-old heating system in five years;
- The standard new heating system five years from now is expected to be an 85 percent efficient model that costs \$4500;
- Within 10 years, the standard new heating system is expected to be a 90 percent efficient model that costs \$5000;
- With the efficiency program influence, the customer opts to scrap its existing old inefficient heating system and replace it today with a new 90 percent efficient model that costs \$5000. The new model is not only more efficient than the old heating system it is replacing, but also more efficient than the new heating system the customer would have bought five years from now.

In this case, as depicted in the bottom of

Table 25, there would be five years of the same level of savings as assumed in the first hypothetical example depicted in Table 24 (i.e., the difference between the old 70 percent and the new efficient 90 percent efficient model). However, unlike in the Table 24 example, there would continue to be savings in Years 6 through 20, though the magnitude of those savings would be lower than in the first five years (i.e., the difference between a standard new 85 percent efficient model and an efficient new 90 percent efficient model). Thus, in the hypothetical example, the NPV of benefits is more than \$1300 greater (\$4140 vs. \$2830) than in the Table 24 example.

On the cost side of things, there would not only be a difference between no baseline cost and the amortized costs of the 90 percent efficient model for the first five years, but also a slightly higher amortized cost in the subsequent 15 years to reflect the difference in cost between a new 85 percent efficient model and a new 90 percent efficient model. Thus, in this hypothetical example, the NPV of costs is also greater—by over \$400 (\$2349 vs. \$1918)—than in the Table 24 example.

The net effect of these changes in costs and benefits is an increase in net benefits per measure of nearly \$900 (i.e., \$1791 vs. \$912) relative to the net benefits of the Table 24 example. It should be noted that the direction of this change is unique to this set of hypothetical assumptions. For example, if the cost of a new 85 percent efficient model in Year 6 was assumed to be \$3500 instead of \$4500 (with the 90 percent efficient model still costing \$5000), the net benefits would be virtually identical to those of the example in Table 24. If the 85 percent efficient model cost only \$2400 (with the 90 percent efficient efficient model still costing \$5000), the measure would actually fall below a 1.00 benefit-cost ratio.

Table 25. Amortization to Address Multiple Baselines for Savings andCosts of Early Replacement

CostsEfficiency Measure Cost\$Standard New Product Cost\$Resource Life\$Existing Product Remaining\$Life\$Real Discount Rate	5000 4500 15 5 3%	Savings Installed Measure Efficiency Standard New Product Efficiency Existing Efficiency Savings Annual Value (Years 1-5) Savings Annual Value (Years 6 and Beyond)	90% 85% 70% \$600 \$124
--	-------------------------------	---	-------------------------------------

A. Mismatched Timing of Costs Incurred under Baseline and Early Replacement Program Scenarios

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20
Baseline	\$0	\$0	\$0	\$0	\$0	\$4500	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$5000
Early Replace	\$5000	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$5000	\$0	\$0	\$0	\$5000

B. Net Costs and Benefits of Early Replacement

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	NPV
Costs																					
Baseline	-	-	-	-	-	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$3882
Early Replace	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$6231
Net Cost	\$407	\$407	\$407	\$407	\$407	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$2349
Benefits	\$600	\$600	\$600	\$600	\$600	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$4140
Net Benefits	\$193	\$193	\$193	\$193	\$193	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$1791
Benefit-Cost Ratio											1.76										

13. Free-Riders and Spillover

This chapter describes how to address free-riders and spillover effects in cost-effectiveness analyses, for those jurisdictions that focus on net savings for those analyses.

13.1 Summary of Key Points

In jurisdictions that focus on net savings for their cost-effectiveness analyses:

- The treatment of free ridership and spillover effects should be a function of the categories of impacts that a jurisdiction chooses to include in the cost-effectiveness test it adopts pursuant to the process outlined in Chapter 3.
- With regard to free riders:
 - Financial incentives paid to free riders are a cost only if the cost-effectiveness test excludes participant impacts; otherwise the value of the financial incentive to the participant offsets the cost of the financial incentive to the utility system. In other words, the net cost of free riders is zero under any test that includes participant impacts.
 - o No benefits from free riders should be included in any cost-effectiveness test.
- With regards to spillover:
 - There are no costs associated with spillover in jurisdictions whose costeffectiveness test includes only utility system impacts. Spillover should increase costs under tests that include participant impacts.
 - O Spillover increases benefits in every test.

Table 26 summarizes which categories of impacts are affected by free-rider and spillover effects, as further discussed below.

Category	Free-F	Riders	Spillover			
	Costs	Benefits	Costs	Benefits		
Utility System Impacts	Increase	n/a	n/a	Increase		
Participant Impacts	Decrease	n/a	Increase	Increase (if applicable)		
Other Impacts	n/a	n/a	Increase (if applicable)	Increase (if applicable)		
Total/Net Impact	Increase only if test <i>excludes</i> participant impacts; otherwise no net effect	No effect under any test	No increase if test includes only utility system impacts; otherwise an increase	Increase under every test		

Table 26. Categories of Impacts Affected by Free-Riders and Spillover

13.2 Applicability and Definitions

This section addresses the economic concepts underpinning how free-ridership and spillover effects should be treated in cost-effectiveness analyses in jurisdictions that choose to focus on net savings. This section does not address the relative merits of focusing on net savings versus focusing on gross savings, as that is beyond the scope of a guidance document focused solely on the construct and application of cost-effectiveness analysis. This section has no relevance to or application for cost-effectiveness analyses in jurisdictions that choose to focus on gross impacts.

Key definitions to consider in applying guidance from this section are as follows:

- **Free-ridership** refers to efficiency program savings that would have occurred in the absence of the program.⁴⁶
- Spillover refers to the installation of efficiency measures or adoption of efficiency practices by customers who did not directly participate in an efficiency program, but were nonetheless influenced by the program to make the efficiency improvement.⁴⁷
- **Gross program impacts** are impacts before or without any adjustments for freeridership and spillover.
- Net program impacts include adjustments for free-ridership and spillover.

13.3 Economic Treatment of Free-Rider Impacts

This section describes which free rider impacts should be included in cost-effectiveness analysis in jurisdictions that focus on net savings, given the categories of impacts that such jurisdictions include in their cost-effectiveness tests.

13.3.1 Utility System Impacts

<u>Benefits</u>: No utility system benefits associated with any savings achieved by free-riders should be included in cost-effectiveness analyses of an efficiency program because the program did not cause those benefits.

⁴⁶ There are three forms of free-ridership: (1) total free-riders—or efficiency program participants who would have installed the same efficiency measures at same time even if the program had not been run; (2) partial free-riders—or participants who would have made some, but not all, of the efficiency investments they made in the absence of the program; and (3) deferred free-riders—participants who would have made the same efficiency investments in the absence of the program, but at a later date (NREL 2014—see: http://www.nrel.gov/docs/fy14osti/62678.pdf).

⁴⁷ Spillover can take multiple forms, including both (1) participant spillover—or savings that were influenced by a customer's participation in efficiency program but were beyond those tracked by the program; and (2) non-participant spillover—or savings that were produced by customers who were influenced by a program even though they did not directly participate in it. Participant spillover can be further subdivided into savings that occur at the same site as savings from program participation (known as "inside spillover") and savings that occur at other sites (typically) owned or operated by the same customer (known as "outside spillover"). Participant spillover can also be subdivided into savings that are from measures or actions that are same as those that were recorded by the program (known as "like spillover") or from different kinds of efficiency measures (known as "unlike spillover"). (NREL 2014—see: http://www.nrel.gov/docs/fy14osti/62678.pdf)

<u>Costs</u>: Any financial incentives paid to free-riders should be treated as a utility system cost, because they are part of the overall cost to the utility of operating an efficiency program. For example, if a customer that receives a \$100 rebate from a utility efficiency program for an efficiency measure that it would have installed absent the program, the utility system has incurred a \$100 cost.

13.3.2 Participant Impacts

<u>Benefits</u>: No participant benefits associated with any savings achieved by free-riders should be included in cost-effectiveness analyses of efficiency programs because the participants would have achieved the same benefits absent the program.

<u>Costs</u>: Financial incentives paid to free-rider participants should be treated as a negative cost to participants because such participants would not have received any such financial support absent the program. This reduction in cost to participants cancels out the cost of free-riders to the utility system. Thus, under cost-effectiveness tests that include both utility system and participant impacts, the net cost of free-riders is zero.

Consider the example in subsection 13.3.1 in which a customer that receives a \$100 rebate from a utility efficiency program for an efficiency measure that it would have installed absent the program. As discussed in subsection 13.3.1, the \$100 is a utility system cost. Thus, if the jurisdiction's cost-effectiveness test included utility system impacts (as all tests must) but did not include participant impacts, there would be a net cost from the free-rider of \$100. However, that changes if the jurisdiction's cost-effectiveness test also includes participant impacts because \$100 cost to the utility system is offset by a \$100 benefit to the free-rider participant. Put another way, under a test that includes both utility system and participant impacts, the \$100 rebate is what is often called a transfer payment. It has distributional impacts—by moving money between customers—but no *net* cost to customers as a whole (which is the perspective that matters under cost-effectiveness tests that include participant impacts as well as utility system impacts).

13.3.3 Other Types of Impacts

<u>Benefits</u>: No other types of benefits associated with any savings achieved by free-riders (other fuel savings, water savings, environmental emission reductions, public health cost savings, poverty reduction, job creation, energy security, etc.) should be included in cost-effectiveness analyses of efficiency programs because they would have been realized absent the program as well.

<u>Costs</u>: Any other types of costs associated with efficiency investments by free-riders should not be included in cost-effectiveness analyses of efficiency programs because they would also have been incurred absent the program.

13.3.4 Summary of Economic Treatment of Free-Riders

Table 27 summarizes the proper economic treatment of free-rider costs and benefits for jurisdictions that focus on net (rather than gross) impacts.

Category	Free-Riders					
	Costs	Benefits				
Utility System Impacts	Increase	n/a				
Participant Impacts	Decrease	n/a				
Other Impacts	n/a	n/a				
Total/Net Impact	Increase only if test <i>excludes</i> participant impacts; otherwise no net effect	No effect under any test				

Table 27. Summary of Economic Treatment of Free Riders

13.4 Economic Treatment of Spillover Effects

This section describes what spillover impacts should be included in cost-effectiveness analysis in jurisdictions that focus on net savings, given the categories of impacts that such jurisdictions include in their cost-effectiveness tests.

13.4.1 Utility System Impacts

<u>Benefits</u>: All utility system benefits associated with spillover effects should be included in cost-effective analyses of an efficiency program because they were caused by the program.

<u>Costs</u>: There are no utility system costs directly associated with spillover effects because, by definition, investments made to produce spillover effects are not subsidized by efficiency programs (i.e., if a customer receives a rebate for installing a measure it is a program participant; spillover effects are produced when customers install measures without taking rebates or other program services).

13.4.2 Participant Impacts

<u>Benefits</u>: In jurisdictions that include participant impacts in their cost-effectiveness test, all spillover participant benefits associated with spillover effects should be included in cost-effectiveness analyses because such effects were caused by the efficiency programs being analyzed.

<u>Costs</u>: All spillover participant costs associated with spillover effects should be included in cost-effectiveness analyses because such effects were caused by the efficiency programs in question.

13.4.3 Other Types of Impacts

<u>Benefits</u>: In jurisdictions that include other types of impacts in their cost-effectiveness test (other fuel impacts, water impacts, environmental impacts, public health impacts, low-income impacts, job impacts, energy impacts, etc.), all other benefits associated with spillover effects should be included in cost-effectiveness analyses because such effects were caused by the efficiency programs under analysis.

<u>Costs</u>: All other types of costs associated with spillover effects should be included in cost-effectiveness analyses because such effects were caused by the efficiency programs under analysis.

13.4.4 Summary of Economic Treatment of Spillover Effects

Table 28 summarizes economic treatment of spillover costs and benefits.

Category	Spillover					
	Costs	Benefits				
Utility System Impacts	n/a	Increase				
Participant Impacts	Increase	Increase (if applicable)				
Other Impacts	Increase (if applicable)	Increase (if applicable)				
Total/Net Impact	No increase if test includes only utility system impacts; otherwise, an increase	Increase under every test				

Table 28. Summary of Economic Treatment of Spillover Effects

References

- Advanced Energy Economy Institute (AEEI). 2014. "Benefit-Cost Analysis for Distributed Energy Resources." T. Woolf et al., Synapse Energy Economics. http://info.aee.net/benefit-costanalysis-for-der-synapse.
- American Council for an Energy-Efficient Economy (ACEEE). 2006. "Valuation of Non-Energy Benefits to Determine Cost-Effectiveness of Whole-House Retrofit Programs: A Literature Review." A061. Prepared by J. Amann. May.
- ———. 2012. "A National Survey of State Policies and Practices for the Evaluation of Ratepayer-Funded Energy Efficiency Programs." U122. M. Kushler et al. http://aceee.org/researchreport/u122.
- ———. 2015. "Multiple Benefits of Business-Sector Energy Efficiency: A Survey of Existing and Potential Measures." IE1501. American Council for an Energy-Efficient Economy. C. Russell. http://aceee.org/research-report/ie1501.
- Analysis Group. 2016. "Value of 'DER' to 'D': The Role of Distributed Energy Resources in Local Electric Distribution System Reliability." Presented by Sue Tierney, CPUC Thought Leaders Forum. April 21.
- Arrow, K. 1996. "Discounting, Morality, and Gaming." Working Paper. Stanford University, Department of Economics. www.fsi.stanford.edu/publications/discounting morality and gaming.
- Avoided Energy Supply Costs Study Group. 2015. "Avoided Energy Supply Costs in New England: 2015 Report." Prepared by R. Hornby, A. Rudkevich, B. Schlensinger, S. Englander, J. Neri, J. Goldis, K. Amoako-Gyan, H. He, A. Rifas, and R. Tabors. April 3. http://www.ct.gov/deep/lib/deep/energy/aescinnewengland2015.pdf.
- Beitel, Annette, Tim Melloch, Bruce Harley, and Alejandra Mejia. 2016. "Technical Reference Manuals: Best Practices from Across the Nation to Inform the Creation of the California Electronic Technical Reference Manual (eTRM)." presented at the ACEEE Summer Study, August.
- California Public Utilities Commission (CPUC). 2001. "California Standard Practice Manual: Economic Analysis of Demand-Side Programs and Projects." www.cpuc.ca.gov/workarea/downloadasset.aspx?id=7741.
- ------. 2016. "Avoided Costs 2016 Interim Update." B. Horii et al., Energy and Environmental Economics, Inc. www.cpuc.ca.gov/WorkArea/DownloadAsset.aspx?id=12504.
- California Public Utility Commission Staff. 2017. "Distributed Energy Resources Cost Effectiveness Evaluation: Societal Cost Test, Greenhouse Gas adder, and Greenhouse Gas Co-Benefits," an Energy Division Staff Proposal. January.
- Ceres. 2012. "Practicing Risk-Aware Regulation: What Every State Regulator Should Know." Prepared by R. Binz, R. Sedano, D. Furey, and D. Mullen. www.raponline.org/wpcontent/uploads/2016/05/ceres-binzsedano-riskawareregulation-2012-apr-19.pdf.
- Consumers Union. 2016. "Show Me the Numbers: a Framework for Balanced Distributed Solar Policies." T. Woolf et al., Synapse Energy Economics. November 10.
- Dasgupta, Partha, Karl-Goran Maler, and Scott Barrett. 2000. "Intergenerational Equity, Social Discount Rates, and Global Warming, Revised." http://dlc.dlib.indiana.edu/dlc/bitstream/handle/10535/7979/Intergenerational%20Equity,% 20Social%20Discount%20Rates%20and%20Global%20Warming.pdf?sequence=1.

Delaware. 2009. Energy Conservation and Efficiency Act of 2009.

District of Columbia. 2008. Clean and Affordable Energy Act.

- E4TheFuture. 2016a. "Occupant Health Benefits of Residential Energy Efficiency." https://e4thefuture.org/wp-content/uploads/2016/11/Occupant-Health-Benefits-Residential-EE.pdf.
- ———. 2016b. "Energy Efficiency Jobs in America." with Environmental Entrepreneurs. https://e4thefuture.org/wpcontent/uploads/2016/12/EnergyEfficiencyJobsInAmerica_FINAL.pdf.
- Eversource Electric. 2017. "2016 Plan Year Report." Evaluated Results, Data Tables, Massachusetts DPU Docket DPU 17-100.
- Florida State Legislature. n.d. Florida Energy Efficiency and Conservation Act, 366.80-366.85 and 403.519.

http://www.leg.state.fl.us/Statutes/index.cfm?App_mode=Display_Statute&Search_String =&URL=0300-0399/0366/Sections/0366.80.html.

- Haeri, Hossein and S. Khwaja. 2013. "Valuing Energy Efficiency: The Search for a Better Yardstick." *Public Utilities Fortnightly*. July 2013.
- ICF International. 2016. "The Locational Value of Energy Efficiency on the Distribution Grid." ACEEE Summer Study. P. Mihlmester and S. Fine.
- Illinois Commerce Commission. 2014. "Commonwealth Edison 2014-2016 EE and Demand Response Plan." Docket No. 13-0495.
- Illinois General Assembly. 2009. *Public Power Act Amendment*. http://www.ilga.gov/legislation/publicacts/fulltext.asp?Name=096-0131.
- International Energy Agency. 2014. "Capturing the Multiple Benefits of Energy Efficiency."
- Lawrence Berkeley National Laboratory (LBNL). 2013. "Assessing Natural Gas Energy Efficiency Programs in a Low-Price Environment." LBNL-6105E. Clean Energy Program Policy Brief. I. Hoffman et al. https://emp.lbl.gov/sites/all/files/lbnl-6105e.pdf.
- ———. 2014. "Current Policies and Practices Related to the Incorporation of Non-Energy Benefits in Energy Saving Performance Contract Projects." E. Stuart, P. Larsen, C. Goldman, D. Gilligan. November 18.
- ———. 2016. "The Future of Electricity Resource Planning." LBNL-1006269. Future Electric Utility Regulation Report Series. Prepared by F. Kahrl, A. Mills, L. Lavin, and N. Ryan. www.emp.lbl.gov/sites/all/files/lbnl-1006269_1.pdf.
- multpl.com. 2017. "10 Year Real Interest Rate." Accessed May 1. www.multpl.com/10-year-realinterest-rate/.
- Michigan. 2016. State of Michigan Act No. 341Public Acts of 2016.
- National Action Plan for Energy Efficiency (NAPEE). 2007. "Guide to Resource Planning with Energy Efficiency." US EPA. www.epa.gov/sites/production/files/2015-08/documents/resource_planning.pdf.
 - ——. 2008. "Understanding Cost-Effectiveness of Energy Efficiency Programs: Best Practices, Technical Methods, and Emerging Issues for Policy-Makers." US EPA.
- National Grid. 2015. "Geo-Targeted DSM: Cost Effectiveness Methodology on a Local Scale." Dunsky Energy Consulting.
- 2016. "Benefit-Cost Analysis Handbook." Niagara Mohawk Power Corporation. Versions 1.0. Reforming the Energy Vision Proceeding. Case 14-M-0101.
- New York Public Service Commission (NY PSC). 2016a. "Order Establishing the Benefit Cost Analysis Framework."

http://documents.dps.ny.gov/public/MatterManagement/CaseMaster.aspx?MatterCaseNo =14-m-0101.

- NMR Group, Inc. 2011. "Massachusetts Program Administrators: Massachusetts Special and Cross-Sector Studies Area, Residential and Low Income Non-Energy Impacts (NEI) Evaluation," August 15, 2011.
- Northeast Energy Efficiency Partnerships (NEEP). 2014. "Cost-Effectiveness Screening Principles and Guidelines." Regional Evaluation, Measurement, and Verification Forum. T. Woolf et al., Synapse Energy Economics. http://www.neep.org/sites/default/files/resources/Forum_C-E_Screening_Guidelines_Final_No_2014.pdf.
- Northwest Electric Power Planning and Conservation Act. 1980. Vol. 94 STAT 2697. https://www.gpo.gov/fdsys/pkg/STATUTE-94/pdf/STATUTE-94-Pg2697.pdf.
- Northwest Power and Conservation Council. 2011. "Overview of the Council's Power Planning Methods." Council Document 2011-02. www.nwcouncil.org/media/29998/2011_02.pdf.
- Oak Ridge National Laboratory. 2002. "Non-Energy Benefits from the Weatherization Assistance Program: A Summary of Findings from the Recent Literature." M. Schweitzer and B. Tonn. ORNL/CON-484. April.
- 2014. "Weatherization Works: Summary of Findings from the Retrospective Evaluation of the U.S. Department of Energy's Weatherization Assistance Program." B. Tonn, et. al. ORNL/TM-2014/338. September.
- Oregon Public Utilities Commission (OR PUC). 1994. "In The Matter Of an Investigation Into The Calculation And Use Of Conservation Cost-Effectiveness Levels." Order No. 94-590,
- ———. 2014. "In the Matter of Energy Trust of Oregon Request for Approval of Exceptions to Cost Effectiveness Guidelines." http://apps.puc.state.or.us/orders/2014ords/14-332.pdf.
- Pindyck, R, and D Rubinfeld. 2001. *Microeconomics*. 5th ed. Upper Saddle River, NJ: Prentice Hall.
- Read, Daniel. 2004. "Intertemporal Choice." In *Blackwell Handbook of Judgment and Decision Making*, 424–443. Oxford: Blackwell Publishing. www.books.google.com/books?hl=en&lr=&id=s73eYI1DRHUC&oi=fnd&pg=PA424&dq=in fo:rynT3IDe1JIJ:scholar.google.com&ots=ndQAmDjmkm&sig=hXcXxr5DFkqCSOSX4uia EIH7dxY.
- Regulatory Assistance Project. 2011. "Valuing the Contribution of Energy Efficiency to Avoided Marginal Line Losses and Reserve Requirements." J. Lazar and X. Baldwin. http://www.raponline.org/wp-content/uploads/2016/05/rap-lazar-eeandlinelosses-2011-08-17.pdf.
- 2012. "Energy Efficiency Cost-Effectiveness Screening: How to Properly Account for 'Other Program Impacts' and Environmental Compliance Costs." T. Woolf et al., Synapse Energy Economics. http://www.synapse-

- ———. 2013a. "Widening Energy Regulators' Circles of Interest: State Examples of Energy Regulations and Processes That Consider Environmental Impacts," B. Hausauer & J. Shenot. January.
- ———. 2013b. "Best Practices in Electric Utility Integrated Resource Planning." R. Wilson and B. Biewald, Synapse Energy Economics. http://www.synapse-energy.com/project/best-practices-electric-utility-integrated-resource-planning.
- ———. 2013c. "Recognizing the Full Value of Energy Efficiency." J. Lazar and K. Colburn. http://www.raponline.org/wp-content/uploads/2016/05/rap-lazarcolburn-layercakepaper-2013-sept-9.pdf.

energy.com/sites/default/files/SynapseReport.2012-11.RAP_.EE-Cost-Effectiveness-Screening.12-014.pdf.

Rhode Island. 2014. Resilient Rhode Island Act.

- SEE Action. 2011a. "Analyzing and Managing Bill Impacts of Energy Efficiency Programs: Principles and Recommendations." DOE/EE-0926. https://www4.eere.energy.gov/seeaction/sites/default/files/pdfs/ratepayer_efficiency_billi mpacts.pdf.
- 2011b. "Using Integrated Resource Planning to Encourage Investment in Cost-Effective Energy Efficiency Measures." DOE/EE-0668. Driving Ratepayer-Funded Efficiency through Regulatory Policies Working Group. J. Shenot, Regulatory Assistance Project. https://www4.eere.energy.gov/seeaction/system/files/documents/ratepayer_efficiency_irp portfoliomanagement.pdf.
- Skumatz Economic Research Associates (SERA). 2006. "Evaluating Attribution, Causality, NEBs, and Cost-Effectiveness in Multifamily Programs: Enhanced Techniques." J. Meyers and L. Skumatz. 2006 American Council for and Energy-Efficient Economy Summer Study on Energy Efficiency in Buildings.
- ———. 2010. "Non-Energy Benefits: Status, Findings, Next Steps, and Implications for Low-Income Program Analyses in California." L. Skumatz, S. Khawaja, R. Krop. Prepared for Sempra Utilities. May, 11.
- ———. 2014. "Non-Energy Benefits/Non-Energy Impacts and Their Role & Values in Cost-Effectiveness Tests: State of Maryland." L. Skumatz. Prepared for Natural Resources Defense Council. March, 31.
 - http://energyefficiencyforall.org/sites/default/files/2014_%20NEBs%20report%20for%20M aryland.pdf.
- 2016. "Non-Energy Benefits / NEBs Winning at Cost-Effectiveness Dominos: State Progress and TRMs." L. Skumatz. 2006 American Council for and Energy-Efficient Economy Summer Study on Energy Efficiency in Buildings.
- Smart Electric Power Alliance (SEPA). 2016. "Beyond the Meter: Addressing the Locational Valuation Challenge for Distributed Energy Resources." Nexant.
- Tennessee Valley Authority. 2015. "Integrated Resource Plan: 2015 Final Report." www.tva.gov/file_source/TVA/Site%20Content/Environment/Environmental%20Stewards hip/IRP/Documents/2015_irp.pdf.
- TetraTech. 2012. "Final Report Commercial and Industrial Non-Energy Impacts Study." Prepared for the Massachusetts Program Administrators. June 29.
- The Mendota Group, LLC. 2014. "Benchmarking Transmission and Distributing Costs Avoided by Energy Efficiency Investments." Prepared for Public Service Company of Colorado. October, 23.
- US Department of Energy (US DOE). 2013. "A Framework for Evaluating the Cost-Effectiveness of Demand Response." Synapse Energy Economics with Regulatory Assistance Project.
- ———. 2016. "Home Rx: The Health Benefits of Home Performance." DOE/EE-1505. J. Wilson et al.

https://www.energystar.gov/sites/default/files/asset/document/Home%20Rx%20The%20 Health%20Benefits%20of%20Home%20Performance%20-%20A%20Review%20of%20the%20Current%20Evidence_FINAL.pdf.

- 2017. "U.S. Energy and Employment Report." January 2017. https://energy.gov/sites/prod/files/2017/01/f34/2017%20US%20Energy%20and%20Jobs %20Report_0.pdf.
- US Office of Management and Budget (US OMB). 2003. "Circular A-4 to the Heads of Executive Agencies and Establishments." Regulatory Analyses.
- Vermont Department of Public Service (VT DPS). 2011. "Comments in the Cost-Effectiveness Screening of Efficiency Measures Workshop Process Related to Non-Energy Benefits,

Discount Rate, Risk Adjustment, and Low-Income Adders," December 9. http://psb.vermont.gov/sites/psb/files/projects/EEU/screening/DPSCostEffectivenessScreeningComments12-9-11.pdf.

- 2014. "Rate and Bill Impacts of Vermont Energy Efficiency Programs." T. Woolf et al., Synapse Energy Economics. http://www.synapseenergy.com/sites/default/files/SynapseReport.2014-04.VT-PSD.VT-EE-Bill-Impacts.13-088.pdf.
- Vermont Public Service Board (VT PSB). 1990. "Vermont Public Service Board Decision in Docket No. 5270: Investigation into Least-Cost Investments, Energy Efficiency, Conservation and Management of Demand for Energy."
- 2014. "Order RE: Energy Efficiency Utility Budgets for Demand Resources Plan." State of Vermont Public Service Board. http://psb.vermont.gov/sites/psb/files/projects/EEU/drp2013/Budget%20Order.pdf.
- Washington State Legislature. 2006. Washington Energy Independence Act. Title 19 RCW 19.285. http://apps.leg.wa.gov/rcw/default.aspx?cite=19.285&full=true.

Appendix A. Traditional Cost-Effectiveness Tests

This appendix provides a description of the tests that are used for assessing EE costeffectiveness: the Utility Cost, Total Resource Cost, Societal Cost, Participant Cost, and Rate Impact Measure tests. While these tests are described in the California Standard Practice Manual, those descriptions are not clear for all purposes, and many jurisdictions have deviated from the tests described there. The descriptions below are intended to provide the theoretical underpinnings of *what should be included* in these tests, which might be different from *what is included* in these tests in practice.

A.1 Overview

This appendix provides information on the three commonly used traditional screening tests: the UCT (also known as the Program Administrator Cost Test); the TRC test; and the SCT.⁴⁸ As discussed in both the introduction to this manual and in Chapter 4, a jurisdiction using the Resource Value Framework could develop a primary cost-effectiveness test that fully aligns with one of these traditional tests—assuming they are appropriately applied according to the principles set forth in Chapter 2 of this NSPM. This appendix describes the key elements of these three traditional tests. Where necessary, users of this manual can cross-reference Chapter 4 with this appendix to help guide considerations of the relationship with the traditional cost-effectiveness tests.

For each of the traditional tests, this appendix provides:

- A description of the test;
- The relevance of the test for cost-effectiveness assessment;
- The costs and benefits covered under each test; and
- limitations of each test.

This appendix also briefly addresses the Participant Cost and Ratepayer Impact Measure tests, as defined by the CaSPM. However, as discussed below, neither the Participant test nor the RIM test are conceptually consistent with the core principles of cost-effectiveness analysis discussed in Chapter 1. Thus, neither is appropriate as a tool for resource investment choices (though they can provide information that is potential useful for other purposes, such as program design).

Table 29 provides a conceptual overview of the traditional cost-effectiveness tests. Table 30 provides a summary of the various costs and benefits that, to be consistent with the analytical perspective each test is intended to represent, should be included in these tests (although they are not always included in practice). Additional information on each test is provided in the sections that follow.

⁴⁸ While most jurisdictions have historically used the CaSPM as the foundation for their cost-effectiveness tests, in practice many jurisdictions have deviated from those tests.

Test	Perspective	Key Question Answered	Summary Approach
Utility Cost	The utility system	Will utility system costs be reduced?	Includes the costs and benefits experienced by the utility system
Total Resource Cost	The utility system plus participating customers	Will utility system costs plus program participants' costs be reduced?	Includes the costs and benefits experienced by the utility system, plus costs and benefits to program participants
Societal Cost	Society as a whole	Will total costs to society be reduced?	Includes the costs and benefits experienced by society as a whole.
Participant Cost	Customers who participate in an efficiency program	Will program participants' costs be reduced?	Includes the costs and benefits experienced by the customers who participate in the program
Rate Impact Measure	Impact on rates paid by all customers	Will utility rates be reduced?	Includes the costs and benefits that will affect utility rates, including utility system costs and benefits plus lost revenues

Table 29. Conceptual Overview of the CaSPM Cost-Effectiveness Tests

Table 30. Costs and Benefits of the CaSPM Cost-Effectiveness Tests

	UCT	TRC Test	SCT	Participant Cost Test	RIM Test
EE Costs:					
Efficiency Program Costs	Yes	Yes	Yes		Yes
Efficiency Portfolio Costs	Yes	Yes	Yes		Yes
Financial Incentive Provided to Participant	Yes	Yes	Yes		Yes
Participant Financial Cost of Efficiency		Yes	Yes	Yes	
Participant Non-Financial Cost of Efficiency		Yes	Yes	Yes	
Participant Increased Resource Consumption		Yes	Yes	Yes	
Societal costs (environmental, health, etc.)			Yes		
Lost Revenues					Yes
EE Benefits:					
Avoided Energy Costs	Yes	Yes	Yes		Yes
Avoided Generation Capacity Costs	Yes	Yes	Yes		Yes
Avoided T&D Capacity Costs	Yes	Yes	Yes		Yes
Avoided T&D Losses	Yes	Yes	Yes		Yes
Wholesale Market Price Suppression Effects	Yes	Yes	lf applicable		Yes
Avoided Environmental Compliance Costs	Yes	Yes	Yes		Yes
Avoided RPS Compliance Costs	Yes	Yes	Yes		Yes
Avoided Credit and Collection Costs	Yes	Yes	Yes		Yes
Participant Resource Savings (fuel, water)		Yes	Yes	Yes	
Participant Non-Resource Benefits		Yes	Yes	Yes	
Reduce Low-income Energy Burden			Yes		
Environmental Benefits			Yes		
Jobs and Economic Development Benefits			Yes		
Societal Health Care Benefits			Yes		
Increased energy security			Yes		
Customer Bill Savings				Yes	

Chapter 6 provides descriptions for the costs and benefits listed here.

A.2 Utility Cost Test

<u>Description</u>: The purpose of the UCT is to indicate whether the benefits of an EE resource will exceed its costs from the perspective of only the utility system. The UCT includes all costs and benefits that affect the operation of the utility system and the provision of electric and gas services to customers. For vertically integrated utilities, this test includes all of the costs and benefits that affect utility revenue requirements. For utilities that are not vertically integrated, this test includes all costs and benefits that affect utility revenue requirements, plus additional costs and benefits associated with market-based procurement of electricity and gas services. The UCT is sometimes referred to as the Program Administrator Cost test, to include those cases where ratepayer-funded EE programs are implemented by non-utility administrators. The UCT is a more accurate name because the costs and benefits included in this test are those that affect the utility system, not those that affect the Program Administrator.

<u>Relevance to EE Assessment</u>: The UCT is useful for identifying the impact of EE on utility system costs and average customer bills, and thus is consistent with the principle that EE is a resource. It is also useful for identifying the extent to which utility investments will provide reduced costs to that same overall group of utility customers, and therefore can have value (among other factors) for informing decisions on relative program priorities, program design (e.g., customer incentive levels) and/or limits on program spending. As discussed in Chapter 3, the UCT should serve as the foundation upon which a jurisdiction's efficiency assessment test is built. From this foundation, other relevant impacts should be added to align the test with the jurisdiction's energy-related policy goals.

<u>Costs Included</u>: The UCT should account for all utility system costs that are incurred to implement the EE resource. This includes all costs that the utility must recover from customers, including: financial incentives for efficiency measures, efficiency program costs, and efficiency portfolio costs.

<u>Benefits Included</u>: The UCT should account for all utility system costs that are avoided by the EE resource. For electricity utilities, this includes avoided energy costs, avoided generation capacity costs, avoided reserves, price suppression effects, avoided transmission costs, avoided distribution costs, avoided ancillary services costs, avoided T&D line losses, avoided environmental compliance costs, avoided RPS compliance costs, avoided credit and collection costs, and the value of reductions in risk and/or increases in system reliability. For gas utilities, this includes avoided gas commodity costs, avoided gas distribution costs, avoided gas storage costs, avoided gas distribution losses, avoided environmental compliance costs, the value of risk mitigation and/or increased reliability, and avoided credit and collection costs.

A.3 Total Resource Cost Test

<u>Description</u>: One of the key principles of cost-effectiveness assessment is that utility EE investments should be evaluated as a resource and compared with other demand-side and supply-side resources. The TRC does so from the combined perspective of the utility system and participants. Thus, this test includes all impacts of the UCT, plus all impacts on the program participants.

<u>Relevance to EE Resource Assessment</u>: The TRC test provides more comprehensive information than the UCT by including the impacts on participating customers. As a result, this test includes impacts on other fuels, which allows for a comprehensive assessment of multi-fuel programs and fuel-switching programs. This test also
conceptually includes other non-energy impacts on participants. This is particularly important for low-income programs.

<u>Costs Included</u>: This TRC test should account for all utility system and program participant costs incurred to implement the EE resource. This includes all costs described above for the UCT, plus any costs incurred by the program participant, including: financial cost to purchase efficiency measures; increased consumption of other fuels; increased O&M costs; and participant non-financial costs.

<u>Benefits Included</u>: This test should account for the utility system and program participant benefits that are experienced because of the EE resource. This includes all benefits described above for the UCT, plus any resources and benefits experienced by the program participant, including: other fuel savings, water savings, participant O&M savings, and all other participant non-resource benefits. The appropriate application of TRC requires that all such participant benefits are fully included in order to ensure symmetry with the inclusion of participant costs.

A.4 Societal Cost Test

<u>Description</u>: The purpose of the SCT is to indicate whether the benefits of an EE resource will exceed its costs from the perspective of society as a whole. This test provides the most comprehensive picture of the total impacts of an EE resource. This test includes all the impacts of the TRC test, plus the additional impacts on society. Note that the CaSPM refers to the SCT as a "variation" of the TRC test (CPUC 2001). Since then, many jurisdictions and many studies have referred to the SCT as a separate test with different implications.

<u>Relevance to EE Resource Assessment</u>: The SCT is useful for identifying the total universe of economic impacts of investment in EE resources. It is particularly apt for jurisdictions that have particular interest in a range of societal considerations, such as environmental or economic development concerns, in addition to an interest in minimizing utility system and efficiency program participant costs.

<u>Costs Included</u>: This test should account for all costs that are incurred to acquire the EE resource. This includes all costs described above for the TRC test, plus any costs incurred by society, including environmental costs and reduced economic development.

<u>Benefits Included</u>: This test should account for all of the benefits that result from the EE resource. This includes all benefits described above for the TRC test plus any benefits experienced by society, including: low-income community benefits, environmental benefits, economic development benefits, and reduced health care costs.

A.5 Participant Cost Test

<u>Description</u>: The intended purpose of this test is to indicate whether the benefits of an EE program will exceed its costs from the perspective of the EE program participant. This test includes all impacts on the program participants, but no other impacts.

<u>Relevance to EE Resource Assessment</u>: The Participant Cost test is not appropriate for assessing the value of EE as a resource because, unlike the other four tests described here, it values benefits based on avoided electricity and gas *rates* rather than on avoided utility system *costs*. That violates the fundamental principle that cost-effectiveness analysis should be "forward-looking" (see Chapter 1) because electric and gas rates are designed to recover both variable (i.e., avoidable) costs and fixed (unavoidable) costs,

some of which were incurred in the past. An example would be the cost of previous capital investments in the T&D system or generating capacity in vertically integrated utilities.⁴⁹

That said, the Participant test can have value for the purpose of informing efficiency program design (e.g., the level of financial incentives to offer prospective participants and/or the need for marketing to better inform participants of non-energy benefits that they may value) by providing insight into energy bill impact on participants.

Note that the US Department of Energy uses a different test to determine whether to include efficiency measures to participants in federally-funded weatherization assistance programs. It uses the savings-to-investment ratio; where the numerator is the present value of net savings in energy, water, non-fuel, or non-water operation and maintenance costs attributable to the proposed energy or water conservation measure, and the denominator is the present value of the cost of the proposed energy or water conservation measure.

A.6 Rate Impact Measure Test

<u>Description</u>: The purpose of this test is to indicate whether an EE resource will increase or decrease electricity or gas rates (i.e., prices). This test includes all of the costs and benefits of the UCT, plus estimates of the utility lost revenues created by EE programs. When regulators take steps to allow utilities to recover the lost revenues of EE programs, through rate cases, revenue decoupling, or other means, then the recovery of these lost revenues will create upward pressure on rates. If this upward pressure on rates exceeds the downward pressure from reduced utility system costs, then rates will increase, and *vice versa*.

<u>Relevance to EE Resource Assessment</u>: The RIM test should not be used for purpose of determining which efficiency resources are cost-effective—i.e., have benefits that exceed their costs—because, like the Participant test, it does not measure changes in net economic costs across a population; rather, it is a measure of distribution equity. Even in that context, the RIM test only considers one of the three factors regulators should consider when exploring distributional equity concerns: rate impacts, bill impacts, and efficiency program participation rates that affect the portion of customers who will experience net increases or decreases in their bills. See Appendix C for a more detailed discussion of how to more holistically conduct and assess the trade-offs associated with rate impacts.

⁴⁹ They may be "avoided" in part by participants, but typically only if a larger portion is then recovered by non-participants. Put another way, a portion of participant benefits is often just a shift in costs from one customer group (participants) to another (non-participants) rather than a true cost savings.

Table 31. Summary of the CaSPM Cost-effectiveness Tests

Test	Purpose	Relevance to EE Assessment
Utility Cost	Indicates the extent to which ratepayer-funded efficiency will reduce costs to that same group of ratepayers; provides a foundation for all efficiency assessment tests	To indicate the impact of efficiency on utility system cost and average customer bills; serves as a foundation for all efficiency assessment tests
Total Resource Cost	Provides a more comprehensive view of EE impacts than the UCT, including impacts of other fuels, which is helpful for multi-fuel programs, and impacts on EE program participants (if properly applied with symmetrical treatment of costs and benefits)	Indicates the total cost of efficiency, regardless of who pays for it
Societal Cost	Most comprehensive test, enabling an assessment of cost-effectiveness based on the universe of costs and benefits of efficiency resource investment	Indicates the full impact of efficiency on society
Participant Cost	Useful in program design, to inform appropriate participant incentives	Not relevant for cost-effectiveness screening
Rate Impact Measure	Indicates whether long-term rates will increase or decrease on average	No appropriate for cost-effectiveness assessment; see Appendix C

Appendix B. Costs and Benefits of Other Types of DERs

This NSPM should serve as a foundation for assessing the cost-effectiveness of DERs. There are, however, important ways in which other types of DERs might need to be treated differently from EE resources. These important DER-specific issues are beyond the scope of this NSPM, but should be addressed by each jurisdiction as they develop cost-effectiveness practices for DERs. This appendix presents an introductory overview of how the types and magnitudes of costs and benefits might differ between EE resources and DERs.

While this NSPM focuses on the assessment of utility EE resources, the core concepts can be applied to other types of utility resources as well. The cost-effectiveness principles described in Chapter 1 and the Resource Value Framework described in Chapter 2 can be used to assess the cost-effectiveness of supply-side resources or distributed energy resources—including EE, demand response, distributed generation, distributed storage, electric vehicles, and strategic electrification technologies.

With regard to DERs, the cost-effectiveness principles and the Resource Value Framework can be used as the foundation for assessing their cost-effectiveness. There are, however, important ways in which other types of DERs might need to be treated differently from EE resources. For example,

- Some costs and benefits of EE might not be applicable to other types of DER, and vice versa. Some of the costs and benefits of EE might have different magnitudes relative to other types of DERs, including time-varying differences and locational differences.⁵⁰
- The policy decision of whether and how to include participant impacts might be different for different types of DERs.
- The approach for addressing rate, bill, and participant impacts might be different for different types of DERs.
- Distributed generation resources can inject power into a distribution grid, while EE resources do not.
- In some jurisdictions, the policy goals supporting other types of DERs might be different from those supporting EE.

These important DER-specific issues are beyond the scope of this NSPM, but should be addressed by each jurisdiction as they develop cost-effectiveness practices for DERs.

This appendix presents an introductory overview of how the types and magnitudes of costs and benefits might differ between EE resources and DERs. The tables below provide an overview of the different types of costs and benefits associated with EE, demand response, distributed generation, and distributed storage. Many of the costs and benefits associated with DERs are the same or similar to those associated with EE. In some cases, however, DERs impose different types of costs or benefits.

⁵⁰ Appendix B provides a comparison of the costs and benefits of EE resources relative to those of other types of DERs.

Table 32 provides an overview of the types of costs and benefits that might be relevant to any type of DER. While most of these were described in Chapter 6, the table also includes some impacts that are not relevant to EE.

		Costs	Benefits	
		Measure costs (utility portion)		Avoided energy costs
		Other financial incentives	-	Avoided generation capacity costs
	Program costs	Other program and administrative costs		Avoided reserves or other ancillary services
		Evaluation, measurement, and		Avoided T&D system investment
sten		verification	Utility System	Avoided T&D line losses
ility Sy	Utility	Porformanco incontivos	Avoided Costs	Wholesale market price suppression
Ē	incentives	Penormance incentives		Avoided RPS or EPS compliance costs
	Integration	Interconnection costs		Avoided environmental compliance costs
	Distribution	Distribution system upgrades		Avoided credit and collection costs
	Capital			Reduced risk
		Measure costs (participant	Low Income	Reduced low-income energy burden
		portion)		Public health benefits
		Interconnection fees	Public	Energy security
-Utility	Participant	Annual O&M		Jobs and economic development benefits
Non	Costs	Costs Participant increased resource	Environmental	Environmental benefits
		consumption		Participant health, comfort, and safety
		Non-financial (transaction) costs	Benefits	Participant resource savings (fuel, water)

Table 32.	Relevant	Costs and	Benefits (of Distributed	Energy	Resources
	Itele vant	00313 4114	Denenta		LICIGY	103001003

This table is presented for illustrative purposes, and is not meant to be an exhaustive list.

Different types of DERs might also have different magnitudes for the same type of cost or benefit. For example, one of the core purposes of EE and distributed generation is to reduce energy consumption from the grid, thereby avoiding energy costs on the utility system. Demand response and storage, however, typically *shift* the timing of energy consumption and therefore tend to reduce capacity costs more than energy costs.

These differences are presented in the tables below using circle icons. The greater the shading of the circle, the more often the costs or benefits are typically associated with the resource.

Table 33 below shows the costs and benefits to the utility system typically associated with EE, demand response, distributed generation, and distributed storage.

		Energy Efficiency	Demand Response	Distributed Generation	Distributed Storage
Cos	ts				
	Measure costs (utility portion)		•	0	0
	Other financial incentives				
tem	Other program and administrative costs			•	
lity Syst	Evaluation, measurement, and verification				
Ūťi	Performance incentives	•	•	•	•
	Interconnection costs	0	0		
	Distribution system upgrades	0	0		
Ben	efits				
	Avoided energy costs				
	Avoided generation capacity costs				
	Avoided reserves or other ancillary services				
	Avoided T&D system investment				
System	Avoided T&D line losses				
Utility \$	Wholesale market price suppression				
	Avoided RPS or EPS compliance costs				
	Avoided environmental compliance costs		•		
	Avoided credit and collection costs	•	•	•	
	Reduced risk				

Table 33. Utility System Costs and Benefits of DERs

This table is presented for illustrative purposes and is not meant to be an exhaustive list.

One of the most notable differences between EE and other DERs is the potential for distributed generation and storage to impose additional distribution system capacity costs and integration costs on the utility system. EE simply reduces energy consumption, while distributed generation and storage often feed electricity into the grid. While low levels of distributed generation and storage are unlikely to impose additional costs on the system, beyond a certain level of penetration, utilities may need to invest in distribution system capacity upgrades. They may also incur integration costs to manage the presence of DERs on the system on a day-to-day basis. For example, system investments may be required to support voltage regulation, upgrade transformers, increase available fault duty, and provide anti-islanding protection (NREL 2013). Integration costs may include scheduling, forecasting, and controlling DERs, as well as procurement of additional ancillary services such as reserves, regulation, and fast-ramping resources.⁵¹

Table 34 provides an indication of the non-utility system costs and benefits associated with different types of DERs. One type of cost that differs from EE is interconnection fees for distributed generation and distributed storage.

⁵¹ The need to procure fast-ramping resources or reserves is due to both the inflexibility of many fossil-fired units and the variability of most renewable generation.

	-	Energy Efficiency	Demand Response	Distributed Generation	Distributed Storage
Cost	5 5			1	
	Measure costs (participant portion)				
۲.	Interconnection fees	0	0	•	
on-Utili	Annual O&M	0	0		
z	Participant increased resource consumption	•	•	•	•
	Non-financial (transaction) costs	O		0	0
Bene	fits			1	
	Reduced low-income energy burden	•	•	•	•
	Public health benefits				
Ð	Energy security				
Non-Utili	Jobs and economic development benefits				
	Environmental benefits				
	Participant health, comfort, and safety	•	0	0	0
	Participant resource savings (fuel, water)		0	0	0

Table 34. Non-Utility System Costs and Benefits of DERs

This table is presented for illustrative purposes and is not meant to be an exhaustive list.

Appendix C. Accounting for Rate and Bill Impacts

The Rate Impact Measure test is not appropriate for cost-effectiveness analyses for several reasons. Nonetheless, the impacts of EE resources on customer rates and bills is sometimes of great interest to regulators and other stakeholders. This appendix describes a better approach for assessing rate and bill impacts of EE resources through long-term independent assessments of rate impacts, bill impacts, and participation rates.

C.1 Multiple Factors Affecting Rate Impacts

Efficiency resources can affect electricity and gas rates in several ways. First, they will create <u>upward</u> pressure on rates as a result of (a) the recovery of efficiency program administration and implementation costs; and (b) the recovery of lost revenues resulting from EE programs.

Second, they will create <u>downward</u> pressure on rates as a result of avoided costs, including:

- reduced generation capacity costs
- reduced T&D costs, including reduced line losses;
- reduced environmental compliance costs;
- reduced utility credit and collection costs;
- reduced wholesale market prices from price suppression effects, in regions with wholesale electricity markets; and
- reduced average fuel costs, in regions without wholesale electricity markets, as a result of reducing the consumption of the marginal fuels.

The net impact of efficiency resources on electricity and gas rates will be a result of all these different factors combined. Some of these impacts (such as recovery of program costs, wholesale market price suppression effects, and reduced average fuel costs) might occur over the short term, while others (such as reduced generation, transmission, and distribution capacity costs) might occur over a longer time period.

Understanding the impact of lost revenues is essential to understanding the impact of efficiency resources on rates. Lost revenues are the main reason why efficiency resources can be highly cost-effective and yet still result in rate increases. An efficiency resource might pass the UCT, where the long-term utility system benefits are significantly greater than the long-term utility system costs, but still result in increased rates if the lost revenues are high enough. This is often the case in practice where many efficiency programs are cost-effective according to the UCT, but not according to the RIM test.⁵²

The recovery of lost revenues is one of the factors that distinguish the impacts of supplyside resources from those of EE resources (as well as all DERs). Supply-side resources do not create lost revenues, because they do not reduce customer consumption.

⁵² The only difference between the Utility Cost test and the RIM test is that the latter includes lost revenues as one of the costs of EE resources.

Therefore, an EE resource might be much more cost-effective than a supply-side resource, but still result in upward pressure on rates as a result of the lost revenues.

Furthermore, the timing and impact on rates due to the recovery of lost revenues will depend upon the frequency of utility rate cases. In the years in between utility rate cases, the base rates are typically not increased to allow for the recovery of lost revenues. Instead, the lost revenues will result in reduced earnings for the utility, all else being equal. However, in those cases where the utility has some form of a decoupling mechanism, rates will be adjusted between rate cases and utility earnings will not be affected by the lost revenues.

The RIM test was originally intended to indicate the impact on rates from EE resources (CPUC 2001, 13). However, this test does not provide useful information regarding efficiency resource cost-effectiveness, as described below.

C.2 Limitations of the Rate Impact Measure Test

One of the main limitations of the RIM test is that it does not provide useful information about what happens to rates as a result of efficiency resource investments. A RIM benefit-cost ratio of less than one indicates that rates will increase (all else being equal), but says little to nothing about the magnitude of the rate impact, in terms of the percent (or ϕ/kWh) increase in rates or the percent (or dollar) increase in bills. In other words, the RIM test results do not provide any context for utilities and regulators to consider the magnitude and implications of the rate impacts.

Another significant problem with the RIM test is that it typically does not result in the lowest cost to customers. Instead, it may lead to the lowest rates (all else being equal, and if the test is applied properly). However, achieving the lowest rates is not the sole or primary goal of efficiency resource assessment. Maintaining low utility system costs, and therefore low customer bills, often has priority over minimizing rates. For most customers, the size of the electricity bills that they must pay is more important than the rates underlying those bills.

In addition, a strict application of the RIM test can lead to perverse outcomes. The RIM test can lead to the rejection of significant reductions in utility system costs to avoid what may be insignificant impacts on customers' rates. For example, a particular efficiency program might offer hundreds of millions of dollars in net benefits under the UCT (i.e., net reductions in utility system costs), but be rejected as not cost-effective if it fails the RIM test. It may well be that the actual rate impact is likely to be so small as to be unnoticeable. Rejecting such large reductions in utility system costs to avoid *de minimus* rate impacts is not in the best interests of customers overall.

Another important problem with the RIM test is that it is not consistent with basic economic theory. The lost revenues from EE are not a new cost created by investments in efficiency resources. Price impacts from lost revenues are caused by the need to recover existing costs over fewer sales. These existing costs that would be recovered through rate increases are not caused by the efficiency resources themselves, they are caused by historical investments in supply-side resources that become fixed costs. In economic terms, these existing fixed costs are referred to as "sunk" costs. In economic theory, sunk costs should not be considered when assessing future investments because they are incurred regardless of whether the future investment is undertaken.

Furthermore, the RIM test results can be misleading. For an efficiency program with a RIM benefit-cost ratio of less than one, the net benefits (in terms of PV\$) will be negative. A negative net benefit implies that the investment will increase costs. However,

as described above, the costs that drive the rate impacts under the RIM test are not new incremental costs associated with efficiency resources. They are existing costs that are already in current electricity or gas rates. Any rate increase caused by lost revenues would be a result of recovering those existing fixed costs over fewer sales, not as a result of incurring new costs. However, efficiency planners frequently present their RIM test results as negative net benefits, implying that the efficiency resource will increase costs, when in fact it will not.

Finally, all electricity and gas resources can result in some form of cross-subsidy. Applying the RIM test to EE resources is inconsistent with how other electricity and gas resources are evaluated for cost-effectiveness.

C.3 Rate Impacts and Customer Equity

In general, efficiency resources will result in lower average customer bills, despite any increase in rates.⁵³ Those customers that participate in an efficiency program will typically experience lower bills, while those that do not participate may experience higher rates and therefore higher bills.⁵⁴ Therefore, the rate impacts of EE resources are not a matter of cost-effectiveness. Instead, they are a matter of customer equity; between customers who participate in efficiency programs and those who do not.

Another limitation of the RIM test is that it does not provide the specific information that efficiency planners and regulators need to assess the equity impacts of efficiency resources. In order to understand equity impacts, it is necessary to simultaneously assess (a) the impacts of efficiency resources on long-term average rates; (b) the impacts of efficiency resources on long-term average customer bills; (c) and the extent to which customers participate in efficiency resource programs (over time) and thereby experience lower bills.

Put another way, regulators and other policymakers need to be able to compare the magnitude of bill reductions to the participating customers against the magnitude of any rate and (therefore) bill increases to non-participating customers and the portion of customers expected to experience such adverse effects. The RIM test does not provide this essential information. It only assesses whether rates will go up or not. It does not divulge the magnitude of the increase; nor does it indicate how many customers will experience the impact as an increase in their bills.

Some of the problems of the RIM test stem from the fact that it attempts to combine cost-effectiveness issues and equity issues into a single calculation. It combines the lost revenues (which are historical, unavoidable costs that drive equity issues) with the resource costs and benefits (which are future, avoidable costs that drive cost-effectiveness issues). By combining cost-effectiveness and equity issues into a single

⁵³ This is not always the case. Many demand response programs can lead to reduced rates, because they involve very little lost revenue recovery. Some EE programs can lead to reduced rates, depending upon program costs, avoided costs, and lost revenue recovery.

⁵⁴ It is important to note that all customers experience some of the benefits of efficiency resources regardless of whether they participate in the programs. In particular, efficiency resources can reduce the need for new generation capacity, reduce wholesale capacity prices, reduce wholesale energy prices, reduce T&D costs, improve system reliability, reduce risk, and more. All of these benefits accrue to all customers. Nonetheless, it is also generally true that efficiency participants will experience greater benefits than non-participants, due to the immediate reduction in their electricity bills.

calculation, the RIM test actually conflates the two issues and provides results that are not meaningful for either one.

The solution to this problem is to undertake two separate analyses. The costeffectiveness analysis should account for all the future, avoidable costs and benefits, using the principles and concepts described in this manual. A separate rate impact and equity analysis can be used to assess the distributional impacts of the EE resource (US OMB 2003, 14), by analyzing the likely long-term impact on rates, bills, and customer participation.

C.4 A Better Approach for Analyzing Rate Impacts

A thorough understanding of the implications of efficiency rate impacts requires analysis of three important factors: rate impacts, bill impacts, and participation impacts.

- **Rate impacts** provide an indication of the extent to which rates for all customers might increase due to efficiency resources.
- **Bill impacts** provide an indication of the extent to which customer bills might be reduced for those customers that install efficiency resources.
- **Participation impacts** provide an indication of the portion of customers will that will experience bill reductions or bill increases. Participating customers will generally experience bill reductions while non-participants might see rate increases leading to bill increases.

Taken together, these three factors indicate the extent to which customers as a whole will benefit from efficiency resources, and also the extent to which efficiency resources may lead to distributional equity concerns. It is critical to estimate the rate, bill and participant impacts properly, and to present them in terms that are meaningful for considering distributional equity issues (SEE Action 2011a).

Rate Impact Estimates

Rate impact estimates should account for all factors that impact rates. This would include all avoided costs that might exert downward pressure on rates, as well as any factors that might exert upward pressure on rates. Any estimates of the impact of lost revenue recovery on rates should (a) only reflect collection of lost revenues necessary to recover fixed costs, and (b) only reflect the actual impact on rates according the jurisdiction's ratemaking practices.

Rate impacts should be estimated over the long term, to capture the full period of time over which the efficiency savings will occur. The study period should include all of the years in which efficiency resources are implemented, plus enough years to include the full measure lives of the last efficiency resources installed. This is necessary to capture the full effect of the downward pressure on rates from avoided generation, transmission, and distribution costs.

Rate impacts should also be put into terms that place them in a meaningful context, so that they can be properly considered and weighed by efficiency planners and regulators. For example, they should be put in terms of \not{e}/kWh impacts, dollars per month, percent of total rates, or percent of total bill.

Rate impacts can be markedly different across different customer types. Therefore, it may be necessary to analyze the rate impacts for different customer sectors. Conducting a rate impact analysis for every customer class is probably too burdensome and not

necessary. Instead, analyses can be conducted for key customer types such as residential, small commercial, and large commercial and industrial.

Bill Impact Estimates

Bill impact estimates should build upon the estimates of rate impacts. While rate impacts apply to every customer within a rate class, bill impacts will vary between participants and non-participants. Further, bill impacts will vary depending upon the type of efficiency program and the amount of efficiency savings from the program. For these reasons, it may be appropriate to estimate bill impacts by efficiency program, or at least the key efficiency programs.

As with rate impacts, bill impacts should be estimated over the long term, to capture the full period of time over which the efficiency savings will occur. The study period should include all of the years in which efficiency resources are implemented, plus enough years to include the full measure lives of the last efficiency resources installed. This is necessary to capture the full effect of the downward pressure on bills from avoided generation, transmission, distribution, and other costs collectively born by ratepayers.

As with rate impacts, bill impacts should also be put into terms that place them in a meaningful context, so that they can be properly considered and weighed by efficiency planners and regulators. For example, they should be put in terms of dollars per month or percent of total bill.

Participation Estimates

Participation estimates should be put in terms of participation rates, measured by dividing efficiency program participants by the total population of customers eligible for the program. Participation rates provide context and more meaningful information relative to a simple number of program participants. Participation rates can also be used to compare participation across programs, across utilities, and across jurisdictions.

Participation rates should be estimated for each year of efficiency resource implementation. They should be compared across several years to indicate the extent to which customers are participating in the programs over time. Participation in multiple programs and across multiple years should be accounted for, and the impacts of participation in multiple efficiency programs by the same customer should be accounted for to the extent possible.

If program participation information is not currently available, it should be collected as soon as possible, so that meaningful estimates can be developed in future years. This type of information is critical for assessing the customer equity issues, and hence the rate impact issues, of efficiency resources.

Many equity concerns driven by rate impacts can be mitigated or even eliminated by promoting widespread customer participation in efficiency programs. Program participation information can be used to ensure that most, and potentially all, customers eventually install efficiency resources of one form or another, and thereby experience net lower bills. Efficiency program administrators could be charged with the responsibility to identify those customers that do not install efficiency resources, and to find ways to reach those customers that have not yet implemented some form of efficiency measure.

C.5 Relationship to the Cost-Effectiveness Analysis

The efficiency resource assessment described in Chapter 3 should provide a comparison of the costs and benefits of certain EE resources. The rate and bill impact analysis should provide an indication of the rate, bill, participation, and equity impacts of those efficiency resources.

Regulators and efficiency planners may wish to consider both analyses to determine whether to invest ratepayer funds in those efficiency resources. This determination could include a qualitative comparison of the trade-offs between cost-effectiveness and rate impacts. For example, regulators and efficiency planners could assess whether any expected long-term rate impacts are warranted in light of the costeffectiveness results, the bill reductions, and the participation rates.

There is no bright line to determine how to balance these different impacts. Instead, this balance will need to be drawn by efficiency planners, ultimately with guidance and final approval of regulators.

Regulators and efficiency planners may choose to modify proposed efficiency programs or portfolios in order to strike a better balance between costeffectiveness and equity issues. As noted

Utilizing Rate, Bill, and Participant Information

A recent study in Vermont estimated that an aggressive, long-term efficiency strategy would produce an average 7 percent reduction in electric bills (net of rate increases) for the more than 95 percent of residential customers who would be expected to participate in programs. The corresponding average increase in bills would be 4–5 percent for the fewer than 5 percent of customers who would not participate (VT DPS 2014).

The Vermont Public Service Board concluded that the estimated rate impact on that portion of customers was acceptable in light of the reduction in bills for participants and the other benefits of EE (VT PSB 2014).

Decision-makers in different jurisdictions might reach different conclusions regarding whether that trade-off would be worth making. However, they cannot make informed decisions unless they see data in this way.

above, one option would be to expand efficiency programs to include more participants and mitigate equity concerns. Another option would be to shift priority from programs that have low participation rates to those that have higher participation rates.

Appendix D. Glossary of Terms

This manual uses several key terms that have specific meaning in the context of the concepts described here.

<u>Avoided costs</u>, refers to the costs of those electricity and gas resources that are deferred or avoided by the energy efficiency resources being evaluated for cost-effectiveness. The avoided costs are what make up the utility system benefits of EE resources.

<u>Distributed energy resources (DERs)</u>, refers to electricity and gas resources that are installed on customers' premises (behind the meter), often to improve customer consumption patterns. These include EE, demand response, distributed generation, storage, plug-in electric vehicles, and more.

<u>Energy efficiency resource</u>, refers to EE technologies, services, measures, or programs funded by, and promoted on behalf of, electricity and gas utility customers.

<u>Impacts</u>, refers to both the costs and the benefits of a supply-side or demand-side resource.

<u>Jurisdiction</u>, refers to states, provinces, utilities, municipalities, or other regions for which EE resources are planned and implemented.

<u>Primary cost-effectiveness test</u>, refers to the cost-effectiveness framework that a jurisdiction most relies upon when choosing the efficiency resources in which to invest ratepayer money.

<u>Regulators/decision-makers</u>, refers to institutions, agents or other decision-makers that are authorized to determine utility resource cost-effectiveness and funding priorities. Such institutions or agents include public utility commissions, legislatures, boards of publicly owned utilities, the governing bodies for municipal utilities and cooperative utilities, municipal aggregator governing boards, and more

<u>Regulatory perspective</u>, refers to the perspective of regulators or other decision makers that oversee efficiency resource investment choices. This perspective is guided by the energy and other applicable policy goals—whether in laws, regulations, organizational policies or other codified forms—under which they operate.

<u>Resource Value Framework</u>, refers to a series of seven steps that can guide any jurisdiction to develop its primary test for assessing EE (and other DERs) cost-effectiveness. The Framework embodies the key principles of cost-effectiveness analyses described in Chapter 1.

<u>Resource Value Test (RVT)</u>, refers to the primary cost-effectiveness test that a jurisdiction has developed using the Resource Value Framework. It embodies all of the key principles of cost-effectiveness analyses, and accounts for that jurisdiction's applicable policy goals.

<u>Utility system</u>, refers to all elements of the electricity or gas system necessary to deliver services to the utility's customers. For electric utilities, this includes, generation, transmission, distribution, and utility operations. For gas utilities, this includes transportation, delivery, fuel, and utility operations. This term refers to any type of utility ownership or management, including investor-owned utilities, publicly-owned utilities, municipal utility systems, cooperatives, etc.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 1 of 302

National Standard Practice Manual

For Benefit-Cost Analysis of Distributed Energy Resources

AUGUST 2020





Acknowledgements

This National Standard Practice Manual for Benefit-Cost Analysis of Distributed Energy Resources (NSPM or the manual) is a publication of the National Energy Screening Project (NESP). The NESP is represented by a stakeholder group of organizations and individuals working to update and improve cost-effectiveness screening practices for distributed energy resources.

This manual incorporates and expands upon the guidance from the 2017 *National Standard Practice Manual for Assessing Cost-Effectiveness of Energy Efficiency Resources*, which presented the NSPM Framework, fundamental benefit-cost analysis principles, and guidance specific to energy efficiency resources.

The NESP is coordinated by E4TheFuture. Funding for the development of this NSPM was made possible by E4TheFuture with support from the US Department of Energy's Weatherization and Intergovernmental Programs Office under Lawrence Berkeley National Laboratory Contract No. DE-AC02-05CH11231. The views and opinions of authors expressed herein do not necessarily state or reflect those of the United States Government or any agency thereof, or The Regents of the University of California.

This manual and related materials, including prior NSPM publications, are available at: www.nationalenergyscreeningproject.org/national-standard-practice-manual/.

Report Authors

This manual was prepared by the following project team:

- Synapse Energy Economics Tim Woolf (Project Lead), Courtney Lane, and Melissa Whited
- Energy Futures Group Chris Neme
- ICF Mike Alter and Steve Fine
- Pace Energy and Climate Center Karl Rábago
- Schiller Consulting Steven R. Schiller
- Smart Electric Power Alliance Kate Strickland and Brenda Chew

The Project was coordinated by Julie Michals at E4TheFuture, with support from Alaina Boyle. The manual was edited by Julie Michals and Tim Woolf.

Advisory Group

E4TheFuture and the project team would like to thank the following individuals for offering their insights and perspectives on the manual and/or participating in Advisory Group meetings. The individuals and their affiliations are listed for identification purposes only. Participation on the Advisory Group does not indicate support for this document in whole or in part.

Adam Scheer, Recurve Andy Satchwell, Lawrence Berkeley National Laboratory Beth Conlin, United States Environmental Protection Agency Christopher Budzynski, Exelon Utilities Courtney Welch, ESource Cyrus Bhedwar, Southeast Energy Efficiency Alliance Dan Cross-Call, Rocky Mountain Institute Dan Violette, Lumina Dana Lowell, MJ Bradley Danielle Sass Byrnett, National Association of Regulatory Utility Commissioners Deborah Reynolds, Washington Utilities and Transportation Commission Don Gilligan, National Association of Energy Service Companies Don Kreis, New Hampshire Consumer Advocate Elizabeth Titus, Northeast Energy Efficiency Partnerships Gregory Dierkers, United States Department of Energy Gregory Ehrendreich, Midwest Energy Efficiency Alliance Greg Wikler, California Efficiency Demand Management Council

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 3 of 302

Jack Laverty, Columbia Gas of Ohio Janet Gail Besser, Smart Electric Power Alliance Jennifer Morris, Illinois Commerce Commission Joe Cullen, Building Performance Association Johanna Zetterberg, United States Department of Energy John Agan, United States Department of Energy John Shenot, Regulatory Assistance Project Julia Dumaine, Connecticut Department of Energy and Environmental Protection Juliet Homer, Pacific Northwest National Lab Kara Podkaminer, United States Department of Energy Kara Saul Rinaldi, Building Performance Association Katherine Johnson, Johnson Consulting Lauren Gage, Apex Analytics Marie Schnitzer, National Grid Mohit Chhabra, Natural Resources Defense Counsel Nadav Enbar, Electric Power Research Institute Nick Dreher, Midwest Energy Efficiency Alliance **Olivia Patterson, Opinion Dynamics** Paula Carmody, Maryland Office of People's Counsel Phil Jones, Alliance for Transportation Electrification Rachel Gold, American Council for Energy Efficient Economy Ric O'Connell, Grid Lab Rick Gilliam, Vote Solar Rodney Sobin, National Association of State Energy Officials Ryan Katofsky, Advanced Energy Economy Sami Khawaja, Cadmus Steven Rymsha, Sunrun Todd Bianco, Rhode Island Public Utilities Commission Tom Stanton, National Regulatory Research Institute Wally Nixon, Arkansas Public Service Commission

TABLE OF CONTENTS

NSF	PM SUMN	1ARY	I
GLC	DSSARY		XVIII
ACF	RONYMS		xxıv
1.	INTROD	UCTION	1-1
	1.1	DISTRIBUTED ENERGY RESOURCES - OVERVIEW	
	1.2	AUDIENCE AND TERMINOLOGY	
	1.3	ROLE OF BENEFIT-COST ANALYSIS	
	1.4	SCOPE OF THIS MANUAL	
	1.5	How to Use this Manual	
PAF	RT I: THE M	ISPM BENEFIT-COST ANALYSIS FRAMEWORK	
2.	BENEFI	T-COST ANALYSIS PRINCIPLES	2-1
	2.1	SUMMARY OF KEY POINTS	
	2.2	THE IMPORTANCE OF PRINCIPLES	
	2.3	THE NSPM PRINCIPLES	
3.	DEVELO	PING BENEFIT-COST ANALYSIS TESTS	3-1
	3.1	SUMMARY OF KEY POINTS	
	3.2	Cost-Effectiveness Testing Perspectives	
	3.3	DEVELOPING A JURISDICTION-SPECIFIC TEST	
	3.4	JURISDICTION-SPECIFIC TESTS AND TRADITIONAL TESTS	
	3.5	DEVELOPING SECONDARY BCA TESTS	
PAF	RT II: DER	BENEFITS AND COSTS	
4.	DER BEI	NEFITS AND COSTS	4-1
	4.1	SUMMARY OF KEY POINTS	
	4.2	UTILITY SYSTEM IMPACTS	
	4.3	GAS UTILITY AND OTHER FUEL SYSTEM IMPACTS	
	4.4	Host Customer Impacts	
	4.5	SOCIETAL IMPACTS	
5.	CROSS-	CUTTING BENEFIT AND COST CONSIDERATIONS	5-1
	5.1	SUMMARY OF KEY POINTS	
	5.2	TEMPORAL IMPACTS	
	5.3	LOCATIONAL IMPACTS	
	5.4	INTERACTIVE IMPACTS	
	5.5	BEHIND-THE-METER VERSUS FRONT-OF-THE-METER IMPACTS	
	5.6	AIR EMISSION IMPACTS.	
	5.7	TRANSFER PAYMENTS AND OFFSETTING IMPACTS	
	5.8	VARIABLE RENEWABLE GENERATION IMPACTS	
	59	WHOLESALE MARKET REVENUES	5-15
	5.10	FREE-RIDERS AND SPILLOVER IMPACTS	
	5 11	DISCOLINT RATES	5-16

PART III: BENEFIT-COST ANALYSIS FOR SPECIFIC DER TYPES

6.	ENERG	EFFICIENCY RESOURCES	6-1
	6.1	SUMMARY OF KEY POINTS	
	6.2	INTRODUCTION	
	6.3	BENEFITS AND COSTS OF ENERGY EFFICIENCY RESOURCES	
	6.4	Key Factors that Affect Energy Efficiency Impacts	
	6.5	COMMON CHALLENGES IN ESTIMATING EE BENEFITS AND COSTS	
	6.6	LOST REVENUES AND RATE IMPACTS	6-11
7.	DEMAN	ID RESPONSE RESOURCES	7-1
	7.1	SUMMARY OF KEY POINTS	
	7.2	INTRODUCTION	
	7.3	BENEFITS AND COSTS OF DEMAND RESPONSE RESOURCES	
	7.4	KEY FACTORS THAT AFFECT DEMAND RESPONSE BENEFITS AND COSTS	
	7.5	COMMON CHALLENGES IN DETERMINING DR BENEFITS AND COSTS	
	7.6	LOST REVENUES AND RATE IMPACTS	
8.	DISTRIB	SUTED GENERATION RESOURCES	8-1
	8.1	SUMMARY OF KEY POINTS	
	8.2	INTRODUCTION	
	8.3	BENEFITS AND COSTS OF DG RESOURCES	
	8.4	FACTORS THAT AFFECT DISTRIBUTED GENERATION BENEFITS AND COSTS	
	8.5	COMMON CHALLENGES IN DETERMINING DG BENEFITS AND COSTS	
	8.6	LOST REVENUES AND RATE IMPACTS	8-12
9.	DISTRIB	UTED STORAGE RESOURCES	9-1
	9.1	SUMMARY OF KEY POINTS	
	9.2	INTRODUCTION	
	9.3	BENEFITS AND COSTS OF DISTRIBUTED STORAGE RESOURCES	
	9.4	Key Factors that Affect Distributed Storage Impacts	
	9.5	COMMON CHALLENGES FOR DETERMINING STORAGE BENEFITS AND COSTS	
	9.6	LOST REVENUES AND RATE IMPACTS	
10.	ELECTR	IFICATION	10-1
	10.1	SUMMARY OF KEY POINTS	10-1
	10.2	INTRODUCTION	
	10.3	BENEFITS AND COSTS OF ELECTRIFICATION	
	10.4	• Key Factors that Affect Electrification Impacts	10-7
	10.5	COMMON CHALLENGES IN ESTIMATING BENEFITS AND COSTS	10-14
	10.6	LOST REVENUES, INCREASED REVENUES, AND RATE IMPACTS	10-16
PAR	T IV: BEI	NEFIT-COST ANALYSIS FOR MULTIPLE DER TYPES	
11.	MULTIP	LE ON-SITE DERS	11-1
	11.1	SUMMARY OF KEY POINTS	
	11.2	INTRODUCTION	
	11.3	BENEFITS AND COSTS OF MULTIPLE ON-SITE DERS	
	11 4	KEY FACTORS THAT AFEECT BENEFITS AND COSTS	

 11.5 COMMON CHALLENGES IN DETERMINING BENEFITS AND COSTS.
 11-6

 11.6 LOST REVENUES AND RATE IMPACTS OF MULTIPLE ON-SITE DERS.
 11-7

	11.7	CASE STUDY: COMMERCIAL GRID-INTERACTIVE EFFICIENT BUILDING	
12.	NON-W	IRES SOLUTIONS	
	12.1	SUMMARY OF KEY POINTS	
	12.2		
	12.3	BENEFITS AND COSTS OF NON-WIRES SOLUTIONS	
	12.4	KEY FACTORS THAT AFFECT BENEFITS AND COSTS	
	12.5	COMMON CHALLENGES IN DETERMINING BENEFITS AND COSTS	
	12.6	LOST REVENUES AND RATE IMPACTS FOR NON-WIRES SOLUTIONS	
	12.7	Case Study: Non-Wires Solution	
13.	SYSTEM	-WIDE DER PORTFOLIOS	13-1
	13.1	SUMMARY OF KEY POINTS	
	13.2	INTRODUCTION	
	13.3	CONSISTENT COST-EFFECTIVENESS TESTS	
	13.4	INTERACTIVE EFFECTS	
	13.5	DER PLANNING OBJECTIVES	
	13.6	MULTIPLE TESTS	
	13.7	RATE IMPACTS OF DER PORTFOLIOS	
14.	DYNAM	IC SYSTEM PLANNING	14-1
	14.1	SUMMARY OF KEY POINTS	
	14.2	INTRODUCTION	
	14.3	INTEGRATED DISTRIBUTION PLANNING	
	14.4	INTEGRATED GRID PLANNING	
	14.5	BCA ISSUES AND CHALLENGES	
APP	ENDIX A.	RATE IMPACTS	A-1
	A.1	THE CAUSE OF RATE IMPACTS	A-1
	A.2	COST-EFFECTIVENESS AND RATE IMPACTS	A-2
	A.3	THE RATE IMPACT MEASURE TEST	A-3
	A.4	BETTER APPROACHES FOR ANALYZING RATE IMPACTS	A-5
	A.5	PRESENTATION OF RATE, BILL, AND PARTICIPATION IMPACTS	A-7
APP	ENDIX B.	TEMPLATE NSPM TABLES	В-1
APP	ENDIX C.	APPROACHES TO ACCOUNTING FOR RELEVANT IMPACTS	C-1
	C.1	INTRODUCTION	C-1
	C.2	MONETARY VALUES	C-2
	C.3	NON-MONETARY VALUES	C-5
	C.4	ALTERNATIVE THRESHOLDS	C-6
	C.5	TOOLS AND STUDIES TO SUPPORT ESTIMATION OF IMPACTS	C-7
APP	ENDIX D.	PRESENTING BCA RESULTS	D-1
	D.1	INTRODUCTION	D-1
	D.2	PRESENT VALUES OF BENEFITS AND COSTS	D-1
	D.3	NET BENEFITS	D-1
	D.4	BENEFIT-COST RATIO	D-2
	D.5	LEVELIZED COSTS	D-4
	D.6	NET LEVELIZED COSTS	D-7
	D.7	MULTIPLE COST-EFFECTIVENESS TESTS	D-10

APPENDIX E.	TRADITIONAL COST-EFFECTIVENESS TESTS	E-1
E.1	OVERVIEW	E-1
E.2	UTILITY COST TEST	E-3
E.3	TOTAL RESOURCE COST TEST	E-3
E.4	SOCIETAL COST TEST	E-4
E.5	Participant Cost Test	E-4
E.6	RATE IMPACT MEASURE TEST	E-5
APPENDIX F.	TRANSFER PAYMENTS AND OFFSETTING IMPACTS	F-1
F.1	SUMMARY OF KEY POINTS	F-1
F.2	INTRODUCTION AND TERMINOLOGY	F-2
F.3	CRITERIA FOR IDENTIFYING OFFSETTING IMPACTS	F-2
F.4	FINANCIAL INCENTIVES TO IMPLEMENT DERS	F-3
F.5	DER Performance Incentives	F-4
F.6	WHOLESALE MARKET PRICE EFFECTS	F-5
F.7	TAX INCENTIVES	F-7
APPENDIX G.	DISCOUNT RATES	G-1
G.1	SUMMARY	G-1
G.2	THE PURPOSE OF DISCOUNT RATES	G-1
G.3	COMMONLY USED DISCOUNT RATES	G-2
G.4	Risk Considerations	G-5
G.5	DETERMINING THE DISCOUNT RATE	G-6
APPENDIX H.	ENERGY EFFICIENCY: ADDITIONAL GUIDANCE	H-1
H.1	ASSESSMENT LEVEL	H-1
H.2	ANALYSIS OF EARLY REPLACEMENT	H-5
H.3	FREE-RIDERS AND SPILLOVER	H-10

REFERENCES

TABLE OF TABLES

Table S-1. NSPM BCA Principles	iv
Table S-2. Developing a Jurisdiction's Primary Test: A 5-Step Process	vi
Table S-3. Potential DER Impacts: Electric Utility System	viii
Table S-4. Potential Benefits and Costs of DERs: Host Customer	ix
Table S-5. Potential Costs and Benefits of DERs: Societal	ix
Table S-6. Potential Benefits and Costs: Electric Utility System	xi
Table S-7. Potential Benefits and Costs of DERs: DER Host Customer	xi
Table S-8. Potential Benefits and Costs of DERs: Societal	xii
Table S-9. Types of Dynamic System Planning Practices	xvii
Table S-10. Guide to Appendices	xvii
Table 1-1. Types of Planning Practices	1-5
Table 1-2. Overview of the National Standard Practice Manual for DERs	1-8
Table 2-1. Fundamental NSPM BCA Principles	2-3
Table 3-1. Developing a Jurisdiction's Primary Test: A 5-Step Process	3-4
Table 3-2. Examples of Policy Goals	3-5
Table 3-3. Commonly Considered Non-Utility System Impacts	3-7
Table 3-4. Different Approaches to Account for All Relevant Impacts	3-11
Table 3-5. Jurisdiction-Specific Test Compared with Traditional Tests	3-14
Table 3-6. Examples of Secondary tests Used to Consider Marginal DERs	3-18
Table 4-1. Potential DER Benefits and Costs: Electric Utility System	4-2
Table 4-2. Potential Benefits and Costs of DERs: Gas Utility or Other Fuel Impacts	4-12
Table 4-3. Potential Benefits and Costs of DERs: Host Customer	4-15
Table 4-4. Treatment of Host Customer Tax Incentives in Different Tests	4-18
Table 4-5. Potential Host Customer Non-Energy Impacts	4-18
Table 4-6. Potential Benefits and Costs of DERs: Societal	4-20
Table 5-1. DER Impacts Typically Affected by Temporal Variation	5-2
Table 5-2. DER Impacts Typically Affected by Locational Variation	5-4
Table 5-3. Potential Offsetting Impacts	5-14
Table 5-4. Summary of Economic Treatment of Free-Riders	5-16
Table 5-5. Considerations for Determining a Discount Rate	5-17
Table 6-1. Potential Impacts of Energy Efficiency: Electric Utility System	6-3
Table 6-2. Potential Impacts of Energy Efficiency: Gas Utility or Other Fuel System	6-4
Table 6-3. Potential Impacts of Energy Efficiency: Host Customer	6-5
Table 6-4. Potential Impacts of Energy Efficiency: Societal	6-5

Table 6-5. Potential Host Customer Non-Energy Impacts of Energy Efficiency	6-8
Table 7-1. Potential Impacts of Demand Response: Electric Utility System	7-5
Table 7-2. Potential Impacts of Demand Response: Gas Utility or Other Fuel System	7-6
Table 7-3. Potential Impacts of Demand Response: Host Customer	7-6
Table 7-4. Potential Impacts of Demand Response: Societal	7-7
Table 7-5. Potential Host Customer Non-Energy Impacts of Demand Response	7-9
Table 8-1. Potential Impacts of Distributed Generation: Electric Utility System	8-3
Table 8-2. Potential Impacts of Distributed Generation: Gas Utility or Other Fuel System	8-4
Table 8-3. Potential Impacts of Distributed Generation: Host Customer	8-4
Table 8-4. Potential Impacts of Distributed Generation: Societal	8-5
Table 8-5. Potential Host Customer Non-Energy Impacts of Distributed Generation	8-7
Table 9-1. Potential Impacts of Distributed Storage: Electric Utility System	9-3
Table 9-2. Potential Impacts of Distributed Storage: Gas Utility or Other Fuel System	
Table 9-3. Potential Impacts of Distributed Storage: Host Customer	
Table 9-4. Potential Impacts of Distributed Storage: Societal	
Table 9-5. Cost-Effectiveness Analysis of Storage Use Cases	9-6
Table 9-6. Potential Host Customer Non-Energy Impacts of Distributed Storage	9-9
Table 10-1. Potential Impacts of Electrification: Electric Utility System	10-3
Table 10-2. Potential Impacts of Electrification: Gas Utility and Other Fuel System Impacts	10-5
Table 10-3. Potential Impacts of Electrification: Host Customer	10-6
Table 10-4. Potential Impacts of Electrification: Societal	10-6
Table 10-5. Potential Host Customer Non-Energy Impacts of Electrification and Electric Vehicles	10-11
Table 11-1. Net Benefits and Costs of GEB Case Study: Utility System Impacts	11-10
Table 11-2. Net Benefits and Costs of GEB Case Study: Host Customer Impacts	11-10
Table 11-3. Net Benefits and Costs of GEB Case Study: Societal Impacts	11-11
Table 12-1. Net Benefits and Costs of NWS Case Study: Utility System Impacts	12-9
Table 12-2. Net benefits and Costs of NWS Case Study: Host Customer Impacts	12-10
Table 12-3. Net Benefits and Costs of NWS Case Study: Societal Impacts	12-10
Table 13-1. Example DER Portfolio Developed to Encourage a Diverse Range of Technologies	13-8
Table 13-2. Example DER Portfolio Developed Using Multiple Planning Objectives	13-12
Table 13-3. Example DER Portfolios Developed Using Different Planning Objectives	13-13
Table 14-1. Types of Planning Practices	14-2
Table A-1. Comparison of Cost-Effectiveness Analysis and Rate Impact Analysis	A-3
Table C-1. Different Approaches to Account for Relevant Impacts	C-1
Table E-1. Conceptual Overview of the Traditional Cost-Effectiveness Tests	E-2

Table E-2. Impacts Included in the Traditional Cost-Effectiveness Tests	E-2
Table E-3. Summary of the Traditional Cost-Effectiveness Tests	E-6
Table F-1. Potential Offsetting Impacts	F-1
Table F-2. Financial Incentives to Participate in DER programs	F-4
Table F-3. DER Performance Incentives	F-5
Table F-4. Treatment of Wholesale Price Effects	F-6
Table F-5. Treatment of Tax Incentives	F-8
Table G-1. Discount Rate Options for Cost-Effectiveness Analyses	G-4
Table G-2. Considerations for Determining a Discount Rate	G-7
Table H-1. Proper Analysis with 25 Percent Fixed Portfolio Costs Included at Portfolio-Level Analysis	H-5
Table H-2. Improper Analysis with 25 Percent Fixed Portfolio Costs Allocated to Individual Programs	H-5
Table H-3. Amortization to Address Mismatched Timing of Baseline and Early Replacement Costs	H-7
Table H-4. Amortization to Address Multiple Baselines for Early Replacement	H-10
Table H-5. Categories of Impacts Affected by Free-Riders and Spillover	H-11
Table H-6. Summary of Economic Treatment of Free-Riders	H-13
Table H-7. Summary of Economic Treatment of Spillover Effects	H-14

TABLE OF FIGURES

Figure S-1. The Regulatory Perspective	V
Figure S-2. Interactive Effects in Grid-Interactive Efficient Building	xiii
Figure S-3. Illustrative Example of NWS Cost-Effectiveness	xv
Figure S-4. Example DERs Sorted by Net Benefit	xvi
Figure 1-1. Process of Conducting BCA for DERs	1-7
Figure 1-2. The NSPM Benefit-Cost Analysis Framework	1-1
Figure 3-1. Cost-Effectiveness Testing Perspectives	3-3
Figure 3-2. Example Jurisdiction-Specific Test Relative to Traditional Tests	3-15
Figure 3-3. Example Use of Secondary Tests for Marginally Cost-Effective DERs	3-18
Figure 5-1. Example of Temporal Impacts on Energy Efficiency Benefits	5-3
Figure 5-2. Example of Locational Impacts on Demand Response Benefits	5-4
Figure 5-3. Example of Interactive Effects on DPV and Storage	5-9
Figure 5-4. Example of Interactive Effects on Energy Efficiency and Demand Response Resources	5-10
Figure 5-5. Hypothetical Example of BTM Versus FOM for a Distributed Storage Resource	5-12
Figure 7-1. Demand Response Classifications, Purposes, and Example Programs	7-4
Figure 11-1. Grid-interactive Efficient Building Flexibility Load Curves	11-3
Figure 11-2. Grid-interactive Efficient Building: Load Curves	11-3
Figure 11-3. Example of Grid-Interactive Efficient Building Cost-Effectiveness	11-11
Figure 11-4. Example of GEB Interactive Effects	11-12
Figure 11-5. PGE 2019 Distributed Resources & Flexible Load Study – Interactive Effects Addressed	11-13
Figure 12-1. Example of NWS Cost-Effectiveness	12-11
Figure 12-2. Example of NWS Locational Benefits	12-12
Figure 13-1. Example DERs Sorted by Net Benefits	
Figure 13-2. GHG Reduction Net Cost Curve	13-6
Figure 13-3. Example Presentation of Long-Term Rate Impacts	13-9
Figure 13-4. Example Presentation of Long-Term Average Combined Bill Impacts	13-10
Figure 13-5. Example Presentation of Long-Term Participation Impacts	13-10
Figure 13-6. Presentation of Multiple BCA Tests: Benefit-Cost Ratios, Multiple DER Types	13-14
Figure 14-1. Comparison of Traditional Distribution Planning to Integrated Distribution Planning	14-3
Figure 14-2. Integrated Grid Planning: Expanding IDP to Include IRP and Transmission Planning	14-5
Figure A-1. Example Presentation of Long-Term Rate Impacts	A-7
Figure A-2. Example Presentation of Long-Term Average Combined Bill Impacts	A-8
Figure A-3. Example Presentation of Long-Term Participation Impacts	A-8
Figure D-1. Example Presentation of Costs, Benefits, and Net Benefits: Hypothetical EE Resources	D-2

Figure D-2. Example Presentation of Benefit-Cost Ratios: Hypothetical Multiple DERs	D-3
Figure D-3. Example Presentation of Levelized Cost of Energy: Hypothetical EE Resources	D-4
Figure D-4. Example Presentation of Levelized Cost of Capacity: Multiple DERs	D-5
Figure D-5. Levelized Cost of GHG Reduction: Multiple Hypothetical DERs	D-6
Figure D-6. Example of GHG Cost Curve: Hypothetical Set of Multiple DER Types	D-7
Figure D-7. Example of GHG Net Cost Curve: Hypothetical Set of Multiple DER Types	D-9
Figure D-8. Presentation of Multiple BCA Tests: Benefit-Cost Ratios—Hypothetical EE Portfolio	D-10
Figure D-9. Presentation of Multiple BCA Tests: Benefit-Cost Ratios—Multiple DER Types	D-11
Figure G-1. Implications of Discount Rates (annual present value dollars)	G-2

NSPM SUMMARY

The purpose of this *National Standard Practice Manual for Benefit-Cost Analysis of Distributed Energy Resources* (NSPM, or the manual) is to help guide the development of jurisdictions' cost-effectiveness test(s) for conducting benefit-cost analyses (BCAs) of distributed energy resources (DERs). BCAs involve a systematic approach for assessing the cost-effectiveness of investments by consistently and comprehensively comparing the benefits and costs of individual or multiple types of DERs with each other and with alternative energy resources.

This manual includes information for conducting BCAs of single and multiple types of DERs and provides use case examples that illustrate BCAs under different combinations and applications of DERs. The DER types covered in this manual are: energy efficiency (EE); demand response (DR); distributed generation (DG); distributed storage (DS); electric vehicles (EV); and increased electrification of buildings including heating and cooling systems.

Distributed Energy Resources (DERs) are resources located on the distribution system that are generally sited close to or at customers' facilities. DERs include EE, DR, DG, DS, EVs, and increased electrification of buildings. DERs can either be on the host customer side of the utility interconnection point (i.e., behind the meter) or on the utility side (i.e., in front of the meter). DERs are mostly associated with the electricity system and can provide all or some of host customers' immediate power needs and/or support the utility system by reducing demand and/or providing supply to meet energy, capacity, or ancillary services (time and locational) needs of the electric grid.

DERs represent a critical component of the evolution of the electricity grid by allowing for a more flexible grid, enabling two-way flows of energy, enabling third parties to introduce and sell new electricity products and services, and empowering customers to optimize their end-uses and consumption patterns to lower their bills and utility costs.

This manual is built around a BCA framework (the NSPM BCA Framework) that defines the steps a jurisdiction can use to develop its primary cost-effectiveness test—the Jurisdiction-Specific Test (JST). The framework also provides guidance on how consider and develop secondary tests, where applicable. The NSPM BCA Framework includes a set of core principles that are the foundation for developing and applying cost-effectiveness tests for BCAs.

The NSPM is policy-neutral in that it does not recommend any specific cost-effectiveness tests or policies, but rather supports BCA practices that align with a jurisdiction's policy goals and objectives. The manual thus serves as an objective, technology-neutral and economically sound

guidance document for regulators, utilities, consumer advocates, DER proponents, state energy offices, and other stakeholders interested in comprehensively assessing the impacts of DER investments.

This manual incorporates and expands upon the guidance from the 2017 *NSPM for Assessing Cost-Effectiveness of Energy Efficiency Resources* (NSPM for EE). Both documents are products of the National Energy Screening Project (NESP), a multi-year effort guided by an advisory group represented by a range of experts with varying perspectives involved in BCA of DERs.

This NSPM provides objective, policy- and technology-neutral, and economically sound guidance for developing jurisdiction-specific approaches to benefit-cost analyses of distributed energy resources.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 14 of 302

Terminology and Applicability of the NSPM

This manual uses many terms that are commonly used within the electricity and gas industries. Key terms are defined in a Glossary and in relevant sections of the manual. Some of the terms used in the manual are more broadly defined than in other applications, as noted below.

NSPM Terminology

Jurisdiction refers broadly to any region or service territory that would be served by the DERs being analyzed. This includes a state, a province, a utility service territory, a city or a town, or some other jurisdiction covered by regulators or other entities that oversee DER initiatives.

Utility refers broadly to any entity that funds, implements, or supports DERs using customer or public funds that are overseen by regulators or other decision-makers. This includes investor-owned utilities; publicly owned utilities (e.g., municipal or cooperative utilities); program administrators; community choice aggregators; regional transmission organizations and independent system operators; federal, state, and local governments; and others. *Utility expenditures* refers to spending by any of these entities on DERs.

Regulator refers broadly to any entity that oversees and guides DER analyses. This includes legislators and their staff; public utility commissions and their staff; boards overseeing public power authorities, municipal or cooperative utilities, or regional grid operators; and federal, state, and local governments.

Host customer refers to any customer that has a DER installed and/or operated on their site. In some cases, these are program participants (such as in a DR or EE program) while in other cases there is no program (such as with EV owners).

Third parties refer to the broad range of independent providers such as aggregators or implementation, service, or technology providers.

The principles and concepts presented in this manual are relevant to:

- 1. DER programs, procurements, or pricing mechanisms associated with expenditures on behalf of the public or utility customers, whether by utilities or others. For simplicity, these are referred to these as 'utility expenditures.'
- 2. Any jurisdiction where DERs are funded, acquired, or otherwise supported by electric or gas utilities or others on behalf of their customers.
- 3. All types of electric and gas utilities, including investor-owned and publicly owned utilities (e.g., municipal or cooperative utilities.)
- 4. All types of utilities, including utilities that are vertically integrated, transmission and distribution (T&D), or distribution-only utilities, or those serving as a distribution platform for host customers to access a variety of energy services and DERs from third parties (e.g., aggregators).
- 5. Single DER and multiple DER BCA analyses, where:
 - *Single-DER analyses* involve assessing *one DER type* in isolation from other DER types, relative to a static set of alternative resources.
 - Multiple-DER analyses involve assessing more than one DER type at the same time relative to a static or dynamic set of alternative resources. Multiple-DER analyses covered in this manual include multiple on-site DERs, non-wires solutions within a specific geographic area, and system-wide DER portfolios.

 Dynamic system planning involves assessing multiple DER types relative to a dynamic set of alternative resources. Under this approach, the goal is to optimize both DERs and alternative utility-scale resources as well. This practice is relatively nascent and still evolving.

While the NSPM addresses BCA for single and multi-DER scenarios, it does not address every nuance or application for DER investments.

Manual Contents

The NSPM includes five parts:

- Part I presents the NSPM BCA Framework, including fundamental principles and guidance on the development of primary and any secondary cost-effectiveness tests.
- Part II describes the full range of potentially relevant DER benefits and costs (i.e., impacts), and presents several cross-cutting considerations on how to account for certain impacts.
- Part III provides guidance on single-DER BCA for various types of DER technologies. These chapters provide guidance on key factors and challenges that affect the impacts of each DER type.
- Part IV provides guidance on multiple-DER analysis. It addresses the three main ways that multiple-DER analysis is conducted: for a customer site; for a geographic region; and for an entire utility service territory. Part IV also addresses, at a high level, dynamic system planning.
- Appendices provide further detail on topics that warrant additional explanation. The appendices also provide information and templates on reporting BCA results.

Part I: The NSPM BCA Framework

Part I presents the NSPM BCA Framework, comprising three elements:

- 1. A set of **fundamental principles** that serve as the foundation for assessing the cost-effectiveness of potential DER investments in an economically sound and policy-neutral manner;
- 2. A **multi-step process** for developing or informing a jurisdiction's primary test—the Jurisdiction-Specific Test (JST)—as guided by the NSPM principles; and
- 3. Guidance on **when and how to use secondary tests** to inform (a) the prioritization of cost-effective DERs, as determined by a primary JST, and (b) decisions around marginally non-cost-effective DERs.

The **NSPM principles** in and of themselves do not determine a jurisdiction's appropriate costeffectiveness test for DERs. The NSPM principles are intended to be applied in a manner that takes into consideration the characteristics and circumstances of each jurisdiction's approach to energy resources and can result in different JSTs for different jurisdictions.

Fundamental BCA Principles

The NSPM provides a set of fundamental BCA principles that represent sound economic and regulatory practices. The NSPM BCA principles presented in Table S-1 set the foundation for developing cost-effectiveness tests for BCA. The principles can be used to guide the application of cost-effectiveness testing, selection of a discount rate, and the reporting of the BCA results, and they can inform the process for prioritizing DERs to be implemented.

The NSPM BCA principles are not mutually exclusive as they contain some overlapping concepts. Further, there may be situations where it is necessary for jurisdictions to make tradeoffs between certain principles depending on specific situations.

Table S-1. NSPM BCA Principles

Principle 1	Treat DERs as a Utility System Resource
	DERs are one of many energy resources that can be deployed to meet utility/power system needs. DERs should therefore be compared with other energy resources, including other DERs, using
	consistent methods and assumptions to avoid bias across resource investment decisions.
Principle 2	Align with Policy Goals Jurisdictions invest in or support energy resources to meet a variety of goals and objectives. The primary cost-effectiveness test should therefore reflect this intent by accounting for the jurisdiction's applicable policy goals and objectives.
Principle 3	Ensure Symmetry Asymmetrical treatment of benefits and costs associated with a resource can lead to a biased assessment of the resource. To avoid such bias, benefits and costs should be treated symmetrically for any given type of impact.
Principle 4	Account for Relevant, Material Impacts Cost-effectiveness tests should include all relevant (according to applicable policy goals), material impacts including those that are difficult to quantify or monetize.
Principle 5	Conduct Forward-Looking, Long-term, Incremental Analyses Cost-effectiveness analyses should be forward-looking, long-term, and incremental to what would have occurred absent the DER. This helps ensure that the resource in question is properly compared with alternatives.
Principle 6	Avoid Double-Counting Impacts Cost-effectiveness analyses present a risk of double-counting benefits and/or costs. All impacts should therefore be clearly defined and valued to avoid double-counting.
Principle 7	Ensure Transparency Transparency helps to ensure engagement and trust in the BCA process and decisions. BCA practices should therefore be transparent, where all relevant assumptions, methodologies, and results are clearly documented and available for stakeholder review and input.
Principle 8	Conduct BCAs Separately from Rate Impact Analyses Cost-effectiveness analyses answer fundamentally different questions than rate impact analyses, and therefore should be conducted separately from rate impact analyses.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 17 of 302

Process for Developing a Primary Jurisdiction-Specific Test

The NSPM presents a step-by-step process for developing a primary cost-effectiveness test (or modifying an existing primary test). Referred to as the 'JST', this test reflects the fundamental BCA principles in Table S-1.

This manual presents the regulatory perspective, which refers to the perspective of regulators or similar entities that oversee utility DER investment decisions. A JST should reflect the regulatory perspective to

ensure proper accounting of the jurisdiction's applicable policy goals—as guided by statutes, regulations, organizational policies, utility resource planning principles and policies, and/or other codified forms under which utilities or energy providers operate.

Figure S-1 illustrates the regulatory perspective relative to traditional cost-effectiveness test perspectives.

Figure S-1. The Regulatory Perspective

Traditional Perspectives



• Three perspectives define the scope of impacts to include in the most common traditional cost-effectiveness tests.

The primary test answers the critical question: Which DERs have benefits that exceed costs and therefore merit utility acquisition or support on behalf of customers?

Utility System Impacts

Regulatory Perspective

- Perspective of public utility commissions, legislators, muni/coop boards, public power authorities, and other relevant decision-makers.
- Accounts for utility system plus impacts relevant to a jurisdiction's applicable policy goals (which may or may not include host customer impacts).
- Can align with one of the traditional test perspectives, but not necessarily.

Table S-2 presents the multi-step process for developing a JST. This process provides the flexibility for each jurisdiction to tailor its primary JST to its own goals and objectives.

Table S-2. Developing a Jurisdiction's Primary Test: A 5-Step Process

STEP 1 Articulate Applicable Policy Goals

Articulate the jurisdiction's applicable policy goals related to DERs.

STEP 2 Include All Utility System Impacts

Identify and include the full range of utility system impacts in the primary test, and all BCA tests.

STEP 3 Decide Which Non-Utility System Impacts to Include

Identify those non-utility system impacts to include in the primary test based on applicable policy goals identified in Step 1:

• Determine whether to include host customer impacts, low-income impacts, other fuel and water impacts, and/or societal impacts.

STEP 4 Ensure that Benefits and Costs are Properly Addressed

Ensure that the impacts identified in Steps 2 and 3 are properly addressed, where:

- Benefits and costs are treated symmetrically.
- Relevant and material impacts are included, even if hard to quantify.
- Benefits and costs are not double-counted.
- Benefits and costs are treated consistently across DER types.

STEP 5 Establish Comprehensive, Transparent Documentation

Establish comprehensive, transparent documentation and reporting, whereby:

- The process used to determine the primary test is fully documented.
- Reporting requirements and/or use of templates for presenting assumptions and results are developed.

When deciding whether to include a benefit or cost in a BCA test, it is important to distinguish between the *definition* versus *application* of the BCA test. Any impact that is deemed to be relevant should be included as part of the definition of the test. In some cases, a benefit or cost may be relevant but not material. *Material* impacts are those that are expected to be of sufficient magnitude to affect the result of a BCA. Impact determined to be immaterial should be documented, but not necessarily included in the application of the BCA test.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 19 of 302

Secondary BCA Tests

The NSPM also provides guidance on how secondary tests can be used to help assess marginally cost-effective DERs or to prioritize across DERs. While a jurisdiction's primary test should be used to inform whether a utility should fund or otherwise support DERs, it does not have to be utilized in a vacuum. In some instances, secondary tests can help enhance regulators' and stakeholders' overall understanding of DER impacts by answering other questions regarding utility DER investments. Different tests provide different information about the costeffectiveness and impacts of DERs. However, secondary tests should be used cautiously to ensure that they do not make the BCA decision-making process burdensome or undermine the purpose of the primary test. This manual does not prescribe any one cost-effectiveness test. Because the JST is based upon each jurisdiction's applicable policy goals, and those goals can vary across jurisdictions, the test may take a variety of forms. Further, depending on a jurisdiction's applicable policy goals, the primary test may or may not align with traditional BCA tests (e.g., the Total Resource Cost test.)

Part II. DER Benefits and Costs and Cross-Cutting Considerations

Part II of the manual presents a catalog of the full range of benefits and costs that may be applicable to specific types of DERs. This catalog can be used as a reference when deciding which types of benefits and costs should be included in a jurisdiction's BCA test.

The catalog of impacts is presented in table format and supported with detailed descriptions of each impact type. Table S-3 shows the range of potential DER impacts to the electric utility system, along with descriptions of each impact. Similarly, Table S-4 and Table S-5 provide a summary of potential host customer and societal impacts, respectively. Part II also addresses natural gas and other fuel system impacts and specific host customer non-energy impacts (NEIs).

Туре	Utility System Impact	Description
Generation	Energy Generation	The production or procurement of energy (kWh) from generation resources on behalf of customers
	Capacity	The generation capacity (kW) required to meet the forecasted system peak load
	Environmental Compliance	Actions to comply with environmental regulations
	RPS/CES Compliance	Actions to comply with renewable portfolio standards or clean energy standards
	Market Price Effects	The decrease (or increase) in wholesale market prices as a result of reduced (or increased) customer consumption
	Ancillary Services	Services required to maintain electric grid stability and power quality
Transmission	Transmission Capacity	Maintaining the availability of the transmission system to transport electricity safely and reliably
	Transmission System Losses	Electricity or gas lost through the transmission system
	Distribution Capacity	Maintaining the availability of the distribution system to transport electricity or gas safely and reliably
	Distribution System Losses	Electricity lost through the distribution system
Distribution	Distribution O&M	Operating and maintaining the distribution system
	Distribution Voltage	Maintaining voltage levels within an acceptable range to ensure that both real and reactive power production are matched with demand
	Financial Incentives	Utility financial support provided to DER host customers or other market actors to encourage DER implementation
	Program Administration	Utility outreach to trade allies, technical training, marketing, and administration and management of DERs
	Utility Performance Incentives	Incentives offered to utilities to encourage successful, effective implementation of DER programs
General	Credit and Collection	Bad debt, disconnections, reconnections
General	Risk	Uncertainty including operational, technology, cybersecurity, financial, legal, reputational, and regulatory risks
	Reliability	Maintaining generation, transmission, and distribution system to withstand instability, uncontrolled events, cascading failures, or unanticipated loss of system components
	Resilience	The ability to anticipate, prepare for, and adapt to changing conditions and withstand, respond to, and recover rapidly from disruptions

Table S-3. Potential DER Impacts: Electric Utility System

Туре	Host Customer Impact	Description
Host Customer	Host portion of DER costs	Costs incurred to install and operate DERs
	Host transaction costs	Other costs incurred to install and operate DERs
	Interconnection fees	Costs paid by host customer to interconnect DERs to the electricity grid
	Risk	Uncertainty including price volatility, power quality, outages, and operational risk related to failure of installed DER equipment and user error; this type of risk may depend on the type of DER
	Reliability	The ability to prevent or reduce the duration of host customer outages
	Resilience	The ability to anticipate, prepare for, and adapt to changing conditions and withstand, respond to, and recover rapidly from disruptions
	Tax incentives	Federal, state, and local tax incentives provided to host customers to defray the costs of some DERs
	Host Customer NEIs	Benefits and costs of DERs that are separate from energy-related impacts
	Low-income NEIs	Non-energy benefits and costs that affect low-income DER host customers

Table S-4. Potential Benefits and Costs of DERs: Host Customer

Table S-5. Potential Costs and Benefits of DERs: Societal

Туре	Societal Impact	Description
Resilience GHG Emissions Other Environmenta	Resilience	Resilience impacts beyond those experienced by utilities or host customers
	GHG Emissions	GHG emissions created by fossil-fueled energy resources
	Other Environmental	Other air emissions, solid waste, land, water, and other environmental impacts
Societal	Economic and Jobs	Incremental economic development and job impacts
	Public Health	Health impacts, medical costs, and productivity affected by health
	Low-Income: Society	Poverty alleviation, environmental justice, and reduced home foreclosures
	Energy Security	Energy imports and energy independence

In addition to describing the range of potential DER impacts, Part II also addresses key cross-cutting benefit and cost issues, including the following:

- Temporal and Locational Impacts of DERs: Several of the benefits and costs of some DERs can vary significantly depending on when the DER operates and where it is located. DER benefits and costs should be estimated using temporal and locational detail sufficient to adequately represent the DER operating patterns and consequent benefits and costs.
- Interactive effects between individual DERs: Some DERs can have interactive effects on other DERs in terms of affecting avoided costs, affecting the magnitude of kWh and kW impacts, and enabling the adoption of other DERs. These interactive effects should be accounted for in BCAs for those instances where they are likely to have a material effect.
NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 22 of 302

- Air emission impacts: Greenhouse gas (GHG) and other air emission impacts will depend upon when the DER operates and which energy resources are displaced at that time. Estimates of GHG and other air emission impacts should account for the temporal and marginal DER impacts in as much detail as necessary to reflect these effects.
- Renewable generation impacts: DERs can support renewable electricity generation by providing grid flexibility and ancillary services. DERs can also reduce (or increase) the need to curtail renewable resources during times when renewable generation exceeds customer load. These impacts on renewable generation should be accounted for when they are expected to have a material effect on the BCA results.
- Discount rates: The choice of discount rate to use for a BCA can often have a very large effect on the result of the analysis. This choice should be guided by the jurisdiction's applicable policy goals and the regulatory perspective.

DER impacts identified for inclusion in a jurisdiction's BCA should ideally be estimated in monetary terms. Monetary values provide a uniform way to compile, present, and compare benefits and costs. While some DER impacts are difficult to quantify in monetary terms—either due to the nature of the impact or the lack of available information about the impacts—approximating hard-to-quantify impacts using best available information is preferable to arbitrarily assuming a value, including assuming that the relevant impacts do not exist or have no value. Further, some approximation may be necessary to ensure symmetry in the treatment of benefits and costs for certain relevant impacts.

Part III: BCA for Specific DER Types

Part III of the NSPM contains five chapters that discuss individual characteristics and impacts of each DER type covered in this manual: EE, DR, DG, DS, and electrification (including managed charging and discharging of EVs). Part III describes and provides guidance on key factors and challenges that affect the impacts of each DER type.

Table S-6, Table S-7, and Table S-8 show the range of benefits and costs in terms of their applicability to each DER. They indicate which impacts are typically a benefit, a cost, or either depending on the specific DER use case. The tables are a compilation of the DER-specific tables presented in Chapters 6–10 of the manual.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 23 of 302

Туре	Utility System Impact	EE	DR	DG	Storage	Electrification
	Energy Generation	•	•	•	•	•
	Capacity	٠	•	•	•	•
6	Environmental Compliance	٠	•	•	•	•
Generation	RPS/CES Compliance	٠	•	•	•	•
	Market Price Effects	٠	•	•	•	•
	Ancillary Services	•	•	•	•	•
	Transmission Capacity	•	•	•	•	•
Transmission	Transmission System Losses	•	•	•	•	•
	Distribution Capacity	٠	•	•	•	•
	Distribution System Losses	•	•	•	•	•
Distribution	Distribution O&M	•	•	•	•	•
	Distribution Voltage	•	•	•	•	•
	Financial Incentives	٠	•	•	•	•
	Program Administration Costs	٠	•	•	٠	•
	Utility Performance Incentives	٠	•	•	٠	•
General	Credit and Collection Costs	٠	•	•	•	•
	Risk	٠	•	•	•	•
	Reliability	٠	•	•	•	•
	Resilience	•	•	•	•	0

Table S-6. Potential Benefits and Costs: Electric Utility System

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type

Table S-7.	Potential	Benefits	and	Costs of	f DERs:	DER	Host	Customer
------------	-----------	----------	-----	----------	---------	-----	------	----------

Туре	Host Customer Impact	EE	DR	DG	Storage	Electrification
Host Customer	Host portion of DER costs	•	•	•	•	•
	Interconnection fees	0	0	•	•	0
	Risk	٠	0	•	•	•
	Reliability	٠	•	•	٠	•
	Resilience	٠	•	٠	٠	•
	Tax Incentives	٠	٠	٠	•	•
	Host Customer NEIs	•	•	•	•	•
	Low-income NEIs	•	•	•	٠	•

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type

Туре	Societal Impact	EE	DR	DG	Storage	Electrification
	Resilience	•	٠	•	•	•
	GHG Emissions	•	•	•	•	•
	Other Environmental	•	•	•	•	•
	Economic and Jobs	•	•	•	•	•
	Public Health	•	•	•	•	•
	Low Income: Society	•	•	٠	•	•
	Energy Security	•	•	•	•	•

Table S-8. Potential Benefits and Costs of DERs: Societal

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type

Part IV: BCA for Multiple DER Types

The manual addresses BCA for different applications where multiple DER types might be combined, including:

- multiple on-site DER types, such as grid-integrated efficient buildings (GEB);
- multiple DER types in a specific geographic location in the form of a non-wires solution (NWS);
- multiple DER types across a utility service territory; and
- dynamic system planning practices that can be used to optimize DERs and alternative resources.

Multiple On-site DERs

Multiple on-site DERs can be installed in a variety of ways:

- On a residential level, utilities programs provide incentives to adopt multiple DER types that can then be used to benefit the customer and the grid.
- On a residential and commercial level, the aggregation of DERs in grid-interactive efficient buildings (GEBs) can provide grid support at scale.
- On a community level, DERs in microgrids and smart neighborhoods can be aggregated to provide grid support at scale.

The potential benefits and costs of multiple on-site DERs will depend on the type of DERs deployed, their capabilities, locational and temporal impacts, seasonal and daily load profiles, resource ownership and control of the DERs (i.e., level of dispatchability), and interactive effects across the DERs. Figure S-2 shows how the interactive effects between distributed photovoltaics and storage and between EE and DR can affect the total benefits of a GEB.



Figure S-2. Interactive Effects in Grid-Interactive Efficient Building

Non-Wires Solutions

These solutions focus on instances where utilities or others seek to install multiple DER types in a specific geographic area for the purpose of deferring or avoiding new investments in distribution or transmission systems. In these cases, cost-effectiveness will be very project-specific, depending on the specific transmission or distribution upgrade being deferred, the length of deferral, the mix of DERs producing the deferral, and a range of other factors. Due to the nature of T&D deferrals and uncertainty of load forecasts, NWS BCAs account for a project's number of years of deferral, which can shift depending on changing load forecasts.

Other key considerations for BCAs of NWSs include:

- When NWS projects are based on existing or new customer-sited DER programs, it is critical to accurately forecast customer participation and adoption, to reduce risk of not meeting requirements.
- Interactive effects should be accounted for, including effects on avoided costs, effects on kWh or kW impacts, and enabling effects.
- DERs geographically deployed to defer a T&D upgrade can have broader impacts on the utility system (e.g., avoided energy and generation capacity costs) as well as broader impacts related to policy objectives (e.g., avoided emissions).

Illustrative Example of BCA for an NWS Project

This manual provides an illustrative example of how a jurisdiction's primary test developed using NSPM can be applied to a hypothetical NWS project. The example assumes that a hypothetical state has developed its primary cost-effectiveness test (or modified its existing primary test) using the 5-step process described in Table S-2.

The state's JST accounts for conventional overarching goals of providing safe, reliable, resilient, and reasonably priced electricity services, as well as the goal of reducing GHG emissions (as articulated in statute). The JST also accounts for host customer impacts.

Non-Wires Solution Case Study Assumptions

In this example, an electric utility is facing the need to upgrade its system infrastructure due to distribution capacity constraints identified in a densely populated geographic area within its service territory. The utility proposes to integrate DERs to serve as a non-wires solution in place of an infrastructure upgrade.

The NWS plan includes the following BTM DERs in residential and commercial buildings:

- Energy efficiency measures (e.g., lighting and controls)
- Demand response (e.g., Wi-Fi-enabled thermostats)
- Distributed photovoltaics
- Distributed storage systems

Jurisdiction-Specific Test: The hypothetical jurisdiction's primary BCA test accounts for utility system, host customer, and GHG emission impacts.

Key assumptions:

- *Non-Coincident Peak:* The distribution need is non-coincident with the overall system peak (e.g., the constrained distribution feeder peaks from 1:00–5:00pm, while system peaks from 5:00–9:00pm).
- *GHG Emissions Reduction:* The system-peak hours entail higher marginal emissions rates than the NWS, which allows the NWS to deliver GHG benefits.
- DER Operating Profiles: The NWS DERs operate in the following ways:
 - \circ $\,$ All DERs are operated to reduce the distribution peak, and some can reduce the system peak as well.
 - Storage charges during the distribution off-peak hours and discharges during the distribution peak hours.
 - DR reduces demand during distribution peak periods and/or shifts load from distribution peak periods to distribution off-peak periods.
 - \circ $\;$ Distributed PV resources generate during a portion of distribution peak period.
 - $\circ~$ EE helps to reduce demand during distribution peak periods.

The example NWS benefits and costs associated with utility system, host customer, and GHG impacts are summarized below and presented in Figure S-3.

- Generation Benefits Some generation benefits (e.g., energy generation, capacity, and ancillary services) accrue from targeting operation of DERs, such as storage and DR, during distribution peak periods. There will be additional benefits that result from some DERs—such as DPV and EE—also operating during other off-peak periods.
- *Transmission Benefits* Some transmission benefits (e.g., capacity and system losses) accrue with the reduced delivery of central generation to customers.
- *Distribution Benefits* The greatest contributor to the overall cost-effectiveness analysis is the direct benefit of operating DERs as much as possible during distribution peak periods.
- GHG Benefits In this example, the GHG emissions are higher during the distribution system peak periods than the other periods. Consequently, the peak demand reductions from the NWS will result in a net reduction in GHG emissions.
- *General Utility Costs* Financial incentives for customers to participate and administrative costs lead to the more substantive general utility costs for this illustrative analysis.
- *Host Customer Impacts* Host customer costs include interconnection fees, transaction costs, and DER costs, while benefits include various non-energy impacts.

Figure S-3 combines the net benefits and costs of utility system, host customer, and GHG impacts. In this case study, locational value plays a central role in the cost-effectiveness of an NWS, as represented by the significant distribution benefits. The BCA indicates that the NWS will have net benefits.



Figure S-3. Illustrative Example of NWS Cost-Effectiveness

System-Wide DER Portfolios

The NSPM provides guidance on how to analyze and prioritize a portfolio of multiple DER types across a utility service territory.

In analyzing portfolios of multiple DER types across a utility service territory, it is important to first establish a single primary cost-effectiveness test that can be used for all DER types. Then, it is useful to articulate the jurisdiction's DER planning objectives, which can include, for example, one or some combination of: implement all cost-effective DERs; implement the lowest-cost DERs; maximize capacity benefits from DERs; encourage a diverse range of DER technologies; encourage customer equity; achieve GHG or electrification goals at lowest cost; and avoid unreasonable rate impacts.

Utilities and others can present the BCA results for DER portfolios in ways that facilitate comparison across DER types, such as:

- DERs can be ranked by benefit-cost ratios or net benefits to indicate the most cost-effective resources.
- Levelized DER costs can be used to directly and consistently compare costs across different DER types.
- Levelized net cost curves can be used to compare and prioritize DERs according to key parameters such as \$/ton GHG reduced.
- Multiple cost-effectiveness tests, in addition to the JST, can provide additional information when analyzing portfolios of multiple DER types.



Figure S-4 presents a list of hypothetical DERS sorted by the net benefits that they provide. This information could be used to identify those DERs that warrant utility support or funding in order to achieve the greatest net benefits for a given level of funding. A similar approach could be used to prioritize BCRs by their benefit-cost ratios, or to prioritize DERs for within a given rate impact cap.

In some cases, a jurisdiction may prefer to invest in a diverse range of DER types on the basis that all DER types contribute benefits in different ways and there is value in promoting a diversity of technologies, as well as reducing

associated system risk. In such a case, regulators might decide to support a minimum amount of each type of DER. This could be achieved by sorting the DER types by net benefits or benefit-cost ratios and selecting the lowest cost options for each type of DER.

Dynamic System Planning

Utilities have conducted traditional distribution system planning for many years to determine how to best to build and maintain the distribution grid. The focus of this practice has been on providing safe, reliable power through the distribution grid at a low cost. It typically has not accounted for DERs as alternatives to traditional distribution system technologies. However, the scope of utility system planning is expanding to manage the increasing complexity of the electricity system, while addressing evolving state policy objectives, changing customer priorities, and increased DER deployment. The manual provides an overview of evolving advanced planning practices that can allow utilities to more effectively and dynamically optimize DERs using *dynamic system planning*.

Table S-9 summarizes several different types of planning practices used by electric and gas utilities. It presents practices according to whether they are used by distribution-only or vertically integrated utilities, and it shows what elements of the utility system are accounted for by each type of practice.

Each type of planning practice uses some form of BCA for comparing and optimizing different resources. Each practice is a type of dynamic system planning described above, where the resources of interest are optimized relative to a dynamic set of alternative resources.

Type of		Planning Practice Accounts for:				
Utility System	Planning Practice	Distribution System	DERs	Transmission System	Utility-Scale Generation	
Distribution-only & vertically integrated	Traditional distribution planning	\checkmark	-	-	-	
	Integrated distribution planning (IDP)	\checkmark	\checkmark	-	-	
Vertically integrated	Transmission planning	-	-	\checkmark	-	
	Integrated resource planning (IRP)	-	\checkmark	-	\checkmark	
	Integrated grid planning (IRP)	√	~	✓	\checkmark	

Table S-9. Types of Dynamic System Planning Practices

Dynamic system planning practices have evolved in recent years to optimize DERs and maximize their value to the system. These include integrated distribution planning (IDP) for distribution-level planning only and integrated grid planning (IGP) for full-system planning.

Appendices

Table S-10 summarizes the appendices that provide further detail on some NSPM topics that warrant additional explanation.

Part V	Appendices	
Appendix A	Rate Impacts	Describes the difference between cost-effectiveness and rate impact analyses, as well as the role of rate, bill, and participation analyses
Appendix B	Template NSPM Tables	Tables that can be used by jurisdictions to document applicable policies and relevant benefits and costs to inform their BCAs
Appendix C	Approaches to Accounting for Relevant Impacts	Provides guidance on options to account for relevant benefits and costs, including hard-to-quantify impacts and non-monetary impacts
Appendix D	Presenting BCA Results	Provides guidance on presenting results in a way that is most useful for making cost-effectiveness decisions
Appendix E	Traditional Cost-Effectiveness Tests	Summarizes the commonly used traditional cost-effectiveness tests from the <i>California Standard Practice Manual</i>
Appendix F	Transfer Payments and Offsetting Impacts	Provides guidance on impacts that appear to be both a benefit to one party and a cost to another party, thereby cancelling each other out
Appendix G	Discount Rates	Describes ways to determine discount rates that are consistent with the jurisdiction's applicable policy goals
Appendix H	Energy Efficiency—Additional Guidance	Describes how to address free-riders and spillover effects where net savings are used; and treatment of early replacement measures

Table S-10. Guide to Appendices

GLOSSARY

Advanced metering infrastructure (AMI): an integrated system of smart meters and data management systems that enables two-way communication between utilities and customers.

Aggregator: a company that negotiates with producers of a utility service such as electricity on behalf of groups of consumers, or which bundles DERs to engage as a single entity—a virtual power plant (VPP)— in power or service markets.

Alternative thresholds: an approach to address hard-to-monetize impacts that allow DERs to be considered cost-effective at pre-determined benefit-cost ratios. (See Appendix C.)

Analysis period: the time period over which cost-effectiveness analysis should be conducted. (See "Expected useful life" and Appendix H.)

Ancillary services: services required to maintain electric grid stability. Typically include frequency regulation, voltage regulation, spinning reserves, and operating reserves, either traded in wholesale energy markets or self-supplied by utilities.

Avoided costs: the costs of those electricity and gas resources (e.g., generation, transmission, and distribution system infrastructure) that are deferred or avoided by the DERs being evaluated for cost-effectiveness. (See Chapter 4.)

Behind the meter (BTM): Also referred to as "customer-facing," BTM includes a range of technologies that customers can install to reduce the host customers' bills and/or improve the operational efficiency of the distribution system, thus also sometimes providing value to the grid. This includes all types of DERs, such as EE, DR, DG, DS, electrification, and EVs.

Benefit-cost analysis (BCA): a systematic approach for comparing the benefits and costs of alternative options to determine whether the benefits exceed the costs over the lifetime of the program or project under consideration. (See Section 1.3.)

Best available: information that is based on acceptable standards of accuracy, reliability and relevancy, is up-to-date and mindful of limitations, is peer-reviewed when appropriate and required, and delivered at an appropriate time in the decision-making process.

Bill impact analysis: indicates the extent to which customer bills are affected for customers that participate in DER programs and those who do not.

Building electrification: substituting electricity for consumption of other fuels, e.g., space heating, water heating, cooling, cooking, drying, and other end-uses, therefore increasing electric system costs. (See "Electrification" and Chapter 10.)

Cost-effectiveness: when investment in a resource is worthwhile; measured by the benefits of investing in a resource being greater than the costs of investing therein.

Demand flexibility: the capability provided by DERs to reduce, shed, shift, modulate, or generate electricity; energy flexibility and load flexibility are often used interchangeably with demand flexibility.

Demand response (DR): DR programs play a role in the operation of the electric grid by reducing or shifting their electricity usage during peak periods in response to time-based rates or other forms of

financial incentives. DR programs are used by some electric system planners and operators as resource options for balancing supply and demand. Methods of engaging customers in DR efforts include offering time-based rates and direct load control programs which provide the ability for power companies to e.g., cycle air conditioners and water heaters on and off during periods of peak demand in exchange for a financial incentive and lower electric bills. (NERC 2011) (See Chapter 7.)

Discount rates: a component of cost-effectiveness analysis which reflects a "time preference"—the relative importance of short- versus long-term impacts. A higher discount rate gives more weight to short-term benefits and costs relative to long-term benefits and costs, while a lower discount rate gives greater weight to long-term impacts. (See Appendix G.)

Dispatchable: means that the timing and level of response is under the control of the utility, either through technical control or by the terms of a contract, or both.

Distributed energy resource (DER): electricity and gas resources sited close to customers that can provide all or some of their immediate power needs and/or can be used by the utility system to either reduce demand or provide supply to satisfy the energy, capacity, or ancillary service needs of the grid (DOE 2019a. These include EE, demand response, distributed generation, storage, plug-in electric vehicles, strategic electrification technologies, and more.

DER pricing mechanism: includes initiatives and performance-based compensation to encourage customers to install and utilize DERs as efficiently as possible through price signals. Examples include time-based pricing (time-of-use or TOU rates, peak time rebates, and critical peak pricing) and net metering compensation.

DER programs: include passive and performance-based programs, initiatives, and policies that encourage customers to adopt DERs. Examples include traditional utility EE programs, traditional utility DR or bring-your-own-device (BYOD) programs, distributed storage incentives, and investments in EV infrastructure.

DER procurement: includes initiatives to procure DERs, whether built by a utility or procured from thirdparty vendors, e.g., competitive and/or technology providers, typically using a competitive procurement process.

Distributed generation (DG): electric generation interconnected to the distribution grid and operating at the distribution level, generally near a load, though sometimes stand-alone. DG includes distributed solar photovoltaic (PV/DPV) technology, combined heat and power (CHP), district heating and cooling, small wind, and biomass and biogas facilities associated with landfills and agricultural operations. (See Chapter 8.)

Distributed storage (DS): technologies used to locally store energy. This manual primarily focuses on BTM resources, such as lithium-ion batteries, but it can also apply more broadly to all distributed storage types (e.g., thermal, including electric water heaters) and chemistries (e.g., lead-acid) as opposed to those connected at transmission (e.g., pumped hydro and compressed air). (See Chapter 9.)

Double-counting: the inclusion of overlapping impacts in analyses, and thus counting one effect more than once.

Effective useful life (EUL): the average time over which a DER measure results in energy savings (or use) including the effects of equipment failure, removal, and cessation of use.

Electric vehicle (EV): a vehicle powered directly by electricity rather than other fuels. These can also in some instances operate as storage devices. See "Vehicle to Grid" and "Electrification" and Chapter 10.

Electrification: increased electrification of end-uses, when beneficial to the utility system as a whole, including the increased integration of electrification including building electrification and EVs. This can be "partial," where some but not all fuel consumption is replaced by electricity (e.g., a plug-in hybrid EV), or "complete" (e.g., a battery electric vehicle). (See Chapter 10.)

Enabling effects: when some DERs make it easier or more cost-effective to adopt other types of DERs. (See Section 11.5.)

Energy efficiency (EE): Resources that include technologies, services, measures, or programs that reduce energy consumption by host customers and that are funded by, promoted, or otherwise supported on behalf of all electricity and gas utility customers. (See Chapter 6.)

Forward-looking analysis: captures the difference between benefits and costs that would occur over the lifetime of the DERs and those what would occur absent the DERs.

Free-ridership: DER savings that would have occurred in the absence of the program.

Front of the meter (FOM): also known as utility-facing, FOM DERs are typically operated to reduce system costs (as opposed to BTM DERs, which are typically operated to reduce customer costs). (See Chapter 5.)

Greenhouse gas (GHG) emissions: gases that trap heat in the atmosphere (including carbon dioxide, methane, nitrous oxide, and fluorinated gases) emitted from human activities, primarily burning fossil fuels for electricity generation, transportation, industrial processes, commercial and residential heating, and agriculture.

Grid-interactive efficient building (GEB): an energy-efficient building that uses smart technologies and on-site DERs to provide demand flexibility while co-optimizing for energy cost, grid services, and occupant needs and preferences in a continuous and integrated way (DOE 2019a). (See Chapter 11.)

Grid modernization: investments in updating the grid to accommodate DERs, multi-way power flows, and/or active management of the distribution grid to achieve reliability and greater efficiency (DOE 2017b). These are often categorized as either utility-facing, which may support increased DER implementation, or customer-facing, which often include a range of DERs.

Host customer: The owner/occupant of the site at which BTM DERs are installed and/or operated. In some cases, these are program participants, e.g., participants in a DR or EE program. In other cases, there is no program, e.g., EV owners.

Incremental analysis: consists of changes that will occur as a result of the DER relative to a scenario where the DER is not in place.

Integrated distribution planning (IDP): a long-run utility planning process that expands on traditional distribution planning and allows for evaluation of both traditional distribution resources and DERs for meeting distribution grid needs (see Table 1-1 and Chapter 14).

Integrated grid planning (IGP): a long-run planning process for vertically integrated utilities that evaluates all resource types (DERs and utility-scale resources) to enable optimization across all levels of the utility system (generation, transmission, and distribution) (see Table 1-1 and Chapter 14).

Integrated resource planning (IRP): a long-run planning process for vertically integrated utilities that evaluates DERs and utility-scale generation for meeting peak and energy demands (see Table 1-1 and Chapter 14).

Long-run: the period covering the full life cycle of the resource being analyzed. The long-run approach is necessary to account for the full benefits and costs of the DER being evaluated, particularly since energy resources, including many DERs, can last decades.

Lost revenues: When DERs reduce consumption of electricity or gas, which can result in fixed costs being spread across a smaller volume of sales, putting upward pressure on rates.

Impacts: both the benefits and the costs of a supply-side or demand-side resource. (See Chapter 4.)

Jurisdictions: states, provinces, utilities, municipalities, or other regions for which DER resources are planned and implemented.

Jurisdiction-Specific Test (JST): the primary test created by a jurisdiction following use of the NSPM BCA Framework. It embodies all of the key principles of cost-effectiveness analyses and accounts for that jurisdiction's applicable policy goals by including impacts identified as relevant to that jurisdiction's goals and objectives.

Levelized costs: represent the average cost per unit of energy required to install and operate an electricity or gas resource. The costs of electricity and gas resources, including DERs, can be put into levelized costs to allow for a relatively simple, direct comparison across different resources.

Material impacts: impacts that are expected to be of sufficient magnitude to affect the result of a BCA.

Microgrid: a group of interconnected loads and DERs within clearly defined electrical boundaries. A microgrid can act as a single controllable entity with respect to the grid and can connect or disconnect from the grid to operate as either grid-connected or an "island." (See Chapter 11.)

Multiple-DER analysis: when multiple-DER types are assessed and evaluated relative to a static set of alternative resources. This approach is more complex than single-DER analysis and is designed to capture the interactive effects of DERs on one another. Multiple-DER analysis can be applied in context of a customer site (e.g. GEB), for a certain geographic area to identify non-wires solutions, and/or across the entire utility system. (See Chapters 11–13.)

Net energy metering (NEM): also known as net metering, tariff design that allows consumers to receive bill credit on a kWh basis for excess generation injected onto the grid for the use of other customers. (See Chapter 8.)

Non-dispatchable: refers to programs and measures without such controls and includes time-varying rates that send price signals to encourage customers to alter their energy usage during particular hours.

Non-energy impacts (NEI): are impacts of DERs other than direct energy and demand impacts. While these impacts can be non-energy benefits and related costs, most are considered benefits (non-energy benefits, or NEBs). Examples include reduced emissions, comfort and productivity improvements, local economic development, and reduced risk of utility service disruptions or price spikes.

Non-pipes solution: alternatives to meeting on-system natural gas demand that delay or avoid the need for investment in traditional resources such as pipelines, storage capacity, winter-peaking services, and distribution system infrastructure.

Non-wires solution (NWS): also known as non-wires alternative (NWA), geotargeting, and market-based alternative or solution. This is a strategy of deploying DERs in a specific geographic area for the purpose of deferring or avoiding new investments in equipment, distribution, or transmission lines. (See Chapter 12.)

Participant Cost Test (PCT): a cost-effectiveness test, as provided in the 2001 *California Standard Practice Manual*, which includes the benefits and costs experienced by host customers.

Participation impacts: indicate impacts participating customers will experience from participating in a DER program, in terms of bill reductions or increases.

Policy neutrality: a core component of the NSPM guidance. The NSPM does not advocate for any policy or cost-effectiveness test, but rather that a jurisdiction applies the NSPM BCA Framework to review its existing policies to develop a cost-effectiveness test which reflects local policy objectives.

Primary cost-effectiveness test: the cost-effectiveness test that a jurisdiction uses to determine whether a DER (or set of DERs) has benefits that exceed costs, and therefore merits acquisition or support from utilities or other energy providers.

Proxies: simple, quantitative values that can be used as indirect indicators for values not monetized by conventional means, which can be applied to any type of benefit or cost that is hard to monetize and is expected to be of significant magnitude. (See Appendix C.)

Rate, bill, participant analysis: indicates the extent to which customers will be affected by DERs, and the extent to which DER investments might lead to distributional equity or cost allocation concerns. See "Rate Impact Analysis," "Bill Impact Analysis," and "Participation Impact Analysis."

Rate impact analysis: assessment of the extent to which investing in a resource will impact customer rates, sometimes in the form of a Ratepayer Impact Measure (RIM) test. This is a separate type of analysis from cost-effectiveness, which assesses whether the benefits of investing in a resource outweigh the costs. See Appendix A.

Regulators and other decision-makers: entities including institutions, agents, or other decision-makers that are authorized to determine utility resource cost-effectiveness and funding priorities, and to oversee and guide DER analyses. Such institutions or agents include public utility commissions, legislatures, boards of publicly owned utilities, the governing bodies for municipal utilities and cooperative utilities, municipal aggregator governing boards, and more.

Regulatory perspective: the perspective of regulators or other agents that oversee resource investment choices, including energy generation and T&D infrastructure. This perspective is guided by the jurisdiction's energy and other applicable policy goals—whether in laws, regulations, organizational policies, or other codified forms—under which it operates.

Revenue shifting: occurs when reduced consumption from DERs shifts the collection of revenues from DER customers to others but does not increase rates. This can be caused by downward pressure on rates created by avoided utility system costs exceeding the upward pressure on rates from reduced sales. Not to be conflated with cost-shifting, which occurs when reduced consumption from DERs shifts collection of revenues from DER customers to others but increases rates. (See Appendix A.)

Secondary cost-effectiveness test: a cost-effectiveness test that supports use of the primary test by helping enhance stakeholders' overall understanding of DER impacts by answering additional questions regarding utility DER investments. The impacts included in secondary tests are driven by the purpose of the secondary analyses, including informing decisions on how to prioritize DERs; informing decisions regarding marginally cost-effective DERs; encouraging consistency across DER types; and considering other types of effects on customers. (See Section 3.3 and Appendix D.)

Single-DER analysis: when one DER type is assessed in isolation from other DER types and is evaluated relative to a static set of alternative resources. (See Chapters 6–10.)

Smart neighborhoods or communities: a set of homes or buildings that include GEB characteristics that can be connected as a neighborhood-style microgrid (DOE 2019c). (See Chapter 11.)

Societal Cost Test (SCT): a cost-effectiveness test, as provided in the 2001 *California Standard Practice Manual*, which includes the benefits and costs experienced by society.

Spillover: installation of DERs by customers who did not directly participate in a DER program but were nonetheless influenced by the program.

Symmetry: a key principle for the treatment of benefits and costs which is necessary to avoid bias toward any one resource, whereby both benefits and costs are included (or excluded) for each relevant impact. If each type of impact is not treated symmetrically, the result will be a sub-optimal selection of resources. (See Chapter 2.)

Time-varying rates: Rate designs that provide different price signals to customers at different times of the day, season, or year, based on differences in underlying costs to the system.

Total Resource Cost Test (TRC): a cost-effectiveness test that includes the benefits and costs experienced by the utility system, plus benefits and costs to the program participants.

Transfer payment: a one-way payment of money for which no money, good, or service is received in exchange. This is uncommon in DER BCAs. (See Appendix F.)

Transparency: a key principle whereby cost-effectiveness practices fully document all relevant inputs, assumptions, methodologies, and results. (See Appendix B and Appendix D for recommended templates and reporting formats.)

Utility: any entity that funds or otherwise supports DERs and is subject to or undertakes a costeffectiveness analysis to inform investment decisions. This includes investor-owned utilities; publicly owned utilities; municipal utilities; cooperative utilities; federal, state, and local governments; nongovernmental organizations; and others.

Utility Cost Test (UCT): a cost-effectiveness test includes the benefits and costs experienced by the utility system. This test is also known as a Program Administrator Cost Test (PACT).

Utility system: all elements of the electricity or gas system necessary to deliver services to the utility's customers. For electric utilities, this includes generation, transmission, distribution, and utility operations. For gas utilities, this includes transportation, delivery, fuel, and utility operations. This term refers to any type of utility ownership or management, including investor-owned utilities, publicly owned utilities, municipal utility systems, cooperatives, etc.

Vehicle to grid (V2G): a two-way flow capability in some EVs that allows EVs to function as storage devices that can push electricity back onto the grid, thus potentially reducing net system peak demands and/or net T&D peak demands. (See "Electrification" and Chapter 10.)

Virtual power plants (VPP): a suite of DERs optimized by software and advanced communication systems that aggregate, control, dispatch, and/or plan deployment to provide services similar to a conventional power plant. (See Chapter 12.)

ACRONYMS

AC	Avoided costs	JST	Jurisdiction-Specific Test
AMI	Advanced metering infrastructure	kW	Kilowatt
BCA	Benefit-cost analysis	kWh	Kilowatt-hour
BCR	Benefit-cost ratio	LBNA	Locational net benefit analysis
BEV	Battery electric vehicle	LCOE	Levelized cost of energy or electricity
втм	Behind the meter	LCSE	Levelized cost of saved energy
BTU	British thermal units	LED	Light-emitting diode
BYOD	Bring your own device	LSE	Load serving entity
вуот	Bring your own thermostat	MLP	Municipal light plant
CES	Clean energy standard	MMBtu	Million British thermal units
СНР	Combined heat and power	MW	Megawatt
C&I	Commercial and industrial	MWh	Megawatt-hour
СОР	Coefficient of performance	NEIs	Non-energy impacts
CPS	Clean peak standard	NEM	Net energy metering
CS	Community solar	NPS	Non-pipes solution
DCFC	Direct current fast chargers	NWS	Non-wires solution
DER	Distributed energy resources	0&M	Operations and maintenance
DG	Distributed generation	PACT	Program Administrator Cost Test
DPV	Distributed solar photovoltaic	РСТ	Participant Cost Test
DR	Demand response	PHEV	Plug-in hybrid electric vehicle
DRP	Distribution resource plan	PV	Solar photovoltaic
DS	Distributed storage	REC	Renewable energy credit
DSP	Distribution system planning	RIM	Rate Impact Measure Test
EE	Energy efficiency	RPS	Renewable portfolio standard
EL	Electrification	RR	Retail rate
EM&V	Evaluation, measurement, and verification	RTO	Regional transmission organization
EV	Electric vehicle	SCT	Societal Cost Test
FIT	Feed-in tariffs	του	Time of use
FOM	Front of the meter	TRC	Total Resource Cost (Test)
FTR	Fixed transmission rights	T&D	Transmission and distribution
GEB	Grid-interactive efficient building	UCT	Utility Cost Test
GHG	Greenhouse gas	VAR	Voltage levels and reactive power
HVAC	Heating, ventilation, and air conditioning	VDER	Value of distributed energy resources
IDP	Integrated distribution planning	VNM	Virtual net metering
IGP	Integrated grid planning	VOS	Value of solar
ΙΟυ	Investor-owned utilities	VPP	Virtual power plants
IRP	Integrated resource planning	V2G	Vehicle to grid
ISO	Independent system operator	WACC	Weighted average cost of capital
ITC	Federal Investment Tax Credit		

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 37 of 302

1. INTRODUCTION

This chapter describes the purpose and scope of this manual. It provides an overview of benefit-cost analysis (BCA) and describes the format of this manual.

1.1 Distributed Energy Resources - Overview

DERs are resources located on the distribution system that are generally sited close to or at customers' facilities. The DERs addressed in this manual include energy efficiency (EE), demand response (DR), distributed generation (DG), distributed storage (DS), electric vehicles (EV), and building electrification.

The electricity and gas industries are increasingly planning for and implementing distributed energy resources (DERs) to meet demand flexibility and other utility system resource needs. This change is driven largely by changing economics, customer preferences and demand, and a range of policy goals and objectives. In this evolving landscape, applying a comprehensive and consistent benefit-cost analysis framework (BCA Framework) to analyze investments in DERs can help jurisdictions implement optimal levels of DERs and avoid uneconomic decisions that can lead to unintended increased costs for customers.

Investments in DERs—whether in the form of programs, procurement, or pricing mechanisms—are made for multiple purposes, either independently or in some combination, to meet a range of policy goals and objectives. These purposes can include, for example, reducing utility system costs, deferring capacity, providing demand flexibility, increasing reliability, reducing energy burdens for low- to moderate-income customers, managing grid power quality, and/or achieving carbon emission reduction goals.

Generally, DERs represent a critical component of the evolution of the electricity grid, allowing for a more flexible grid, enabling two-way flows of energy, enabling third parties to introduce and sell new electricity products and services, and empowering customers to optimize their end-uses and consumption patterns to lower their bills and utility costs. Additionally, the flexible loads that DERs can provide or support are increasingly being considered by states as an

option to enable the integration of variable-generation renewable energy resources to achieve a carbon-free grid.

This manual sets forth the NSPM BCA Framework and supporting concepts and guidance for sound, comprehensive, and balanced assessment of the benefits and costs of DERs. While the manual does not address every nuance or scenario for DER investments, the NSPM can help jurisdictions identify the full range of DERs whose benefits exceed their costs and develop BCA practices that are most appropriate for their jurisdictions given their goals and objectives. The NSPM can help jurisdictions identify the full range of DERs whose benefits exceed their costs and develop BCA practices that are most appropriate for their jurisdictions given their goals and objectives.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 38 of 302

1.2 Audience and Terminology

The NSPM is intended to be relevant to all jurisdictions, and to all entities that have a role overseeing and guiding DER decision-making.

This manual is relevant to all DERs that are funded, acquired, or otherwise supported by electric or gas utilities (or other energy providers) on behalf of their customers.

The NSPM applies to all types of utilities, including those in regions where utilities are vertically integrated, distribution-only, or serving as a distribution platform for host customers to access a variety of energy services and DERs from third parties (e.g., aggregators.)

This manual uses many terms that are commonly used within the electricity and gas industries. Key terms are defined in a Glossary and in relevant sections of the manual. Some of the terms used in the manual are more broadly defined than in other applications, as noted in the text box below.

KEY TERMS USED THROUGHOUT THIS MANUAL

Jurisdiction refers broadly to any region or service territory that would be served by the DERs being analyzed. This includes a state, a province, a utility service territory, a city or a town, or some other jurisdiction covered by regulators or other entities that oversee DER initiatives.

Utility refers broadly to any entity that funds, implements, or supports DERs using customer or public funds that are overseen by regulators or other decision-makers. This includes investor-owned utilities; publicly owned utilities; municipal or cooperative utilities; program administrators; community choice aggregators; regional transmission organizations and independent system operators; federal, state, and local governments; and other energy service providers. *Utility expenditures* refers to spending by any of these entities on DERs.

Regulator refers broadly to any entity that oversees and guides DER analyses. This includes legislators and their staff; public utility commissions and their staff; boards overseeing public power authorities, municipal or cooperative utilities, or regional grid operators; and federal, state, and local governments.

Host customer refers to any customer that has a DER installed and/or operated on their site. In some cases, these are program participants (such as in a DR or EE program) while in other cases there is no program (such as with EV owners).

Third parties refer to the broad range of independent providers such as aggregators or implementation, service, or technology providers.

1.3 Role of Benefit-Cost Analysis

Benefit-cost analysis is a systematic approach for assessing the costeffectiveness of investments by comparing the benefits and costs of alternative options. It is widely used by businesses for deciding whether to proceed with projects, investments, programs, initiatives, or other courses of action. The analysis entails identifying all the relevant benefits and costs of a project and determining whether the benefits exceed the costs over the lifetime of the expected program or project.

BCA is frequently used by utilities, both for making internal resource investment decisions and to justify resource investment decisions to

DER BCAs answer the fundamental question: Which resources have benefits that exceed costs and therefore merit utility acquisition or support on behalf of their customers? regulators and other stakeholders. BCA has been used for many years to demonstrate the costeffectiveness of specific DER investments and is central to several utility planning practices used to optimize resource decisions, as described further below.

DER BCAs can provide a systematic way to compare DERs consistently and comprehensively with each other and with alternative resources. DER BCAs answer the fundamental question: *Which resources have benefits that exceed costs and therefore merit utility acquisition or support on behalf of their customers?* DER BCAs can also answer other related and overlapping questions, such as:

- How can DERs be used to reduce utility/power system costs?
- How can DERs be used to meet applicable policy goals (e.g., providing benefits to low-income customers, reducing consumption of other fuels, and reducing carbon emissions)?
- How will DERs affect host customers? How likely is it that host customers will adopt DERs?
- How cost-effective is one DER type relative to other types for a specific use case?
- How should investment in different DER types be prioritized?
- Which DERs should be implemented considering budgetary or other constraints?

Jurisdictions interested in potential rate impacts from DERs can also conduct rate, bill, and participation analyses. These analyses may be useful for informing decisions regarding utility investment in or support of DERs and can be complementary to BCAs. Rate impact analyses, however, answer different questions than BCAs and therefore should be conducted separately from BCAs. Section 2.3 and Appendix A provide further information on the distinction between BCAs and rate impact analyses.

1.4 Scope of this Manual

This NSPM is designed to provide objective, policy- and technology-neutral guidance that regulators, utilities, consumer advocates, DER proponents, state energy offices, and other stakeholders can apply using a systematic approach to develop BCA practices that inform decisions regarding which DERs merit acquisition or support from utilities.

This manual incorporates and expands upon the 2017 *NSPM for Assessing Cost-Effectiveness of Energy Efficiency Resources* (NSPM for EE).¹

Types of DERs and Applications Covered in this Manual

DER refers to electricity and gas resources that include EE, DR, DG, distributed storage, plug-in EVs, electrification technologies, and others. DG includes distributed photovoltaics (DPV), combined heat power (CHP), distributed wind, distributed hydro, and distributed biomass.

DERs can be funded or otherwise supported by utilities or other energy providers through several mechanisms. The three primary mechanisms include:

• *DER programs:* This includes programs, initiatives, and policies implemented by utilities or others to encourage the adoption of DERs. Examples include utility EE programs, utility DR or bring your own device (BYOD) programs, DG programs, and distributed storage programs, and investments in EV infrastructure.

¹ For more information, see www.nationalenergyscreeningproject.org/national-standard-practice-manual/.

- *DER procurement:* This includes initiatives to procure DERs, whether built by a utility or procured from third-party vendors, typically using a competitive procurement process.
- *DER pricing mechanism:* This includes pricing mechanisms designed to compensate DERs for their value to the grid, including time-varying rates (e.g., time-of-use rates, peak time rebates, and critical peak pricing) and distributed generation tariffs (e.g., net energy metering tariffs).

The NSPM focuses on DER programs because they are commonly subjected to administrative costeffectiveness assessments. Nonetheless, the principles, concepts, and guidance offered in this manual apply to DER procurement and pricing mechanisms as well. This manual is neutral regarding the use of any of the above mechanisms or approaches to supporting investment in DERs.²

The NSPM focuses on electricity DERs because these are the most common types of DERs. Nonetheless, the principles, concepts, and guidance in the manual are also applicable to natural gas DERs.

The NSPM addresses DERs that affect multiple fuel types, including electricity, natural gas, oil, propane, and gasoline (i.e., applications where a DER, or combinations of DERs, affect multiple fuel types).

The concepts and guidance in the NSPM can be applied to both behind-the-meter (BTM) and front-of-the-meter (FOM) DERs, defined as:

- *Behind the meter:* These are DERs installed at host customer facilities, where the DER is installed on the host customer side of the utility service connection (i.e., behind the utility retail meter).
- *Front of the meter:* These are DERs installed on the distribution utility system near customer loads, but not at host customer facilities.

The NSPM is neutral regarding whether a DER should be located in front of or behind the meter.

Different Levels of DER Cost-Effectiveness Analysis

DERs can be analyzed for cost-effectiveness at three different levels (LBNL 2018):

- Single-DER analysis: when one DER type is assessed in isolation from other DER types and is evaluated relative to a static set of alternative resources. This approach has been used for many years and is relatively simple compared to the multi-DER approaches described below. Chapters 6-10 include guidance for single-DER analysis of each DER type.
- Multiple-DER analysis: when multiple DER types are assessed and evaluated together, again
 relative to a static set of alternative resources. This approach is more complex than single-DER
 analysis and is designed to capture the interactive effects of DERs on one another. Multiple-DER
 analysis can be applied when analyzing (a) multiple DER types that are located at one customer
 site, for example in the context of grid-interactive efficient buildings; (b) multiple DER types that
 are located in a certain geographic area to identify non-wires solutions; and (c) multiple-DER
 types that are located across the entire utility system. Chapters 11–13 provide considerations
 and case studies for multiple-DER analysis.
- *Dynamic system planning:* when multiple DER types are assessed and evaluated relative to a dynamic set of alternative resources. Under this approach, the goal is to optimize both DERs and alternative utility-scale resources as well. Dynamic system planning practices to integrate DERs

² Some DER practices do not fall neatly into these categories. For example, demand response resources might be achieved through either utility programs or pricing mechanisms. Net metering and similar policies could be described as either a utility program or a pricing mechanism. These distinctions are not relevant for the purposes of this manual.

are relatively nascent and still evolving. Chapter 14 provides an overview of BCA issues related to dynamic system planning.

Electric and Gas System Planning Practices

DERs can also be analyzed using different planning practices. Table 1-1 summarizes several different types of planning practices used by electric and gas utilities.³ It presents practices according to whether they are used by distribution-only or vertically integrated utilities, and it shows what elements of the utility system are accounted for by each type of practice. While distribution planning is applicable to both distribution-only and vertically integrated utilities, it is only the latter which also focus on transmission and resource planning.

Each of these types of planning practices uses some form of BCA for comparing and optimizing different resources. Each of these practices are a type of dynamic system planning described above, where the resources are optimized relative to a dynamic set of alternative resources.

Table	1-1.	Types	of F	Planning	Practices
i a sic		.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	.	ion in the second	

Type of Utility System	Planning Practice	Planning Practice Accounts for:			
		Distribution System	DERs	Transmission System	Utility-Scale Generation
Distribution- only & vertically integrated	Traditional distribution planning	\checkmark	-	-	-
	Integrated distribution planning (IDP)	✓	\checkmark	-	-
Vertically integrated	Transmission planning	-	-	\checkmark	-
	Integrated resource planning (IRP)	-	✓	-	\checkmark
	Integrated grid planning (IGP)	\checkmark	\checkmark	✓	\checkmark

Dynamic utility planning practices are an area of growing importance. For example, the National Association of Regulatory Utility Commissioners (NARUC) and the National Association of State Energy Officials (NASEO) Task Force on Comprehensive Electricity Planning is exploring frameworks for greater alignment between long-term electricity resource planning and distribution system planning (NARUC, NASEO 2020).

This manual primarily focuses on single- and multiple-DER analyses since these are the most common BCA applications in use today. Chapter 14 provides an overview of dynamic planning practices and additional context on the evolution of system planning.

³ Jurisdictions have adopted a range of terms to describe planning processes. This manual uses IDP to describe enhanced distribution planning processes that dynamically optimize DERs and IGP to describe aligned full-system planning processes that incorporate generation, transmission, and distribution, including DERs.

BCAs, Implementation Strategies, and EM&V

In addition to BCAs, utilities and others use DER implementation strategies and evaluation, measurement, and verification (EM&V) practices to plan for DERs. The DER implementation strategies include designing effective programs, procurement, and pricing mechanisms to help obtain or promote DERs, and the DER EM&V practices include the assessment of DER performance to determine actual values for achieved benefits and costs. The three approaches work together in an iterative fashion over time to help utilities plan for DERs.

Appropriate implementation strategies and EM&V practices are critical to the successful implementation of DERs, and they are each evolving with the changing roles of single and multiple DERs. For example:

- *DER Implementation Strategies* are changing with some program designs moving from siloed DER funding streams to more integrated procurement of DERs and innovative pricing mechanisms (SEE Action 2020b; SEA Action 2020c).
- *Evaluation, Measurement, and Verification (EM&V)* practices are evolving with the increased use of advanced metering and data analytics, both associated with a growing focus on DERs providing demand flexibility value in terms of temporal and locational impacts (SEE Action 2020a).

The NSPM is agnostic on use of any particular implementation strategy or EM&V approach. Coordinated evolution of the three areas can help to avoid unintended barriers to investment in DERs.



NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 43 of 302

1.5 How to Use this Manual

This manual includes five parts:

- **Part I** presents the NSPM Benefit-Cost Analysis Framework, including fundamental principles and guidance on the development of primary and any secondary cost-effectiveness tests.
- **Part II** describes the full range of potentially relevant DER benefits and costs (i.e., impacts), and presents several cross-cutting considerations on how to account for certain impacts.
- **Part III** provides guidance on single-DER analysis for various types of DER technologies. These chapters discuss key factors that affect BCAs for each type of DER, identify BCA challenges and provide guidance on key issues.
- **Part IV** provides guidance on both multiple-DER analysis and dynamic system planning. It addresses the three main ways that multiple-DER analysis is conducted: for a customer site; for a geographic region; and for an entire utility service territory. Part IV also addresses dynamic system planning practices.
- **Appendices** provide further detail on some topics that warrant additional explanation. This includes information and template tables on reporting BCA results.

The process for conducting a BCA for DERs is summarized Figure 1-1, which also shows where to find information and guidance on the BCA process in specific Parts or Appendices of this manual.

Develop Primary	Develop Secondary	Conduct/Implement	Present BCA Results
BCA Test	BCA Test(s)	BCA	
 Identify applicable policy goals Decide which benefits and costs to include in test Refer to: Parts I-II Appendices A-F 	 Decide which benefits and costs to include in any secondary test(s) Decide how the secondary tests will be used Refer to: Parts I-II Appendices A-F 	 Determine values for each benefit and cost in test(s) Determine which discount rate(s) to use Conduct individual DER or multiple DER BCA Refer to: Parts III-IV Appendices C, F, G, H 	 Present test(s), values of benefits and costs, key assumptions Present results Refer to: Parts I, II and IV Appendices B and D

Figure 1-1. Process of Conducting BCA for DERs

In addition to navigating this manual using Figure 1-1 above, below presents the organization of this document by major Parts, Chapters and Appendices, with a brief description of topics covered.

Users of this manual can refer to both Figure 1-1 and Table 1-2 to help navigate key sections or topics in this manual.

Part/ Chapter	Торіс	Description
Chapter 1	Introduction	Describes the purpose and scope of this manual, with an overview of BCA
Part I	The NSPM Benefit-Cost Ana	lysis Framework
Chapter 2	Benefit-Cost Analysis Principles	Describes fundamental BCA principles that serve as the foundation for remainder of this manual
Chapter 3	Developing BCA Tests	Provides guidance on how to develop a jurisdiction's primary cost-effectiveness test, and any secondary tests
Part II	DER Benefits and Costs	
Chapter 4	DER Benefits and Costs	Presents a catalog of the full range of benefits and costs that may be applicable to specific types of DERs
Chapter 5	Cross-Cutting Benefit and Cost Considerations	Discusses a variety of issues and considerations that span several of the benefits and costs listed in Chapter 4
Part III	Specific DER Types – BCA Iss	sues and Guidance
Chapter 6	Energy Efficiency Resources	
Chapter 7	Demand Response Resources	These chapters describe and provide guidance on key factors that affect the
Chapter 8	Distributed Generation Resources	benefits and costs relevant to specific DER technologies (EE, DR, DG, DS and Electrification)
Chapter 9	Distributed Storage Resources	
Chapter 10	Electrification	
Part IV	Multiple DER Types – BCA Is	sues and Guidance
Chapter 11	Multiple On-Site DERs	Describes how to apply BCA principles and concepts to situations with multiple DER types per customer site, such as grid interactive efficient buildings
Chapter 12	Non-Wires Solutions	Describes how to apply BCA principles and concepts to situations with multiple DER types in a geographic region, such as non-wires solutions
Chapter 13	System-wide DER Portfolios	Describes how to apply BCA principles and concepts to situations with multiple DER types across a utility service territory
Chapter 14	Dynamic System Planning	Provides a brief overview of the key concepts of integrated distribution planning and integrated grid planning
Part V	Appendices	
Appendix A	Rate Impacts	Describes the difference between cost-effectiveness and rate impact analyses, and the role of rate, bill, and participation analyses
Appendix B	Template NSPM Tables	Tables that can be used by jurisdictions to document applicable policies and relevant benefits and costs to inform their BCA
Appendix C	Approaches to Accounting for Relevant Impacts	Provides guidance on options to account for relevant benefits and costs, including hard-to-quantify impacts and non-monetary impacts
Appendix D	Presenting BCA Results	Provides guidance on presenting results in a way that is most useful for making cost-effectiveness decisions
Appendix E	Traditional Cost-Effectiveness Tests	Summarizes the commonly used traditional cost-effectiveness tests from the <i>California Standard Practice Manual</i>
Appendix F	Transfer Payments and Offsetting Impacts	Provides guidance on impacts that appear to be both a benefit to one party and a cost to another party, thereby cancelling each other out
Appendix G	Discount Rates	Describes ways to determine discount rates that are consistent with the jurisdiction's applicable policy goals
Appendix H	Energy Efficiency—Additional Guidance	Describes how to address free-riders and spillover effects in cases for which net savings are used; and treatment of early replacement measures
References		

Table 1-2. Overview of the National Standard Practice Manual for DERs

PART I: THE NSPM BENEFIT-COST ANALYSIS FRAMEWORK

Overview

This part of the manual presents the NSPM BCA Framework that can be used by jurisdictions to assess the cost-effectiveness of DERs.

The NSPM BCA Framework is composed of the following three elements, as shown in Figure 1-2:

- 1. A set of **fundamental principles** that serve as the foundation for jurisdictions assessing the cost-effectiveness of potential DER investments in an economically sound and policy-neutral manner.
- 2. A **multi-step process** for developing a jurisdiction's primary test—a Jurisdiction-Specific Test (JST)—or reviewing whether or to what extent a jurisdiction's current test(s) aligns with the NSPM's fundamental principles.
- 3. Guidance on when and how to use **secondary tests** to (a) inform the prioritization of cost-effective DERs, as determined by a primary JST, and (b) consider marginally non-cost-effective DERs.

Chapter 2 introduces the fundamental principles that serve to guide BCA and inform development of a jurisdiction's primary cost-effectiveness test. Chapter 3 lays out a multi-step process for developing a jurisdiction's primary test and provides guidance on development and use of secondary tests. Part I also references other parts of the NSPM that describe a wide range of considerations and inputs for assessing the cost-effectiveness of specific, individual DERs or multiple DERs.

Figure 1-2. The NSPM Benefit-Cost Analysis Framework

Fundamental BCA Principles Multi-Step Process to Develop a Primary Cost-effectiveness Test

When and How to Use Secondary Cost-Effectiveness Tests

2. BENEFIT-COST ANALYSIS PRINCIPLES

This chapter describes fundamental BCA principles that serve as the foundation for both developing a jurisdiction's primary test and applying the related guidance provided in other sections of the NSPM.

2.1 Summary of Key Points

- Principles are commonly adopted in industries to establish foundational standards by which an industry conducts its practice.
- The NSPM principles serve as a foundation for economically sound and policy-neutral BCA practice to use in the assessment of DER investments.
- The NSPM principles can guide the selection of relevant benefits and costs for a jurisdiction to include in its primary test, as well as any secondary cost-effectiveness tests.
- The NSPM principles can also be used to guide the application of cost-effectiveness testing, selection of a discount rate, and the reporting of the BCA results.
- The NSPM principles can inform the process for choosing and prioritizing DERs to be implemented.

2.2 The Importance of Principles

Principles are commonly adopted in industries to establish foundational standards by which an industry conducts its practice, such as in the fields of accounting, economics, medical practice, and psychology. For example, utility regulators have used the Bonbright Principles for utility ratemaking purposes for decades (Bonbright 1961).

Principles help to guide sound practice and support more efficient decision-making, in particular when applied to the analysis of new situations. When a set of foundational principles is understood, accepted, and applied in an industry, the principles support the decision-makers in making decisions in a clear, transparent, and efficient manner.

2.3 The NSPM Principles

The NSPM principles defined in this manual serve as a foundation for conducting DER BCAs. Developed and informed by DER industry stakeholders,⁴ the NSPM principles are intended to be policy-neutral,⁵

⁴ DER industry stakeholders include those on the NSPM Advisory Group listed on the Acknowledgements page.

⁵ The NSPM is policy-neutral in that it can be used by any jurisdiction, regardless of what policies it has adopted. The manual does not advocate or reject any specific policies but rather sets forth that a jurisdiction's cost-effectiveness practices should align with the jurisdiction's policy goals and objectives.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 47 of 302

technology-neutral, and in line with sound economic practices. They are foundational in guiding regulators and others in the assessment of DER cost-effectiveness.

The NSPM principles provide a construct to guide jurisdictions in their review of *existing* cost-effectiveness practice by testing for alignment with the principles and determining where modifications may be needed to improve alignment. Alternatively, the principles can be applied in developing a *new* primary cost-effectiveness test from the ground up.

The NSPM principles are summarized in Table 2-1 and discussed in the following text. In applying the principles, it is important to note that:

- The NSPM principles are not mutually exclusive as they contain some overlapping concepts; and
- It is up to the users of the NSPM to decide if they will adhere to all the principles in all situations or, for example, if it is necessary to make tradeoffs between certain principles in certain situations.

The NSPM principles in and of themselves do not determine what is a jurisdiction's appropriate cost-effectiveness test for DERs. The NSPM principles are intended to be applied in a manner that takes into consideration the characteristics and circumstances of each jurisdiction's approach to energy resources and can result in different JSTs for different jurisdictions.

Table 2-1. Fundamental NSPM BCA Principles

Principle 1	Treat DERs as a Utility System Resource DERs are one of many energy resources that can be deployed to meet utility/power system needs. DERs should therefore be compared with other energy resources, including other DERs, using consistent methods and assumptions to avoid bias across resource investment decisions.
Principle 2	Align with Policy Goals Jurisdictions invest in or support energy resources to meet a variety of goals and objectives. The primary cost-effectiveness test should therefore reflect this intent by accounting for the jurisdiction's applicable policy goals and objectives.
Principle 3	Ensure Symmetry Asymmetrical treatment of benefits and costs associated with a resource can lead to a biased assessment of the resource. To avoid such bias, benefits and costs should be treated symmetrically for any given type of impact.
Principle 4	Account for Relevant, Material Impacts Cost-effectiveness tests should include all relevant (according to applicable policy goals), material impacts including those that are difficult to quantify or monetize.
Principle 5	Conduct Forward-Looking, Long-term, Incremental Analyses Cost-effectiveness analyses should be forward-looking, long-term, and incremental to what would have occurred absent the DER. This helps ensure that the resource in question is properly compared with alternatives.
Principle 6	Avoid Double-Counting Impacts Cost-effectiveness analyses present a risk of double-counting of benefits and/or costs. All impacts should therefore be clearly defined and valued to avoid double-counting.
Principle 7	Ensure Transparency Transparency helps to ensure engagement and trust in the BCA process and decisions. BCA practices should therefore be transparent, where all relevant assumptions, methodologies, and results are clearly documented and available for stakeholder review and input.
Principle 8	Conduct BCAs Separately from Rate Impact Analyses Cost-effectiveness analyses answer fundamentally different questions than rate impact analyses. Cost-effectiveness analyses should therefore be conducted separately from rate impact analyses.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 49 of 302

PRINCIPLE 1 TREAT DERS AS A UTILITY SYSTEM RESOURCE

DERs are resources that can be used to defer or avoid spending on traditional utility distribution, transmission, and/or generation resources. DERs are also used for increasing reliability, improving power quality, distribution balancing, or enabling the addition of renewables onto the grid. As such, if a utility or other entity is supporting a DER(s)—such as through a direct incentive, time-of-use (TOU) rate, or net metering tariff—and regardless of whether it owns or controls the DER(s), the BCA should ensure consistent treatment with other DERs and resources, including use of consistent cost-effectiveness principles, methodologies, and assumptions. Absent consistent assessment, utilities risk supporting certain resources over others, resulting in potential increased costs for utility customers and missed opportunities for achieving applicable policy goals.

Absent consistent BCA assessment, utilities risk over-investing in some resources while underinvesting in others, resulting in potential increased costs for utility customers and missed opportunities for achieving applicable policy goals.

This principle necessitates that the full range of utility system impacts serve as the foundation of a jurisdiction's primary cost-effectiveness test, upon which other (non-utility system) impacts may be added where applicable to relevant policy goals (per the *Align with Policy Goals* principle). Refer to STEP 2 in Chapter 3 for guidance on applying this principle, and Chapter 4 for a description of utility system impacts.

PRINCIPLE 2 ALIGN WITH POLICY GOALS

Utilities and others often invest in or otherwise support DERs to meet a variety of goals, including policy goals. As such, a jurisdiction's primary cost-effectiveness test should account for the jurisdiction's applicable policy goals. The choice of investments in DERs and other utility resources can materially affect the costs, timeframe, and ability to achieve policy goals. Thus, cost-effectiveness analyses should inform and guide resource choices in that context to ensure alignment with established policies.

Regulators typically have broad statutory authority to set rates that are fair, just, and reasonable; ensure that utilities provide customers with safe and reliable services; ensure a fair and reasonable return on equity; and generally, guide utility actions that are in the public interest. Regulators also typically operate in the context of other relevant jurisdictional policies, some of which affect the implementation of DERs.

Energy and other applicable policy goals evolve over time, responding to changes in the energy industries and society, as well as changing perspectives from policymakers, regulators, utilities, and other industry stakeholders. As such, identifying applicable policies for a jurisdiction is not a static process, but one that should be periodically updated to reflect evolving policy priorities. Therefore, a jurisdiction's cost-effectiveness test(s) may need to periodically evolve as well. Policy goals can be articulated in many different ways, including but not limited to legislation; executive orders; regulations; commission or board guidelines, standards, or orders; a utility's resource planning principles and policies; and requirements of other governing agencies within a jurisdiction. In some cases, a jurisdiction may have different policy goals for different DER types.⁶ STEP 1 in Chapter 3 and Chapter 13 provide guidance for how jurisdictions can analyze DER types as consistently as possible even if different goals exist. Appendix B provides template tables that regulators, utilities, and others can use to take inventory of their applicable policy goals and associated relevant impacts.

PRINCIPLE 3 ENSURE SYMMETRY

Symmetrical treatment in the accounting of benefits and costs is necessary to avoid bias toward any one resource, whereby both benefits and costs are included (or both excluded) for each relevant type of impact.

For example, accounting for utility system impacts should ensure that the full range of utility system benefits and costs be included in the cost-effectiveness analysis. Similarly, to the extent that accounting for host customer impacts and any societal impacts are relevant to a jurisdiction (per its articulated applicable policies, as provided by the *Align with Policy Goals* principle), then both the benefits and costs should be accounted for. If a jurisdiction's applicable policy goals *do not dictate* that DER host customer impacts be accounted for, or are silent on this treatment, then regulators can determine with stakeholder input whether to include host customer impacts, as long as the treatment for the costs and the benefits is consistent and symmetric.

If host customer costs are included in the test, then host customer benefits should also be included (even if certain impacts are hard to quantify and monetize, per the *Account for Relevant, Material Impacts* principle below.) Conversely, the *Ensure Symmetry* principle holds that if a jurisdiction decides not to account for any material host customer benefits, then it should not include any host customer costs.

If each type of impact is not treated symmetrically, the result will be a sub-optimal selection of resources, where the BCA test will result in bias in favor of or against DERs. This will lead to higher than necessary costs imposed upon utility customers and missed opportunities to achieve policy goals.

Ensuring the symmetrical treatment of impacts for both utility and non-utility system impacts is further discussed and illustrated in Chapter 3: STEP 4. If a jurisdiction's applicable policy goals *do not dictate* that DER host customer impacts be accounted for, or are silent on this treatment, then regulators can determine with stakeholder input whether to include host customer benefits and costs, as long as the treatment for the costs and the benefits is consistent and symmetric.

PRINCIPLE 4 ACCOUNT FOR RELEVANT, MATERIAL IMPACTS

Those DER impacts that are deemed to be relevant and material should be accounted for in costeffectiveness tests. The relevant, material impacts should ideally be estimated in monetary terms so that they can be put into useful BCA metrics, such as the present value of revenue requirements, net benefits, and benefit-cost ratios.

As with many utility system resources, some DER benefits and costs are challenging to quantify and monetize. Data may not be readily available, studies to establish values may require a considerable amount of time and/or resources to conduct, and such studies might still result in significant

⁶ Where jurisdictions currently evaluate different DER types using different cost-effectiveness tests or assumptions, it is usually because their BCA practices were developed in different contexts (e.g., different utility commission dockets) without a uniform approach—not because there are any inherent reasons for using different tests for different DERs.

Relevant impacts are those defined by utility system benefits and costs (according to the *Treat DERs as a Utility System Resource* principle), and any non-utility system impacts identified based on applicable policy goals (the *Align with Policy Goals* principle).

Material impacts are those that are expected to be of sufficient magnitude to affect the result of a BCA. If an impact is determined to be immaterial, it should be noted in the BCA. uncertainty. Similarly, there may be cases where some impacts are not counted at all because they are not properly identified, understood, or valued.

Nonetheless, DER benefits and costs that are relevant and are expected to be material should be accounted for even if they are difficult to quantify and monetize. There are a variety of ways to account for some impacts without necessarily monetizing them—such as the use of proxies, use of values from other jurisdictions where applicable, or use of alternative thresholds. Accounting for hard-toquantify impacts qualitatively is another approach to consider. (See Chapter 3: STEP 3 and Appendix C.) In

general, any benefit or cost value used in DER BCAs should be based on a logical, documented, justified method and should not be arbitrary or contrived.

When deciding whether to include a benefit or cost in a BCA test, it is important to distinguish between the definition of the BCA test and the application of the test. Any impact that is deemed to be relevant (as identified by applying Principles 1-3) should be included as part of the definition of the BCA test. In some cases, a benefit or cost may be relevant but not material, and therefore would not need to be included in the application of the BCA test. In this case, the impact is included in the definition of the BCA test but is not included in the application of it. This distinction is especially important when using the same BCA test across multiple DER types and when using the same test over many years.

Using best available information to approximate hard-toquantify impacts, or accounting for impacts qualitatively, is preferable to assuming that those benefits or costs do not exist or have no value. For example, ancillary services are utility system impacts that are material for some DER types, but not others. Therefore they should be included in the *definition* of any BCA test. If a jurisdiction deems that the impacts of a DER on ancillary services are immaterial, the impacts should be set to zero in the application of the BCA test. Ancillary services are thus included in the *definition* of the BCA test but not in the *application* of the test for that DER. If a DER's ancillary services impacts are material, then the impacts should be included in the *application* of the test. This further allows for potential changes in impacts for a particular DER over time.

PRINCIPLE 5 CONDUCT FORWARD-LOOKING, LONG-TERM, INCREMENTAL ANALYSES

Cost-effectiveness analysis of DERs should be forward-looking, capturing the difference between benefits and costs that would occur over the life of the DERs and those that would occur absent the implementation of the DERs.

This principle embodies three inter-related concepts:

• Forward-looking impacts: BCAs should include only forwardlooking impacts because historical costs, also known as sunk costs, cannot be changed and will remain in place under any The term *long-run* refers to the period covering the full life cycle of the resource being analyzed, i.e., for a period of time sufficient to capture the full stream of benefits and costs associated with the resources under analysis. future scenario. Therefore, historical costs are not relevant when comparing future DER implementation scenarios.⁷

- Long-run benefits and costs: BCAs should have a study period that is long enough to include long-run benefits and costs of DERs. This approach is necessary to account for the full benefits and costs of the DER being evaluated, particularly since energy resources, including many DERs and their alternatives, can last decades and thus resource decisions made today can affect costs and benefits far into the future.
- Incremental impacts: BCAs should consider only incremental impacts, i.e., the changes that will occur because of the DERs, relative to a scenario where the DERs are not in place. This concept is applicable to both benefits and costs.

This principle is consistent with utilities having a responsibility to meet utility customer needs in a safe, reliable, and least-cost way over the long term, as well as regulators having a responsibility to protect customers over both the short term and the long term. Over-emphasis on short-term costs may lead to an increase in long-term costs for customers.

PRINCIPLE 6 AVOID DOUBLE-COUNTING IMPACTS

When defining and estimating the values of different DER impacts, it is important to recognize the potential for overlap between some impacts to avoid counting any of them more than once. In particular, caution is required to ensure that costs included in the cost-effectiveness test are not somehow captured in the benefits included in the test, and vice versa. Examples of such overlaps are:

- Utility system risk impacts might overlap with other utility system impacts such as avoided generation costs;
- Costs for complying with environmental regulations might overlap with societal environmental costs; and
- Public health impacts might overlap with either host customer health and safety impacts or societal environmental costs.

Chapter 4 further identifies areas of potential overlap across impacts.

PRINCIPLE 7 ENSURE TRANSPARENCY

DER BCAs require many detailed assumptions and methodologies, and they typically produce detailed results. For regulators, utilities, and other stakeholders to properly assess and understand BCAs—and therefore to ultimately ensure that BCA conclusions are reasonable and robust—key inputs, assumptions, methodologies, and results should be clearly documented in sufficient detail.

Transparent documentation helps to ensure that the approach to cost-effectiveness analysis is consistent with fundamental principles, regulatory objectives, and applicable policy goals. It also facilitates and expedites regulatory and stakeholder understanding and review of cost-effectiveness analyses.

⁷ Historical costs do have important implications for rate impacts and potential cost-shifting between customers. These costs should be considered in a separate rate impact analysis. (See Appendix A.)

Transparency also entails ensuring that stakeholder input allows for review and discussion of the BCA assumptions, methods, and results.

Chapter 3: STEP 5 provides guidance on applying the *Ensure Transparency* principle, with Appendix B and Appendix D presenting template tables for documenting applicable goals, identifying relevant impacts, and presenting BCA results.

PRINCIPLE 8 CONDUCT BCAS SEPARATELY FROM RATE IMPACT ANALYSES

This manual focuses on determining the cost-effectiveness of resource investments. However, many jurisdictions are concerned about rate impacts and this concern is often raised in the context of cost-effectiveness analyses.

DERs, like all energy resources, can affect customer rates. Some DERs are likely to increase rates, while other DERs are more likely to reduce rates. Some DERs might have little effect on rates at all. The extent to which DERs will impact rates depends upon many factors, including the deployment of the DERs, the extent to which they reduce or increase utility system costs, the extent to which they provide utility system benefits, and the extent to which they reduce or increase utility sales.

Cost-effectiveness analyses and rate impact analyses serve different purposes, and it is important to delineate and understand the differences between them to ensure that they both provide meaningful information for DER decision-making. Specifically, these different analyses answer fundamentally different questions: Cost-effectiveness analyses and rate impact analyses serve different purposes, and it is important to delineate and understand the differences to ensure that the different types of analyses provide meaningful information for DER decision-making.

- Benefit-cost analyses answer the question: *Which DERs are expected to provide benefits that exceed costs for all customers on average?*
- Rate impact analyses answer the question: *How much will DERs increase or decrease customer rates, and if so by how much?*

Another key difference between BCAs and rate impact analyses has to do with how the impacts are distributed across customers. While BCAs indicate the extent to which DERs will provide net benefits to customers on average (regardless of how the benefits and costs are distributed across different customers), rate impact analyses indicate how DERs will affect customer rates and therefore any distributional and equity issues created by DERs.

The Rate Impact Measure (RIM) test is one of the traditional cost-effectiveness tests that has been used to assess the rate impacts of DERs. However, because the RIM Test accounts for historical, sunk costs, and is not based on a forward-looking, incremental analysis, the information it provides is of limited value for informing resource investment decisions (see Appendix A). The RIM Test can be used simply to determine whether a DER is likely to increase or reduce rates, and/or to determine whether a rate impact analysis is needed to better understand equity concerns. For example, if the results of the RIM Test are positive or suggest that the rate impacts might be small (e.g., with a benefit-cost ratio of 0.95), then regulators and other stakeholders might decide that there is no need for a rate impact analysis. If the results of the RIM test are much lower, then regulators and utilities might decide to look into rate impacts and equity issues further.

If a jurisdiction decides that a rate impact analysis is warranted, then it is important to conduct a separate long-term rate impact analysis, which includes a bill and participation analysis as well. These analyses address different equity issues:

- *Rate impacts* indicate the extent to which rates for all customers might increase or decrease due to DERs.
- *Bill impacts* indicate the extent to which customer bills might be affected by DERs, both for host customers and other customers.
- *Participation impacts* indicate the portions of customers that will experience bill reductions or bill increases.

Taken together, these three factors indicate the distributional impact on customers from DERs, and therefore the extent to which DERs might lead to equity concerns. It is critical to estimate the rate, bill and participant impacts properly, and to present them in terms that are meaningful for considering distributional equity issues. (See Appendix A.)

Further, while BCAs and rate, bill, and participation analyses should be conducted separately, such analyses should be conducted within a common docket or proceeding to help provide a comprehensive assessment of both cost-effectiveness and equity issues.

3. DEVELOPING BENEFIT-COST ANALYSIS TESTS

This chapter describes provides guidance on how to develop a jurisdiction's primary cost-effectiveness test, the Jurisdiction-Specific Test (JST), as well as any complementary secondary tests to help inform resource investment decisions.

3.1 Summary of Key Points

- A policy-neutral and systematic multi-step process can be used to develop a jurisdiction's primary test for assessing DERs, referred to as the Jurisdiction-Specific Test.
- The JST should be based on the regulatory perspective, the perspective of those that oversee utility DER investment decisions, as guided by a jurisdiction's applicable policy goals. This stands in contrast to traditional cost-effectiveness tests that were not necessarily designed to take this regulatory perspective or applicable policy goals into account.
- The JST used to evaluate DERs answers the fundamental question: Which DERs have benefits that exceed costs and therefore merit acquisition or support from utilities or others?
- Secondary cost-effectiveness tests can be used to provide additional information, represent different perspectives, and answer different questions.

3.2 Cost-Effectiveness Testing Perspectives

3.2.1 Traditional Perspectives

The cost-effectiveness tests traditionally used to evaluate DERs to date consider benefits and costs from different perspectives, and therefore answer different cost-effectiveness questions. The perspectives of the traditional tests, as provided in the 2001 *California Standard Practice Manual* are as follows (See Appendix E):

- Utility Cost Test (UCT), also known as the Program Administrator Cost Test (PACT), which includes the benefits and costs experienced by the utility system.
- Total Resource Cost (TRC) Test, which includes the benefits and costs experienced by the utility system, plus benefits and costs to host customers.
- Societal Cost Test (SCT), which includes the benefits and costs experienced by society.
- *Participant Cost Test (PCT)*, which includes the benefits and costs experienced by host customers. This test supports program design and host customer investment decisions.

• *Rate Impact Measure (RIM) Test*, which indicates whether rates are likely to increase or decrease as a result of DER investments, and therefore primarily represents the perspective of non-host customers.

Over the years, most states have adopted variations of the UCT, SCT, and TRC Test for cost-effectiveness testing, primarily for EE resources. In most cases, these tests in their adopted forms do not adhere to the theoretical definitions of the traditional tests (e.g., such as state use of a modified TRC test) (NESP 2019). In addition, the traditional tests generally do not utilize the perspective of regulators and other decision-makers that are responsible for reviewing DER investment decisions to ensure that they provide net benefits and meet applicable policy goals.⁸

An objective of the NSPM is to support jurisdictions' review of their traditional test practices by applying the NSPM principles and determining the extent of alignment with the principles. This process can help to identify specific modifications or refinements to current BCA practice to improve alignment or can inform the development of a new primary test—a jurisdictionspecific test – that is unique to the jurisdiction.

3.2.2 The Regulatory Perspective

This manual presents the regulatory perspective, which refers to the perspective of regulators or similar entities that oversee utility DER investment decisions. This perspective is guided by the jurisdiction's applicable policy goals and objectives—whether in laws, regulations, organizational policies, or other codified forms in which utilities or energy providers operate. As such, the regulatory perspective reflects the *Align with Policy Goals* principle.

The regulatory perspective is typically broader than the utility perspective, in that it accounts for applicable policy goals (beyond the basic utility goals of providing affordable, safe, reliable electricity services). The regulatory perspective may be narrower than the societal perspective, in that it may not account for some societal impacts if not articulated in statute, regulatory decisions, or other governing policy documents.

The regulatory perspective reflects the utility system perspective plus the values of the jurisdiction's relevant policy goals and objectives. The regulatory perspective could theoretically align with a traditional test perspective, though is more likely to be unique to that jurisdiction. Figure 3-1 illustrates the regulatory perspective relative to the perspectives underlying the traditional cost tests.

The regulatory perspective reflects a jurisdiction's applicable goals that can be articulated in different ways, including but not limited to: legislation; executive orders; regulations; commission or board guidelines, standards or orders; utility resource planning principles and policies; and requirements of other jurisdictional governing agencies.

⁸ The California Standard Practice Manual explicitly notes that policy goals are an "integral" part of cost-effectiveness analyses, but that it does not address policy goals: "The implementing agencies... have traditionally utilized open public processes to incorporate the diverse views of stakeholders before adopting externality values and policy rules which are an integral part of the cost-effectiveness evaluation. These policy rules are not a part of this manual" (CA PUC 2001, page 6). The NSPM BCA Framework differs in that it sets forth the principle that policy goals should be accounted for in determining the primary test for cost-effectiveness analyses.

Figure 3-1. Cost-Effectiveness Testing Perspectives



effectiveness tests.



 Can align with one of the traditional test perspectives, but not necessarily.

The regulatory perspective is necessary to ensure that a jurisdiction's relevant policy goals are considered in its cost-effectiveness test in order to avoid sub-optimal resource selections with respect to stated goals. Otherwise the jurisdiction might incur additional costs to meet its policy goals.

Developing a Jurisdiction-Specific Test 3.3

Developing a jurisdiction's primary cost-effectiveness test-the JST—is foundational to the BCA process.

Table 3-1 presents the key steps to developing a jurisdiction's primary test. The following subsections describe each of these steps in more detail. Further information on implementing BCA tests is presented in Chapters 6–14, and examples of reporting BCA results are presented in Appendix D.

Because the primary test is based upon each jurisdiction's applicable policy goals, and those goals can vary across jurisdictions, the JST may take a variety of forms across jurisdictions. This outcome can have the disadvantage of making it difficult to compare BCA results across jurisdictions absent sufficient transparency and information. Yet, it has the advantage of allowing each jurisdiction to tailor its primary test to reflect its own goals and objectives.

A jurisdiction's primary test—the Jurisdiction Specific Test-should reflect the regulatory perspective to ensure proper accounting of the full range of utility system impacts and reflect the jurisdiction's applicable policy goals. Such a primary test can be used to answer the critical question: Which DERs have benefits that exceed costs and therefore merit utility acquisition or support on behalf of customers?
Table 3-1. Developing a Jurisdiction's Primary Test: A 5-Step Process

STEP 1 Articulate Applicable Policy Goals

Articulate the jurisdiction's applicable policy goals related to DERs.

STEP 2 Include All Utility System Impacts

Identify and include the full range of utility system impacts in the primary test, and all BCA tests.

STEP 3 Decide Which Non-Utility System Impacts to Include

Identify those non-utility system impacts to include in the primary test based on applicable policy goals identified in Step 1:

• Determine whether to include host customer impacts, low-income impacts, other fuel and water impacts, and/or societal impacts.

STEP 4 Ensure that Benefits and Costs are Properly Addressed

Ensure that the impacts identified in Steps 2 and 3 are properly addressed, where:

- Benefits and costs are treated symmetrically.
- Relevant and material impacts are included, even if hard to quantify.
- Benefits and costs are not double-counted.
- Benefits and costs are treated consistently across DER types.

STEP 5 Establish Comprehensive, Transparent Documentation

Establish comprehensive, transparent documentation and reporting, whereby:

- The process used to determine the primary test is fully documented.
- Reporting requirements and/or use of templates for presenting assumptions and results are developed.

Note: The 5-step process is not necessarily chronological in order and often requires iteration.

STEP 1 ARTICULATE APPLICABLE POLICY GOALS

This first step applies the *Align with Policy Goals* principle, which states that a jurisdiction's primary costeffectiveness test should account for applicable policy goals. Documenting applicable goals at the outset of either developing a JST or reviewing current BCA tests and practices is necessary to ensure that the cost-effectiveness test explicitly and properly accounts for such goals, and therefore the jurisdiction's regulatory perspective.

Policies typically take the form of mandates or objectives.

 Policy Mandates: Policy goals can include specific mandates for investing in or supporting DERs. For example, legislation might require utilities to achieve a certain amount of efficiency savings or distributed generation capacity by a specific year. In these cases, legislators have determined (or presumed) that the benefits associated with their policy directive exceed the costs. In these cases, utilities need to comply with the mandate, provided that a funding mechanism is available. Nonetheless, regulators and others may conduct BCAs for statutorily required DERs to (a) select the most cost-effective DERs within the statutorily mandated requirements by comparing the cost-effectiveness of different DERs and different implementation approaches; and (b) identify ways to reduce costs and increase benefits of those resources.

Policy Objectives: Policy goals can be specific about meeting certain objectives without
identifying how they should be met. For example, a jurisdiction might have a policy goal of
increasing demand flexibility or serving low- and moderate-income customers, without
specifying any particular technology or way to meet the goal. In these cases, BCAs should
identify the most cost-effective resources for achieving the goal by comparing the attributes,
including cost-effectiveness, of the different resources being evaluated to meet the intended
objectives.

Examples of energy and non-energy related policy goals that may be applicable in some jurisdictions are provided in Table 3-2.⁹

Table 3-2. Examples of Policy Goals

Common Overarching Goals: Provide safe, reliable, reasonably priced electricity and gas services; support fair and equitable economic returns for utilities; promote customer equity; protect/reduce energy burden for low-income and vulnerable customers.

Resource Goals: Reduce electricity and gas system costs; develop least-cost energy resources; improve system reliability and resiliency; reduce system risk; promote resource diversity; increase energy independence; reduce price volatility; provide demand flexibility.

Other Applicable Goals: Ensure stable energy markets; reduce environmental impact of energy consumption; promote jobs and local economic development; improve health associated with reduced air emissions and better indoor air quality.

Non-utility system impacts should be included *only if consistent with achieving applicable policy goals*. In such cases, policy goals may be very general. For example, a jurisdiction might have policy goals to reduce low-income energy burdens, enhance local economic development, or reduce greenhouse gas (GHG) emissions, without specific utility system targets or requirements for each goal. Whether to include such general policy goals may be up to interpretation by regulators, and any such decisions should be informed by meaningful stakeholder input. Metrics can be used to indicate how well policy goals are being achieved, and the contribution of each DER to achieving those goals.

In some cases, it may be challenging to identify and articulate a jurisdiction's applicable DER policy goals, such as when jurisdictions have different policy goals for different DER types, making it difficult to establish a single primary test for all DER types. In these cases, there are several approaches for determining which policy goals to account for in the primary DER test (see STEP 4 and Section 3.5.)

Finally, the decision on how best to align policy objectives consistently across multiple DER analyses could have significant implications for the primary DER test and the DER assessments. Stakeholder input can help to inform the regulator's determination of the most appropriate approach (see STEP 5.)

⁹ This list is not intended to be exhaustive, nor is it intended to imply a recommendation of any policies for any jurisdiction. It is intended to illustrate the types of policies that jurisdictions typically establish.

STEP 2 INCLUDE ALL UTILITY SYSTEM IMPACTS

This step of including all the utility system impacts in a JST applies the *Treat DERs as a Utility System Resource* principle, which sets forth that DERs be compared consistently with other utility resources. This emphasis on utility system impacts is important because these are the impacts that directly affect the utility customers that are supporting utility-sponsored DERs in one form or another. Including the utility system impacts in a costeffectiveness test ensures that the test will, at a minimum, indicate the extent to which total utility system costs will be reduced (or increased) by the DER.

Utility system benefits and costs provide the foundation for every JST.

- The term "utility system" here is used to represent the entire energy system that provides services to retail customers.
- The three most historically applied costeffectiveness tests, the UCT, the TRC, and the SCT, all include all utility system impacts, at a minimum.

This step requires accounting for all utility system impacts that will be affected by the DER. Specifically:

- For electric utilities, utility system impacts include the generation, transmission, and distribution of electricity services.
- For gas utilities, utility system impacts include the transportation, storage, and distribution of gas services.

Utility system impacts should be accounted for irrespective of the type of utility structure (e.g., investorowned or publicly owned utilities) or market structure. In particular, for a jurisdiction with competitive wholesale markets and distribution-only electric utilities, it is important to account for the impacts on generation, transmission, and distribution because all these services will be affected by the DER and distribution customers have to pay for all those services—even if the utility funding or supporting the DER provides only distribution services.¹⁰ Utility system impacts are generally defined as follows:

- Utility system benefits typically include all the utility system costs that would be avoided or deferred by implementing the DER. These avoided costs are one of the more important and sometimes challenging inputs to any BCA of DERs and will significantly affect the results of the analyses. Therefore, it is essential to ensure that avoided cost estimates are comprehensive, up to date, informed by stakeholders, and ultimately reviewed and approved by regulators or other decision-makers.
- Utility system costs typically include the portion of the DER paid by the utility, other financial or technical support provided to host customers, and interconnection costs and distribution system upgrades not paid by the DER owner. These costs also generally include any other utility system costs associated with DER implementation, administration, marketing, evaluation, measurement, and verification.

Accounting for Policies that affect Utility System Impacts: As an example, consider a jurisdiction that has a policy goal of reducing GHG emissions from the utility sector by 30% by 2030. Utilities are going to have to incur costs to meet that goal with or without the use of DERs. In this case, the cost of achieving this policy goal, and the associated benefits, are considered utility system impacts and should be included in the JST. See STEP 2 for further guidance.

¹⁰ Additionally, it is important that the avoided cost impact assessment of the BCA be aligned with utility planning assumptions in order to be consistent with "actual" avoided costs. For example, if the BCA uses one method to calculate capacity reduction needs, but the utility uses another method and calculates a different value, the BCA will either over- or underestimate what the utilities actual avoided costs end up being (which is dependent on the utility's specific planning and operational practices).

See Chapter 4 for descriptions of DER benefits and costs within each of these categories, and Chapter 5 for discussion of cross-cutting considerations and factors that can affect the magnitude of impacts.

STEP 3 DECIDE WHICH NON-UTILITY SYSTEM IMPACTS TO INCLUDE

This step applies the *Align with Policy Goals* principle, which sets forth that a JST should account for applicable policy goals. Once a jurisdiction's applicable policies have been identified and articulated in STEP 1, and utility system benefits and costs are identified and accounted for in STEP 2, this next step involves deciding which non-utility benefits and costs to include in the test, based on applicable policy goals. These non-utility impacts can be categorized as host customer impacts and societal impacts.

In some cases, the determination of whether to include certain non-utility system impacts is fairly straightforward. For example, legislation establishing a specific DER program might explicitly state that the goals of the program are to create jobs and improve public health. Or a state might have statutory language requiring them to consider impacts such as environmental benefits in resource decision-making. In other cases, the determination may be less clear. This is often the case regarding host customer benefits and costs, where policies do not necessarily articulate whether they should be accounted for in a primary cost-effectiveness test. The decision of which non-utility system impacts should be included in a JST, in order to reflect policy goals, will need to be made by regulators with appropriate input from stakeholders.

Table 3-3 presents an illustrative list of non-utility impacts that could be included in a primary test to the extent they are relevant to a jurisdiction. See Chapter 4 for descriptions of DER benefits and costs within each of these categories, and Chapter 5 for discussion of cross-cutting considerations and factors that can affect the magnitude of impacts.

Non-Utility Impact	Description
Other fuel impacts	Impacts on fuels that are not provided by the relevant utility, for example, electricity (for a gas utility), gas (for an electric utility), oil, propane, gasoline, and wood
Host customer impacts	Host customer portion of DER costs and host customer non-energy impacts (NEI), such as impacts on productivity, comfort, health and safety, mobility, and more
Impacts on low-income customers	Impacts that are different from or incremental to non-low-income customer impacts such as energy affordability and poverty alleviation
Environmental impacts	Impacts associated with GHG emissions, criteria pollutant emissions, land use, solid waste, etc.; includes only those impacts not embedded in the utility cost of compliance with environmental regulations, which should always be treated as a utility system cost
Public health impacts	Impacts on public health; includes health impacts that are not included in host customer impacts or environmental impacts and includes benefits in terms of reduced healthcare costs
Economic development and jobs	Impacts on direct and indirect economic development and jobs
Energy security	Reduced reliance on fuel or energy imports from outside the state, region, or country

Table 3-3. Commonly Considered Non-Utility System Impacts

This table is presented for illustrative purposes and is not meant to be an exhaustive list or applicable in every jurisdiction.

STEP 4 ENSURE THAT BENEFITS AND COSTS ARE PROPERLY ADDRESSED

STEP 4 in developing a jurisdiction's primary test involves ensuring that the impacts identified in STEP 2 and STEP 3 are properly addressed, whereby:

- Benefits and costs are treated symmetrically;
- Relevant and material impacts are included, even if hard to quantify;
- Benefits and costs are not double-counted; and
- Benefits and costs are treated consistently across DER types.

Benefits and Costs are Treated Symmetrically

The *Ensure Symmetry* principle in Chapter 2 sets forth that cost-effectiveness practices be symmetrical, where both benefits and costs are included (or in some cases may be excluded) for each relevant type of impact. This principle applies to different categories of impacts as follows:

- Utility System Impacts: If the full range of utility system costs are included, then the full range of utility system benefits should be included in the cost-effectiveness analysis, even if hard to quantify (consistent with the Account for Relevant, Material Impacts principle).
- *Host (participant) customer impacts:* If regulators and others have decided to account for host customer impacts in the jurisdiction's primary BCA test (per its articulated applicable policies, as provided by the *Align with Policy Goals* principle) then both costs and benefits experienced by those customers should be included in the test. On the cost side, this would most commonly be the host customer portion of the DER installation costs. On the benefits side, depending on the resource, there may be a variety of non-energy benefits experienced by the host customer (e.g., improved reliability, improved comfort, improved productivity, etc.). If regulators and others are reluctant to include any or some portion of the material host customer benefits in the BCA, then the host customer costs should be treated consistently (see example below.)
- Other non-utility system impacts: If a jurisdiction's applicable policies indicate that a certain category of costs should be included in its cost-effectiveness test (e.g., other fuel, water, low income, environmental, public health, economic development) then the comparable benefits for that category should be captured in the test as well, and vice versa.

Example: Ensuring Symmetry in the Treatment of Host Customer Benefits and Costs

The *Ensure Symmetry* principle provides that benefits and costs should be accounted for symmetrically in BCAs, whether relevant to utility system or non-utility system impacts. The following example illustrates the importance of this principle, and how treatment of host customer impacts can affect the BCA results of a DER.

In this example, it costs a total of \$9,000 to install a DER measure at a host customer site, comprised of total measure cost (\$7,500) and utility administrative cost (\$1,500). The host customer pays the DER cost (\$5,625), which complements the utility incentive (\$1,875). The host customer is expected to experience non-energy benefits (\$4,000).*

In Scenario A, the BCA accounts for the host customer costs (\$5,625), but not the host customer benefits (\$4,000). This asymmetrical treatment of costs and benefits suggest the program is not cost-effective, with a BCA of 0.67. Yet this scenario fails to recognize the non-energy related benefits to the host customer (e.g., increased comfort, reduced indoor air pollution, increased asset value, etc.).

In Scenario B, symmetry is achieved by including both the host customer costs and benefits. In this case, a jurisdiction has conducted analyses to estimate monetary values of the non-energy benefits (see Appendix C). This symmetrical treatment suggests the program is cost-effective with a BCA ratio of 1.11.

In Scenario C, neither host customer costs nor benefits are included in the BCA. The scenario assumes that the jurisdiction has decided to not include host customer impacts in its primary test. If a jurisdiction is reticent to quantify host customer benefits, then exclusion of both costs and benefits assures symmetrical treatment of impacts. This symmetrical treatment suggests the program is cost-effective with a BCA ratio of 1.78.

	Asymmetrical	Symme	etrical
Costs and Benefits	A. Host Customer Costs Included , Benefits Excluded	B. Host Customer Benefits and Costs Both Included	C. Host Customer Benefits and Costs Both Excluded
DER Costs		-	
Utility System Costs	\$7,500		
 Rebate/incentive 	\$1,875	\$1,875	\$1,875
 Administrative Costs 	\$1,500	\$1,500	\$1,500
Host Customer Costs	\$5,625	\$5,625	
Total Costs Accounted for:	\$9,000	\$9,000	\$3,375
DER Benefits			
Utility System Benefits	\$6,000	\$6,000	\$6,000
Host Customer Non-Energy Benefits		\$4,000	
Total Benefits Accounted for:	\$6,000	\$10,000	\$6,000
Net Benefit	(\$3,000)	\$1,000	\$2,625
Benefit-Cost Ratio:	0.67	1.11	1.78

Illustrative Example: Treatment of Host Customer Costs and Benefits

Both Scenarios B and C adhere to the *Ensure Symmetry* principle, and therefore either would be appropriate to apply to a JST, depending upon how the jurisdiction chooses to account for host customer impacts. This example does not include other possible non-utility system impacts, for simplicity, but those impacts would also be included in the JST if consistent with applicable policy goals.

*Even if regulators and others decide to include host customer impacts in its primary test, the host customer benefits from bill savings should not be included in the test because (a) bill savings represent lost revenues, which should not be included in BCA tests (per the *Conduct BCAs Separately from Rate Impact Analyses* principle), and (b) bill savings overlap considerably with utility system avoided costs which are already included in the test, and double-counting should be avoided (per the *Avoid Double-Counting Impacts* principle.) Applying the principle of symmetry sometimes requires estimating "net" impacts for certain types of benefits. For example, if environmental or economic development benefits are included in the cost-effectiveness framework, it is important to also include environmental and economic development costs as well.

Account for Relevant Impacts, even if Hard to Quantify

Once a jurisdiction has identified which relevant impacts of DERs to include in its primary costeffectiveness test, these impacts should be accounted for in monetary terms to the extent that it is practical to do so. If an impact is deemed to be immaterial to the test (i.e., to be so small that it is not likely to affect the outcome of test) then it does not need to be included when applying the primary test. This is especially applicable if it would be burdensome or expensive to develop reasonable inputs for the impact.

For example, ancillary services are clearly a utility system impact and therefore should be included in any BCA test. However, for some DERs the impacts on ancillary services might be immaterial and, therefore, do not need to be included when applying the primary test. For other DERs, ancillary services might be material and therefore should be included when applying the test. In both cases, ancillary services are included in the *definition* of the primary test. Yet for one DER, they are not estimated for the *application* of the test because they are deemed immaterial, while in another case they are estimated.

Some approximation may be necessary to establish values of certain relevant impacts that are deemed to be material. Table 3-4 summarizes six different approaches that can be used to account for material impacts of DERs (see Appendix C). While some of these approaches represent approximations and include some uncertainties, it is better to use the best available approximation for a material impact than to assume it does not exist or that its value is zero. Any benefit or cost value used in DER BCAs should be based on a logical, documented, justified method and should not be arbitrary or contrived.

While some DER impacts are difficult to quantify in monetary terms—either due to the nature of the impact or the lack of available information about the impacts—approximating hardto-quantify impacts is preferable to assuming that the relevant benefits and costs do not exist or have no value.

Approach	Description
Jurisdiction-specific studies	Jurisdiction-specific studies on DER benefits and costs offer what can be the most accurate approach for estimating and monetizing relevant impacts.
Studies from other jurisdictions	If jurisdiction-specific studies are not available, studies from other jurisdictions or regions, as well as national studies, can be used for estimating and monetizing relevant impacts.
Proxies	If monetized impacts are not available, well-informed and well-designed proxies can be used as a simple substitute. For example, the benefits of a DER could be increased by a specified percentage (e.g., x% adder) to account for difficult-to-quantify benefits.
Alternative thresholds	Pre-determined thresholds for determining cost-effectiveness that are different from a 1.0 benefit-cost ratio (BCR) can be used as a simple way to account for relevant impacts that are not otherwise included. For example, resources with a BCR of 0.95 could be deemed cost-effective as an alternative to monetizing difficult-to-quantify benefits.
Accounting for non-monetized impacts	Relevant qualitative information can be used to consider impacts that cannot be monetized.

Table 3-4. Different Approaches to Account for All Relevant Impacts

Avoid Double-Counting Benefits and Costs

The Avoid Double-Counting Impacts principle requires not counting impacts more than once and not ignoring impacts because they appear to overlap with others. This can be achieved by clearly defining and valuing each benefit and cost to identify where there might be overlaps or gaps.

Accounting for risk and resilience provides an example of how double-counting can occur. Utilities and host customers might both experience risk and resilience benefits from DERs. For cost-effectiveness tests that account for both utility system and host customer impacts, it is important to ensure that some or all risk and resilience benefits are not counted more than once.

Treat All DERs Consistently

The *Treat DERs as a Utility System Resource* principle states that all DERs types should be compared with other energy resources, and should be evaluated using consistent cost-effectiveness principles, methodologies, and assumptions. This begins with using the same primary cost-effectiveness test for all DERs, whether a jurisdiction is conducting separate single-DER analyses or multi-DER analyses.

In some cases, STEP 1 of the process, *Articulate Applicable Policy Goals*, may be challenging to apply where jurisdictions have different policy goals for different DER types, making it difficult to establish a single primary test for all DER types. For example, a jurisdiction might have a statute or a commission order indicating that the environmental benefits of distributed PV should be accounted for in BCAs, but no such requirement for EE or other DER types. In these cases, there are several options regulators can use to determine which policy goals to account for in the primary test for all DERs:

• Use a narrow interpretation of policy goals, where the primary test accounts for only those policy goals that are applicable to *every* DER type.

If different primary tests are used for different DER types, the BCAs may lead to sub-optimal DER choices, resulting in increased costs to customers and missed opportunities to achieve policy goals.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 66 of 302

- Use a broad interpretation of policy goals, where the primary test accounts for every policy goal that is applicable to *any* DER type.
- Do something in between where the *highest priority* policy goals associated with *any* DER type are included regardless of how frequently they apply to different DER types.

The decision on which approach to take could have significant implications for the primary DER test and the DER assessments. Consistent with the *Transparency* principles, stakeholder input is important for informing the regulator's determination of how best to align policy objectives consistently across the multiple DER analysis.

Regulators, with stakeholder input, may need to interpret legislation, regulations, orders, and other policy directives to determine what they imply about the primary costeffectiveness test.

Appendix B provides template tables that jurisdictions can use to articulate and assess applicable policy goals.

If a jurisdiction is unable to develop a single primary DER test, secondary tests could be used to promote consistency in the BCA of all DERs. For example, the primary test could be based on a broad interpretation of policy goals, while a secondary test could be based on a narrow interpretation of policy goals. (See Section 3.5.)

Treating DERs consistently in BCAs also requires that benefit and cost assumptions are consistent across DER types. The inputs and per-unit values used for the impacts of different DER types should be the same or based on the same methodologies and assumptions, accounting for differences in magnitude, timing, or location where warranted. For example, the values for avoided energy or avoided generation capacity for any given time or location should be the same for all DER types.

STEP 5 ESTABLISH COMPREHENSIVE, TRANSPARENT DOCUMENTATION

This final step in developing a primary BCA test applies the *Ensure Transparency* principle, whereby costeffectiveness practices document all relevant inputs, assumptions, methodologies, and BCA results. Such documentation can include, for example:

- An inventory of applicable policy goals used to identify impacts for the primary test;
- A description of the utility system benefit and cost assumptions, including sources of values;
- A description of the non-utility system benefit and cost assumptions, including sources of values;
- BCA modeling parameters such as the study period and discount rates; and
- A summary of the results of the BCA, including all the benefits and costs, the monetary results, the non-monetary results, and the justification of the ultimate resource decision.

Appendix B provides template tables a jurisdiction can use to take inventory of its applicable policies, and to help identify associated benefits and costs. Chapter 13 and Appendix D provide guidance on how to present BCA results in ways that are most informative and most useful in making cost-effectiveness decisions. It also provides guidance on how to present results to support efforts to prioritize across different DER types.

Further, the *Ensure Transparency* principle extends to the *process* of developing a jurisdiction's costeffectiveness practices. This includes primary and secondary test considerations, where the above information is shared and reviewed with input from interested stakeholders to allow for robust review and input. Such stakeholders include utilities, evaluators, consumer advocates, low-income representatives, state agencies, DER industry representatives, environmental advocates, and others. Stakeholder input is important at the outset of the BCA process for assessing and interpreting applicable policy directives, as they can provide critical viewpoints regarding the value of DERs in the context of the jurisdiction's policy goals.

Stakeholder input can be achieved through a variety of means, as appropriate for a given jurisdiction, such as via:¹¹

- Rulemaking processes,
- Generic jurisdiction-wide dockets,
- Utility commission orders on specific DERs, and/or
- Working groups or technical sessions.

The process should address objectives based on current jurisdiction policies and should also be flexible to address new or modified policies that are adopted over time.

Some jurisdictions may wish to incorporate input from government agencies or representatives that do not typically make decisions regarding DER cost-effectiveness but would nonetheless have insights on the jurisdiction's applicable policy goals. For example, a state's public utility commission may wish to incorporate input from that state's department of environmental protection or department of health and human services.

3.4 Jurisdiction-Specific Tests and Traditional Tests

When a jurisdiction develops its own JST using the NSPM BCA Framework, or refines an existing test using the NSPM process, it might result in a test that is different from one of the traditional tests or it might result in a test that is identical to the theoretical definition of a traditional test. Table 3-5 provides a high-level comparison of the use of a JST relative to the traditional tests.

¹¹ In many jurisdictions, BCA processes involve proceedings with expert testimony, where supporting information, assumptions and practices must be admissible and reliable, and must stand up to discovery, and where applicable, to cross-examination and litigation.

Test	Perspective	Key Question Answered	Categories of Benefits and Costs Included
Jurisdiction- Specific Test	Regulators or decision-makers	Will the cost of meeting utility system needs, while achieving applicable policy goals, be reduced?	Includes the utility system impacts, plus those impacts associated with achieving applicable policy goals
Utility Cost Test*	The utility system	Will utility system costs be reduced?	Includes the utility system impacts
Total Resource Cost Test	The utility system plus host customers	Will utility system costs and host customers' costs collectively be reduced?	Includes the utility system impacts, plus host customer impacts
Societal Cost	Society as a whole	Will total costs to society be reduced?	Includes the utility system impacts, plus host customer impacts, plus societal impacts such as environmental and economic development impacts

Table 3-5. Jurisdiction-Specific Test Compared with Traditional Tests

*Also referred to as the Program Administrator Cost (PAC) Test.

Figure 3-2 presents illustrative primary JSTs, as compared to traditional tests. The figure shows that:

- 1. In all cases, the full range of utility system impacts are accounted for, consistent with the *Treat DERs as a Utility System Resource* principle.
- 2. JST 1, 2, and 3 provide examples of where a JST aligns with the UCT, the TRC test, and the SCT, respectively, *assuming* in each case the JST reflects that jurisdiction's specific applicable policy goals.
- 3. In the JST 4 example, the jurisdiction's specific policies lead to a primary test that is different from any of the traditional tests. This example assumes that the hypothetical jurisdiction has statutes, regulations, orders, or other policy directives that require or suggest that these impacts be accounted for when assessing DER cost-effectiveness.

Importantly, Figure 3-2 illustrates the flexibility provided by the JST.



Figure 3-2. Example Jurisdiction-Specific Test Relative to Traditional Tests

3.5 Developing Secondary BCA Tests

While a jurisdiction's primary test should be used to inform whether a utility should fund or otherwise support DERs, it does not have to be utilized in a vacuum. In some instances, secondary tests can help enhance regulators' and stakeholders' overall understanding of DER impacts by answering other questions regarding utility DER investments.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 70 of 302

The key uses for secondary tests include:

- Informing decisions on how to prioritize DERs (including reviewing DERs from perspectives other than the regulatory perspective, such as the host customer perspective);
- Informing decisions regarding marginally costeffective DERs; and
- Encouraging consistency in BCA analyses across different DER types.

While the primary test, the JST, should be used as the best indication of costeffectiveness, and thus given the most weight, resource selection may also be supported by one or more secondary tests. For example, secondary tests can indicate the relative priority of different DERs, or how to treat DERs that are marginally cost-effective.

Different tests provide different information about the cost-effectiveness and impacts of DERs. However, secondary tests should be used cautiously to ensure that they do not make the BCA decisionmaking process burdensome or undermine the purpose of the primary test.

There is a range of options that regulators, utilities, and other stakeholders can consider for secondary tests. Examples include:

- Utility Cost Test. The UCT might be useful as a secondary test to provide a benchmark to compare all DER types. This test provides a useful indication of how DERs will affect utility system costs.
- Societal Cost Test. The SCT might be useful as a secondary test to indicate how cost-effective DERs would be when accounting for the full range of potential benefits and costs.
- A "Narrow Secondary test" based on a narrow interpretation of the jurisdiction's policy goals, where the test accounts for only those policy goals that are applicable to every DER type.
- A "Broad Secondary test" based on a broad interpretation to the jurisdiction's policy goals, where the test accounts for every policy goal that is applicable to any DER type. This will help achieve consistency but may give undue weight to some policy goals.
- A secondary test that includes (or excludes) a particular non-utility system impact that regulators and stakeholders want to investigate separately from the primary test. This might be useful for conducting a sensitivity analysis of that non-utility system impact.

Decisions on which impacts to include in secondary tests should be driven by the purpose(s) of the secondary analyses. The main purposes of secondary tests are discussed below.

3.5.1 Inform Decisions on How to Prioritize DERs

While the primary test indicates which DERs merit utility acquisition on behalf of customers, some jurisdictions may choose not to invest in all cost-effective DERs due to funding limits, equity concerns, or other constraints. In these cases, regulators, utilities, and others may need to make choices among cost-effective DERs to decide which ones to implement.

Regulators and stakeholders may choose to prioritize programs based on the results of the primary test. For example, DERs could be ranked according to their net benefits or benefit-cost ratios, and the lowercost ones could be chosen over the higher-cost ones.

Alternatively, regulators with input from stakeholders may choose to use a secondary test to help prioritize DERs. For example, the UCT could be useful for prioritizing DERs because it focuses on the direct benefits and costs to the customers that are funding the utility-sponsored DER. Or regulators may

decide not to invest in DERs that, for the host customer, have a very short payback period (e.g., less than one year as indicated by the PCT). (See Chapter 13.)

3.5.2 Inform Decisions Regarding Marginally Cost-Effective Resources

In cases where a DER is marginally cost-effective, secondary tests can be used to decide whether the utility should fund or otherwise support the DER. Secondary tests provide regulators and other stakeholders with additional information to assess whether certain DER investments will provide net benefits.

For example, consider a jurisdiction where stakeholders are divided on whether to account for the benefit of GHG reductions in the primary test, and the regulators decide not to account for them. In this case, a secondary test that does account for the benefits of GHG reductions might be useful. If a proposed DER has a benefit-cost ratio under the primary test of just below 1.0, and a benefit-cost ratio under the secondary test of nearly 2.0, then regulators with input from stakeholders might choose to invest in this DER because of the GHG benefits. Alternatively, if the benefit-cost ratio under the primary test is 0.85, and 1.05 under the secondary test, regulators and stakeholders might choose to not invest in this DER.¹²

If secondary tests are used to inform decisions regarding marginally cost-effective DERs, regulators and other stakeholders may want to establish protocols for how they should be used. Figure 3-3 presents an example of protocols that could be used for this purpose.

¹² In addition to showing how including a particular impact in the BCA test will affect the BCA results, sensitivity analyses can also be used to show how different estimates for an impact (e.g., low, medium, high estimates) will affect the BCA results.



Figure 3-3. Example Use of Secondary Tests for Marginally Cost-Effective DERs

Table 3-6 presents some examples of how secondary tests could be used to make decisions regarding marginally cost-effective DERs. In this example, the primary test includes all utility system impacts, low-income impacts, and other fuels. The secondary test includes those impacts plus benefits of reducing GHG emissions. The conclusions are summarized as follows:

- *Case 1.* The primary test indicates that the DER is cost-effective. There is no need to apply a secondary test. The DER is accepted for utility funding or support.
- *Case 2.* The DER is accepted for utility funding or support even though the primary test indicates that the DER(s) is not cost-effective.
- *Case 3.* The primary test indicates that the DER is marginally cost-effective, but the secondary test indicates that the DER is not cost-effective even after accounting for GHG emissions. The DER is not accepted for utility funding or support.
- *Case 4.* The primary test indicates that the DER is not even marginally cost-effective, and so is not accepted for utility funding or support despite the secondary test indicating cost-effectiveness.

	Case 1	Case 2	Case 3	Case 4
Primary test (JST)	2.1	0.96	0.96	0.8
Secondary test	Not needed	1.8	0.98	1.8
Investment Decision	accept	accept	reject	reject

Table 3-6. Examples of Secondary tests Used to Consider Marginal DERs

Using secondary tests in this way may diminish the prominence of the primary JST test, as shown in Case 2. On the other hand, the secondary test alone may not determine investment decisions, as shown in Case 4. This implies that secondary tests should be used carefully and with full documentation of the process and bases for making investment decisions.

3.5.3 Promote Consistency Across Multiple DER Types

As described in Chapter 2, the *Treat DERs as a Utility System Resource* principle requires that all DER types be evaluated using consistent cost-effectiveness principles, methodologies, and assumptions.

If a jurisdiction analyzes multiple DER types together, then the best way to ensure consistency across DER types is to use the same primary cost-effectiveness test for all DER types. See Part IV for guidance and case studies on developing multiple-DER tests. If a jurisdiction is having difficulty developing a single primary test to evaluate all DERs, (for instance, due to attributes of each DER or how different DERs are addressed in the jurisdiction's policies) then secondary tests can again be helpful to promote consistency across DERs. Several options are available for secondary tests. (See Section 3.5.)

A Narrow Test

If regulators and others choose to apply a single primary test that accounts for the *broadest* set of policy goals, then a secondary test that accounts for a *narrower* set of policy goals associated with one DER type could be designed to indicate how the BCA results would be different for that DER type under that test.

If that DER type is found to be cost-effective under both tests, then no more investigation is needed. If that DER is found to be cost-effective under the primary test, but not under the narrower secondary test, additional investigation might be warranted. The DER might be deemed cost-effective if the secondary test results indicate that it is marginally cost-effective (for example, with a benefit-cost ratio slightly below 1.0). The DER might be deemed not cost-effective if the secondary test results indicate that it is not even marginally cost-effective (for example, with a benefit-cost ratio well below 1.0).

A Broad Test

If regulators choose to apply a single primary test that accounts for the *narrowest* set of policy goals, then a secondary test that accounts for a *broader* set of policy goals associated with one DER type could be designed. Again, if that DER type is found to be cost-effective under both tests, then no additional investigation is warranted. If not, then regulators and others might want to consider the results of the secondary test.

3.5.4 Other Uses for Secondary Tests

There are other situations where a cost-effectiveness test that is not the primary test is useful for purposes other than cost-effectiveness analyses. For example, the UCT can be helpful in mitigating equity concerns through program design. The financial incentives offered to DER host customers could be capped at a level equal to the utility system avoided costs. This would prevent customers that do not host DERs from paying more than the benefits they receive from the DER resource.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 74 of 302

PART II: DER BENEFITS AND COSTS

Overview

This part of the manual describes the types of benefits and costs that can be created by DERs.

Chapter 4 presents a catalog of the range of benefits and costs that might be applicable to any DER type, including a brief description of each type of impact. The benefits and costs are organized by:

- electric utility system,
- gas utilities and other fuel systems,
- host customer, and
- societal impacts.

Chapter 5 discusses a variety of considerations that span several of the benefits and costs described in Chapter 4. These include:

- temporal impacts,
- locational impacts,
- interactive impacts,
- behind-the-meter versus front-of-the-meter impacts,
- air emission impacts,
- transfer payments and off-setting impacts,
- variable renewable generation impacts,
- wholesale market revenues,
- free-riders and spillover impacts, and
- discount rates.

Together, Chapters 4 and 5 feed into Part III (Chapters 6–10) which describe whether and how the range of benefits and costs apply to different DER technologies.

4. DER BENEFITS AND COSTS

This chapter presents a catalog of the range of benefits and costs that might be applicable to any DER type, including a brief description of each type of benefit or cost. Chapter 5 describes some additional issues that span multiple types of benefits and costs. Chapters 6 through 10 describe how these benefits and costs apply to different DER types.

4.1 Summary of Key Points

- DERs can have a wide range of impacts that can affect the electric utility system, the gas utility and other fuels systems, host customers, and society.
- While some impacts are definitively a benefit or cost for several DER types, other impacts can be either a benefit or a cost depending on factors, such as where the DER is located and when and how it operates.
- For DERs funded or otherwise supported by *electric* utilities, all BCAs should include the electric utility system impacts, and may include gas utility and other fuel system impacts, host customer impacts, and/or societal impacts (depending on the jurisdiction's applicable policy goals).
- For DERs funded or otherwise supported by *gas utilities or other fuel* vendors, all BCAs should include the gas utility or other fuel system impacts respectively, and may include the electric utility impacts, host customer impacts, and/or societal impacts (depending on the jurisdiction's applicable policy goals).
- Information presented in this chapter is based on the assumption that each DER type is implemented and operated in isolation from other DER types. When a BCA encompasses multiple DER types, the impacts might be different than those presented in this chapter. For example, some costs might become benefits and vice versa, and the net impact may be different than the sum of the individual DER impacts. Part IV of this manual provides further discussion on the implications of multiple DER cost-effectiveness analyses.

4.2 Utility System Impacts

4.2.1 Summary

Table 4-1 presents a list of the potential electric utility system impacts of DERs. This table includes impacts on the electric utility system that occur when either electric or gas utilities support DERs that have impacts on electricity consumption.

In some situations, these impacts might be in the form of costs, while in other situations they may be in the form of benefits (i.e., avoided costs). For example, EE programs will typically reduce electricity generation, creating a benefit, while electrification resources will typically increase electricity generation, creating a cost. As another example, DG that is fueled by renewable resources will typically reduce environmental compliance costs, but DG that is fueled by fossil fuels will typically increase environmental compliance costs.

Chapters 6–10 provide details on the impacts presented below for each DER type.

Туре	Utility System Impact	Description
	Energy Generation	The production or procurement of energy (kWh) from generation resources on behalf of customers
	Capacity	The generation capacity (kW) required to meet the forecasted system peak load
Comonation	Environmental Compliance	Actions to comply with environmental regulations
Generation	RPS/CES Compliance	Actions to comply with renewable portfolio standards or clean energy standards
	Market Price Effects	The decrease (or increase) in wholesale market prices as a result of reduced (or increased) customer consumption
	Ancillary Services	Services required to maintain electric grid stability and power quality
Transmission	Transmission Capacity	Maintaining the availability of the transmission system to transport electricity safely and reliably
	Transmission System Losses	Electricity or gas lost through the transmission system
	Distribution Capacity	Maintaining the availability of the distribution system to transport electricity or gas safely and reliably
Distribution	Distribution System Losses	Electricity lost through the distribution system
Distribution	Distribution O&M	Operating and maintaining the distribution system
	Distribution Voltage	Maintaining voltage levels within an acceptable range to ensure that both real and reactive power production are matched with demand
	Financial Incentives	Utility financial support provided to DER host customers or other market actors to encourage DER implementation
	Program Administration	Utility outreach to trade allies, technical training, marketing, and administration and management of DERs
	Utility Performance Incentives	Incentives offered to utilities to encourage successful, effective implementation of DER programs
General	Credit and Collection	Bad debt, disconnections, reconnections
	Risk	Uncertainty including operational, technology, cybersecurity, financial, legal, reputational, and regulatory risks
	Reliability	Maintaining generation, transmission, and distribution system to withstand instability, uncontrolled events, cascading failures, or unanticipated loss of system components
	Resilience	The ability to anticipate, prepare for, and adapt to changing conditions and withstand, respond to, and recover rapidly from disruptions

Table 4-1. Potential DER Benefits and Costs: Electric Utility System

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 77 of 302

4.2.2 Generation Impacts

Energy Generation

Definition: Expenses from the production or procurement of energy (i.e., kWh) from generation resources on behalf of customers. These expenses include the fuel cost and variable operations and maintenance (O&M) costs. Energy generation costs can vary by season and time of day.

Discussion: In general, DERs will create energy generation benefits when they reduce the amount of electricity utilities need to produce or procure in order to meet load and will create costs if they require higher levels of energy generation. An exception to this occurs during periods of negative pricing whereby consuming grid energy (e.g., storage or EV charging) results in a benefit and curtailing grid energy consumption results in a cost.

The direction and scale of this impact fluctuates over time. In terms of system conditions, the negative pricing example shows how the real-time balance in supply and demand will dictate the direction of the impact while the level of system load will drive the scale of the impact (i.e., generally higher during peak periods).

Generation Capacity

Definition: The amount of generation (i.e., kW) required to meet the forecasted peak load, ¹³ which typically includes an additional reserve margin. A utility will either need to build generation capacity or procure it (for instance through bilateral contracts or wholesale market purchases) to ensure it has sufficient generation capacity to meet its planning requirement.

Discussion: The effect of a DER on generation capacity is limited to instances when a DER's operation is coincident with the utility system peak load. If a DER's operation results in a net increase in load (such as with electrification) during the system peak, the utility will incur additional generation capacity costs. Alternatively, if a DER's operation results in a net decrease in load (e.g., EE savings; curtailment through demand response; PV generation; injections from storage) during the system peak, the utility will derive a benefit in the form of lower generation capacity needs.

Environmental Compliance

Definition: There are several environmental regulations that impact the utility system. These can include federal regulations like the Clean Air Act, regional mandates such as the Regional Greenhouse Gas Initiative (RGGI), or state and local mandates. These regulations typically require a reduction in emissions from energy generation or place limits on allowable levels of emissions from new or existing resources.

Discussion: While some environmental regulations place a fixed limit on the amount of emissions for new or existing resources, determining a DER's net effect on this impact typically requires an analysis of the marginal emissions rate of the resource being displaced by the DER (see Section 5.6). For example, if a DG resource's output occurs when the marginal supply resource for the utility system is an emitting resource, it will result in a benefit. Alternatively, if a DPV resource's generation occurs when a non-emitting resource (e.g., solar, wind, nuclear) is the marginal resource, then the DPV resource would provide neither a benefit nor a cost.

¹³ Jurisdictions may have alternative approaches for defining peak (e.g., single peak hour, a certain percentage of top load hours, etc.).

In some cases, it is important to consider and avoid double-counting impacts when, for example:

- Criteria pollutants are regulated through a cap-and-trade program—the environmental compliance impacts might already be included in the energy generation cost. It is important to ensure that any such embedded environmental compliance impacts are not double-counted by including them in this category.
- Considering the relationship between environmental compliance impacts and societal environmental impacts—environmental compliance impacts are the direct impacts in dollar terms that will be experienced by the utility and passed on to all customers through revenue requirements. Therefore, these impacts are utility system impacts that should be included in all BCA tests.¹⁴ Societal environmental impacts, on the other hand, are the impacts on the environment that occur after environmental compliance regulations have been met. Therefore, for those BCA tests that include societal environmental impacts, it is important to distinguish them from environmental compliance impacts in order to prevent double-counting.
- Distinguishing between environmental compliance impacts and societal environmental impacts for the purpose of rate, bill, and participation analyses—environmental compliance impacts are utility system impacts that will affect rates and therefore should be included in rate, bill, and participation analyses. Societal environmental impacts, on the other hand, are not utility system impacts, will not affect rates, and should not be included in rate, bill, and participation analyses.

In estimating environmental compliance impacts, a BCA should account for all regulations that have already been promulgated and are in effect. It should also account for all regulations that are expected to be in effect over the study period (Regulatory Assistance Project 2012). If there is uncertainty about future environmental regulations, then this can be addressed through sensitivity analyses or expected value calculations. (For example, if the likelihood of future promulgation is 70 percent, the environmental compliance cost can be multiplied by 70 percent.)

Renewable Portfolio Standard Compliance

Definition: A renewable portfolio standard (RPS) is a regulatory mandate to increase production of energy from renewable sources such as wind, solar, biomass, and other alternatives to fossil fuel-based electric generation. This impact may also include similar regulatory mechanisms like clean energy standards (CES) that more generally focus on zero-emissions resources (e.g., nuclear, hydro), or clean peak standards (CPS) that encourage clean energy resources that can mitigate the utility system peak.

Discussion: In jurisdictions that have adopted an RPS or similar mechanism expressed as a percentage of electric sales, DERs can reduce RPS compliance costs either by reducing the RPS target by virtue of lowering overall electricity demand or increasing the level of qualified renewable generation.¹⁵ Alternatively, if a DER has the effect of increasing electricity demand (e.g., electrification), it will require additional renewable purchases and therefore increase RPS compliance costs.

Wholesale Market Price Effects

Definition: In jurisdictions with competitive wholesale electricity markets, wholesale market prices are a function of the demand of buyers and the marginal costs of suppliers at any given instant. When DERs

¹⁴ Except for the PCT.

¹⁵ The utility must be able to claim the renewable energy attribute by retaining ownership of renewable energy credits (REC) from renewable DG to realize the benefit of reduced RPS compliance costs. In some cases, the host customer may choose to retain the RECs and thus renewable DG would not reduce the utility's RPS compliance costs.

reduce (or increase) the demand for electricity, they reduce (or increase) the wholesale market prices, which creates benefits (or costs) for all customers participating in the wholesale market at that time. A similar effect can be seen in wholesale natural gas markets from the reduction in natural gas demand from DERs (AESC Study Group 2018).

Wholesale market price effects are different from energy generation, generation capacity, and ancillary services impacts. The former is the impact on *prices* from a change in demand, which will affect all buyers from the wholesale markets. The latter is the impact on the *costs* associated with consuming a different amount of a commodity.

Discussion: DERs can impact wholesale market prices either in the form of demand (e.g., DPV treated as a utility load modifier) or supply (e.g., demand response participation directly in the wholesale market). This market price effect might be relatively small, but the impacts are felt by all customers buying from the market at the time of the effect, and therefore the total impact can be significant. This impact typically lasts for only a short period before the market adjusts to the new supply/demand balance (AESC Study Group 2018).

Market price effects from DERs located in a given utility service territory may extend beyond the borders of that service territory because of the regional nature of most wholesale markets, which tend to encompass multiple utility service territories. Thus, regulators need to decide whether to include the market price effects for (a) only the utility service territory in question, (b) the entire jurisdiction under the regulator's purview, or (c) for the entire wholesale market region. This decision should be based on the regulatory perspective and applicable policy goals for each jurisdiction.

Wholesale market price effects are sometimes referred to as a "transfer payment" between competitive suppliers and wholesale market buyers because the benefit to one is directly offset by the cost to another. However, this is a misleading conclusion. While it is true that the impact on wholesale market buyers' costs is offset by the impact on suppliers' profits, both impacts should be included in a BCA because both the buyer's costs and the supplier's profits are a part of the benefits and costs of the electricity resource. This issue is addressed further in Section 5.7 and Appendix F.

Ancillary Services

Definition: Ancillary services are those services required to maintain electric grid stability. They typically include frequency regulation, voltage regulation, spinning reserves, and operating reserves. These services are either traded in wholesale energy markets or self-supplied by utilities.

Description: As with other impacts, a DER's net effect on ancillary services depends on how the DER operates and what the real-time system conditions are at the time of its operation. Some resources may be actively dispatched to provide ancillary services (for instance, storage providing frequency regulation). Alternatively, even if a DER's operation is not directly in response to a signal to provide ancillary services, it may nevertheless create an impact. For example, during times when load is ramping up quickly and/or generation resources are ramping down quickly, DERs can provide additional operating reserves, fast frequency response, or ramping services.

4.2.3 Transmission Impacts

Transmission Capacity

Definition: Transmission capacity refers to the availability of the electric transmission system to transport electricity in a safe and reliable manner. In areas with insufficient transmission capacity

available to support transmission of lowest-cost electricity, there will be transmission congestion costs due to the need to utilize higher-cost generation to avoid the transmission constraint.

Discussion: As with generation capacity, a DER's impact on transmission capacity depends on how it operates during the times coincident with the transmission peaks. If a DER increases load at the time of the transmission system peak, it will result in added costs. Alternatively, if a DER reduces load at the time of the transmission system peak, it will result in reduced costs.

DERs may reduce transmission capacity costs in two ways:

- DERs may passively defer needed transmission capacity investments if their operation for other purposes (e.g., host customer bill management) results in lower load at the same time the transmission facilities are at their peak. In these instances, the DERs may be attributed with a system-wide average for the transmission capacity benefit provided.
- DERs may actively defer transmission capacity needs as part of a geographically targeted NWS. The value of active deferrals is typically based on the actual deferral value of the avoided transmission project (i.e., the costs avoided if the wires investment is deferred for a certain number of years). There is often a minimum cost threshold for transmission projects to be considered for an NWS; therefore, the value of active deferrals is typically higher than that of passive deferrals.

Some ISOs/RTOs allow for wholesale market participants to trade fixed transmission rights to help them manage transmission congestion costs. Some DERs might be able to create benefits by reducing transmission congestion and costs of fixed transmission rights. Costs of fixed transmission rights are typically included in wholesale energy market prices and therefore may not need to be included as a separate impact.

Transmission System Losses

Definition: A portion of all electricity produced at electric generation facilities is lost as it travels across transmission lines. Line losses grow exponentially with higher levels of load, and as such it is important that calculations account for marginal loss rates when determining this impact.

Discussion: To the extent DERs offset utility-scale energy generation (e.g., DG meeting host customer load) they will help avoid electricity transmission and as a result reduce the cost associated with transmission line losses. However, to the extent utility-scale energy generation is needed to meet any incremental load from DERs (e.g., electrification or storage and EV charging) the costs associated with transmission line losses will increase. The magnitude of the impact will depend on the amount of transmission-level load at the time of the DER's operation.

As with electric transmission line losses, gas transmission systems also experience "pipe losses," though they tend to be much smaller in magnitude (in percentage terms) compared to electric losses.

4.2.4 Distribution Impacts

Distribution Capacity

Definition: Distribution capacity refers to substation and distribution line infrastructure necessary to meet customer electric demand, and as such the net impact will depend on the cost associated with the specific type of distribution infrastructure being affected. If customer demand exceeds distribution capacity, it will require investments to increase distribution capacity to a level that preserves safety and

reliability. The net effect of DERs on distribution capacity depends on how they operate during the distribution system peaks.

Discussion: DERs can either actively or passively help defer or eliminate the cost of needed investments by reducing net load during peak hours. With respect to passive benefits, a DER may have the effect of reducing net load despite operating for some other purpose (e.g., host customer bill management). In terms of active deferrals, a utility may incentivize DERs through pricing, programs, or procurements to provide distribution capacity benefits.

Alternatively, DER's might increase distribution capacity costs if the local distribution system does not have sufficient hosting capacity (i.e., if a given feeder cannot accommodate more DERs without impacting system operation under existing control and infrastructure configurations). For example, if a DER consumes electricity from the grid during times of the distribution peak load or injects electricity onto the grid during times of minimum load (and therefore create voltage issues) it would have the effect of creating a cost to invest in the necessary distribution infrastructure to avoid these issues.

Distribution System Losses

Definition: A portion of all electricity produced at electric generation facilities is lost as it travels across the distribution system to the final point of consumption. This includes losses on the distribution lines and transformers. Line losses expand exponentially as the system experiences higher levels of load, so cost-effectiveness calculations should account for marginal loss rates.

Discussion: The net effect of a DER's operation on distribution line and transformer energy losses depends on the relative balance between load and net DER output. For example, if the net impact of DERs is a reduction of load at the feeder level, then there can be net reductions in line and transformer energy losses, and vice versa.

Like electric distribution line losses, gas distribution systems also experience "pipe losses," though they tend to be much smaller in magnitude (in percentage terms) compared to electric losses.

Distribution O&M

Definition: Utilities must incur O&M expenses to maintain the safe and reliable operation of distribution facilities. This includes maintenance of substations, wires and poles, and repairs and replacements. Some portion of distribution O&M expenses are variable, which means the expense incurred by a utility is a function of the volume of energy transfers through the system.

Discussion: When DERs reduce electricity consumption, they will typically reduce the energy transfers through distribution facilities. This creates a benefit by reducing variable distribution O&M expenses. Alternatively, when DERs increase electricity consumption, they might increase distribution O&M expenses. DERs that are intermittent in nature can lead to increased distribution costs due to the need to manage energy flows to maintain voltage and frequency within acceptable limits.

Distribution Voltage

Definition: Voltage regulation is necessary to ensure reliable and continuous electricity flow across the power grid. Voltage on the distribution system must be maintained within an acceptable range to ensure that both real and reactive power production are matched with demand (RMI 2015).

Discussion: DERs can either exacerbate or help address emerging voltage issues on the distribution system. Quantifying this impact requires analysis of when a DER will operate relative to real-time system

conditions. Additionally, care must be taken to ensure any impact associated with distribution voltage is not double-counted for another impact (e.g., ancillary services; distribution capacity).

4.2.5 General Impacts

Financial Incentives

Definition: Utility financial support provided to DER host customers or other market actors (e.g., retailers, contractors, distributors, manufacturers, integrators, and aggregators) to encourage DER implementation.

Discussion: Financial incentives may come in various forms, including incentives or rebates, buy-downs of interest rates for financing a portion of DER costs, payments to support trade ally reporting on sales of DERs, funding or co-funding of marketing of DER equipment by trade allies, and sales bonuses provided to retail or contractor sales staff for selling DER equipment.

Program Administration Costs

Definition: Program administration costs are those incurred by the utility related to the design, implementation, and evaluation of a DER program or initiative.

Discussion: These costs may come in a variety of forms, including costs to support utility outreach to trade allies, technical training, other forms of technical support, marketing, administration, and management of DER programs and/or portfolios of programs. Administration costs also often include evaluation, measurement, and verification studies to inform either DER program design or retrospective assessment of DER performance.

Utility Performance Incentives

Definition: In many jurisdictions, utilities are offered shareholder incentives for meeting specific performance metrics related to the success of DER programs. These performance incentives represent a cost associated with the delivery of the DER program.

Discussion: DER performance incentives can take many forms, including shared savings mechanisms, payments for meeting energy savings targets, payments for meeting capacity savings targets, or combinations of the above. Performance incentives can take the form of rewards, or penalties, or both.

Credit and Collection Costs

Definition: Costs associated with customers who are deficient on energy bill payments, including notices and support provided to customers in arrears, shutting off service and turning it back on, carrying costs associated with arrears, and writing off bad debt.

Discussion: To the extent a DER has the effect of lowering a host customer's energy bill, it may reduce the probability of the customer falling behind or defaulting on bill payment obligations and therefore result in a utility benefit. This may be a particularly important benefit of DER programs targeted to low-income customers.

Risk

Definition: The utility system can be exposed to several categories of risk, including operational (e.g., equipment breakdown or damage) technology, cybersecurity, financial, legal, regulatory, and reputational risks.

Discussion: DERs can reduce utility system risk in several ways. Key among them are that DERs:

- create a more diverse portfolio of resources that can meet customers' energy needs (all other things being equal, diversity typically reduces risk);
- reduce uncertainty in forecasts of future loads and related capital investment needs;
- reduce exposure to potential future fuel price volatility associated with other resource types (particularly natural gas, oil, and/or coal-fired generation) (Ceres 2012); and
- can transfer some financial risk away from the utility and non-participating customers to the host customers, in cases where host customers are putting up the capital to install the DER.

Also, as a resource that can be implemented in many relatively small increments, DERs provide more optionality than large central generation facilities. For example, once a utility decides to build a 200 MW natural gas-fired power plant to meet customer demand, it will be locked into that resource for the three or four years it takes to construct the power plant. If DERs are installed instead of that plant, then the utility will have those three of four years to determine whether it needs those 200 MW. If load growth turns out to be lower than expected over this time, then the utility might be able to reduce costs to customers by installing less than 200 MW of DER capacity.

DERs can also increase utility system risk in several ways. For example, some DERs may:

- have risks associated with host customers' willingness and ability to respond to DR signals;
- pose technical performance risks, particularly for new and emerging technologies;
- create a cybersecurity risk by virtue of creating a larger number of entry points to the electricity network;
- increase electricity or gas demand, which might increase risks to those utilities; or
- create risk in cases where DERs are controlled by customers or independent technology providers, because the utility does not directly control the operation of those resources.

It is important to avoid double-counting of risk, reliability, and resilience impacts.

Reliability

Definition: The U.S. Department of Energy defines reliability as the ability of the system or its components to withstand instability, uncontrolled events, cascading failures, or unanticipated loss of system components (DOE 2017c, page 4-1). Reliability has typically been tracked by utilities using metrics related to the frequency, duration, and extent of power outages experienced by customers (e.g., system average interruption duration index, system average interruption frequency index, and customer average interruption duration index).

Discussion: By lowering loads or increasing generation on the grid, DERs can reduce the probability and/or duration of customer service interruptions. The magnitude of the value of this benefit will vary, with less value to systems that have excess capacity or newly installed capacity, and greater value to systems that are short of capacity or have a large amount of aging infrastructure. It is important to avoid double-counting of risk, reliability, and resilience impacts.

Resilience

Definition: NARUC defines resilience as "Robustness and recovery characteristics of utility infrastructure and operations, which avoid or minimize interruptions of service during an extraordinary and hazardous event" (NARUC 2019).

Discussion: DERs can create resilience impacts in several ways. Some DERs, such as storage and DG, can provide power to the host customer, the local grid, or a micro-grid during an outage. Some DERs can increase the resilience of the overall energy system by providing black start (starting without power from the grid) or ramping capabilities. Some DERs can increase resilience by reducing the amount of load that needs to be served to recover from an outage. It is important to avoid double-counting of risk, reliability, and resilience impacts.

4.2.6 Cross-Cutting Utility System Impacts

Some impacts of DERs are not specific to any one utility system impact as they apply to more than one impact. As such they are not included in Table 4-1 because they are not mutually exclusive, i.e., they are cross-cutting impacts. These cross-cutting impacts, described below, will typically affect the magnitude of some of the other impacts presented in Table 4-1. They can be accounted for by either modifying the magnitude of those impacts or accounting for them separately. Either way, it is important to ensure that there is no double-counting of these cross-cutting impacts with other utility system impacts.

Enabling other DERs

Some DERs can make it easier or more cost-effective to adopt other types of DERs, thereby creating a benefit. (See Section 5.4.) Examples of how this can occur include:

- Distributed storage technologies can be combined with distributed PV or wind to improve the economics of each by storing excess generation during periods of low electricity costs and discharging from the storage device during periods of high electricity costs (ACEEE 2020). This arrangement may also have the effect of increasing hosting capacity.
- Multiple DERs can be combined in an NWS to help defer or avoid a distribution investment that could not be deferred or avoided by a subset of those DERs.
- Commercial and industrial lighting controls that are installed as part of an EE initiative can also be used as a DR resource.

Grid Flexibility

Operational flexibility of the electric grid is becoming increasingly valuable for grid operators to meet the needs of an increasingly dynamic grid. Some DERs can provide savings, generation, or ancillary services quickly, thereby enhancing flexibility of grid operations. Some DERs allow for shifting and shaping loads, thereby allowing for meeting peak demands with greater flexibility (typically referred to as demand flexibility). Some DERs can operate using two-way communications with grid operators, thereby enhancing their ability to respond quickly and flexibly to price and other signals. (DOE 2019a; SEE Action 2020b; RAP 2019b.)

Grid flexibility impacts will typically affect the magnitude of the DER impacts presented in Table 4-1. For example, grid flexibility might reduce generation capacity costs, ancillary services costs, transmission costs, and distribution costs. Grid flexibility might also increase risk, reliability, and resilience benefits.

The economic value of grid flexibility has not been studied as much as the values of some of the other utility system impacts of DERs. Further, the economic value of grid flexibility will depend upon many factors specific to the DER being studied, including the timing and use case of the DER operation; the location of the DER on the grid; the type of grid services provided by the DER; the expected service life of the DER; and the value of other resources that would provide comparable services (SEE Action 2020c).

Impacts Associated with Time-of-Use Rates

Definition: TOU rates provide customers with an incentive to shift load from high-cost peak periods to low-cost off-peak periods. TOU rates are designed to be revenue neutral for the utility (i.e., they provide the utility with the same amount of revenue as they would receive in the absence of the TOU rates).¹⁶ TOU rates sometimes require utility system costs in the form of administration, metering, and billing costs. TOU rates provide utility system benefits by reducing demand during high-cost hours.

Discussion: TOU rates can result in a shifting of cost recovery from customers with low-cost consumption patterns to those with higher-cost consumption patterns. Cost-shifting between customers is a different matter from cost-effectiveness and should not be accounted for in BCAs. It is nonetheless of great importance to regulators and other stakeholders. Those regulators concerned about cost-shifting from TOU rates should conduct rate, bill, and participation analyses to assess the implications of those rates. (See Chapter 2 and Appendix A.)

Market or Technology Transformation

Some utility DER initiatives can transform markets and technologies, resulting in lower costs and new opportunities to expand deployment. A common example is ratepayer-funded utility EE programs. These programs have been shown to lower the cost of new technologies and increase customer awareness and acceptance to the point where little to no utility incentives are needed for certain technologies. Some efficient end-uses, such as efficient lighting technologies, have been so widely adopted and commercialized as a result of EE programs that they no longer need to be supported by utility EE programs.

4.3 Gas Utility and Other Fuel System Impacts

4.3.1 Summary

There are two situations where DERs will have impacts on natural gas and other fuels. First, when natural gas utilities implement or otherwise support gas DERs there will be impacts on the natural gas utility system and perhaps other fuels. Second, when electric utilities implement or otherwise support DERs there are sometimes impacts on natural gas and other fuels. This section describes the impacts created by either of these two situations.

Table 4-2 presents a list of the potential impacts on gas utilities from gas utility DERs. For example, an EE program offered by a natural gas utility can result in gas utility fuel, variable O&M, and capacity benefits. Such programs can also result in costs associated with financial incentives, program administration, and utility performance incentives.

¹⁶ TOU rates are often designed to be revenue neutral to the utility based on historical consumption patterns. When customers change consumption patterns based on the TOU price signals, the revenue recovered from those customers will be different from historical revenue collections. In these cases, the TOU rate in itself is not revenue neutral.

Table 4-2 also includes the potential impacts on gas utilities and other fuels from electric utility DERs. Examples of gas and other fuel impacts from electric utility DERs include the reduced consumption of natural gas space heating that results from electric utility EE programs that provide air sealing to reduce air conditioner loads, the increased consumption of natural gas or other fuels from DR programs that rely upon back-up generators, the increased consumption in natural gas or other fuels resulting from an electric utility CHP program, and the reduced consumption in gasoline as a result of electric utility EV programs.

In some situations, these impacts might be in the form of costs, while in other situations they may be in the form of benefits. For example, a natural gas DR program might create natural gas capacity benefits, while an electric DR program that relies upon natural gas for back-up generation might create natural gas capacity costs.

Chapters 6–10 provide details on the impacts presented below for each DER type.

Туре	Gas Utility or Other Fuel Impact	Description	
	Fuel and Variable O&M	The fuel and O&M impacts associated with gas or other fuels	
	Capacity	The gas capacity required to meet forecasted peak load	
Energy	Environmental Compliance	Actions required to comply with environmental regulations	
	Market Price Effects	The decrease (or increase) in wholesale prices as a result of reduced (or increased) customer consumption	
	Financial Incentives	Utility financial support provided to DER host customers or other market actors to encourage DER implementation	
	Program Administration Costs	Utility outreach to trade allies, technical training, marketing, and administration and management of DERs	
	Utility Performance Incentives	Incentives offered to utilities to encourage successful, effective implementation of DER programs	
General	Credit and Collection Costs	Bad debt, disconnections, reconnections	
	Risk	Uncertainty including operational, technology, cybersecurity, financial, legal, reputational, and regulatory risks	
	Reliability	Maintaining the gas or other fuel system to withstand instability, uncontrolled events, cascading failures, or unanticipated loss of system components	
	Resilience	The ability to anticipate, prepare for, and adapt to changing conditions and withstand, respond to, and recover rapidly from disruptions	

Table 4-2. Potential Benefits and Costs of DERs: Gas Utility or Other Fuel Impacts

4.3.2 Energy Impacts

Fuel and Variable O&M

Definition: Utilities and other energy suppliers incur expenses from the procurement of fuel (e.g., natural gas, oil, propane) on behalf of customers, including any related variable O&M costs. This can include the costs related to the transport of natural gas from delivery points located on interstate and intrastate pipelines to customers.

Discussion: Some DERs will create fuel and variable O&M benefits when they reduce the amount of natural gas, fuel oil, propane, and wood that utilities and energy suppliers need to produce or procure to meet customer demand. Among the most common examples are EE measures such as insulation, air sealing, high efficiency boilers and furnaces, and electrification (e.g., heat pumps). Alternatively, some DERs like fossil fuel-powered combined heat and power systems or fuel cells may increase consumption of these other fuels. Furthermore, electric efficiency resources can reduce the "waste heat" from inefficient lighting, refrigeration, or air flow components, thereby increasing the need for other fuels used for building space heating.

Capacity

Definition: Natural gas utilities contract for firm pipeline transport and storage capacity to meet customer demand. This ensures that pipeline capacity is available when needed to distribute natural gas to customers with firm service contracts.

Discussion: The impact for DERs depends on how their operation affects other fuel consumption during peak hours. For example, gas EE or DR programs—potentially as part of a non-pipes solution (NPS)—can lower demand for gas supply during peak times and create benefits. Alternatively, if DERs increase demand on other fuel capacity during peak times (e.g., CHP; fuel cells), it will create an added cost for the utility.

Environmental Compliance

Definition: There are costs associated with environmental regulations for gas utility and other fuel systems (e.g., caps on methane emissions). This is comparable to the environmental compliance costs for electricity resources described earlier. (See Section 4.2.2.)

Discussion: As with other impacts, the net effect of a DER on environmental compliance depends on how it affects consumption of natural gas and other fuels: if it lowers consumption, it will produce a benefit, and vice versa. (See Section 4.2.2.)

Wholesale Market Price Effects

Definition: Wholesale markets for natural gas, oil, propane, and other fuels are a function of the magnitude of demand and the marginal costs of supply-side resources. When DERs reduce (or increase) the demand for other fuels, they reduce (or increase) the wholesale market prices, which creates benefits (or costs) for all customers participating in the wholesale market at that time (AESC Study Group 2018). This is similar to the wholesale market price effects for wholesale electricity markets. (See Section 4.2.2.)

Discussion: (See Section 4.2.2.)

4.3.3 General Impacts

Financial Incentives

Definition: Utility financial support provided to DER host customers or other market actors (e.g., retailers, contractors, distributors, manufacturers, integrators, and aggregators) to encourage DER implementation.

Discussion: Financial incentives may come in various forms, including rebates for DER technologies, buydowns of interest rates for financing a portion of DER costs, payments to support trade ally reporting on sales of DERs, funding or co-funding of marketing of DER equipment by trade allies, and sales bonuses provided to retail or contractor sales staff for selling DER equipment.

Program Administration Costs

Definition: Program administration costs are those incurred by the utility related to the design, implementation, and evaluation of a DER program or initiative.

Discussion: These costs may come in a variety of forms, including costs to support utility outreach to trade allies, technical training, other forms of technical support, marketing, administration, and management of DER programs and/or portfolios of programs. Administration costs also often include evaluation, measurement, and verification studies to inform either DER program design or retrospective assessment of DER performance.

Utility Performance Incentives

Definition: In many jurisdictions, utilities are offered shareholder incentives for meeting specific performance metrics related to the success of DER programs. These performance incentives represent a cost associated with the delivery of the DER program.

Discussion: DER performance incentives can take many forms, including shared savings mechanisms, payments for meeting energy savings targets, payments for meeting capacity savings targets, or combinations of the above. Some jurisdictions might include penalties for not meeting performance targets.

Credit and Collection Costs

Definition: Costs associated with customers who are deficient on energy bill payments, including notices and support provided to customers in arrears, shutting off service and turning it back on, carrying costs associated with arrears, and writing off bad debt.

Discussion: To the extent a DER has the effect of lowering a host customer's energy bill, it can reduce the probability of the customer falling behind or defaulting on bill payment obligations and therefore result in a utility benefit. This might be a particularly important benefit of DER programs targeted to low-income customers.

Risk

Definition: The utility system can be exposed to several categories of risk, including operational (e.g., equipment breakdown or damage) technology, cybersecurity, financial, legal, regulatory, and reputational risks. This impact is comparable to the risk impacts for electric utility systems described earlier. (See Section 4.2.5.)

Discussion: Some DERs can reduce risks associated with natural gas and other fuels, while others might increase risks. (See Section 4.2.5.)

Reliability

Definition: The U.S. Department of Energy defines reliability as the ability of the system or its components to withstand instability, uncontrolled events, cascading failures, or unanticipated loss of system components (DOE 2017c, page 4-1).

Discussion: DERs that reduce natural gas consumption during peak periods (e.g., EE, DR, electrification) can improve reliability through eliminating or reducing demand spikes and pipeline constraints. This can help to reduce the probability and/or likely duration of natural gas and other fuel delivery interruptions. There could be some overlap between this benefit and the benefits of reduced risk and avoided capacity costs. Therefore, any assessment of the value of increased reliability would need to ensure that there is no double-counting of overlap with such other benefits.

Resilience

Definition: The National Renewable Energy Laboratory defines resilience as "The ability to anticipate, prepare for, and adapt to changing conditions and withstand, respond to, and recover rapidly from disruptions through adaptable and holistic planning and technical solutions" (NREL 2019, page 2).

Discussion: Natural gas infrastructure is susceptible to the impacts of climate change, including sea level rise and flooding that can cause loss of service and in extreme cases damage to pipelines. However, there is limited research available on the impacts of DERs to natural gas system resilience.

4.4 Host Customer Impacts

4.4.1 Summary

Table 4-3 presents a list of the potential host customer impacts of DERs. The term "host customer" is used to refer to a customer that installs a DER in their home or business. The host customer might be a participant in a utility-sponsored DER program, a customer who installs DERs with the assistance of a third party, or a customer who installs DERs in response to price signals.

In some situations, these impacts can be in the form of benefits, while in other situations they can be in the form of costs. In most cases, the portion of DER costs, any transaction costs, and any interconnection fees will represent costs to the host customers. For some of the impacts listed, such as risk or host customer NEIs, the impacts can be in the form of either benefits or costs. Chapters 6–10 provide details on the impacts presented below for each DER type.

Туре	Host Customer Impact	Description
	Host portion of DER costs	Costs incurred to install and operate DERs
	Host transaction costs	Other costs incurred to install and operate DERs
	Interconnection fees	Costs paid by host customer to interconnect DERs to the electricity grid
	Risk	Uncertainty including price volatility, power quality, outages, and operational risk related to failure of installed DER equipment and user error; this type of risk may depend on the type of DER
Host	Reliability	The ability to prevent or reduce the duration of host customer outages
customer	Resilience	The ability to anticipate, prepare for, and adapt to changing conditions and withstand, respond to, and recover rapidly from disruptions
	Tax incentives	Federal, state, and local tax incentives provided to host customers to defray the costs of some DERs
	Host Customer NEIs	Benefits and costs of DERs that are separate from energy-related impacts
	Low-income NEIs	Non-energy benefits and costs that affect low-income DER host customers

 Table 4-3. Potential Benefits and Costs of DERs: Host Customer

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 90 of 302

4.4.2 Host Customer Energy Impacts

Host Customer Portion of DER Costs

Definition: Host customer costs include those costs incurred to install and operate a DER.

Discussion: Utility financial incentives (e.g., rebates) are often provided to defray some of the incremental costs of DERs. In those cases, the host customer DER cost includes only the portion of the DER cost not covered by the financial incentive.

Host Customer Transaction Costs

Definition: Transaction costs associated with the acquisition and installation of DERs.

Discussion: These costs can include time spent collecting information, obtaining quotes from multiple vendors, filing paperwork, and applications for rebates and other financing mechanisms. This impact will always manifest itself as a cost for the host customer.

Interconnection Fees

Definition: Costs associated with the utility and/or ISO/RTO interconnection process paid for by the host customer.

Discussion: Utilities typically require some DERs (e.g., DG and energy storage) to pay costs to interconnect to the utility grid. This usually requires host customers (or their vendors) to provide an application with relevant data on equipment, sizing, energy production, and location to ensure they meet necessary standards. Utilities charge an interconnection fee to cover the cost of the application and inspection process. In some cases, more detailed engineering analyses are required with the host customer being responsible for the additional costs.

Similarly, if a DER owner seeks to participate in the wholesale market and is subject to an ISO/RTO interconnection process, they may incur interconnection costs from the ISO/RTO. In some cases, host customers might incur both the ISO/RTO and utility interconnection costs. It is important to ensure that this cost is not double-counted with the host customer portion of DER costs (many installers will submit and pay for the interconnection fee on behalf of customers as part of the total cost of the system).

Risk

Definition: Host customers may face changes in risk levels related to price volatility, power quality, outages, and operational risk associated with the failure of installed DER equipment and user error. This type of risk may depend on the type of DER.

Discussion: DERs may increase or reduce risks to host customers. Reduced risks result from lowering a customer's exposure to high prices by reducing consumption during peak periods and reducing a customer's exposure to fossil fuel price volatility. Increased risks might result from technology performance, particularly in situations where DERs participating in wholesale electricity markets are exposed to non-performance penalties. It is important to avoid double-counting of utility and host customer risk impacts.

Reliability

Definition: Reliability refers to the ability of host customers to access energy supply when needed.

Discussion: DERs that can provide customers with access to power during a system outage (e.g., DG plus storage) can increase reliability. There could be overlap between this benefit and the benefit of resilience, and therefore any assessment of the value of increased reliability would need to ensure that there is no double-counting with other impacts. It is important to avoid double-counting of utility and host customer reliability impacts, as well as for risk, reliability, and resilience impacts.

Resilience

Definition: Resilience refers to the ability of host customers to avoid, mitigate, or quickly respond to power outages.

Discussion: DER systems with ride-through or islanding capacity¹⁷ (e.g., storage) can be used to provide energy in the event of a power outage and create a resiliency benefit in the form of continued supply of electricity for medical and health uses, continued business operations, safety and security, and other functions. In addition, some DER measures can offer short-term resilience benefits (e.g., in case of efficiency, high levels of insulation allow customers to remain comfortable without heating fuel for longer periods). It is important to avoid double-counting of utility, host customer, and societal resilience impacts, as well as for risk, reliability, and resilience impacts.

Tax Incentives

Definition: Federal, state, and local tax incentives are sometimes available to host customers to defray the costs of some DERs. One example is the federal solar investment tax credit to encourage investments in distributed solar energy.

Discussion: Whether to include host customer tax incentives in a BCA depends upon the particular BCA test used. If the host customer impacts are outside of the test, then host customer tax incentives are not relevant. If both host customer impacts and taxpayer impacts are within the scope of the test, then host customer tax incentives should not be included because they offset each other. If host customer impacts are within the scope of the test but taxpayer impacts are not, then host customer tax incentives should be included in the test because there is no offsetting impact. (See Section 5.7 and Appendix F.)

Table 4-4 indicates how these factors play out in different tests. The JST can either align with one of any of the traditional tests below, and have consistent treatment of tax incentives, or it can be a unique test that requires different treatment. In the example presented below the JST does not include taxpayers within the scope of the test.

¹⁷ The term "ride-through" capacity refers to the ability of DERs to be able to operate during a power outage. The term "islanding" capacity refers to the ability to isolate a set of customers to operate independently from the rest of the grid, e.g., with a micro-grid.

Test Used	Host Customer Impacts Within Scope?	Taxpayer Impacts Within Scope?	Include Tax Incentives?
Utility Cost Test	_	—	_
Total Resource Cost Test	✓	—	✓
Jurisdiction Specific Test: Example	✓	—	✓
Societal Cost Test	✓	✓	_

Table 4-4. Treatment of Host Customer Tax Incentives in Different Tests

4.4.3 Host Customer Non-Energy Impacts

Definition: DERs can create a variety of non-energy impacts (NEIs) for host customers that are separate from the energy saved or produced by DERs. Table 4-5 presents a summary of host customer NEIs that might potentially be created by DERs. These impacts can sometimes be in the form of benefits and sometimes costs. The presence, direction, and magnitude of these impacts will depend upon many factors, including the type of DER (e.g., EE, DR, DG, storage, electrification, EVs), the specific DER technology (e.g., EE lighting versus EE building conditioning), the type of host customer (e.g., low-income, residential, commercial, industrial), and more.

Host Customer NEI	Summary Description
Transaction costs	Costs incurred to adopt DERs, beyond those related to the technology or service itself (e.g., application fees, time spent researching, paperwork)
Asset value	Changes in the value of a home or business as a result of the DER (e.g., increased building value, improved equipment value, extended equipment life)
Productivity	Changes in a customer's productivity (e.g., changes in labor costs, operational flexibility, O&M costs, reduced waste streams, reduced spoilage)
Economic well-being	Economic impacts beyond bill savings (e.g., reduced complaints about bills, reduced terminations and reconnections, reduced foreclosures—especially for low-income customers)
Comfort	Changes in comfort level (e.g., thermal, noise, and lighting impacts)
Health & safety	Changes in customer health or safety (e.g., fewer sick days from work or school, reduced medical costs, improved indoor air quality, reduced deaths)
Empowerment & control	The satisfaction of being able to control one's energy consumption and energy bill
Satisfaction & pride	The satisfaction of helping to reduce environmental impacts (e.g., one of the reasons why residential customers install rooftop PV)

Table 4-5. Potential Host Customer Non-Energy Impacts

Discussion: Some NEIs can be difficult to quantify and monetize. By their nature, DER non-energy costs and costs are less well known and harder to research than utility system impacts and host customer direct benefits or costs. In addition, the host customer NEIs might vary considerably depending upon the type of customer, the customer's facility, the type of DER, and the operation of the DER. Further, there has been little empirical research on host customer NEIs for some DERs, such as DPV, storage, and EVs.

Nonetheless, the fact that host customer NEIs can be difficult to quantify and monetize does not mean that they should be ignored in cost-effectiveness analyses. First, the *Account for Relevant, Material Impacts* principle states that impacts that are relevant (i.e., within the scope of a BCA test) and material

(i.e., are expected to have a material impact on the results) should be accounted for in the BCA, regardless of whether they are hard to quantify. (See Chapter 2, Section 3.4, and Appendix C.)

Second, the *Ensure Symmetry* principle requires that cost-effectiveness practices be symmetrical, where both benefits and costs are included for each relevant type of impact, even if difficult to quantify. This means that if a BCA test includes host customer costs, then it must also include host customer benefits, including non-energy benefits. Similarly, if a jurisdiction decides that host customer NEIs should not be included in a BCA test, then other host customer impacts should not be included either.

Low-Income Host Customer Non-Energy Impacts

Definition: Low-income customers experience similar NEIs as non-low-income customers, but in some cases, they are distinctly different.

Discussion: Low-income NEIs can come in two forms. First, low-income NEIs include the same *types* of NEIs as realized by non-low-income residential participants (e.g., economic well-being, comfort, health, safety). The *magnitude* of some of these benefits are often greater for low-income customers than for non-low-income customers. This is because the condition of the low-income housing stock is often worse and/or because the economic stress under which low-income customers live can result in greater sacrifice of amenity (e.g., comfort) absent efficiency investments.

Second, some host customer NEIs are unique, or largely unique, to this subset of customers. Examples include reduced home foreclosures and reduced need to move residence as a result of unpaid bills.

Host Customer Bill Savings

Definition: DERs typically result in bill savings for the host customer. In some cases, e.g., electrification, the DER might increase the host customer electricity bill but decrease the costs of other fuels such as natural gas or gasoline, resulting in a net decrease in energy costs for the host customer.

Discussion: Host customer bill savings should not be included in cost-effectiveness tests used to determine which DERs warrant utility support on behalf of all utility customers. Host customer bill savings overlap significantly with utility system benefits, which are already accounted for in the utility system impacts in BCA tests. As such, including them in a BCA would double-count some of those impacts.¹⁸ Further, host customer bill savings will result in rate impacts, which should be analyzed separately from cost-effectiveness analyses. (See Chapter 2 and Appendix A).

Host customer bill savings should, however, be included in the PCT because that test is designed to represent the actual impacts on host customers, including bill savings. In the PCT, the bill savings are the primary benefits of DERs, but the utility system benefits are not included at all. Similarly, host customer bill savings should be accounted for in the RIM Test because that test is designed to identify how DERs will impact rates.

¹⁸ Host customer bill savings are driven by the rates that the customer pays for generation, transmission, and distribution, which are typically based on historical, embedded costs. Utility system benefits are based on future, marginal generation, transmission, and distribution costs.
4.5 Societal Impacts

4.5.1 Summary

Table 4-6 presents a list of the potential societal impacts of different DER types.

In some situations, these impacts might be in the form of costs, while in other situations they may be in the form of benefits (i.e., avoided costs). For example, EE programs will typically reduce electricity generation, creating a benefit, while electrification resources will typically increase electricity generation, creating a cost. As another example, DG that is fueled by renewable resources will typically reduce environmental compliance costs, but DG that is fueled by fossil-fuels will typically increase environmental compliance costs.

Chapters 6–10 provide details on the impacts presented below for each DER type.

Table 4-6 presents a list of the potential societal impacts for different DER types. These impacts are relevant for parties other than the electric utility, gas utility, other fuel provider, or host customer.

Туре	Societal Impact	Description
Societal	Resilience	Resilience impacts beyond those experienced by utilities or host customers
	GHG Emissions	GHG emissions created by fossil-fueled energy resources
	Other Environmental	Other air emissions, solid waste, land, water, and other environmental impacts
	Economic and Jobs	Incremental economic development and job impacts
	Public Health	Health impacts, medical costs, and productivity affected by health
	Low Income: Society	Poverty alleviation, environmental justice, and reduced home foreclosures
	Energy Security	Energy imports and energy independence

Table 4-6. Potential Benefits and Costs of DERs: Societal

4.5.2 Types of Societal Impacts

Resilience

Definition: Same as for electric and gas utility or other fuel systems.

Discussion: Society can realize DER resilience benefits that are above and beyond the benefits that accrue to utilities, fuel suppliers, and host customers. Some DERs (e.g., DG combined with storage) allow for critical facilities such as hospitals, fire stations, police stations, water treatment facilities, and more to continue providing services during a planned or unplanned power outage. The services that these critical facilities provide to society go beyond the benefits enjoyed by the host customers themselves. It is important to avoid double-counting between societal, utility system, and host customer resilience benefits.

Greenhouse Gas Emissions

Definition: GHG emissions are created from a variety of sources, including production, transmission, and distribution of both electricity and natural gas; industrial processes; heating of commercial and residential buildings; and transportation. Societal GHG emissions represent the emissions that occur

after compliance with environmental regulations and requirements. These societal emissions are referred to as "externalities" because the impacts are external to the monetary prices of the goods that create them.

Discussion: The magnitude of DER GHG impacts will depend upon the marginal GHG emissions rate of the resources that are affected by the DER, which in turn will depend upon when the DER operates. The marginal GHG emission rates from a utility system can vary considerably across hours, days, and months, and can also vary considerably across utility systems and utility control areas. (See Sections 5.2 and 5.6.)

For DERs that impact multiple fuels (e.g., electrification, EVs) it is important to identify the environmental impact of all fuels affected by the resource. Electrification and EVs will typically increase the GHG emissions from the electricity industry but will reduce the GHG emissions from other fuel sources. Both effects are necessary to derive the net impact on GHG emissions.

It is important to distinguish between environmental compliance impacts and societal GHG impacts for the following reasons:

- Environmental compliance impacts are utility system impacts that should be included in all BCA tests.¹⁹ Societal GHG impacts, on the other hand, are the impacts on the environment that occur after environmental compliance regulations have been met and should be included only in those BCA tests that include societal GHG impacts.
- In the context of rate, bill, and participation analyses, environmental compliance impacts are utility system impacts that will affect rates and therefore should be included in rate, bill, and participation analyses. Societal GHG impacts, on the other hand, are not utility system impacts, will not affect rates, and therefore are not appropriate to include in rate, bill, and participation analyses.

Other Environmental

Definition: Energy resources can have a variety of environmental impacts beyond GHG emissions. These include other air emissions, liquid and solid waste emissions, land use, water use, and more. Societal environmental impacts represent the impacts that occur *after* compliance with environmental regulations and requirements. These societal impacts are referred to as "externalities" because the impacts are external to the monetary prices of the goods that create them.

Discussion: As with GHG emissions, other environmental impacts from DERs typically depend upon the energy resources that are affected by the DER, which in turn will depend upon when the DER operates. Therefore, other environmental impact estimates should be based on as much temporal granularity as possible. (See Sections 5.2 and 5.6.)

Similarly, it is important to identify the other environmental impacts of all fuels that are affected by the DER, such as the impacts of gasoline consumption that are affected by EVs.

As with GHG emissions, it is important to distinguish between environmental compliance impacts and societal environmental impacts, both for determining which impacts to include in BCA test and for conducting rate, bill, and participation impacts.

¹⁹ Except for the PCT.

Economic Development and Jobs

Definition: The value of any incremental economic development and jobs provided by DERs.

Discussion: Economic development impacts from energy resource investments include three categories of impacts:

- *Direct impacts:* Jobs and economic activity associated with constructing, installing, and operating the energy resource.
- Indirect impacts: Jobs and economic activity associated with additional work and revenue that such programs funnel to the supply chains associated with the direct impacts. These supply chains include contractors, builders/developers, equipment vendors, product retailers, distributors, manufacturers, and other elements.
- *Induced impacts:* Jobs and economic activity created by the re-spending of the newly hired workers who gained employment in the direct or indirect impacts categories.

DERs contribute to these three categories of impacts, primarily through two different phases. The first phase is during the installation of the DER, which might last a year or two. The second phase is during the operation of the DER, which lasts many years over the full operating life of the DER. In the second phase, most of the job and economic activity impacts are created when the host customers spend the money that they have gained from reduced energy bills (ACEEE 2019a).

All investments in energy resources will have economic development impacts. While DERs will typically increase economic development, they will displace other energy resources that also would have increased economic development. Estimates of DER economic development and job impacts should account for both the impacts of DERs and the energy resources that they displace, in order to adhere to the *Ensure Symmetry* principle.

Economic development can be expressed in several ways, including employment (in job-years), gross domestic product (in \$), personal income (in \$), or state tax revenues (in \$). Since these different expressions of economic development are interrelated and overlapping, a jurisdiction cannot simply take the sum of them individually to derive the net impact. Consequently, the choice of which way to express economic development can have a significant effect on the monetary value of this impact.

Monetary estimates of economic development impacts should not be added to the monetary costeffectiveness analysis results, because they represent a different type of economic impact (Synapse 2019, Appendix B). The economic development benefits represent economic activity in the state, which is different from the customer and societal impacts included in an energy efficiency program BCA.

The number of job-years is a potentially useful metric to present alongside BCA results, because job growth is easily understood and relatively easy to isolate from the other indicators.

Public Health

Definition: Some energy resources create health impacts for populations impacted by fuel extraction, combustion, and transportation. These health impacts have implications for (a) the health and well-being of the affected populations, (b) the societal investment required in medical facility infrastructure, and (c) the economic productivity of the affected populations (RAP 2013a).

Discussion: There is potential for considerable overlap between other environmental impacts and public health impacts. It is important to define and address both types of impacts carefully to avoid double-counting or under-counting of impacts.

Low-Income: Society

Definition: Low-income community or societal impacts that go beyond those realized by host customers. Some examples include poverty alleviation, local environmental justice benefits, improving low-income community strength and resiliency, and reduced home foreclosures.

Discussion: There is potential for considerable overlap between low-income host customer impacts and low-income societal impacts. It is important to address both types of impacts carefully to avoid double-counting or under-counting of impacts. For example, any societal impacts from reduced foreclosures must be incremental to the host customer impacts related to foreclosures.

Energy Security

Definition: DER investments that reduce imports of various forms of energy help advance the goals of energy independence and security.

Discussion: There is potential for overlap between energy security and utility system reliability and risk. It is important to address both types of impacts carefully to avoid double-counting or under-counting of impacts.

5. CROSS-CUTTING BENEFIT AND COST CONSIDERATIONS

This chapter discusses a variety of benefit and cost considerations that span several of the impacts listed in Chapter 4 and across different types of DERs discussed in Chapters 6 through 10.

5.1 Summary of Key Points

- Temporal and Locational Impacts of DERs: Several of the benefits and costs of some DERs can vary significantly depending on when the DER operates and where it is located. DER benefits and costs should be estimated using temporal and locational detail sufficient to adequately represent the DER operating patterns and consequent benefits and costs.
- Interactive effects between DER types: Some DER types can have interactive effects on other DERs in terms of affecting avoided costs, affecting the magnitude of kWh and kW impacts, and enabling the adoption of other DER types. BCAs should account for these interactive effects for those instances where the effects are likely to have a material effect.
- Behind versus in front of the meter DERs: Some DERs are located in front of the meter and are primarily operated to reduce utility system costs, while others are located behind the meter and are primarily operated to reduce customer costs. While the impact categories for these types of DERs are similar, there may be important differences in operation between utility needs and host customer needs that should be identified and accounted for.
- Air emission impacts: GHG and other air emission impacts will depend upon when the DER operates and which energy resources are utilized differently at that time. Estimates of GHG and other air emission impacts should account for the temporal and marginal DER impacts in as much detail as necessary to reflect these effects.
- Transfer payments and offsetting impacts: There are some situations in DER BCAs where a DER benefit experienced by one party is exactly offset by a corresponding DER cost experienced by another party, and therefore should be excluded from the BCA. For some BCA tests, financial incentives and tax incentives might be offsetting effects and therefore should be excluded from the BCA.
- *Renewable generation impacts*: DERs can affect renewable generation by providing grid flexibility and ancillary services to help with increasing amounts of intermittent generation from these resources. DERs can also reduce (or increase) the need to curtail renewable resources during times when renewable generation exceeds customer load. These impacts on renewable generation should be accounted for when they are expected to have a material effect on the BCA results.
- Wholesale market revenues: Some DERs are eligible to participate in wholesale electricity markets, which provide revenues to host customers or DER aggregators. These revenues should not be included in most cost-effectiveness tests because this would result in double-counting of the energy, generation capacity, and ancillary services benefits to the utility system. The one exception is the PCT because this test does not account for utility system benefits and the

wholesale market revenues are experienced by the host customers in addition to the bill savings.

• *Discount rates*: The choice of discount rate to use for a BCA might have a large effect on the result of the analysis. This choice should be guided by the jurisdiction's applicable policy goals and the regulatory perspective.

5.2 Temporal Impacts

DER impacts can vary depending on the specific timing of the DER's operation, because of temporal variation in system conditions (LBNL 2019). For example, generation costs and air emissions vary considerably over time depending upon which power plant is on the margin. As another example, DERs can provide greater benefits to the system if they relieve transmission and/or distribution capacity constraints during times of peak loading, but alternatively could result in added transmission and/or distribution capacity costs if their operation increases load during peak periods. Table 5-1 indicates which DER impacts are most likely to have temporal variation.

Utility System Impacts	Gas and Other Fuel Impacts	Host Customer Impacts	Societal Impacts
Energy generation	Fuel and variable O&M	Risk	GHG emissions
Generation capacity	Capacity	Reliability	Other environmental
Environmental compliance	Environmental compliance	Resilience	Public health
Market price effects	Risk		Resilience
Ancillary services	Reliability		
Transmission capacity	Resilience		
Distribution capacity			
Distribution voltage			
Risk			
Reliability			
Resilience			

Table 5 1. Delt impacts rypically Anected by Temporal Valiation	Table 5-1. DER	Impacts T	ypically	Affected b	y Temporal	Variation
---	----------------	-----------	----------	------------	------------	-----------

Figure 5-1 provides a hypothetical example of the effect that temporal variations can have on an EE resource's benefits. This example presents six categories of benefits for an EE resource, and how they might vary depending upon whether the benefits are based on annual averages, averages for four periods (winter on- and off-peak and summer on- and off-peak), or hourly data. This example assumes the EE resource provides a larger portion of savings during peak periods than off-peak periods.

In this example:

- The energy generation benefits are higher in the four periods case and higher still in the hourly case because these cases allow for a better breakout of peak versus off-peak hours.
- The T&D capacity benefits are also higher in the four periods case and higher still in the hourly case because these cases allow for a better breakout of peak versus off-peak hours.
- The GHG benefits are slightly lower in the four periods case and lower still in the hourly case. This is because for this hypothetical electricity system the higher emission resources are on the margin during peak periods and this effect is captured better with more granular data.
- The generation capacity, other utility, and host customer benefits do not change as a result of more granular temporal information.

In practice, the magnitude and direction of the impacts presented here could vary from this illustrative example, depending upon the DER, its operating pattern, its location, and other factors. For example, for an EE resource that provides a larger portion of savings during off-peak periods relative to peak periods, the energy and T&D benefits might be lower as a result of more granular data.



Figure 5-1. Example of Temporal Impacts on Energy Efficiency Benefits

DER benefits and costs should be estimated using enough temporal detail to adequately represent the DER operating patterns and consequent benefits and costs. For some DERs, such as EE and DR, it may be sufficient to use four periods per year to adequately capture temporal impacts; whereas for other DERs, such as storage or vehicle-to-grid (V2G) it might be necessary to use hourly or sub-hourly periods.

Although the goal of measuring impacts with greater temporal granularity is to derive a more accurate estimate of actual impacts, it may also entail greater levels of uncertainty. For example, while there are several industry references available that provide prototypical DER performance on an hourly basis, there is greater uncertainty in deriving these prototypical performance curves on a sub-hourly basis.

5.3 Locational Impacts

DER impacts can vary depending on where they are located on the distribution system. Like the temporal impacts discussed in Section 5.2, locational impacts vary depending on dynamic system conditions (e.g., the cost to generate, transmit, and distribute electricity). For example, transmission and/or distribution capacity constraints may make it more valuable for DERs to be sited in locations where their operation can help alleviate the constraint. Alternatively, a DER located in an area of the distribution system with more limited hosting capacity may require distribution system upgrades to preserve distribution safety and reliability. Table 5-2 indicates which DER impacts are most likely to have locational variation.

Utility System Impacts	Gas and Other Fuel Impacts	Host Customer Impacts	Societal Impacts
Energy generation	Fuel and variable O&M	Risk	Resilience
Generation capacity	Capacity	Reliability	Other environmental
Environmental compliance	Environmental compliance	Resilience	Public health
Transmission capacity	Risk		
Distribution capacity	Reliability		
Distribution O&M	Resilience		
Distribution voltage			
Interconnection costs			
Risk			
Reliability			
Resilience			

Figure 5-2 provides a hypothetical example of the effect that locational variations can have on a DER's benefits. This example presents six categories of benefits for a DR resource, and how they might vary depending upon whether the benefits are based on the system average, a location where distribution capacity is not constrained, and a location where distribution capacity is constrained.

In this example, the distribution benefit is the only impact that changes across the three cases:

- For the system average, there is a moderate distribution benefit based on the average benefits across the utility system.
- For the unconstrained location, the distribution benefit is zero, leaving only the transmission benefit.
- For the constrained location, the distribution benefit is higher than the system average and the unconstrained location.

In practice, the magnitude and direction of the impacts presented here could vary from this illustrative example, depending upon the DER, its operating pattern, its location, and other factors.



Figure 5-2. Example of Locational Impacts on Demand Response Benefits

DER benefits and costs should be estimated using enough detail to adequately represent the DER locational impacts and consequent benefits and costs. For some DERs where distribution benefits are a relatively small portion of the total benefits, such as EE, it may be sufficient to use system average impacts. For other DERs where more targeted distribution benefits are a significant portion of the benefits (such as DR, storage, and NWSs) it is more important to capture the locational details.

Some impacts might vary by location for some utilities but not others. For example, if energy generation is settled on an ISO/RTO zonal basis and the utility's service territory only spans a single zone, then this impact would not vary depending on where the DER is interconnected. However, if the utility's service territory spans several zones, then the energy generation impacts might vary across those zones.

Although the goal of measuring impacts with greater locational granularity is to derive a more accurate estimate of actual impacts, it may also entail greater levels of uncertainty. For example, while a utility may be able to readily determine impacts at the system level, there is more complexity involved with determining these impacts on a more granular locational level (e.g., having to forecast load growth, DER adoption, and DER operating profiles on a feeder level). BCAs must also consider that locational needs — and hence likely impacts — may change over time due to dynamic system conditions.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 103 of 302

ESTIMATING LOCATION IMPACTS – EXAMPLES

California and New York initial efforts to estimate locational impacts can serve as a useful reference for other jurisdictions seeking to develop more robust frameworks for estimating locational DER impacts.

CALIFORNIA

Section 769 of the California Public Utilities Code requires the investor-owned utilities (IOUs) to prepare Distribution Resource Plans (DRPs) that identify optimal locations for the deployment of DERs. The IOUs apply a Locational Net Benefit Analysis (LNBA) methodology, which takes the statewide averaged avoided costs for T&D and unbundles these values into specific sub-categories, such as distribution voltage and power quality, distribution reliability and resiliency, transmission, flexible resource adequacy procurement, and renewables integration. The LNBA Working Group released a Long-Term Refinements Report to the Commission providing recommendations on further refinements and improvements to the LNBA methodology (SDGE, SCE, and PGE 2018). The report addressed many topics and identified six priority topics: (1) locational avoided energy value; (2) locational avoided capacity value; (3) locational avoided line losses value; (4) incorporation of reactive power priority (VAR profiles); (5) automatic input of DER profiles; and (6) locational avoided transmission value.

NEW YORK

As part of the Reforming the Energy Vision initiative, the New York State Public Service Commission established the Value of Distributed Energy Resources (VDER) proceeding and developed a new DER tariff called the Value Stack. The Value Stack was designed initially to replace retail rate net energy metering for some types of DERs, such as community DG and other BTM projects for which customers were already subject to demand metering but the tariff then became more broadly applicable to all DERs. The Value Stack was designed to capture values across five distinct areas: (1) wholesale energy; (2) wholesale capacity; (3) environmental value (i.e. carbon emissions); (4) distribution system-wide value; and (5) targeted locational distribution value (NY PSC 2019).

Like California's LNBA efforts, the last two Value Stack components seek to reflect a more accurate approximation of the distribution value DERs can provide. To derive each of these two value components, the utilities were required to "deaverage" their system-wide marginal cost of service studies into two separate values measuring the impact on avoided distribution costs. The distribution system-wide value, referred to as the Distribution Reduction Value, provides compensation for the average value in avoiding distribution costs at the system level. Alternatively, the more targeted locational value, referred to as the Locational System Relief Value, is only available for DERs located in utility-identified load pockets where these resources are more likely to avoid forecasted distribution upgrades. For both the Distribution Reduction Value and Locational System Relief Value, DERs will only be compensated for their performance during certain hours that are determined to represent the highest system-wide or locational need.

5.4 Interactive Impacts

While Chapter 4 provides an overview of each individual impact, this section focuses on how multiple DERs might influence their impacts collectively. These interactive effects may manifest themselves in three areas: (1) marginal system costs, (2) the magnitude of kW or kWh impacts of other DERs, and (3) enablement of other DER types.

BCA practitioners do not necessarily need to identify all of the impacts of each DER that is used in combination with other DERs. All that matters is the total impacts of all the DERs combined. Nonetheless, it is important to consider the concepts described below in order to understand how the DERs can affect each other in order to determine the total impacts to be input to the BCA.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 104 of 302

5.4.1 Impacts on Marginal System Costs

Interactive Effects of Marginal System Costs

Marginal system costs refer to the costs associated with serving the next kW of load. These costs vary over time as a result of changes to the system (e.g., a dynamic supply-demand balance; quality of system infrastructure; changing resource mix; etc.). Marginal system costs are important in DER BCAs because they determine many of the key DER benefits, such as energy and capacity benefits.

When multiple DERs and multiple DER types are installed on a system they might affect the marginal system costs, and thereby affect the avoided costs, of each other. While this effect on marginal system costs may be small and immaterial for each DER or type of DER, the net effect may become increasingly significant as DER deployment increases.

One prominent example of this effect is the duck curve resulting from high deployment of solar resources. As solar resources provide generation during the hours of the day with sunlight, the result will be a lower net load during those hours that might otherwise have been considered the peak hours (and as such, result in lower system marginal costs and lower avoided costs). With enough solar deployment, this effect could be large enough to shift the peak period from the afternoon to the early evening when the solar generation dissipates. As a result of this phenomenon, the impact other resources (e.g., efficient air conditioning) have on marginal system costs during the afternoon hours will be lower than it would have otherwise been at that time.

This type of interactive effect is not limited to just the duck curve. Another example is when a DR program's size and effectiveness is significant enough to materially change either the timing or magnitude of peak generation, transmission, or distribution demands. If this is the case, it will fundamentally affect the on-peak avoided costs of other DERs. Similarly, greater adoption of EE resources will put downward pressure on the marginal cost of energy, which will reduce avoided energy costs for other DER types.

In some instances, TOU rates can be used to help DERs respond to changes in marginal system costs. For example, TOU rate structures could be modified over time to ensure that the peak periods for pricing purposes remain consistent with the peak periods that shift as a result of changes to marginal system costs from DERs.

Accounting for Interactive Effects of Marginal System Costs in DER BCAs

DER cost-effectiveness analyses should account for these interactive effects on marginal system costs where they are likely to have a significant impact on the results of the analysis. These interactive effects will be increasingly significant in jurisdictions with growing DER deployment.

Effectively capturing these interactive effects requires incorporation of DER load impacts into system planning (i.e., IRP, IDP, or IGP) to consider both DERs and supply-side resources as potential solutions to meet system needs. (See Chapter 14.)

For those jurisdictions and utilities that do not use IDP practices, there are several ways to consider or account for impacts on the avoided costs of other DERs.

The first step is to decide whether this effect should be accounted for in the cost-effectiveness analysis. In many cases with relatively low deployment of DERs, the impacts of some DERs on the avoided costs of other DERs might be small. This is the assumption that has been used in many EE and DER costeffectiveness analyses in the past. For jurisdictions that decide to account for this effect, there are several approaches to account for how some DERs can affect the avoided cost of other DERs:

- One option is to *estimate which DER types are likely to be the most cost-effective*. For example, if EE and DR resources are likely to be the most cost-effective DER types, then these could be assumed to be implemented first. These resources could be assumed to be in place when the avoided costs for other DERs are estimated. As another example, all the low-cost EE, DR, DG, electrification, and storage technologies could be installed first, then the mid-cost DERs, and then the high-cost DERs.
- Another option is to *modify the avoided costs over time* based upon the assumed implementation of DERs. In other words, the estimate of avoided costs in the second year of the study period could assume a certain amount of DERs are installed in the first year of the study period, etc.
- A further option is to *iterate using different assumptions for avoided costs* under different scenarios. For example, the first set of scenarios could assess the cost-effectiveness of DERs without modifications to the avoided costs. The next set of scenarios could use modified avoided costs that assume that all the cost-effective DERs identified in the first set are in place. This process could continue several times until an equilibrium is reached or there is confidence that the impacts on avoided costs have been sufficiently addressed.

Each of these options has limitations that make them much less accurate than using IDP to dynamically estimate the impacts of DER on marginal system costs. Nonetheless, for those jurisdictions and utilities that do not use IDP, they might be better than ignoring the interactive effects that DERs have on marginal system costs.

5.4.2 Effects on the Magnitude of kWh or kW Impacts of Other DERs

There are many ways that one DER might affect the magnitude of kWh or kW impacts of other DERs. For example:

- When EE resources reduce a customer's peak demand, they will likely reduce that customer's ability to reduce peak demand through DR or storage resources.
- When DG and storage resources are installed by the same customer, the combined benefits of both resources operating together are likely to exceed the benefits of each DER operating without the other. The storage resource will allow the customer to sell DG output back to the grid at times when prices are highest.
- A building with DR resources (e.g., thermostats; hot water heaters) might affect the dispatch parameters of a storage resource within the same building. (See Chapter 11.)
- A utility that sources an NWS may seek to use a portfolio of DERs whose combined interactive effects are best suited to meet the system need. (See Chapter 12.)

Figure 5-3 presents an illustrative example of how interactive effects might influence the benefits of DPV and storage resources. The bar on the left indicates the potential benefits of DPV and storage when each is the only resource installed on a site. The bar on the right indicates the potential benefits of DPV and storage when they are both installed on the same site behind the same meter. This example assumes that the costs of each DER type remain unchanged when they are installed together, for simplicity.

In this example, the energy, generation capacity, and T&D benefits are greater in the case with DPV and storage combined. This is because the storage can be charged and discharged at times that reflect the off-peak and peak times on the system. In practice, the magnitude and direction of the impacts presented here could vary from this illustrative example, depending upon the DERs, their operating patterns, their location, and other factors.



Figure 5-3. Example of Interactive Effects on DPV and Storage

Figure 5-4 presents a different example of how interactive effects might influence the benefits of DERs. The bar on the left indicates the potential benefits of EE and DER when each is the only resource installed on a site. The bar on the right indicates the potential benefits of EE and DR when they are both installed on the same site behind the same meter. This example assumes that the costs of each DER type remain unchanged when they are installed together, for simplicity.

In this example, the generation capacity and T&D benefits are lower in the case with EE and DR combined. This is because the EE resource assumed in this example will reduce the host customer demand during the generation and T&D peak periods, thereby reducing the potential for DR to reduce customer demands at those times. In practice, the magnitude and direction of the impacts presented here could vary from this illustrative example, depending upon the DERs, their operating patterns, their location, and other factors.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 107 of 302



Figure 5-4. Example of Interactive Effects on Energy Efficiency and Demand Response Resources

Interactive effects should be accounted for in BCAs for those instances where they are likely to have a material effect. Several resources are available to help address this issue, including ACEEE 2019b; ACEEE 2020; LBNL 2020b; RAP 2019a.

5.4.3 Enabling Other DER Types

Some DERs can make it easier or more cost-effective to adopt other types of DERs. One example is when a distributed storage resource is combined with DG to improve the latter's economics. A host can do this by storing excess PV generation during periods of low electricity costs and/or demand and discharging from the storage device during periods of high electricity costs or demand, as with the duck curve example discussed earlier (ACEEE 2020). Another example may entail an NWA where the utility requires a portfolio of multiple DER types (e.g., EE, DR, and storage) to meet a system need that could otherwise not be met cost-effectively by a single DER type.

Accounting for this enabling effect is more straightforward when both the enabled and enabling DER types are installed at the same time. In these cases, the cost-effectiveness analysis should account for the collective impacts of all DER types by recognizing the interactive effects on marginal system costs and on the magnitude of kWh or kW impacts from each DER type. (See Sections 5.4.1 and 5.4.2.)

Alternatively, there may be a situation where one DER type (the enabling DER) is installed prior to the other DER type (the enabled DER). In other words, the enabling DER has a "latent capability." For example, commercial and industrial lighting controls that are initially installed as part of an EE program could later be used as DR resources if that provides a greater economic opportunity.

One option to account for this potential is through probabilistic analysis:

- Estimate the probability of the second DER type being installed. This may be formed based on market research forecasting system-level DER growth expectations and/or more granular analyses on the propensity for individual customers to adopt specific DER types.
- Forecast the expected impact of interactive effects. Determine what the net impact would be from these DER types interacting. This should account for expectations around when the second DER types may be installed.
- *Calculate a probability-weighted net impact.* Multiply the results of the first two steps to derive a value to include in the cost-effectiveness analysis.

5.5 Behind-the-Meter Versus Front-of-the-Meter Impacts

All the DER types covered in this manual could be located BTM. Only DG (e.g., community solar, CHP, distributed wind) and energy storage (e.g., batteries, compressed air energy storage, and pumped hydro) can be located FOM. The principles, concepts, and guidance provided in this manual are generally applicable to both BTM and FOM DERs. This section discusses how the benefits and costs of DERs might vary depending upon whether the DER is located BTM or FOM.

The key differences between BTM and FOM DERs are (a) BTMs are typically owned and controlled by host customers to reduce their energy costs, while FOMs are typically owned and controlled by utilities to reduce utility system costs; and (b) FOM DERs do not have host customers, so they do not create host customer impacts. These differences have four implications:

- BTM DER costs are sometime paid for by the host customer (e.g., DPV and storage), but FOM DER costs are typically paid for by the utility and eventually passed on to all utility customers.
- BTM DERs provide host customer benefits (e.g., host customer reliability) while FOM DERs do not.
- BTM and FOM DERs might be operated at different times due to the different economic incentives to the host customer versus the utility. For example, host customers with a noncoincident demand charge are likely to operate BTM storage at the time of the customer peak demand, which might not be the same time as the utility generation, transmission, or distribution peak periods. A more sophisticated rate design accounting for the time-varying nature of energy, capacity, transmission, and distribution costs, should incentivize a customer to operate BTM storage in a way that is more closely aligned with how a utility would operate a FOM storage.
- BTM DERs might be treated differently than FOM DERs in the operation of wholesale markets. While ISOs/RTOs have expanded opportunities for BTM DERs to participate in wholesale electricity markets, there may still be limitations on the ability of BTM DERs to provide and be compensated for certain wholesale market services (e.g., ancillary services).

Figure 5-5 presents a hypothetical example of how the benefits of a distributed storage resource might be different if it is located behind the meter versus in front of it. This example assumes that the BTM storage is operated by the host customer primarily to reduce bills in response to energy-based TOU rates, while the FOM storage is operated primarily to reduce generation and T&D peak period costs. In this example:

- The energy and GHG benefits of the FOM DER are less than those of the BTM DER because the host customer operated the BTM storage to reduce their energy bill in response to TOU rates based on hourly energy costs.
- The generation capacity and T&D benefits of the FOM DER are greater than those of the BTM DER because the utility operated the FOM DER to reduce generation, transmission, and distribution peak costs.
- The host customer benefits do not exist for the FOM DER.
- The other utility benefits of the FOM DER are greater than those for the BTM DER because the utility is able to use the FOM storage to sell ancillary services into the wholesale electricity markets.²⁰



Figure 5-5. Hypothetical Example of BTM Versus FOM for a Distributed Storage Resource

5.6 Air Emission Impacts

Determining the air emission impacts of a DER depends on at least two key factors. First, the BCA must clearly identify the time period for which to measure this impact (e.g., hourly, daily, monthly, etc.). The temporal granularity with which the BCA measures the impact could materially change how the DER is credited with air emission impacts relative to its actual impact. For example, if a BCA measures air emission impacts on a daily or monthly basis, the net impact will be averaged out across hours with varying air emission profiles, leading to a potential overstatement or understatement of the DER's air emission benefit/cost based on its actual performance.

Second, and related to the first factor, is understanding to the extent possible what the marginal emissions rate is of the resource being displaced by the DER's operation. For example, if a distributed PV resource operates at a time when the marginal resource is one that causes air emission costs, then the PV resource would be credited with the air emission benefit associated with displacing the emitting

²⁰ In practice, the magnitude and direction of the impacts presented here could vary from this illustrative example, depending upon the DERs, their operating patterns, their location, and other factors.

resource. Alternatively, if the PV resource is operating during a period of abundant PV production and the marginal resource is another PV resource, then there would be no corresponding air emission benefit since each resource has the same impact.

There are several analytical tools that can make probabilistic approximations of marginal emissions at hourly/diurnal/seasonal time and regional locational scales. For example, the U.S. Environmental Protection Agency (US EPA) offers the AVoided Emissions and geneRation Tool (AVERT) tool, which can be used to evaluate county, state, and regional emissions displaced at electric power plants by EE and renewable energy policies and programs. (See Appendix C.5).

Electricity system marginal emission rates can vary considerably over time as the mix of power plants and power plant dispatch changes over time. It is important that the system marginal emission rates account for expected changes to the electricity system over the course of the DER BCA study period.

Unlike generation resources that have an emissions rate associated with each kWh of generation, there is an additional layer of complexity for determining this impact for distributed storage resources and EVs (i.e., resources that can charge and discharge energy). For these resources, there are three factors affecting the direction and magnitude of the air emission impact: (1) the marginal emissions rate of the resource used to charge the resource; (2) the marginal emissions rate of the resource displaced by the discharge of the resource; and (3) the round-trip efficiency of the resource (i.e., energy losses associated with a charge-discharge cycle). The potentially significant intra- and inter-day variation in the marginal emissions rates requires particular attention to these first two factors to effectively determine the net impact on GHG and other air emissions.

5.7 Transfer Payments and Offsetting Impacts

Some DER BCA studies use the term "transfer payment" to describe the situation where one party experiences a cost and another a commensurate benefit (CA PUC 2001). This term has a specific meaning in economics, and is defined as follows:

A transfer payment is a one-way payment of money for which no money, good, or service is received in exchange. Transfer payments commonly refer to efforts by local, state, and federal governments to redistribute money to those in need. Typical examples of transfer payments include government programs such as Social Security, Medicare, student grants, and unemployment compensation.²¹

In many cases, the transfer payments identified in DER BCAs are not the same thing as transfer payments as defined by economic theory. They are not a one-way payment of money for which no money, good, or service is received in exchange, and they are not driven by government social service programs.

Nonetheless, there are some situations in DER BCAs where a DER cost experienced by one party is exactly offset by a corresponding DER benefit experienced by another party. In some situations, it may be appropriate to exclude both impacts from a BCA because the net impact is zero, but in other cases it may not be appropriate to exclude both impacts because the two impacts do not truly offset each other.

In this manual, the term "off-setting impacts" is used to refer to the situation where a DER cost (or benefit) experienced by one party is exactly offset by a corresponding DER benefit (or cost) experienced by another party and it is therefore appropriate to exclude both impacts from the BCA.

²¹ <u>http://www.businessdictionary.com/definition/transfer-payment.html.</u>

Offsetting impacts can be identified by considering two criteria. If both of these two criteria are met, then the benefits and costs in question are offsetting impacts:

- The DER cost in question is not a part of the total cost of acquiring the DER. Similarly, the DER benefit in question is not a part of the total cost of acquiring alternatives to the DER.
- Both the party incurring the cost and the party receiving the benefit are within the scope of the cost-effectiveness test being used.

Table 5-3 presents a summary of the DER impacts that are sometimes considered offsetting impacts. A "no" entry indicates that the impact is not an offsetting impact, while a "yes" entry indicates that the impact is an offsetting impact. (See Appendix F.)

BCA Test Used	Financial Incentives to Host Customers	DER Performance Incentives	Wholesale Market Price Effects	Tax Incentives
Utility Cost Test	No	No	No	No
TRC Test	Yes	No	No	No
Jurisdiction-Specific Test	Yes	No	No	Yes
Societal Cost Test	Yes	No	No	Yes

Table 5-3. Potential Offsetting Impacts

In this example, the JST is assumed to include host customer impacts but not taxpayer impacts.

Offsetting impacts should not be included in the BCA test because the benefit offsets the cost. Impacts that are not an offsetting impact should be included in the BCA test because the benefit does not offset the cost.

5.8 Variable Renewable Generation Impacts

Increasing levels of variable renewable generation can create new dynamics for the electricity grid, resulting in opportunities for DERs to either provide benefits by supporting the integration of these resources or cause additional costs by exacerbating system needs.

There are three primary factors to consider when determining the variable renewable generation impact of DERs:

- DR, distributed storage, and EVs (both in charge and discharge mode) can provide low-cost flexible resources that either take up excess demand or reduce demand to ease ramping needs on the system.
- In locations where substantial solar PV deployment leads to significant ramping of net load as the sun begins to set (the duck curve), there will be a greater need for resources to help mitigate the scale of this ramping need.
- If there are renewable curtailment requirements or periods of negative pricing due to an abundance of solar PV or wind generation relative to load, then a DER might be able to have additional value by shifting electricity grid consumption to these negative pricing hours. Conversely, DERs that operate during these periods might increase, rather than decrease, some of the utility system costs.

To effectively account for variable renewable generation impacts, a BCA should (a) account for the operational profile of the DER at a granular level (e.g., hourly), (b) account for intermittent renewable generation resources when determining utility system benefits, (c) account for ancillary services benefits related to system ramping from intermittent generation, and (d) account for periods with negative prices or when renewable resources are otherwise required to curtail operation.

5.9 Wholesale Market Revenues

Several types of DERs can participate in wholesale electricity markets in some parts of the country. Sometimes a host customer will participate directly in such markets, and sometimes DER aggregators will combine DERs from a number of host customers to participate in such markets. As DERs become increasingly common and wholesale markets become increasingly flexible, it is likely that DER will have expanded opportunities to participate in these markets over time.

When DERs participate in wholesale electricity markets, it is important to properly characterize the revenues from those markets in the DER BCA. In regions of the country with wholesale electricity markets, the energy, capacity, and ancillary services benefits should be included as utility system impacts. These benefits are typically based on the wholesale market prices of those services. Thus, the DERs' wholesale market benefits should be accounted for in all tests that include utility system impacts.

Since the energy, capacity, and ancillary services benefits of DERs are already included in the utility system benefits, there is no *additional* benefit associated with the revenues that host customers receive from wholesale markets. Therefore, revenues from DER participation in wholesale markets should not be included in most cost-effectiveness tests. If these revenues to host customers are included in cost-effectiveness tests, these benefits would be counted twice, which would violate the *Avoid Double-Counting Impacts* principle. The one exception is the PCT.²² This test is explicitly designed to identify the impacts on host customers. The benefits for this test do not include the utility system benefits that are included in the other tests. Instead, the benefits include customer bill savings. Furthermore, the host customer revenues from wholesale markets are real benefits that are experienced by host customers in addition to the bill savings. In this case, it is appropriate to include wholesale market revenues as one of the benefits in the test. While this might appear to double-count the benefits of reducing wholesale market costs, it is nonetheless an accurate depiction of the impacts on host customers.

5.10 Free-Riders and Spillover Impacts

For jurisdictions that focus on net savings (i.e. savings net of free-rider and spillover effects), the decision of whether to include participant impacts in the jurisdiction's BCA has important implications for how to treat free-ridership and spillover effects. The treatment of net savings in BCA is particularly applicable to efficiency (See Appendix H).

If a jurisdiction *does not* include participant impacts in its cost-effectiveness test, rebates to free-riders should be considered as a cost. If the jurisdiction *does* include participant impacts in its cost-effectiveness test, cost-effectiveness analyses should not include free-rider costs because the participant's portion of the measure costs are part of the baseline (i.e., would have been incurred

²² The PCT can provide useful information regarding the likelihood of customers adopting DERs, either on their own or with support from utility initiatives. This information can be helpful for designing DER initiatives, determining how much financial support to offer host DER customers, and forecasting future deployment of DERs. (See Appendix E.)

absent the program) and utility costs associated with rebates paid to free-riders is offset by a participant benefit (i.e., receipt of the rebate). By definition, jurisdictions focusing on net savings will not derive benefits from free-riders.

With respect to spillover effects, if a jurisdiction *does not* include participant impacts in its costeffectiveness test, then its cost-effectiveness analyses should not include costs associated with spillover effects because all such costs would be borne entirely by spillover customers. If the jurisdiction *does* include participant impacts in its cost-effectiveness test, cost-effectiveness analyses should capture the entire measure cost because spillover customers incur this entire cost in the process of investing in efficiency measures. By definition, jurisdictions focusing on net savings will derive benefits from spillover effects.

Table 5-4 provides a summary of guidance on the inclusion, or not, of free-rider and spillover benefits and costs in cost-effectiveness tests for jurisdictions focused on net impacts.

Category	Free-Riders		Spillover		
	Costs		Costs	Benefits	
Utility system impacts	Increase	n/a	n/a	Increase	
Participant impacts	Decrease	n/a	Increase	Increase (if applicable)	
Other impacts	n/a	n/a	Increase (if applicable)	Increase (if applicable)	
Total/Net impact	Increase only if test excludes participant impacts; otherwise no net effect	No effect under any test	No increase if test includes only utility system impacts; otherwise an increase	Increase under every test	

Table 5-4. Summary of Economic Treatment of Free-Riders

5.11 Discount Rates

A discount rate is typically used in a BCA to convert future dollars into present value dollars. The choice of discount rate can have a significant impact on present value dollars and therefore on the results of the BCA. (See Appendix G.)

A discount rate reflects a particular time preference, which is the relative importance of short- versus long-term impacts. A higher discount rate gives more weight to short-term benefits and costs relative to long-term benefits and costs, while a lower discount rate weighs short-term and long-term impacts more equally.

Different economic actors can have differing discount rates, based on their own time preferences. Further, different resource types (DERs and other energy resources) have different costs of capital and different risk profiles, which are two of the factors affecting discount rates.

Despite these differences, a single discount rate is typically used for conducting a BCA. A single discount rate is typically applied to all the benefits and costs, even though the benefits and costs are associated with energy resources with different costs of capital and risk profiles. Similarly, a single discount rate is

typically applied to a BCA, even though different parties affected by the resources in the BCA have different time preferences.²³

There are three categories of discount rates typically considered for DER assessments: the utility's weighted average cost of capital (WACC), a discount rate reflecting an average customer time preference, and a societal discount rate. A fourth option is some combination of these three categories.

The choice of discount rate is a decision that should be informed by the jurisdiction's applicable policy goals. Further, the choice of discount rate should reflect the ultimate objective of the cost-effectiveness analysis. Therefore, the regulatory perspective should be used to determine the appropriate discount rate.

Based on the considerations described above, regulators should determine a discount rate that best reflects the jurisdiction's regulatory perspective, with input from stakeholders. Table 5-5 offers suggestions for how this determination might be made (See Appendix G).

Consideration	If the answer is "yes"
Time Preference Considerations:	
Does the regulatory perspective suggest the same time preference as utility investors?	Choose a discount rate equal to the utility WACC.
Does the regulatory perspective suggest placing a higher value on long-term impacts than utility investors?	Choose a discount rate less than the utility WACC.
Does the regulatory perspective suggest the same time preference as that of all utility customers?	Choose a discount rate that is represents all utility customers on average.
Does the regulatory perspective suggest the same time preference as that of society?	Choose a societal discount rate.
Does the regulatory perspective suggest placing a lower value on long-term impacts than society does?	Choose a discount rate greater than a societal discount rate, or at the high end of the range of societal discount rates.
Risk Considerations (for use in situations where resource-sp	pecific risks are not accounted for in the BCA inputs):
Will DERs result in a net reduction in risk relative to alternatives?	Choose a relatively low-risk discount rate, such as the societal discount rate.
Will DERs result in a net increase in risk relative to alternatives?	Choose a relatively high discount rate.

Table 5-5. Considerations for Determining a Discount Rate

²³ Sensitivities can inform analysis of the effect of different discount rates. A single discount rate is typically used for any scenario or sensitivity.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 115 of 302

Part III: BENEFIT-COST ANALYSIS FOR SPECIFIC DER TYPES

Overview

This part of the manual includes separate chapters addressing each type of DER, including:

- energy efficiency,
- demand response,
- distributed generation,
- distributed storage, and
- electrification.

Each chapter includes a summary of key points; a description of the DER type; a summary of the utility system, host customer, and societal impacts; a discussion of the various factors that affect benefits and costs; and guidance on some key challenges of conducting BCAs for each DER type.

Each chapter discusses the benefits and costs of each DER type as if it were operated in isolation of the other DER types. This is partly for clarity and partly for readers and stakeholders who are interested in single-DER analyses. Part IV of the manual addresses issues that arise when multiple DERs are combined in different ways.

Each chapter includes a set of tables identifying the potential impacts of the DER type by utility system, host customer, and societal impacts. These tables are compiled across all the DERs in the Summary of this manual (pp xi–xii) to show the potential impacts across all the DER types.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 116 of 302

6. ENERGY EFFICIENCY RESOURCES

This chapter describes the benefits and costs most relevant to energy efficiency (EE) resources. It identifies key factors that affect EE benefits and costs and provides guidance on addressing common challenges with EE cost-effectiveness analyses.

6.1 Summary of Key Points

- EE resources include technologies, services, measures, or programs that reduce end-use energy consumption by host customers, and that are funded by, promoted by, or otherwise supported on behalf of all electricity and gas utility customers.
- EE resources provide utility system benefits in the form of avoided generation and T&D costs, and other benefits such as reduced risk, and improved reliability and resilience of the capacity system. The costs typically include program administration costs, financial incentives, and utility performance incentives, where relevant.
- The temporal and locational benefits of EE can vary considerably across seasons and during different hours of the day, and as well as by the physical location on the energy distribution network.
- Some electric EE resources will reduce other fuel consumption resulting in other fuel benefits, while others will increase other fuel consumption resulting in other fuel costs.
- EE resources can create a variety of host customer NEIs, depending upon the technology and the host customer. The EE NEIs can include transaction costs, other resource Impacts, asset value, productivity, economic well-being, comfort, health & safety, and satisfaction/pride.
- EE resources can offer significant reductions in air emissions. As with all DERs, estimates of air emission impacts should account for the time period when the EE resource operates and the marginal emission rates of the utility system at that time.
- For the purpose of deciding which EE resources to invest in or otherwise support on behalf of utility customers, BCAs should be applied at the program, sector, and/or portfolio levels.
- Addressing the impacts of EE resources throughout their full effective useful life requires properly defining counterfactual (baseline) conditions for both savings and costs, particularly with respect to early-replacement projects.
- EE resources will typically create lost revenues. These might lead to increased rates, depending upon the magnitude of lost revenues, the magnitude of EE utility system costs, and the magnitude of utility system benefits. Rate impacts should not be included in BCAs, but should instead be accounted for in rate, bill, and participation analyses.

6.2 Introduction

For the purpose of this manual, EE resources include technologies, services, measures, or programs that reduce end-use energy consumption by host customers.²⁴ EE resources broadly include both (a) investments that provide the same level of (or better) service with lower energy use; and (b) conservation measures that may result in some reductions in amenity (e.g. removal of 2nd refrigerator, adjustment to thermostat setpoints through behavior programs, etc.).

Some efficiency measures and programs can offer at least the potential to be DR measures in the future. Examples include controllable "smart thermostats," variable speed drives on motors, and lighting controls. Such measures can both (1) enable customers' loads to be controlled through dispatchable DR in the future; and (2) enhance customers' ability to respond to price signals to lower demand during hours when energy is expensive and/or when there are capacity constraints on the system.

The fundamental BCA principles described in Chapter 2 can be used to identify the relevant benefits and costs for EE, as with all DER types. In many cases, identifying whether a particular benefit or cost is relevant to EE requires defining the specific EE technology and use case.

6.3 Benefits and Costs of Energy Efficiency Resources

The tables in this section summarize the full range of potential benefits and costs of EE resources (see Chapter 4 for definitions of impacts). Each impact is described as a benefit, a cost, or either, depending on the most common applications or use cases of this technology. There might be some less-common applications where a cost could be a benefit, or vice versa. The tables include notes on applicability that provide further explanation for those impacts that may be either a cost or benefit. Table 6-2 presents the impacts of EE on the gas utility system. All EE utility system impacts presented in these tables should be included in BCA tests. The remaining EE impacts presented in Table 6-3 (Host Customer) and Table 6-4 (Societal) should be included in BCA tests to the extent they are relevant to the jurisdiction's applicable policy goals (see Chapter 3).

²⁴ Some investments on the utility side of the meter that reduce customers' end-use consumption, such as conservation voltage regulation, are included in this definition.

Туре	Utility System Impact	Benefit or Cost	Notes/Typical Applicability	
	Energy Generation	•	Always a benefit because EE reduces electricity generation	
	Generation Capacity	•	A benefit because EE can reduce generation peaks; exceptions are EE measures that do not reduce system capacity requirements	
.	Environmental Compliance	•	A benefit because EE reduces electricity generation	
Generation	RPS/CES Compliance	•	Always a benefit because EE reduces electricity sales	
	Market Price Response	•	Always a benefit in jurisdictions with wholesale energy and/or capacity markets	
	Ancillary Services	•	Can be a modest benefit	
Transmission	Transmission Capacity	•	A benefit because EE can reduce transmission peaks; same exceptions as for generation capacity	
	Transmission Line Losses	•	Always a benefit because EE reduces transmission volumes	
	Distribution Capacity	•	A benefit because EE can reduce distribution peaks; same exceptions as for generation capacity	
Distribution	Distribution Line Losses	•	Always a benefit because EE reduces distribution volumes	
	Distribution O&M	•	Always a benefit because EE reduces distribution volumes	
	Distribution Voltage	•	A benefit for conservation voltage regulation programs	
	Financial Incentives	•		
	Program Administration Costs	•	Always a cost, where relevant	
	Utility Performance Incentives	•		
	Credit and Collection	•	A benefit because customer savings make bill payment easier, especially for low-income programs	
General	Risk	•	Always a net benefit because EE is lower risk than supply-side resources	
	Reliability	•	A benefit (if not already captured in avoided generation and/or avoided T&D categories)	
	Resilience	•	A benefit where relevant – e.g. when large portions of a system need reconstruction, a history of efficiency investments reduces amount of capacity that requires reconstruction	

Table 6-1. Potential Impacts of Energy Efficiency: Electric Utility System

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

Table 6-2 presents the potential benefits and costs of EE on gas utilities and other fuel systems. Gas utility EE programs are most likely to reduce natural gas consumption. Electric utility EE programs can sometimes reduce and sometimes increase natural gas and other fuel consumption.

Туре	Non-Electric Energy System Impact	Gas Utility EE Programs	Electric Utility EE Programs that Affect Other Fuels	Notes/Typical Applicability
Other Fuel: Energy	Fuel and Variable O&M	•	•	Always a benefit for gas EE; electric EE programs that reduce other fuels will create a benefit ²⁵
	Capacity	•	•	A benefit for gas EE; exceptions are for measures that don't save in winter and/or during distribution system peak hours; electric EE programs that reduce other fuels create a benefit
	Environmental Compliance	•	٠	Always a benefit for gas EE; electric EE
	Market Price Response	•	•	benefit
Other Fuel: General	Financial Incentives	•	0	Always a cost for gas EE; electric EE
	Program Administration Costs	•	0	affect this impact
	Utility Performance Incentives	•	o	A cost in jurisdictions that offer them; electric EE programs that reduce other fuels will not affect this impact
	Credit and Collection Costs	•	•	A benefit for gas EE, particularly for low- income programs; electric EE programs that reduce other fuels create a benefit
	Risk	•	•	Always a benefit for gas EE; electric EE programs that reduce other fuels create a benefit
	Reliability	•	•	A benefit for gas EE (exceptions the same as for those capacity described above); electric EE programs that reduce other fuels create a benefit
	Resilience	•	•	May be a benefit for gas EE; electric EE programs that reduce other fuels create a benefit

Table 6-2. Potential Impacts of Energy Efficiency: Gas Utility or Other Fuel System

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

²⁵ Electric EE programs can sometimes indirectly increase the use of other fuels, e.g., when more efficient lighting produces less heat thereby requiring additional space heating. In these cases there will be a cost to the other fuel.

Table 6-3 presents the potential host customer impacts of EE (see Section 4.4).

Туре	Host Customer Impact	Benefit or Cost	Notes/Typical Applicability
	Host Customer Portion of DER Costs	•	Typically a cost; exceptions are when utility program pays the full measure cost (e.g. low-income programs) or there is no measure cost (e.g. behavior programs or disposal of extra appliances)
	Interconnection Fees	0	Not applicable for EE
	Risk EE reduces exposure to	EE reduces exposure to future fuel price volatility	
Customer	Reliability	•	Lower loads enable back-up capacity (storage or generation) to be smaller and less expensive
	Resilience	•	Some EE measures may offer short-term resilience benefits (e.g. high levels of insulation allow customers to remain comfortable without heating fuel for longer periods)
	Tax Incentives	•	Potentially a benefit for those resources where they apply
	Host Customer NEIs	•	A benefit or cost depending on the NEI (see Section 6.4.4)
	Low-Income NEIs	•	A benefit; applies only to low-income EE programs

 Table 6-3. Potential Impacts of Energy Efficiency: Host Customer

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

Table 6-4 presents the potential societal impacts of EE.

Table 6-4. Potentia	l Impacts of Energy	Efficiency	Societal
---------------------	---------------------	-------------------	----------

Туре	Societal Impact	Benefit or Cost	Notes/Typical Applicability
Societal	Resilience	•	A benefit because improving resilience for the grid and host customers can also alleviate stress on a variety of public institutions during responses to catastrophes
	GHG Emissions	•	A benefit; magnitude depends on electric generation mix and any other fuel savings
	Other Environmental	•	
	Economic and Jobs	•	A benefit as EE resources are usually more labor intensive than electric and/or other energy supply
	Public Health	•	A benefit resulting from reduced environmental emissions; need to insure no double-counting of host customer NEIs and societal environmental impacts
	Low Income: Society	•	For low-income EE programs there can be societal benefits associated with environmental justice and community stability
	Energy Security	•	A benefit to the extent that efficiency reduces reliance on fuels with security concerns, e.g., imported fossil fuels

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

6.4 Key Factors that Affect Energy Efficiency Impacts

6.4.1 Technology Characteristics

EE differs from other DERs in various ways that have important implications for benefit-cost analysis. EE is an immensely diverse and modular resource: there are hundreds if not thousands of different types of EE measures, addressing numerous energy uses in a variety of different applications and buildings. They can also be promoted through a wide range of programs, implementation options, and market transactions. This modularity and diversity of EE means that comprehensive portfolios of EE programs can produce significant levels of load reduction in most every hour of the year.

Some EE measures can enable grid flexibility. For example, smart thermostats or networked lighting controls can enable dispatchable DR or facilitate customers' ability to respond to price signals. BCA for EE should consider the potential enabling of other DERs to meet system needs such as grid flexibility or other objectives.

6.4.2 Technology Operating Profile

The temporal and locational value of energy efficiency savings can vary considerably across seasons and during different hours of the day, and by the physical location on the energy distribution network. Thus, the temporal and locational attributes of EE have important implications for benefit-cost analysis, as these attributes will dictate the extent to which EE produces energy, generating capacity, and T&D savings.

The average hourly load shape of when energy savings occur will depend on the mix of efficiency measures in the efficiency program portfolio, the host customers to whom the measures are being promoted, the local climate, and a variety of other factors. For example, residential lighting savings will typically be higher in the winter than in the summer, a larger portion of residential cooling savings than of commercial cooling savings will occur in the evening, and industrial motor or process efficiency savings will tend to be more consistent across the hours of industrial operations than either residential or commercial energy savings.

6.4.3 Other Fuel Impacts

Table 6-2 presents the potential benefits and costs of EE on gas utilities and other fuel systems. Gas utility EE programs should always reduce gas consumption. Electric utility EE programs will often reduce other fuel consumption resulting in other fuel benefits. However, some electric efficiency measures can lead to increases in consumption of other fuels.

As shown in Table 6-2 non-electric cost savings resulting from efficiency improvements can extend beyond just commodity (i.e., fuel) savings. For example, when efficiency lowers natural gas consumption at times of gas system peak demand, it can help defer the potential for future capital investments to upgrade the gas transmission and/or distribution system infrastructure. There may be similar "capacity" savings associated with reduced infrastructure for distribution of petroleum products.

6.4.4 Host Customer Non-Energy Impacts

There can be a variety of host customer NEIs from EE investments. Although host customer NEIs can be either benefits or costs, depending on the EE measure, program, or host customer, the impacts in most efficiency program portfolios are primarily benefits.

Table 6-5 presents a summary of host customer NEIs that might potentially be created by EE resources. These impacts can sometimes be in the form of benefits and sometimes costs, depending on the most common applications for EE resources. The presence, direction, and magnitude of these impacts will depend upon many factors, including the EE program, the type of host customer (e.g., low-income, residential, commercial, industrial), and more.

Category	Cost or Benefit	Examples	
Transaction costs	•	 Application fees and processes 	
Other Resource Impacts	•	 Reduced water consumption/cost Reduced waste streams 	
Asset value	•	 Equipment functionality/performance improvement Equipment life extension Increased building value Increased ease of selling or leasing building 	
Productivity	•	 Improved labor productivity Reduced operation and/or maintenance (O&M) costs Reduced spoilage/defects Reductions in occupant turnover and/or increased occupancy rates in leased buildings Impact of improved aesthetics, comfort, etc. on product sales 	
Economic well- being	•	 Fewer bill-related calls to utility Fewer utility disconnections and reconnections Reduced foreclosures (especially for low-income host customers) Fewer moves (especially for low-income host customers) Other manifestations of improved economic stability (especially for low-income host customers) 	
Comfort	•	 Thermal comfort Noise reduction Light quality 	
Health & safety	•	 Improved well-being due to reduced incidence of illness—chronic (e.g., asthma) or episodic (e.g., hypothermia or hyperthermia) Reduced medical costs (emergency room visits, drug prescriptions) Fewer sick days (work and school) Reduced deaths Reduced insurance costs (e.g., for reduced fire, other risks) 	
Satisfaction/Pride	•	 Contribution to addressing environmental concerns 	

Table 6-5. Potential Host Customer Non-Energy Impacts of Energy Efficiency

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; O = not relevant for this resource type.

Specific examples of EE host customer non-energy benefits include:

• Other Resource/Water savings: certain EE measures, such as low-flow showerheads, pre-rinse spray valves for commercial dish cleaning, and efficient clothes washers save energy by reducing the amount of hot water (and therefore the amount of water) needed to meet customers' needs.

- Asset value building durability improvements: for example, sealing and insulating attics can reduce or eliminate ice dams which occur when melting snow on a roof refreezes as ice at the edge of the roof, backing up other melting snow to the point where it can leak and damage the roof, walls, and other structural elements of the home.
- Productivity benefits
 - O&M cost savings: for example, efficient LED lighting measures not only save electricity but are longer-lived than the standard/baseline lighting products they replace and, therefore, eliminate some future replacement costs that would otherwise be incurred.
 - Business productivity and/or profitability improvements: for example, higher quality LED lighting that can enhance aesthetics of retail product displays, ventilation efficiency improvements that can also improve indoor air quality and worker productivity, reductions in heating bills that can reduce vacancy rates for multi-family building owners, and reductions in industrial waste streams that can reduce energy consumption while lowering disposal costs.
- *Improvements to economic well-being:* for example, EE programs, particularly those targeted to low-income customers, can reduce stress associated with challenges in paying bills, reduce foreclosures, reduce the number of moves from one home or apartment to another.
- *Comfort improvements:* for example, sealing and insulating a home not only saves energy but improves the comfort of the home by reducing drafts and enabling more even distribution of heat.
- *Health and safety improvements:* for example, many home efficiency retrofit programs routinely test for, identify, and address problems such as back-drafting of combustion gases (including carbon monoxide) from fossil fuel-fired heating or water heating systems and mold problems associated with improper ventilation.

Examples of EE host customer non-energy costs include some losses of amenity (e.g., when second refrigerators are voluntarily removed and recycled), increased maintenance costs (e.g., for operating and maintaining CHP equipment), and aesthetic concerns.

For more information see: Tetra Tech 2011; Tetra Tech 2012; Skumatz 2014; NEEP 2017; and NESP 2019.

6.4.5 Air Emissions Impacts

EE resources can reduce air pollution emissions by reducing the amount of generation that fossil fuelfired power plants will need to produce.

EE resources that affect multiple fuels can also have impacts on other fuels. For example, building envelope improvements can produce both electric cooling savings in the summer and natural gas (or other fossil fuel) heating savings in winter—with both forms of savings producing emission reductions.

As discussed in Chapter 5, air emission impacts should account for two important factors: (a) the time period for which the DER operates; and (b) the marginal emission rates of the utility system at the time when the DER operates.

The *Ensure Symmetry* principle requires accounting for both reduced and increased air emissions from EE resources, to the extent that they occur.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 125 of 302

6.5 Common Challenges in Estimating EE Benefits and Costs

There are a variety of challenges in estimating benefits and costs for EE resources. Fundamentally, conducting BCA for EE resources requires ensuring that the NSPM Principles are appropriately applied, in particular with regard to ensuring alignment with applicable policies and symmetrical treatment of relevant benefits and costs. These issues are addressed extensively in Chapters 2 and 3. While it may be useful to understand costeffectiveness at all four of these levels, when deciding which DERs merit utility acquisition they should be evaluated at the program, sector, or portfolio level.

Other common issues covered in this section are: (1) the level of aggregation of EE impacts at which it is appropriate to assess cost-effectiveness; and (2) how to assess the impacts of early replacement measures (i.e., those installed prior to the end of the useful life of the existing equipment.) More detailed discussions of both of these issues, as well as treatment of fixed and variable costs, changing cost and/or benefit baselines, and both free-rider and spillover effects, is provided in Appendix H.

6.5.1 Assessment Level for Benefit-Cost Analysis

The cost-effectiveness of EE resources can be evaluated at several different levels of assessment. Assessments can focus on individual measures, individual customer-specific projects, individual programs combining multiple measures and/or projects, sectors (e.g., all residential or all business programs), and/or on a portfolio of programs across all sectors.

Cost-effectiveness analyses for EE resources should be applied – i.e., used to screen investments as worthwhile or not – at the program, sector, or portfolio levels. Applying cost-effectiveness results at the measure and project levels can have perverse implications, as further described in Appendix H. In some cases, it could reduce the overall net economic benefits of efficiency investments for the following reasons:

- A customer's interest in a non-cost-effective measure may be key to persuading the customer to install a package of measures that are cost-effective in aggregate;
- A customer's interest in a non-cost-effective measure may be key to the development of a relationship with the customer that can lead to installation of cost-effective measures in the future; and/or
- Installation of a non-cost-effective measure may be necessary in order to technically or safely enable the installation of other cost-effective measures.

Another disadvantage of assessing cost-effectiveness at the measure or project level is that it can be difficult to account for NEIs, hard-to-monetize impacts, or additional considerations at the measure level. Some NEIs, such as improved health and safety, are obtained through a package of multiple measures, and it is impractical to apply such impacts on each measure.

6.5.2 Early Replacement Measures

Many efficiency program portfolios include what are commonly called early replacement measures. These are measures in which an efficient new product is used to replace an inefficient existing piece of equipment that is still functioning and would not otherwise have been replaced—at least not until sometime in the future.

For example: an existing and functioning T12 commercial light fixture, that would normally not be replaced for another four years, is replaced this year through an efficiency program by a new LED light fixture with an assumed life of 12 years. In this example, the initial savings are a function of the difference in efficiency between the existing T12 and a new LED fixture, and the initial cost is the full cost of the LED fixture. However, the initial savings level would only last for the estimated four years of remaining life for the existing T12, and the installation of the LED fixture today would eliminate the cost that otherwise would have been incurred four years from now when the T12 would have been replaced absent the efficiency program.

A key to proper assessment of the cost-effectiveness of early replacement EE measures is to develop reasonable estimates of what would have occurred absent the EE program (i.e., the baseline)—both for savings and costs.

- With respect to savings, programs should employ what is commonly called a dual baseline. In the case of the T12 replacement example, that would mean (a) four years of savings equal to the difference in efficiency between the existing T12 and the new LED and then (b) eight years of savings equal to the (likely smaller) difference in efficiency between the new fixture that would have been purchased four years from now and the new LED installed today.
- With respect to cost, in jurisdictions that include participant impacts, the BCA needs to consider not only the upfront cost of the new LED fixture installed, but also the cost savings for the customer of not having to buy whatever new fixture they otherwise would have purchased four years from now. However, to align the baseline timeline with the efficiency investment timeline (i.e., to ensure an "apples to apples" assessment of the difference in costs) only a portion of the future cost that was eliminated should be treated as a cost savings. Put another way, if the baseline product that would have been installed four years from now would have had an expected life of 12 years, it is only appropriate to include eight years' worth of the cost savings (to align the cost comparison with the life of the LED measure installed today). See Appendix H for further details.

6.6 Lost Revenues and Rate Impacts

Lost revenues and potential rate impacts are not appropriate to include in cost-effectiveness analyses, and should instead be analyzed separately using rate, bill, and participation analyses. (See Section 2.3 and Appendix A.) In conducting BCAs, therefore, lost revenues should be identified so that they can be properly excluded from BCAs and properly included in rate, bill, and participation analyses.

In general, several key factors affect the extent to which DERs might create rate impacts:

- Increases in utility system costs will put upward pressure on rates.
- Reductions in utility system costs will put downward pressure on rates.
- Reductions in sales from DER resources will put upward pressure on rates.
- Increases in sales from DER resources will put downward pressure on rates.
- Rate design will affect the amount of lost or increased revenues created by the DER.

EE resources will typically increase utility system costs because of EE program costs, but they will reduce utility system costs by avoiding generation, transmission, and distribution costs (see Table 6-1).

EE resources will typically create lost revenues by (a) allowing host customers to reduce their energy bills by reducing their energy consumption; and (b) allowing host customers to reduce their energy bills by reducing demand charge payments, where such charges are in place.

Regulators and other stakeholders that are concerned about EE rate impacts might consider (a) conducting a long-term rate, bill, and participation analysis; (b) seeking EE program designs likely to mitigate equity concerns; (c) examining ways to expand customer participation, especially among low-and moderate-income customers; and (d) periodically reviewing impacts as EE programs are implemented.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 128 of 302

7. DEMAND RESPONSE RESOURCES

This chapter describes the benefits and costs most relevant to demand response (DR) resources. It identifies key factors that affect DR benefits and costs and provides guidance on addressing common challenges with for DR cost-effectiveness analyses.

7.1 Summary of Key Points

- DR resources include a broad range of technologies, programs, and actions that can be used to control or manage electricity or gas demands, including dispatchable and non-dispatchable DR resources, DR for economic purposes, DR for reliability purposes, and time-varying rates.
- DR resources that result in load-shifting will create benefits during the targeted peak periods and costs during the periods to which loads are shifted.
- DR benefits will significantly depend upon the resource's operating profile, including how the resource will respond to price and dispatch signals.
- DR resources do not typically result in reduced consumption of other fuels. Some DR resources might increase the use of other fuels if the host customer relies upon some form of backup generation.
- DR resources provide avoided generation and avoided T&D benefits to the utility system, though in some cases load shifting may results in costs. Other benefits can include reduced risk and improved resilience. Costs typically include program administration costs, financial incentives and performance incentives, where relevant.
- DR resources can create several host customer NEIs, depending upon the technology and the host customer. These DR NEIs can include transaction costs, asset value, productivity, economic well-being, comfort, satisfaction/pride.
- DR resources can sometimes reduce air emissions by reducing the amount of fossil-fired generation or by shifting loads from high-emission periods to low-emission periods. DR resources can sometimes increase air emissions, for example, when DR resources rely upon fossil-fueled back-up power, or when DR resources shift loads from low-emission periods to high-emission periods.
- Many DR benefits will depend upon the counterfactual baseline for host customers.
- Some DR resources can provide benefits related to multiple grid services.
- DR resources will typically create lost revenues. These might lead to increased rates, depending upon the magnitude of lost revenues, the magnitude of DR utility system costs, and the magnitude of utility system benefits. Rate impacts should not be included in BCAs, but should instead be accounted for in rate, bill, and participation analyses.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 129 of 302

7.2 Introduction

For the purposes of this manual, DR is a broad term that encompasses a variety of different technologies and programs that can provide a range of capabilities and services (e.g., shape, shift, shed, or shimmy). DR is a temporary and/or voluntary change in demand, and in that way differs from demand-focused energy efficiency measures which seek to obtain a permanent reduction in energy use during both peak and off-peak periods.

As the grid hosts an ever-expanding array of DR technologies, programs, and functionalities, the role and potential purposes of DR programs will also expand. Originally developed primarily to address coincident system

From the utility perspective, DR is a change in the power consumption by a customer that assists the utility to better match the demand for power with economic supply.

demand peaks of relatively short duration, DR programs are evolving to address additional system needs, such as: mitigating demand fluctuations associated with EV charging; addressing rapid increases or decreases in demand associated with variable generation; ramping service to mitigate rapid-start requirements on generators, non-coincident peak reduction, locational peak reductions, and others.

The fundamental BCA principles described in Chapter 2 can be used to identify the relevant benefits and costs for DR, as with all DER types. In many cases, identifying whether a particular benefit or cost is relevant to DR requires defining the specific DR technology and use case.

Demand Response Services

DR resources include programs and actions taken to change energy use during specified time periods or under specific conditions, with the aim of cost-effectively altering customer usage and system demand.

Historically, the term "demand response" referred primarily to short-term reductions in electricity demand created at customer premises in response to utility price signaling or load control. From the utility perspective, DR is a change in the power consumption by a customer that assists the utility to better match the demand for power with economic supply. Because utility supply costs can rise dramatically during relatively brief peak demand periods, DR can be a cost-effective alternative to generation supply. Increasingly, DR can be a cost-effective solution to infrastructure upgrades and grid constraints as part of an NWS. (See Chapter 12.)

DR can also be used to increase consumption during particular time periods in order to make greater use of certain resources. For example, DR can be used to encourage customers to pre-cool a building in advance of a period of system peak demand. Finally, behavioral DR with technology advancements and paired with TOU rates can change customer energy usage behavior and can modify customer load profiles.

DR services have historically been characterized as applying to either load reduction, load-shifting, ancillary services, or load-shaping. A more recent DR taxonomy and analytic framework groups these services in categories to account for the increasing range of DR services and to facilitate comparisons between the cost and value associated with diverse and flexible loads. These core categories are: Shape, Shift, Shed and Shimmy (LBNL 2016, 1-1 to 1-2).

• Shape captures DR that reshapes customer load profiles through price response or on behavioral campaigns—"load-modifying DR"—with advance notice of months to days.
- Shift represents DR that encourages the movement of energy consumption from times of high demand to times of day when there is surplus of renewable generation. Shift could smooth net load ramps associated with daily patterns of solar energy generation.
- Shed describes loads that can be curtailed to provide peak capacity and support the system in emergency or contingency events—at the statewide level, in local areas of high load, and on the distribution system, with a range in dispatch advance notice times.
- *Shimmy* involves using loads to dynamically adjust demand on the system to alleviate short-run ramps and disturbances at timescales ranging from seconds up to an hour.

Demand Response Classifications

Figure 7-1 presents a classification of DR resources, differentiated by dispatchability (e.g., dispatchable versus non-dispatchable), purpose (e.g., economic benefits or reliability), and by types of program (e.g., TOU rates, critical peak pricing, direct load control) (NERC 2011):

- Dispatchable versus Non-Dispatchable: When used with DR resources, the term "dispatchable" means that the timing and level of response is under the control of the utility, either through technical control or by the terms of a contract, or both. Examples of dispatchable forms of DR include air conditioning switches, thermostats, or water heater programs that respond directly to a utility signal. The term "non-dispatchable" refers to programs and measures without such controls and includes time-varying rates that send price signals to encourage customers to alter their energy usage during particular hours.
- Program Purposes: DR resources may be designed and called upon for several purposes, and often provide multiple services simultaneously. Figure 7-1 shows the primary purposes served by various types of DR. The services provided by DR may include providing economic benefits by enabling a more cost-effectively scheduled energy mix and/or market purchases and by contributing to reliability by providing or reducing requirements for ancillary services and peak demand capacity, including during emergency or high-cost events. Non-dispatchable DR resources like time-varying rates can help shape loads to achieve economic, reliability, and customer-savings benefits.



Figure 7-1. Demand Response Classifications, Purposes, and Example Programs

Source: NERC 2011.

7.3 Benefits and Costs of Demand Response Resources

The tables in this section summarize the full range of potential benefits and costs of DR resources (see Chapter 4 for definitions of impacts). Each impact is described as a benefit, a cost, or either, depending on the most common applications of DR. In less-common applications, there may be cases where a cost could be a benefit, or vice versa. The tables include notes on applicability that provide further explanation for those impacts that may be either a cost or benefit.

Table 7-1 presents the potential benefits and costs of DR on the electric utility system, while Table 7-2 presents the impacts of DR on the gas utility system. All DR utility system impacts presented in these tables should be included in BCA tests. The remaining DR impacts presented in Table 7-3 (Host Customer) and Table 7-4 (Societal) should be included in BCA tests to the extent they are relevant to the jurisdiction's applicable policy goals (see Chapter 3).

Туре	Utility System Impact	Benefit or Cost	it Notes/Typical Applicability st	
	Energy Generation	•	A benefit for DR that reduces electricity generation, but load-shifting might result in costs	
	Generation Capacity	•	A benefit because reduced system peak demand is frequently the primary objective of DR	
Generation	Environmental Compliance	•	A benefit for DR that reduces electricity generation, but load-shifting might result in costs	
	RPS/CES Compliance	•	A benefit for DR that reduces electricity generation, but load-shifting might result in costs	
	Market Price Response	•	A benefit since DR tends to target higher-priced supply in wholesale markets; depends on generation market operation	
	Ancillary Services	٠	A benefit due to load reductions during peak periods	
	Transmission Capacity	•	A benefit due to load reductions during peak periods	
Transmission	Transmission System Losses	•	A benefit due to load reductions during peak periods	
Distribution	Distribution Capacity	٠	A benefit due to load reductions during peak periods; however, circuit-	
	Distribution System Losses	•	level peaks are not always aligned with system peaks, and thus load- shifting to address system peaks could result in increased peak	
	Distribution O&M	•	demand at the circuit or substation level, and vice versa	
	Distribution Voltage	•	A benefit when DR is used to manage voltage fluctuations on the grid	
	Financial Incentives	•	Always a cost, where relevant	
	Program Administration Costs	•	Always a cost, where relevant	
	Utility Performance Incentives	•	A cost (where jurisdictions have performance incentives)	
General	Credit and Collection	•	A benefit because customer savings make bill payment easier, especially for low-income customers	
	Risk	•	A benefit due to reduced load during peak periods	
	Reliability	•	A benefit due to better asset utilization of generation resources and enhanced grid flexibility	
	Resilience	•	Potentially a benefit due to reduced restart load	

Table 7-1. Potential Impacts of Demand Response: Electric Utility System

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

Table 7-2 presents the potential benefits and costs of DR on gas utilities and other fuel systems. Gas utility DR programs are most likely to reduce gas consumption, but there may be examples of gas load-shifting programs that slightly increase gas consumption. Electric utility DR programs will typically increase other fuel consumption if they rely upon back-up generation from other fuels. Electric utility DR programs will rarely, if ever, reduce other fuel consumption.

Туре	Non-Electric Energy System Impact	Gas Utility DR Programs	Electric Utility DR Programs that Affect Other Fuels	Notes/Typical Applicability
Other Fuel: Energy Other Fuel: General	Fuel and Variable O&M	•	•	Back-up generation or load-
	Capacity	•	•	shifting from DR can increase costs
	Environmental Compliance	•	•	and associated impacts of other
	Market Price Response	•	•	fuels
	Financial Incentives (e.g., rebates)	•	0	
	Program Administration Costs	•	0	A cost for gas utility DR programs; not relevant otherwise
	Utility Performance Incentives	•	0	
	Credit and Collection Costs	•	0	A benefit for gas utility DR programs; not relevant otherwise
	Risk	•	•	Back-up generation or load-
	Reliability	•	•	shifting from electric utility DR can increase risk and reduce reliability
	Resilience	•	•	and resilience of other fuels

Table 7-2. Potential Impacts of Demand Response: Gas Utility or Other Fuel System

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

Table 7-3 presents the potential host customer impacts of DR.

Table 7-3. Potential Impacts of Demand	Response: Host Customer
--	--------------------------------

Туре	Host Customer Impact	Benefit or Cost	Notes/Typical Applicability
Host Customer	Host portion of DER costs	•	A cost when they exist
	Interconnection Fees	ο	Not relevant for DR
	Risk	ο	DR has little net risk impact on the host customer
	Reliability	•	A benefit or a cost, ddepending on electric and thermal ride-through capacity (e.g., the ability of the building to stay cool or warm during period
	Resilience	•	of reduced electricity consumption); load-shifting or back-up generation might result in costs
	Tax Incentives	•	A benefit, where applicable
	Host Customer NEIs	•	A benefit or cost depending upon the NEI (see Section 7.4.4)
	Low-income NEIs	•	A benefit for low-income DR programs only

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; O = not relevant for this resource type.

Table 7-4 presents the potential societal impacts of DR.

Туре	Societal Impact	Benefit or Cost	Notes/Typical Applicability
	Resilience	•	
Societal	GHG Emissions	•	Potentially benefits due to reduced use during neak periods
	Other Environmental	•	especially where most inefficient and most expensive generator
	Economic and Jobs	•	are on the targeted margin, or where DR is used to help integrate variable renewable generation; potentially costs if load-shifting
	Public Health	•	and backup generation create environmental and public health
	Low Income: Society	•	costs
	Energy Security	•	

|--|

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; O = not relevant for this resource type.

7.4 Key Factors that Affect Demand Response Benefits and Costs

DR involves a change in load at a customer's premises, and thus the technical capabilities of the DR resource can be as diverse as the end-uses providing it. DR programs can also be designed to use a variety of communication technologies, incentive levels, customer engagement strategies, and performance requirements. All of these have important implications for cost-effectiveness.

DR programs are increasingly being implemented to address an expanding range of grid system conditions and issues. For example, DR programs aimed at *increasing* load in order to reduce renewable energy curtailment have been implemented, in addition to traditional DR programs focused on load reductions. Understanding and characterizing the problems that DR seeks to address is the first step in evaluating cost-effectiveness.

DR cost-effectiveness can also be profoundly impacted by whether a customer decides to simply reduce load, shift it to other hours, or use back-up generators to replace grid-supplied energy. Other important considerations for cost-effectiveness include the capabilities and performance characteristics of DR technologies themselves, as well as the ways that these resources are deployed by DR program managers.

7.4.1 Technology Characteristics

The type of benefits and costs provided by DR resources depend on two key technology characteristics: DR performance and DR predictability.

DR performance includes factors such as the speed, precision, and duration capabilities of the resource. For example, more traditional forms of DR (such as one-way water heating programs) may be called upon only to provide load reductions for capacity needs, while more advanced technologies with two-way communications (such as two-way grid-interactive water heaters) can be called upon to provide a broader range of services, including very precise frequency regulation, while also providing greater visibility to the DR program manager.

Understanding and characterizing the problems that DR seeks to address is the first step in evaluating costeffectiveness.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 135 of 302

Further, whether the DR resource can modulate load in both directions (both increasing and decreasing load) and the location of the resource on the system (for location-targeted reductions).

DR predictability depends on the level of control the utility has over the DR resource. In some cases, a utility request to its customers to reduce usage during on-peak hours costs relatively little to the utility and also likely creates relatively small participant costs (since reduction in use is voluntary). This type of DR program is not very predictable, and reductions could occur

The level of customer awareness and engagement will substantially impact the effectiveness of all programs.

almost anywhere on the system. In contrast, contract- and tariff-based DR programs offer precision in benefit and cost estimation, but also increased program costs in the form of higher incentives.

A residential thermostat control program provides a useful example. The level of peak demand reduction—and the predictability of that reduction—depends on whether the program is implemented through dispatchable control or utility messaging with voluntary, non-dispatchable customer response.

However, even dispatchable programs may incorporate customer override or opt-out provisions. If such provisions are widely exercised by host customers, the predictability and effectiveness of DR programs can be reduced. Prospective BCAs should account for this customer behavior, and proactive measures designed to reduce opt-outs may add to program costs (Navigant 2017; O'Leary 2016).²⁶

In general, the level of customer awareness and engagement will substantially impact the effectiveness of DR programs. Building customer awareness often involves multiple and iterative education campaigns across multiple channels. Participation rates are a function of customer type, awareness, program choices, incentive rates, program complexity, hardware and software interactions and costs, and other factors—all of which could impact program costs and cost-effectiveness.

7.4.2 Technology Operating Profile

The cost-effectiveness of DR programs is a function of the load changes provided during the targeted time period, but also of any load changes in other hours. For example, participating DR customers during a summer peak event may have their air conditioners adjusted by a few degrees, but post-event may cause an unexpected spike in load induced by the DR event. In addition, customers who receive notice of impending super-peak pricing periods, for example, may pre-cool their buildings to ride through such events. It is conceivable that costs related to shifted load may overwhelm benefits of desired responsive behaviors. In every case, the extent to which customers respond and to which they shift load also affects (e.g., through pre-cooling) program cost-effectiveness.

7.4.3 Other Fuel Impacts

In some cases, the host customer response to a DR measure will be to rely on alternatives or substitute energy resources, such as self-generation. The simple case involves operation of a local dispatchable generator, such as a fossil-fueled reciprocating engine. Impacts associated with use of these resources, even if not under utility control, will affect DR program cost-effectiveness from a societal perspective (in terms of pollutants) and the host customer perspective (in terms of costs).

²⁶ Some argue that opt-out opportunities are a net benefit to program effectiveness because they increase enrollment and customer satisfaction.

7.4.4 Host Customer Non-Energy Impacts

Table 7-5 presents a summary of host customer NEIs that DR resources might potentially create. These impacts can sometimes be in the form of benefits and sometimes costs. The presence, direction, and magnitude of these impacts will depend upon many factors, including the DR program, the DR use case, the type of host customer (e.g., low-income, residential, commercial, industrial), and more.

Table 7-5 indicates whether each category of NEIs is typically a cost, a benefit, or either in most common applications for DR resources. The actual direction of the impact, however, may vary and should be determined on a case-by-case basis.

Category	Benefit or Cost	Examples
Transaction costs	•	 Application fees and processes
Asset value	•	 Equipment functionality/performance improvement Increased building value
Productivity	•	 Labor costs and productivity O&M costs
Economic well-being	•	 Fewer bill-related calls to utility Fewer utility shut-offs, reconnects
Comfort	•	Thermal comfortLight quality
Satisfaction/Pride	•	 Contribution to addressing environmental concerns

Table 7-5. Potential Host Customer Non-Energy Impacts of Demand Response

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

7.4.5 Air Emissions Impacts

DR resources can sometimes reduce air emissions by reducing the amount of fossil-fired generation or by shifting loads from hours with high-emission resources on the margin to hours with low-emission resources on the margin.

DR resources can sometimes increase air emissions, for example, when DR resources rely upon fossilfueled back-up power, or when DR resources shift loads from hours with low-emission resources on the margin to hours with high-emission resources on the margin.

As discussed in Chapter 5, estimates of air emission impacts should account for two important factors: (a) the time period for which the DER operates; and (b) the marginal emission rates of the utility system at the time when the DER operates.

The *Ensure Symmetry* principle requires accounting for both reduced and increased air emissions from EE resources, to the extent that they occur. (See Chapter 2.)

7.5 Common Challenges in Determining DR Benefits and Costs

7.5.1 Determining the Operating Profile

When assessing a DR program, the program classification, the type of technology deployed, program purpose, and the mechanisms used all help inform its cost-effectiveness. These component elements will ultimately characterize the operating profile that forms the basis for evaluation of the DR program.

The assessment of the DR program should be based on a detailed understanding of technologies to be used. Modeling of program impacts, program participation rates, and customer behavior over a range of future scenarios can support development of use-informed operating profiles.

7.5.2 Determining the Counterfactual Host Customer Baseline

Counterfactual customer baselines represent what the host customer load patterns would have been in the absence of the DR resource. DR resource savings are determined by comparing counterfactual baselines to the host customer load patterns with the DR in operation. It can sometimes be difficult to determine the counterfactual baseline for estimating DR savings.

Historical customer load patterns can be used to forecast a baseline, but historical customer data is of limited use. For example, customers might make significant changes to their buildings or facilities that affect load patterns; customers might install other DER types that will affect load patterns; or there might be structural shifts in the economy (e.g., a recession) that would affect customer load patterns.

Approaches and considerations to determining baseline usage include:

- Conduct data collection and analysis well in advance of the BCA for determining a host customer counterfactual baseline. This enables collection of baseline customer data and avoids introducing self-selection bias into the data. In addition, program developers can improve the accuracy of counterfactual estimates by undertaking customer research and segmentation analysis. Once programs are established, most DR programs have a large stock of nonparticipating customers that can be studied as a control group for establishing estimates of the counterfactual load patterns. However, self-selection bias should be recognized and mitigated in these control groups.
- Use of a control day or control customer within the program that does not receive an event signal, or ideally use of actual host customer load patterns on non-event days to construct a baseline. These options help eliminate self-selection bias concerns associated with using non-participant control groups.

Evaluation of data and results from peer utilities and programs can also help establish *a priori* estimates of DR savings. In some cases, measuring cost-effectiveness at the portfolio level may be more reliable, even if more general, than results obtained from program-by-program analysis.

7.5.3 Accounting for Provision of Multiple Services through Demand Response Programs

DR programs can provide value for multiple grid services. For example, operation of a DR program could provide either (a) wholesale generation capacity value in an organized market, (b) relief to heavily loaded distribution circuits, or (c) both. Accounting for these multiple impacts requires allocating

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 138 of 302

benefits and costs across the different grid services, similar to the approach for allocating "joint and common" impacts. Accurately evaluating these shared impacts requires determining a reasonable and transparent method for allocating those benefits and costs—one that avoids double-counting or under-counting.

Dispatchable DR measures allow for targeted application to various services. But they also raise tradeoff issues, as with other DERs such as storage. Dispatchable DR measures allow for targeted application to various services. But they also raise tradeoff issues...

Evaluating a DR resource requires attention to which services it can provide and how it will balance potential conflicts that prevent the resource from fully meeting all of its intended use cases (e.g., operation during one part of the day rendering the resource unavailable for another service later that day).

The process for determining the overall impact of DR programs that provide multiple services involves the following steps:

- 1. Evaluate the program and impacts as a whole. All benefits and all costs, wherever incurred or generated should be addressed on a cumulative basis to determine overall cost-effectiveness.
- 2. Select a consistent and reliable method for allocating program costs. Some benefits and costs may be directly allocable. For example, capacity market reservation fees for DR programs should be allocated to the generation market component of program costs.

In other cases, more general allocators may be appropriate. For example, if the implementing utility incurs generation costs that are 25 percent of total distribution revenue requirement and distribution capacity costs that represent 75 percent of those costs, these values can be used as a basis for allocating benefits and costs of the DR program. Ultimately, the allocation method should bear some relationship to costs avoided and incurred.

7.6 Lost Revenues and Rate Impacts

Lost Revenues and Rate Impacts of Demand Response in General

Lost revenues and potential rate impacts should not be included in cost-effectiveness analyses. Instead, DER lost revenues and rate impacts should be analyzed separately using rate, bill, and participation analyses (see Section 2.3 and Appendix A). In conducting BCAs, therefore, lost revenues should be identified so that they can be properly excluded from BCAs and properly included in rate, bill, and participation analyses.

In general, several key factors affect the extent to which DERs might create rate impacts:

- Increases in utility system costs will put upward pressure on rates.
- *Reductions in utility system costs* will put downward pressure on rates.
- *Reductions in sales* from DER resources will put upward pressure on rates.
- Increases in sales from DER resources will put downward pressure on rates.
- *Rate design* will affect the amount of lost or increased revenues created by the DER.

DR resources will typically increase utility system costs because of DR program costs, but they will reduce utility system costs by avoiding generation, transmission, and distribution costs.

DR resources will typically create lost revenues by (a) allowing host customers to reduce their energy bill by reducing, shifting, or shaping their energy consumption; and (b) allowing host customers to reduce their energy bill by reducing demand charge payments, where such charges are in place; and (c) providing host customers with bill credits for reducing demand, e.g., through a peak-time rebate program.

Where DR resources result in load shifts, DR lost revenues should be reduced by revenues generated during the period of increased usage.

Regulators and other stakeholders concerned about DR rate impacts might consider: (a) conducting a rate, bill, and participation impact analysis; (b) seeking DR program designs likely to mitigate equity concerns; (c) examining ways to expand customer participation, especially among low- and moderate-income customers; and (d) periodically reviewing impacts as DR programs are implemented.

Lost Revenues and Rate Impacts of Non-Dispatchable DR

DR programs based on non-dispatchable DR (e.g., time-varying rates) can increase bills for customers who are unable or unwilling to alter consumption during on-peak price periods. This can lead to reduced bills for customers who respond to price signals and increased bills for those who do not. These bill impacts may be the result of more efficient price signals that more accurately allocate costs according to customer consumption patterns, but they may also raise important equity concerns.

These impacts on customers who experience increased bills versus those that experience reduced bills should be assessed in a rate, bill, and participation analysis. In this case, the program participants (i.e., host customers) are those who respond to the non-dispatchable DR, and the non-participants are those who do not respond. Assessing DR program rate impacts, therefore, requires customer segmentation to understand which customers are able and willing to respond the most, and which customers respond the least. Such analyses should be periodically updated. This is because customer willingness and ability to respond to DR signals will likely evolve over time as grid modernization investments increase tools such as customer information portals and appliances with advanced controls become more common.

8. DISTRIBUTED GENERATION RESOURCES

This chapter describes the benefits and costs most relevant to distributed generation (DG) resources. It identifies key factors that affect DG benefits and costs and provides guidance on addressing common challenges with DG cost-effectiveness analyses.

8.1 Summary of Key Points

- Distributed generation includes a range of technology types, including DPV, CHP, distributed wind, distributed hydro, and distributed biomass.
- DG benefits and costs will depend upon whether the DG output is expected to only serve on-site load, only inject from the facility to the grid, or both.
- DG resources can either reduce or increase T&D costs, depending upon the hosting capacity on the utility system. Hosting capacity analyses can inform assessment of the direction and magnitude of these impacts and should account for potential changes to the system over the long run.
- Some DG resources, such as DPV, distributed wind, and distributed hydro, do not typically affect consumption of other fuels. Other DG resources, such as CHP and distributed biomass, can increase consumption of other fuels.
- Some DG resources, such as DPV, distributed wind, and distributed hydro, can reduce air emissions. Other DG resources, such as CHP and distributed biomass, might reduce or increase air emissions, depending upon the fuel used, the fuel avoided, and marginal emissions from the electricity grid.
- While DG resources sometimes require increased costs associated with the electricity they inject onto the grid, they do not create any additional costs associated with back-up services, supplemental services, or the host customer using the grid as if it were a battery.
- DG tariffs can create lost revenues, either by reducing host customer demand or by providing credits that reduce host customer bills. Lost revenues will not exceed what the host customer's bill would have been in the absence of the DG resource.
- Lost revenues from DG resources might lead to increased rates, depending upon the magnitude of lost revenues, the magnitude of DG utility system costs, and the magnitude of utility system benefits. Rate impacts should not be included in BCAs but should instead be accounted for in rate, bill, and participation analyses.

8.2 Introduction

For the purposes of this manual, DG resources include electric generation technologies interconnected to the distribution grid and operating at the distribution level. This generally means DG operates near a load, though some operate as stand-alone resources not close to a particular host customer's load. DG

technologies include DPV, CHP, district heating and cooling, distributed wind, biomass and biogas facilities associated with landfills and agricultural operations, and others.

The fundamental BCA principles described in Chapter 2 can be used to identify the relevant benefits and costs for DG, as with all DER types. In many cases, identifying whether a particular benefit or cost is relevant to DG requires defining the specific DR technology and use case. For example, DG resources that never inject energy to the grid, but instead consume all generation on site, have different impacts from those that do inject energy back onto the grid.

8.3 Benefits and Costs of DG Resources

The tables in this section summarize the full range of potential benefits and costs of DG resources (see Chapter 4 for definitions of impacts). Each impact is described as a benefit, a cost, or either, depending on the most common applications of this technology. There might be some less-common applications where a cost could be a benefit, or vice versa. The tables include notes on applicability that provide further explanation for those impacts that may be either a cost or benefit.

Table 8-1 presents the potential benefits and costs of DG on the electric utility system, while Table 8-2 presents the impacts of DG on the gas utility system. All DG utility system impacts presented in these tables should be included in BCA tests. The remaining DG impacts presented in Table 8-3 (Host Customer) and Table 8-4 (Societal) should be included in BCA tests to the extent they are relevant to the jurisdiction's applicable policy goals (see Chapter 3).

Туре	Utility System Impact	Benefit or Cost	Notes, or Typical Applicability	
	Energy Generation	•		
	Generation Capacity	•		
	Environmental Compliance	•	Typically benefits because DG reduces electricity generation and system peak demands	
Concration	RPS/CES Compliance	•		
Generation	Market Price Response	•		
	Ancillary Services	•	A benefit or a cost, depending upon DG technology and system conditions; magnitude of benefits and costs may change with deployment rate of DG on particular parts of the distribution system	
Transmission	Transmission Capacity	•	A benefit or a cost because DG can increase or decrease transmission peak demand	
	Transmission System Losses	•	A benefit because DG reduces transmission volumes	
Distribution	Distribution Capacity	•	A benefit or a cost because DG can increase or decrease distribution peak demand	
	Distribution System Losses	•	A benefit because DG reduces distribution volumes	
	Distribution O&M	•	May add to or relieve congestion and grid management costs	
	Distribution Voltage	•	A benefit or a cost, depending upon location-specific grid conditions	
	Financial Incentives	•		
	Program Administration Costs	•	A cost to the extent they are relevant	
	Utility Performance Incentives	•		
General	Credit and Collection Costs	•	A benefit because customer savings make bill payment easier, especially for low-income customers	
	Risk	•	DPV reduces some system risks but adds complexity to system operations	
	Reliability	•	Variable DG can support reliability or impose reliability costs at high deployment levels	
	Resilience	•	Grids with DG should be easier to restore; islandable systems or systems with ride-through capacity may have faster restoration and reduced recovery times	

Table 8-1. Potential Impacts of Distributed Generation: Electric Utility System

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

Table 8-2 presents the potential benefits and costs of DG on gas utilities and other fuel systems. Gas utility DG programs might reduce or increase gas consumption, depending upon the facility and the fuel used. Some electric utility DG resources will not increase natural gas or other fuel consumption (e.g., DPV) but others can (e.g., natural gas-fired CHP).

Туре	Non-Electric Energy System Impact	Gas Utility DG Programs	Electric Utility DG Programs that Affect Other Fuels	Notes, or Typical Applicability	
	Fuel and Variable O&M	•	•	DG that uses gas or other fuels	
Other	Capacity	•	•	to generate electricity	
Energy	Environmental Compliance	•	•	increases fuel use and	
2.1.0.87	Market Price Response	•	•	associated impacts	
Other Fuel: General	Financial Incentives (e.g., rebates)	•	0	Typically costs for gas utility DR programs; not relevant	
	Program Administration Costs	•	0		
	Utility Performance Incentives	•	0	otherwise	
	Credit and Collection Costs	•	•		
	Risk	•	•	DG that uses gas or other fuels	
	Reliability	•	•	to generate electricity can create costs for these impacts	
	Resilience	•	•		

Table 8-2. Potential Impacts of Distributed Generation: Gas Utility or Other Fuel System

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

Table 8-3 presents the potential host customer impacts of DG.

	Table 8-3. Potential Im	pacts of Distributed	Generation: Host	Customer
--	-------------------------	----------------------	-------------------------	----------

Туре	Host Customer Impact	Benefit or Cost	Notes, or Typical Applicability
Customer	Measure Costs (customer)	٠	
	Interconnection Fees	•	
	Risk	•	Reduced dependence on utility but may involve technology operating risk
	Reliability	•	Typically not relevant for renewable DG; potentially a benefit for
	Resilience	•	СНР
	Tax Incentives	•	Potentially a benefit, where relevant
	Host Customer NEIs	•	A benefit or a cost depending upon the NEI (see Section 8.4.4)
	Low-income NEIs	•	A benefit for low-income host customers only

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

Table 8-4 presents the potential societal impacts of DG.

Туре	Societal Impact	Benefit or Cost	Notes, or Typical Applicability
Societal	Resilience	•	Typically not relevant for renewable DG; potentially a benefit for CHP
	GHG Emissions	•	Renewable DG will typically reduce system emissions; CHP can also reduce
	Other Environmental	•	system emissions depending upon fuel source and generation displaced
	Economic and Jobs	•	Typically a net benefit
	Public Health	•	Renewable DG will typically offer benefits; CHP can also offer benefits depending upon fuel source and generation displaced
	Low Income: Society	•	A benefit, depending on siting and low-income participation
	Energy Security	•	A benefit for renewable DG; potentially a cost for non-renewable DG, depending on fuel source

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

8.4 Factors that Affect Distributed Generation Benefits and Costs

8.4.1 DG Technology Characteristics

DG output typically has two possible outcomes:

- Generation that occurs during periods of on-site consumption will typically serve that load. In this manual, the term "load reduction" is used in the context of DG to refer to those periods when the DPV generation serves the host customer load.
- Generation that exceeds on-site consumption requirements will be injected onto the grid through the point of the customer's connection to the grid—typically at the meter. In this manual, the term "injection" is used in the context of DG to refer to the instantaneous injection onto the grid, even for very small time periods. Further, the term "excess generation" is used to refer to generation that is not used to offset host customer consumption from the grid over the course of that customer's billing period.²⁷ For example, for a customer with a monthly billing period, the excess generation will be the net effect of all the load reductions and injections that occur over that month.

Over a typical day, month, or year, some DG technologies might alternate frequently between generating for load reduction, injection to the grid, or both.

DG output that occurs during a time when the grid is down is often curtailed in order to prevent unwanted energizing of the grid for safety reasons. DG systems operating in an islandable/microgrid system may continue to serve load during grid outages.²⁸

²⁷ In this manual the term "billing period" refers to the period used for reading customer meters and billing customers for their consumption. This is typically one month, but other options are possible.

²⁸ The Microgrid Institute identifies islandable microgrids as microgrids that are fully interconnected to a local utility grid and are capable of both consuming power from, and supplying power to, the utility grid. They can maintain some level of service during a utility outage. Operators remain tethered to the utility grid and switch seamlessly back and forth, drawing energy when they need it, and selling it back to the utility when they have surplus.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 145 of 302

8.4.2 DG Technology Operating Profile

Characterizing DG benefits and costs should start with an assessment of whether the DG output is expected to serve only on-site load, is only injected from the facility to the grid, or both.

Serve on-site load only: Some DPV systems are sized so as to serve only on-site load and never inject energy onto the grid. The cost-effectiveness analysis for such systems should be generally the same as that for any load-modifying resource, such as energy efficiency.

Inject to the grid only: Some DPV systems might be installed by commercial developers for the purpose of only injecting onto the grid. These DPV system operators sell their output to the local utility, to a direct access customer, or to an ultimate customer through the utility and/or the wholesale power system. The cost-effectiveness analysis for these systems should be generally the same as that for a generator located on the distribution system.

Serve on-site load and inject to the grid: Many DPV systems both serve on-site load and inject energy onto the grid. These systems can simultaneously alternate between these modes instantaneously. This flexibility makes it difficult to distinguish between costs, benefits, and lost revenues. Different DG tariffs treat load reduction versus electricity generation differently; therefore, it is important to account for these mechanisms when identifying benefits, costs, and lost revenues of DPV resources. (See Section 8.5.1.)

8.4.3 Other Fuel Impacts

DG resources can have impacts on other fuel use, depending on the resource type:

- DPV systems typically do not have any impacts on non-electric fuels.
- Distributed wind and hydro resources typically do not have any impacts on non-electric fuels.
- CHP and district energy systems often rely on natural gas or biomass for primary fuel.
- Some DG resources rely upon biomass, biogas, or other fuels.

8.4.4 Host Customer Non-Energy Impacts

Table 8-5 presents a summary of host customer NEIs that might potentially be created by DG resources. These impacts can sometimes be in the form of benefits and sometimes costs. The presence, direction, and magnitude of DG NEIs will depend upon many factors, including the type of host customer (e.g., low-income, residential, commercial, industrial), and more.

Table 8-5 indicates whether each category of NEIs is typically a benefit, a cost, or either in most common applications for DG resources. The actual direction of the impact, however, may vary and should be determined on a case-by-case basis.

Category	Benefit or Cost	Examples
Transaction costs	•	- Application fees and processes, neighborhood association requirements
Asset value	•	- Increased building value
Economic well-being	•	 Fewer bill-related calls to utility Fewer utility shut-offs, reconnects
Customer Empowerment & Control	•	 Psychological benefit in the form of personal empowerment Energy independence
Satisfaction/Pride	•	- Contribution to addressing environmental concerns

Table 8-5. Potential Host Customer Non-Energy Impacts of Distributed Generation

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; O = not relevant for this resource type.

8.4.5 Air Emissions Impacts

Many DG systems installed in the United States are DPV systems, powered by emissions-free solar energy. For these systems, reductions in environmental compliance costs can create benefits to the utility system and reductions in emissions can create benefits to society. For other kinds of DG, such as gas-, biomass-, gasoline- or diesel-powered generators, there may be either increased or reduced air emissions depending upon the DG fuel, the displaced fuel, the DG operating patterns, and more.

The location and effects of air emissions may vary among DG technologies and according to operating profiles. Biomass-fueled DG may avoid upstream utility emissions and, for example, methane emissions from anaerobic digestion of wastes. However, these systems typically generate electricity through combustion of the biomass fuel, producing local emissions that have local incremental impacts.

CHP systems that operate efficiently on methane gas may also avoid air emissions by displacing largescale generation. While CHP systems produce local emissions, the net system-wide efficiency and emissions reductions benefits may yield a net benefit for this type of DG.

The *Ensure Symmetry* principle requires accounting for both avoided and incremental emissions in order to accurately account for the emissions impacts of DG systems that produce emissions.

8.4.6 Cumulative Effects: The Duck Curve

Unless coupled with energy storage systems, all PV systems in a particular area will start and stop generating in correlation with available sunlight. Large quantities of PV, including both utility-scale and DPV, can therefore have large cumulative impacts on the grid associated with the decline of solar generation as the sun sets. This often occurs at the time when residential system demand is in or approaching the period of peak demand and can create a rapid rise in system-wide energy demand and an urgent need for load-following resources. The effect, which when plotted on a graph appears like the silhouette of a duck, may be a source of increased demand for generation, capacity, and ancillary services. Even if the effect is not significant across an entire grid system, clustering of DPV systems could create the problem at smaller scales.

The duck curve effect of PV systems might impose other costs on the electricity system to respond to the rapid rise of resources needed to meet the declining solar resource. If this effect is expected to be large enough to make a material difference in the BCA results, then these additional costs should be

accounted for in the BCA. Any such estimates of these costs should account for other resources that are likely to be available mitigate the costs, including DR, storage, and resources providing ancillary services.

8.5 Common Challenges in Determining DG Benefits and Costs

8.5.1 Implications of DG Tariffs on Lost Revenues

Many jurisdictions have created DG tariffs that determine how a host customer will be compensated for DG output. These DG tariffs are designed in many different ways, and the specific design of a DG tariff will affect the amount of lost revenues created by the DG resource.

The diversity and complexity of DG tariffs raise two important BCA questions. To what extent does the DG tariff used in the jurisdiction create lost revenues? To what extent does the DG tariff used in the jurisdiction create utility system costs?²⁹

DG Tariff Design Options

For BCA purposes it is useful to describe DG tariffs as falling into two categories:

- Net billing approaches, where a portion of the DG output is used to offset the host customer electricity consumption, while the remaining output (i.e., the excess generation) is compensated at a specific compensation rate (NREL 2017a): This approach can be applied using traditional meters and therefore has been widely used to date. Net billing approaches can include traditional net metering, where the compensation rate for the excess generation is equal to the host customer retail rate. They can also include alternative rates to compensate excess generation.³⁰
- *Buy-all/sell-all* approaches, where all the DG output is compensated at a certain rate (PNNL 2018): In these cases, the DG output does not offset the host customer's electricity load. This approach requires meters that measure the output of the DG.

Both net billing and buy-all/sell-all mechanisms can offer a variety of compensation rates. The compensation rates can equal the host customer's retail rates (in the case of net metering); they can equal utility system avoided costs; they can equal a rate that is designed to cover all or some of the cost of installing and operating the DG; or they can be based on something else. Further, both net billing and buy-all/sell-all options can be used for community or virtual solar projects.³¹

²⁹ Any costs created by DG resources should be included in a DG BCA, but lost revenues, and the rate impacts that they create, should be addressed separately in a rate, bill, and participation analysis. (See Section 2.3 and Appendix A.)

³⁰ Some studies describe net metering as a separate category from net billing (PNNL 2018). In this manual the term "net billing" is used to describe net metering as well as net billing. Both types of tariffs have the same general design where a portion of the DG output is used to offset the host customer electricity load, and the excess generation is compensated separately. Net billing and net metering typically differ (a) in the compensation rate and (b) in the period when customer consumption and DG output are netted.

³¹ Community or virtual solar projects are when a solar facility is not installed on a host customer premises but is instead installed externally and shared among subscribers. Subscribers typically receive credits on their electric bill for energy produced by their share of the solar facility.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 148 of 302

For both net billing and buy-all/sell-all tariffs, participating customers are typically offered bill credits as compensation for the DG output. In the case of net billing, the bill credits apply to only the excess generation whereas for buy-all/sell-all, the bill credits apply to all of the DG output.

One of the key design features of net billing tariffs is the treatment of excess generation. Excess generation is typically determined over the billing period because net billing tariffs rely upon traditional meters that cannot

distinguish excess generation at a more detailed level. In most net billing tariffs, any excess generation in one billing period is used to create a bill credit to offset the host customer bill in the following billing period. These bill credits can roll forward from one billing period to the next until they are used up. Many jurisdictions place a limit on how long the bill credits can be rolled forward (i.e., they apply a netting period over which the excess generation will be compensated).³²

Billing credits remaining at the end of the netting period can be treated different ways. For example, sometimes the host customer can choose to provide extra DG bill credits to other utility customers; sometimes the value of remaining bill credits is paid directly to host customers; and sometimes the remaining bill credits are simply retired (i.e., they are not used for any purpose).

Identifying Lost Revenues and Utility System Costs

A few key considerations can help identify the extent to which DG tariffs might create lost revenues or utility system costs.

First, lost revenues are created two ways. Lost revenues are created when the host customer's demand is met by the DG output, thereby reducing the net amount of electricity that the customer consumes and is billed for. Lost revenues can also be created from the bill credits generated as a result of the DG tariff. These bill credits allow host customers to pay lower bills in just the same way that reduced consumption does. Since lost revenues can be created from either reduced host customer electricity load or from the bill credits, it is not necessary to determine the extent to which a DG resource offsets a customer's load or exports electricity to the grid in order to estimate lost revenues.

Second, lost revenues should be measured relative to the total electricity bill that the host customer would have experienced if not for the DG output (i.e., the counterfactual host customer bill). It is best to determine the counterfactual bill over the full netting period because this will capture the effects of rolling credits across billing periods. Further, lost revenues can never exceed the host customer's counterfactual bill over the full netting period. In other words, revenues can only be considered lost if they would have existed in the counterfactual case.

Third, utility system costs are created only if there is some increase in utility revenue requirements. If a DG tariff results in an increase in utility revenue requirements, then there will be a utility system cost resulting from that tariff. If not, then there will not be any costs created by the DG tariff.³³

Fourth, if the total compensation paid for the DG output exceeds the host customer counterfactual bill, i.e., exceeds the lost revenues, then the difference might create a new utility system cost. Whether this difference

Utility system costs are created only if there is some increase in utility revenue requirements.

One of the key design features of net billing tariffs is the treatment of excess generation.

³² In this manual, the term "netting period" refers to the period over which customers are allowed to carry forward their billing credits. This is typically one year, but other options are possible.

³³ This point refers only to utility system costs created by the DG tariff compensation mechanism. Other utility system costs, such as those indicated in Table 8-1, can be created by the DG resource.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 149 of 302

creates a new utility system cost depends upon how the DG bill credits are treated at the end of the netting period, as described below.

These considerations lead to the following conclusions:

- Lost revenues from DG resources will equal the amount of DG output times the compensation rate for the DG output.
 - For buy-all/sell-all DG tariffs, the compensation rate for this purpose is simply the compensation rate offered for the DG output and used to create bill credits.
 - For net billing DG tariffs, the compensation rate for this purpose has two components. The DG output that is used to reduce the host customer load is compensated at the host customer retail rate, while the excess DG output is compensated at the excess generation compensation rate. For net metering these two compensation rates will be the same, but for other forms of net billing they will be different.
- Lost revenues from DG resources will not exceed the host customer counterfactual bills over the DG tariff netting period.
- If a DG resource is expected to be compensated at an amount that exceeds the lost revenues, then the implications of this outcome will depend upon how any such difference is treated by the utility at the end of the netting period:
 - If the host customer is allowed to provide any excess bill credits to other utility customers, then those bill credits will create lost revenues by reducing the recipient's bills.
 - If the value of the remaining bill credits is provided to the host customer in the form of a payment, then this payment represents a new incremental utility system cost.
 - If the bill credits remaining at the end of the netting period are simply retired, then they will not create lost revenues or utility system costs.

To date, few jurisdictions have established DG tariffs that allow host customers to be paid for bill credits that exceed their counterfactual bill. Thus, there are few circumstances where a DG tariff would create a utility system cost. In most cases, the lost revenues will be equal to the DG output times the compensation rate(s) for that output, with a cap equal to the magnitude of the host customer's counterfactual bill.

8.5.2 Accounting for Transmission and Distribution Impacts

DG resources have the potential to reduce T&D costs because they are sited close to customer load and result in less electricity being transmitted from centralized generation sources to the host customer. On the other hand, some DG resources have the potential to increase T&D costs when the level of interconnected DG exceeds the hosting capacity of the system.

In addition, any single DG facility is not likely to have much impact on T&D costs on its own, particularly small DG facilities, and particularly when DG facilities are installed on systems with excess T&D capacity. However, when multiple DG facilities are installed over time, they might have significant positive or negative effects on T&D costs. In some utility systems, new DG resources might have very little impact

All T&D impacts created by DG resources should be accounted for in cost-effectiveness analyses. on T&D costs for several years, but then the system might reach a tipping point where system upgrade costs become required.

These factors create challenges for determining the magnitude and the direction of DG resources' impacts on the T&D system.

All T&D impacts created by DG resources should be accounted for in cost-effectiveness analyses, as with any DER type (see Section 2.3 and Section 3.3). The magnitude and direction of these impacts can vary across utility systems, depending upon the hosting capacity on the system and any changes to hosting capacity over time. Therefore, the utility hosting capacity should be considered when determining whether new DG resources will likely reduce or increase distribution system costs. The likelihood of such impacts will be greater on circuits that have been identified as approaching maximum capacity or that are experiencing unusually high load growth. Marginal cost of service studies can also be used as a preliminary tool for assessing impacts. In addition, tracking patterns of DG development may help prioritize hosting capacity analyses.

BCAs should include study periods that are long enough to capture the full long-run costs and benefits of the DERs, consistent with the *Conduct Forward-Looking, Long-term, Incremental Analyses* principle (see Section 2.3). This may require study periods that are 20 or 30 years long. Study periods this long allow for an assessment of the long-run impacts on the utility system hosting capacity. If the hosting capacity is expected to be sufficient in the short run but insufficient in the long run, then that expectation should be factored into the determination of the DG effects on distribution system costs.

Hosting capacity analyses should account for estimated deployment levels of DG resources over time. They should also account for estimated deployment levels of other DER types, as well as other factors such as load growth and distribution system upgrades that would occur regardless of DG deployment.

8.5.3 Accounting for Grid Services to the Host Customer

Under current U.S. federal law and regulation, substantially replicated and applicable in most U.S. states, utilities are obligated to provide back-up and supplementary power services and rates to certain qualified facilities. This includes distributed qualified facilities. This raises the question of whether the back-up or supplementary power services provided to DG host customers represents an impact that should be included in a BCA.

Similarly, when DG resources inject electricity onto the grid it raises the question of whether host customers benefit from using the grid essentially as a battery to allow them to take advantage of their excess generation, and whether this effect represents an impact that should be included in a BCA.

All distribution upgrade costs created by DG resources, or any DER, should be accounted for in costeffectiveness analyses (see Section 2.3 and Section 3.3). Regarding the two questions raised above:

- If the utility has to incur incremental costs in order to enable a customer to inject DG generation onto the grid, then those costs should be accounted for in the interconnection or distribution system upgrade costs.
- If the DG injections do not require any such incremental costs, then there are no costs associated with back-up services, supplemental services, or the host customer using the grid as if it were a battery.

8.5.4 Community Solar Projects

Community solar projects allow customers to subscribe to a designated share of the output of a solar system sited remotely from their premises. Community solar projects offer the advantage of allowing customers that do not have adequate facilities for siting DPV to benefit from DPV projects. They can be especially useful in supporting low-income customer participation in DPV.

The Absence of a Host Customer Load

Community solar projects differ from other types of DPV because there is no host customer whose load would be reduced by the community solar projects. This raises the question of how the lack of a host customer load affects the lost revenues and utility system costs created by the community solar DG tariff. If there is no host customer load to offset, does the community solar project create lost revenues for the utility?

Community solar projects typically create bill credits that are provided to the participants in the program. These credits are then used to reduce the participants' bills. These reduced bills result in lost utility revenues in the same way that reduced bills from reduced customer load create lost revenues. Therefore, the lost revenues from community solar projects will be equal to the project's output times the community solar compensation rate. These lost revenues should not be included in a BCA but should be accounted for in rate, bill, and participation analyses. Community solar projects will not create utility system costs as a result of the DG tariff.³⁴ (See Section 8.5.1.)

Avoided Transmission and Distribution Costs

Unlike on-site DPV, remote solar projects use the grid to transmit the electricity they produce. This raises the question of whether they can avoid the transmission or distribution costs that on-site DPV facilities can avoid.

DERs can either increase or decrease utility system infrastructure costs, depending upon the loading characteristics and hosting capacity available on the distribution system. As with any type of DER, community solar projects should account for the distribution system hosting capacity in order to determine the actual impact the facility will have on the grid. (See Section 4.2.)

Some DERs will have different impacts depending upon the location of the DER on the distribution grid, and BCAs should attempt to account for this locational value wherever it is likely to have a material effect on the BCA results. There may be some locations where community solar projects increase infrastructure costs and some locations where it decreases them. (See Section 5.3.)

8.6 Lost Revenues and Rate Impacts

Lost revenues and potential rate impacts should not be included in cost-effectiveness analyses. Instead, DER lost revenues and rate impacts should be analyzed separately using rate, bill, and participation analyses. (See Section 2.3 and Appendix A.) In conducting BCAs, therefore, lost revenues should be identified so that they can be properly excluded from BCAs and properly included in rate, bill, and participation analyses.

In general, several key factors affect the extent to which DERs might create rate impacts:

³⁴ Community solar projects can create other utility system costs, such as those presented in Table 8-1.

- Increases in utility system costs will put upward pressure on rates.
- *Reductions in utility system costs* will put downward pressure on rates.
- Reductions in sales from DER resources will put upward pressure on rates.³⁵
- Increases in sales from DER resources will put downward pressure on rates.
- *Rate design* will affect the amount of lost or increased revenues created by the DER.

DG resources will typically increase utility system costs because of grid upgrade costs or program administration costs, but will reduce utility system costs by avoiding generation, transmission, and distribution costs (see Table 8-1).

DG resources will typically create lost revenues by (a) allowing host customers to reduce their energy bill by reducing their energy consumption; (b) allowing host customers to reduce their energy bill by reducing demand charge payments, where such charges are in place; and (c) providing host customers with bill credits for excess DG output. The lost revenues from DG resources will depend upon the DG tariff in effect, among other factors. (See Section 8.5.1.)

Regulators and other stakeholders that are concerned about DG rate impacts could consider the following: (a) conduct a long-term rate, bill, and participation impact analysis; (b) seek and implement DG tariffs likely to mitigate equity concerns (e.g., by keeping compensation rates relatively low); (c) seek ways to expand customer participation through programs like community solar projects, especially among low- and moderate-income customers; and (d) periodically review impacts as DG deployment rates increase. (Consumers Union 2016.)

³⁵ Some DG tariffs create a different dynamic regarding rate impacts. Some DG tariffs provide customers with bill credits for the DG output, which can be rolled over across billing periods or be used to support virtual or community solar programs. These bill credits are sometimes a better indication of DG lost revenues than reduced sales from the DG resource. (See Chapter 8.)

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 153 of 302

9. DISTRIBUTED STORAGE RESOURCES

This chapter describes the benefits and costs most relevant to distributed storage (DS). It identifies key factors that affect distributed storage benefits and costs and provides guidance on addressing common challenges with distributed storage cost-effectiveness analyses.

9.1 Summary of Key Points

- Distributed storage resources include lithium-ion batteries, lead-acid batteries, and thermal storage such as electric water heaters.
- Perhaps more than any other DER type, the benefits and costs of distributed storage resources will depend significantly on the use case of the storage device.
- The use case for a storage device will depend upon many things, such as the technology charging and discharging capabilities, the host customer rate design, whether the operation is controlled by the host customer versus the utility, and more.
- Like other DERs, the ability of distributed storage resources to "stack" multiple benefits can significantly affect its benefits and costs. However, evolving wholesale electricity market rules combined with the inability of distributed storage to sustain an indefinite charge/discharge requires consideration of potential tradeoffs between different impacts.
- Distributed storage resources can either reduce or increase air emissions, depending upon the use case for the storage device, the marginal emission rates during charging times, the marginal emission rates during the discharging time, and the round-trip efficiency of the storage resource.
- The benefits of distributed storage will depend upon how it will balance potential conflicts that prevent it from fully meeting all of its intended use cases (e.g., operation during one part of the day rendering the resource unavailable for another service later that day).
- Distributed storage resources will typically create lost revenues as a result of the ultimate bill savings to host customers. These might lead to increased rates, depending upon the magnitude of lost revenues, the magnitude of distributed utility system costs, and the magnitude of utility system benefits. Rate impacts should not be included in BCAs, but should instead be accounted for in rate, bill, and participation analyses.

9.2 Introduction

For the purposes of this manual, distributed storage resources include lithium-ion batteries , lead-acid batteries, and thermal storage such as electric water heaters.³⁶ Storage technologies like pumped hydro

³⁶ EVs can also act as a storage resource. These are addressed in Chapter 10.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 154 of 302

and compressed air tend to be connected at the transmission, not the distribution, level and are therefore not considered distributed storage.

It is imperative to define the storage device use case, i.e. how it will be implemented and operated, to determine the relevant benefits and costs. Any cost-effectiveness evaluation of distributed storage should consider both the advantages it provides (e.g., dispatchability) and some of the inherent limitations of the technology (e.g., an inability to indefinitely charge/discharge energy; the losses involved in roundtrip efficiency; etc.).

There are many factors affecting a storage resource's cost-effectiveness, including resource location, resource ownership and control, the

It is imperative to define the use case i.e. how the resource will be implemented and operated—for the distributed storage resource to determine the relevant benefits and costs.

resource's expected operating profile, and considerations around interconnection. Taken together, these factors will influence not only which impacts are relevant, but also whether they amount to a benefit or cost.

This chapter primarily focuses on distributed storage resources that are located behind a host customer's meter (i.e. coupled with load). One key difference between BTM and FOM storage is that, in general, BTM storage can provide a greater number of distinct benefits to host customers than FOM storage. For example, FOM storage resources cannot provide bill management (e.g., demand charge reduction or TOU rate arbitrage) benefits for customers (see Chapter 5). Another key difference is that FOM storage may offer greater economies of scale than BTM storage.

9.3 Benefits and Costs of Distributed Storage Resources

The tables in this section summarize the full range of potential benefits and costs of distributed storage resources (see Chapter 4 for definitions of impacts). Each impact is described as a benefit, a cost, or either, depending on the most common applications of this technology. There might be some less-common applications where a cost could be a benefit, or vice versa. The tables include notes on applicability that provide further explanation for those impacts that may be either a cost or benefit.

Table 9-1 presents the potential benefits and costs of storage on the electric utility system, while Table 9-2 presents the impacts of storage on the gas utility system. All storage utility system impacts presented in these tables should be included in BCA tests. The remaining storage impacts presented in Table 9-3 (Host Customer) and Table 9-4 (Societal) should be included in BCA tests to the extent they are relevant to the jurisdiction's applicable policy goals (see Chapter 3).

Туре	Utility System Impact	Benefit or Cost	Notes, or Typical Applicability		
	Energy Generation	•	A cost because storage technologies generally require more energy to charge than what they discharge		
	Generation Capacity	•	A benefit, depending upon the storage use case and the electric utility's ability to affect the operation of the storage device; otherwise, a cost if storage device charges during peak periods		
Generation	Environmental Compliance	•	A benefit or cost depending upon system environmental profile during charging and discharging times		
	RPS/CES Compliance	•	A cost because storage technologies generally require more energy to charge than what they discharge		
	Market Price Response	•	A benefit or cost depending upon market conditions during charging and discharging times		
	Ancillary Services	•	A benefit or cost depending upon the storage use case and the electric utility's ability to affect the operation of the storage device		
Transmission	Transmission Capacity	•	Potentially benefits depending upon the storage use case and the electric utility's ability to affect the operation of the storage device:		
	Transmission Line Losses	•	otherwise, potentially costs if storage device charges during transmission peak periods		
	Distribution Capacity	•			
Distribution	Distribution Line Losses	•	Potentially benefits depending upon the storage use case and the electric utility's ability to affect the operation of the storage device;		
Distribution	Distribution O&M	•	otherwise, potentially costs if storage device charges during distribution peak periods		
	Distribution Voltage	•			
	Financial Incentives	•			
	Program Administration Costs	•	Typically costs to the extent they are relevant		
	Utility Performance Incentives	•			
General	Credit and Collection Costs	•	A benefit because customer savings make bill payment easier, especially for low-income customers		
	Risk	•	Potentially benefits depending upon the storage use case and the		
	Reliability	•	electric utility's ability to affect the operation of the storage		
	Resilience	•	technology during peak or emergency periods		

Table 9-1. Potential	Impacts of D	Distributed Storage:	Electric Utility	System
		0		

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

Table 9-2 presents the potential impacts of distributed storage on gas utility or other fuel systems. It illustrates that (a) there are no distributed storage resources for natural gas utilities, and (b) electric utility storage resources do not affect other fuels.³⁷

³⁷ This table is included for consistency with the other tables in Part III.

Туре	Non-Electric Energy System Impact	Gas Utility Storage Resources	Electric Utility Storage Resources that Affect Other Fuels	
	Fuel and Variable O&M			
Other	Capacity			
Energy	Environmental Compliance			
LIICIBY	Market Price Response			
	Financial Incentives (e.g., rebates)	Not applicable.	Not applicable. Electric utility storage resources do	
	Program Administration Costs	There are no distributed storage		
Other	Utility Performance Incentives	resources for natural gas utilities		
Fuel: General	Credit and Collection Costs	resources for natural gas atimites.		
	Risk			
	Reliability			
	Resilience			

Table 9-2. Potential Impacts of Distributed Storage: Gas Utility or Other Fuel System

Table 9-3 presents the potential host customer impacts of distributed storage (see Section 4.4).

Туре	Host Customer Impact	Benefit or Cost	Notes, or Applicability	
	Host portion of DER costs	•	Costs to the output those suist	
Host R Customer R T H	Interconnection Fees	•	Costs to the extent they exist	
	Risk	•		
	Reliability	•	Benefits depending upon the storage use case and the extent	
	Resilience	•		
	Tax Incentives	•	Potentially a benefit for those resources where they apply	
	Host Customer NEIs	•	A benefit or cost depending on the NEI (see Section 9.4.4)	
	Low-Income NEIs	•	A benefit for low-income storage only	

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

Table 9-4 presents the potential societal impacts of distributed storage.

Туре	Societal Impact	Benefit or Cost	Notes, or Applicability
	Resilience	•	Potentially a benefit depending upon the storage use case and the extent to which critical customers control the device
Societal	GHG Emissions	•	Benefits or costs depending upon system environmental profile during
	Other Environmental	•	charging and discharging times
	Economic and Jobs	•	Potentially a net benefit or net cost depending on use case and net utility system benefits
	Public Health	•	Same as GHG Emissions and Other Environmental
	Low Income: Society	•	Typically a benefit depending on siting and low-income participation
	Energy Security	•	Potentially a benefit or cost depending on use case and net utility system benefits

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

9.4 Key Factors that Affect Distributed Storage Impacts

Determining a distributed storage resource's cost-effectiveness depends upon a clear identification of the use case(s) it intends to target. Although some impacts from distributed storage are clearly a benefit or cost, there are numerous impacts where the net effect varies from project to project. Table 9-5 provides an example of analyzing potential benefits of energy storage based on a BCA of a range of possible use cases.

Use Case		Estimated Share of 1766 MW Recommendation		Millions \$		Benefit/Cost
		%	MW	Combined Benefits (Market Revenue + System Benefits)	Cost	Ratio
Investor-Owned Distributed Stor	Utility Grid Mod Asset: age at Utility Substations	40%	707	1301	387	3.36
Municipal Light	Plant Asset	10%	177	446	97	4.60
Load Serving Entity/Competitive Electricity Supplier Portfolio Optimization		8%	141	158	77	2.05
	C&I Solar + Storage	6%	106	103	58	1.78
Behind the Meter	Residential Storage	4%	71	19	53	0.49
	Residential Storage Dispatched by Utility	5.5%	96	129	39	2.43
Merchant	Alternative Technology Regulation Resource	1.5%	28	45	15	3.00
	Storage + Solar	10.5%	185	373	102	3.66
	Stand-alone Storage or Co- Located with Traditional Generation Plant	9.5%	168	405	92	4.40
Resilience/Microgrid		5%	87	133	48	2.77

Table 9-5. Cost-Effectiveness Analysis of Storage Use Cases

Source: State of Massachusetts 2017.

The box below provides three illustrative examples of how the relevant impacts for a distributed storage resource might vary between different use cases.

Illustrative Distributed Storage Use Case Tradeoffs

Example 1: The storage resource seeks to provide multiple host customer benefits in the form of reliability, resilience, and bill management. For all three of these benefits, the storage resource must maintain a certain state of charge (i.e. how full the storage resource is) to be able to provide the services when needed. If, however, there is a forecasted system reliability or resilience event that requires the storage resource to remain fully charged, then the resource would not be capable of simultaneously discharging to manage the customer's bill.

Example 2: The storage resource seeks to provide both system generation and distribution capacity, but there will need to be a clear understanding of relative priority of these services for instances when the resource cannot provide both in full. For example, if the distribution capacity need is from 12:00–4:00 PM while the generation capacity need is from 4:00–8:00 PM, a storage resource with four hours of duration would not be able to fully meet both obligations (It would have insufficient time to recharge the resource between providing each service). A BCA will need to contemplate what mechanisms exist to determine which service the storage resource prioritizes in the event of a conflicting obligation.

Example 3: The storage resource seeks to provide system frequency regulation (an ancillary service) to the bulk power system and distribution deferral. Since frequency regulation requires a resource to be able to either inject or withdraw, the ability for the storage resource to provide both services depends on the timing and direction of the needed responses. For example, if there is a temporal overlap of the needs (e.g., 4:00–6:00 PM) and the frequency regulation dispatch requires the storage resource to charge while the distribution deferral dispatch requires the storage resource to discharge, then the storage resource will not be able to simultaneously provide both services using the same capacity.

9.4.1 Technology Characteristics

The specific technical characteristics (largely based on the type and chemistry) of a distributed storage resource may affect its ability to provide certain services. For example, some distributed storage assets may have durations (the amount of time the asset can sustain a continuous charge or discharge at a fixed level) that are less than the minimum requirement to provide a service such as generation capacity through the organized wholesale market.³⁸ In addition, the specific technology may affect other eligibility requirements the distributed storage resource must meet to provide certain services, such as being synchronized to the electricity system. As a result, any BCA should consider how the technical characteristics of a distributed storage resource affect which impacts are relevant, including the relative scale of each.

9.4.2 Technology Operating Profile

Distributed storage can operate in several different modes, including electricity withdrawal (i.e., consumption), electricity injection, and load modulation. Understanding distributed storage impacts requires an understanding of when the distributed storage asset plans to operate in each mode, as well as an articulation of the specific benefits and costs attributable to each mode. For example, a distributed storage resource would likely exhibit load-shifting behavior in order to provide a service like distribution

³⁸ To meet minimum run-time requirements, however, a distributed storage resource could lower its discharge amount (kW) to increase its effective duration.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 160 of 302

capacity but might instead operate to modulate load (i.e., real-time fluctuations) when providing ancillary services that require a more dynamic response.

While these operational modes allow distributed storage to provide additional value in the form of flexibility, they will also affect the useful lifetime of the resource. For example, as the number of cycles (switching from charging to discharging) and depth of discharge (the ratio of how much the storage resource is charged and discharged relative to its full capability) increase, the expected lifetime of the storage asset will decrease. Consequently, a BCA for distributed storage should consider the expected useful lifetime of the resource as that will affect how far into the future the BCA should account for relevant impacts. Understanding distributed storage impacts requires an understanding of when the distributed storage asset plans to operate in each mode, as well as an articulation of the specific benefits and costs attributable to each mode.

In addition, a BCA should consider the fact that distributed storage resources have duration limitations, since they cannot indefinitely charge or discharge. These resources also have less than 100 percent roundtrip efficiency, as some energy is lost between the resource charging and discharging. These characteristics make it imperative to understand both how and when the distributed storage resource will operate because the benefits provided and costs incurred during the times of operation will affect the cost-effectiveness of the resource. For example, if the value for discharge is sufficiently higher than the cost incurred to charge, then there may be sufficient energy arbitrage opportunity despite losing energy in the process. This example highlights the importance of understanding the time-varying nature of the distributed storage resource's operation and associated benefits and costs. These benefits and costs are very dependent on the host customer's rate structure.

9.4.3 Other Fuel Impacts

Distributed storage—in isolation from other DER types—will not have any impacts on natural gas or other fuels because (a) there are currently no distributed storage resources for natural gas utilities, and (b) electric utility storage resources do not affect other fuels.

Distributed storage resources might have impacts on natural gas or other fuels when combined with electrification DERs. In these cases, storage could help alleviate electric system spikes, potentially paving the way for more economic electrification. Any such impacts on natural gas supplies and emissions should be evaluated on a case-by-case basis.

9.4.4 Host Customer Non-Energy Impacts

Table 9-6 presents a summary of host customer NEIs that distributed storage resources might potentially create. These impacts can sometimes be in the form of benefits and sometimes costs. The presence, direction, and magnitude of distributed storage NEIs will depend upon many factors, including the use case for the storage, the type of host customer (low-income, residential, commercial, industrial), the host customer's needs, and the rate design in place.

Table 9-6 indicates whether the category of NEIs is typically a cost, a benefit, or either in most common applications. The actual direction of the impact, however, may vary and should be determined on a case-by-case basis.

Category	Benefit or Cost	Examples
Transaction costs	•	 Researching technologies, application fees and processes
Asset value	•	- Increased building value
Productivity	•	 Productivity gains from consistent power source Loss of productivity due to non-host customer obligation*
Economic well-being	•	 Stable operating costs Resource may add to customer debt profile
Comfort	•	 Thermal comfort may be positively or negatively impacted Noise pollution from distributed storage resource
Health & safety	•	 Safety concerns associated with operation of storage device Provide electricity to critical health devices during outage
Empowerment & Control	•	 Psychological benefit in the form of personal empowerment Energy independence Inconvenience of non-host customer obligations
Satisfaction/Pride	•	- Contribution to addressing environmental concerns

Table 9-6. Potential Host Customer Non-Energy Impacts of Distributed Storage

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource.

9.4.5 Air Emissions Impacts

Distributed storage resources can either reduce or increase air emissions, depending upon the use case for the storage device, the marginal emission rates during charging times, the marginal emission rates during the discharging time, and the round-trip efficiency of the storage resource.

As discussed in Chapter 5, air emission impacts should account for two important factors: (a) the time period for which the DER operates; and (b) the marginal emission rates of the utility system at the time when the DER operates. Distributed storage devices can potentially have significant intra- and inter-day variation, making it especially important to account for the detailed time periods that they are expected to operate.

The *Ensure Symmetry* principle requires accounting for both reduced and increased air emissions from EE resources, to the extent that they occur.

9.4.6 Interconnection Location and Process

The cost- effectiveness of a distributed storage resource fundamentally depends on its interconnection parameters, both in terms of resource location and the process for gaining utility approval to be built.

Interconnection location

Like other DERs, where distributed storage is interconnected will affect its cost-effectiveness. First, the set of relevant impacts for a distributed storage resource depends on whether the resource is interconnected in front of or behind the customer's meter. For example, if a distributed storage resource is interconnected in front of the customer's meter, then it is incapable of providing the host customer with reliability or resilience benefits. (See Section 5.5.)

Second, and related to the first, the distributed storage resource may be eligible for certain financial incentives (e.g., the federal Investment Tax Credit, or ITC) if it is co-located with solar PV and a certain amount of the storage resource's charging is supplied by the solar PV system.

Finally, a distributed storage resource's benefits will depend on where utility system needs are in relation to the resource's interconnection location. For example, if a distributed storage resource is in an area where there are no distribution capacity needs, then even if the resource has the technical capability to provide that service, it should not be credited with that benefit. (See Section 5.3.)

Interconnection process

There are three key aspects of the interconnection process that could affect a distributed storage resource's cost-effectiveness:

- Assumptions regarding the storage resource's intended operation could impact interconnection costs and hosting capacity limits.
- Assumptions regarding the storage resource's ability to inject energy onto the grid, which could impact its ability to provide certain ancillary services that require injections of energy rather than withdrawal.
- Assumptions regarding how the interconnection process allows distributed storage to utilize unused interconnection rights of an existing resource (e.g., solar PV, assuming all engineering concerns are met), which could lead to lower costs by virtue of a more streamlined interconnection process. For example, if an existing solar PV system has 5 MW of injection rights from the utility, a co-located storage resource could install a control system that ensures the total aggregate injection of the storage and solar PV resources never exceeds 5 MW.

9.4.7 Compensation Mechanisms

There are two key factors affecting distributed storage compensation mechanisms: market rules and rate design.

Market rules

Market rules—which can broadly be applied to both organized wholesale markets and any utility programs or procurements—determine how distributed storage resources capture value for responding to price and dispatch signals to deliver system benefits. For wholesale markets, FERC Order 841 (FERC 2018) requires all ISOs and RTOs to develop energy storage participation models allowing this resource type (including FOM and BTM storage) to provide and be compensated for all wholesale market services they are technically capable of providing. This order creates additional opportunities for distributed storage to capture wholesale market value.

Separately, utility programs and procurements can provide distributed storage resources with compensation to deliver local value. An important consideration for these types of programs and procurements is whether they allow the distributed storage resource to seek alternative sources of revenue when not being utilized by the utility.

Rate design

When located behind the customer's meter (i.e., co-located with load), the applicable retail rate will likely affect the storage device's use case, which will affect its benefits and costs. For example, time-

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 163 of 302

varying rate structures and demand charges provide price signals that could encourage charging when rates are lower and discharging when rates are higher, or by reducing the customer's peak demand.

As more jurisdictions adopt sophisticated rate structures, it will become increasingly important to assess how those rate structures inform a storage resource's operational profile, and consequently its impact on cost recovery for the utility. For example, if a jurisdiction has a TOU rate that charges higher prices for electricity consumption during peak hours, there may be a growing number of storage resources operated to minimize bill impacts from on-peak rates. If rates do not accurately reflect costs on the system, the use of storage for TOU rate arbitrage could have negative impacts on system costs and electricity rates. In general, utilities may have greater levels of confidence from a planning and operations perspective if they directly own and control a resource.

9.4.8 Resource Ownership and Control

The entity that owns and controls distributed storage may affect the resource's cost-effectiveness. In some cases the entity owning the storage might be different from the entity controlling it.

In terms of ownership, the primary effect on cost-effectiveness will be the allocation of various benefits or costs. For example, while a utility-owned distributed storage resource may entail the utility bearing all integration costs and accruing all benefits, a customer-owned distributed storage resource may involve some level of benefit and cost sharing.

Distributed storage control may also affect which impacts are relevant for the resource. In general, utilities may have greater levels of confidence from a planning and operations perspective if they directly own and control a resource. In these instances, the utility will have a clear line of sight into which services it seeks the distributed storage resource to provide and can control it accordingly. However, if control resides with the customer or another third party, the utility must rely on market-based mechanisms (e.g., dynamic pricing; competitive procurements; etc.) to elicit the desired response from the distributed storage resource.

This dynamic informs cost-effectiveness in two key regards. First, the level of control the utility requires on the distributed storage resource will affect the overall cost because more advanced control technologies may be more expensive. Second, if the utility does not directly control the distributed storage resource, it may assign the resource a decremented value to "compensate" for the reduced level of utility confidence that the resource will deliver the targeted service(s), which could lead to unnecessary investments in additional resources as a result.

9.5 Common Challenges for Determining Storage Benefits and Costs

9.5.1 Determining the Operating Profile

As described earlier, as part of a BCA it is critical to determine how and when a distributed storage resource intends to operate because the temporal nature of its operation will materially affect its benefits and costs. While it is generally easier to predict the effect on net loading from DERs such as solar PV and EE, determining this effect for distributed storage is more complex given its potentially dynamic operation. For example, the multiple potential operational modes for distributed storage adds complexity to determining when the resource will operate in each. Further, while the resource may plan

to provide a certain set of services at the time of its interconnection, it may seek to adjust its operations in the future to target other services as a result of changing market dynamics or policies.³⁹

If the utility directly owns and operates the distributed storage resource, or if it has an executed contract with the third-party distributed storage owner to provide certain services, it will likely have greater confidence over the resource's planned operational profile. This confidence will allow it to more conclusively determine the net benefit or cost of each relevant impact.

Alternatively, if the storage resource is owned and operated by a non-utility party, the utility may seek to collect certain information—including as part of the interconnection process—that will enable it to more effectively determine the resource's planned operational profile. In the case of non-utility ownership and operation, jurisdictions should continue to identify best practices for the types of information needed to determine how the storage resource will likely operate in the future based on proposed operational characteristics.

9.5.2 Determining the Counterfactual Host Customer Baseline

If distributed storage is used to modify load as if it were a DR resource, there will be challenges akin to those discussed in Section 7.5.2 for how to determine the counterfactual baseline to measure the quantity of DR provided. Determining a storage resource's performance as a DR resource requires an assessment of what the customer's energy consumption would have been absent the storage resource.

To the extent a distributed storage resource is coupled with load (i.e., behind the same meter), it is critical to discern what impact the storage resource's operation has on the customer's load and how much of this impact is due to an intentional action to meet the dispatch instructions for the storage resource.

Determining the counterfactual baseline will require an understanding of the storage resource's typical operational profile, similar to how baselines are used to determine a customer's typical consumption profile. Alternatively, if the storage resource has its own production meter separate from the customer's load meter, then the storage resource's operation can effectively be separated from customer load, allowing for a more direct analysis of changes in the customer's net load due to dispatch instructions.

9.5.3 Accounting for Provision of Multiple Services

While all DERs can stack multiple values (see Section 5.3), distributed storage offers unique considerations: Its dispatchability allows for targeted application to various services, but its duration limitations warrant consideration about potential tradeoffs in services it can provide. Evaluating a distributed storage resource requires attention to which services it will likely provide and how it will balance potential conflicts that prevent the resource from fully meeting all of its intended use cases (e.g., operation during one part of the day rendering the resource unavailable for another service later that day).

This is an emerging focus area in the industry, but further work is needed to more formally define a framework that stipulates rules governing

Evaluating a distributed storage resource requires attention to which services it will likely provide and how it will balance potential conflicts that prevent the resource from fully meeting all of its intended use cases.

³⁹ If the distributed storage resource changes its originally intended operating profile but does not seek to increase the amount of power it can inject onto the system, it may not require a new interconnection review.

distributed storage resource's ability to provide multiple services. The California Public Utilities Commission (CPUC) and New York Public Service Commission (NY PSC) have focused on whether the storage resource is providing reliability-based services and whether services can be differentiated in terms of timing or the storage resource's capacity dedicated to meet each service (CPUC 2018; NYPSC 2018). Ultimately, while there may be standardized aspects of a framework guiding the ability of distributed storage resources to stack multiple services, there may need to be variation across jurisdictions given the unique market, policy, and regulatory environment of each.

9.6 Lost Revenues and Rate Impacts

Lost revenues and potential rate impacts should not be included in cost-effectiveness analyses. Instead, DER lost revenues and rate impacts should be analyzed separately using rate, bill, and participation analyses. (See Section 2.3 and Appendix A.) In conducting BCAs, therefore, lost revenues should be identified so that they can be properly excluded from BCAs and properly included in rate, bill, and participation analyses.

In general, several key factors affect the extent to which DERs might create rate impacts:

- Increases in utility system costs will put upward pressure on rates.
- *Reductions in utility system costs* will put downward pressure on rates.
- *Reductions in sales* from DER resources will put upward pressure on rates.
- Increases in sales from DER resources will put downward pressure on rates.
- *Rate design* will affect the amount of lost or increased revenues created by the DER.

Distributed storage resources will typically increase utility system costs because of financial incentives or program administration costs, but will reduce utility system costs by avoiding generation, transmission, and distribution costs (see Table 9-1).

Distributed storage resources will typically create lost revenues by (a) allowing host customers to reduce their energy bill by reducing their energy consumption, especially during high-price hours; and (b) allowing host customers to reduce their energy bill by reducing demand charge payments, where such charges are in place. The lost revenues from distributed storage resources will depend upon many factors, including the storage device's use case and the rate design in place.

Regulators and other stakeholders that are concerned about distributed storage rate impacts could consider the following: (a) conduct a long-term rate, bill, and participation impact analysis, (b) seek and implement rate designs and storage initiatives likely to mitigate equity concerns, (c) seek ways to expand customer participation, especially among low- and moderate-income customers, and (d) periodically review impacts as distributed storage deployment rates increase.
10. ELECTRIFICATION

This chapter describes the benefits and costs most relevant to electrification resources. It identifies key factors that affect electrification benefits and costs and provides guidance on addressing common challenges with electrification cost-effectiveness analyses.

10.1 Summary of Key Points

- Electrification resources can include technologies to replace other fuels in buildings, as well as technologies to replace other fuels in the transportation sector, such as EVs.
- Electrification resources will increase net electric utility system costs because they require increased electricity generation. However, they will also reduce costs associated with the other fuels that they replace. Consequently, electrification resources are likely to be cost-effective only when the benefits of other fuels are included in a jurisdiction's cost-effectiveness analyses.
- The amount of added costs to the electric grid due to electrification will depend upon when the technologies are utilized, which in turn will be influenced by the host customer rate structure. This is particularly true for electrification measures whose demands can be most flexibly managed by customers, such as EVs.
- The added costs to the electric grid due to most electrification technologies may also be reduced when combined with DR, such as "managed charging" of EVs and direct load control of heat pumps.
- EVs with V2G capability can further mitigate increased costs to the grid as a result of electrification, and potentially even reduce net *electric utility* system costs, because of their ability to function as storage.
- Electrification measures can reduce net air emission impacts (both GHG and other pollutants), as long as the marginal emissions from the electricity grid are low enough relative to the marginal emissions of the displaced fuel.
- Different charging levels for EVs—particularly the prevalence and use of fast charging with short duration draws of large amounts of power—can potentially impact T&D capacity needs and costs.
- Electrification resources will typically create increased revenues for the electric utility. These might lead to reduced electricity rates, depending upon the magnitude of increased revenues, utility system costs, and utility system benefits. Rate impacts from electrification resources are more appropriately assessed using rate, bill, and participation analyses (Appendix A).
- Electrification resources will sometimes create lost revenues for the gas utility. These might lead to increased rates, depending upon the magnitude of lost revenues and the magnitude of gas utility system benefits.

10.2 Introduction

For the purposes of this manual, electrification is defined as the conversion of energy use from a nonelectric fuel source to electricity. This includes:

- Electrification of buildings and industry (for any of a wide range of potential end-uses, including space heating, water heating, and industrial processes).
- Transportation electrification (e.g., light-, medium- and heavy-duty EVs; electric flight and electric marine applications).

Electrification can be "partial" (e.g., a plug-in hybrid electric vehicle (PHEV) or a heat pump retrofit in a home where the existing fossil fuel furnace or boiler remains as a secondary system) or "complete" (e.g., a battery electric vehicle (BEV) or a home switching entirely to electric heat).

The fundamental BCA principles described in Chapter 2 can be used to identify the relevant benefits and costs for electrification, as with all DER types. In many cases, identifying whether a particular benefit or cost is relevant to electrification requires defining the specific electrification technology and use case.

10.3 Benefits and Costs of Electrification

The tables in this section summarize the full range of potential benefits and costs of electrification resources (see Chapter 4 for definitions of impacts). Each impact is described as a benefit, a cost, or either, depending on the most common applications of this technology. There might be some less-common applications where a cost could be a benefit, or vice versa. The tables include notes on applicability that provide further explanation for those impacts that may be either a cost or benefit.

Table 10-1 presents the potential benefits and costs of electrification on the electric utility system, while Table 10-2 presents the impacts of electrification on the gas utility system. All electrification utility system impacts presented in these tables should be included in BCA tests. The remaining electrification impacts presented in Table 10-3 (Host Customer) and Table 10-4 (Societal) should be included in BCA tests to the extent they are applicable to the jurisdiction's policy goals (see Chapter 3).

Table 8-1 shows that many electrification measures have the potential to function as DR resources. In addition, EVs with V2G capability can function as storage resources (see Chapter 9). The characterization of impacts in Table 10-1 as benefits or costs addresses only the electrification features of these technologies because (a) not all electrification measures can provide DR and/or storage functionality; (b) for any DR or storage functionality to have impact it must be activated through a utility system initiative; and (c) not all EVs, heat pumps, electric water heaters and/or other electrification program will get enrolled in complementary DR and/or storage initiatives. The impacts of DR and storage are addressed in Chapters 7 and 9, respectively. The impacts of multiple DERs—which would encompass the combination of electrification and DR, for example—are discussed in Part IV of this manual.

Туре	Utility System Impact	Benefit or Cost	Notes, or Typical Applicability	
	Energy Generation	•	A cost because electrification increases electricity generation. Cost for many measures can be reduced through economic dispatch using DR and further reduced through use of storage capabilities of V2G EVs. (See Chapters 7 and 9.)	
	Generation Capacity	•	A cost because most uncontrolled electrification measures will add some demand on system peak (electric heat in summer peaking system is a possible exception). Resulting capacity cost for many measures can be reduced through DR; it can be eliminated or even made negative (i.e., a grid benefit) if storage capability of V2G EVs is utilized. (See Chapters 7 and 9.)	
General	Environmental Compliance	•	By adding load to the grid, electrification can increase electric costs of compliance (but reduce other fuel costs of compliance).	
	RPS/CES Compliance	•	By increasing electricity load, the quantity of renewables needed to meet RPS increases.	
	Market Price Response	•	Any increase in electricity consumption will increase market clearing prices where there are competitive wholesale markets.	
	Ancillary Services	•	By itself, electrification could increase ancillary services costs. However, both EVs and water heaters offer the ability to provide ancillary services when enabled through DR; if that capability is utilized, this can become a benefit. (See Chapter 7.)	
Transmission	Transmission Capacity	•	Most uncontrolled electrification measures will add some demand at transmission peak time (electric heat in summer peaking region a possible exception). Resulting capacity cost for many measures can be reduced through DR and eliminated or even made negative (i.e. a grid benefit) if storage capability of V2G EVs is utilized. (See Chapters 7 and 9.)	
	Transmission System Losses	•	Any consumption increase will increase losses.	
	Distribution Capacity	•	Most uncontrolled electrification measures will add some demand at distribution peak time (electric heat in summer peaking area is a possible exception). Resulting capacity cost for many measures can be reduced through DR and eliminated or even made negative (i.e. a grid benefit) if storage capability of V2G EVs is utilized. (See Chapters 7 and 9.)	
Distribution	Distribution System Losses	•	Any consumption increase will increase losses.	
	Distribution O&M	•	Any consumption increase will increase O&M.	
	Distribution Voltage	•	Added loads will make distribution voltage more challenging to keep at desired levels.	

Table 10-1. Potential Impacts of Electrification: Electric Utility System

Туре	Utility System Impact	Benefit or Cost	Notes, or Typical Applicability		
General	Financial Incentives	•			
	Program Administration Costs	•	Costs, where relevant		
	Utility Performance Incentives	•			
	Credit and Collection Costs	•	A benefit because other fuel savings may make it easier for customers with electrified end-uses to afford electricity bills.		
	Risk	•	Adds risk to electric grid but may be offset by reduced risk associated with displaced fuel(s)		
	Reliability	•	By adding load to the grid, electrification will decrease electric system reliability. For many measures that effect can be reduced through DR; it can be eliminated or made negative (i.e., a grid benefit) if storage capability of V2G EVs is utilized. (See Chapters 7 and 9.)		
	Resilience	ο	Electrified building end-uses do not affect electric system resilience; EVs functioning in V2G mode, could improve resilience by functioning as storage. (See Chapter 9.)		

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; O = not relevant for this resource type.

Table 10-2 presents the potential benefits and costs of electrification on gas utilities and other fuel systems. By definition, electrification programs decrease the use of other fuels.

Туре	Non-Electric Energy System Impact	Benefit or Cost	Notes/Typical Applicability	
	Fuel and Variable O&M	•	Any decrease in demand for other fuels caused by electrification will produce other fuel cost reductions.	
Other Fuel: Energy	Capacity		To the extent electrification causes a decrease in peak demand for other fuels, it can avoid capital investment in system capacity. Electrification of new construction also eliminates capital cost associated with connection to other fuel delivery systems.	
	Environmental Compliance	•	By reducing consumption of other fuels, electrification may reduce cost of compliance with environmental regulations for those other fuels (depending on how the regulations are structured).	
	Market Price Response	•	Any decrease in consumption of other fuels as a result of electrification will lower market clearing prices where there are competitive wholesale markets.	
	Financial Incentives (e.g., rebates)	•	A cost to other fuel systems only if applicable to those systems (e.g. for natural gas non-nine solutions)	
	Program Administration Costs	•		
	Utility Performance Incentives		(0,	
Other	Credit and Collection Costs	•	If total energy bills across all fuels decline, customers may be better able to pay all bills.	
Fuel: General	Risk	•	Lower consumption resulting from electrification should reduce risk (e.g. of exposure to future fuel price volatility).	
	Reliability	•	Lower consumption of displaced fuel should increase reliability of supply of that fuel.	
	Resilience	•	Reduced reliance on displaced fuels should reduce the amount of infrastructure for delivery of that fuel that needs to be replaced as a result of storms or other catastrophes.	

Table 10-2. Potential Impacts	of Electrification: Gas Utilit	y and Other Fuel System Impacts
-------------------------------	--------------------------------	---------------------------------

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; O = not relevant for this resource type.

Table 10-3 presents the potential host customer impacts of electrification.

Туре	Host Customer Impact	Benefit or Cost	Notes/Typical Applicability
	Host Customer Portion of DER Costs	•	Both the cost of the electric products (e.g., EVs or heat pumps) and possible costs to upgrade electric service necessary to use them (including EV charging equipment and related electrical upgrades)
	Interconnection Fees	0	Potentially a cost for V2G, otherwise not applicable
Customer	Risk •		Potentially a cost due to reduced electricity fuel diversity; potentially a benefit due to reduced volatility of other fuel prices
	Reliability	•	EVs can function in times of gasoline shortages and heat pumps can keep buildings heated if there are problems with fossil fuel access or with fossil fuel heating systems; conversely, there can be reliability issues tied to power outages.
	Resilience	•	V2G storage capability can be a benefit to host customers if used as back-up power during grid outages; otherwise not applicable.
	Tax Incentives	•	Potentially a benefit where relevant.
	Host Customer NEIs	•	Benefit or cost depending on NEI (see Section 10.4.5)
	Low-income NEIs	•	For low-income electrification only

Table 10-3. Potential Impacts of	Electrification: Host Customer
----------------------------------	--------------------------------

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

Table 10-4 presents the potential societal impacts of electrification.

Table 10-4. Potential Im	pacts of Electrification: Societal
--------------------------	------------------------------------

Туре	Societal Impact	Benefit or Cost	Notes/Typical Applicability
Societal	Resilience	•	Depends upon whether reduced gas consumption affects critical customers and whether increased electricity consumption stresses the grid
	GHG Emissions	•	Depends on use case and hourly environmental profile of
	Other Environmental	•	displaced by appliance/vehicle
	Economic and Jobs	•	Potentially a net benefit or net cost depending upon fuels displace
	Public Health	•	Same as GHG emissions and other environmental
	Low Income: Society	•	Potentially a benefit depending on siting and low-income participation
	Energy Security	•	Potentially a benefit depending upon the extent that petroleum products are being displaced

• = typically a benefit for this resource type; • = typically a cost for this resource type; • = either a benefit or cost for this resource type, depending upon the application of the resource; \circ = not relevant for this resource type.

10.4 Key Factors that Affect Electrification Impacts

10.4.1 Technology Characteristics

Electrification can be applied to a variety of end-uses, including EVs; electric heating, water heating, cooling, cooking, drying and other end-uses in homes and business; and electric processes for industrial customers. There are also a number of different technology options for each of those end-uses. Put simply, not all EVs or heat pumps or electric water heaters or electric cooktops are the same. The different characteristics of those technologies can affect the cost-effectiveness of electrification in a variety of ways.

First, the efficiency of an electrification technology can have important implications for its costeffectiveness. The more efficient the electrification measure, the lower the added costs to the electric system for the same amount of displaced fossil fuel. For example, a heat pump can be two to four times more efficient at providing electric heat than electric resistance heating systems. Thus, electrifying space heating with a heat pump may add 50 percent to 75 percent less cost to the grid than electrification with electric resistance heat. Of course, if there is an added cost associated with greater electric efficiency—as is the case with heat pumps, which are much more expensive than electric resistance heat.

The change in Source Energy consumption resulting from an electrification measure is equal to the total amount of energy required to produce (in a power plant) and deliver (across power lines) the electricity consumed by an electrification measure, minus the gasoline, natural gas, fuel oil, propane or other fuels displaced by the measure.

Specifically, 1 kWh of electricity has 3,412 BTUs of energy, while a fossil fuel fired power plant can burn 7,000 (roughly 50% efficient) to 11,000 BTUs (roughly 30% efficient) or more of fuel to produce one kWh. The amount of fuel burned to produce a kWh is called the power plant's "heat rate." In 2018, the average heat rate for gas-fired power plants was 7,822 BTUs/kWh, or an average efficiency of about 44% (EIA 2019a). There are also losses associated with delivering a kWh of electricity from power plants to homes and businesses, averaging about 5% in the United States (EIA 2019b).

Many electrification measures (e.g., high performance heat pumps) can be considered "energy efficiency" measures because they reduce total source energy consumption.⁴⁰ On the other hand, some potential electrification measures (e.g., electric resistance heating systems) can increase source energy consumption.⁴¹ It is in fact possible for some electrification measures that increase source energy consumption to be cost-effective, while some that reduce source energy consumption may not be cost-effective. As such, in cases where a jurisdiction adopts policies that preclude promotion of electrification measures unless they reduce total energy consumption (whether source energy or site energy)

⁴⁰ For example, on an electric grid with an average marginal heat rate of 8,000 Btu/kWh and 8 percent average line losses, a high-performance heat pump with an average coefficient of performance (COP) of 3.0 would consume 24 percent less *source energy* per unit of heat produced than a 90 percent efficient propane furnace. COP is the ratio of energy output (in the form of heat in this example) to energy input (electricity in this example).

⁴¹ For example, on an electric grid with an average marginal heat rate of 8000 Btu/kWh and 8 percent average line losses, electric resistance heating, which has a coefficient of performance (COP) of 1.0, would consume more than twice as much source energy per unit of heat produced than a 90 percent efficient propane furnace.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 173 of 302

consumption)⁴² any such constraints are policy choices and not necessarily reflections of cost-effectiveness.

Second, many electrification measures have the potential to be managed by customers to minimize costs to the grid (especially if they see hourly price signals) and/or to be controlled directly through DR programs. Some electrification measures—namely EVs and electric water heaters—are particularly well-suited to load management through DR programs. Unlike other electrification measures, EVs with V2G capability offer the potential to actually reduce net system peak demands and/or net T&D peak demands.

Third, while many EVs allow only one-way electricity flows (from the grid to the vehicle for charging), some have two-way flow capability (also allowing vehicle-to-grid flows). EVs with that unique feature can function as storage resources for the grid (similar to other storage options discussed in Chapter 9). This allows not only charging during times of low prices and/or low net demand, but also the ability to push electricity back onto the grid during times of high prices and/or high net demand. Thus, unlike other electrification measures, EVs with V2G capability offer the potential to actually *reduce* net system peak demands and/or net T&D peak demands.

EVs with V2G capability even have the potential to reduce total electric generation (energy) costs — if costs avoided when they put power onto the grid are high enough to offset the costs of increased consumption when they are taking power off the grid.⁴³ Of course, the extent to which V2G storage capabilities will actually affect system peak demands and/or energy costs will depend on the extent to which the vehicle owners both participate in utility system initiatives to utilize the storage capabilities and have their vehicles plugged in and accessible to the grid at the times their storage capabilities may be needed.

Finally, different types and even different models of electrification technology can have significant impacts on host customer non-energy impacts. For example, advanced air source heat pumps can improve comfort relative to fossil fuel heating systems, while less advanced systems can create comfort problems, particularly in colder climates. Similarly, electric induction cooktops can enhance cooking performance relative to gas or propane cooktops, while standard electric coil cooktops offer less heat control than gas or propane (though all electric cooktops provide health and safety benefits relative to gas or propane). Both high performance heat pumps and induction cooktops typically cost more than standard electric alternatives. Thus, in jurisdictions whose policies dictate that host customer impacts be included in cost-effectiveness assessments, it will be important to be clear about which electric technology is being assessed and to characterize the specific attributes of each technology.

⁴² The concept of "source efficiency" can be important when the marginal generation on the grid is fossil fuel-powered. In such cases, there are tradeoffs—affecting both cost and environmental emissions—between increased fossil fuel consumption on the grid and reductions in direct fossil fuel consumption by consumers in their vehicles, homes, and businesses. Source efficiency loses relevance to the extent that renewable energy sources such as wind and solar—whose "fuel" has no cost and which produce no emissions—are the marginal generation resource on the grid.

⁴³ Consider the following hypothetical example in which total electricity used for driving an EV was 3000 kWh per year, where the EV storage functionality was used to put 500 kWh per year onto the grid when energy prices averaged \$0.25/kWh (a cost savings of \$125), and where the average energy cost during times the EV was taking its 3500 kWh off the grid for charging was \$0.03/kWh (a cost of \$105). In this hypothetical, the net generation cost for the EV would be -\$20 (or a \$20 cost savings), before accounting for any capacity cost savings.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 174 of 302

10.4.2 Technology Operating Profile

Electrification impacts on the grid will depend considerably on their load shape, which is the amount of energy consumed during different seasons of the year as well as during different hours of the day. Electrification load shapes differ substantially from one end-use to another. For example, electricity consumption for residential space heating is naturally concentrated in the winter, and often greater at night when outdoor temperatures are lowest. This pattern will be fairly similar on weekdays and weekends.⁴⁴ In contrast, electricity consumption for electrified industrial processes will typically be the same in all seasons of the year, and even relatively consistent across all operating hours of the day (e.g., similar across all 16 hours of a two-shift manufacturing facility). However, depending on whether the facility operates five, six or seven days a week, there could be significant differences between weekdays and weekends.

There can also be differences in load shapes between different kinds of technology used for a given electric end-use. For example, the shape of the electric load for space heating will be different for electric resistance heating, standard air source heat pumps, and cold climate air source heat pumps—as described below:

- For electric resistance heat, the consumption in any given hour is largely linearly related to outdoor temperature because the efficiency of resistance heat never changes.
- Heat pump efficiency declines as the outdoor temperature declines; thus, its consumption increases more quickly as outdoor temperatures drop. While it is still more efficient than electric resistance heat, the efficiency gain relative to resistance heat declines as the temperature declines.
- At a low enough outdoor temperature, heat pumps may no longer be able to meet the heating needs of the building and may need to rely on electric resistance back-up heat. The outdoor temperature at which that change-over occurs will depend on the heat pump.
- Standard air source heat pumps rely entirely on electric resistance back-up heat when the outdoor temperature drops below a certain point (e.g., below 5, 10, 20, or 30 degrees Fahrenheit, depending on the heat pump model).
- High performance, cold climate, air source heat pumps can perform at their stated nameplate capacity in heat pump mode down to 5 degrees Fahrenheit, and they can continue to provide some heat down to -10 or -15 degrees Fahrenheit or even lower (although at a lower efficiency).
- Ground source heat pumps can produce heat relatively efficiently regardless of outdoor air temperature.

Similarly, as discussed in 10.5.2 below, load shapes for EVs can be different depending on the extent to which customers rely on home chargers, have access to chargers at work, and the type of charger that they use (level 1, level 2 and/or direct current fast charging).

Finally, as discussed above as well as in 10.5.1, load shapes for a given electric end-use can vary significantly depending on rate design (e.g., whether customers pay time-varying rates or flat rates), as well as the extent to which their electrified loads are managed or controlled through DR programs.

⁴⁴ There may be some differences between weekdays and weekends if customers set back thermostats during weekday work hours when no one is home.

10.4.3 Enabling of DR Capability

Many electrification measures can be managed by the host customer to minimize costs (if the host customer has the appropriate rate structure) and/or controlled through DR initiatives. Both customer load management and DR initiatives can reduce—potentially significantly—the generation and/or T&D capacity costs that electrification measures impose on the electric system.

If DR capabilities are enabled at the time the electrification measures are purchased or installed, the additional capacity, T&D, and/or reliability benefits should be included in cost-effectiveness assessments. The same is true of the storage resource capability of EVs with V2G capability. However, even if the DR and/or V2G storage capability is not enabled at the time of the electrification measure purchase or installation, there is a value associated with the latent capability of electrification measures to provide DR and/or storage capability to the grid in the future. That value, though potentially difficult to quantify, should be reflected in cost-effectiveness assessments of electrification.

10.4.4 Other Fuel Impacts

Electrification means substituting (i.e., increasing) electricity consumption for the consumption of other fuels, and therefore will increase electric system costs (with the possible exception of EVs with V2G capability, as discussed above.) While management or control of electrified loads can minimize the *magnitude* of the electric system cost increases associated with most electrification measures, it cannot eliminate them altogether. Thus, again with the possible exception of EVs with V2G capability, electrification measures are likely to be deemed cost-effective only if the benefits of their corresponding reductions in gasoline, diesel fuel, natural gas, fuel oil, propane and/or other fuels are included in a jurisdiction's BCA test.

Electrification measures are likely to be deemed costeffective only if the benefits of their corresponding reductions in gasoline, diesel fuel, natural gas, fuel oil, propane and/or other fuels are included in a jurisdiction's cost-effectiveness test and cost-effectiveness analyses.

As shown in Table 10-2, for jurisdictions that include other fuel impacts in their BCA tests (consistent with jurisdictional energy policies), it is important to recognize that the other energy system cost savings resulting from electrification can extend beyond just commodity (or fuel) savings. For example, when electrification lowers demand at times of gas system peak demand, it can help defer the potential for future capital investments to upgrade the gas transmission and/or distribution system infrastructure. There may be similar "capacity" savings associated with reduced infrastructure for distribution of petroleum products.

In addition, electrification of buildings can obviate the need for connecting those buildings to the natural gas distribution system (including installation of a gas meter), saving the cost of such connections. This can occur in the following example scenarios:

- In the case of new construction, when all end-uses that might otherwise have used natural gas—space heating, water heating, drying, cooking, etc.—are instead met entirely with electricity. In such cases, the builder would save the cost of connecting the building to the gas distribution system. Any ongoing fixed costs (meter reading, billing, etc.) of supplying natural gas to the home in subsequent years would also be avoided.
- In the case of fuel-switching existing buildings using fuel oil, propane, or other unregulated fuels to either (a) natural gas which is not currently locally available or (b) electricity which is already available at the building. In such cases, assessments of the relative cost-effectiveness of the

natural gas conversion should include consideration of the cost of bringing the gas distribution infrastructure to the building (whether relatively modest because gas is already "at the curb" or more substantial because the gas pipe would need to be brought to the town and then to the building) as well as ongoing fixed annual costs of supplying gas to the building.

10.4.5 Host Customer Non-Energy Impacts

Table 10-5 presents a summary of host customer NEIs that might potentially be created by electrification and EVs resources. These impacts can sometimes be in the form of benefits and sometimes costs. The presence, direction, and magnitude of electrification or EV NEIs will depend upon many factors, including the use case for the technologies, the type of host customer (low-income, residential, commercial, industrial), the host customer's needs, and the rate design in place.

Table 10-5 indicates whether the category of NEIs is typically a cost, a benefit, or either in most common applications. The actual direction of the impact, however, should be determined on a case-by-case basis.

Host Customer NEI	Electrification	EVs	Examples
Transaction costs	•	•	 Electrification: Application fees and processes EVs: reduced need to stop to re-fuel when charging at home or work EVs: increased time to re-fuel when charging on the road
Asset value	•	0	- Electrification: Increased building value
Productivity & operational flexibility	0	•	- EVs: Reduced O&M costs
Comfort	•	•	Electrification: Thermal comfort may be better or worseElectrification and EVs: Noise reduction
Health & safety	•	0	- Electrification: Reduced indoor air pollution
Empowerment & control	•	•	 Electrification and EVs: Psychological benefit from empowerment Electrification and EVs: Energy independence
Satisfaction & pride	•	•	 Electrification and EVs: Contribution to addressing environmental issues

Table 10-5. Potential Host Customer Non-Energy Impacts of Electrification and Electric Vehicles

• = typically a benefit for this example; • = typically a cost for this example; • = either a benefit or cost for this example, depending upon the application of the resource; \circ = not relevant for this example.

Key examples include:

- Transaction Costs can be reduced for owners of EVs. The ability to charge an EV at home and/or at work can eliminate the need to stop to refuel at gas stations. Conversely, driving range limitations can be considered a cost by EV customers because the much longer times required for recharging (relative to refueling a gas tank) can make it challenging to use such vehicles for some long-distance trips. These concerns may diminish as the range of EVs continues to improve and access to fast-charging facilities increases.
- Operation and Maintenance cost savings are an important benefit of EVs. EVs do not need oil changes and they tend to have fewer moving parts and therefore fewer mechanical problems as

the vehicle ages. O&M costs for EVs can be as much as half the costs for internal combustion engine driven vehicles (Logtenberg et al. 2018).

- *Heating Comfort* can be either a benefit or cost, depending on the type of heating system installed. For example, customers can experience comfort issues when buildings are retrofitted with heat pumps that are not well designed for their climates (ERS 2014). On the other hand, that concern can be addressed by matching heat pump selection to the climate (e.g., cold climate air source heat pumps or ground source heat pumps for cold climates). In addition, heat pumps with advanced features, such as variable speed inverter compressors, room temperature sensors, and variable air flow can improve comfort by keeping temperatures steadier.
- Cooling/Dehumidification Comfort is often a benefit. For example, heat pump water heaters both cool and dehumidify spaces. Also, when a heat pump is installed for space heating in homes that do not have (and would not otherwise have installed) cooling, the ability to also cool a room or house or business becomes an added amenity (though with added electric system costs).
- Noise Reduction is usually a benefit. EVs run more quietly than internal combustion engine driven vehicles. Advanced heat pumps that modulate the amount of heat provided to a room or building are often quieter than fossil fuel furnaces or boilers that are typically either 100 percent "on" (or 100 percent "off"). On the other hand, heat pump water heaters can be noisier than gas water heaters (though standard electric resistance water heaters are not).
- Cooking Control can be either a benefit or cost, depending on the type of electric cooking technology installed. Gas or propane cooktops allow instant heat as well as instant reaction to changes in the amount of heat being applied to a pot or pan. Electric coil cooktops cannot provide such instantaneous control; however, electric induction cooktops can. In addition, electric cooktops, whether coils or induction technology, can reach high heat—i.e. boil a large pot of water—more quickly than gas cooktops (Hope 2019 and Livchak et al. 2019).
- *Health and Safety* is generally a benefit of electrification. For example, electric appliances eliminate any indoor air pollution that would otherwise result from the use of fossil fuel burning appliances. That is perhaps most notable for fossil fuel cooktops, but also for heating and water heating appliances that are not properly vented, have cracked heat exchangers, and/or that experience back-drafting. Also, because electric induction cooktops use electromagnetism to produce heat and cook food in metal pots or pans, it is the pot or pan that becomes the "heater." When the pot or pan is removed from the cooktop there is no heat being produced, reducing chances of inadvertent fires or burns.
- Driving Responsiveness can be increased by EVs, which typically react more quickly than internal combustion engine-driven vehicles when accelerating and have very good torque (DOE 2020).

10.4.6 Air Emissions Impacts

Electrification measures can reduce net air emission impacts of both GHG and other pollutants, in cases where the marginal emissions from the electricity grid are low enough relative to the marginal emissions of the displaced fuel. Achieving aggressive carbon reduction goals may require a substantial amount of electrification because for some sectors, such as natural gas and transportation, there are few alternatives for significantly reducing GHG emissions at a reasonable cost (Synapse 2018).

As discussed in Chapter 5, DER air emission estimates should account for two important factors: (a) the time period for which the DER operates; and (b) the marginal emission rates of the utility system at the

time when the DER operates. Further, the *Ensure Symmetry* principle requires accounting for both reduced and increased air emissions from DERs.

For electrification measures, there are several additional factors to account for regarding emission rates:

- There is a variety of fuels that can potentially be displaced by electrification (e.g., gasoline), and the marginal emission rates for those fuels should be accounted for.
- The efficiency of the baseline technology (e.g., an internal combustion engine vehicle or fossil fuel burning furnace, boiler, or water heater) will affect the fuel savings and therefore the air emission rates.
- The efficiency of the electric technology (e.g., the EV, heat pump, or heat pump water heater) will affect the fuel savings and therefore the air emission rates.

EVs with V2G capability can potentially have significant intra- and inter-day variation, making it especially important to account for the detailed time periods in which they are expected to operate.

10.4.7 Increased Electricity Costs

As discussed above, with the possible exception of EVs with V2G capability, electrification measures will increase electric system costs by increasing the amount of electric energy that needs to be produced and by adding to electric system and/or T&D system peak demands.

The extent to which increased consumption of electric energy adds cost to the system will be driven by a variety of factors including:

- The operating profiles of the electrification measures, including the extent to which electrification loads are coincident with current system and/or T&D peaks;
- The extent to which the operating profiles are affected by electric rate structures and/or DR measures designed to shift consumption to off-peak and lower cost hours;
- Variable energy costs, generation capacity costs, and T&D capacity costs, which often vary from season to season as well as during different times of day within the same region; and
- The amount of excess system generation capacity and/or excess T&D capacity on the grid.

All of these factors could be highly variable from one service territory to another.

Estimation of potential impacts on system and/or T&D peak demands entails consideration of more than just current grid load profiles. Jurisdictions should also consider how current peaks might change over time both absent electrification (e.g. "duck curve" effects associated with increasing deployment of solar generating capacity) and/or as a result of electrification (e.g. by adding enough electric heat to change a system from summer-peaking to winter-peaking).

In the case of electric heat pumps, it is also important to understand the potential for adding load during both winter peaks and summer peaks. Specifically, there may be cases where the heat pumps are adding cooling capacity to buildings that did not previously have it and would not have added it absent the electrification initiative.⁴⁵

⁴⁵ It is important to not just assume that the baseline for a building currently without cooling would continue to be no cooling. Indeed, several evaluations of cold climate heat pump programs in New England found that many homes participating in such programs were initially interested in adding air conditioning and were convinced by the programs to purchase heat pumps instead of air conditioners (ERS 2014).

10.5 Common Challenges in Estimating Benefits and Costs

10.5.1 Seasonal and Daily Load Profiles

As discussed in the preceding sections, the impacts electrification measures will have on the grid will depend upon their use case, particularly their load shape. Load shapes have been used to estimate impacts in EE program cost-effectiveness analyses for decades. However, it is potentially more challenging for electrification measures than for efficiency measures.

This challenge can be addressed, in part, by utilizing end-use load shapes already in use for EE analyses, with the following potential modifications:

- To the extent that electrification is only partial—e.g. where electric heat pumps are only partially displacing fossil fuel heating, and the fossil fuel heating systems remain to help serve loads during very cold temperatures—EE load shapes will need to be modified.
- To the extent that any electrified loads will likely be modified by customers in response to new rate structures (e.g., time-varying rates)⁴⁶ and/or controlled through different forms of DR, load shapes will need to be modified.
- Finally, load shape assumptions will need to be developed for EVs whose charging patterns can be highly variable and influenced by rate design (SEPA 2019b), the availability of charging infrastructure for EVs, the fast-evolving technology (particularly around EV batteries), and other factors.

Overall, jurisdictions will need to develop assumptions to address each of these modifications or needs, and initial assumptions will likely involve significant uncertainty. However, that uncertainty can be mitigated over time as analyses are performed and used to update assumptions.

10.5.2 Charging Methods Used by EVs Consumers

The impacts EVs will have on the grid depend on the charging methods used by EV consumers:

- Level 1 chargers have relatively modest draws (up to 3 kW) of long duration (often 12 hours or more to fully charge a light-duty vehicle).
- Level 2 chargers can draw 3 to 22 kW and usually charge EVs in three to four hours.
- DC Fast Chargers (DCFC) have draws of 50 kW to 120 kW (Superchargers) and can charge most EVs to 80 percent of capacity within 30–60 minutes.
- Some light-duty vehicles are now accepting charging rates at up to 350 kW and the industry is currently developing standards for up to 1 MW charging for heavy-duty vehicles (SEPA 2019b).

Over an entire service territory, the differences in draw amount and duration of draw associated with the different charging options may be smoothed out if charging times are diversified across a large population of customers. However, high use of DCFC could reduce such diversification, at least on certain localized elements of the distribution system. This would cause the average EV load shape in some geographic areas to be different than it would be if rapid charging were not as widely used. In other words, large "spikes" in consumption from DCFC can potentially have significant effects on peak

⁴⁶ Which would lead to a different pattern of usage relative to an existing electric heating load shape that was based on a simpler rate structure.

demands of localized elements of the distribution system that serve them. (The average EV load shape for parts of the grid with high usage of rapid charging may be very different than for other parts of the grid or for the system as a whole.)

Even if use of a DCFC charging station were more evenly spread across all hours of the day (e.g., because it is serving large multi-family buildings, a truck stop, fleet vehicle facility, other institutional customers, or any other location with many vehicles charging at different times) the impact on the grid could be different than more dispersed use of level 1 and level 2 chargers as a result of the greater geographic concentration of charging.

This challenge can be addressed by considering the potential differences in grid impacts of different levels of charging, particularly the T&D system impacts, and estimating the expected mix of charging methods.⁴⁷ This assessment should include charging needs for large commercial fleets. To the extent that any adverse effects would be mitigated by rate design, DR, non-wires alternatives, and/or BTM storage, such mitigation should also be factored into analyses.

10.5.3 Impacts on Curtailment of Renewable Resources

In some areas, during some hours of the year, local generation from wind, solar and other generation resources (such as must-run generators) can exceed local area demand. If transmission capacity is insufficient to move all of the excess generation to other areas, renewable generation might need to be curtailed. To the extent that electrification measures are installed close to those transmission-constrained renewables, the curtailment of such renewable resources can be reduced. (See Section 5.3.)

The reduction in curtailments is clearly a benefit to the owners of the renewable resources because they will receive payments that they otherwise would not have received. Further, to the extent that electrification can utilize resources that otherwise would have been curtailed, the marginal cost to serve the electrified end-uses with such resources is zero. It is important to note, however, that from the broader electric system perspective there are no recognized cost reductions due to greater utilization of curtailed resources; there are just no additional costs to serve electrified load during the hours that renewables would otherwise have been curtailed. This is because no new generation needed to be constructed and no new variable costs need to be incurred to meet the needs of the electrified loads in the hours during which curtailments would otherwise have occurred. All else equal, however, electrification in such regions will be more cost-effective than in regions with no hours during which the marginal cost of electricity is zero. Geo-targeting of electrification to areas experiencing or forecast to experience significant renewables curtailment could be more cost-effective than system-wide electrification initiatives.

A potential exception to there being no electric system cost reductions resulting from reducing renewable energy curtailment would be if utilities and others are considering adding transmission and/or distribution capacity in order to reduce such curtailments. In such cases, electrification measures could provide a cost savings to the electric system if they displace or defer any planned transmission additions. Such avoidance or deferrals would improve the comparative economics of geo-targeted electrification even more. Capturing such

Over an entire service territory, the differences in draw amount and duration of draw associated with the different charging options may be smoothed out if charging times are diversified across a large population of customers.

⁴⁷ Differences in charging efficiencies can also affect grid costs and should also be considered.

impacts in cost-effectiveness assessments of electrification measures will require:

- Identifying planned transmission investments;
- Estimating the costs and expecting timeline for such investments;
- Estimating the degree to which such investments could be deferred through increased local deployment of electrification measures and resulting reductions in renewables curtailment; and
- Assigning a value of investment deferral to each kW of added load from electrification (roughly speaking by dividing the value of investment deferral by the amount of electrification needed to produce the deferral).

10.6 Lost Revenues, Increased Revenues, and Rate Impacts

Lost revenues and potential rate impacts should not be included in cost-effectiveness analyses. Instead, DER lost revenues and rate impacts should be analyzed separately using rate, bill, and participation analyses. (See Section 2.3 and Appendix A.) In conducting BCAs, therefore, lost revenues should be identified so that they can be properly excluded from BCAs and properly included in rate, bill, and participation analyses.

In general, several key factors affect the extent to which DERs might create rate impacts:

- Increases in utility system costs will put upward pressure on rates.
- Reductions in utility system costs will put downward pressure on rates.
- *Reductions in sales* from DER resources will put upward pressure on rates.
- Increases in sales from DER resources will put downward pressure on rates.
- *Rate design* will affect the amount of lost or increased revenues created by the DER.

Electrification resources will typically increase electric utility system costs because of financial incentives or other program costs, and can increase utility system costs by increasing generation, transmission, and distribution costs (see Table 9-1). Electrification will typically reduce gas utility system costs by reducing fuel, variable O&M, and other gas costs (see Table 9-2).

Electrification will result in increased electricity sales and therefore increased electric utility revenues. The increase in electric revenues will exert downward pressure on electric rates by spreading the recovery of fixed costs over a broader base of sales. On the other hand, reduced consumption of natural gas will create upward pressure on natural gas rates because of the need to recover fixed and/or sunk costs over a smaller volume of gas sales.

Regulators and other stakeholders that are concerned about electrification impacts on gas utility rates could consider the following: (a) conduct a long-term gas utility rate, bill, and participation impact analysis, (b) seek and implement rate designs and electrification initiatives likely to mitigate equity concerns, (c) seek ways to expand customer participation, especially among low- and moderate-income customers, and (d) periodically review impacts as electrification technology deployment rates increase. Any such analyses should consider the reductions in electric utility rates as well as the increases in gas utility rates.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 182 of 302

Part IV: Benefit-Cost Analysis for Multiple DER Types

Overview

This part of the manual includes four chapters addressing different ways of combining multiple DER types, including:

- multiple on-site DER types, such as grid-integrated efficient buildings (GEBs);
- multiple DER types in a specific geographic location in the form of an NWS;
- multiple DER types across a utility service territory; and
- dynamic system planning practices that can be used to optimize DERs and alternative resources.

Each chapter includes a summary of key points, a description of how the multiple DER types might be used together, a discussion of key factors in determining benefits and costs, and guidance on how to address common challenges in determining benefits and costs.

Chapter 11 provides a case study of a GEB, with an illustration of how to consider interactive effects between DER types.

Chapter 12 provides a case study of an NWS, with an illustration of how to consider locational impacts of DER types.

Chapter 13 provides several examples of how to present BCA results to help prioritize DERs across a utility service territory.

Chapter 14 provides an overview of the evolving system planning practices, such as integrated distribution planning and integrated grid planning.

11. MULTIPLE ON-SITE DERS

In some cases, utilities, customers, and/or third-parties may seek to install multiple on-site DERs, which may take form at the building, facility, campus, or neighborhood level. This chapter describes how to holistically account for the benefits and costs of multiple on-site DERs. It identifies key factors that affect those impacts and provides guidance on how to address challenges in determining benefits or costs.

11.1 Summary of Key Points

- Multiple on-site DERs might be installed in a variety of ways:
 - On a residential level, utilities are offering programs that give incentives to adopt multiple DER types that can then be used to benefit the customer and the grid.
 - On a residential and commercial level, grid-interactive efficient buildings (GEBs) have the potential to aggregate DERs and provide grid support at scale.
 - On a community level, microgrids and smart neighborhoods also have the potential to aggregate DERs and provide grid support at scale.
- The potential benefits and costs of multiple on-site DERs will depend on the type of DERs deployed, their capabilities, locational and temporal impacts, seasonal and daily load profiles, resource ownership and control of the DERs (i.e., level of dispatchability), and interactive effects across the DERs.
- Cost-effectiveness analyses of multiple on-site DERs should account for whether the initiative is new or is based on existing resources or programs with incremental benefits and costs.
- A key factor influencing cost-effectiveness calculations is the granularity of data desired and available for analysis. This is dependent on the site, existing infrastructure investments, and level of telemetry.

11.2 Introduction

Multiple (or aggregated) on-site DERs can take place in different forms, such as at the residential, commercial, and community levels—including microgrids and smart neighborhoods. These are each described below:

Residential and Commercial: Multiple DERs

On the residential level, the adoption of multiple DERs and other smart home technologies can control load. This approach enables utilities to leverage the combined effects of residential loads that may be too small or not flexible enough to address individually. Multiple DER programs or strategies to integrate technologies include Wi-Fi-enabled thermostats, appliances, other home devices (e.g., water heaters, refrigerators, air conditioners) and advanced metering infrastructure (AMI).

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 184 of 302

Examples include EE/DR integration and expanded integrated DER offerings in the form of residential solar-plus-storage offerings, bring-your-own-thermostat (BYOT) to bring-your-own-device (BYOD) DR programs, and broader smart home technology and EV adoption.⁴⁸ These technologies provide new customer engagement strategies for utilities and expanding the capabilities for customers to control and manage their load, and opportunities to provide value to the grid.

Commercial buildings can also serve as controllable loads that can provide value to grid for utilities. Buildings can account for up to 80 percent of the peak demand on the grid, indicating that their A GEB is defined as an energy efficient building with smart technologies characterized by the active use of DERs to optimize energy use for grid services, occupant needs and preferences, and cost reductions in a continuous and integrated way (DOE 2019c).

interconnection and interaction with the grid is not only important but potentially more cost-effective than residential DER integration due to the scale of grid support that they could provide (EIA 2018a).

Grid-interactive Efficient Buildings

GEBs are an example of the optimization that can occur from multiple DER types at one site or defined location, and the grid support that can be provided by residential and commercial buildings. A GEB is defined as an energy efficient building with smart technologies characterized by the active use of DERs to optimize energy use for grid services, occupant needs and preferences, and cost reductions in a continuous and integrated way (DOE 2019c). GEBs use efficient, connected, smart, and flexible resources such as energy storage, distributed generation, DR, and EE to efficiently shed, shift, and modulate load, thereby easing constraints on the grid and providing ancillary services as an option (DOE 2019a). (See Chapter 7.) Figure 11-1 illustrates changes in building load profiles for each of the operational modes.

⁴⁸ BYOT offerings allow customers to receive credit for their pre-approved and enrolled smart thermostat devices in exchange for enabling utility or third-party control to provide grid benefits. BYOD programs offer customers opportunities to participate in DR programs with the choice of multiple controllable devices, such as a thermostat, water heater, pool pump, EV charger, and batteries.



Figure 11-1. Grid-interactive Efficient Building Flexibility Load Curves

Source: DOE 2019a. Note the gray curve represents an example baseline building load and the green, blue, purple, and orange curves show the resulting building load.

GEBs use analytics supported by sensors and controls to optimize energy use for occupant preferences, utility price signals, and available on-site DERs. For example, Figure 11-2 illustrates representative daily average load profiles for a building by comparing scenarios including energy efficiency, on-site DPV generation, and a GEB scenario featuring an optimized blend of energy efficiency, solar PV, and demand flexibility (DOE 2019a). As seen in Figure 11-2, the addition of demand flexibility (represented as load-shedding/shifting) to efficiency (green line) and distributed generation (solar PV, black dotted line) results in a flattened and reduced building net load profile (blue line), which provides grid support (DOE 2019a).





Source: DOE 2019a.

Microgrids

Microgrids are composed of a group of interconnected loads and DERs within clearly defined electrical boundaries. A microgrid can act as a single controllable entity with respect to the grid, and it can connect or disconnect from the grid to operate in both grid-connected and "island" mode.

Microgrids can provide back-up power and added reliability and resilience to a single or networked system of loads within a defined boundary. Increasing disturbances caused by severe weather events, wildfires, and natural disasters are resulting in more emphasis on multiple on-site DERs such as microgrids as potential reliability and resilience solutions, especially for critical infrastructure (DOE 2019c and NARUC 2019).

Microgrids can contain a broad range of DER types, and configurations can range from a single customer microgrid to more complex designs that serve multiple customers, such as campus or neighborhood-style microgrids:

- A single customer microgrid is located BTM or at a point of common coupling that serves one customer or building's load and acts as a single controllable entity. Customers may install microgrids to meet their own needs and/or support neighboring or system needs.
- A campus microgrid is similar to a single customer microgrid except it serves multiple buildings that can be contiguous or non-contiguous. Universities, hospitals, and military bases are common examples.
- Multi-customer microgrids can serve multiple users' loads that are on contiguous or noncontiguous properties. The ownership, operation, and/or management of the microgrid assets vary between the customers and a third-party developer. Examples include mixed-use real estate development and data centers (SEPA 2019a).

Other variables that are useful in defining a microgrid include: size or capacity of the project, types of assets included in the project, primary operating mode, point of interconnection (BTM versus FOM), public rights of way, and on-site generation versus imported electricity (SEPA 2019a).

Smart Neighborhoods, Campuses, and Districts

There is a growing opportunity to expand the optimization of DER types integrated at a defined location across a set of homes or buildings into a smart neighborhood/community, campus, or district. Smart neighborhoods/communities will often include GEB characteristics such as energy efficient systems, smart appliances, connected devices, and flexible DERs (such as DPV and storage) that can be connected as a neighborhood-style microgrid (DOE 2019c). Campuses (such as universities, hospitals, or military bases) and districts (including mixed commercial, retail, and residential buildings) also can optimize demand flexibility and multiple DERs across buildings on a particular site.

11.3 Benefits and Costs of Multiple On-Site DERs

The potential benefits and costs of aggregated DERs on site will depend on the choice of DERs deployed. Chapter 4 provides descriptions of the potential benefits and costs of DERs (including utility-system, host customer, and societal impacts). Chapters 6–10 includes the benefits and costs for EE, DR, DG, storage, and electrification, each of which may be included as part of multiple on-site DERs. Benefits and costs will depend upon many factors, including the DER types used in the home or building(s), their size and

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 187 of 302

layout, their operational patterns, and the existing capabilities at each site (i.e., the portfolio of passive and dispatchable resources).

The specific technologies utilized per site will affect their costeffectiveness. Depending on the project and DER technologies used, not all benefit or cost categories may apply. Further, to assess the cost-effectiveness per defined location, accounting for other contributing factors is important. Such factors include locational and temporal values, the potential interactive effects of different DER types, resource ownership and control, and seasonal and daily load profiles. (See Chapter 5.) Benefits and costs will depend upon many factors, including the DER types used in the home or building(s), their size and layout, their operational patterns, and the existing capabilities at each site (i.e., the portfolio of passive and dispatchable resources).

11.4 Key Factors that Affect Benefits and Costs

11.4.1 Integration and Cross-Coordination of Program Design

Integration of DERs at a defined location or into a single program has the potential to increase overall efficiencies and benefits, in addition to creating new value streams and/or value propositions.

Cost-effectiveness analysis should consider whether multiple on-site DERs are based on existing resources or programs, where incremental benefits and costs can be leveraged, or whether they include a new program or procurement. Program design cost efficiencies may be realized through reduced upfront investments, which can increase annual cost savings. This dynamic is especially prevalent when considering retrofitting current building stock to develop GEBs. Further, GEBs are most cost-effective on a larger scale where loads can be more easily smoothed to result in greater savings.

Program design factors affecting GEBs' cost-effectiveness include:

- location;
- measurement and verification;
- market integration (e.g., program administration, DER aggregation/communication, inverter interface, and cybersecurity); and
- codes and standards development.

For example, in the case of demand flexibility provided by GEBs, traditional utility program design and planning approaches may result in program savings estimates that differ greatly from forecasted planning estimates due to the time-sensitive value of grid services, as well as programs including measures/resources with long expected useful lives (10–15 years) (SEE Action 2020c). This challenge may be addressed through accounting for benefits across the full expected useful lives of resources and newer approaches to program design, such as pay-for-performance instead of one-time, upfront payments (SEE Action 2020c).

11.4.2 Determining All Host Customer Impacts

The different DERs' operational characteristics pose challenges to the utility program managers when examining the utility system but also have the potential to impact customers. (See Chapter 4 and Table 4-3, in addition to Chapters 6–9.) Determining host customer impacts, including non-energy impacts, will require additional considerations beyond those discussed earlier when analyzing multiple DERs and the scale or conditions of the site.

For example, there are a number of considerations to account for when examining host customer impacts for multiple DER types at the building level. Older and more complex buildings may have higher operating costs and may require larger upfront costs to interconnect and increase efficiency (RMI 2019). Building loads will also vary and have different use cases based on factors such as the nature of their use or type, in addition to existing HVAC systems, EV charging infrastructure, existing energy management systems, and other installed technologies. These factors will all affect customer costs and savings.

11.4.3 Existing Infrastructure Investments, Visibility, and Control

A key factor influencing BCA calculations for multiple on-site DERs is the granularity of data desired and available for analysis. Data availability depends upon the site, existing infrastructure investments, and level of telemetry. Establishing avoided cost profiles and measuring program impacts will depend on the level of granularity of available data (temporal and locational). Existing on-site technology may be limited in providing the level of data granularity desired for establishing avoided cost profiles, and measuring performance or program impacts, which can impact cost-effectiveness analysis. Such limitations will need to be understood by the jurisdiction and planned for upfront.

Improving data granularity may require different approaches/technologies, such as building energy management systems, AMI submeters, or load disaggregation. The need for technology capabilities to monitor and potentially control DERs at a site may require additional investment for the utility or customer and may ultimately impact the upfront costs and administrative fees for the site manager and the utility. Advancements in technology and broader adoption may continue to expand capabilities for obtaining granular data of multiple DERs at a site, helping to overcome this challenge over time.

11.5 Common Challenges in Determining Benefits and Costs

The installation of multiple on-site DERs is an emerging area of research, with common challenges in determining cost-effectiveness as projects and programs are implemented by utilities, customers, or third-party providers.

11.5.1 Locational and Temporal Value

The assessment of cost-effectiveness for multiple on-site DERs depends on where the DERs are located, when they generate or increase/reduce consumption, and the resulting benefits and costs. (See Sections 5.2 and 5.3.)

The locational value of DERs has significant implications for multiple on-site DERs, such as GEBs or microgrids. In the case of GEBs providing demand flexibility and the ability to adjust loads and provide local capacity savings, locational benefits are likely to be higher (SEE Action 2020c). Furthermore, locational impacts and their associated economic value should be modeled to account for transmission and generation system values (see Section 5.3). The analysis should account for the interactive effects between distribution and bulk power system impacts (SEE Action 2020c).

The temporal value of DERs also has significant implications for some of the primary benefits and costs of multiple on-site DERs: energy generation, generation capacity, transmission capacity, distribution capacity, and environmental benefits and losses, among others. For example, in a GEB with two or more

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 189 of 302

DER types providing demand flexibility, the combined load shape impacts on generation and T&D capacity demand should reasonably reflect the interaction of these resources with each other (SEE Action 2020c).

In sum, the benefits and costs of multiple DERs per site should be estimated using enough locational and temporal detail to adequately represent the DER operating patterns and consequent benefits and costs.

11.5.2 Interactive Effects

Different DER types can have interactive effects on each other, including effects on avoided costs and effects on kWh or kW impacts, and enabling other DERs. These interactive effects should be accounted for in the BCA of initiatives that promote multiple DER types per site, if such impacts are determined to be material.

- *Marginal System Costs*: When a large amount of DERs are installed in one region, they can affect the avoided costs of other DERs within that region. The best way to account for this effect is through dynamic system planning (see Chapter 14). In the absence of dynamic planning, different approaches can be used to approximate the interactive effects on avoided costs (see Section 5.4.1). In the case of initiatives promoting multiple on-site DERs, the interactive effects on marginal system costs are likely to be relatively small, until the initiatives reach high rates of deployment.
- *kWh and kW Effects*: When multiple DER types are deployed, the operation of one DER type might affect the kWh or kW impacts of other DER, depending on their types. In the case of initiatives that install multiple on-site DERs, it will be important to estimate the interactive effects that may occur due to their proximity and depending on the types of DERs (see Section 5.4.2). Evaluation efforts for these initiatives should specifically investigate how different DER types affect the kWh and kW impacts of other DERs, so that better information will be available over time.
- Enabling Effects: Some DERs can make it easier or more cost-effective to adopt other types of DERs. In the case of initiatives that install multiple on-site DERs, the initiative itself is designed to facilitate multiple DER types, and it is expected that some DERs will help make other DERs more cost-effective. For instance, in the case of GEBs, analysis can first capture major interactions between pairs of DERs, such as DR and EE, or DPV with on-site storage. Therefore, these enabling effects should be factored into the design of the initiative, and the cost-effectiveness analysis should account for that design. (See Section 5.4.3.)

11.6 Lost Revenues and Rate Impacts of Multiple On-Site DERs

Lost revenues and potential rate impacts should not be included in cost-effectiveness analyses. Instead, DER lost revenues and rate impacts should be analyzed separately using rate, bill, and participation analyses. (See Section 2.3 and Appendix A.) In conducting BCAs, therefore, lost revenues should be identified so that they can be properly excluded from BCAs and properly included in rate, bill, and participation analyses.

In sum, the benefits and costs of multiple DERs per site should be estimated using enough locational and temporal detail to adequately represent the DER operating patterns and consequent benefits and costs. In general, several key factors affect the extent to which DERs might create rate impacts:

- Increases in utility system costs will put upward pressure on rates.
- *Reductions in utility system costs* will put downward pressure on rates.
- *Reductions in sales* from DER resources will put upward pressure on rates.
- Increases in sales from DER resources will put downward pressure on rates.
- *Rate design* will affect the amount of lost or increased revenues created by the DER.

The potential rate impacts of programs offering multiple on-site DERs will depend on many factors, including the choice of DERs deployed; the magnitude of DER savings, generation, and consumption; DER deployment; DER utility system costs; and utility system avoided costs.

Rate impact analyses of programs offering multiple on-site DERs should assess the rate impacts of all DER types in combination. This holistic approach will provide the best indication of the actual rate impacts on customers. (See Chapters 6–10 and Appendix A.)

11.7 Case Study: Commercial Grid-Interactive Efficient Building

This section provides an illustrative example of a GEB to demonstrate key factors and challenges of BCA for multiple on-site DERs. Interactive effects are highlighted for discussion, and in addition, real world project examples and further resources are included for further consideration.

11.7.1 Case Study Assumptions

This illustrative case study describes a program for commercial GEBs to provide demand flexibility benefits to the grid. The goal of the program is to integrate clean resources during system peak hours and meet jurisdiction goals to reduce GHG emissions. The utility provides performance compensation for occupants in the building that provide demand flexibility during system peak periods. The GEB program is provided by an investor-owned utility regulated by a commission.⁴⁹

⁴⁹ The sources and types of information used are illustrative but are based on typical experience and data.

Commercial GEB Case Study Assumptions

DER Types: The GEB program includes the following BTM DERs:

- Energy efficiency (e.g., building envelope, HVAC, and lighting and controls)
- Demand response (e.g., direct load controls, programmable thermostats)
- Distributed photovoltaics
- Distributed storage systems

The Jurisdiction-Specific Test: The hypothetical jurisdiction's primary BCA test includes utility system impacts, host customer impacts, and GHG impacts.

Key Assumptions:

- Non-coincident Peak: the distribution feeder hosting the GEB has a peak demand that is non-coincident with the overall system peak (e.g., distribution feeder hosting the GEB peaks at 1:00-5:00pm, while system peaks from 5:00-9:00pm).
- *GHG Emissions Reduction:* the system-peak hours entail higher marginal emissions rates than the GEB, which allows the GEB to deliver GHG benefits.
- *DER Operating Profiles:* the GEB's BTM DERs operate in the following interactive ways:
 - Storage charges during the system off-peak hours, including any excess solar PV generation from the building, and discharges during system peak.
 - \circ ~ DR shifts load from on-peak system hours to off-peak system hours.
 - Solar helps meet on-site customer load, with any excess generation being used to charge the storage resource. Solar's benefits are concentrated during system off-peak hours, and its benefit to system peak diminishes as the sun sets unless paired with storage.
 - EE has a general downward trajectory on usage.

11.7.2 Case Study Benefit-Cost Analysis

Using the above JST and assumptions for this illustrative GEB program, the benefits and costs are examined below across an array of utility system, host customer, and societal impacts.

Table 11-1 presents the utility system impacts; Table 11-2 presents the host customer impacts; and Table 11-3 presents the societal impacts of this illustrative GEB program. The net impacts are presented in aggregate in Figure 11-3.

Туре	Utility System Impact	Cost or Benefit	Notes	
	Energy Generation	•	Significant generation benefits accrue in the form of:	
	Capacity	•	 Demand flexibility, as the GEB shifts building load away from system neak hours, with load-shifting also providing capacity. 	
	Environmental Compliance	•	during system peak times	
Generation	RPS/CES Compliance	•	 Excess solar from PV is used to charge the GEB's storage 	
	Market Price Effects	•	systems during system off-peak hours, and storage systems are scheduled to discharge during system-peak hours	
	Ancillary Services	•	 GEB using DR direct load control to shift load during system- peak hours to off-peak time 	
	Transmission Capacity	•	Transmission benefits accrue as the GEB's combined DERs	
Transmission	Transmission System Losses	•	reduce delivery of central generation to customers (DPV helps meet on-site customer load, with any excess generation being leveraged by the storage system).	
	Distribution Capacity	•	Not honofite for distribution imposts operus, over though pool	
Distribution	Distribution System Losses	•	demand for feeder is non-coincident with system peak. The net	
Distribution	Distribution O&M	•	benefits would be greater if the distribution feeder hosting the	
	Distribution Voltage	•	GEB has a coincident peak with the system.	
	Financial Incentives	•		
	Program Administration Costs	•		
	Utility Performance Incentives	0		
- ·	DG tariffs	0	other general utility costs (financial and performance	
General	Credit and Collection Costs	•	incentives, and administration costs) to net to an overall	
	Risk	٠	general utility cost in this example.	
	Reliability	•		
	Resilience	•		

Table 11-1. Net Benefits and Costs of GEB Case Study: Utility System Impacts

• = a benefit for this example. • = a cost for this example. • = not relevant for this example.

Table 11-2. Net Benefits and Costs of GEB Case Study: Host Customer Impacts

Туре	Host Customer Impact Cost or Benefit		Notes
	Host portion of DER costs	•	
	Host transaction costs	•	
	Interconnection fees	•	This example jurisdiction includes both host
Host Customer	Risk	•	customer costs and benefits to ensure symmetry
	Reliability	•	in treatment of impacts in the JST.
	Resilience	•	
	Host Customer NEIs	•	

• = a benefit for this example. • = a cost for this example. • = not relevant for this example.

Туре	Societal Impact	Cost or Benefit	Notes
	GHG Emissions	•	GHG emissions reductions accrue as the GEB helps offset resources with higher marginal emission rates.
	Other Environmental	0	
Societal	Resilience	0	
Impacts	Economic and Jobs	0	This example jurisdiction does not include these
	Public Health	0	applicable policies.
	Low Income: Society	0	
	Energy Security	0	

Table 11-3. Net Benefits and Costs of GEB Case Study: Societal Impacts

• = a benefit for this example. O = not relevant for this example.

Figure 11-3 presents the utility system impacts, the host customer impacts, and the GHG impacts as required in this example jurisdiction's JST. As indicated, the combination of multiple on-site DERs in this GEB example results in an overall net benefit.





11.7.3 Key Challenge: Interactive Effects

Identifying interactive effects between DERs are a common challenge in BCAs for GEBs. The primary interactive effect of GEBs is the impact that each DER has on the magnitude of kW or kWh impacts of other DERs. This illustrative GEB case study considers major enabling interactions between pairs of DERs, such as DR and EE, or solar PV with on-site storage, to aid cost-effectiveness analysis. These interactive effects are illustrated in Figure 11-4.



Figure 11-4. Example of GEB Interactive Effects

Figure 11-4 demonstrates how, in this GEB case study, accounting for interactive effects of DERs led to an increase in benefits. The left bar illustrates the sum of individual DER benefits without consideration of interactive effects. (See Section 5.4 for individual DER graphs.) This can be juxtaposed with the bar on total benefits with interactive effects (right bar), which illustrates the total benefits accrued due to the combination of DERs after accounting for interactive effects. Interactive effects can either enhance or reduce benefits, as illustrated in Figure 11-4, with distributed PV and storage as well as with EE and DR. Considerations of the interactive effects of these pairings include:

- Pairing solar PV with storage systems improves the PV's benefits by storing excess PV generation during periods of low electricity costs and/or demand and discharging from the storage device during periods of high electricity costs or demand.
- Combined DG and storage resources may influence the storage resource's dispatch pattern based on the performance of the DG resource (e.g., firming the capacity of a solar PV resource).
- Efficiency measures implemented in the building reduce the amount of demand flexibility capacity available to respond during events.
- Although not indicated in Figure 11-4, a building with DR resources (e.g., controllable thermostats, hot water heaters) could affect the dispatch parameters of a storage resource within the same building.

The text box below provides a real-world example of the dynamics of interactive effects. Additional information on GEB projects can be found in ACEEE 2019b, DOE 2019a, NEEP 2020, and SEE Action 2020b.

Portland General Electric's Smart Grid Test Bed Pilot and Demand Flexibility

Interactive effects can be further considered through a real-world example. Portland General Electric (PGE) launched its Smart Grid Test Bed pilot in 2019 to test demand flexibility, including 69 MW in summer and 77 MW in winter, to address a 2021 capacity gap (PGE 2019). The pilot aims to integrate customer-sited smart grid technologies including smart thermostats, water heaters, electric vehicle chargers, and batteries for demand flexibility. The pilot includes over 20,000 customers in three Oregon neighborhoods (served by three distribution substations) (PGE 2020; SEE Action 2020b).

The pilot is focused on testing DR programs and smart-home technology incentives to build towards future initiatives (e.g., virtual power plants), as well as to accelerate customer-hosted DER adoption (PGE 2020). PGE also coordinates with the Energy Trust of Oregon on energy efficiency and rooftop solar. On the commercial side, PGE is testing direct installation of smart thermostats, EV charging, and storage (SEE Action 2020).

The pilot includes an opt-out peak time rebate (PTR) program, which includes day-ahead notice of called events and a rebate of \$1.00 per kWh in reduced energy consumption during the event (PGE 2018; SEE Action 2020; PGE 2020). Additionally, the pilot will test how to shift customers to an opt-in load control program, utilize distribution substation-level data to inform DER modeling and distribution system planning, and explore DERs as NWSs (SEE Action 2020).

Included in its 2019 Integrated Resource Plan, PGE commissioned a comprehensive DER forecast study to enhance understanding of flexible load, including a focus on interactive effects between programs and resources (PLMA 2019). PGE plans to incorporate the analysis across the organization, including in its test bed activities. The *Distributed Resources & Flexible Load Study* (developed by Navigant/Guidehouse) included analysis on the interactive effects among resources, with interactive effects referring to "the effects of one distributed resource on the load shape of another distributed resource, beyond the simple addition of the two resources' load shapes" (Navigant/Guidehouse 2019). Figure 11-5 includes the interactive effects addressed in the study.



Figure 11-5. PGE 2019 Distributed Resources & Flexible Load Study – Interactive Effects Addressed

The study highlights the following interactive effects:

- Solar + storage modeling captured interactions between resources at a single customer site and optimized for storage charging (from on-site PV) and discharging to maximize both peak shaving and customer value from bill management. The model simulations included hourly energy flows between the PV and energy storage systems, the customer's end-uses, and the grid. Simulations also optimized operations to maximize customer value, such as considering: different rate schedules (e.g., TOU pricing), customer segments, and varied solar and storage systems configurations. Value streams for resilience, direct load control, demand charge avoidance, TOU pricing arbitrage, and available rebates were considered when optimizing host customer impacts. Assumed baseline load shape not affected by behavior change.
- Light-Duty Vehicles (LDV) and DR modeling focused on the participation of LDV in a direct-load control DR program for aggregate peak reduction.
- Pricing (TOU) and Other DER modeling focused on the influence of TOU pricing on DR, solar PV, and LDV:
 - DR modeling considered TOU participation's impact on other DR programs.
 - o DPV and storage modeling optimized storage charging and discharging to optimize customer bill savings.
 - LDV modeling assumed an average electricity cost decrease due to off-peak EV charging. Modeling also considered the impacts of TOU pricing on EVs, including on total customer cost of ownership (relative to an internal combustion engine vehicle), and the impacts of shifting charging to off-peak.

(Navigant/Guidehouse 2019 and Navigant/Guidehouse 2020)

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 196 of 302

12. NON-WIRES SOLUTIONS

Utilities and others sometimes seek to install multiple DER types in a specific geographic area for the purpose of deferring or avoiding new investments in distribution or transmission systems. This chapter describes the benefits and costs most relevant to such a non-wires solution (NWS). It identifies key factors that affect NWS impacts and provides guidance on addressing common challenges in determining NWS benefits and costs.

12.1 Summary of Key Points

- NWSs include initiatives where utilities or others seek to install multiple DER types in a specific geographic area for the purpose of deferring or avoiding new investments in distribution or transmission systems.
- Cost-effectiveness analyses of NWS initiatives should account for the specific use case of the initiative, including the transmission or distribution upgrade being deferred, the length of deferral, the mix of DERs producing the deferral, and a range of other factors.
- The ability of an NWS initiative to defer transmission or distribution costs can shift depending on changing loads and load forecasts. Much of the option value of an NWS initiative is linked to the extent to which forecasted load growth ultimately materializes.
- Locational values of T&D impacts are among the driving factors behind the NWS, and therefore should be accounted for in NWS BCAs.
- Cost-effectiveness analyses of NWS initiatives should accurately forecast customer adoption and participation because risks from not meeting requirements pose challenges to the system.
- Whether an NWS initiative is new or is based on existing resources or programs with incremental benefits and costs should be accounted for in the NWS BCA.
- Cost-effectiveness analyses of NWS initiatives should account for interactive effects of DER types, especially the interactive effects on the total kW and kWh impacts of the DERs.
- NWS initiatives may have broader impacts on the utility system—beyond the avoided T&D costs. Costeffectiveness analyses of NWS initiatives should account for all relevant impacts included in a jurisdiction's JST.

This manual uses the term "nonwires solution" to refer to all of these applications and defines NWS as "an electricity grid investment or project that uses non-traditional T&D solutions, such as distributed generation, energy storage, energy efficiency, demand response, and grid software and controls, to defer or replace the need for specific equipment upgrades, such as T&D lines or transformers, by reducing load at a substation or circuit level" (Navigant 2017).

12.2 Introduction

The concept of deploying DERs in specific geographic locations in order to defer or displace investment in transmission and/or distribution system upgrades has been referred to as: "targeted demand-side management," "geo-targeting," "non-wires alternative," "non-transmission alternative," and "non-wires solution." This manual uses the term "non-wires solution" to refer to all of these applications and defines NWS as "an electricity grid investment or project that uses non-traditional T&D solutions, such as distributed generation, energy storage, energy efficiency, demand response, and grid software and controls, to defer or replace the need for specific equipment upgrades, such as T&D lines or transformers, by reducing load at a substation or circuit level" (Navigant 2017).⁵⁰ While DERs can be located in any area and provide benefits to both the transmission and distribution grids, they are considered an NWS when specifically associated with the intentional deferral or replacement of utility T&D infrastructure upgrades.

NWSs can come in different sizes, have different deferral objectives, and utilize different DER technologies:

- *NWS Example 1: Geo-Targeted Existing EE & DR Programs:* Leveraging 300 kW EE and DR via existing customer programs that include resources such as Wi-Fi-enabled thermostats, heat pump water heaters, and window air conditioners to defer a substation feeder upgrade.
- *NWS Example 2: Large FOM Battery Storage*: Installing a 2 MW, 8 MWh battery system to defer distribution line upgrades and address thermal constraints on a feeder.
- *NWS Example 3: Combined FOM and BTM DERs:* Leveraging 52 MW of utility and customer-side resources such as EE, DR, DG (such as solar PV), and energy storage, among others, to defer a new substation upgrade.

This chapter discusses how the different characteristics of an NWS can affect its cost-effectiveness. The chapter focuses on electric NWSs, but the concepts apply equally to non-pipe solutions for gas utilities.

12.3 Benefits and Costs of Non-Wires Solutions

The potential benefits and costs of an NWS investment or project will depend on the choice of DERs deployed. (See Chapter 4.) Chapters 6–10 describe the benefits and costs for specific DERs, including EE, DR, DG, storage, and electrification, each of which may be a component of an NWS. This chapter provides guidance on how the cost-effectiveness of multiple DER types should be considered in the context of an NWS.

Depending on the NWS project and DER technology utilized, not all benefit or cost categories may apply. Additionally, as outlined in Chapter 5, conducting a BCA for an NWS project involves addressing other

⁵⁰ NWSs differ from virtual power plants (VPP) and microgrids. VPPs rely on software and advanced communication systems to aggregate, control, dispatch, plan, and optimize a suite of DERs to provide services similar to a conventional power plant. Microgrids comprise a group of interconnected loads and DERs within clearly defined electrical boundaries. A microgrid can act as a single controllable entity with respect to the grid, and it can connect or disconnect from the grid to operate in both grid-connected and "island" mode. Some NWS projects include VPPs and microgrids, which have the potential to reduce constraints on existing T&D infrastructure and help avoid the needs for system upgrades. However, a distinction can be drawn based on the purpose and goals of a project. NWSs covered in this manual refer to projects that defer or replace grid infrastructure upgrades, while VPPs and microgrids are traditionally developed for a variety of other purposes (E4TheFuture, PLMA, SEPA 2018).

contributing factors such as the characteristics of the infrastructure constraint, NWS technology characteristics, and locational and temporal values, among others.

12.4 Key Factors that Affect Benefits and Costs

12.4.1 Characteristics of Infrastructure Constraint

NWS cost-effectiveness depends on various infrastructure constraints, including the specific T&D infrastructure constraint, the size of constraint, and the season and time of constraint.

- T&D infrastructure: An NWS can be targeted to any element of the T&D system, such as a single transformer serving one large building, a substation, or a high-voltage transmission line serving a broader regional need. The specific constraint will impact cost-effectiveness: all else equal (including the cost of the NWS), the more expensive the traditional solution (per kW) the greater the benefit of the NWS deferral.⁵¹
- *Constraint size:* NWS cost-effectiveness will also depend on the size (both time duration and MW) of the specific constraint. For instance, an NWS could seek to address a limited summer peak capacity constraint or a broader regional load capacity growth forecast.
- Season and time: The season and time of the T&D peak being addressed will also impact costeffectiveness, because DERs operate at different times depending on the season and/or time of
 day. An NWS will provide greater system peak benefits if the included DERs' operational
 patterns align with the system peak because peak reduction achieved locally will also have value
 at the system level. Furthermore, an NWS can target different deferral time horizons of a T&D
 investment: short-term deferral (1–3 years), longer-term deferral (5–10 years), or complete
 displacement (at least for the foreseeable future). All else equal, the longer an NWS deferral,
 the greater the benefit. The number of years of deferral of an NWS project can also shift while
 the NWS is being deployed as forecasts of the T&D need are updated and refined (see Section
 12.5). Such changes can either move up the expected need date or push it back, where typically
 the latter is the case and allows for additional time to further refine forecasts.

12.4.2 Selected Technology Characteristics and Capabilities

NWSs can include a broad range of DER types, sizes/capacities, and locations (FOM or BTM). Diverse technologies included in an NWS will have different benefits and costs that they can simultaneously provide while addressing localized T&D constraints. For example, DR often only provides T&D and potentially system peak savings. In contrast, DERs such as storage and vehicles with V2G capabilities can provide T&D peak savings, system peak savings, and potentially economic dispatch capabilities to shift consumption from high energy cost hours to lower cost hours. Other DERs such as EE and DG can also provide energy benefits, in addition to T&D and system peak savings.

An NWS can include a single DER, but NWSs more commonly include multiple DERs. An NWS can be based on a combination of different technologies with varying predictability when it comes to their net

⁵¹ A critical component of cost-effectiveness analysis for NWS includes T&D capacity benefits, with the value of (active) deferrals per peak kW typically based on the actual deferral value of the avoided transmission or distribution project (e.g. the amount of revenue requirement that will not be collected if the wires investment is deferred for a certain number of years). There is often a minimum cost threshold for T&D projects to be considered for NWA solutions; therefore, the value of active deferrals is typically higher than passive deferrals.

loading. When these DERs are combined as an integrated solution, it will be important to account for the performance of different DERs in relation to other inputs such as customer participation (in the case of DR), weather (e.g., solar PV performance), and resource ownership and control (e.g., storage systems and their operational profile). (See Chapters 6–10.) Beyond understanding each DER that may be included in an NWS portfolio, it is important to understand how the portfolio will perform in aggregate in response to the needs of the T&D constraint.

12.4.3 Existing Programs or Procurement

In some cases, cost-effectiveness analyses must consider whether an NWS portfolio includes DER types that are based on an existing program or a new program. If a DER is based on existing programs (such as existing EE and DR programs) that can be leveraged to alleviate localized T&D constraints, the BCA should include the NWS impacts that are *incremental* to the existing programs. For new NWS programs, the BCA should include the *entire* cost of the NWS program.

If there are existing system-wide EE or DR programs included in an NWS, for example, it is easier and less expensive (all else equal) to leverage existing trained contractors with technology and customer expertise, relationships with trade allies, marketing materials, etc., to further recruit customers into an existing program. However, it is also important to consider the fact that the impacts associated with the existing programs will be experienced regardless of the NWS. The net effect of these two factors will depend on the DER initiative, its historical deployment rates in the geo-targeted area, and a range of other factors. (See Section 12.5 for additional discussion on customer adoption and participation.)

12.4.4 Accounting for Other Electric Utility System and Non-Electric Impacts

In addition to T&D deferral benefits, NWS cost-effectiveness should account for other electric utility system and non-electric impacts, to the extent they are relevant to a jurisdiction given its applicable energy policy goals (see Chapters 2 and 3). Many DERs deployed as part of an NWS will not only affect the localized T&D constraint targeted by the NWS, but also costs associated with other aspects of the electric system. All such electric system impacts should be included in assessments of the cost-effectiveness of an NWS (see Chapter 4).

Many DERS deployed as part of an NWS will also have impacts beyond the electric system, including host customer consumption of other fuels, other non-energy impacts experienced by host customers and a variety of societal impacts such as changes in GHG emissions. All such non-electric system impacts that are included in the jurisdiction's primary cost-effectiveness test should also be reflected in cost-effectiveness assessments of NWSs (see Chapter 4). (See Chapters 6–10 for guidance on individual DERs.)

12.5 Common Challenges in Determining Benefits and Costs

12.5.1 Determining Locational and Temporal Value of DERs in an NWS

DER benefits and costs depend significantly on where they are located within the system and when they generate or increase/reduce consumption. The locational value of DERs has significant implications for T&D impacts, and therefore, is a primary criterion when planning, implementing, and evaluating an NWS portfolio. (See Sections 5.2–5.3.)

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 200 of 302

Different approaches exist today for calculating locational value, which include a more site-specific approach versus a more system-wide approach. The temporal value of DERs also has implications for several important impacts, including: energy, generation capacity, transmission capacity, distribution capacity, and environmental impacts. (See Sections 5.2 and 5.3.)

Utilities and others will need to consider the granularity of available data (temporal and locational) for the purpose of properly assessing and assigning a numerical value to a location under evaluation. The

The locational value of DERs has significant implications for T&D impacts, and therefore, is a primary criterion when planning, implementing, and evaluating an NWS portfolio.

accuracy and confidence of determining locational and temporal values is dependent on the data used to inform the analysis. For many utilities, the ability to gather and leverage granular data may be limited by existing infrastructure, software capabilities, and devices along the grid.

In sum, the benefits and costs of NWSs should be estimated using sufficient locational and temporal detail to adequately represent the DER operating patterns and consequent benefits and costs.

12.5.2 Accounting for Option Value and Determining Project Lifetimes

Due to the nature of T&D deferrals and uncertainty of load forecasts, it can be challenging to identify the number of years that T&D investments will be deferred by an NWS. This deferral can shift over time depending on changing load forecasts independently of the NWS (e.g., due to economic variability). For example, if the proposed solution is planned for a 5-year deferral, an unexpected benefit may arise if the proposed solution ultimately provides 10 years of deferral, and vice versa. Much of the option value is linked to load forecasting and the extent to which forecasted load growth ultimately materializes.

Load forecasts tend to be more uncertain and less accurate the further out they go in time. In these situations, DERs (such as EE or DR) deployed in an NWS can provide the opportunity to recalibrate load forecasts over time. This occurs due to the ability to incrementally implement an NWS portfolio (e.g., a project manager can implement EE and DR first, followed by energy storage, or recruit participants gradually for participation in a DR program) without high upfront deployment costs. As load forecasts decrease or increase in relation to the T&D capacity constraint, DER deployment can be adjusted. This modular nature of DER deployment can buy time to reassess needs (and options for meeting those needs) in a way that is often not possible with less modular capital investments in the T&D system. This flexibility provides value to the utility system.

12.5.3 Interactive Effects

Different DER types can have interactive effects on each other, including effects on avoided costs and effects on kWh or kW impacts, and enabling other DERs. It is important that these interactive effects be accounted for when assessing the cost-effectiveness of NWS initiatives, if such impacts are determined to be material. In theory, the benefits and costs of an NWS would not be different than the benefits and costs of the multiple DERs that compose the NWS, with interactive effects considered. The objectives of an NWS—to defer or displace investment in transmission and/or distribution system upgrades—require clear definition of both how DERs will be operated and the deferred or avoided infrastructure investment cost.

• *Marginal System Costs.* When large numbers of DERs are installed in one region, they can affect the avoided costs of other DERs within that region. The best way to account for this effect is through dynamic system planning (see Chapter 14). In the absence of dynamic planning,

different approaches can be used to approximate the interactive effects on avoided costs (see Section 5.4.1). In the case of NWS initiatives, the interactive effects on avoided costs are likely to be small, unless and until the initiatives reach high rates of deployment.

- *kWh and kW Effects*. When multiple DER types are deployed, the operation of one DER type might affect the kWh or kW impacts of other DER types. In the case of NWS initiatives, it is possible that the operation of one DER will affect the kWh or kW impacts of another type because of their proximity. This is especially true for storage resources that could be dispatched in a way that accounts for the operation of other DERs. For NWS initiatives, therefore, it may be important to estimate these interactive effects. Evaluation efforts for these initiatives should specifically investigate how different DER types affect the kWh and kW impacts of other DERs, so that better information will be available over time. (See Section 5.4.2.)
- Enabling Effects. Some DERs can make it easier or more cost-effective to adopt other types of DERs. In the case of NWS initiatives, the initiative itself is designed to include multiple DER types, and it is expected that some DERs will help make other DERs more cost-effective. Therefore, these enabling effects should be factored into the design of NWS initiatives, and the cost-effectiveness analysis should account for that design. (See Section 5.4.3.)

12.5.4 Evaluating and Measuring NWS Impacts

Identifying both baseline forecasts and incremental impacts of NWSs can sometimes be challenging. The forecast used to identify the T&D capacity need must be the same as the forecast from which geo-targeted DER deployment increases will be measured for costs.

For example, consider a substation upgrade that was forecast to be needed five years from now. The forecast was based on historical localized load growth during a time when system-wide EE programs were producing 1.0 percent peak savings per year. However, the utility is now planning for 1.5 percent new peak savings per year from system-wide EE programs (and any additional geo-targeted EE is measured relative to that new 1.5 percent level). In this case, the resources needed to defer the substation upgrade would inappropriately appear to be greater and more extensive (and therefore more expensive) than they should be because the forecast need will have omitted system savings of 0.5 percent per year.

Additionally, in order to assign benefits and costs to the NWS, the incremental impacts of geo-targeted DERs must be measured relative to expected system-wide DER initiatives. Measurement options to account for this include strategies to ensure NWS DER adoption is additive and properly counted as incremental. An example strategy would be comparing the anticipated participation rates of the NWS in the geo-targeted area to account for historical DER participation rates in the same area with existing system-wide programs.

12.5.5 Accounting for System Reliability and Risk

Uncertainty of DER performance poses potential risk to the utility system, particularly when the DERs are used to defer a specific transmission or distribution asset. This risk is a countervailing effect to option value (see Section 12.5.1). If DERs do not perform when needed along the grid and consequently do not alleviate the identified grid constraint, this poses risk to the grid operator and the customers served. The variability and intermittent nature of DERs can add greater uncertainty into load forecasting and operation of the system. Utilities may thus perceive a risk when relying upon DERs to meet a system need compared to a traditional solution. For example, as discussed above, an NWS comprising
customer-side load reductions may fail if customer recruitment and participation levels are not met. Likewise, a storage system may have construction delays or not perform at the required level to meet the NWS need. If an NWS does not perform as needed, the utility may face reliability penalties. When assessing the risk impacts of a DER, the net risk impacts should be applied, accounting for both the risk benefits and risk costs.

This risk of non-performing DERs can be mitigated as follows:

- One strategy entails modular deployment of DERs with continuous check-ins on progress and whether those DERs are meeting goals.
- Another strategy includes a "no-regrets" strategy, by deploying DERs that have numerous other benefits beyond addressing grid constraints.

The assessment of risk assumed in a BCA will depend on the approach used by the NWS planner and operator.

When NWSs are based on existing or new customer-sited DER programs, it is critical to accurately forecast customer adoption. Some customer-sited DERs (such as DR, BTM storage, and V2G) for NWSs require customers and operators to understand their dispatch commitments, but risks from customers failing performance requirements are difficult to predict and quantify. For example, customers participating in DR programs are often incentivized to participate, while not as many customers receive penalties for failing to meet commitments. In the case of an NWS depending on customer resources to perform during critical times, the costs for not performing may be much higher to the utility in the form of outage duration and frequency cost impacts.

Another option is to establish penalties in the developers' contracts for DER procurement. A similar option is to establish utility performance incentives to ensure that the utility will successfully procure the DERs or compensate customers for what is not procured. These mechanisms to address risk should be incorporated into the design of the NWS program and into the BCA for that program.

12.6 Lost Revenues and Rate Impacts for Non-Wires Solutions

Lost revenues and potential rate impacts should not be included in cost-effectiveness analyses. Instead, DER lost revenues and rate impacts should be analyzed separately using rate, bill, and participation analyses. (See Section 2.3 and Appendix A.) In conducting BCAs, therefore, lost revenues should be identified so that they can be properly excluded from BCAs and properly included in rate, bill, and participation analyses.

In general, several key factors affect the extent to which DERs might create rate impacts:

- Increases in utility system costs will put upward pressure on rates.
- *Reductions in utility system costs* will put downward pressure on rates.
- *Reductions in sales* from DER resources will put upward pressure on rates.
- Increases in sales from DER resources will put downward pressure on rates.
- *Rate design* will affect the amount of lost or increased revenues created by the DER.

The potential rate impacts of NWS initiatives will depend on many factors, including the choice of DERs deployed; the magnitude of DER savings, generation, and consumption; the DER deployment; the DER utility system costs; and the utility system benefits.

Rate impact analyses of NWS initiatives should assess the rate impacts of all DER types in combination. This holistic approach will provide the best indication of the actual rate impacts on customers. (See Chapters 6–10 and Appendix A.)

12.7 Case Study: Non-Wires Solution

The objective of this case study is to demonstrate the cost-effectiveness analysis of multiple DER types in a specific geographic area, deployed for the purpose of deferring or avoiding new investments in distribution or transmission systems.

12.7.1 Case Study Assumptions

In this illustrative scenario, a utility is facing the need to upgrade significant infrastructure due to distribution capacity constraints identified on parts of the system in a geographic area. The utility plans to integrate DERs to serve as an NWS in place of the infrastructure upgrade. The constraint area is densely populated with residential and commercial customers. The geographic area is in an investor-owned utility's service territory regulated by a commission.⁵²

Non-Wires Solution Case Study Assumptions

DER Types: The NWS plan includes the following BTM DERs in residential and commercial buildings:

- Energy efficiency measures (e.g., lighting and controls)
- Demand response (e.g., Wi-Fi-enabled thermostats)
- Distributed photovoltaics
- Distributed storage systems

The Jurisdiction-Specific Test: The hypothetical jurisdiction's primary BCA test includes utility system impacts, host customer impacts, and GHG impacts.

Key Assumptions:

- *Non-Coincident Peak:* The distribution need is non-coincident with the overall system peak (e.g., the constrained distribution feeder peaks from 1:00–5:00 pm, while system peaks from 5:00–9:00 pm).
- *GHG Emissions Reduction:* The system-peak hours entail higher marginal emissions rates than the NWS, which allows the NWS to deliver GHG benefits.
- DER Operating Profiles: The NWS DERs operate in the following ways:
 - \circ $\,$ All DERs are operated to reduce the distribution peak, and some can reduce the system peak as well.
 - Storage discharges during the distribution peak hours and charges during the system off-peak hours.
 - DR reduces demand during distribution peak periods and/or shifts load from distribution peak periods to system off-peak periods.
 - Distributed PV resources generate during distribution peak periods and during a portion of system peak periods.
 - \circ EE helps to reduce demand during distribution peak periods, as well as system peak periods.

⁵² The sources and types of information used are hypothetical/illustrative but are based on actual/typical experience and data.

12.7.2 Case Study Benefit-Cost Analysis

Using the above JST and assumptions for this illustrative NWS program, the benefits and costs are examined below across an array of utility system, host customer, and societal impacts.

Table 12-1 presents the utility system impacts; Table 12-2 presents the host customer impacts; and Table 12-3 presents the societal impacts of this illustrative NWS program. The net impacts are presented in aggregate in Figure 12-1.

Туре	Utility System Impact	Cost or Benefit	Notes
	Energy Generation	•	Energy generation benefits occur as a result of the
	Capacity	•	reduced energy consumption due to the DERs.
	Environmental Compliance	•	includes EE savings and DPV output that overlap with
Generation	RPS/CES Compliance	•	system generation peak.
	Market Price Effects	•	primarily to reduce distribution system peaks, and
	Ancillary Services	•	therefore might provide less generation capacity benefit than other combinations of DERs would.
Transmission	Transmission Capacity	•	Some transmission benefits result from the reduced
Transmission	Transmission System Losses	•	delivery of central generation to customers.
	Distribution Capacity	•	
Distribution	Distribution System Losses	•	The largest benefit of the NWS is the direct benefits of geo-targeting DERs (operating at the necessary time) to
Distribution	Distribution O&M	•	alleviate distribution constraints.
	Distribution Voltage	•	
	Financial Incentives	•	
	Program Administration Costs	•	
	Utility Performance Incentives	0	
	DG tariffs	0	Benefits from reliability, resilience, and risk are outweighed by other general utility costs (financial and
General	Credit and Collection Costs	•	performance incentives, and administration costs) to net
	Risk	•	to an overall general utility cost in this example.
	Reliability	•	
	Resilience	•	

Table 12-1. Net Benefits and Costs of NWS Case Study: Utility System Impacts

• = typically a benefit for this example; • = typically a cost for this example; O = not relevant for this example.

Туре	Host Customer Impact	Cost or Benefit	Notes
	Host portion of DER costs	•	
	Host transaction costs	•	
	Interconnection fees	•	
Llost Customer	Risk	•	This example jurisdiction includes both host
Host Customer	Reliability	•	in treatment of impacts in the JST.
	Resilience	•	
	Host Customer NEIs	•	
	Low-income NEIs	•	

Table 12-2. Net benefits and Costs of NWS Case Study: Host Customer Impacts

• = typically a benefit for this example; • = typically a cost for this example; O = not relevant for this example.

Туре	Societal Impact	Cost or Benefit	Notes
Societal	GHG Emissions	•	GHG emissions reductions accrue as the NWS helps offset resources with higher marginal emission rates.
	Other Environmental	0	
	Resilience	0	
Impacts	Economic and Jobs	0	This example jurisdiction does not include these
	Public Health	0	societal impacts in its JST based on its review of its applicable policies.
	Low Income: Society	0	
	Energy Security	0	

• = typically a benefit for this example; • = typically a cost for this example; O = not relevant for this example.

Figure 12-1 presents the utility system costs and benefits, low-income customer benefits, and GHG impacts, as required by this example jurisdiction's JST. As indicated, the T&D benefits represent a large portion of the benefits in this example, and this NWS initiative will result in net benefits.

Figure 12-1. Example of NWS Cost-Effectiveness



In this case study, locational value plays a central role in the cost-effectiveness of an NWS, as represented by the significant T&D benefits in Figure 12-2. The assessment of NWS cost-effectiveness depends on where the DERs are located, when they provide services, and the resulting benefits and costs.

Figure 12-2 illustrates the effect that locational benefits of DERs can have on this hypothetical NWS. It presents (a) the benefits of each DER type without accounting for locational impacts (i.e., assuming system average T&D benefits); (b) the locational benefits associated with this NWS initiative; and (c) and the total benefits after accounting for locational impacts. The DR and distributed storage resources would be expected to have the greatest locational benefits because they can be operated at the times of distribution and transmission system peaks.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 207 of 302



Figure 12-2. Example of NWS Locational Benefits

The text box below provides a real-world example of the dynamics of interactive effects. Additional information on NWSs can be found in E4TheFuture, PLMA, SEPA 2018.

EXAMPLE PROJECT: NATIONAL GRID – TIVERTON NWA PILOT

In 2012, National Grid launched the Tiverton NWA Pilot to defer a \$2.9 million, six-year feeder project with expectations of cumulatively meeting a 1 MW goal. This project also sought to test whether geographically targeted EE and DR resources (including Wi-Fi thermostats, heat pump water heaters, and window air conditioners) could defer the needs for a new substation feeder upgrade serving 5,200 majority residential customers in two communities in Rhode Island. The proposed upgrade was necessary to address distribution grid constraints due to summer weather-induced peaking.

The Tiverton NWA Pilot deferred the \$2.9 million feeder project over five years; however, the project was not able to fully realize the 1 MW summer load reduction goal. As noted below, the NWA pilot achieved an overall benefit-cost ratio of 1.40, with each year proving to be cost-effective (except for 2018, which was previously identified as the post-pilot final evaluation for related costs only). The benefit-cost calculations were calculated using the Total Resource Cost test.

The Tiverton NWS pilot example explores key factors and challenges discussed in the chapter above, including option value, load forecasts, and customer adoption. Due to slower than forecasted load growth, as well as cooler 2017 summer temperatures, the substation upgrade was deferred beyond the original date of 2017. National Grid continues to monitor loading on the feeder.

System Reliability Procurement (SRP) - Tiverton/Little Compton								
Summary of Cost Effectiveness (\$000)								
	2012 2013 2014 2015 2016 2017 2018 Overall							Overall
BENEFITS	\$179.0	\$1,325.4	\$1,033.3	\$1,281.1	\$687.7	\$568.0	\$0.0	\$5,074.6
Focused Energy Efficiency Benefits	\$90.2	\$1,015.1	\$716.7	\$1,024.8	\$435.0	\$66.94	\$0.0	\$3,348.7
SRP Energy Efficiency Benefits	\$88.8	\$310.4	\$136.8	\$78.0	\$88.1	\$341.6	\$0.0	\$1,043.7
Demand Reduction Benefits	\$0.0	\$0.0	\$5.6	\$6.8	\$5.3	\$11.3	\$0.0	\$28.9
Deferral Benefits	\$0.0	\$0.0	\$174.2	\$171.5	\$171.5	\$148.2	\$0.0	\$635.3
COSTS	\$133.4	\$672.4	\$569.3	\$1,029.4	\$611.1	\$510.9	\$90.8	\$3,617.4
Focused Energy Costs	\$46.6	\$331.1	\$195.8	\$529.3	\$280.1	\$281.3	\$0.0	\$1,664.1
System Reliability Procurement Costs	\$86.8	\$341.3	\$373.5	\$500.2	\$331.0	\$229.6	\$90.8	\$1,953.3
BENEFIT/COST RATIO	1.34	1.97	1.81	1.24	1.13	1.11	-	1.40

Summary of Cost-Effectiveness for the Tiverton NWS Pilot Project

NOTES:

• Focused EE benefits include the NPV (over the life of these measures) of all TRC benefits associated with EE measures installed in that year that are being focused to the Tiverton/Little Compton area.

SRP EE benefits include all TRC benefits associated with EE measures in each year that would have been installed as part of the statewide EE programs.

o DR benefits represent the energy and capacity benefits associated with the demand reduction events projected to occur in each year.

o Deferral benefits are the NPV benefits associated with deferring the wires project (substation upgrade) for a given year in \$2014.

- EE costs include PP&A, Marketing, STAT, Incentives, Evaluation and Participant Costs associated with statewide levels of EE that have been focused to the Tiverton/Little Compton area. For the purposes of this analysis, they are derived from the planned ¢/Lifetime kWh in Attachment 5, Table E-5 of each year's EEPP in the SF EnergyWise and Small Business Direct Install program. These are the programs through which measures in the SRP pilot will be offered.
- SRP costs represent the SRPP budget which is separate from the statewide EEPP budget, as well as SRP participant costs. The SRP budget includes PP&A, Marketing, Incentives, STAT and Evaluation. All benefits and costs are in \$current year except for the deferral benefits. 2012-2017 numbers have been updated to reflect year-end data. 2018 numbers reflect year-end projections.

Source: E4TheFuture, PLMA, SEPA 2018, based on data from National Grid (The Narragansett Electric Company), System Reliability Procurement 2019 Report, October 2018.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 209 of 302

13. SYSTEM-WIDE DER PORTFOLIOS

This chapter provides guidance on how to analyze a portfolio of multiple DER types across a utility service territory, including guidance on how to prioritize across DERs.

13.1 Summary of Key Points

- In analyzing portfolios of multiple DER types, it is important to first establish a single primary cost-effectiveness test that can be used for all DER types.
- In analyzing portfolios of multiple DER types, it is important to account for interactive effects between multiple DER types, including enabling effects, savings effects, and avoided cost effects.
- In analyzing portfolios of multiple DER types, it is useful to articulate the jurisdiction's DER planning objectives, for example: implement all cost-effective DERs; implement the lowest-cost DERs; maximize capacity benefits from DERs; encourage a diverse range of DER technologies; encourage customer equity; achieve GHG or electrification goals at lowest cost; and avoid unreasonable rate impacts.
- In analyzing portfolios of multiple DER types, it can be useful to present the BCA results in ways that facilitate comparison across DER types. For example:
 - DERs can be ranked by benefit-cost ratios or net benefits to indicate the most cost-effective DERs.
 - Levelized DER costs can be used to directly and consistently compare costs across different DER types.
 - Levelized net cost curves can be used to compare and prioritize DERs according to key parameters such as \$/ton GHG reduced.
 - Multiple cost-effectiveness tests, in addition to the JST, can be used to provide additional information when analyzing portfolios of multiple DER types.
- Any analysis of the rate impacts from portfolios of multiple DER types should assess the rate, bill, and participation impacts of all DER types in combination.

13.2 Introduction

Chapters 6 through 10 discuss how DER types can be assessed in isolation using *single-DER analysis*. This chapter refers to how to assess all DER types within a jurisdiction or service territory using *multiple-DER analysis*. Chapter 14 discusses how IDP can be used to optimize multiple DERs and supply-side resources.

A jurisdiction's primary cost-effectiveness test should be designed to answer the key question: *Which DERs have benefits that exceed costs and therefore merit utility funding or support on behalf of customers?* (See Chapter 3.) Once this universe of cost-effective DERs has been defined and identified,

some jurisdictions may wish to address a secondary question: Which of those cost-effective DERs should be funded or supported by a utility on behalf of customers? This chapter provides guidance on how to address this second question.

13.3 Consistent Cost-Effectiveness Tests

When analyzing a portfolio of multiple DER types across a utility service territory, it is especially important to apply consistent costeffectiveness principles, methodologies, and assumptions across all DER types. Otherwise, utilities risk over-investing in some DER types and under-investing in others, resulting in increased costs When analyzing the costeffectiveness of a portfolio of DERs across a utility service territory, it is helpful to begin by articulating the jurisdiction's DER planning objectives. This allows regulators, utilities, and others to identify the criteria that will be used to prioritize across different DER types.

for utility customers and missed opportunities for achieving applicable policy goals.

The best way to ensure consistency across DER types is to use the same primary cost-effectiveness test for all types. A single primary test for all DERs can be developed using the NSPM BCA Framework following the same steps used to develop a primary test for a single DER type. (See Section 3.4.)

In some cases, STEP 1 of the NSPM BCA Framework—*Articulate Applicable Policy Goals*—may be challenging to apply where jurisdictions have different policy goals for different DER types. This can make it difficult to establish a single primary test for all DER types. In these instances, the primary DER test could be based on the narrowest set of policy goals, the broadest set of policy goals, or something in between. (See Section 3.5.3.)

If a jurisdiction is unable to develop a single primary test to evaluate to all DERs, secondary tests can help promote consistency across DERs. Several options are available for secondary tests. (See Section 3.5.3, Chapter 13, and Appendix D.)

13.4 Interactive Effects

Different DER types can have interactive effects on each other, including effects on avoided costs and effects on kWh or kW impacts, and enabling other DERs. These interactive effects should be accounted for when assessing the cost-effectiveness of initiatives that promote multiple DER types, if such impacts are determined to be material.

- *Marginal System Costs:* When a large number of DERs are installed in one region, they can affect the avoided costs of other DERs within that region. The best way to do account for this effect is through IDP (see Chapter 14). In the absence of IDP, different approaches can be used to approximate the interactive effects on avoided costs (see Section 5.4.1). In the case of multiple DER types across a service territory, the interactive effects on avoided costs could be significant, especially under high rates of DER deployment.
- *kWh and kW Effects:* When multiple DER types are deployed, the operation of one DER type might affect the kWh or kW impacts of other DER types. In the case of multiple DER types across a service territory, it is possible that the operation of one DER will affect the kWh or kW impacts of another type, especially in the case of distributed storage resources. In these cases, it will be important to estimate these interactive effects (see Section 5.4.2). Evaluation efforts for these initiatives should specifically investigate how different DER types affect the kWh and kW impacts of other DERs, so that better information will be available over time.

• Enabling Effects: Some DERs can make it easier or more cost-effective to adopt other types of DERs. In the case of multiple DER types across a service territory, it is unlikely that some DERs will help make other DERs more cost-effective. This effect is most likely to occur for multiple DERs per site or for NWS applications. If there are any enabling effects, however, they should be accounted for in the cost-effectiveness analysis (see Section 5.4.3).

13.5 DER Planning Objectives

13.5.1 DER Planning Objectives

When analyzing the cost-effectiveness of a portfolio of DERs across a utility service territory, it is helpful to begin by articulating the jurisdiction's DER planning objectives. This allows regulators, utilities, and others to identify the criteria that will be used to prioritize across different DER types.⁵³

In general, a jurisdiction's policy goals should inform DER planning objectives. Planning objectives are different from policy goals, however, in that they give priority to certain goals. For example, two jurisdictions might have a policy goal of reducing GHG emissions, but one jurisdiction might decide to make decarbonization a planning objective while another one might not. The jurisdiction with decarbonization as a planning objective might choose to prioritize DERs on the basis of GHG reduction potential (for example, by considering levelized \$/ton GHG reduced).

There are many DER objectives that jurisdictions could use for planning purposes. In those cases, it might be useful to revisit those planning objectives in light of evolving goals, new DER technologies and services, and other developments in the electricity and gas industries.

One DER planning objective would be to implement all cost-effective DERs. However, some regulators might be cautious about implementing all cost-effective DERs because of budget constraints, rate impacts, or other concerns. In these cases, alternative planning objectives can be used to design system-wide DER portfolios. Examples of potential planning objectives include:

- Implement the most cost-effective DERs.
- Encourage a diverse range of DER technologies.
- Encourage customer equity.
- Achieve GHG goals at lowest cost.
- Avoid unreasonable rate impacts.
- Achieve multiple planning objectives.

The following sections offer guidance on how to present and evaluate the BCA results in order to identify those DERs that best meet these planning objectives.

13.5.2 Objective: Implement the Most Cost-Effective DERs

Jurisdictions with a planning objective to implement the most cost-effective DERs can do so by maximizing either net benefits or benefit-cost ratios of the primary DER cost-effectiveness test for the

⁵³ In general, DER planning objectives should be consistent with the planning objectives applied to other utility resources.

jurisdiction. Figure 13-1 shows the cost-effectiveness results for a set of illustrative DERs, in terms of net benefits.⁵⁴



Figure 13-1. Example DERs Sorted by Net Benefits

These results could be used to prioritize across DERs and select the most cost-effective ones. For example, utilities or others could accept the most cost-effective DERs, in terms of either net benefits or benefit-cost ratios, until a certain implementation budget cap is reached. As another example, utilities or others could accept the most cost-effective DERs, in terms of either net benefit-cost ratios, until a certain rate impact cap is reached.⁵⁵

If net benefits are used to determine the most cost-effective DERs, then this will result in the largest amount of net benefits. If, instead, benefit-cost ratios are used to determine the most cost-effective

Source: Appendix D.

⁵⁴ The DERs and cost-effectiveness results presented in these figures are purely illustrative and not based on specific DERs in a specific jurisdiction. Actual cost-effectiveness results could be significantly different from those presented here. In addition, actual results will differ depending upon the cost-effectiveness test used. Further, some DER types are bundled together to keep the figures from being too complex. (See Appendix D.)

⁵⁵ After ranking the DER BCA results in this way, one might draw the conclusion that all of the most cost-effective DER (FOM DG in this example) should be acquired before beginning to acquire the next most cost-effective DER (large C&I in this example). However, this approach would undermine other DER planning objectives, such as encouraging customer equity and encouraging a diverse range of DER types and technologies. It would also run counter to conventional practices, which include acquiring multiple DER types in parallel.

DERs, then this will result in implementing those DERs that maximize the benefits for each dollar spent. There may be instances where the result is the same or similar.

13.5.3 Objective: Achieve GHG Goals at Lowest Cost

Jurisdictions with a planning objective to achieve GHG goals can do so by prioritizing the DERs that reduce GHG emissions at the lowest cost. This objective might be appropriate in jurisdictions that have clearly defined GHG reduction targets.

Net cost curves allow for a ranking of DERs according to what they cost to achieve a particular benefit (See Appendix D). A GHG net cost curve can be used to identify the lowest cost DERs available for reducing GHG emissions.⁵⁶

Figure 13-2 presents an example GHG net cost curve for an illustrative set of DERs.⁵⁷ Net cost curves such as this are created with the following steps:

- Each DER's costs are put into levelized terms. Levelized costs include (a) the costs of the DER over its economic operating life, amortized over that lifetime and discounted back to the first year, divided by (b) the total lifetime energy produced.
- Each DER's benefits are also put into levelized terms. For the purpose of making a GHG cost curve, the GHG benefits are excluded from these levelized benefits. This allows for the presentation of net levelized costs required to achieve those GHG benefits (in \$/ton GHG).
- The net levelized cost is determined by subtracting the levelized costs from the levelized benefits.
- The DERs are ordered from lowest net levelized cost to highest.
- The vertical axis presents the net levelized cost of each DER (in \$/ton GHG).
- The horizontal axis presents the amount of GHG savings (in tons GHG) from each DER.

Appendix D describes levelized costs, the uses and limitations of cost curves, and the illustrative DERs used to create this figure.

The DERs whose net costs are negative in Figure 13-2 represent those DERs that are cost-effective without accounting for GHG emissions. The DERs whose net costs are positive represent those that are not cost-effective without accounting for GHG emissions, but that might be after accounting for these emissions.

⁵⁶ For a useful example of levelized net cost curves, see McKinsey 2013.

⁵⁷ The DERs and cost-effectiveness results presented in this figure are purely illustrative and not based on specific DERs in a specific jurisdiction. Actual cost-effectiveness results could be significantly different from those presented here. In addition, actual results will differ depending upon the cost-effectiveness test used. Further, some DER types are bundled together to keep the figures from being too complex. (See Appendix D.)



Figure 13-2. GHG Reduction Net Cost Curve

Source: These are hypothetical examples. All data used here are discussed further in Appendix D.

Net cost curves such as this one can be used for several purposes. For example:

- They can be used to prioritize DERs according to those that will reduce GHG emissions at the lowest cost. The leftmost DERs can be prioritized over those on the right.
- They can be used to determine which are the lowest-cost DERs available to meet a particular GHG target. Figure 13-2 includes a dashed vertical line indicating the relevant GHG target for this hypothetical jurisdiction. The DERs to the left of that line represent the lowest cost options for meeting this GHG target.
- They can be used to identify the marginal net cost of complying with a specific target. Figure 13-2 includes a dotted horizontal line that indicates the marginal resource needed to meet the relevant GHG target for this hypothetical jurisdiction. In this example, the marginal resource is EV initiatives for public transportation. As indicated in the previous figure, the cost of this DER is roughly 80 \$/ton GHG. This marginal cost could be used as an input to the BCA, where the value in \$/tons is used as the cost of achieving the jurisdiction's GHG target.

13.5.4 Objective: Encourage Customer Equity

Jurisdictions with a planning objective to encourage customer equity can do so by giving priority to those DERs that reach a broad range of customers, serve multiple customer types, or serve otherwise hard-to-reach customers.

Examples of DERs that might be prioritized over others in order to promote customer equity include:

- Low-income EE programs that might be less cost-effective than other DERs but provide important equity benefits to low-income customers.
- DR programs for residential customers to be consistent with similar programs for commercial and industrial customers, even if the former costs more than the latter.
- DPV initiatives such as community solar projects that allow customers to enjoy the benefits of DPV even if they do not own or occupy a building that can support DPV technologies. Such initiatives could also be designed to serve low-income or other underserved customers.
- Locating EV chargers in economically distressed areas or in public places that are not well-served by privately funded EV chargers.

13.5.5 Objective: Encourage a Diverse Range of DERs

Some regulators and others might prefer to encourage a diverse range of DER types based on the logic that all DER types deserve some utility support because they all contribute benefits in different ways and there is value in promoting a diversity of technologies. Diversity across technologies can also help reduce system risk.

In this case, regulators might decide to support a minimum amount of each type of DER. This could be achieved by sorting the DER types by net benefits or benefit-cost ratios, as indicated in Figure 13-1, and selecting the lowest cost options for each type of DER. This might result in utilities implementing some DER types that are less cost-effective than other DER types that are not implemented.

Table 13-1 presents one example of how such a portfolio might be developed. It presents five DER types and several DERs within each type.⁵⁸ The DERs are sorted by type, and for each DER type the results are sorted by benefit-cost ratios. In this hypothetical jurisdiction, the planning objective is to support the most cost-effective DERs while encouraging a range of DER technologies. This DER portfolio is designed to include those DERs that have a benefit-cost ratio of 1.6 or greater or are needed to support DER technology diversity.⁵⁹ As indicated with green shading in Table 13-1, several DERs are included in the portfolio because they support technology diversity, despite having benefit-cost ratios less than the planning threshold.

⁵⁸ The DERs and cost-effectiveness results presented in these figures are purely illustrative and not based on specific DERs in a specific jurisdiction. Actual cost-effectiveness results could be significantly different from those presented here. In addition, actual results will differ depending upon the cost-effectiveness test used. Further, some DER types are bundled together to keep the figures from being too complex. (See Appendix D.)

⁵⁹ This cost-effectiveness "floor" was chosen simply to illustrate one way to prioritize DERs.

DER Type	DER	Net Benefits (million PV\$)	BCR	Reason for Including or Excluding
DG	Large C&I	36	1.9	Include: BCR high
DG	FOM	17	1.7	Include: BCR high
DG	Small C&I	22	1.7	Include: BCR high
DG	Residential	12	1.5	Include: needed for diversity
DR	C&I - low cost	30	1.6	Include: BCR high
DR	Residential - low cost	15	1.5	Include: needed for diversity
DR	C&I - high cost	10	1.3	Exclude: BCR low, not needed for diversity
DR	Residential - high cost	2	1.1	Exclude: BCR low, not needed for diversity
Storage	C&I - low cost	20	1.5	Include: needed for diversity
Storage	C&I - high cost	10	1.3	Exclude: BCR low, not needed for diversity
Storage	Residential - low cost	4	1.2	Include: needed for diversity
Storage	Residential - high cost	1	1.1	Exclude: BCR low, not needed for diversity
EE	Large C&I - prescriptive	41	2.2	Include: BCR high
EE	Large C&I - custom	42	2.1	Include: BCR high
EE	Residential Appliances	38	1.8	Include: BCR high
EE	Low-Income	14	1.6	Include: BCR high
EE	Residential Retrofit	20	1.5	Include: needed for diversity
EE	Small C&I	14	1.3	Include: needed for diversity
Electrification	C&I Heat Pumps	7	1.7	Include: BCR high
Electrification	EVs - public transport	15	1.5	Include: needed for diversity
Electrification	EVs - trucks	2	1.4	Exclude: BCR low, not needed for diversity
Electrification	EVs - cars & light duty	6	1.2	Include: needed for diversity
Electrification	Res. Heat Pumps	1	1.1	Include: needed for diversity

Table 13-1. Example DER Portfolio Developed to Encourage a Diverse Range of Technologies

Source: These are hypothetical examples (see Appendix D).

13.5.6 Objective: Avoid Unreasonable Rate Impacts

Jurisdictions that want to avoid unreasonable rate impacts could do so by conducting rate, bill, and participation analyses to supplement the cost-effectiveness analyses (See Section 2.3 and Appendix A).

In this context, it is useful to conduct a rate, bill, and participation analyses on different DER portfolios to see whether the combined rate impacts of all DER types are reasonable. If rate impacts are estimated to be unreasonably high under one scenario, then other scenarios can be developed to either include more DERs that reduce rates, fewer DERs that increase rates, or both.

Figure 13-3 provides an example of a useful way to present the results of a long-term rate impact analysis.⁶⁰ This example shows the estimated long-term rate impacts, in terms of percent changes in rates, for six different DER types. It also shows the long-term rate impacts of all the DERs combined. In

⁶⁰ The DERs and results presented in this figure are purely illustrative and not based on specific DERs in a specific jurisdiction. Actual results could be significantly different from those presented here. Further, results will vary within the DER types presented here; the DER types are bundled together to keep the figures from being too busy. (See Appendix A.)

this example, the long-term rate increases from EE and DG resources are essentially offset by the rate reductions from the other DER types.



Figure 13-3. Example Presentation of Long-Term Rate Impacts

Figure 13-4 provides an example of a useful way to present the results of a long-term bill impact analysis.⁶¹ This analysis includes the average bills across all utility customers, including both DER host customers and other customers. Also, for those resources that affect multiple fuel types, such as electrification and EVs, this analysis reflects the impact on the combined bills for all the fuels affected by the DER.

⁶¹ See Footnote 63.



Figure 13-4. Example Presentation of Long-Term Average Combined Bill Impacts

Figure 13-5 provides an example of a useful way to present the results of a long-term participation analysis.⁶² It presents five years of historical DER participation rates, as well as 10 years of projections of future participation rates. Participation rates present the number of participants in a program divided by the total eligible participants. Ideally, the participation rates would account for customers who participate in more than one program or install more than one type of DER. That would allow for the presentation of the percentage of DER host versus non-host customers, which would help inform discussions about customer equity.





⁶² The DERs and results presented in this figure are purely illustrative and not based on specific DERs in a specific jurisdiction. Actual results could be significantly different from those presented here. Further, results will vary within the DER types presented here; the DER types are bundled together to keep the figures from being too busy. (See Appendix A.)

13.5.7 Multiple Planning Objectives

Many jurisdictions are likely to have multiple DER planning objectives. In these cases, regulators can establish a set of criteria to use in designing the preferred DER portfolio. For example, a jurisdiction might have the following set of planning criteria:

- Every DER portfolio must, at a minimum, meet statutory targets for specific DER types.
- The preferred DER portfolio should encourage customer equity by giving priority to DERs that serve a broad range of customers, provide balance across customer types, and serve otherwise hard-to-reach customers.
- The preferred DER portfolio should seek to encourage technology diversity by including at least a minimum amount of support for each type of DER.
- The preferred DER portfolio should result in reasonable long-term average rate impacts, commensurate with the benefits provided by the DERs and the customer participation in DERs.
- The preferred DER portfolio should be designed to meet the jurisdiction's GHG goals.

With planning criteria such as these in place, utilities and others can then investigate different combinations of DERs with the goal of meeting the criteria as much as possible.

Table 13-2 provides an example of a DER portfolio developed using multiple planning objectives to determine which DERs should be funded or otherwise supported by utilities. The DERs are sorted from high to low benefit-cost ratios. The top 10 DERs are included in the portfolio because they have high net benefits and benefit-cost ratios. Other DERs are included in the portfolio because they are cost-effective, and they help meet other planning objectives. Three DER were not included in the portfolio even though they are cost-effective, because they had low benefit-cost ratios and were not needed to meet planning objectives.

DER Type	DER	Net Benefits (million PV\$)	BCR	Reason for Including
EE	Large C&I - prescriptive	41	2.2	High BCR & net benefits
EE	Large C&I - custom	42	2.1	High BCR & net benefits
DG	Large C&I	36	1.9	High BCR & net benefits
EE	Residential Appliances	38	1.8	High BCR & net benefits
DG	Small C&I	22	1.7	High BCR & net benefits
Electrification	Heat pumps - commercial	7	1.7	High BCR & net benefits
DG	FOM	17	1.7	High BCR & net benefits
EE	Low-Income	14	1.6	High BCR & net benefits
DR	C&I - low cost	30	1.6	High BCR & net benefits
DR	Residential - low cost	15	1.5	High BCR & net benefits
Electrification	EVs - public transport	15	1.5	High BCR & net benefits
Storage	C&I - low cost	20	1.5	High BCR & net benefits
EE	Residential Retrofit	20	1.5	High BCR & net benefits
DG	Residential	9	1.5	High BCR & net benefits
Electrification	EVs - trucks	2	1.4	Needed to meet GHG target
EE	Small C&I	14	1.3	Customer equity
Storage	C&I - high cost	10	1.3	Exclude from portfolio
DR	C&I - high cost	10	1.3	Exclude from portfolio
Storage	Residential - low cost	4	1.2	Technology diversity & customer equity
Electrification	EVs - cars & light duty	6	1.2	Needed to meet GHG target
Storage	Residential - high cost	1	1.1	Exclude from portfolio
DR	Residential - high cost	2	1.1	Exclude from portfolio
Electrification	Heat pumps - residential	1	1.1	Needed to meet GHG target

Table 13-2. Example DER Portfolio Developed Using Multiple Planning Objectives

Source: These are hypothetical examples (see Appendix D).

13.5.8 Testing the Planning Objectives

It can be useful to test the implications of a jurisdiction's planning objectives by considering the key results of the BCA alongside the key results of the rate, bill, and participation analysis. Regulators, utilities, and others can review these key results to assess tradeoffs between cost-effectiveness and customer equity.

Table 13-3 presents an example comparison of the key results of several DER portfolios developed with different planning objectives. It shows the results of the BCA (in terms of benefit-cost ratios and net benefits), as well as the results of the rate, bill, and participation analysis. The table presents five scenarios: (1) including all cost-effective DERs; (2) focusing on multiple planning objectives; (3) focusing on diverse DER technologies; (4) focusing on the most cost-effective DERs due to budget constraints; and (5) focusing on a mix of DERs that minimizes negative rate impacts.

Planning Objective	Benefit-Cost Ratio	Net Benefits (million PV\$)	Rate Impact	Average Bill Impact	Participation Rates
1. All Cost-Effective DERs	1.5	376	0.6%	-2.6%	28%
2. Multiple Planning Objectives	1.7	353	0.8%	-2.3%	26%
3. Encourage Diverse Technologies	1.8	348	0.9%	-2.4%	21%
4. Most Cost-Effective DERs	2.2	320	0.5%	-2.0%	19%
5. Minimize Negative Rate Impacts	1.9	290	0.2%	-1.8%	16%

Table 13-3. Example DER Portfolios Developed Using Different Planning Objectives

Source: These are hypothetical examples (see Appendix D).

Regarding the key results across scenarios:

- Benefit-cost ratios will typically increase as some of the less cost-effective DERs are removed from the portfolio.
- Net benefits will typically decline as some of the less cost-effective DERs are removed from the portfolio.
- The change in rate impacts will depend upon which DERs are included in the portfolio. The direction and magnitude of rate impacts will vary by DER.
- Average bill impacts will typically decline as DERs are removed from the portfolio. All DERs whose utility system benefits exceed utility system costs will reduce average bills.
- The participation rates will decline as DERs are removed from the portfolio. All else being equal, fewer DERs will result in fewer participants.

Regarding the key results for each scenario:

- 1. *All Cost-Effective DERs:* The benefit-cost ratio is the lowest, but the net benefits are the highest. The participation rate is also the highest because this scenario includes the greatest number of DERs.
- 2. *Multiple Planning Objectives:* The rate impacts are higher than those for the All Cost-Effective DERs scenario because the DERs that were removed for this scenario were DG and storage DERs that have downward pressure on rates.
- 3. *Encourage Diverse Technologies:* The rate impacts are highest because the DERs that were removed for this scenario were DG and storage—DERs that have downward pressure on rates.
- 4. *Most-Cost-Effective DERs:* The benefit-cost ratio is the highest, but this scenario has relatively low net benefits and participants.
- 5. *Minimize Negative Rate Impacts:* The rate impacts are the lowest of all scenarios, but the participation rates are also the lowest. This illustrates the tradeoff between minimizing rate impacts and increasing participation rates, both of which affect customer equity.

This sort of portfolio analysis can help regulators, utilities, and other stakeholders assess the implications of their planning objectives. The results presented in Table 13-3 are illustrative. Actual results will depend upon multiple factors and could vary significantly from those presented here both in direction and magnitude.

13.6 Multiple Tests

In some instances, multiple tests can be helpful when developing portfolios of multiple DER types. Different tests provide different information, and there may be situations where that additional information is helpful in understanding, prioritizing, and deciding among DER options.

Figure 13-6 presents an illustrative example for how to present the results of multiple tests for multiple DER types. It includes benefit-cost ratios for five DER types according to three traditional tests and a JST. The dashed line indicates the benefit-cost ratio of one, which is the threshold above which DERs are cost-effective. Presenting the results this way allows for a quick assessment of how cost-effective the different DER types are according to different tests.



Figure 13-6. Presentation of Multiple BCA Tests: Benefit-Cost Ratios, Multiple DER Types

Source: These are hypothetical examples. All data used here are discussed further in Appendix D.

This example uses a hypothetical JST that includes the following impacts: utility system, host customer, other fuel, low-income, and GHG emissions.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 223 of 302

13.7 Rate Impacts of DER Portfolios

Rate impact analyses should be conducted separately from cost-effectiveness analyses. Rate impacts and cost-effectiveness are two separate issues and analyzing them separately provides much more meaningful information than combining them in a single analysis. (See Chapter 2 and Appendix A.)

The potential rate impacts of programs offering multiple DER portfolios will depend on many factors, including the choice of DERs deployed; the magnitude of DER savings, generation, and consumption; the DER deployment; the DER utility system costs; and the utility system avoided costs.

Rate impact analyses of portfolios with multiple DER types should assess the rate, bill, and participation impacts of all DER types in combination. This offers the advantage of seeing how rate impacts of one DER type might offset, or exacerbate, the rate impacts of other DER types. (See Part III, Chapters 6–10 for further information on individual technology rate impacts, and Appendix A for additional information on rate impacts.)

14. DYNAMIC SYSTEM PLANNING

Dynamic planning practices allow utilities to more dynamically optimize DERs and maximize their value to the system. These types of dynamic planning processes—referred to as integrated distribution planning (IDP) for distribution-level planning only and integrated grid planning (IGP) for full-system planning—are complex, and best practices are still evolving. This chapter provides a brief overview of the key concepts associated with these emerging planning practices.

14.1 Summary of Key Points

- The scope of utility system planning is expanding to manage the increasing complexity of the electricity system, while addressing evolving state policy objectives, changing customer priorities, and increased DER deployment.
- IDP can enable utilities to proactively plan the distribution system in a more dynamic way that better reflects the benefits and costs of DERs. To capture the full value potential of DERs, utilities require enhanced analytical capabilities.
- IGP requires alignment of the objectives, assumptions, and planning horizons between IDP, resource planning, and transmission planning.
- New planning components such as more granular forecasts, improved interconnection
 procedures, hosting capacity analysis, and locational value assessment can enhance traditional
 planning by informing a more comprehensive framework capable of addressing the full range of
 DER value to the grid. These enhanced practices can help identify values for otherwise hard-toquantify impacts.
- Each of these types of planning practices uses some form of BCA for comparing and optimizing different resources.

14.2 Introduction

Chapters 6 through 10 discuss how different DERs can be assessed in isolation using *single-DER analysis*. Chapters 11 through 13 discuss how *multiple-DER analysis* can be used to assess DERs relative to a static set of alternative resources. This chapter discusses advanced planning practices that can allow utilities to more effectively and dynamically optimize DERs using *dynamic system planning*. These practices are complex and best practices are evolving.

Table 14-1 summarizes several different types of planning practices used by electric and gas utilities.⁶³ It presents practices according to whether they are used by distribution-only or vertically integrated utilities, and it shows which elements of the utility system are accounted for by each type of practice.

⁶³ Jurisdictions have adopted a range of terms to describe planning processes. This manual uses the term "IDP" to describe enhanced distribution planning processes that dynamically optimize DERs and uses the term "IGP" to describe enhanced fullsystem planning processes that incorporate generation, transmission, and distribution, including DERs.

Type of		Planning Practice Accounts for:				
Utility System	Planning Practice	Distribution System	DERs	Transmission System	Utility-Scale Generation	
Distribution-only	Traditional distribution planning	\checkmark	-	-	-	
integrated	Integrated distribution planning (IDP)	\checkmark	\checkmark	-	-	
	Transmission planning	-	-	\checkmark	-	
Vertically integrated	Integrated resource planning (IRP)	-	\checkmark	-	\checkmark	
5	Integrated grid planning (IGP)	\checkmark	\checkmark	\checkmark	\checkmark	

Table 14-1. Types of Planning Practices

Utilities of all types have conducted traditional distribution system planning for many years to determine how to best build and maintain the distribution grid. The focus of this practice has been on providing safe, reliable power through the distribution grid at a low cost. It typically has not accounted for DERs as alternatives to traditional distribution system technologies.

IDP has recently evolved as a more comprehensive way for distribution-only utilities to incorporate DERs into traditional distribution system planning. It allows for evaluation of both traditional distribution resources and DERs for meeting distribution grid needs (ICF 2016).

Integrated resource planning (IRP) has been practiced for many years by vertically integrated utilities. It focuses on meeting forecasted peak and energy demands through a combination of DERs and utility-scale generation over a long-term planning period. IRP allows for the optimization of both utility-scale resources and DERs for meeting bulk system needs, but historically it has not been used to investigate how DERs can be used to optimize distribution system costs (RAP 2013c).

IGP has recently evolved as a more comprehensive way for vertically integrated utilities to incorporate DERs into traditional IRP practices. IGP allows for evaluation of all resource types (DERs and utility-scale supply-side resources) to enable optimization across all levels of the utility system (generation, transmission, and distribution) (HECO 2018).

This chapter focusses on IDP and IGP practices because these are evolving practices to better account for opportunities from DERs. In addition to expanding scope beyond traditional practices, IDP and IGP practices account for new priorities such as grid flexibility, resilience, decarbonization, and optimizing grid services. Some jurisdictions are increasingly requiring utilities to go beyond analyzing the impact of DERs on their system to further study how DERs can be optimized and integrated to provide a variety of services on the grid.

Many states are also increasingly interested in grid modernization planning. This practice is similar to traditional distribution planning and IDP, but also incorporates a variety of traditional and emerging utility-facing distribution technologies that can expand grid capabilities, reduce some distribution costs, and enable increasing levels of DER integration (EPRI 2015). For example, as part of the DOE's vision for Next Generation Distribution System platform (DSPx), distribution utilities would make investments in grid modernization technologies in order to more effectively integrate and utilize DERs (DOE 2017a).

Grid modernization planning includes BCA as one component of the decision-making process, but it also includes a least-cost, best-fit component to address some decisions where BCAs are not feasible or

warranted (DOE 2017a).⁶⁴ More detailed discussion of grid modernization planning is beyond the scope of this manual.

14.3 Integrated Distribution Planning

Traditional distribution planning is squarely focused on the engineering analysis required to identify needs on the distribution system and typically involves three steps: (1) develop a single load and DER forecast; (2) conduct distribution engineering analyses to assess the current distribution system and determine long-term needs; and (3) finalize a distribution investment roadmap.

As shown in Figure 14-1, IDP represents a significant enhancement over traditional planning and incorporates new elements and practices that result in a more holistic planning process. For example, instead of using a traditional, single deterministic load and DER forecast, IDP uses multiple scenario-based forecasts. Separately, the distribution engineering analysis is expanded to include elements of hosting capacity and interconnection studies, the latter of which overlaps with the utility interconnection process. The process also incorporates additional steps to assess DER value, including analysis of location-specific DER value and consideration of different DER sourcing mechanisms.



Figure 14-1. Comparison of Traditional Distribution Planning to Integrated Distribution Planning

Source: ICF 2016.

The following summarizes the core components of IDP:

• Load and DER Forecasting: Forecasting is evolving so that utilities are better able to identify system needs and address uncertainties about DER adoption and operation. Accounting for DER in load forecasts, increasing spatial and temporal granularity, and introducing probabilistic and

⁶⁴ The least-cost, best fit approach is sometimes applied to utility projects where a utility has already made the decision that it needs to make an investment to meet a particular need, and the only remaining question is which technology will be the best fit at the lowest cost. A key difference between least-cost, best-fit and BCA approaches is that the former does not require an assessment of the benefits of the project because the need has already been established (DOE 2017a.)

scenario planning help to better reflect DER impacts in forecasting and support more informed grid investments.

- *Hosting Capacity Analysis:* Hosting capacity is the amount of DERs that the distribution system can accommodate without adversely impacting power quality or reliability under current configurations and without requiring infrastructure upgrades. Hosting capacity analysis aims to enable greater DER integration by providing an improved understanding of constraints on the system and identifying where upgrades may be needed to support higher DER levels.
- Interconnection: System planners recognize the need to improve the efficiency and transparency of DER interconnection procedures to manage the workflow associated with a growing number of interconnection applications and meet customer expectations.
- Locational Net Benefits Analysis: The ability to measure and map the location of DER performance against utility planning criteria is a key benefit of assessing locational value in distribution planning. It enables the use of DERs to eliminate or defer system upgrades through NWSs and can inform other efforts including program design and more accurate pricing signals.
- Sourcing DER-Provided Services: Utilities can source services from DERs through a combination of mechanisms including programs, procurement, and pricing. Locational value assessments can inform the design of programs, where utilities provide targeted incentives to resources that deliver needed services. Other sourcing mechanisms, such as setting price signals for DER services through retail rates (time-varying rates, tariffs, etc.) or soliciting DER services through competitive procurements can also be developed to accurately and transparently reflect the temporal and locational value of DERs.

14.4 Integrated Grid Planning

Part of the progression beyond IDP is to move toward greater alignment with IRP and transmission planning to enable fully integrated system planning (see Figure 14-2, purple box). It becomes increasingly important to achieve this alignment with higher levels of DER deployment and greater provision of generation and transmission services from DERs. Comprehensive planning that accounts for these interactions enables consideration of DER to meet needs at all levels of the system. Figure 14-2. Integrated Grid Planning: Expanding IDP to Include IRP and Transmission Planning



Aligning IDP with Other Planning Processes

* Part of Interconnection Process

Source: ICF 2016.

To achieve the full value potential of DERs, utilities need to both advance their planning processes (i.e. IDP and full-system planning) and enhance operational capabilities. Without the requisite operational capabilities, utilities will be unable to account for and optimize the DER value that is otherwise identified through the planning process. As such, utilities need to enhance DER visibility (i.e. sensing and measurement) and control capabilities to effectively manage an increasingly complex and dynamic distribution system with higher DER deployment. By enabling DER value realization at the distribution level, it should by extension also unlock DER value potential at all levels of the system as IGP becomes more effectively integrated into whole-system planning processes.

14.5 BCA Issues and Challenges

There are a number of challenges to reaching fully integrated planning. An expanding set of planning objectives—such as clean energy, resilience, flexibility, equity and affordability, and more—widens the scope of planning activities and requires new methods, criteria, and models.

These additional objectives, combined with the increasing complexity of the electricity system, leads to the need for more sophisticated analytical tools and advanced modeling capabilities. The collection of more granular data is needed to support advanced analysis, but it requires consideration of protecting system security and customer privacy. Increased coordination and new information exchanges are

needed within the utility, between the utility and other agencies with planning responsibility, and with stakeholders through increased transparency at critical steps in the planning process.

To address these challenges, regulators, utilities, and other stakeholders can:

- Work collaboratively to gain an understanding of the engineering analysis that is central to distribution planning. They can expand from there to incorporate advanced methods (e.g., hosting capacity analysis, non-wires alternatives, locational value, etc.) and common assumptions that support alignment and optimization across the system.
- Leverage pilots and demonstration projects to build operational experience and test new concepts, including new DER sourcing methods (e.g., competitive procurements, pricing strategies, and program design) that accurately communicate DER value to the system.
- Phase implementation of new elements and analytical capabilities in a "walk, jog, run" manner that allows for a gradual increase in the complexity of planning practices in accordance with desired objectives. They can also prepare for planning enhancements to occur in lockstep with broader evolution of the distribution system and culminate in a final stage characterized by active grid optimization.
- Eventually, work to identify the objectives, process steps, and timing cycles of distribution, transmission, and generation planning and seek consistency across these planning functions to allow for a fully integrated whole-system planning process.

To learn more about evolving electricity planning processes, NARUC and NASEO created a Joint Task Force on Comprehensive Electricity System Planning. The initiative offers resources to help states explore key topic areas related to aligned planning (NARUC, NASEO 2020). For specific information about how utility commissions can oversee the development of IDPs, see the Mid-Atlantic Distributed Resources Initiative (MADRI) white paper on "Integrated Distribution Planning for Electric Utilities: Guidance for Public Utility Commissions" (MADRI 2019).

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 230 of 302

Appendix A. Rate Impacts

This appendix explains how DERs can lead to rate impacts, and how rate impacts are different from costeffectiveness and why the two should be examined using separate analyses. It also describes how to conduct a meaningful rate, bill, and participation impact analysis.

A.1 The Cause of Rate Impacts

DERs, like all utility resources, will impact customer rates. Some DERs are likely to increase rates, while other DERs are likely to reduce rates. Some may have very little effect on rates at all. The extent to which DERs will impact rates depends upon many factors, including the extent to which they reduce or increase utility system costs and the extent to which they reduce or increase utility sales.

In general, electricity and gas utility rates are set to equal utility system costs (i.e., revenue requirements) divided by sales (i.e., billing determinants). Therefore, analyzing DER rate impacts requires determining how DERs affect utility system costs and utility sales.

The extent to which DERs will create rate impacts typically depends upon several key factors:

- Increases in utility system costs will put upward pressure on rates. These might include, for example: program administration costs; financial incentives to developers and customers; and performance incentives. (See Table 4-1.)
- *Reductions in utility system costs* will put downward pressure on rates. These might include, for example: avoided energy costs, avoided generation capacity costs; avoided environmental compliance costs; avoided RPS/CES compliance costs; market price suppression effects; avoided transmission costs; and avoided distribution costs. (See Table 4-1.)
- Reductions in sales from DER resources will put upward pressure on rates. Reduced sales lead to lost revenues which might require increased rates in order to recover fixed costs over fewer sales.⁶⁵
- Increases in sales from DER resources will put downward pressure on rates. Increased sales lead
 to increased revenues, which might allow rates to be reduced because the increased revenues
 can help pay for fixed costs over more sales.
- *Rate design* will affect the amount of lost or increased revenues created by the DER. Fixed customer charges mitigate the effect of DERS on lost revenues or increased revenues. Similarly, demand charges might mitigate the effect of DERs on lost revenues or increased revenues, depending upon how the charge is structured and the extent to which the DER allows a customer to reduce payments for the demand charge.

⁶⁵ Some DG tariffs create a different dynamic regarding rate impacts. Some DG tariffs provide customers with bill credits for the DG output, which can be rolled over across billing periods or be used to support virtual or community solar programs. These bill credits are sometimes a better indication of DG lost revenues than reduced sales from the DG resource. (See Chapter 8.)

Rate impacts do not necessarily occur immediately upon installation and operation of DERs. Rates need to first be adjusted to reflect changes in costs or changes in sales. Some rate adjustments might occur fairly quickly (e.g., in the case of reconciling rate riders to recover EE program costs, or in the case of fuel and purchased power costs that are automatically passed through to customers in rate riders). Other rate adjustments might occur over a slightly longer period, e.g., in jurisdictions where there is decoupling, performance-based ratemaking adjustments, or lost revenue recovery mechanisms. Other rate adjustments might occur as a result of a rate case (for instance, in the case of reduced generation, transmission, and distribution costs, or in the case of sales adjustments). In the absence of rate adjustments, or until such time as all the rate adjustments are made, the impacts of DERs on costs and sales will be experienced by utility shareholders, not customers.

DER rate impacts can vary from year to year. Many DER costs are experienced in early years, whereas many DER benefits are experienced throughout the DER operating life. Consequently, rate impacts are sometimes greater in the short term than over the long term. A comprehensive understanding of rate impacts requires a long-term analysis of all the factors described above, consistent with the study period used for BCAs.

A.2 Cost-Effectiveness and Rate Impacts

In some cases, DER host customers will experience lower bills while those that do not install DERs may experience higher rates and therefore higher bills. Consequently, the rate impacts of DERs are a matter of customer equity between DER host customers and other customers.

Cost-effectiveness analyses, on the other hand, do not address customer equity. Cost-effectiveness analyses are focused on the benefits and costs to customers and the utility system as a whole.

Cost-effectiveness analyses should be conducted separately from rate impact analyses for the following reasons:

- First, cost-effectiveness analyses should account for only future, incremental benefits and costs, as required by the *Conduct Forward-Looking, Long-term, Incremental Analyses* principle. Rate impacts are driven by lost revenues, but lost revenues from DERs are not a new, incremental cost created by investments in those resources. Rate impacts from lost revenues are caused by the need to recover existing costs over fewer sales. These existing costs that would be recovered through rate increases are not caused by the DERs themselves: They are caused by historical investments in other utility resources that become fixed costs. In economic terms, these existing fixed costs are referred to as "sunk" costs. In economic theory, sunk costs should not be considered when assessing future investments because they are incurred regardless of whether the future investment is undertaken.
- Second, cost-effectiveness analyses are intended to answer a different question than rate impact analyses. Cost-effectiveness analyses are intended to answer the key question of *which utility DER investments are expected to have benefits that exceed costs*. Rate impact analysis are intended to answer the question of *how much will utility DER investments impact rates for one group of customers compared to another*. Attempting to answer these two questions in a single analysis conflates the two questions and thus does not provide helpful information on either one. The RIM Test, for example, combines DER lost revenues (which drive cost-shifting) with DER benefits and costs (which drive cost-effectiveness), to provide a set of metrics in the form of benefit-cost ratios and net benefits. However, these metrics say very little about cost-shifting and they confuse the issue of cost-effectiveness by including distributional effects.

Table A-1 presents a comparison of cost-effectiveness analyses and rate impact analyses.

Key Considerations	Cost-Effectiveness Analysis	Rate Impact Analysis
Answers the question:	Which utility DER investments are expected to have benefits that exceed costs? Cost-effectiveness indicates the extent to which different utility investments will reduce utility costs and achieve other policy goals, regardless of how the benefits and costs are distributed across different customers.	How much will utility DER investments impact rates for one group of customers compared to another? ⁶⁶
Results of the analysis are expressed as:	Present value of revenue requirements, benefit-cost ratios, and net benefits. These metrics are important for regulators and other stakeholders to understand cost- effectiveness, but do not provide any information relevant to rate impacts.	Long-term impacts on rates (in g/kWh or percent changes to rates) or in terms of long-term bill impacts (in \$ per month or percent changes to bills). These metrics are important for regulators and other stakeholders to understand rate impacts but do little to inform benefit-cost analyses.

Table A-1. Comparison of Cost-Effectiveness Analysis and Rate Impact Analysis

Finally, it is important to note that many utility resource investments create rate impacts and customer equity issues. In addition, there are other aspects of electricity and gas regulation that create inequities, such as rate designs that are not fully cost-based. The issues discussed in this manual refer to only those equity issues that are created by DERs. Rate and equity concerns about DERs should be considered in light of the broader point that rate impacts and customer equity issues occur for many reasons beyond just DERs.

A.3 The Rate Impact Measure Test

The *California Standard Practice Manual* presents the Rate Impact Measure Test to account for rate impacts in EE cost-effectiveness analyses (CA PUC 2001). The RIM Test is the same as the UCT except that it includes the lost revenues from DER programs as one of the costs of the DER (or it includes the increased revenues as one of the benefits). (See Appendix E.)

Most jurisdictions have rejected use of the RIM Test for EE cost-effectiveness analyses, but many regulators and stakeholders continue to consider the results of this test and struggle with how to understand and address the rate impacts of EE resources. In recent years some studies of DER cost-effectiveness have used the RIM Test to assess the cost-effectiveness and rate impacts of other DER types.⁶⁷

⁶⁶ Fully understanding the impacts across different customers requires a comparison of the bill impacts on host customers versus the bill impacts of other customers. (See Section A.4.)

⁶⁷ For an overview of BCA tests used for distributed solar resources, see ICF 2018.

The RIM Test can be useful for two purposes, both of which are related to rate impacts but not costeffectiveness:

- To determine whether a DER or set of DERs is likely to increase or decrease rates; and
- To help inform whether a long-term rate, bill, and participation impact analysis is warranted.

For example, if a RIM Test result indicates that rates are likely to be reduced by DERs (with a RIM benefit-cost ratio of greater than 1.0) or rates are likely to be increased a small amount (with a RIM benefit-cost ratio of less than but close to 1.0) then there may be no need to conduct a rate, bill, and participation analysis. If, on the other hand, the RIM Test results indicate that rates are likely to be reduced by a significant amount (such as with a RIM benefit-cost ratio of less than 0.9) then a rate, bill, and participation analysis might be warranted.

Limitations of the RIM Test for assessing the cost-effectiveness of DERs include the following reasons:

- Cost-effectiveness analyses should account for only future, incremental benefits and costs, as required by the *Conduct Forward-Looking, Long-term, Incremental Analyses* principle. The RIM Test accounts for sunk costs (i.e., lost revenues) and as such is inappropriate to use for benefit-cost analysis.
- The RIM Test attempts to answer two different questions in a single analysis, which conflates the two questions and thus does not answer either one.
- The RIM Test does not provide useful information about what happens to rates, in terms of the magnitude of impact, as a result of DER investments. A RIM benefit-cost ratio of less than one (1.0) indicates that rates will increase (all else being equal) but does not inform the extent of the rate impact—either in terms of the percent (or ¢/kWh) increase in rates or the percent (or dollar) increase in bills. In other words, the RIM Test results do not provide any context for regulators and stakeholders to consider the magnitude and implications of the rate impacts.
- Application of the RIM Test will not result in the lowest cost to customers. Instead, it may lead to the lowest rates (all else being equal). However, achieving the lowest rates is not the sole or primary goal of DER planning. Maintaining low utility system costs, and therefore low customer bills, may warrant priority over minimizing rates.
- Application of the RIM Test can lead to perverse outcomes. The RIM Test can lead to the
 rejection of significant reductions in utility system costs to avoid what may be insignificant
 impacts on customers' rates. For example, a DER might offer millions of dollars in net benefits
 under the UCT (i.e., net reductions in utility system costs) but be rejected as not cost-effective if
 it fails the RIM Test. It may well be that the actual rate impact would be so small as to be
 unnoticeable. Rejecting such large reductions in utility system costs to avoid *de minimus* rate
 impacts is not in the best interests of customers overall.
- Lastly, the RIM Test results can be misleading. For a DER investment with a RIM benefit-cost ratio of less than one (1.0), the net benefits (in terms of present value dollars) will be presented as negative benefits. A negative net benefit implies that the DER investment will increase costs. However, as described above, the costs that drive the rate impacts under the RIM Test are not new incremental costs associated with DERs. They are existing costs that are already in current electricity or gas rates. Any rate increase caused by lost revenues would be a result of recovering those existing fixed costs over fewer sales, not as a result of incurring new costs. However, utilities and others frequently present their RIM Test results as negative net benefits, implying that the DER investment will increase costs, when in fact it will not.

A.4 Better Approaches for Analyzing Rate impacts

A thorough understanding of the implications of DER rate impacts and cost-shifting requires analysis of three important factors: rate impacts, bill impacts, and participation impacts.

- *Rate impacts* indicate the extent to which rates for all customers might increase or decrease due to DERs.
- *Bill impacts* indicate the extent to which customer bills might be reduced for those customers that install DERs.
- *Participation impacts* indicate the portion of customers that will experience bill reductions or bill increases. DER host customers will typically experience bill reductions while other customers might see rate increases leading to bill increases.

Taken together, these three factors indicate the extent of the impact on customers from DERs, and also the extent to which DERs might lead to distributional equity concerns. It is critical to estimate the rate, bill, and participation impacts properly, and to present them in terms that are meaningful for considering distributional equity issues (SEE Action 2011).

Rate, bill, and participation analyses should be performed for DERs at an aggregated level, partly for the need to keep the analyses simple and accessible, and partly because customers will experience the rate impacts from DERs in an aggregated fashion. Further, regulators and other stakeholders may prefer to see rate, bill, and participation analyses that account for the impacts of all DERs combined because this offers the advantage of seeing how rate impacts of one DER type might offset, or exacerbate, rate impacts of other DER types.

A.4.1. Rate Impact Estimates

Rate impact estimates should account for all factors that impact rates. This would include all factors that might exert downward pressure on rates, as well as all factors that might exert upward pressure on rates. Any estimates of the impact of lost revenue recovery on rates should reflect collection of only those lost revenues necessary to recover fixed costs. In addition, rate impact estimates should reflect only the actual impact on rates according to the jurisdiction's ratemaking practices, such as decoupling, lost revenue recovery mechanisms, and the frequency of rate cases.

Rate impacts should be estimated over the long term, to capture the full period over which the efficiency savings will occur. The study period should include all the years in which DERs are installed and operational.

Rate impacts should also be put into terms that place them in a meaningful context, so that they can be properly considered and weighed by regulators and other stakeholders. For example, they should be put in terms of g/kWh impacts, dollars per month, percent of total rates, or percent of total bill.

Rate impacts can be markedly different across different customer types. Therefore, it may be necessary to analyze the rate impacts for different customer sectors. Conducting a rate impact analysis for every customer class is probably too burdensome and not necessary. Instead, analyses can be conducted for key customer types such as residential, small commercial, and large commercial and industrial.

A.4.2. Bill Impact Estimates

Bill impact estimates should build upon the estimates of rate impacts. While rate impacts apply to every customer within a rate class, bill impacts will vary between DER host customers and other customers. Further, bill impacts will vary depending upon the customer, the customer's rate design, the type of DER, and the way the DER is operated. For these reasons, it may be sufficient to conduct relatively high-level bill impact analyses.

As with rate impacts, bill impacts should be estimated over the long term, to capture the full period over which the DER is installed and operational. Bill impacts should also be put into terms that place them in a meaningful context, so that they can be properly considered and weighed by DER planners and regulators. For example, they should be put in terms of dollars per month or percent of total bill.

A.4.3. Participation Estimates

Participation estimates should be put in terms of participation rates, measured by dividing DER host customers by the total population of customers. Participation rates provide context and more meaningful information relative to a simple number of program participants. Participation rates can also be used to compare participation across DER initiatives, DER types, across utilities, and across jurisdictions.

Participation rates should be estimated for each year of DER implementation. They should be compared across many years to indicate the extent to which customers are participating in DERs over time. Participation in multiple DERs and across multiple years should be accounted for to the extent possible.

If program participation information is not currently available, it should be collected as soon as possible so that meaningful estimates can be developed in future years. This type of information is critical for assessing the customer equity issues, and hence the rate impact issues, of DERs.

Many equity concerns driven by rate impacts can be mitigated or even eliminated by promoting greater customer adoption of DERs, particularly EE, DR, and community solar programs that can be offered to all customers. Program participation information can be used to encourage more customers to install DERs and thereby experience lower bills. Utilities and other entities promoting DERs could be charged with the responsibility to reach those customers that have not yet implemented some form of DER.

A.4.4. Using the Results

There is no bright line to determine how to balance the cost-effectiveness results with the rate, bill, and participation analysis results. Nonetheless, the results of both analyses can be used to inform that balance. Regulators and other stakeholders can use these two types of analyses to assess whether any expected long-term rate impacts are warranted in light of the cost-effectiveness results, the bill reductions, and the participation rates.

Further, there are many different ways to address concerns about rate impacts once they are well understood. Regulators and other stakeholders may choose to modify proposed DER resources in order to strike a better balance between cost-effectiveness and equity issues. One option is to expand DER resources (especially EE and DR programs that can be offered to any customer) to include more participants and mitigate equity concerns. Another option is to revisit some of the DER designs to find ways to deliver them at lower cost (e.g., finding third-party financing or requiring host customers to pay

larger portions of the DER cost). Another option is to shift priority from DERs that have low participation rates to those that have higher participation rates.

A.5 Presentation of Rate, Bill, and Participation Impacts

Figure A-1 provides an example of a useful way to present the results of a long-term rate impact analysis. This example shows the estimated long-term rate impacts, in terms of percent changes in rates, for six different DER types. It also shows the long-term rate impacts of all the DERs combined. In this example, the long-term rate increases from energy efficiency and distributed generation resources are essentially offset by the rate reductions from the other DER types.



Figure A-1. Example Presentation of Long-Term Rate Impacts

Figure A-2 provides an example of a useful way to present the results of a long-term bill impact analysis. This analysis includes the average bills across all utility customers, i.e., it includes the bills of both DER host customers and other customers. Also, for those resources that affect multiple fuel types, such as electrification and EVs, this analysis reflects the impact on the combined bills for all the fuels affected by the DER.



Figure A-2. Example Presentation of Long-Term Average Combined Bill Impacts

Figure A-3 provides an example of a useful way to present the results of a long-term participation analysis. It presents five years of historical DER participation rates, as well as 10 years of projections of future participation rates. Ideally, the participation rates would account for customers who participate in more than one program or install more than one type of DER. That would allow for the presentation of the percentage of DER host versus non-host customers, which would help inform discussions about customer equity.





The DERs and long-term rate, bill, and participation results presented in these figures are purely illustrative and not based on specific DERs in a specific jurisdiction. Actual results could be significantly different from those presented here. Further, results will vary within the DER types presented here—the DER types are bundled together to simplify the figures.

Note also that for a given set of DERs these results will not vary depending upon which test is used because rate and bill impact estimates include only utility system benefits and costs; other benefits and costs will not affect rate or bill impacts.
Appendix B. Template NSPM Tables

This appendix provides example template tables that jurisdictions can use to take inventory of their applicable policies goals and objectives and to help identify relevant benefits and costs to inform development of their primary Jurisdiction-Specific Test.

The template tables presented or referenced in this appendix are intended to support jurisdictions in their efforts to articulate key assumptions and underlying information to inform development of their primary cost-effectiveness test (or modification to an existing test) using the NSPM Framework, and reporting results of their BCAs. These template tables can help to provide transparency in assumptions and processes used to conduct BCA in a jurisdiction.

Policy Inventory Tables

The Policy Inventory Template Spreadsheet Tables can help support jurisdictions in applying Step 1 of the NSPM BCA Framework, as well as informing Steps 2–4. The tables can help guide jurisdiction efforts to take inventory of applicable energy policies and identify associated relevant impacts to develop their primary test—the Jurisdiction-Specific Test—and to compare these to the jurisdiction's current cost-effectiveness testing practices.

This template table can be downloaded (in excel) at: <u>https://nationalenergyscreening.org/resources/templates/</u>

Summary Benefit-Cost Analysis Results Table (for single DER)

This template table, illustrated on next page, can be used by jurisdictions to document the benefit and cost results from their BCA for a single DER program/project. Results are summarized by impact, by impact category, and in total, as well as net benefits and overall BCR.

The table provides the flexibility to display results of JSTs from any jurisdiction by providing space for that jurisdiction to include host customer and societal impacts to the extent that they are relevant (according to applicable policies) and included in that JST. There is also space to include non-monetized results of quantitative or qualitative data for economic development, market transformation, and other impacts.

This template table can be downloaded (in excel) at: <u>https://nationalenergyscreening.org/resources/templates/</u>

Template Table: Summary Benefit-Cost Analysis Results

Single DER Program/Project Name:				Date:				
A. Monetized Electric Utility System Impacts	Benefits	6	Costs	(A. cont.)	Bene	fits	Cos	ts
Generation: Energy Generation				Distribution: O&M				
Generation: Capacity				Distribution: Voltage				
Generation: Environmental Compliance				General: Financial Incentives				
Generation: RPS/CES Compliance				General: Program Administration Costs				
Generation: Market Price Effects				General: Utility Performance Incentives				
Generation: Ancillary Services				General: DG tariffs				
Transmission: Capacity				General: Credit and Collection Costs				
Transmission: System Losses				General: Risk				
Distribution: Capacity				General: Reliability				
Distribution: System Losses				General: Resilience				
				A. Sub-Total Electric Utility System Impacts	\$	-	Ś	-
B. Monetized Host Customer Impacts	Benefits	6	Costs	C. Monetized Societal Impacts	Bene	fits	Cos	ts
DER costs (host)				Resilience				
Transaction costs (host)				GHG Emissions				
Interconnection Fees		1		Other Environmental	In	cludes	L. Cociotal	
Risk	Includ	le Host		Economic and Jobs		nacts t	oevtent	
Reliability	Custome	t thou are		Public Health	th	iev are	part of	
Resilience	nart	they are		Low Income		JS	T	
Power Quality		51 331		Energy Security				
Non-Energy Impacts (Host)				Energy Security Benefits			1	
Non-Energy Impacts (Low-income)								
B. Sub-Total Host Customer Impacts	\$-	\$	-	C. Sub-Total Societal Impacts	\$	-	\$	-
D. Total Monteized Benefits and Costs	Total Pre	sent Value	e (\$)	Benefit-Cost Ratio				
Total Benefits	\$		-					
Total Costs	\$		-					
Net Benefits or Costs	\$		-	0.0				
E. Non-Monetized and Qualitative Assessments								
Economic Development and Job Impacts			GHG Emissions Include Societal t Other Environmental Include Societal acts Economic and Jobs Impacts to extent acts Public Health Impacts to extent Low Income JST Impacts to extent Energy Security Impacts to extent Impacts to extent Energy Security Benefits Impacts to extent Impacts to extent - C. Sub-Total Societal Impacts \$ - 'alue (\$) Benefit-Cost Ratio Impacts - - - - 0.0 Impacts					
Market Transformation Impacts			e qualitativ	e considerations and describe how considered				
Other Non-Monetized Impacts		Provide	quantitati	ve information, qualitative considerations, and describe hov	v consider	red		
COST-EFFECTIVENESS DETERMINATION:				Do DER Benefits Exceed Costs?		[Yes	/No]	

Jurisdictions can use these template tables as is or can expand and modify them to meet their specific needs in documenting key assumptions and results from their BCA.

Appendix C. Approaches to Accounting for Relevant Impacts

This appendix describes different approaches that jurisdictions can use to account for relevant benefits and costs, including those that are hard to quantify or hard to monetize. It also describes how to use quantitative and qualitative information in the absence of monetary values. The appendix provides a sample of existing tools and resources/studies for how to quantify impacts, as well as some information on general methodologies.

C.1 Introduction

Benefits and costs of DER investment may be estimated in monetary or other non-monetary quantitative terms or qualitative values. Although using monetary values provides consistency for direct comparison of DER impacts, some impacts are hard to monetize. Jurisdictions also face constraints to conducting rigorous jurisdiction-specific impact studies and may therefore choose to apply other approaches such as estimating values based on studies from other jurisdictions, determining proxies, or using alternative thresholds. This appendix discusses these considerations and also provides examples of existing studies and tools for quantifying DER impacts.

For impacts that are hard to monetize, identifying other approaches to including these impacts in tests as quantitative or qualitative data can be necessary in order to treat benefits and costs symmetrically.

Table C-1 summarizes five approaches that can be used to account for relevant and material impacts of DERs that a jurisdiction has chosen to include in its cost-effectiveness test. These are each discussed in the following sections.

Approach	Description
Monetary Approaches:	
Jurisdiction-specific studies	Rigorous jurisdiction-specific studies on DER impacts offer the potentially most accurate approach for estimating and monetizing relevant impacts.
Studies from other jurisdictions	If jurisdiction-specific studies are not available, studies from other jurisdictions or regions, or national studies, can be used for estimating and monetizing impacts.
Proxies	If monetized impacts are not available, well-informed and well-designed proxies can be used as a simple substitute (e.g., % adders).
Non-Monetary Approaches:	
Alternative thresholds	Pre-determined thresholds, e.g., BCRs that are different from one (1.0), can be used as a simple way to account for relevant impacts that are not otherwise included.
Accounting for non-monetized impacts	Relevant qualitative information can be used to estimate impacts that cannot be monetized.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 241 of 302

C.2 Monetary Values

All DER impacts that a jurisdiction has chosen to include in its cost-effectiveness tests should ideally be estimated in monetary terms. Monetary values provide a uniform way to compile, present, and compare benefits and costs. While some DER impacts are difficult to quantify in monetary terms—either due to the nature of the impact or the lack of available information about the impacts—informed approximation of hard-to-quantify impacts (i.e., not using arbitrary values) is preferable to assuming that the relevant benefits and costs do not exist or have no value. Further, some approximation may be necessary to ensure symmetry in the treatment of benefits and costs for certain relevant impacts.

Estimating impacts in monetary terms for utility system impacts, other fuel impacts, and non-utility system impacts require the following considerations:

- For electric and gas utility system impacts, monetary values should be in terms of utility revenue requirements so that the values will indicate the effect of the benefits and costs on the utility customers in each year. Costs that occur on an annual basis and are collected by utilities each year should be accounted for in the year they are collected from utility customers. Capital costs should be amortized over the regulatory book life of the investment. This will turn a single capital investment in one year into a stream of annual revenue requirements over that book life. Those annual revenue requirements should properly reflect the ratemaking treatment for capital costs in the relevant jurisdiction, including recovery of depreciation, taxes, debt, and equity.
- *For other fuels impacts,* monetary values should be based upon the retail price of those fuels. The retail price typically reflects the annual marginal costs of providing those resources.
- For other non-utility system impacts, monetary terms should be used as much as possible. For example, host customer impacts and third-party impacts should be put in monetary terms for the years in which they are incurred.

C.2.1. Jurisdiction-Specific Studies

Jurisdiction-specific studies are the most rigorous and reliable way to estimate the benefits and costs of DERs in that jurisdiction. These studies typically use local information to the greatest extent possible, by utility, by state, by province, or by the relevant Regional Transmission Organization/Independent System Operator. These studies are derived from, or at least be consistent with, the most recent integrated resource planning studies available, wherever they exist. More information on tools and studies is available in Section C.5 below.

Utility System Impacts

Avoided cost studies. These studies are frequently conducted to quantify DER benefits, in particular the utility system benefits. These studies are typically jurisdiction-specific to ensure that they accurately capture the operations and the impacts of the utility system where the DERs will be installed. The studies should be comprehensive by addressing the full range of utility system impacts (see Chapter 4). Further, best practice for conducting avoided cost studies includes use of all relevant information available at the time of study, ensuring periodic updates to reflect the most recently available information, and transparency in underlying input assumptions used to calculate the avoided costs.

Avoided cost studies require many detailed assumptions and comprehensive/complex methodologies, and they typically produce detailed results. For regulators, utilities, and other stakeholders to properly

assess and understand avoided cost studies and implications for BCAs—to ultimately ensure that BCA conclusions are reasonable and robust—key inputs, assumptions, methodologies, and results should be clearly documented in sufficient detail to allow for review by key stakeholders and ultimately by regulators via a jurisdiction's specific process.⁶⁸

Host Customer and Societal Impacts

Impact Studies: Host customer (participant) impact studies estimate a range of benefits to customers that invest in a DER. To date, participant impact studies mostly pertain to the impacts of energy efficiency programs, and frequently they apply to one or more program sectors such as residential or industrial.

Host customer and societal impact studies may use one or a combination of several approaches to data collection, including self-reporting e.g., through surveys, direct measurement such as from a meter, and using secondary data collected from other industries, such as insurance filings. Once collected, this data may be applied to assess program impacts through basic aggregation, engineering analyses to extrapolate impacts beyond the data collected, and complex economic models.

As with avoided cost studies, conducting host customer and societal impact studies can involve comprehensive studies, which typically entail statistical analyses and triangulation of various estimation methods. For regulators, utilities, and other stakeholders to properly assess and understand these impacts and their implications for BCAs, key inputs, assumptions, methodologies, and results should be clearly documented in sufficient detail to allow for review by key stakeholders and ultimately by regulators through a jurisdiction's specific process.

C.2.2. Studies from Other Jurisdictions

In some cases and for some impacts, a jurisdiction-specific study might not be available for some of the information needed for a BCA. In these cases, it may be appropriate to use results from other jurisdictions. This could include studies prepared for other utilities, other states, other jurisdictions, or other regions. It could also include regional or national studies that do not necessarily focus on any one jurisdiction or region.

However, DER planners must take care to ensure that the value of a particular benefit or cost in another jurisdiction is equal to, or sufficiently comparable to, the value in the jurisdiction of interest. If not, it may be necessary to adjust values from other jurisdictions before using them. For example, labor costs in one part of the country might be significantly different from other parts of the country. These differences can be accounted for by adjusting values according to an agreed-upon methodology.

Lawrence Berkeley National Laboratory has developed a list of jurisdiction-specific studies currently used to quantify NEIs in EE cost-effectiveness tests, as well as information on the transferability of the values and/or methodologies produced in the studies (LBNL 2020a).

⁶⁸ See as an example the New England Avoided Energy Supply Cost studies (AESC Study Group 2018). Another example is the California Public Utility Commission cost-effectiveness calculator that embeds the state's official avoided costs in a model to calculate cost-effectiveness (CPUC 2016).

C.2.3. Proxies

For the purpose of DER BCAs, a proxy is a simple, quantitative value that can be used as a substitute for a value that is not monetized by conventional means. Proxies can be applied to any type of benefit or cost that is hard to monetize and is expected to be of significant magnitude.

Proxy values are typically based on professional judgment; but they should not be developed or perceived as arbitrary values. Proxies should be developed by making informed approximations based upon the best information currently available regarding the relevant impact. This should include a review of relevant literature on the specific impact, as much quantification of the impact that is both feasible and reasonable, a review of proxy values used by other jurisdictions, and consideration of conditions specific to the relevant jurisdiction.

To date, proxies have most frequently been used to account for DER benefits such as low-income benefits, participant non-energy benefits, or risk benefits (NESP 2019). However, proxies can be used for all DER impacts.

Level of Application

Proxy values can be developed for different levels of application, ranging from a single proxy value that applies to an entire portfolio of DERs to different proxy values for each DER impact.

When choosing the level of detail to apply to a proxy, there may be a tradeoff between accuracy and feasibility. Proxies that are more detailed are likely to more accurately represent the magnitude of the specific impact in question. However, proxies that are more detailed are also likely to require more information and greater costs to develop.

One advantage of more detailed proxies is that they are more transferrable across programs, across utilities, and over time. For example, an impact-level proxy such as improved health and safety, applied to residential retrofit efficiency programs, is likely to be generally applicable to other residential retrofit programs and remain relatively constant over time. Conversely, a sector-level proxy to account for all participant non-energy benefits for the residential sector should, in theory, be different for different programs and could change over time as the mix of distributed resources changes over time.

Type of Proxy

Several different types of proxies can be used to account for DER impacts.

- *Percentage Adder:* A percentage adder approximates the value of non-monetized impacts by scaling up impacts that are monetized. This type of proxy is the simplest and easiest to apply.
- *Electricity Savings Multiplier (\$/MWh):* An electricity savings multiplier approximates the value of non-monetized benefits or costs relative to the quantity of electricity saved by a DER.
- Gas Savings Multiplier (\$/therm): A gas savings multiplier approximates the value of nonmonetized benefits or costs relative to the quantity of gas saved by a DER. It offers the same advantages and disadvantages of electricity multipliers.
- *Fuel Savings Multiplier (\$/MMBtu):* A fuel multiplier approximates the value of non-monetized benefits or costs relative to the total quantity of fuel saved by a DER, regardless of the type of fuel saved (e.g., electricity, gas, oil, propane).
- *Customer Adder (\$/customer):* A customer adder (or subtraction) approximates the value of non-monetized benefits relative to the number of customers served by a DER program.

 Measure Multiplier (\$/measure): A measure multiplier (positive or negative) approximates the value of non-monetized benefits or costs relative to the number of measures implemented as part of a DER program.

As with the choice of level of application for a proxy, the choice of which type can result in a tradeoff between accuracy and feasibility. Proxies that are more focused (e.g., by measure, by customer, or by fuel) are more likely to accurately represent the magnitude of the specific impact in question. However, proxies that are more focused are also likely more difficult and expensive to develop.

C.3 Non-Monetary Values

Once all efforts to monetize DER impacts have been considered and exhausted, the following steps can be used to consider additional quantitative and qualitative non-monetary information.

Distinguish between whether and how to include an impact

Regulators and stakeholders will sometimes decide not to include a certain benefit or cost in the primary test because the impact is assumed to be too small, too uncertain, or too hard to quantify. A better approach is to first decide *whether* to include impacts in cost-effectiveness tests based on the fundamental cost-effectiveness principles, and then decide separately *how* to value or otherwise account for the impacts.

This distinction between whether to include something and how to include it is very important. This is partly because it allows for a more transparent application of the fundamental BCA principles, partly because an impact that is small and immaterial today might become large and material in the future, and partly because an impact that is small and immaterial for one DER might be large and material for another DER. Once the decision has been made regarding whether to include an impact in the primary test, then a secondary decision can be made as to how to account for it. If an impact is assumed to be so small as to be immaterial, then it should be included in the test but assigned a value of zero.

Provide as much quantitative evidence as possible

For those impacts that remain non-monetized, it may be possible to put them into quantitative terms. Quantitative values generally provide more concrete information for decision-makers to consider, relative to qualitative values or no values at all. Quantitative values of DER impacts should be documented in detail, along with justification for why and how the values are relevant to the costeffectiveness analysis.

For example, jurisdictions that choose to include job impacts might want to present this impact in terms of the number of job-years, rather than a monetized value for jobs. Regulators and distribution planners could then compare different energy resources according to how many job-years are created by each one.

Establish metrics to create quantitative data for future analyses

Metrics offer a quantitative way to assess the extent of a benefit or cost, in the absence of monetary values for this purpose. For example, if the utility does not monetize safety, resilience, or power quality benefits for its BCA, regulators can establish metrics to indicate the extent to which these benefits will be experienced. The data obtained over time for these metrics can be used for BCAs in future years.

Provide as much qualitative evidence as possible

Those impacts that are not monetized or quantified should be addressed qualitatively. Qualitative information can provide some information for decision-makers to consider, relative to no information at all. For those DER impacts that are addressed qualitatively, distribution planners should develop and present as much qualitative evidence as possible regarding those impacts. This evidence should also include a justification for why the considerations are relevant to the cost-effectiveness analysis.

For example, a jurisdiction might choose to consider incremental market transformation benefits without quantifying or monetizing such benefits. In this case, regulators or distribution planners would consider the incremental market transformation benefits, without necessarily estimating what those benefits are either in terms of energy savings or dollar savings.

Decide upon the implications of the quantitative and qualitative evidence

Any non-monetized impacts of DERs should be presented along-side the monetary impacts. This allows the regulators and other decision-makers to compare the monetized, quantitative, and qualitative factors and evidence to decide whether a DER is cost-effective. In some cases, the monetary results alone might be sufficient to make this decision (for instance, if the monetary benefits exceed the monetary costs, and all the non-monetary evidence indicates there will be additional benefits). The cost-effectiveness decision might also be easy if the monetary benefits are slightly less than the monetary costs, but the non-monetary benefits are clearly significant enough to make up the difference.

In other cases, the decision might not be so clear. For example, if the monetary benefits do not exceed the costs, then the non-monetary benefits are not necessarily significant enough to make up the difference. In these cases, regulators and other decision-makers should make a cost-effectiveness determination, based on all the evidence presented, and with input from relevant stakeholders.

Document and justify the decision

Finally, the cost-effectiveness decision should be fully documented and justified. This is necessary to provide transparency regarding the decision for the resource in question, and to provide guidance on how similar decisions will be made in future BCAs.

C.4 Alternative Thresholds

Alternative thresholds are another approach for addressing hard-to-monetize impacts. They allow DERs to be considered cost-effective at pre-determined benefit-cost ratios that are different from one (1.0). Regulators can apply a benefit-cost ratio of greater than one (1.0) to account for DER costs that have not been monetized, or a benefit-cost ratio of less than one (1.0) to account for benefits that have not been monetized.

Alternative thresholds are, by design, a simplistic way of recognizing that the hard-to-monetize impacts are significant enough to influence the cost-effectiveness analysis. The primary advantage of this approach is that it does not require the development of specific monetary or proxy values. Rather, it provides a general reflection of the regulators' willingness to be flexible in accounting for certain impacts.

Using alternative benchmarks can essentially have the same effect as applying a proxy value. For example, an alternative portfolio-level benefit-cost ratio benchmark of 0.9 is equivalent to a portfolio-

level benefit multiplier of 11 percent; and an alternative benefit-cost ratio benchmark of 0.8 is equivalent to a benefit multiplier of 25 percent.

Regulators should ensure that alternative thresholds are as transparent as possible and are established prior to the cost-effectiveness analysis. Ideally, regulators should articulate which resources the alternative thresholds can be applied to, what the threshold is, and the basis for the threshold chosen.

C.5 Tools and Studies to Support Estimation of Impacts

A variety of tools and studies exist to help local utilities, regulators, policymakers, and other stakeholders estimate and document quantified impacts of DER programs, policies, and investments in their jurisdictions.⁶⁹

Studies on Multiple DERs

Most research on impact quantification to date exists for EE programs, so many of the resources listed in this section pertain to energy efficiency. However, there is a growing list of publications to support estimation of impacts for multiple DERs, including:

- <u>How Distributed Energy Resources Can Improve Resilience in Public Buildings: Three Case Studies</u> <u>and a Step-by-Step Guide</u> (DOE 2019d),
- <u>Energy Efficiency and Distributed Generation for Resilience: Withstanding Grid Outages for Less</u> (DOE 2019a),
- <u>Capturing More Value from Combinations of PV and Other Distributed Energy Resources</u> (RAP 2019a), and
- The Value of Resilience for Distributed Energy Resources: An Overview of Current Analytical Practices (NARUC 2019).

Tools

EPA maintains the <u>Publications, Tools, and Data for State, Local, and Tribal Governments webpage that</u> <u>includes resources to support energy policy-making, program implementation, and evaluation.</u> The tools available on this page include the AVoided Emissions and geneRation Tool (AVERT), which supports air emissions impact estimation (2018), the Environmental Benefits Mapping and Analysis Program (BenMAP-CE), which calculates the economic and public health impacts of air pollution (2018), and the CO-Benefits Risk Assessment (COBRA) Health Impacts Screening and Mapping Tool for estimating the health and economic benefits of clean energy policies (2018).

Impacts studies

The studies listed below provide a sample of existing methodologies for estimating one or multiple EE impacts at the utility system, host customer, or society level. Therefore, they are organized below by the level of impact studied.

⁶⁹ This list of tools resources is current as of the Summer 2020 publication of this NSPM and is not an exhaustive list of available resources.

Utility System Impacts studies

- Avoided Energy Supply Components in New England: 2018 Report (Synapse Energy Economics, Resource Insight, Les Deman Consulting, North Side Energy, and Sustainable Energy Advantage. 2018)
- Keeping the Lights On: Energy Efficiency and Electric System Reliability (ACEEE, 2018)
- <u>Everyone Benefits: Practices and Recommendations for Utility System Benefits of Energy</u> <u>Efficiency. (ACEEE, 2015)</u>
- Valuing the Contribution of Energy Efficiency to Avoided Marginal Line Losses and Reserve Requirements (RAP, 2011)

Multiple benefits studies (Host Customer Impacts, Societal Impacts)

- <u>Making Health Count: Monetizing the Health Benefits of In-Home Services Delivered by Energy</u> <u>Efficiency Programs (ACEEE, May 2020)</u>
- Public Health Benefits per kWh of Energy Efficiency and Renewable Energy (EPA, 2019)
- <u>Cost-Effectiveness Tests: Overview of State Approaches to Account for Health and</u> <u>Environmental Benefits of Energy Efficiency (ACEEE, 2018)</u>
- <u>Saving Energy, Saving Lives: The Health Impacts of Avoiding Power Plant Pollution with Energy</u> <u>Efficiency (ACEEE, 2018)</u>
- <u>Analysis of the Public Health Impacts of the Regional Greenhouse Gas Initiative (Abt Associates,</u> 2017)
- Occupant Health Benefits of Residential Energy Efficiency (E4TheFuture, 2016)
- Home Rx: The Health Benefits of Home Performance (DOE, 2016)
- State and Utility Pollution Reduction Calculator Version 2 (SUPR 2) (ACEEE, 2016)
- <u>ACEEE State Policy Toolkit: Guidance on Measuring the Economic Development Benefits of</u> <u>Energy Efficiency (ACEEE, 2019)</u>
- Quantifying the Multiple Benefits of Energy Efficiency and Renewable Energy (US EPA, 2018)
- <u>Non-energy Benefits in State Cost-Effectiveness Tests Reducing Bias in Consideration of Energy</u> <u>Efficiency as a Resource (Skumatz, 2018)</u>
- The Cost of Saving Electricity Through Energy Efficiency Programs Funded by Utility Customers: 2009-2015 (Hoffman, I., C. Goldman, S. Murphy, N. Mims, G. Leventis, and L. Schwartz., 2018.)
- <u>Assessing the Cost Effectiveness of Energy Efficiency Portfolios (LBL, 2017)</u>
- Non-energy Impacts Approaches and Values: An Examination of the Northeast, Mid-Atlantic, and Beyond (NEEP, 2017)
- Evaluating and Quantifying the Non-Energy Impacts of Energy Efficiency (LBL Webinar, 2016)
- <u>Recognizing the Value of Energy Efficiency's Multiple Benefits (ACEEE, 2015)</u>
- <u>Recognizing the Full Value of Energy Efficiency (RAP, 2013)</u>
- <u>Energy Efficiency Cost-Effectiveness Screening How to Properly Account for 'Other Program</u> <u>Impacts' and Environmental Compliance Costs (RAP, 2012)</u>

Appendix D. Presenting BCA Results

This appendix provides guidance on how to present results in ways that are most informative and most useful in making cost-effectiveness decisions. It also provides guidance on how to present results to support efforts to prioritize across different DER types.

D.1 Introduction

There are multiple ways of presenting the results of BCAs, each with different strengths and limitations. The primary options are described in the following subsections.

The examples below illustrate how to present some of the most useful information from costeffectiveness analyses. The DERs and cost-effectiveness results presented in examples below are purely illustrative and not based on specific DERs in a specific jurisdiction. Actual cost-effectiveness results could be significantly different from those presented here. In addition, actual results will differ depending upon the cost-effectiveness test used. Further, some DER types are bundled together to keep the figures from being too complex.

D.2 Present Values of Benefits and Costs

The present value of the benefits and costs serve as the foundation for BCA results. It is often useful to start by presenting the costs separately from the benefits before combining them into net benefits or benefit-cost ratios.

It is also useful to present the stream of annual present value costs and the stream of annual present value benefits to provide a complete picture of how they play out over time.

Then the two streams of annual present value benefits and costs should be accumulated over the study period to determine the cumulative present value of benefits and the cumulative present value of costs.

It is also useful to present the annual streams of *cumulative* present values of benefits and costs, in order to indicate which year, if ever, the cumulative benefits exceed the cumulative costs. The number of years it takes for the cumulative benefits to exceed cumulative costs provides an indication of the "payback" period. In general, resources with shorter payback periods are more cost-effective than those with longer payback periods.

D.3 Net Benefits

Net benefits are equal to the difference between the cumulative present value of benefits and the cumulative present value of costs, as per the formula below. This metric is useful as a benchmark for determining cost-effectiveness: If a DER's net benefits are greater than zero, its benefits exceed its costs, and it is considered cost-effective.

Net Benefits (dollars) = NPV Σ benefits - NPV Σ costs

Figure D-1 provides an example of a useful way to present costs, benefits, and net benefits of an illustrative set of EE programs. Similar graphs could also be developed for other DER programs, other EE programs and/or combinations of EE and other DER programs. In this hypothetical example, the Large C&I program offers the greatest net benefits, and the Deep Energy Retrofit program has negative net benefits and is therefore not cost-effective. Of course, actual EE programs might have very different results than the examples presented here.





The net benefits metric provides important information that is not provided by a benefit-cost ratio, by indicating the magnitude of the benefits to be gained by the efficiency resource. For example, a BCR of 2.2 does not indicate how much money will be saved by the resource. It might save \$1 million, \$10 million, or \$100 million.

On the other hand, the net benefits of DERs cannot easily be used to compare DERs across different utilities and jurisdictions. A large utility would naturally expect to have higher net benefits than a small utility for a comparable DER.

D.4 Benefit-Cost Ratio

A benefit-cost ratio is equal to the ratio of the cumulative present value of benefits to the cumulative present value of costs, as per the formula below. This metric is useful as a simple benchmark for determining cost-effectiveness: If a DER's BCR exceeds 1.0, its benefits exceed its costs, and it is considered cost-effective.

Benefit-Cost Ratio = $\frac{\text{NPV} \sum \text{benefits (dollars)}}{\text{NPV} \sum \text{costs (dollars)}}$

The BCR metric can be useful for comparing DERs with each other (i.e., a higher BCR indicates one resource is "more cost-effective" than another), because it effectively normalizes the results for DERs of

different sizes. This metric is also useful for comparing DERs across utilities and jurisdictions of different sizes, again because it effectively normalizes the results for any differences in size.⁷⁰

Figure D-2 provides an example of a useful way to present benefit-cost ratios. This example includes a variety of hypothetical DER types. If multiple DERs are presented in a single graph in this way, it is important that the cost-effectiveness test and the inputs for each DER are consistent. For example, if the host customer benefits and costs are included in the costs for one DER type, then they should be included for all DER types.





The BCR metric provides important information that is not provided by a net benefits metric. It does this by indicating the relative effectiveness of the money spent on the resource. i.e., how many dollars of benefits are received per dollar spent. For example, a net benefit of \$10 million in present value dollars (PV\$) does not indicate how much money was needed to generate those net benefits. It could have cost \$90 million, with benefits of \$100 million and a BCR of 1.1. Or it could have cost \$4 million, with benefits of \$15.

⁷⁰ However, in making such comparisons it is important to recognize that different utilities and jurisdictions might have different avoided costs, i.e., different benefits for the same amount of savings. Different jurisdictions might also include different impacts in their resource assessment test.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 251 of 302

D.5 Levelized Costs

D.5.1. Levelized Cost Calculations

The costs of electricity and gas resources, including DERs, can be put into levelized costs to allow for a relatively simple, direct comparison across different resources. Levelized costs represent the average cost per unit of energy required to install and operate an electricity or gas resource. They include the costs of the resource over its economic operating life, amortized over the lifetime and discounted back to the first year divided by the total lifetime energy produced (Lazard 2019).

Levelized costs for electricity generation resources are commonly referred to as levelized cost of electricity (LCOE) (EIA 2019). Levelized costs for efficiency resources are commonly referred to as levelized cost of saved energy (LCSE) (LBNL 2018). They are calculated using the following formulas:

LCOE = (capital recovery factor) * (resource lifetime costs) / (annual generation, in kWh) LCSE = (capital recovery factor) * (resource lifetime costs) / (annual electricity savings, in kWh) Capital recover factor = [r*(1+r)n] / [(1+r)n-1] r = the discount rate n = the resource lifetime

This manual uses the term "levelized cost of energy" to refer to both LCOE and LCSE interchangeably.

Figure D-3 provides an example of a useful way to present LCOE for EE resources.

Figure D-3. Example Presentation of Levelized Cost of Energy: Hypothetical EE Resources



While the levelized cost of energy is a useful way to present the *costs* of electricity and gas resources, it only tells part of the story. It says little about the *benefits* of the resources. It puts all the costs in terms of generation benefits (MWh), but says nothing about generation capacity (kW) benefits. It also says nothing about T&D benefits, risk, reliability, other fuel, host customer, or societal benefits.

Levelized costs can also be used to indicate capacity impacts by calculating a levelized cost of capacity (in \$/kW). This is especially useful for resources whose primary function is to provide capacity services. It is calculated the same way as the LCOE, except that annual generation is replaced with annual peak

savings. Figure D-4 provides an example of a useful way to present levelized cost of capacity for several illustrative DERs.⁷¹





Similarly, levelized costs can also be used to indicate carbon reduction impacts by calculating a levelized cost of carbon reduction (in \$/ton of GHG reduced). This is especially useful for indicating which resources are most effective at reducing carbon emissions for the dollar spent. It is calculated the same way as the levelized cost of energy, except that annual generation is replaced with annual GHG reductions (or increases). Figure D-5 provides an example of a useful way to present levelized cost of GHG reduction for several illustrative DERs.

⁷¹ This figure does not include levelized cost of capacity for electrification DERs because these resources typically do not save capacity.



Figure D-5. Levelized Cost of GHG Reduction: Multiple Hypothetical DERs

LCOE, capacity, and GHG reduction all provide different information about the impacts of DERs. For those jurisdictions interested in avoiding capacity costs and reducing GHG emissions, in addition to avoiding energy costs, it would be most informative to calculate and consider all three of these metrics.

Finally, it is important to recognize the limitations of levelized costs of any type. They do not account for certain benefits that might be included in a jurisdiction's primary test, including risk, other fuel impacts, low-income benefits, host customer benefits, or societal benefits. Therefore, while levelized costs have some usefulness for comparing different DERs to each other, they should not be used in isolation for determining whether a DER is cost-effective. Also, while levelized costs can help prioritization across DERs, it may be important to also consider other comparisons that account for other differences in DER benefits.

D.5.2. Levelized Cost Curves

DER cost curves offer a useful way to compare DERs. Cost curves are a graphic depiction of levelized costs where a bar chart is used to depict (a) the DER levelized costs, sorted lowest to highest cost, and (b) the magnitude of the savings for each DER. Typically, the levelized costs are represented on the vertical axis, while the magnitude of savings is presented on the horizontal axis, where the width of each bar indicates the amount of savings.

There are many types of cost curves that can be created to inform DER cost-effectiveness decisionmaking. Figure D-6 presents an example of a GHG cost curve.



Figure D-6. Example of GHG Cost Curve: Hypothetical Set of Multiple DER Types

Cost curves like the one above are subject to the same limitations as levelized costs. They present only costs. They do not account for the full complement of DER benefits and they only account for the one benefit displayed along the horizontal axis, which in this case is GHG benefits. Therefore, while cost curves have some usefulness for comparing different DERs to each other, they should not be used in isolation for determining whether a DER is cost-effective. And while they can help inform prioritization across DERs, it may be important to also consider other comparisons that account for other differences in DER benefits.

D.6 Net Levelized Costs

D.6.1. Net Levelized Cost Calculations

As discussed above, levelized cost calculations divide the levelized measure or program cost of a DER by *just one* of the many benefits of each DER—whether kWh reduction, peak kW reduction, gas thermal reduction, GHG emissions reduction, or any other metric of interest. That limits the value and applicability of levelized costs calculations. For example, a DER that provides both energy savings and system peak demand savings may be more cost-effective (either greater net economic benefits and/or

higher benefit-cost ratio) but look more expensive on a levelized cost per kW of system peak reduction basis than a different DER that provides only system peak kW benefits.

That limitation can be addressed through one important modification: subtracting the value of the "other benefits" (i.e. all benefits other than the one by which levelized costs are being divided) from the DER measure or program cost used in a standard levelized cost calculation. The result is a *net levelized cost*. As with levelized costs, net levelized costs can be computed for any DER impact or benefit of interest—whether kWh saved, system peak kW reduction, distribution kW reduction, gas thermal reduction, GHG emissions reduction, or any other metric.

D.6.2. Net Levelized Cost Curves

Net levelized cost curves present all the relevant benefits and costs, where one of the benefits is displayed along the horizontal axis and the net levelized costs are presented along the vertical axis. These are similar to the levelized cost curves discussed above except that net levelized costs are presented.

Figure D-7 presents an example of a net levelized cost curve for GHG reductions. It uses all the same information as the example in Figure D-6, except that all the benefits but GHG emission benefits are subtracted from the DER costs before they are levelized. The DERs whose net costs are negative in the figure represent those DERs that are cost-effective even without accounting for the value of avoided GHG emissions. The DERs whose net costs are positive represent those that are not cost-effective without accounting for the value of avoided GHG emissions but might be cost-effective once the value of such emission reductions is included in the BCA.



Figure D-7. Example of GHG Net Cost Curve: Hypothetical Set of Multiple DER Types

Net cost curves such as this one can be used for several purposes. For example:

- Net cost curves can be used to prioritize DERs according to those that will reduce GHG emissions at the lowest cost. The leftmost DERs can be prioritized over those on the right.
- Net cost curves can be used to determine which are the lowest-cost DERs available to meet a particular GHG target. Figure D-7 includes a dashed vertical line indicating the hypothetical GHG emission reduction target for this hypothetical jurisdiction. The DERs to the left of that line represent the lowest cost options for meeting this GHG target.
- Net cost curves can be used to identify the marginal net cost of complying with a specific target. Figure D-7 includes a dotted horizontal line that indicates the marginal resource needed to meet the GHG target for this hypothetical jurisdiction. In this example, the marginal resource is EV initiatives for public transportation. As indicated in the previous figure, the cost of this DER is roughly 80 \$/ton GHG. This marginal cost could be used as an input to the BCA; where the value in \$/tons is used as the cost of achieving the jurisdiction's GHG target.
- Net cost curves can also be used to estimate how much of a resource is cost-effective for a given avoided cost value. For example, at a GHG emission reduction value of \$50/ton, all the

resources to the left of the C&I heat pumps program in Figure D-7 are cost-effective. Similar assessments can be used, for example, in comparing non-wires solutions—when their costs are expressed in net levelized cost per distribution kW—to the cost of a substation or other traditional T&D capacity upgrade and/or the value of deferring such an upgrade.

Levelized cost and net cost curves help allow for a clear comparison and ranking across DERs and DER types. The information regarding the amount of savings available from each DER is useful for determining how they meet a jurisdiction's DER goals, relative to each other. This information can be especially useful for jurisdictions seeking to develop portfolios of multiple DER types across a utility service territory, as discussed in Chapter 13.

D.7 Multiple Cost-Effectiveness Tests

D.7.1. Multiple Tests for One DER Type

Some jurisdictions might choose to use secondary tests, as described in Section 3.5. In some cases, it might be helpful to present more than one secondary test. Different tests provide different information, and there may be situations where that additional information is helpful in analyzing and deciding among DER options.

Figure D-8 presents an example of how multiple test results can be used to provide additional information about DER benefits and costs. It shows the BCA results for a hypothetical EE portfolio in terms of benefit-cost ratios, for a hypothetical JST as well as the traditional UCT, TRC, and SCT. Presenting the results this way allows for a quick assessment of how cost-effective this DER is according to different tests.



Figure D-8. Presentation of Multiple BCA Tests: Benefit-Cost Ratios—Hypothetical EE Portfolio

The primary test (JST) used in this example is assumed to include the following impacts: utility system, host customer, other fuel, low-income, and GHG emissions.

For this hypothetical EE portfolio:

- UCT results: The EE is very cost-effective, and the utility system benefits exceed the utility system costs by a factor of two.
- TRC results: The EE is less cost-effective than under the UCT because the TRC Test includes host customer impacts, which are predominantly costs in this example. The TRC also includes low-income impacts and other fuel impacts.
- JST results: The EE is more cost-effective than under the UCT because the JST includes lowincome benefits, other fuel impacts, and GHG benefits—it is more cost-effective than the TRC because it includes GHG benefits.
- SCT results: The EE is more cost-effective than under the JST because the SCT includes other environmental, public health, economic development, and energy security benefits.

D.7.2. Multiple Tests and Multiple DER Types

Some BCA practitioners might choose to use secondary tests and apply them across multiple DER types. Again, different tests provide different information, and there may be situations where that additional information is helpful in analyzing and deciding among DER options.

Figure D-9 presents the BCA results for hypothetical DERs in terms of benefit-cost ratios. Presenting the results this way allows for a quick assessment of how cost-effective the different DER types are according to different tests.



Figure D-9. Presentation of Multiple BCA Tests: Benefit-Cost Ratios—Multiple DER Types

This example uses the same primary test used in the previous sub-section. The primary test is assumed to include the following impacts: utility system, host customer, other fuel, low-income, and GHG emissions.

For this hypothetical EE scenario:

• The results are the same as those presented and explained in the previous section.

For this hypothetical DR scenario:

- UCT results: The DR is reasonably cost-effective—the utility system benefits exceed the utility system costs by a factor of 1.5.
- TRC results: The DR is slightly less cost-effective than under the UCT because the TRC results include some small net costs to the host customer.
- JST results: The DR is slightly more cost-effective than under the UCT because the JST includes low-income and GHG benefits.
- SCT results: The DR is slightly more cost-effective than under the JST because the SCT includes other environmental, public health, economic development, and energy security impacts.

For this hypothetical PV scenario:

- UCT results: The PV is very cost-effective—the utility system benefits exceed the utility system costs by a factor of three. This is because the utility costs are relatively low since most PV costs are borne by the host customer.
- TRC results: The PV is much less cost-effective than under the UCT because the TRC Test includes host customer impacts that are predominantly costs and the host customer costs cover the majority of PV costs.
- JST results: The PV is the more cost-effective than under the TRC Test because the JST includes GHG benefits.
- SCT results: The PV is less cost-effective than under the UCT because the SCT includes host customer costs. It is more cost-effective than under the JST test because the SCT includes other environmental, public health, economic development, and energy security impacts.

For this hypothetical storage scenario:

- UCT results: The storage is reasonably cost-effective—the utility system benefits exceed the utility system costs by a factor of 1.6.
- TRC results: The storage is much less cost-effective than under the UCT because the TRC Test includes host customer impacts, which are predominantly costs in this case.
- JST results: The storage is less cost-effective than under the UCT because the JST includes host customer costs and the storage resource causes a slight increase in GHG emissions because it creates a net increase in electricity generation.
- SCT results: The storage is less cost-effective than under the UCT because the SCT includes host customer costs. It is more cost-effective than under the JST test because the SCT includes other environmental, public health, economic development, and energy security impacts.

For this hypothetical electrification scenario:

- UCT results: The electrification is not cost-effective—with a benefit-cost ratio of 0.9. This occurs because this test does not account for the other fuel savings, which are the primary rationale for the electrification.
- TRC results: The electrification is more cost-effective than the UCT because the TRC Test accounts for the other fuel savings.
- JST results: The electrification is more cost-effective than under the TRC Test because the JST includes GHG emission impacts.
- SCT results: The electrification is more cost-effective than under the JST because the SCT also includes other environmental, public health, economic development, and energy security impacts.

For this hypothetical EV scenario:

- UCT results: The EV is not cost-effective—with a benefit-cost ratio of 0.8. This occurs because this test does not account for the other fuel savings.
- TRC results: The EV is more cost-effective than the UCT because the TRC Test accounts for the other fuel savings.
- JST results: The EV is more cost-effective than under the TRC Test because the JST includes GHG emission impacts.
- SCT results: The EV is more cost-effective than under the JST because the SCT also includes other environmental, public health, economic development, and energy security impacts.

Appendix E. Traditional Cost-Effectiveness Tests

This appendix provides a description of the traditional tests used for assessing DER cost-effectiveness and is intended to provide the theoretical underpinnings of what should be included in the traditional tests, which sometimes differs from actual practice. Chapter 3 describes how use of the NSPM BCA Framework to develop a jurisdiction's primary test compares to these traditional tests. This appendix also emphasizes the distinction between cost-effectiveness analysis and rate impact analysis and refers to Appendix A for more detailed information and guidance on rate impact analyses.

E.1 Overview

This appendix provides information on the five traditional screening tests: the Utility Cost Test, (also known as the Program Administrator Cost Test); the Total Resource Cost Test; the Societal Cost Test; the Participant Cost Test; and the Rate Impact Measure Test. These traditional tests are presented in the *California Standard Practice Manual* and have been used to assess cost-effectiveness for efficiency resources for several decades.⁷²

As discussed in both the introduction to this manual and in Chapter 3, a jurisdiction using the NSPM BCA Framework could develop a primary cost-effectiveness test that fully aligns with one of these traditional tests, or that it is unique to the jurisdiction, depending on the outcome of applying the principles set forth in Chapter 2 to the specific jurisdiction. This appendix describes the key elements of these traditional tests.

For each of the traditional tests, this appendix provides:

- A description of the test;
- The relevance of the test for cost-effectiveness assessment;
- The benefits and costs covered under each test; and
- Limitations of each test.

Table E-1 provides a conceptual overview of the traditional cost-effectiveness tests. Table E-2 provides a summary of the various benefits and costs that, to be consistent with the analytical perspective each test is intended to represent, should be included in these tests. Additional information on each test is provided in the sections that follow.

⁷² While most jurisdictions have historically used the CaSPM as the foundation for their cost-effectiveness tests, in practice many jurisdictions have deviated from those tests and developed modified versions.

Test	Perspective	Key Question Answered	Impacts Accounted For
Utility Cost	The utility system	Will utility system costs be reduced?	Includes the benefits and costs experienced by the utility system
Total Resource Cost	The utility system plus participating customers	Will utility system costs plus program participants' costs be reduced?	Includes the benefits and costs experienced by the utility system, plus benefits and costs to program participants
Societal Cost	Society as a whole	Will total costs to society be reduced?	Includes the benefits and costs experienced by society as a whole
Participant Cost	Customers who participate in a program	Will program participants' costs be reduced?	Includes the benefits and costs experienced by the customers who participate in the program
Rate Impact Measure	Impact on rates paid by all customers	Will utility rates be reduced?	Includes the benefits and costs that will affect utility rates, including utility system benefits and costs plus lost revenues

Table E-1. Conceptual Overview of the Traditional Cost-Effectiveness Tests

Table E-2. Impacts Included in the Traditional Cost-Effectiveness Tests

	UCT	TRC Test	SCT	РСТ	RIM Test
Electric Utility System Impacts	√	✓	✓	-	✓
Gas Utility System Impacts	1	1	✓	-	✓
Other Fuel Impacts	-	✓	✓	✓	-
Host Customer Impacts	-	✓	✓	✓	-
Societal Impacts	-	-	✓	-	-
Host Customer Bill Savings	-	-	-	✓	✓

In the case of the PCT and the RIM Test, while these are traditionally referred to as cost-effectiveness tests, they should not be used to answer the key question of which DERs should be funded or otherwise supported by utilities on behalf of customers. Instead, the PCT should be used to assist with program design and estimating customer participation, and the RIM Test should be used (a) to determine whether rates are likely to increase or decrease, and (b) to determine whether to conduct a rate, bill, and participation analysis.

Each of these tests are discussed in turn below.

E.2 Utility Cost Test

Description: The purpose of the UCT is to indicate whether the benefits of a DER resource will exceed its costs from the perspective of only the utility system.⁷³ The UCT includes all benefits and costs that affect the operation of the utility system and the provision of electric and gas services to customers. For vertically integrated utilities, this test includes all of the benefits and costs that affect utility revenue requirements. For utilities that are not vertically integrated, this test includes all benefits and costs that affect utility revenue requirements, plus additional benefits and costs associated with market-based procurement of electricity and gas services.

Costs Included: The UCT should account for all utility system costs that are incurred to implement the DER resource.

Benefits Included: The UCT should account for all utility system costs that are avoided by the DER resource.

Relevance to DER Assessment: The UCT is useful for identifying the impact of a DER on utility-system costs and average customer bills, and thus is consistent with the principle that DERs are a resource. It is also useful for identifying the extent to which utility investments will provide reduced costs to that same overall group of utility customers, and therefore can have value (among other factors) for informing decisions on relative program priorities, program design (e.g., customer incentive levels) and/or limits on program spending. As discussed in Chapter 3 the UCT should serve as the foundation upon which a jurisdiction's DER assessment test is built. From this foundation, other relevant impacts should be added to align the test with the jurisdiction's applicable policy goals.

E.3 Total Resource Cost Test

Description: One of the key principles of cost-effectiveness assessment is that utility DER investments should be evaluated as a resource and compared with other demand-side and supply-side resources. The TRC does so from the combined perspective of the utility system and participants. Thus, this test includes all impacts of the UCT, plus all impacts on the program participants.

Costs Included: The TRC Test should account for all utility system and program participant costs incurred to implement the DER resource.

Benefits Included: The TRC Test should account for the utility system and program participant benefits that are experienced because of the DER resource.

Relevance to DER Resource Assessment: The TRC Test provides more comprehensive information than the UCT by including the impacts on participating customers. As a result, this test includes impacts on other fuels, which allows for a comprehensive assessment of multi-fuel programs and fuel-switching programs. This test also conceptually includes other non-energy impacts on participants. This is particularly important for low-income programs.

⁷³ The UCT is sometimes referred to as the Program Administrator Cost test in the case of BCA of ratepayer-funded energy efficiency programs where programs are implemented by non-utility administrators. The UCT is a more accurate name because the benefits and costs included in this test are those that affect the utility system, not those that affect the Program Administrator.

E.4 Societal Cost Test

Description: The purpose of the SCT is to indicate whether the benefits of a DER resource will exceed its costs from the perspective of society as a whole. This test provides the most comprehensive picture of the total impacts of a DER resource. This test includes all the impacts of the TRC Test, plus the additional impacts on society. The CaSPM refers to the SCT as a "variation" of the TRC Test (CPUC 2001). Since then, many jurisdictions and many studies have referred to the SCT as a separate test with different implications.

Costs Included: The SCT should account for all costs that are incurred to acquire the DER resource, including all utility system and all non-utility system costs.

Benefits Included: The SCT should account for all of the benefits that result from the DER resource, including all utility system and all non-utility system benefits.

Relevance to DER Resource Assessment: The SCT is useful for identifying the full range of economic impacts on society resulting from the investment in DER resources. It is particularly apt for jurisdictions that have particular interest in a range of societal considerations, such as environmental or economic development concerns, in addition to an interest in minimizing utility system and efficiency program participant costs.

E.5 Participant Cost Test

Description: The intended purpose of this test is to indicate whether the benefits of a DER program will exceed its costs from the perspective of the DER program participant. This test includes all impacts on the program participants, but no other impacts.

Costs Included: The PCT should account for all costs that are incurred by the host customer to install, operate, and maintain the DER, including direct costs and non-energy costs.

Benefits Included: The PCT should account for all benefits experienced by the host customer, including bill savings and non-energy benefits. This test should not include avoided utility system costs as a benefit.

Relevance to DER Resource Assessment: The PCT is not appropriate for assessing the value of DER as a utility system resource because, unlike the other four tests described here, it values benefits based on avoided electricity and gas *rates* rather than on avoided utility system *costs*. That violates the *Conduct Forward-Looking, Long-term, Incremental Analyses* principle, which states that cost-effectiveness analysis should be forward-looking, because electric and gas rates are designed to recover both variable (avoidable) costs and fixed (unavoidable) costs, some of which were incurred in the past. An example would be the cost of previous capital investments in the T&D system or generating capacity in vertically integrated utilities.⁷⁴

That said, the PCT can have value for the purpose of informing program design (e.g., the level of financial incentives to offer prospective participants and/or the need for marketing to better inform

⁷⁴ They may be "avoided" in part by participants, but typically only if a larger portion is then recovered by non-participants. Put another way, a portion of participant benefits is often just a shift in costs from one customer group (participants) to another (non-participants) rather than a true cost savings.

participants of non-energy benefits that they may value) by providing insight into energy bill impact on participants.⁷⁵

E.6 Rate Impact Measure Test

Description: The purpose of this test is to indicate whether a DER resource will increase or decrease electricity or gas rates (i.e., prices). This test includes all of the benefits and costs of the UCT, plus estimates of the utility lost revenues created by DER programs. When regulators take steps to allow utilities to recover the lost revenues of DER programs, through rate cases, revenue decoupling, or other means, then the recovery of these lost revenues will create upward pressure on rates. If this upward pressure on rates exceeds the downward pressure from reduced utility system costs, then rates will increase, and *vice versa*.

Costs Included: The RIM Test should account for all utility system costs that are incurred to implement the DER resource. It should also include lost revenues as a cost. The lost revenues are equal to the host customer bill savings.

Benefits Included: The RIM Test should account for all utility system costs that are avoided by the DER resource.

Relevance to DER Resource Assessment: As provided by the NSPM principles in Chapter 2, and further explained in Appendix A, the RIM Test is designed to address rate impacts and therefore answers fundamentally different questions than does a cost-effectiveness analysis. As such, the RIM Test is limited in determining which DERs are cost-effective—i.e., have benefits that exceed their costs and therefore warrant utility acquisition on behalf of utility customers. This is because the RIM Test does not measure changes in net economic costs across a population; rather, it is a measure of distribution equity. Even in that context, the RIM Test considers only one of the three factors that regulators should consider when exploring distributional equity concerns. It does not consider bill impacts or participation rates, which are critical for understanding distributional equity.

Table E-3 summarizes the purpose of each of the traditional tests, and the relevance to DER assessment.

⁷⁵ The U.S. Department of Energy uses a different test to determine whether to include efficiency measures to participants in federally funded weatherization assistance programs. It uses the savings-to-investment ratio; where the numerator is the present value of net savings in energy, water, non-fuel, or non-water operation and maintenance costs attributable to the proposed energy or water conservation measure, and the denominator is the present value of the cost of the proposed energy or water conservation measure.

Test	Purpose	Relevance to DER Assessment
Utility Cost	Indicates the extent to which ratepayer-funded resources will reduce costs to that same group of ratepayers; provides a foundation for all resource assessment tests	Indicates the impact of the resource on utility system cost and average customer bills
Total Resource Cost	Provides a more comprehensive view of DER impacts than the UCT, including impacts of other fuels, which is helpful for multi-fuel programs, and impacts on DER program participants (if properly applied with symmetrical treatment of benefits and costs)	Indicates the total cost of a resource, regardless of who pays for it
Societal Cost	Most comprehensive test, enabling an assessment of cost- effectiveness based on the universe of benefits and costs of the resource investment	Indicates the full impact of a resource on society
Participant Cost	Useful in program design, to inform appropriate participant incentives	Not relevant for cost-effectiveness testing
Rate Impact Measure	Indicates whether long-term rates will increase or decrease on average	Not appropriate for cost- effectiveness assessment

Table E-3. Summary of the Traditional Cost-Effectiveness Tests

Appendix F. Transfer Payments and Offsetting Impacts

This appendix provides guidance on how to identify whether certain benefits and costs offset each other and should therefore be excluded from the BCA.

F.1 Summary of Key Points

- The term "transfer payment" is used in economics to refer to a one-way payment of money for which no money, good, or service is received in exchange. Typical examples include government payments for programs such as Social Security, Medicare, and unemployment.
- The term "transfer payment" is sometimes used in the context of DER BCAs to refer to a situation where a cost to one party is exactly offset by a corresponding benefit to another party. This raises the question of whether transfer payments should be excluded in the BCA because they cancel each other out.
- In many cases, the transfer payments identified in DER BCAs are not the same thing as transfer payments as defined by economic theory (i.e., they are not a one-way payment of money for which no money, good, or service is received in exchange) and they are not driven by government programs.
- There are some situations in DER BCAs where a DER cost experienced by one party is exactly offset by a corresponding DER benefit experienced by another party. This manual refers to these situations as "offsetting impacts," in order to avoid confusion with transfer payments as defined by economic theory.
- Offsetting impacts in DER BCAs can be identified by considering two criteria. If both of these criteria are met, then the benefits and costs in question are offsetting impacts:
 - The cost in question is not a part of the total cost of the DER and is not a part of the costs avoided by the DER.
 - Both the party incurring the cost and the party receiving the benefit are within the scope of the cost-effectiveness test being used.
- Table F-1 presents a summary of the DER impacts that are sometimes considered transfer payments. A "no" entry indicates that the impact is not an offsetting impact, while a "yes" entry indicates that the impact is an offsetting impact.

BCA Test Used	Financial Incentives to Host Customers	DER Performance Incentives	Wholesale Market Price Effects	Tax Incentives
Utility Cost Test	No	No	No	Not Relevant
TRC Test	Yes	No	No	No
Jurisdiction-Specific Test	Depends	No	No	Depends
Societal Cost Test	Yes	No	No	Yes

Table F-1. Potential Offsetting Impacts

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 268 of 302

F.2 Introduction and Terminology

Some DER BCA studies use the term "transfer payment" to describe this situation where one party experiences a cost and another a commensurate benefit (CA PUC 2001). This term has a specific meaning in economics, and is defined as follows:

> A transfer payment is a one-way payment of money for which no money, good, or service is received in exchange. Transfer payments commonly refer to efforts by local, state, and federal governments to redistribute money to those in need. Typical examples of transfer payments include government programs such as Social Security, Medicare, student grants, and unemployment compensation.⁷⁶

In this manual, the term "offsetting impacts" is used to refer to the situation where a DER cost (or benefit) experienced by one party is exactly offset by a corresponding DER benefit (or cost) experienced by another party and it is appropriate to exclude both impacts from the BCA.

In many cases, the transfer payments identified in DER BCAs are not the same thing as transfer payments as defined by economic theory. They are not a one-way payment of money for which no money, good, or service is received in exchange, and they are not driven by government social service programs.

Nonetheless, there are some situations in DER BCAs where a DER cost experienced by one party is exactly offset by a corresponding DER benefit experienced by another party. In some situations, it may be appropriate to exclude both impacts from a BCA because the net impact is zero, but in other cases it may not be appropriate to exclude both impacts because the two impacts do not truly offset each other.

In this manual, the term "off-setting impacts" is used to refer to the situation where a DER cost (or benefit) experienced by one party is exactly offset by a corresponding DER benefit (or cost) experienced by another party and it is appropriate to exclude both impacts from the BCA. The term "transfer payments" is not used further in this manual, in order to avoid confusion with the economic definition of that term.

The following section describes two criteria for determining when it is appropriate to treat two impacts as offsetting impacts and when it is not. The sections after that discuss several examples of impacts that are sometimes considered offsetting impacts and provide guidance on how to determine whether they should be included or excluded in a BCA test.

F.3 Criteria for Identifying Offsetting Impacts

There are several factors that determine whether two impacts offset each other and should therefore be excluded from the BCA test. These include the nature of the impact and the scope of the costeffectiveness test being used. These two criteria can be used to determine whether two impacts are offsetting impacts:

- The cost in question is not a part of the total cost of the DER and is not a part of the costs of the resources avoided by the DER.
- Both the party incurring the cost and the party receiving the benefit are within the scope of the BCA test. If both parties are not within the scope of the test, then the cost and the benefit will

⁷⁶ <u>http://www.businessdictionary.com/definition/transfer-payment.html.</u>

not offset each other. Determining whether this is the case requires answering the following questions:

- What categories of impacts are included the jurisdiction's primary costeffectiveness test?
- Which party is incurring the cost? Are impacts on that party among the categories of impacts included in the jurisdiction's primary test?
- Which party is receiving the benefit? Are impacts on that party among the categories of impacts included in the jurisdiction's primary test?

Any set of benefits and offsetting costs should be assessed using these two criteria in order to determine whether they should be treated as offsetting impacts in BCA. A set of benefits and costs must meet both criteria in order to be treated as offsetting impacts.

Offsetting impacts should not be included in the BCA test because the benefit offsets the cost. Impacts that are not an offsetting impact should be included in the BCA test because the benefit does not offset the cost.

F.4 Financial Incentives to Implement DERs

Utilities and other DER program administrators offer rebates or other financial inducements to participate in DER programs. Common examples include incentives for efficiency measures or for agreements to participate in DR programs. (See Chapter 4.)

The cost of a financial incentive is experienced by all customers that pay for the DER program. The benefit of a financial incentive is experienced by the DER host customer.

Table F-2 applies the criteria described above to determine whether financial incentives are an offsetting impact. As indicated in the table, financial incentives are not offsetting impacts in the UCT but are in tests that include host customer impacts:

- Under the UTC, financial incentives are not an offsetting impact because one of the parties (host customers) are not a part of the test. In this case, financial incentives should be included in the test as one of the costs of the DER.
- Under cost-effectiveness tests that include host customers, financial incentives are an offsetting impact because they are not a part of the total cost of the DER. Instead, they are an exchange between two parties to help host customers overcome the market barriers to the DER. In this case, financial incentives should not be included in the test. Instead, the DER cost (including costs to both the utility and host customers) should be included in the test.

Table F-2. Financial incentives to Participate in DER programs	Table F-2.	Financial	Incentives	to Participate	e in DER	programs
--	------------	-----------	------------	----------------	----------	----------

Criterion	Answer	Criterion Met?	Analysis and Implication
1. Is the cost a part of the total cost of acquiring the DER? Is the cost a part of the costs avoided by the DER?	No	Yes	Incentives to induce customers to install DERs are not a part of the costs to acquire DERs. Instead, they are a one-way payment to help host customers overcome market barriers to DERs.
2. Are both the party incurring the cost and the party receiving the benefit within the scope of the cost- effectiveness test?	Depends on the test used	Depends on the test used	Party Incurring Cost: Financial incentives to participate in DER programs are a utility system cost. All cost-effectiveness tests must include all utility system costs. Thus, the party incurring this cost is always within the scope of the test. Party Receiving Benefit: Host customers are typically the recipients of financial incentives. Their impacts are included in some tests but not all.

Conclusions:

<u>Utility Cost Test</u>: The financial incentive is not an offsetting impact because the party receiving the benefit is outside the scope of the test.

<u>Tests that include host customer impacts</u>: The financial incentive is an offsetting impact, where the cost of the incentive is offset by the benefit to the host customer.

Implications:

<u>Utility Cost Test</u>: The financial incentive should be included in the BCA test as a utility system cost.

<u>Tests that include host customer impacts</u>: The financial incentive should be included in the BCA test as a utility system cost. The host customer impacts should include only the portion of the measure cost paid by the host customer.

F.5 DER Performance Incentives

Some utilities or DER program administrators are eligible to receive performance incentives for the successful implementation of DERs. The most common example is when electric and gas utilities are subject to performance incentive mechanisms for EE programs. Other more recent examples include performance incentive mechanisms to support DG resources, NWSs, or EV infrastructure investments. (See Chapter 4.)

The costs of performance incentives are experienced by all customers that pay for the DER program. The benefits of performance incentives are experienced by the utility or DER program administrator.

Table F-3 applies the two criteria to determine whether performance incentives are an offsetting impact. As indicated in the table, performance incentives are not offsetting impacts because they are part of the total cost of implementing the DER. Financial incentives are equivalent to profits, which are commonly included as a cost associated with energy resources, as well as other types projects subject to a BCA.

This conclusion applies to all cost-effectiveness tests, because the parties experiencing the costs (all customers) and the parties experiencing the benefits (utilities or program administrators) are within the utility system and therefore should be included in all cost-effectiveness tests. (See Chapter 2.)

Table F-3. DER Performance Incentives

Criterion	Answer	Criterion Met?	Analysis and Implication		
1. Is the cost a part of the total cost of acquiring the DER? Is the cost a part of the costs avoided by the DER?	Yes	No	DER performance incentives are a form of profit paid for the acquisition of DERs. Profits are part of the cost of acquiring DERs and other energy resources. Thus, performance incentives do not meet this criterion and are not offsetting impacts.		
 Are both the party incurring the cost and the party receiving the benefit within 	Yes	Yes	Party Incurring Cost: The costs of DER performance incentives are paid by the customers within the utility system. All cost- effectiveness tests must include all utility system impacts. Thus, the party experiencing the cost is always within the scope of the test.		
the scope of the cost- effectiveness test?		163	Party Receiving Benefit: The recipients of DER performance incentives are also part of the utility system. All cost-effectiveness tests must include all utility system impacts. Thus, the party experiencing the benefit is always within the scope of the test.		
Conclusion:					
All tests: DER performance incentives are not an offsetting impact because they are a part of the cost of the DER.					
Implication: All tests: DER performance incentives should always be included in BCA tests as utility system costs					

F.6 Wholesale Market Price Effects

Investments in some DERs can cause wholesale energy and capacity market prices to decline as a result of reduced demand. This short-term reduction in market prices are typically relatively small but the impacts are felt by all customers buying from the market at the time of the effect, and therefore the total impact can be significant. Some DERs cause increases in energy and peak demand, which has the opposite effect of increasing market prices. (See Chapter 4.)

When DERs cause market prices to be reduced, the costs are experienced by all generators that are selling into the market at the time of the effect, and the benefits are experienced by all customers purchasing from the market at the time of the effect. Conversely, when DERs cause market prices to increase, the costs are experienced by all customers purchasing from the market at the time of the effect, and the benefits are selling into the market at the time of the effect.

Table F-4 applies the two criteria described above to determine whether wholesale market prices effects are an offsetting impact. As indicated in the table, wholesale market price effects are not an offsetting impact because they are part of the cost of the resources avoided by the DERs.

Wholesale market price effects are essentially a reduction (or increase) in suppliers' profits. Profits are a part of the cost of acquiring energy resources and are commonly included in BCAs as a cost, without any offsetting effect on the parties that benefit from the profits. It would be inconsistent to consider changes to wholesale suppliers' profits as an offsetting impact, and therefore exclude them from the BCA, when the profits of other entities that provide energy resources are included in the BCA.

The wholesale market price effects are not an offsetting impact for any cost-effectiveness test, because the parties experiencing the benefits and costs are all within the utility system and all cost-effectiveness tests should include benefits and costs to the utility system. (See Chapter 4.)

Table F-4. Treatment of Wholesale Price Effects

Criterion	Answer	Criterion Met?	Analysis and Implication	
1. Is the cost a part of the total cost of acquiring the DER? Is the cost a part of the costs avoided by the DER?	Yes	No	Reductions (or increases) in wholesale energy prices from DERs will lead to reduced (or increased) profits for the competitive electricity and gas suppliers. Profits are part of the costs of the resources avoided by the DER. Thus, they do not meet this criterion and are not an offsetting impact.	
 Are both the party incurring the cost and the party receiving the benefit within 	Yes	Yes	<i>Party Incurring Cost</i> : ⁷⁷ Competitive electricity and gas suppliers who experience reductions in revenues and profits from this effect are part of the utility system. All cost-effectiveness tests must include all utility system costs. Thus, the party experiencing the cost is always within the scope of the test.	
the scope of the cost- effectiveness test?			Party Receiving Benefit: Market buyers who receive the benefits of lower prices are also part of the utility system. All cost-effectiveness tests must include all utility system costs. Thus, the party experiencing the benefit is always within the scope of the test.	
Conclusion:				

All tests: Changes in wholesale market clearing prices are not an offsetting impact because the profits affected by these changes are part of the costs of acquiring the alternatives to the DER.

Implication:

All tests: Reductions in market clearing prices should be included in all BCA tests as utility system benefits. Increases in market clearing prices should be included in all BCA tests as utility system costs.

In addition, it is useful to consider this issue in the context of the wholesale markets themselves. From the perspective of wholesale market goals, it does not make sense to exclude suppliers' profits from the benefits (or costs) of a DER, for several reasons:

- Wholesale electricity markets are established with the goal of reducing costs to customers. This is achieved by using competitive forces to make suppliers provide their products as efficiently as possible. One way to make suppliers more efficient is to create pressure for them to reduce profits. Therefore, it does not make sense to exclude the benefits of reduced supplier profits from the BCA when this is one of the goals of competitive wholesale markets.
- Another goal of competitive markets is to shift some of the risks of energy resources from customers to the suppliers who have a comparative advantage to manage and mitigate those risks. Shifting risks in this way is clearly a benefit of wholesale electricity markets. Therefore, it does not make sense to exclude this benefit from the BCA by assuming that the profits to suppliers are exactly offset by the costs borne by customers.
- The fundamental market mechanism of balancing demand and supply depends not only on a robust set of options to supply electricity but also the ability for demand to respond to the changes in supply. DERs provide benefits to wholesale markets by allowing for more flexibility in the demand for supplies, i.e., by providing a more dynamic demand curve. Therefore, it does not

⁷⁷ The discussion in this example focuses on cases in which market prices are lowered. However, the results of the same for DERs that may increase market clearing prices. In those cases, both the bearer of the costs (market buyers) and the recipients of the benefits (market sellers) are part of the utility system and therefore within the scope of all cost-effectiveness tests.

make sense to exclude this benefit from the BCA by excluding the effect of the more dynamic demand curve.

F.7 Tax Incentives

Some DERs are eligible for incentives in the form of reduced local, state, or federal taxes. Common examples include federal tax breaks for certain highly efficient EE measures and federal and state tax breaks for installing distributed solar technologies. (See Chapter 4.)

The cost of tax incentives is experienced by the taxpayers who ultimately supply the incentives, and the benefit is experienced by host customers.

Table F-5 applies the two criteria to determine whether tax incentives are an offsetting impact. As indicated in the table, tax incentives are offsetting impacts in some tests but not others. Tax incentives are not a part of the total cost of the DER. Instead, they are one-way payment from the government to individual consumers purchasing eligible DERs. Thus, they meet this criterion for being an offsetting impact.⁷⁸

However, for a tax incentive to be an offsetting impact both parties experiencing the impacts must be a part of the cost-effectiveness test.

- For the UCT: Tax incentives are not offsetting impacts because host customers and taxpayers are not included in this test.
- For the TRC Test: Tax incentives are not offsetting impacts because the party incurring the cost (taxpayers) are outside the scope of this test.
- For the SCT: Tax incentives are an offsetting payment because the scope of this test typically includes all taxpayers and host customers.
- For other tests that include host customer impacts: Tax incentives are offsetting effects if the scope of the test includes taxpayers. The decision to include certain taxpayers in the BCA test might depend upon the jurisdiction. For example, a state regulator may determine that the state's JST should encompass only impacts on its state taxpayers—and therefore would treat local or state tax incentives as offsetting impacts, but federal tax incentives as benefits. Similarly, a municipal utility may consider local tax incentives to be within the scope of its JST, but state or federal incentives to be outside its scope. In that example, a local tax incentive would be treated as an offsetting impact, but a state or federal tax incentive would be treated as a benefit.

⁷⁸ Tax incentives are the only example discussed in this appendix that fits the economic definition of a transfer payment.
Table F-5. Treatment of Tax Incentives

Criterion	Answer	Criterion Met?	Analysis and Implication							
1. Is the cost a part of the total cost of acquiring the DER? Is the cost a part of the costs avoided by the DER?	No	Yes	Tax incentives are not a part of the total cost of the DER. Instead, they are one-way payment from the government to individual consumers purchasing eligible DERs.							
2. Are both the party incurring the cost and the party receiving the benefit within the scope of the cost- effectiveness test? Depends on the test used Depends on the test used Depends on the test used Party Incurring Cost: Taxpayers are the party incurring the cost. Whether they are within the scope of the cost-effectiveness test 2. Are both the party incurring the cost and the party receiving the benefit within the scope of the cost- effectiveness test? Depends on the test used Depends on the test used Depends test used Can depend on both whether the tax credit is local, state, or federal, as well as what impacts are included in the test. Party Receiving Benefit: Host customers are typically the recipients of tax incentives. Their impacts are included in some tests but not all.										
Conclusion: Utility Cost Test: Tax incentives a TRC Test: Tax incentives are not a this test. Societal Cost Test: Tax incentives Participant Cost Test: Tax incentives cope of this test. Other tests that include host cust	Conclusion: Utility Cost Test: Tax incentives are not relevant because host customers and taxpayers are not included in this test. TRC Test: Tax incentives are not an offsetting impact because the party incurring the cost (taxpayers) are outside the scope of this test. Societal Cost Test: Tax incentives are an offsetting payment because the scope of this test includes all taxpayers. Participant Cost Test: Tax incentives are not offsetting impacts because the party incurring the cost (taxpayers) are outside the scope of this test. Other tests the tipelude best externess impacts. Tax incentives are not offsetting effects if the scope of the test includes turpayers.									
Utility Cost Test: Tax incentives should not be included as a benefit or a cost. TRC Test: Tax incentives should be included as a benefit to the host customer. Societal Cost Test: Tax incentives should not be included as a benefit or a cost.										

Other tests that include host customer impacts: Tax incentives should not be included as a benefit or a cost, unless the federal, state, or local taxpayers are outside the scope of the test.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 275 of 302

Appendix G. Discount Rates

This appendix provides guidance on how to determine a cost-effectiveness test discount rate that is consistent with the objective(s) of the cost-effectiveness analysis and the jurisdiction's applicable policy goals. The concepts described in this appendix are applicable to all DERs.

G.1 Summary

- The discount rate reflects a particular "time preference," which is the relative importance of short- versus long-term impacts. A higher discount rate gives more weight to short-term benefits and costs relative to long-term benefits and costs, while a lower discount rate weighs short-term and long-term impacts more equally.
- Different economic actors may have differing discount rates, based on their own time preferences. However, the same discount rate should be used for assessing and comparing different DERs in order to allow for direct comparisons across all resource types.
- There are three categories of discount rates typically considered for DER assessments: WACC, average customers' discount rate, and societal discount rate. A fourth option is some combination of these three categories.
- The choice of discount rate is a decision that should be informed by the jurisdiction's applicable policy goals. Therefore, the regulatory perspective should be used to determine the appropriate discount rate.
- The following steps can assist regulators in determining the discount rate for their costeffectiveness test(s):
 - Articulate the jurisdiction's applicable policy goals.
 - Consider the relevance of a utility's WACC.
 - Consider the relevance of the average utility customer discount rate.
 - Consider the relevance of a societal discount rate.
 - Consider an alternative discount rate.
 - Consider risk implications.
 - Based on these considerations, determine a discount rate that best reflects the jurisdiction's regulatory perspective.

G.2 The Purpose of Discount Rates

Discount rates are an essential aspect for assessing any multi-year project or investment. They allow analysts to compare benefits and costs that occur over different time periods.

Some utility costs, such as power plant siting, licensing, and construction, occur in the short term. Other utility costs such as fuel and O&M stretch into the long-term future. A power plant

takes a few years to build, and then generates electricity for decades. On the other hand, many DER resources can be implemented within a year or two, and then operate for many years thereafter.

The key point is that dollars at different times in the future are not directly comparable values; they are apples and oranges. Applying discount rates turns benefits and costs in different years into comparable values. The choice of discount rates is a critical element of any long-term cost-effectiveness analysis because it has large impacts on the results. This is especially true when the analysis involves long-lived resources.

Figure G-1 illustrates how DER benefits (for instance, avoided generating fuel costs) can be affected by different discount rates. This example starts with an annual fuel costs savings of \$10 per year over the course of a 20-year period. The top, blue line indicates the magnitude of the future avoided fuel costs assuming no discount rate. The other lines present the annual present value of the avoided fuel benefit, depending upon the discount rate used. As indicated, higher discount rates will dramatically reduce the value of avoided fuel savings benefits in Year 20, while lower discount rates have a much smaller impact.



Figure G-1. Implications of Discount Rates (annual present value dollars)

Note: These benefits are presented as real dollars (i.e., adjusted for inflation), and the discount rates are real discount rates.

G.3 Commonly Used Discount Rates

G.3.1. Different Perspectives and Time Preferences

As described earlier in this manual, the regulatory perspective is an important concept for determining a jurisdiction's primary cost-effectiveness test as well as the associated discount rate. The regulatory perspective includes the full scope of issues for which regulators and other relevant decision-makers are responsible. It is typically based upon statutes, regulations, executive orders, commission orders, and ongoing policy discussions.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-001B Page 277 of 302

However, the regulatory perspective can take into consideration other relevant perspectives, including:

- Investor-Owned Utility Perspective: The utility WACC is typically used to indicate the time preference for investor-owned utilities (i.e., reflects the time preference of utility investors, based on the utility authorized return on equity, cost of debt, and debt-to-equity ratio). The key goal of utility investors is to maximize the returns on their investments. Therefore, the time preference of utility investors is not necessarily the same as the time preference of utility customers, or the regulatory time preference.
- Publicly Owned Utility Perspective: Publicly owned utilities, such as public power authorities, municipal utilities, and cooperatives, usually have a different time preference than investor-owned utilities. First, the cost of capital for publicly owned utilities is typically based solely on debt, and therefore is much lower than the WACC of investor-owned

In some ways, the time preference from a regulatory perspective is aligned with utility customers' time preference. In both cases, time preference should be consistent with the objective of identifying those resources that will best serve customers. The time preference from the regulatory perspective, however, captures two additional considerations. First regulators have a responsibility to ensure that utility resources will meet applicable policy goals. Second, regulators have a responsibility to consider both current and future customer interests. For both of these reasons, the regulatory perspective should place a higher value on long-term benefits and costs than the utility customer perspective.

utilities. Second, publicly owned utilities' perspectives, by design, are likely to be more aligned with the time preference of utility customers as a whole.

- Utility Customer Perspective: An objective of utility cost-effectiveness analysis is to
 identify those resources that will best serve utility customers over the long term, while
 also achieving applicable policy goals of the jurisdiction. Thus, the utility customer time
 preference is an important consideration in determining the appropriate discount rate
 for analyses. There are at least two challenges to using customer-focused discount
 rates. First, the customers' cost of capital is only one factor that will influence the
 customers' time preference and thus they may place a different time preference on
 dollars spent on DERs relative to dollars spent on other products or other investments.
 Second, the customer cost of capital varies considerably across customer classes, and
 also across customers within classes. Any one cost-effectiveness test, however, can use
 only one discount rate. Therefore, to the extent that the customer cost of capital is used
 to inform the determination of a discount rate, it should be an average cost of capital
 that represents the broad range of utility customers.
- *Host Customer Perspective:* The host customers' perspective is directly relevant when applying the PCT because the goal of that test is to indicate the impact on host customers only. For this test, a discount rate reflecting the host customers' time preference would be appropriate.⁷⁹

⁷⁹ The Participant Cost Test can be useful as a secondary test to provide useful information regarding the likelihood of customers adopting DERs, either on their own or with support from utility initiatives. This information can be helpful for designing DER initiatives, determining how much financial support to offer host DER customers, and forecasting future deployment of DERs.

Table G-1 summarizes several types of discount rates that could be used for energy resource cost-effectiveness assessment. For each type of discount rate, it indicates the time preference represented by that rate, a range of typical values, some brief notes, and sources.

Type of Discount Rate	Potential Indicator of Time Preference	Typical Values	Notes and Sources
Societal	Societal cost of capital, adjusted to consider intergenerational equity or other societal values	<0% to 3%	In addition to low-risk financing, government agencies have a responsibility to consider intergenerational equity, which suggests a lower discount rate (US OMB 2003). Society's values regarding environmental impacts might warrant the use of a negative discount rate (Dasgupta, Maler, and Barrett 2000).
Low-Risk	Interest rate on 10-year U.S. Treasury Bonds	-1.0% to 3%	Over the past 20 years the real interest rate on 10-year U.S. Treasury Bonds ranged between roughly -1.0% and 3.0% percent (multpl.com).
Utility Customers on Average	Customers' opportunity cost of money	varies	Customers' opportunity costs can be represented by either the cost of borrowing or the opportunity costs of alternative investments (Pindyck and Rubinfeld 2001, 550). The real rate on long-term government debt may provide a fair approximation of a discount rates for private consumption (US OMB 2003).
Publicly Owned Utility	Publicly owned utility's cost of borrowing	3% to 5%	Publicly owned utility costs of capital are available from the Federal Energy Regulatory Commission Form 1, Securities Exchange Commission 10k reports, and utility annual reports.
Investor- Owned Utility	Investor-owned utility's weighted average cost of capital	5% to 8%	Investor-owned utility costs of capital are available from the Federal Energy Regulatory Commission Form 1, Securities Exchange Commission 10k reports, and utility Annual Reports.

Table G-1. Discount Rate Options for Cost-Effectiveness Analys
--

Note: Typical values of discount rates are in real terms, as opposed to nominal. Real interest rates take into account the effects of inflation whereas nominal rates have not been adjusted for inflation. Real discount rates should always be applied to real cash flows, and nominal discount rates should always be applied to nominal cash flows. The utility cost of capital should be after-tax.

The typical values presented in Table G-1 are provided for illustrative purposes only; other values outside these ranges are also possible. Other points to consider include that: these values can change over time according to changing economic conditions; there are multiple options for determining a low-risk discount rate; and different utility customers will have different time preferences, which can be determined in multiple ways.

Some practitioners recommend that the choice of discount rate should reflect the perspective represented by the cost-effectiveness test in use. For example, the U.S. Department of Energy and Environmental Protection Agency's National Action Plan on Energy Efficiency (NAPEE 2007, 5-4) states that:

- The societal discount rate should be applied when using the SCT.
- The utility weighted average cost of capital should be applied when using the UCT, the TRC Test, or the Rate Impact Measure (RIM) Test.
- A customer discount rate should be used when applying the PCT.

While there is some logic to the concept of matching the discount rate to the perspective of the test used, this logic must be applied carefully. First, it is important to recognize the role of the applicable policies in developing the cost-effectiveness test and in determining the appropriate

time preference. Second, it is important to be clear on whose perspective is actually represented in particular discount rates.

G.3.2. Role of the Cost of Capital and Other Considerations

In general, the cost of capital is a key factor in determining discount rates. It indicates the time value of money (or the opportunity cost for alternative investments) for the relevant entity. However, cost of capital is not the only factor that dictates the appropriate discount rate to use for DER investments.

For example, the primary objective of a utility cost-effectiveness analysis is to identify those utility resources that will best serve customers over the long term, while also achieving applicable policy goals of the jurisdiction. In light of this objective, the time preference for cost-effectiveness analysis should account for more than just the cost of capital; it should also account for the value of utility service over the long term and applicable policy goals. In other words, important utility services (such as providing safe and reliable power) and important policy goals (such as protecting low-income customers or promoting economic development) are all factors that affect the time preference relevant to the cost-effectiveness analysis.

This point that a discount rate used for cost-effectiveness analysis could reflect more than just the cost of capital is at least one basis for the application of the societal discount rate. That rate, which is used in multiple applications, reflects more than simply the cost of capital to society. It also reflects societal values and priorities, such as long-term benefits to society, achieving societal goals, addressing the needs and interests of multiple entities across society, and more — some or all of which may be considerations included in a jurisdiction's policies.

G.4 Risk Considerations

Risk is often cited as an important factor to consider when determining a discount rate, because risk can affect the value that one might place on long-term versus short-term impacts. However, risk can be represented in different ways in a cost-effectiveness analysis, and it is important to be careful that any treatment of risk in the discount rate recognizes how risk is addressed in the rest of the analysis to ensure that there is no double-counting or under-counting of risk.

Risks can vary considerably across different types of utility resources. For example, EE resources tend to create relatively low risk; generators create different amounts of capital cost, siting, and construction risks; fossil-fueled generators create price escalation and volatility risks; and T&D facilities impose their own kinds of risks (Ceres 2012).

In general, it is preferable to account for such resource-specific risks separately and explicitly for each resource type, rather than embed it in a discount rate. Discount rates are applied to all resources in a cost-effectiveness analysis. Applying a single discount rate to all resources to reflect risks associated with any one of those resources, could conflate the treatment of resource-specific risk with the overall choice of time preference. Instead, resource-specific risk should be accounted for in developing the benefit cost and inputs to the cost-effectiveness analysis.

G.5 Determining the Discount Rate

Ultimately, the choice of discount rate is a decision regarding how much weight to give to longterm versus short-term benefits and costs. This decision should be driven by the regulatory perspective, and each jurisdiction should determine a discount rate based on its own policies and goals. Regulators/decision-makers can take the following steps to make this determination.

Step A: Articulate Policy Goals

Chapter 3 describes how regulators should identify and articulate policy goals. Those same policy goals should be articulated and applied when determining the discount rate.

Step B: Consider the Utility Investor Perspective

Regulators should consider whether the utility WACC represents the regulatory time preference. This decision can be based on answers to these questions: Is the utility investor time preference consistent with the jurisdiction's regulatory perspective and policy goals, is the utility investor time preference the appropriate time preference for resource planning, and does the utility WACC accurately reflect the cost of capital of DERs and the other resources being assessed? If the answer to any of these questions is "no," then alternative discount rates should be utilized, such as those based on customer or societal perspectives.

Step C: Consider the Average Customer Discount Rate

Regulators should consider whether the average customer discount rate represents the regulatory time preference. Should the discount rate be based on the average utility customer time preference? Does this time preference adequately address applicable policy goals and interests of future utility customers?

Step D: Consider the Societal Discount Rate

Regulators should also consider whether a societal discount rate is appropriate for the primary cost-effectiveness test, i.e., is a societal time preference consistent with the jurisdiction's applicable policy goals?

Step E: Consider an Alternative Discount Rate

Regulators should also consider whether to use a discount rate that is not tied to any one of the three perspectives described above. The regulatory perspective may be different from the perspective of utility investors, customers, and society. Thus, the regulatory time preference and discount rate could be different as well.

Step F: Consider Risk Implications

Resource-specific risk issues are best accounted for in estimating the costs of each resource, for example in the resource-specific cost of capital, as adjustments to a resource's benefits or costs, and/or in the avoided cost portfolio modeling process. Nonetheless, there may be situations where the DER benefits or costs do not properly reflect resource-specific risks. For example, the full set of risks associated with avoided costs (e.g., risks associated with avoided fuel costs, risks associated with construction costs) are sometimes not captured in the cost-effectiveness inputs.

In such situations, regulators and other decision-makers may choose to apply a discount rate to reflect the risk benefits (or risk costs) of DER resources, if those benefits or costs are not otherwise accounted for in the inputs to the analysis.

Step G: Determine a Discount Rate

Based on the considerations described above, regulators should determine a discount rate that best reflects the jurisdiction's regulatory perspective. Table G-2 offers suggestions for how this determination might be made.

Consideration	If the answer is "yes"
Time Preference Considerations:	
Does the regulatory perspective suggest the same time preference as utility investors?	Choose a discount rate equal to the utility WACC.
Does the regulatory perspective suggest placing a higher value on long-term impacts than utility investors?	Choose a discount rate less than the utility WACC.
Does the regulatory perspective suggest the same time preference as that of all utility customers?	Choose a discount rate that is represents all utility customers on average.
Does the regulatory perspective suggest the same time preference as that of society?	Choose a societal discount rate.
Does the regulatory perspective suggest placing a lower value on long-term impacts than society does?	Choose a discount rate greater than a societal discount rate, or at the high end of the range of societal discount rates.
Risk Considerations (for use in situations where resource-sp	ecific risks are not accounted for in the BCA inputs):
Will DERs result in a net reduction in risk relative to alternatives?	Choose a relatively low-risk discount rate, such as the societal discount rate.
Will DERs result in a net increase in risk relative to alternatives?	Choose a relatively high discount rate.

Table G-2. Considerations for Determining a Discount Rate

Appendix H. Energy Efficiency: Additional Guidance

This appendix includes guidance on several detailed BCA issues not covered in Chapter 6 (Energy Efficiency Resources) of this manual but which were addressed in the 2017 NSPM for EE and which may still be useful for some jurisdictions.

These BCA issues or considerations are:

- H.1 Assessment Level;
- H.2 Analysis of Early Replacement (of functioning equipment);
- H.3 Free-Riders and Spillover.

H.1 Assessment Level

The cost-effectiveness of efficiency resources can be assessed at several levels of aggregation. Assessments can focus on individual measures, individual customer-specific projects, individual programs combining multiple measures and/or projects, sectors (e.g. all residential or all business programs), or portfolios of programs (across all sectors). This appendix discusses the advantages and disadvantages of conducting cost-effectiveness analyses at each of those levels. It also discusses the level at which fixed costs should be included in analyses.

Assessment level options include the following, and are each described in turn:

- Measure-level
- Project -level
- Program-level
- Sector-level
- Portfolio-level

H.1.1. Measure-Level Assessment

Resource assessment at the measure level means that each individual measure promoted by an efficiency program must be cost-effective on its own. Screening at the measure level is the most restrictive application of the cost-effectiveness tests.

Measure-level application of cost-effectiveness requirements will essentially guarantee that every measure included in an efficiency program will be cost-effective on its own. However, application of cost-effectiveness requirements at that level can have perverse implications. In some cases, it could reduce the overall net economic benefits of efficiency investments. That can occur for any of the following reasons:

• A customer's interest in a non-cost-effective measure may be key to persuading the customer to install a package of measures that are cost-effective in aggregate. In such cases, the flexibility to

promote the non-cost-effective measure as part of a package will lead to greater overall net benefits.

- A customer's interest in a non-cost-effective measure may be key to the development of a relationship with the customer that can lead to installation of cost-effective measures in the future. In that sense, promotion of the non-cost-effective measure can be analogous to a marketing investment.
- Installation of a non-cost-effective measure may be necessary in order to technically or safely enable the installation of other cost-effective measures. An example of this would be the installation of non-cost-effective mechanical ventilation in order to make indoor air quality acceptable when tightening up a building.

Another disadvantage of requiring all measures to be cost-effective is that it can be difficult to account for non-energy impacts, hard-to-monetize impacts, or additional considerations at the measure level. Some non-energy impacts, such as improved health and safety, are obtained through a package of multiple measures, and it is impractical to apply such impacts to each measure.

H.1.2. Project-Level Assessment

Resource assessment at the project level means that the combination of measures implemented together in a package for an individual customer must be cost-effective on its own. Project-level assessments are typically conducted only for projects undertaken by larger business customers for which the transaction cost of a site-specific assessment can be justified.

Project-level application of cost-effectiveness requirements will essentially guarantee that every project included in an efficiency program will be cost-effectiveness on its own. However, application of cost-effectiveness requirements at that level can have some (though fewer) of the perverse implications of measure-level cost-effectiveness requirements. Specifically, supporting the implementation of a non-cost-effective package of measures of interest to a customer can facilitate development of a relationship with customer that can produce a more cost-effective project later. Also, depending on whether and how participant non-energy benefits are included in cost-effectiveness assessments, the full value of non-energy benefits of a project may not be captured in project-level cost-effectiveness assessments.⁸⁰

H.1.3. Program-Level Assessment

Resource assessment at the program level means that the measures and/or projects within a program must be cost-effective collectively. Some individual measures and/or projects may not be cost-effective on their own but could still be included in the program if the overall program were cost-effective.

The primary advantage of this approach is that it best represents the benefits and costs of initiatives that combine a set of actions (e.g., marketing, education, technical support, financial support, etc.) into

⁸⁰ The focus of this discussion is solely on the use of cost-effectiveness analysis to determine which investments merit acquisition from either utility system or broader perspectives. Efficiency programs targeted to large business customers often present benefits and costs to individual customers from the customer's perspective (i.e. using retail energy prices rather than avoided system costs, as well as considering customer non-energy benefits that may or may not be part of a jurisdiction's cost-effectiveness test). Similarly, some low-income programs base the determination of which measures to install on the savings-to-investment ratio (i.e., benefit-to-cost ratio) derived using the customer's retail rate. The merits of such customer-focused analyses are fundamentally different from those discussed here regarding utility system resource analyses.

a single package offered to customers. In addition, resource assessment at the program level avoids the problems noted above regarding missing the interrelationships between measures. These include technical connections and the ability to engage customers in ways that can lead to increasing net economic benefits, as well as the ability to properly capture customer non-energy benefits where warranted.

A disadvantage of this approach is that a program might include one or more measures that are not individually cost-effective and are not needed to account for the concerns addressed above. This has the effect of decreasing to some extent the overall cost-effectiveness of the program. However, this concern can be addressed with sound program design. Efficiency program planners and designers should include only those efficiency measures that effectively contribute to achieving the specific goals of the program.

One other potential concern with program-level screening is that it might preclude certain special programs that address important objectives at the sector or portfolio level. For example, pilot programs to test new and unproven program designs might not appear cost-effective but might provide future sector or portfolio benefits that cannot be identified in the present. For that reason, jurisdictions that apply program-level screening may want to allow these types of programs to be considered in a sector-level assessment.

H.1.4. Sector-Level Assessment

Resource assessment at the sector level means that the programs within a sector (e.g., low-income, residential, commercial and industrial)⁸¹ must be cost-effective collectively. Some programs may not be cost-effective on their own but could still be implemented if the combined impact of all of the programs targeted to a given sector were cost-effective.

The primary advantage of this approach is that it indicates the benefits and costs of initiatives to provide a package of efficiency services to an entire sector. This may allow for non-cost-effective programs to be provided to a sector for the purpose of providing a complete set of efficiency services to that sector—an objective often driven by concerns about equitable access to efficiency programs across a large range and number of customers.

The primary disadvantage of this approach is that it could result in the inclusion of efficiency measures or programs that are not individually cost-effective, thereby decreasing the economic value of the suite of programs for that sector.

H.1.5. Portfolio-Level Assessment

Evaluation at the portfolio level means that the programs within a portfolio (i.e., combining all programs together) must be cost-effective collectively. Some programs may not be cost-effective on their own but could still be pursued if the combined impact of all of the programs was cost-effective.

⁸¹ Some jurisdictions treat low-income programs as their own "sector," because of the special consideration often given to such customers in program design and delivery. Others treat low-income programs as part of the residential sector. Alternatively, though commercial and industrial customers could be considered to be different "sectors," most efficiency programs targeted to business customers do not differentiate between those two groups of customers, creating what are called business, non-residential, or commercial & industrial (C&I) sector programs. For the purpose of this manual, lowincome, residential, and C&I are noted as three sectors of interest for illustrative purposes only. The conceptual discussion in this section applies regardless of whether low-income is treated as its own sector or as part of the residential sector and regardless of whether commercial and industrial are treated as their own sectors or combined.

The primary advantage of this approach is that it indicates the benefits and costs of the entire suite of EE programs.

The primary disadvantage of this approach is that it could result in implementing efficiency measures or programs that are not cost-effective, thereby decreasing the economic value of the overall portfolio.

H.1.6. Properly Accounting for Fixed and Variable Costs

A variety of costs are incurred in the acquisition of efficiency resources. It is important that those costs be included at the proper analytical level—e.g., measure, program, sector and/or portfolio—when analyzing the economics of efficiency resources. In a nutshell, only EE costs that are variable at a given analytical level should be included in cost-effectiveness analysis for that level because they are the only costs that can be avoided as a result of the analysis. EE costs that are largely fixed at a particular analytical level should not be "allocated" or otherwise included *at that level*. Doing so could lead to rejection of investments whose marginal benefits exceed their marginal costs, thereby lowering net economic benefits. That does not mean that EE costs that are fixed at a given analytical level should be omitted or ignored altogether. Instead, they can and should be included at higher level analyses at which they are variable and therefore are avoidable.

For example, when assessing the economics of efficiency measures, one should include only costs that largely increase or decrease in proportion to the number of measures installed. That will obviously include the cost of the measures themselves and could also include some program costs that are largely variable. Examples would include rebate processing costs, if the program administrator is paying a vendor a price for every rebate processed, and inspection costs if the program is committed to inspecting a certain percentage of all projects.⁸² However, other EE program costs that are either largely fixed or do not change in proportion to program participation levels, such as the costs of marketing⁸³ or managing and evaluating the program, should not be included in the economic analysis of individual measures. Rather, they should be included only in a program-level cost-effectiveness assessment.

Similarly, portfolio costs that are either largely fixed or do not change in proportion to the number of programs or participation levels in those programs should not be allocated to programs for the purpose of analyzing the economics of individual programs. Rather, they should only be included at portfolio-level cost-effectiveness analysis. Such costs can include portfolio-level marketing, management, and evaluation costs.

The tables below illustrate the importance of accounting for largely fixed costs at the proper analytical level. Table H-1 shows that for each of five programs analyzed, the benefits exceed the variable costs of the programs. When largely fixed portfolio costs (equal to about 25 percent of the sum of the five program costs) are added to the sum of the variable impacts of the five programs, the portfolio itself is shown to be cost-effective, providing total net benefits of \$800,000.

⁸² Alternatively, if the program is committed to inspecting enough projects to get a statistically valid sample, such that the number of inspections would not change significantly or at all between a level of 2000 and 10,000 participants, then such inspection costs should be treated as largely fixed and captured at the program level rather than at the measure level.

⁸³ Marketing costs can be somewhat variable in the sense that more marketing should lead to more participation. However, that relationship is rarely linear with the number of measures installed. In addition, and perhaps more importantly, program marketing budgets are often treated as largely fixed. That is, while marketing can play an important role in driving program participation, the costs of marketing do not go up and down as the number of participants goes up and down.

	Benefits (\$000)	Costs (\$000)	Net benefits (\$000)	Positive net benefits?
Program 1	\$500	\$250	\$250	Yes
Program 2	\$300	\$200	\$100	Yes
Program 3	\$1000	\$400	\$600	Yes
Program 4	\$500	\$300	\$200	Yes
Program 5	\$1000	\$850	\$150	Yes
Sum of all programs	\$3300	\$2000	\$1300	Yes
Portfolio-level costs	\$0	\$500	-\$500	
Total portfolio impacts	\$3300	\$2500	\$800	Yes

Table H-1. Proper Analysis with 25 Percent Fixed Portfolio Costs Included at Portfolio-Level Analysis

Table H-2 shows that when the fixed portfolio-level costs are improperly allocated as 25 percent "adders" to each of the programs, the fifth program is no longer seen as cost-effective. If that program is then removed from the portfolio, but with portfolio costs remaining unchanged, the portfolio net benefits decline by \$150,000 (i.e., the marginal impact of the fifth program on the portfolio) to \$650,000.⁸⁴ In short, including fixed costs at the improper level can reduce the economic benefits of efficiency resource acquisition.

	Benefits (\$000)	Costs (\$000)	Net benefits (\$000)	Positive net benefits?
Program 1	\$500	\$313	\$188	yes
Program 2	\$300	\$250	\$50	yes
Program 3	\$1000	\$500	\$500	yes
Program 4	\$500	\$375	\$125	yes
Program 5	\$1000	\$1063	-\$63	no
Sum of all programs	\$3300	\$2500	\$800	yes
Portfolio-level costs	I	ncluded as a	dder for each pi	rogram
Total portfolio if non-cost- effective programs excluded	\$2300	\$1650	\$650	yes

Table H-2. Improper Analysis with 25 Percent Fixed Portfolio Costs Allocated to Individual Programs

H.2 Analysis of Early Replacement

Early replacement occurs when a functioning piece of equipment is replaced with a more efficient model before it normally would have been replaced. This section provides guidance on how to analyze the

⁸⁴ Removing the fifth program would require a reallocation of the fixed portfolio cost to the remaining four programs (i.e. each of the remaining four programs would now be allocated a larger portion of the fixed portfolio costs). In this example, the four remaining programs would still all be cost-effective even after absorbing this larger allocation. However, under a different set of example programs, it is possible that the resulting larger allocation of fixed costs would render another program cost-ineffective.

benefits and costs of such early replacement efficiency measures. It also addresses why costeffectiveness analysis of early replacement measures and programs requires special attention, as compared to other common measure categories.

Efficiency measures typically fall into one of four categories:

- *New Construction:* in which a building is going to be constructed, and an efficiency program prompts developers, builders, or contractors to install more efficient products or use more efficient construction practices than they otherwise would have.
- *Time-of-Sale/Natural Replacement:* in which a product is going to be sold and purchased, such as when an appliance breaks down and needs to be replaced, and an efficiency program is designed to persuade a vendor to sell and/or a customer to purchase a more efficient product than they otherwise would have.
- *Retrofit:* in which efficiency programs incentivize customers to install new efficiency measures in an existing space, such as an un-insulated attic.
- *Early Replacement:* in which an existing inefficient product is functioning and would not otherwise be replaced until a future year, and an efficiency program prompts a customer to replace it with a more efficient product sooner than he or she otherwise would have.

For the first three of those efficiency measure classifications, the cost impacts are commonly felt only in the first year (i.e., the incremental cost of an efficiency upgrade over a standard measure that would otherwise have been purchased or the full cost of a retrofit measure). The savings are thus simply the difference between the baseline efficiency and the new efficiency that will recur annually for the life of the measure.

Characterization of both the costs and savings of early replacement measures can be more complicated for two reasons:

- Early replacement changes the timing of costs relative to when they could be incurred in the baseline scenario (i.e., absent the early replacement)—at least in cases where a jurisdiction chooses to include participant benefits and costs; and
- That change in timing can lead to the need to account for multiple baseline assumptions (assumptions that change over time) for both costs and savings.

This section provides guidance on how to account for changes in the timing of costs, and accounting for multiple baselines for both costs and savings/benefits.

Accounting for Changes in the Timing of Costs

Under an early replacement scenario, there is the initial full cost of the replacement product. However, there are also potential cost savings from not having to buy the new product that would otherwise have been purchased several years into the future.

Consider, for example, the following hypothetical early replacement scenario:

- The customer has a 10-year-old and still functioning heating system with a 70 percent efficiency rating, and the heating system is normally assumed to last 15 years;
- Absent an efficiency program influence, the customer is expected to replace its 10-year-old heating system in five years with a new 90 percent efficient model that will cost \$5,000;
- With the efficiency program influence, the customer decides to scrap its existing inefficient heating system and replace it today with a new 90 percent efficient model that costs \$5,000.

In this case, there would be only five years of savings from the early replacement. If the costeffectiveness test includes participant impacts, the net cost of the efficiency resource is equal to the \$5000 initial cost of the early replacement *minus the NPV of the benefit of deferring a new purchase from the beginning of Year 6 to the beginning of Year 16.*⁸⁵ It is critically important that the reduction in cost associated with deferring the next new purchase be incorporated into cost-effectiveness analyses. To not account for it would result in markedly overstating the costs of early replacement measures and programs.⁸⁶

Calculating the value of that deferral requires a cost amortization approach. This serves to align the mismatched timing of costs under the baseline condition and the early replacement condition, as illustrated in Table H-3.

In short, the amortizing or annualizing of the different purchase times under the baseline and early replacement scenarios has the effect of lining up costs so that the only difference is five years of annualized costs under the early replacement scenario. (The annualized cost under the baseline and early replacement scenarios are the same in Years 6 through 20, cancelling each other out.) Importantly, that also aligns the cost analysis with the benefits analysis (i.e., both benefits and costs occur only in Years 1 through 5).

Fable H-3. Amortization to Address Mismatched	l Timing of Baseline ar	nd Early Replacement Costs
--	-------------------------	----------------------------

	<u>Savings</u>	
\$5000	Installed Measure Efficiency	90%
\$5000	Standard New Product Efficiency	90%
15	Existing Efficiency	70%
5	Savings Annual Value (Years 1-5)	\$600
3%	Savings Annual Value (Years 6 and Beyond)	\$0
	\$5000 \$5000 15 5 3%	Savings\$5000Installed Measure Efficiency\$5000Standard New Product Efficiency15Existing Efficiency5Savings Annual Value (Years 1-5)3%Savings Annual Value (Years 6 and Beyond)

Mismatched Timing of Costs Incurred under Baseline and Early Replacement Program Scenarios

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20
Baseline	-	-	-	-	-	\$5000	-	-	-	-	-	-	-	-	-	-	-	-	-	\$5000
Early Replace	\$5000	-	-	-	-	-	-	-	-	-	-	-	-	-	-	\$5000	-	-	-	\$5000

Net Costs and Benefit	s of Early Retirement	Calculated through	Cost Amortization

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	NPV
Costs	Costs																				
Baseline	-	-	-	-	-	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$4313
Early Replace	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$6231
Net	\$407	\$407	\$407	\$407	\$407	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	\$1918
Benefits	\$600	\$600	\$600	\$600	\$600	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	\$2830
Net Benefits	\$141	\$141	\$141	\$141	\$141	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	\$912
Benefit-Cost	3enefit-Cost Ratio									1.48											

⁸⁵ Year 6 is when the customer would otherwise have had to buy a new replacement heating system; Year 16 is when the customer will have to replace the new heating system that was just installed.

⁸⁶ Again, this is only an issue if the cost-effectiveness test includes participant impacts. If it does not, the change in timing of costs associated with future equipment purchases is not relevant.

Accounting for Multiple Baselines for Both Costs and Savings

Unlike in the more straightforward example above, there can also be differences between the cost and efficiency of the early replacement measure that is installed today and the standard new product that would have otherwise been installed five years from now. For example, consider the following modifications to the hypothetical scenario outlined above:

- The customer has a 10-year-old and still functioning heating system with a 70 percent efficiency rating;
- This class of products is normally assumed to last 15 years, so absent an efficiency program influence, the customer is expected to replace its 10-year-old heating system in five years;
- The standard new heating system five years from now is expected to be an 85 percent efficient model that costs \$4500;
- Within 10 years, the standard new heating system is expected to be a 90 percent efficient model that costs \$5000;
- With the efficiency program influence, the customer opts to scrap its existing old inefficient heating system and replace it today with a new 90 percent efficient model that costs \$5000. The new model is not only more efficient than the old heating system it is replacing, but also more efficient than the new heating system the customer would have bought five years from now.

In this case, as depicted in the bottom of

Table H-4, there would be five years of the same level of savings as assumed in the first hypothetical example depicted in Table H-3 (i.e., the difference between the old 70 percent and the new efficient 90 percent efficient model). However, unlike in the Table H-3 example, there would continue to be savings in Years 6 through 20, though the magnitude of those savings would be lower than in the first five years (i.e., the difference between a standard new 85 percent efficient model and an efficient new 90 percent efficient model). Thus, in the hypothetical example, the NPV of benefits is more than \$1300 greater (\$4140 vs. \$2830) than in the Table H-3 example.

On the cost side of things, there would not only be a difference between no baseline cost and the amortized costs of the 90 percent efficient model for the first five years, but also a slightly higher amortized cost in the subsequent 15 years to reflect the difference in cost between a new 85 percent efficient model and a new 90 percent efficient model. Thus, in this hypothetical example, the NPV of costs is also greater—by over \$400 (\$2349 vs. \$1918)—than in the Table H-3 example.

The net effect of these changes in benefits and costs is an increase in net benefits per measure of nearly \$900 (i.e., \$1791 vs. \$912) relative to the net benefits of the Table H-3 example. It should be noted that the direction of this change is unique to this set of hypothetical assumptions. For example, if the cost of a new 85 percent efficient model in Year 6 was assumed to be \$3500 instead of \$4500 (with the 90 percent efficient model still costing \$5000), the net benefits would be virtually identical to those of the example in Table H-3. If the 85 percent efficient model cost only \$2400 (with the 90 percent efficient model still costing \$5000), the measure would actually fall below a 1.00 benefit-cost ratio.

Table H-4. Amortization to Address Multiple Baselines for Early Replacement

Costs		Savings	
Efficiency Measure Cost	\$5000	Installed Measure Efficiency	90%
Standard New Product Cost	\$4500	Standard New Product Efficiency	85%
Resource Life	15	Existing Efficiency	70%
Existing Product Remaining Life	5	Savings Annual Value (Years 1-5)	\$600
Real Discount Rate	3%	Savings Annual Value (Years 6 and Beyond)	\$124

Mismatched Timing of Costs Incurred under Baseline and Early Replacement Program Scenarios

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20
Baseline	\$0	\$0	\$0	\$0	\$0	\$4500	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$5000
Early Replace	\$5000	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$0	\$5000	\$0	\$0	\$0	\$5000

Net Costs and Benefits of Early Replacement

	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	NPV
Costs																					
Baseline	-	-	-	-	-	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$366	\$3882
Early Replace	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$407	\$6231
Net Cost	\$407	\$407	\$407	\$407	\$407	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$41	\$2349
Benefits	\$600	\$600	\$600	\$600	\$600	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$124	\$4140
Net Benefits	\$193	\$193	\$193	\$193	\$193	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$83	\$1791
Benefit-Cost Ratio											1.76										

H.3 Free-Riders and Spillover

This section describes how to address free-riders and spillover effects in cost-effectiveness analyses, for those jurisdictions that focus on net savings in BCAs.

In jurisdictions that focus on net savings for their cost-effectiveness analyses:

- The treatment of free-ridership and spillover effects should be a function of the categories of impacts that a jurisdiction chooses to include in the cost-effectiveness test it adopts pursuant to the NSPM BCA Framework outlined in Chapters 2–3.
- With regard to free-riders:
 - Financial incentives paid to free-riders are a cost only if the costeffectiveness test excludes participant impacts; otherwise the value of the financial incentive to the participant offsets the cost of the financial incentive to the utility system. In other words, the net cost of free-riders is zero under any test that includes participant impacts.
 - No benefits from free-riders should be included in any cost-effectiveness test.
- With regards to spillover:
 - There are no costs associated with spillover in jurisdictions whose costeffectiveness test includes only utility system impacts. Spillover should increase costs under tests that include participant impacts.

• Spillover increases benefits in every test.

Table H-5 summarizes which categories of impacts are affected by free-rider and spillover effects, as further discussed below.

Category	Free-F	liders	Spillover			
	Costs	Benefits	Costs	Benefits		
Utility System Impacts	Increase	n/a	n/a	Increase		
Participant Impacts	Decrease	n/a	Increase	Increase (if applicable)		
Other Impacts	n/a	n/a	Increase (if applicable)	Increase (if applicable)		
Total/Net Impact	Increase only if test <i>excludes</i> participant impacts; otherwise no net effect	No effect under any test	No increase if test includes only utility system impacts; otherwise an increase	Increase under every test		

Table H-5. Categories of Impacts Affected b	y Free-Riders and Spillove
---	----------------------------

H.3.1. Applicability and Definitions

This section addresses the economic concepts underpinning how free-ridership and spillover effects should be treated in cost-effectiveness analyses in jurisdictions that choose to focus on net savings. This section does not address the relative merits of focusing on net savings versus focusing on gross savings, as that is beyond the scope of a guidance document focused solely on the construct and application of cost-effectiveness analysis. This section has no relevance to or application for cost-effectiveness analyses in jurisdictions that choose to focus on gross impacts.

Key definitions to consider in applying guidance from this section are as follows:

- *Free-ridership* refers to efficiency program savings that would have occurred in the absence of the program.⁸⁷
- *Spillover* refers to the installation of efficiency measures or adoption of efficiency practices by customers who did not directly participate in an efficiency program but were nonetheless influenced by the program to make the efficiency improvement.⁸⁸

⁸⁷ There are three forms of free-ridership: (1) total free-riders—or efficiency program participants who would have installed the same efficiency measures at the same time even if the program had not been run; (2) partial free-riders—or participants who would have made some, but not all, of the efficiency investments they made in the absence of the program; and (3) deferred free-riders—participants who would have made the same efficiency investments in the absence of the program, but at a later date (NREL 2017b).

⁸⁸ Spillover can take multiple forms, including both (1) participant spillover—or savings that were influenced by a customer's participation in efficiency program but were beyond those tracked by the program; and (2) non-participant spillover—or savings that were produced by customers who were influenced by a program even though they did not directly participate in it. Participant spillover can be further subdivided into savings that occur at the same site as savings from program participation (known as "inside spillover") and savings that occur at other sites (typically) owned or operated by the same

- *Gross program impacts* are impacts before or without any adjustments for free-ridership and spillover.
- *Net program impacts* include adjustments for free-ridership and spillover.

H.3.2. Economic Treatment of Free-Rider Impacts

This section describes which free-rider impacts should be included in cost-effectiveness analysis in jurisdictions that focus on net savings, given the categories of impacts that such jurisdictions include in their cost-effectiveness tests.

Utility System Impacts

Benefits: No utility system benefits associated with any savings achieved by free-riders should be included in cost-effectiveness analyses of an efficiency program because the program did not cause those benefits.

Costs: Any financial incentives paid to free-riders should be treated as a utility system cost, because they are part of the overall cost to the utility of operating an efficiency program. For example, if a customer that receives a \$100 rebate from a utility efficiency program for an efficiency measure that they would have installed absent the program, the utility system has incurred a \$100 cost.

Participant Impacts

Benefits: No participant benefits associated with any savings achieved by free-riders should be included in cost-effectiveness analyses of efficiency programs because the participants would have achieved the same benefits absent the program.

Costs: Financial incentives paid to free-rider participants should be treated as a negative cost to participants because such participants would not have received any such financial support absent the program. This reduction in cost to participants cancels out the cost of free-riders to the utility system. Thus, under cost-effectiveness tests that include both utility system and participant impacts, the net cost of free-riders is zero.

Consider the example above in which a customer that receives a \$100 rebate from a utility efficiency program for an efficiency measure that they would have installed absent the program. As discussed above, the \$100 is a utility system cost. Thus, if the jurisdiction's cost-effectiveness test included utility system impacts (as all tests must) but did not include participant impacts, there would be a net cost from the free-rider of \$100. However, that changes if the jurisdiction's cost-effectiveness test also includes participant impacts because \$100 cost to the utility system is offset by a \$100 benefit to the free-rider participant. Put another way, under a test that includes both utility system and participant impacts, the \$100 rebate is what is often called a transfer payment. It has distributional impacts — by moving money between customers — but no *net* cost to customers as a whole (which is the perspective that matters under cost-effectiveness tests that include participant impacts as well as utility system impacts).

customer (known as "outside spillover"). Participant spillover can also be subdivided into savings that are from measures or actions that are same as those that were recorded by the program (known as "like spillover") or from different kinds of efficiency measures (known as "unlike spillover") (NREL 2017b).

Other Types of Impacts

Benefits: No other types of benefits associated with any savings achieved by free-riders (other fuel savings, water savings, environmental emission reductions, public health cost savings, poverty reduction, job creation, energy security, etc.) should be included in cost-effectiveness analyses of efficiency programs because they would have been realized absent the program as well.

Costs: Any other types of costs associated with efficiency investments by free-riders should not be included in cost-effectiveness analyses of efficiency programs because they would also have been incurred absent the program.

Summary of Economic Treatment of Free-Riders

Table H-6 summarizes the proper economic treatment of free-rider benefits and costs for jurisdictions that focus on net (rather than gross) impacts.

Category	Free-Riders					
	Costs	Benefits				
Utility System Impacts	Increase	n/a				
Participant Impacts	Decrease	n/a				
Other Impacts	n/a	n/a				
Total/Net Impact	Increase only if test <i>excludes</i> participant impacts; otherwise no net effect	No effect under any test				

Table H-6. Summary of Economic Treatment of Free-Riders

H.3.3. Economic Treatment of Spillover Effects

This section describes what spillover impacts should be included in cost-effectiveness analysis in jurisdictions that focus on net savings, given the categories of impacts that such jurisdictions include in their cost-effectiveness tests.

Utility System Impacts

Benefits: All utility system benefits associated with spillover effects should be included in cost-effective analyses of an efficiency program because they were caused by the program.

Costs: There are no utility system costs directly associated with spillover effects because, by definition, investments made to produce spillover effects are not subsidized by efficiency programs (i.e., if a customer receives a rebate for installing a measure it is a program participant; spillover effects are produced when customers install measures without taking rebates or other program services).

Participant Impacts

Benefits: In jurisdictions that include participant impacts in their cost-effectiveness test, all spillover participant benefits associated with spillover effects should be included in cost-effectiveness analyses because such effects were caused by the efficiency programs being analyzed.

Costs: All spillover participant costs associated with spillover effects should be included in cost-effectiveness analyses because such effects were caused by the efficiency programs in question.

Other Types of Impacts

Benefits: In jurisdictions that include other types of impacts in their cost-effectiveness test (other fuel impacts, water impacts, environmental impacts, public health impacts, low-income impacts, job impacts, energy impacts, etc.), all other benefits associated with spillover effects should be included in cost-effectiveness analyses because such effects were caused by the efficiency programs under analysis.

Costs: All other types of costs associated with spillover effects should be included in cost-effectiveness analyses because such effects were caused by the efficiency programs under analysis.

Summary of Economic Treatment of Spillover Effects

Table H-7 summarizes economic treatment of spillover benefits and costs.

Category	Spillover	
	Costs	Benefits
Utility System Impacts	n/a	Increase
Participant Impacts	Increase	Increase (if applicable)
Other Impacts	Increase (if applicable)	Increase (if applicable)
Total/Net Impact	No increase if test includes only utility system impacts; otherwise, an increase	Increase under every test

Table H-7. Summary of Economic Treatment of Spillover Effects

REFERENCES

- American Council for An Energy-Efficient Economy (ACEEE). 2020. *Integrating Energy Efficiency, Solar and Battery Storage in Utility Programs*. Prepared by Srivastava, Bastian, Amann, Gold, and Grossberg. <u>https://www.aceee.org/sites/default/files/pdfs/b2001.pdf</u>.
- -----. 2019a. Guidance on Measuring the Economic Development Benefits of Energy Efficiency: State Policy Toolkit.
- -----. 2019b. Integrated Energy Efficiency and Demand Response Programs. Prepared by D. York, G. Relf, and C. Waters. <u>https://www.aceee.org/research-report/u1906.</u>
- Avoided Energy Supply Costs (AESC) Study Group. 2018. Avoided Energy Supply Components in New England: 2018 Report. Prepared by Synapse Energy Economics, Resource Insight, Les Deman Consulting, North Side Energy, and Sustainable Energy Advantage. June 1.
- Bonbright, James. 1961. Principles of Public Utility Rates. New York: Columbia University Press.
- The Brattle Group (Brattle). 2019. The National Potential for Load Flexibility. Value and Market Potential Through 2030. Prepared by R. Hledik, A. Faruqui, T. Lee, and J. Higham. <u>https://brattlefiles.blob.core.windows.net/files/16639_national_potential_for_load_flexibility_-____final.pdf</u>
- California Public Utilities Commission (CPUC). January 11, 2018. Proposed Decision of Commissioner Peterman: Decision on Multiple-Use Application Issues. Filed under rulemaking 15-03-011. http://docs.cpuc.ca.gov/PublishedDocs/Published/G000/M202/K234/202234451.pdf.
- -----. 2017. Distributed Energy Resources Cost Effectiveness Evaluation: Societal Cost Test, Greenhouse Gas adder, and Greenhouse Gas Co-Benefits, an Energy Division Staff Proposal. January.
- -----. 2016. Avoided Costs 2016 Interim Update. B. Horii et al., Energy and Environmental Economics, Inc. www.cpuc.ca.gov/WorkArea/DownloadAsset.aspx?id=12504.
- —. 2001. California Standard Practice Manual: Economic Analysis of Demand-Side Programs and Projects. <u>www.cpuc.ca.gov/workarea/downloadasset.aspx?id=7741</u>.
- Ceres. 2012. Practicing Risk-Aware Regulation: What Every State Regulator Should Know. Prepared by R. Binz, R. Sedano, D. Furey, and D. Mullen. <u>www.raponline.org/wp-content/uploads/2016/05/ceres-binzsedano-riskawareregulation-2012-apr-19.pdf</u>.
- Consumers Union. 2016. *Show Me the Numbers: A Framework for Balanced Distributed Solar Policies.* T. Woolf et al., Synapse Energy Economics. November 10.
- Dasgupta, Partha, Karl-Goran Maler, and Scott Barrett. 2000. Intergenerational Equity, Social Discount Rates, and Global Warming, Revised.
- E4TheFuture, Peak Load Management Alliance, and Smart Electric Power Alliance. 2018. *Non-Wires Alternatives: Case Studies from Leading U.S. Projects*. <u>https://sepapower.org/resource/non-wires-alternatives-case-studies-from-leading-u-s-projects/</u>.

- E4TheFuture. 2016. "Occupant Health Benefits of Residential Energy Efficiency." <u>https://e4thefuture.org/wp-content/uploads/2016/11/Occupant-Health-Benefits-Residential-EE.pdf</u>.
- Electric Power Research Institute (EPRI). 2015. *The Integrated Grid: A Benefit-Cost Framework. Final Report*. Prepared by K. Forsten.
- Energy Futures Group. 2015. Energy Efficiency as a T&D Resource: Lessons Learned from Recent U.S. Efforts to Use Geographically Targeted Efficiency Programs to Defer T&D Investments. Published by Northeast Energy Efficiency Partnerships. Neme, Chris and Jim Grevatt. (<u>https://neep.org/energy-</u> <u>efficiency-transmission-and-distribution-resource-using-geotargeting</u>).
- Energy Innovation: Policy and Technology LLC. 2016. *Grid Flexibility: Methods for modernizing the power grid*. Prepared by S. Aggarwal and R. Orvis. <u>https://energyinnovation.org/wp-</u> <u>content/uploads/2016/05/Grid-Flexibility-report.pdf.</u>
- Energy Resource Solutions (ERS). May 2014. *Emerging Technology Program Primary Research Ductless Heat Pumps*. Prepared for the Northeast Energy Efficiency Partnerships.
- Federal Energy Regulatory Commission (FERC). February 15, 2018. Electric Storage Participation in Markets Operated by Regional Transmission Organizations and Independent System Operators. Docket Nos. RM16-23-000 and AD16-20-000. <u>https://www.ferc.gov/whats-new/commmeet/2018/021518/E-1.pdf</u>.
- Hawaii Electric Companies (HECO). 2018. *Planning Hawaii's Grid for Future Generations: Integrated Grid Planning*. March.
- Hope, Paul. 2019. "Gas or Electric Range: Which Is Better?" *Consumer Reports.* (<u>https://www.consumerreports.org/ranges/gas-or-electric-range-which-is-better/</u>).
- ICF International (ICF). 2018. *Review of Recent Cost-Benefit Studies for Net Metering and Distributed Solar*. Prepared for U.S. DOE. May.
- -----. 2016. Integrated Distribution Planning. Prepared for the Minnesota Public Service Commission. August.
- Interstate Renewable Energy Council (IREC). 2017. *Optimizing the Grid, A Regulator's Guide to Hosting Capacity Analysis for Distributed Energy Resources.*
- Institute for Policy Integrity. 2018. *Toward Resilience. Defining, Measuring, and Monetizing Resilience in the Electricity System.* Prepared by B. Unel, Ph.D. and A. Zevin.
- Lawrence Berkeley National Laboratory (LBNL). 2020a. *Applying Non-Energy Impacts From Other Jurisdictions in Cost-Benefit Analyses of Energy Efficiency Programs: Resources for State for Utility Customer-Funded Programs.* Prepared by M. Sutter, J. Mitchell-Jackson, S. Schiller, L. Schwartz, I. Hoffman. Electricity Markets and Policy Group. May.
- -----. 2020b. A conceptual Framework to Describe Energy Efficiency and Demand Response Interactions. Prepared by A. Satchwell, P. Cappers, J. Deason, S. Forrester, N. Frick, B. Gerke, M. Piette.
- —. 2019. Current Developments in Retail Rate Design: Implications for Solar and Other Distributed Energy Resources. Prepared by A. Satchwell, P. Cappers, G. Barbose. Electricity Markets and Policy Group. July.

- —. 2018. The Cost of Saving Electricity Through Energy Efficiency Programs Funded by Utility Customers: 2009-2015. Prepared by I. Hoffman, C. Goldman, S. Murphy, N. Mims, G. Levantis, L. Schwartz. Electricity Markets and Policy Group. June.
- -----. 2016. California Demand Response Potential Study Final Report: Charting California's Demand Response Future: Final Report on Phase 2 Results. November.
- —. 2014. Financial Impacts of Net-Metered PV of Utilities and Ratepayers: A Scoping Study of Two Prototypical U.S. Utilities. Prepared by A. Satchwell, A Mills, G. Barbose, R. Wiser, P. Cappers, N. Darghouth. Environmental Energy Technologies Division. September.
- Livchak et al. 2019. *Residential Cooktop Performance and Energy Comparison Study*. Frontier Energy. (https://cao-94612.s3.amazonaws.com/documents/Induction-Range-Final-Report-July-2019.pdf).
- Logtenberg, R., Pawley, J., and Saxifrage, B. 2018. *Comparing Fuel and Maintenance Costs of Electric and Gas Powered Vehicles in Canada*. 2 Degrees Institute.

(<u>https://www.2degreesinstitute.org/reports/comparing_fuel_and_maintenance_costs_of_electric_and_gas_powered_vehicles_in_canada.pdf</u>).

McKinsey & Company (McKinsey). 2013. Pathways to a Low Carbon Economy: Version 2 of the Global Greenhouse Gas Abatement Cost Curve.

- Multpl.com. nd. 10 Year Real Interest Rate. https://www.multpl.com/10-year-real-interest-rate.
- National Action Plan for Energy Efficiency (NAPEE). 2007. *Guide to Resource Planning with Energy Efficiency*. US EPA. www.epa.gov/sites/production/files/2015-08/documents/resource_planning.pdf.
- National Association of Regulatory Commissioners (NARUC) and National Association of State Energy Offices (NASEO). 2020. *Task Force on Comprehensive Electricity System Planning*. <u>https://www.naruc.org/taskforce/.</u>
- NARUC. 2019. "The Value of Resilience for Distributed Energy Resources: An Overview of Current Analytical Practices." Prepared by Converge Strategies. <u>https://pubs.naruc.org/pub/531AD059-</u> <u>9CC0-BAF6-127B-99BCB5F02198</u>.
- National Energy Screening Project (NESP). 2019. *Database of State Efficiency Screening Practices*. <u>https://www.nationalenergyscreeningproject.org/state-database-dsesp/</u>.
- 2017. National Standard Practice Manual for Evaluating the Cost-Effectiveness of Energy Efficiency Resources (NSPM for EE), Edition 1, May.
- National Renewable Energy Laboratory (NREL). 2019. *Resilience Roadmap: A Collaborative Approach to Multi-Jurisdictional Resilience Planning*. <u>https://www.nrel.gov/docs/fy19osti/73509.pdf</u>.
- -----. 2017a. *Grid-Connected Distributed Generation: Compensation Mechanism Basics*. Prepared by O. Zinamen, A. Aznar, C. Linville, N. Dargouth, T. Dubbeling, E. Bianco. October.
- —. 2017b. "Chapter 21: Estimating Net Savings Common Practices: Methods for Determining Energy-Efficiency Savings for Specific Measures." The Uniform Methods Project: Methods for Determining Energy Efficiency Savings for Specific Measures. Update to 2014 publication. Prepared by D.M. Violette, P. Rathbun. NREL/SR-7A40-68578. <u>https://www.nrel.gov/docs/fy17osti/68578.pdf</u>.
- Navigant Consulting. 2019. *Distributed Resource and Flexible Load Study: Integrated Resource Planning System-Level Report*. Prepared for Portland General Electric (PGE). April. Accessed via PGE Integrated Resource Plan.

- —. 2017. Cutting Peak Demand: Two Competing Paths and Their Effectiveness." Guidehouse. Prepared by S. Tobias. https://guidehouse.com/-/media/www/site/insights/energy/2017/iepec_ngriddriepec2017.pdf.
- New York Public Service Commission (NY PSC). December 2019. New York State Standardized Interconnection Requirements and Application Process for New Distributed Generators and Energy Storage Systems for 5 MW or Less Connected in Parallel with Utility Distribution Systems. <u>http://www3.dps.ny.gov/W/PSCWeb.nsf/96f0fec0b45a3c6485257688006a701a/dcf68efca391ad60</u> <u>85257687006f396b/\$FILE/December%202019%20SIR%20-%20FINAL%20-%20Clean.pdf</u>.
- December 13, 2018. Order Establishing Energy Storage Goal and Deployment Policy. Case 18-E-0130: In the Matter of Energy Storage Deployment Program. http://documents.dps.ny.gov/public/Common/ViewDoc.aspx?DocRefId=%7BFDE2C318-277F-4701-B7D6- http://common/ViewDoc.aspx?DocRefId=%7BFDE2C318-277F-4701-318-277F-4701-B7D6-C70FCE0C6266}.
- North American Electric Reliability Corporation (NERC). 2011. Demand Response Availability Data System (DADS): Phase I and II Final Report.
- Northeast Energy Efficiency Partnerships (NEEP). 2020. *Grid-Interactive Efficient Buildings (GEBs) Tri-Region Status Report*. January. https://neep.org/sites/default/files/resources/NEEP%20GEBs%20Report Final.pdf
- -----. 2017. Non-Energy Impacts Approaches and Values: An Examination of the Northeast, Mid-Atlantic, and Beyond.
- O'Leary, K. 2016. "Three Ways Opt-Out Can Improve Your Demand Response Program." *Energy Hub,* May 24. https://www.energyhub.com/blog/three-ways-opt-out-can-improve-your-demandresponse-program.
- Olinsky-Paul, Todd. 2019. Energy Storage: The New Efficiency How States Can Use Energy Efficiency Funds to Support Battery Storage and Flatten Costly Demand Peaks. Publication of Clean Energy Group. <u>https://www.cleanegroup.org/wp-content/uploads/energy-storage-the-new-efficiency.pdf</u>.
- Peak Load Management Alliance (PLMA). 2019. *The Future of Distributed Energy Resources: A Compendium of Industry Viewpoints.* Produced by PLMA Thought Leadership and Distributed Energy Resource Integration Groups. <u>https://www.peakload.org/assets/resources/PLMA-Future-of-DER-Compendium.pdf.</u>
- Pindyck, R, and D Rubinfeld. 2001. Microeconomics. 5th ed. Upper Saddle River, NJ: Prentice Hall.
- Portland General Electric Company (PGE). 2020. *PGE Smart Grid Test Bed*. Website. Accessed June 2020: <u>https://www.portlandgeneral.com/our-company/energy-strategy/smart-grid/smart-grid-test-bed.</u>
- —. 2019. Integrated Resource Plan. July. <u>https://www.portlandgeneral.com/our-company/energy-strategy/resource-planning/integrated-resource-planning.</u>
- —. 2018. Advice No. 18-14: NEW Schedule 13 Opt-Out Residential Demand Response Testbed Pilot and Application. Submitted to P.U.C. Oregon No. E-18. October 25. <u>https://edocs.puc.state.or.us/efdocs/UAA/uaa173123.pdf.</u>
- Regulatory Assistance Project (RAP). 2019a. "Capturing More Value from Combinations of PV and Other Distributed Energy Resources." Shenot, Linville, Dupuy, Britkoski.

https://www.raponline.org/knowledge-center/capturing-more-value-from-combinations-of-pvand-other-distributed-energy-resources/.

- -----. 2019b. "Flexibility for the 21st Century Power System." Linville, Lazar, Shipley, Littel, and Farnsworth. October.
- —. 2013a. Recognizing the Full Value of Energy Efficiency. J. Lazar and K. Colburn. <u>http://www.raponline.org/wp-content/uploads/2016/05/rap-lazarcolburn-layercakepaper-2013-sept-9.pdf</u>.
- ----. 2013b. Widening Energy Regulators' Circles of Interest: State Examples of Energy Regulations and Processes That Consider Environmental Impacts. B. Hausauer & J. Shenot. January.
- —. 2013c. Best Practices in Electric Utility Integrated Resource Planning. R. Wilson and B. Biewald, Synapse Energy Economics. http://www.synapse-energy.com/project/best-practices-electric-utilityintegrated-resource-planning.
- —. 2012. Energy Efficiency Cost-Effectiveness Screening: How to Properly Account for 'Other Program Impacts' and Environmental Compliance Costs. T. Woolf et al., Synapse Energy Economics. http://www.synapse-energy.com/sites/default/files/SynapseReport.2012-11.RAP_.EE-Cost-Effectiveness-Screening.12-014.pdf.
- Rocky Mountain Institute (RMI). 2019. "Value Potential for Grid-Interactive Efficient Buildings in the GSA Portfolio: A Cost Benefit Analysis." Prepared by M. Jungclaus, C. Carmichael, and P. Keuhn. http://www.rmi.org/GEBs_report.
- —. 2015. "The Economics of Demand Flexibility. How 'Flexiwatts' Create Quantifiable Value for Customers and the Grid." Prepared by P. Bronski, M. Dyson, M. Lehrman, J. Morris, and H. Touati <u>https://rmi.org/insight/the-economics-of-demand-flexibility-how-flexiwatts-create-quantifiablevalue-for-customers-and-the-grid/</u>.
- Sandia National Laboratories. 2015. Energy Infrastructure Resilience: Framework and Specific Metrics. Presentation prepared for the U.S. Department of Energy's National Nuclear Security Administration. (<u>https://www.energy.gov/sites/prod/files/2015/01/f19/SNLResilienceApril29.pdf</u>).
- Skumatz Economic Research Associates (SERA). 2014. "Non-Energy Benefits/Non-Energy Impacts and Their Role & Values in Cost-Effectiveness Tests: State of Maryland." Prepared by L. Skumatz for Natural Resources Defense Council. http://energyefficiencyforall.org/sites/default/files/2014_%20NERs%20report%20for%20Marylan

http://energyefficiencyforall.org/sites/default/files/2014_%20NEBs%20report%20for%20Maryland. pdf.

- Smart Electric Power Alliance (SEPA). 2019a. "Microgrids: The Role of Microgrids in the Regulatory Compact." Prepared by J. Leader and R. Tucker. https://sepapower.org/resource/microgrids-the-role-of-microgrids-in-the-regulatory-compact/.
- —. 2019b. "Residential Electric Vehicle Rates that Work: Attributes that Increase Adoption." Published in partnership with The Brattle Group, E4theFuture, and Enel X. (sepapower.org/resource/residential-electric-vehicle-time-varying-rates-that-work-attributes-thatincrease-enrollment/).
- San Diego Gas and Electric, Southern California Edison, and Pacific Gas and Electric (SDGE, SCE, and PGE). 2018. "Locational Net Benefit Analysis Working Group Long Term Refinements Final Report." Presented to the California Public Service Commission, Rulemaking 14-08-013. January.

- State and Local Energy Efficiency Action Network (SEE Action). 2020a forthcoming. Advancing Performance Assessments of Demand Flexibility from Grid-interactive Efficient Buildings: Issues and Considerations. Prepared by: Steven R. Schiller, Lisa Schwartz, and Sean Murphy, Lawrence Berkeley National Laboratory. Available upon publication at: <u>https://emp.lbl.gov/publications.</u>
- -----. 2020b. *Grid-Interactive Efficient Buildings: An Introduction for State and Local Governments*. Prepared by: Lisa Schwartz and Greg Leventis, Lawrence Berkeley National Laboratory.
- —. 2020c. Determining Utility System Value of Demand Flexibility from Grid-Interactive Efficient Buildings. Prepared by: Tom Eckman, Lisa Schwartz, and Greg Leventis, Lawrence Berkeley National Laboratory. April
- —. 2011. "Analyzing and Managing Bill Impacts of Energy Efficiency Programs: Principles and Recommendations." DOE/EE-0926. <u>https://www4.eere.energy.gov/seeaction/sites/default/files/pdfs/ratepayer_efficiency_billimpacts.</u> <u>pdf</u>.
- State of Massachusetts. 2017. "State of Charge: Massachusetts Energy Storage Initiative." Published in partnership with the Massachusetts Department of Energy Resources, Massachusetts Clean Energy Center, Customized Energy Solutions, Ltd., Alevo Analytics, Sustainable Energy Advantage, LLC, Daymark Energy Advisors, and Strategen. <u>https://www.mass.gov/files/2017-07/state-of-charge-report.pdf</u>.
- Synapse Energy Economics (Synapse). 2019. New Hampshire Cost-Effectiveness Review: Application of the National Standard Practice Manual to New Hampshire. Prepared for the New Hampshire Evaluation, Measurement and Verification (EM&V) Working Group. Erin Malone, Tim Woolf, Steve Letendre. October 14, 2019.
- —. 2018. Decarbonization of Heating Energy Use in California Buildings: Technology, Markets, Impacts, and Policy Solutions. Prepared by A.S. Hopkins, K. Takahashi, D. Glick, M. Whited for the Natural Resources Defense Council. (www.synapse-energy.com/sites/default/files/Decarbonization-Heating-CA-Buildings-17-092-1.pdf).
- TetraTech. 2012. *Final Report Commercial and Industrial Non-Energy Impacts Study.* Prepared for the Massachusetts Program Administrators. June 29.
- —. 2011. Final Report Massachusetts Special and Cross-Sector Studies Area, Residential and Low-Income Non-Energy Impacts Evaluation. Prepared for the Massachusetts Program Administrators. August.
- U.S. Department of Energy (DOE). Forthcoming-a. (GMLC). *Benefit-Cost Analysis for Utility-Facing Grid Modernization Investments: Trends, Challenges, and Considerations.* Grid Modernization Laboratory Consortium, prepared by Synapse Energy Economics.
- -----. Forthcoming-b. *Modern Distribution Grid: Decision Guide Volume IV*.
- -----. 2020. *Electric Vehicle Benefits*. Office of Energy Efficiency and Renewable Energy. (www.energy.gov/eere/electricvehicles/electric-vehicle-benefits).
- —. 2019a. Grid-Interactive Efficient Buildings Technical Report Series: Overview of Research Challenges and Gaps. Prepared by M. Neukomm, V. Nubbe, and R. Fares. December. <u>https://www1.eere.energy.gov/buildings/pdfs/75470.pdf.</u>

- —. 2019b. Grid-Interactive Efficient Buildings Technical Report Series: Whole Building Controls, Sensors, Modeling and Analytics. Prepared by M. Neukomm, V. Nubbe, and R. Fares. December. <u>https://www1.eere.energy.gov/buildings/pdfs/75478.pdf</u>.
- —. 2019c. How Distributed Energy Resources Can Improve Resilience in Public Buildings: Three Case Studies and a Step-by-Step Guide. Prepared by J. Agan, B. Holleman, and S. Gheewala. September. <u>https://www.energy.gov/sites/prod/files/2019/10/f67/distributed-energy-resilience-publicbuildingsv2.pdf.</u>
- —. 2019d. Energy Efficiency and Distributed Generation for Resilience: Withstanding Grid Outages for Less. DOE/GO-102019-5186. July. https://www.energy.gov/sites/prod/files/2019/06/f64/EEDG-Resilience.PDF
- —. 2017a. Modern Distribution Grid: Decision Guide Volume III.
- —. 2017b. Summary of Electric Distribution System Analyses with a Focus on DERs. Grid Modernization Laboratory Consortium. <u>https://epe.pnnl.gov/pdfs/Summary of electric distribution system analyses April 10 FINAL.pdf</u>
- —. 2017c. Transforming the Nation's Electricity System: The Second Installment of the Quadrennial Energy Review. Chapter IV: Ensuring Electricity System Reliability, Security, and Resilience. <u>https://www.energy.gov/policy/downloads/quadrennial-energy-review-second-installment.</u>
- U.S. Energy Information Administration (EIA). 2019a. Average Operating Heat Rate for Selected Energy Sources. <u>https://www.eia.gov/electricity/annual/html/epa_08_01.html</u>.
- ----. 2019b. How much electricity is lost in electricity transmission and distribution in the United States? <u>https://www.eia.gov/tools/faqs/faq.php?id=105&t=3).</u>
- —. 2018a. Annual Energy Outlook.
- —. 2018b. "Major utilities continue to increase spending on U.S. electric distribution systems." Today in Energy. <u>https://www.eia.gov/todayinenergy/detail.php?id=36675.</u>
- U.S. Office of Management and Budget (US OMB). 2003. *Circular A-4 to the Heads of Executive Agencies and Establishments*. Regulatory Analyses.

Public Service Company of New Hampshire d/b/a Eversource Energy Docket No. IR 22-042

Date Request Received: November 01, 2022 Data Request No. RR 1-002 Date of Response: December 16, 2022 Page 1 of 3

Request from: New Hampshire Public Utilities Commission

Request:

<u>Performance Incentive</u>: The Commission wants to understand the history of planned versus actualized energy efficiency savings and how the PI structure has incentivized exceptional performance. The Commission requests that the Joint Utilities submit one response to each of the following questions:

- 1. Please fill out the table in the attached excel (Tab: Request II.1 PI) In order to create a comprehensive Energy Efficiency Program-wide dataset on each utility's planned versus actual expenditures, savings, and PI. Please provide planned and actual data for each program, for each completed program year since 2016. Please also calculate the PI if it had been earned beginning at 100% of the planned savings, not the lower amount, for each program for each year. Please comment on all instances where a PI Cap was reached and what the payment would have been if there had been no cap.
- 2. How do other New England states where the joint utilities operate or have visibility construct performance incentives or other mechanisms to reward Utilities for exceptional work implementing their respective state energy efficiency policies and programs?
- 3. Please summarize the results of the PI working group and provide any comments reflecting changes in circumstances since the report was issued.
- 4. Please summarize how the current minimum and maximum PI thresholds were determined.

Response:

1.

11/30/2022 Response:

The Utilities were granted an extension for the response to this question.

12/16/2022 Response:

Please see Excel Attachment RR 1-002D for the summation of the utility information provided to the Commission in reporting requirement number 6 on August 1, 2022, in this docket. The

Public Service Company of New Hampshire d/b/a Eversource Energy Docket No. IR 22-042

Date Request Received: November 01, 2022 Data Request No. RR 1-002

Date of Response: December 16, 2022 Page 2 of 3

Utilities have presented results to illustrate the impact of retroactively applying a theoretical 100 percent threshold to actual results from 2016 through 2021. However, caution should be taken so as to not misinterpret the results. Similar to the context provided to reporting requirement 2.2a, please note that:

• Minimum thresholds impact the establishment of savings and benefits targets; high minimum thresholds tend to provide downward pressure on goals;

• Performance Incentives are not guaranteed and must be earned by the Utilities through successful program implementation and the achievement of established goals and thresholds;

• Had different minimum thresholds been in force during the program years in question, the behavior of utility administrators and the outcomes resulting would likely have differed from what is portrayed here *ex post facto*;

• Performance incentives were not designed or earned on a program-by-program basis but on the performance of each sector (2016-2019) and then later on the basis of the portfolio as a whole (2020-2021).

- Please see Attachments RR 1-002A and RR 1-002B for the current performance incentive mechanisms for Massachusetts and Connecticut (see page 27). For other states, please refer to the several linked files within the "Useful Materials" of the Performance Incentive Working Group's webpage on the Commission's website, located at <u>https://www.puc.nh.gov/EESE%20Board/EERS_Working_Groups.html#pi</u>.
- 3. Please refer to Attachment RR 1-002C for a copy of the performance incentive working group report, which can also be found on the Commission's website at: https://www.puc.nh.gov/EESE%20Board/EERS_Working_Groups.html

From the Executive Summary of the report:

The PI Working Group met in order to review the current, and alternative, PI calculation methodologies and to recommend an appropriate PI framework to be implemented for the 2020 period. The Working Group considered including potential metrics to encourage electric system peak load reductions and to increase participation by low income groups and households in energy efficiency programs. The discussions of the PI Working Group occurred over a sixteen-month period between January 2018 and July 2019, and the salient documents from these discussions are posted to the Commission website. A significant portion of the Working Group's time was spent studying and

Public Service Company of New Hampshire d/b/a Eversource Energy Docket No. IR 22-042

Date Request Received: November 01, 2022 Data Request No. RR 1-002 Date of Response: December 16, 2022 Page 3 of 3

revising minimum PI thresholds, calculation methodologies, and developing a more comprehensive and transparent framework for calculating PI that constitutes a good replacement for the existing methodology.

The PI Working Group finished its report on July 31, 2019, and the Utilities submitted it to the Commission as Attachment M to the 2020 NHSaves Energy Efficiency Plan update filing made to the Commission on September 13, 2019, in Docket No. DE 17-136, tab 143. This Plan and subsequent settlement, including application of the PI Working Group recommendations, was reviewed and approved by the Commission in Order No. 26,323 for application beginning in program year 2020.

The NHSaves programs remain consistent with those offered in 2020 when the new PI framework was first applied to program design. The PI structure continues to be an effective mechanism for encouraging the Utilities to achieve clearly defined goals and policy objectives and maximize savings produced by the state's energy efficiency programs.

4. The Performance Incentive Working Group, consisting of various stakeholders and consultants, collaborated over the course of two years and arrived at consensus recommendations, which were subsequently agreed to in settlement and approved by the Commission in Order No. 26,323 on December 31, 2019, as referenced in the response to question RR 1-002 part 3, with application beginning in program year 2020.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-002A Page 1 of 1

A. Performance Incentives (PI) Calculation

The 2022-2024 Statewide Electric PI pool includes a standard savings component, a component for benefits generated by measures that are installed in selected Environmental Justice Communities and selected Boston zip codes ("equity"), and a component for benefits generated by strategic electrification measures.

1 Statewide Electric 2022 - 2024 Total Design Level PI (In 2022\$)		\$ 118,620,000
	REMAINING PORTFOLIO ("STANDARD") SAVINGS COMPONENT	
2	Statewide Electric Standard Benefits	\$ 3,273,891,720 Excluding CLC
3	Electric PI Allocated to Savings Component	\$ 41.315.450
4	Electric Standard Savings Payout Rate	\$ 0.0126
5	Standard Savings Component Threshold Level	75%
	EQUITY SAVINGS COMPONENT	
6	Statewide Electric Equity Benefits	\$ 1,167,124,299 Excluding CLC
7	Electric PI Allocated to Equity Savings Component	\$ 17,674,475
8	Electric Equity Savings Payout Rate	\$ 0.0151 120% of Standard Rate
9	Equity Savings Component Threshold Level	85%
	ELECTRIFICATION SAVINGS COMPONENT	
10	Statewide Electric Electrification Benefits	\$ 1,587,737,267 Excluding CLC
11	Electric PI Allocated to Electrification Savings Component	\$ 24,044,074
12	Electric Electrification Savings Payout Rate	\$ 0.0151 120% of Standard Rate
13	Electrification Savings Component Threshold Level	60%
	VALUE COMPONENT	
14	Statewide Electric Total Portfolio Program Costs	\$ 2,447,711,629 Excluding CLC
15	Statewide Electric Total Net Benefits	\$ 3,581,041,656 Line 20-Line 14
16	Electric PI Allocated to Value Component	\$ 35,586,000 30% of total pool
17	Electric Value Payout Rate	\$ 0.0099 Line 16/Line 15
18	Electric Value Threshold Level	75%
19	Statewide Electric 2022 - 2024 Design Level Performance Incentives	\$ 118,620,000 Line 3 + Line 7 + Line 11 + Line 16
20	Statewide Total Benefits	\$ 6,028,753,286 Line 2 + Line 6 + line 10
21	Statewide Weighted Portfolio Threshold	73% ((Line 2 * Line 5) + (Line 6*Line 9) + (Line 10*Line 13))/Line 15
	Additional Calculations	
22	Total Residential & Income Eligible Equity Benefits in 2024	\$ 409,274,846
23	Total Residential & Income Elgible Portfolio Benefits in 2024	\$ 1,394,540,401
24	% Residential & Income Eligible Equity Benefits of total Residential & Income Eligible Benefits	29.3%
25	Target	25.0%



Energy and Technology Policy

10 Franklin Square • New Britain, CT 06051

www.ct.gov/deep

Affirmative Action/Equal Opportunity Employer

June 1, 2022

DEPARTMENT OF ENERGY AND ENVIRONMENTAL PROTECTION

FINAL DETERMINATION: APPROVAL WITH CONDITIONS OF THE 2022-2024 CONSERVATION AND LOAD MANAGEMENT PLAN

The Connecticut Department of Energy and Environmental Protection (DEEP) hereby issues the following Determination modifying and approving, with conditions, the 2022-2024 Conservation and Load Management (C&LM) Plan based on input received from the Energy Efficiency Board (EEB), requests for information issued by DEEP, public input sessions, and subsequent written comments. DEEP thanks the members of the EEB, the energy efficiency vendor and contractor community, Eversource Energy, The United Illuminating Company (UI), the Southern Connecticut Gas Company, the Connecticut Natural Gas Corporation, (collectively the Utilities), members of the public, and the rest of the stakeholder community who participated in this process, for the thoughtful consideration evident in their feedback regarding the 2022-2024 C&LM Plan.

I. <u>Procedural History</u>

Pursuant to Conn. Gen. Stat. § 16-245m, the Utilities, in consultation with the EEB, develop and submit to DEEP a combined electric and gas Conservation and Load Management Plan to implement "cost-effective energy conservation programs, demand management, and market transformation initiatives" for the state of Connecticut. The C&LM Plan is a three-year plan, including a detailed budget, that is supplemented by annual updates. DEEP is required, in an uncontested proceeding, to approve, modify or reject the three-year C&LM Plan and the detailed budget submitted with the Plan, including any annual updates submitted by the Utilities.

Connecticut's energy efficiency programs have a strong track record of success and have been recognized as some of the best in the nation. In 2021, the programs generated over \$62 million in savings for Connecticut residents and businesses, eliminating the need for 114 megawatts of electricity generation capacity, enough to power approximately 43,138 homes for a year, and avoiding over 196,807 tons of carbon dioxide emissions.¹ The energy-saving investments funded by the 2022-2024 Plan are projected to provide over \$1.7 billion in cumulative benefits.²

The C&LM Plan is funded by ratepayer funds collected through the Conservation Adjustment Mechanism (CAM), revenues from the Regional Greenhouse Gas Initiative (RGGI) pursuant to Conn. Agencies Reg. § 22a-174-31, and proceeds from C&LM savings sold into the ISO New England Forward Capacity Market (FCM). In each Plan year, the Utilities collect ratepayer funds from each electric end use customers through the CAM charge at a rate of not more than six mills per kilowatt hour of electricity and

 ¹ See Energy Efficiency Board 2021 Programs and Operations Report, March 1, 2022, available at: <u>https://energizect.com/connecticut-energy-efficiency-board/about-energy-efficiency-board/annualreports</u>
 ² See Proposed 2022-2024 Conservation and Load Management Plan, November 1, 2021, available at: <u>http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/04d115cb68d338b7852587880</u> 07091c9?OpenDocument.

for gas customers at a rate of no more than the equivalent of four and six-tenth cents per hundred cubic feet of natural gas.³

The EEB voted to approve the 2022-2024 C&LM Plan text on September 8, 2021 and approved the budget on October 13, 2021.⁴ Subsequently, the Utilities filed the proposed 2022-2024 C&LM Plan and 2022 Program Savings Document with DEEP on November 1, 2021.⁵ The EEB provided several opportunities for public input throughout the development of the 2022-2024 C&LM Plan. The first of these occurred on March 10, 2021, while the C&LM Plan was in an earlier phase of development. On August 11 and 18, 2021, the EEB hosted two additional public input sessions to collect feedback on a draft of the 2022-2024 C&LM Plan. All comments received during the August 2021 are summarized in Appendix B of the 2022-2024 C&LM Plan, along with responses from both the Utilities and EEB.⁶

In its review of the proposed 2022-2024 Plan, DEEP submitted Requests for Information to the Utilities on November 3, 2021 soliciting additional information on a variety of topics.⁷ The Utilities submitted responses to these Requests on November 12 and 16, 2021.⁸ Following the DEEP Technical Meetings held on November 16 and 18, 2021, DEEP submitted additional Requests for Information to the Utilities on November 30, 2021.⁹

Pursuant to Conn. Gen. Stat. § 16-245m(d)(1), DEEP is required to initiate an uncontested proceeding to approve, modify, or reject a C&LM Plan, which may include a public meeting. DEEP held three public meetings to inform its review of the 2022-2024 C&LM Plan: two Technical Meetings on November 16

⁸ See Utility Responses to DEEP Requests for Information, available at:

³ Historically, for electric and gas customers, when there are unused ratepayer funds at the end of a Plan year, the Utilities carry over up to five percent of the current annual budget related to the CAM into the next year. Any unspent funds from the CAM in excess of the five-percent threshold is returned to ratepayers through the following year's CAM. RGGI funds and FCM funds are not subject to the five-percent carryover threshold, and may be carried over without a cap. *See* Conn. Gen. Stat. § 16-245m(d)(1) and Final Decision, *Application of The Connecticut Light And Power Company And The United Illuminating Company For Approval Of A Conservation Adjustment Mechanism*, Docket No. 13-11-14, at 6, May 20, 2014; Final Decision, *PURA Investigation of the Annual Conservation Adjustment Mechanisms filed by: Connecticut Natural Gas Corporation, the Southern Connecticut Gas Company and Yankee Services Gas Company*, Docket No. 14-03-01, at 6, May 20, 2014.

⁴ See minutes from the September 8, 2021 and October 13, 2021 meetings of the Energy Efficiency Board, *available at*: <u>https://energizect.com/sites/default/files/2021-10/Minutes%20Sept%202021%20EEB.pdf</u> (September) and <u>https://energizect.com/sites/default/files/2021-12/Oct%202021%20EEB%20Minutes_f.pdf</u> (October)

⁵ See Proposed 2022-2024 Conservation and Load Management Plan, November 1, 2021, *available at*: <u>http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/04d115cb68d338b7852587880</u> 07091c9?OpenDocument.

⁶ See Proposed 2022-2024 Conservation and Load Management Plan (Appendix B), November 1, 2021, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/04d115cb68d338b7852587880 07091c9?OpenDocument

⁷ See DEEP Requests for Information: Questions and Comments Regarding Proposed 2022-2024 Conservation and Load Management Plan, November 3, 2021, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/abcc7396b7044cd5852587830 0446ea6/\$FILE/Requests%20for%20Information%20-%202022-2024%20CLM%20Plan.pdf

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/7091cd5a6937ac52852587900 05a4d16?OpenDocument (RFI #BETP1-9) and

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/6cadc3d093b715be852587920 0580264?OpenDocument (RFI #BETP10-28)

⁹ See Requests for Information: Questions and Comments Regarding Proposed 2022-2024 Conservation and Load Management Plan, November 30, 2021, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/d963f3c1f089626e8525879d00 67a5d3/\$FILE/Second%20Requests%20for%20Information%20-%202022-2024%20C&LM%20Plan.pdf

and 18, 2021 and one Public Input Session on November 17, 2021.¹⁰ The Technical Meetings included presentations from the Utilities and expert stakeholders regarding various elements of the 2022-2024 C&LM Plan, as well as opportunities for public comment.¹¹ The Public Input Session was held in the evening and included a high-level overview of the C&LM Plan from the Utilities and additional opportunities for public comment.¹² So as not to duplicate efforts, DEEP stated in its October 26 Notice that all comments received through the EEB public processes would be considered in DEEP's review of the C&LM Plan.¹³ Following the November 2021 meetings, DEEP issued a Notice of Request for Written Comments regarding the proposed 2022-2024 C&LM Plan.¹⁴ These comments are summarized in Attachment C. In addition to regular public meetings held by DEEP and the EEB throughout the year, these public participation opportunities provided an avenue for stakeholders to engage in the development of the 2022-2024 C&LM Plan.

DEEP issued an Interim Decision with respect to the 2022-2024 C&LM Plan budget on December 23, 2021. In this Interim Decision, DEEP approved the total budget for the 2022-2024 C&LM Plan and provided that DEEP would issue a Determination on the Plan text, subject to compliance conditions, to the Utilities at a later date.¹⁵

On April 12, 2022, DEEP issued a Draft Determination approving the Plan text with Conditions of Approval.¹⁶ Following the issuance of the Draft Determination, DEEP convened a public meeting to

¹¹ See DEEP Notice of Technical Meetings and Public Input Session, October 26, 2021, *available at*: <u>http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/e4415567bfc5b3e28525878200</u> <u>4d0852/\$FILE/Notice%20of%20Technical%20Meeting%20-%202022-2024%20CLM%20Plan.pdf</u>. Recordings of the Technical meetings are available here: <u>https://ctdeep.zoom.us/rec/play/WCdG84uKS96n2x-</u>

¹² See DEEP Notice of Technical Meetings and Public Input Session, October 26, 2021, *available at*: http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/e4415567bfc5b3e28525878200 4d0852/\$FILE/Notice%20of%20Technical%20Meeting%20-%202022-2024%20CLM%20Plan.pdf. A recording of the Public Input Session is available here: https://ctdeep.zoom.us/rec/play/kCpwKDLuHnAuZp2-

<u>1uApHl64bJczhRs6KAcuqkuQMdd6lTkEBGx_zwb74V7SmxsFmicNzh0VhkOV-_0f.M5DYF7R-</u> <u>Hl7Eajy4?continueMode=true</u>. A slide deck from the Public Input Session is available here: <u>https://portal.ct.gov/-</u> /media/DEEP/energy/ConserLoadMgmt/20211117-Public-Input-Session.pdf

¹⁰ See DEEP Notice of Technical Meetings and Public Input Session, October 26, 2021, *available at*: <u>http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/e4415567bfc5b3e28525878200</u> <u>4d0852/\$FILE/Notice%20of%20Technical%20Meeting%20-%202022-2024%20CLM%20Plan.pdf</u>.

gjZuROpKeb3euh4VdU4s1Pfe3MXz9x9oQNaLgyTgToHI_RDzAMfLdesaiLbZWS8ww.lUvzeQ55sAxsCmcG?co ntinueMode=true (November 16) and here:

https://ctdeep.zoom.us/rec/play/ltR67izDoCJUBryEr3nMcHRMdxjziWuD38V0wh_DkV36dc74Ymf14a4DlNkH7a k8ZbAumtnrLLKciVdu.wWBY0dm24hAbl-6J?continueMode=true&_x_zm_rtaid=fSaU-

jGkSa6eJMonBwzGSg.1637337535995.253bf4dffb9e6955e3ed0b3b70047c0c&_x_zm_rhtaid=623 (November 18). Slide decks from the Technical Meetings are available here: https://portal.ct.gov/-

[/]media/DEEP/energy/ConserLoadMgmt/20211116-CLM-Technical-Meeting.pdf (November 16) and here: https://portal.ct.gov/-/media/DEEP/energy/ConserLoadMgmt/20211118-CLM-Technical-Meeting.pdf (November 18).

¹³ See Id.

¹⁴ See DEEP Notice of Request for Written Comments, November 19, 2021, available at:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/f13e422d410439cc8525879200 57b0f3/\$FILE/Notice%20of%20Request%20for%20Written%20Comments%20-%202022-

^{2024%20}C&LM%20Plan.pdf. All written comments submitted in response to this request can be found at: http://www.dpuc.state.ct.us/DEEPEnergy.nsf/\$EnergyView?OpenForm&Start=1&Count=30&Expand=7.1&Seq=3

¹⁵ See DEEP Approval and Determination of the 2022-2024 Conservation and Load Management Plan Budget, December 23, 2021, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/871ac6d7479e1de8852587b80 0409cf8?OpenDocument

¹⁶ See DEEP Draft Determination: Approval With Conditions of the 2022-2024 Conservation and Load Management Plan, April 12, 2022, *available at*:
discuss the Draft, its associated attachments, and Conditions of Approval.¹⁷ DEEP also provided an opportunity for stakeholders to submit written comments through April 27, 2022.¹⁸ Comments received in response to the Draft Determination are summarized in Attachment E.

II. <u>2022-2024 C&LM Plan Priorities</u>

The proposed 2022-2024 C&LM Plan is structured around three priorities – equity, decarbonization, and energy affordability – and highlights the ways in which C&LM offerings advance these priorities.¹⁹ DEEP's determination supports these priorities, providing several findings and Conditions of Approval intended to further advance equity, decarbonization, and energy affordability through the C&LM Plan.

a. Equity

The proposed 2022-2024 C&LM Plan asserts a commitment to ensuring that, "the Portfolios are equitable in their distribution of programs and benefits across the state, including communities and neighborhoods, market segments, and customer types."²⁰ The Utilities propose to fulfill this commitment in a number of ways during the 2022-2024 Plan term, including through the development of new equity metrics, increased marketing to non-English speakers, supporting minority- and women-owned vendors, reaching priority communities with education and engagement initiatives, and ensuring that energy efficiency workforce opportunities are available in underserved communities.

Through its Equitable Energy Efficiency (E3) Proceeding, DEEP is working with the Utilities, EEB, and stakeholders to assess equity in C&LM programs and develop strategies to ensure equitable program delivery. There has already been progress on the recommendations from the first phase of the E3 Proceeding and DEEP looks forward exploring new issues in future phases.²¹ In particular, the EEB has acted on the E3 recommendation to hire a Diversity, Equity, and Inclusion Consultant who will play an important role in moving this work forward.²²

This Determination builds upon these efforts through Conditions of Approval directing the Utilities to, among other things, conduct targeted outreach to underserved communities in subsequent rounds of the Community Partnership Initiative, set goals for delivering workforce development offerings in distressed municipalities, and implement programmatic changes to make Connecticut's energy efficiency programs more accessible.

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/41b827a51e5366bb852588220 0719b45?OpenDocument. A recording of this meeting is available here:

https://ctdeep.zoom.us/rec/play/JD6l0q_TrAbn4AY_L4bZow136EF4jtm1-

xwhnX2tJFTTq9q9sAPoz5qPTPmNrpZD-n0YnNzJdA nhcfo.yLY5PZwTfYd0jpKc?continueMode=true

¹⁸ Written comments submitted in response to the Draft Determination are available here:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/04d115cb68d338b7852587880 07091c9?OpenDocument

²⁰ See Id.

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/ea714e056ac86bef8525882200 71e982?OpenDocument

¹⁷ See DEEP Notice of Public Meetings and Request for Written Comments – Draft Determination on the 2022-2024 C&LM Plan, April 12, 2022, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/\$EnergyView?OpenForm&Start=1&Count=30&Expand=7.1&Seq=11 ¹⁹ See Proposed 2022-2024 Conservation and Load Management Plan, November 1, 2021, available at:

²¹ See Equitable Energy Efficiency Proceeding Phase 1 Progress Report, December 28, 2021, available at: https://portal.ct.gov/-/media/DEEP/energy/ConserLoadMgmt/Fall-2021-E3-Progress-Report.pdf

²² The EEB issued a Request for Proposals for a DEI Consultant on November 17, 2021, *available at:* <u>https://energizect.com/sites/default/files/2021-12/DEI%20Consultant%20RFP.pdf</u>

During this Plan term, there may be opportunities to coordinate with the implementation of other funding sources, including increased Weatherization Assistance Program (WAP) funds pursuant to the Infrastructure Investment and Jobs Act (IIJA), to enhance the delivery of energy efficiency programs to low-income and other underserved communities. This Determination calls for DEEP, the Utilities, the EEB, and other stakeholders to seek opportunities to collaborate and coordinate to take full advantage of these opportunities. Additionally, this Determination provides further guidance on the use of RGGI funds to ensure equitable distribution in Section III.e., below.

b. Decarbonization

Decarbonization is the second key priority of the proposed 2022-2024 C&LM Plan. Although the proposed Plan includes several strategies to enhance the C&LM portfolio's impact on emissions, DEEP's recent report that Connecticut is not on track to meet economy-wide greenhouse gas emissions reduction targets established in the Global Warming Solutions Act (GWSA) demonstrates that more ambitious action is required.²³ As outlined later in this Determination, greenhouse gas emissions from use of fossil fuels in residences needs to be reduced several times more rapidly from 2018 to 2030 than they had been from 2001 to 2018; and emissions from use of these fuels in commercial buildings must be reduced even more rapidly. Specifically, the 2018 Connecticut Greenhouse Gas Emissions Inventory found that, based on the most recent greenhouse gas emission data available, emissions from residential, commercial and industrial buildings increased from 2017-2018, which is not in line with the approximately 30% reduction in such emissions that will be required to meet the 45% reduction in greenhouse gas emissions needed economy-wide by 2030.²⁴ The updated Comprehensive Energy Strategy (CES), which began development in early 2022, will provide a roadmap for some of this work. Pursuant to Governor Lamont's Executive Order 21-3, the CES must include strategies to, among other things, provide for more affordable heating and cooling for Connecticut residents and businesses and achieve reductions in greenhouse gas emissions from residential and commercial buildings and industrial facilities as required to meet GWSA targets.²⁵ There are critical steps that the C&LM Plan must take to remove barriers to increased decarbonization and position C&LM programs to be responsive to eventual CES directives. This Determination outlines a set of incremental steps designed to pave the way to more widespread building decarbonization. These steps include:

- Modifying cost-effectiveness testing to capture the impact of emissions on the energy system;
- Enabling customers to select the most efficient measures; and,
- Aligning C&LM incentives with statewide climate policy goals.

These steps will build toward a planned phase-out of all residential natural gas equipment incentives during this Plan term. This incremental approach reflects the need to identify and address existing economic, logistical, and other barriers to widespread adoption of renewable thermal measures, including heat pumps, and other natural gas alternatives as a critical step to prepare for phase-out of natural gas equipment incentives. In the near-term, the Utilities are directed to take certain steps to remove barriers to heat pump adoption, including strengthening workforce development and customer education initiatives and aligning incentives with the cost effectiveness testing approach adopted herein. DEEP will continue to work with the EEB, Utilities, and stakeholders throughout the 2022-2024 Plan term to phase natural gas equipment incentives out of the residential portfolio. A planned timeline for this phase-out is described in the sections below and in the Conditions of Approval (Attachment A).

c. Energy Affordability

²³ See DEEP 2018 Greenhouse Gas Emissions Inventory, October 2021, *available at*: <u>https://portal.ct.gov/DEEP/Climate-Change/CT-Greenhouse-Gas-Inventory-Reports</u>

²⁴ See DEEP 2018 Greenhouse Gas Emissions Inventory, October 2021, available at:

https://portal.ct.gov/DEEP/Climate-Change/CT-Greenhouse-Gas-Inventory-Reports

²⁵ See Governor Lamont Executive Order No. 21-3, December 16, 2021, *available at*: <u>https://portal.ct.gov/-/media/Office-of-the-Governor/Executive-Orders/Lamont-Executive-Orders/Executive-Order-No-21-3.pdf</u>

Energy Affordability is the third key priority of the proposed 2022-2024 C&LM Plan. With Connecticut facing some of the highest average annual energy costs in the nation, the state's energy efficiency programs are vital to managing residents' energy burdens. The average Connecticut household spends \$3,600 per year on energy bills, the second highest average in the country.²⁶ Low-income households spend 8-19 percent of their annual income on energy bills, which is well above the commonly accepted 6 percent threshold for affordability.²⁷

The proposed Plan includes several enhancements to improve energy affordability, including deploying additional residential and commercial customer education resources, providing vendors with tools to reach priority customers, enhanced promotion of loan products for businesses, and targeting customers in distressed communities.

The sections below outline areas where work towards this priority could be improved and associated Conditions of Approval. These include continued targeted outreach to arrearage and shutoff customers, improvements to the Utilities' proposed residential concierge service offering, and measures to streamline commercial and industrial projects.

There are interconnections among these three priorities, and they should not be considered in isolation. For example, data from the Energy Information Agency indicates that the least efficient, highest-emitting homes are concentrated in low-income communities that simultaneously face high energy burdens and significant barriers to receiving weatherization and energy retrofit services.²⁸ The 2022-2024 C&LM Plan and this Determination recognize and seek to address that intersection.

III. <u>Findings</u>

a. Statutory Consistency

DEEP is authorized to review and approve the C&LM Plan to ensure that it contains cost-effective energy conservation, demand management, and market transformation initiatives.²⁹ DEEP finds that the Plan, as modified by this Determination, is consistent with Conn. Gen. Stat. § 16-245m, Conn. Gen. Stat. § 16a-35k, Connecticut's Comprehensive Energy Strategy, Connecticut's Integrated Resources Plan, and the Global Warming Solutions Act.

In approving, with modifications and conditions, the 2022-2024 C&LM Plan, and ensuring coordination of this Plan with other state plans, DEEP is supporting the state's responsibility to coordinate state plans and functions to maximize the use of energy efficient systems and minimize the environmental impact of energy production and use, pursuant to Conn. Gen. Stat. § 22a-1a. Consistent with Conn. Gen. Stat. § 16a-3a, the C&LM Plan advances Connecticut's progress in ensuring that energy resource needs are first met "through all available energy-efficiency and demand reduction resources that are cost-effective, reliable, and feasible," which will "minimize the cost of all energy resources to customers over time." Additionally, the C&LM Plan contains steps to meet the emissions reductions targets established in the

²⁶ See U.S. Department of Energy Low-Income Energy Affordability Data (LEAD) Tool, *available at*: <u>https://www.energy.gov/eere/slsc/maps/lead-tool</u>

²⁷ See Id. The 6 percent affordability threshold assumes that a household should spend 30 percent of its income on shelter costs, and energy costs should be no more than 20 percent of shelter costs.

²⁸ See comments submitted by Acadia Center regarding DEEP's Draft Determination on the 2022-2024 C&LM Plan, April 27, 2022, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/036f5cb6b6a05b3c8525883300 33182c?OpenDocument

²⁹ CGS 16-245m(d)(1)

Global Warming Solutions Act,³⁰ the goals and recommendations of the Governor's Council on Climate Change (GC3),³¹ and the goal of weatherizing 80 percent of residential units by 2030 established in Conn. Gen. Stat. § 16-245m(d)(1).

b. Collaboration with Energy Efficiency Board and Coordinated Policy Priorities

The C&LM Plan reflects Connecticut's energy and environmental policy priorities, provides details about program designs refined through collaboration with the EEB, identifies strategies for reaching targeted customer segments, and catalyzes technologies that will advance Connecticut's energy efficiency goals. DEEP concurs with these areas of focus and appreciates the Utilities' and the EEB's efforts to ensure the Plan's priorities reflect Connecticut's energy and environmental strategies and statutes.

c. Accountability and Transparency

Progress made toward statutory and C&LM-related goals cited herein is reported through a variety of channels, including compliance filings with DEEP, quarterly and year-end reporting to the EEB, the statewide energy efficiency dashboard, and reports such as the annual Equitable Distribution Report, among others.³² In its review of the 2020 C&LM Plan Update, DEEP directed the Utilities to develop a list of quarterly program reports required for program oversight.³³ This list, submitted by the Utilities in June 2020, outlines the reporting structure to which the Utilities adhered during the 2019-2021 C&LM Plan term.³⁴

However, stakeholders have noted challenges accessing relevant reporting and commented that the Plan lacks clarity regarding how the programs as proposed are serving the three priorities for this C&LM Plan term: equity, decarbonization, and affordability.³⁵

As further described in the Conditions of Approval, DEEP is directing the Utilities to continue the reporting practices outlined in their response to Condition of Approval No. 18 to the 2020 C&LM Plan Update, with some modifications to ensure that reports are accessible to stakeholders and clearly demonstrate how the programs are advancing equity, decarbonization, and affordability.

³⁰ See Public Act No. 08-98 "An Act Concerning Connecticut Global Warming Solutions, available at: https://www.cga.ct.gov/2008/ACT/PA/2008PA-00098-R00HB-05600-PA.htm

³¹ See Governor's Council on Climate Change, Phase 1 Report: Near-Term Actions, January2021, available at: https://portal.ct.gov/-/media/DEEP/climatechange/GC3/GC3 Phase1 Report Jan2021.pdf

 ³² For a full inventory of program data and reporting resources, *see* this September 2021 EEB Technical Consultant presentation: <u>https://app.box.com/s/m8885iktcpb5zxodi0t9lem2qablljbx/file/857348044627</u>
³³ See DEEP's Conditional Approval of the 2020 C&LM Plan Update (Appendix A), February 11, 2020, *available*

³³ See DEEP's Conditional Approval of the 2020 C&LM Plan Update (Appendix A), February 11, 2020, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/16d2e80a4a780ab78525850b0 057ec6a?OpenDocument

³⁴ See Utilities' response to Condition of Approval No. 18 of the 2020 C&LM Plan Update, June 30, 2020, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/23661838715798ce852585970 06ce1e1?OpenDocument

³⁵ See for instance, written and verbal comments submitted to DEEP by CT Green Building Council, Henry Auer, Vincent Giordano, Melissa Leonard (Naugatuck Conservation Commission), Shannon Laun (Conservation Law Foundation). Written comments *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/\$EnergyView?OpenForm&Start=1&Count=30&Expand=7.1&Seq=5. Recordings of verbal comments available at: <u>https://ctdeep.zoom.us/rec/play/kCpwKDLuHnAuZp2-1uApHl64bJczhRs6KAcuqkuQMdd6ITkEBGx_zwb74V7SmxsFmicNzh0VhkOV-_0f.M5DYF7R-Hl7Eajy4?continueMode=true</u>

Additionally, in its review of the 2022-2024 C&LM Plan, the EEB is working to catalogue Plan commitments and establish a system for tracking progress toward these commitments over time.³⁶ DEEP looks forward to working with the EEB, Technical Consultants, and Utilities to develop a transparent process for reporting progress on these commitments.

d. Ability to Respond to New Funding Sources

The federal Infrastructure Investment and Jobs Act (IIJA), enacted on November 15, 2021, includes several programs and funding opportunities that have the potential to bring additional federal investment to Connecticut that will contribute to meeting the measures and objectives in the C&LM Plan.³⁷ Securing additional federal funding will require DEEP, the Utilities, and other stakeholders to be nimble and responsive to new funding opportunities.

To ensure a coordinated response, DEEP is directing the Utilities, through this Determination, to identify IIJA funding opportunities that are relevant to the C&LM Plan and for which the Utilities are eligible to apply, and to coordinate with DEEP on these opportunities. Further direction on this plan is provided in the Conditions of Approval.

DEEP is further directing the Utilities to coordinate with DEEP and other stakeholders to develop approaches to "braid" (or combine multiple sources of) funding for the C&LM programs and the federally funded Weatherization Assistance Program (WAP), so that increased funds from IIJA are leveraged to build WAP into a sustainable program that addresses energy affordability in distressed communities and environmental justice areas, targeted workforce development in populations disproportionately affected by climate change, and carbon emissions reduction through energy efficiency and renewable energy deployment in communities with poor social determinants of health.

e. Determination of Equitable Distribution

Pursuant to Conn. Gen. Stat. § 16-245ee, the DEEP Commissioner, "shall determine that an equitable amount of the funds administered by [various boards] are to be deployed among small and large customers with a maximum average monthly peak demand of one hundred kilowatts in census tracts in which the median income is not more than sixty per cent of the state median income." A definition of "equitable" is not established in the statute, which grants the DEEP Commissioner the authority to "determine the equitable share."

DEEP conducts regular, retrospective analyses of C&LM funds through its Equitable Distribution Reports, the most recent of which covers the 2018 program year. Historically, these reports have determined that equitable distribution occurs when, "census tracts (receive) an equal or greater percentage of total incentives than that same tract's contributions as a percentage of total billed collections."³⁸ Using that definition, DEEP's review of the most recent data analyzed found that the overall distribution of revenues was not equitably deployed in 2018.³⁹ DEEP is currently reviewing data from 2019 and 2020

https://app.box.com/s/ht3zucrs01vm9izl0tlzow57p9afcakn/file/905319037330

³⁶ While the development of this commitment tracker is still in progress, a presentation describing the process of commitment tracking can be found here:

³⁷ See Public Act 117-58, An Act to authorize funds for Federal-aid highways, highway safety programs, and transit programs, and for other purposes, November 15, 2021, *available at*: https://www.congress.gov/117/plaws/publ58/PLAW-117publ58.pdf

³⁸ See 2018 Equitable Distribution: Report on the Equitable Distribution of Conservation and Load Management and Renewable Energy Funds in Connecticut, July 2021, *available at*: <u>https://portal.ct.gov/-/media/DEEP/energy/ConserLoadMgmt/2018-EQD.pdf</u>

³⁹ See Id.

and will analyze data from 2021 when it becomes available in 2022. However, DEEP is also reexamining the definition of "equitable distribution."

Given the directive of Conn. Gen. Stat. § 16-245ee, DEEP has been working with the EEB, Utilities, and other stakeholders on an Equitable Energy Efficiency (E3) Proceeding that sets forth strategies to increase deployment of energy efficiency programs to underserved communities. Among these strategies is an examination of current Equitable Distribution Reporting practices to ensure that the definition of "equitable distribution" is accurate and comprehensive. As a preliminary step, the 2019 Equitable Distribution Report will contain new lines of analysis, including an assessment of Home Energy Solutions (HES) and Home Energy Solutions-Income Eligible (HES-IE) in communities of color, census tracts with high average energy burdens, and areas with high rates of arrearages and utility shutoffs.⁴⁰

Some recommendations from Phase I of the E3 Proceeding were incorporated into the proposed 2022-2024 C&LM Plan, while others are still in progress.⁴¹ The E3 Proceeding is scheduled to continue with Phase 2 in 2022 with the support of a Diversity, Equity, and Inclusion (DEI) Consultant. Once in place, the DEI Consultant will work with the EEB and DEEP to thoroughly review how "equitable distribution" is both measured and achieved in C&LM programming.⁴²

The 2022-2024 C&LM Plan includes a parity analysis of revenues and budgets across customer classes. With some variation year-to-year, commercial and industrial customers account for 52 percent of the budget and 52 percent of revenues, residential income-eligible customers account for 14 percent of the budget and 13 percent of revenues, and market-rate residential customers account for 34 percent of the budget and 35 percent of revenues. These projections suggest near-parity for the electric sector.⁴³

During the public comment periods, some stakeholders called for budget allocations based on equity rather than parity, and enhanced funding for low-income programs.⁴⁴ Although equitable distribution of C&LM funding has historically been based on parity, equitable distribution, and not parity, is statutorily required.⁴⁵ Parity can be useful to ensure that, for example, residential rate classes and commercial rate classes are not unduly cross-subsidizing each other. But determining equitable distribution of benefits within a rate class requires more than an examination of parity according to income level. Additionally, in the event that income-eligible programs have over- or under-utilized their resources, the Utilities should re-allocate funds from market rate programs to ensure that income-eligible programs remain accessible.

Benchmarking of budget allocations for low-income programs in other states indicates that Connecticut's low-income spending as a percentage of its total energy efficiency budget is comparable to regional

⁴⁰ See DEEP Final Determination, Equitable Energy Efficiency Phase I Goals and Actions, July 21, 2021, *available at*: <u>https://portal.ct.gov/-/media/DEEP/energy/ConserLoadMgmt/Final-E3-Phase-I-Determination.pdf</u>

⁴¹ See DEEP Equitable Energy Efficiency Proceeding Phase 1 Progress Report, December 28, 2021, available at: https://portal.ct.gov/-/media/DEEP/energy/ConserLoadMgmt/Fall-2021-E3-Progress-Report.pdf

⁴² See EEB Request for Proposals: Consultant on Diversity, Equity, and Inclusion to the CT Energy Efficiency Board, November 17, 2021 (revised December 2, 2021), *available at*: <u>https://energizect.com/sites/default/files/2021-12/DEI%20Consultant%20RFP.pdf</u>

⁴³ See Proposed 2022-2024 Conservation and Load Management Plan, November 1, 2021, *available at*: <u>http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/04d115cb68d338b7852587880</u> 07091c9?OpenDocument

⁴⁴ *See*, for instance, public comments provided by The Connecticut Green Building Council, Melissa Kops, and Kimberly Stoner, summarized in Appendix C and Summary of public comments from the Energy Efficiency Board's August 2021 public input sessions, August 23, 2021, *available at*: https://app.box.com/s/43qlddtxd0cxg4z9z4iaon4k0fc1utbh/file/853820676721

⁴⁵ See Conn. Gen. Stat. § 16-245ee.

neighbors.⁴⁶ However, other states employ policies such as minimum low-income spending requirements and more expansive definitions of low-income to enhance funding directed at low-income programs.⁴⁷

DEEP has also examined federal policy and guidance regarding equity in low-income spending. The Biden Administration's Justice40 Initiative (Justice40), for example, is an effort of the federal government to deliver at least 40% of the overall benefits from Federal investments in climate and clean energy to disadvantaged communities.⁴⁸ In Interim Guidance, issued on July 20, 2021, the "disadvantaged communities" at the center of the Justice40 Initiative were defined as "a group of individuals living in geographic proximity to one another, or a geographically dispersed set of individuals (such as migrant workers or Native Americans) where either type of group experiences common conditions."⁴⁹ Those common conditions might include low-income or high poverty, high unemployment or underemployment, high housing and/or transportation costs, racial and/or ethnic segregation, or disproportionate impacts from climate change.⁵⁰ Justice40's broad definition of "disadvantaged communities" goes beyond the population of residential income-eligible customers included in the parity analysis in the 2022-2024 C&LM Plan.

Given this difference, DEEP strongly encourages the EEB to work with the DEI Consultant, once hired, to investigate how future equity analyses can be conducted in alignment with Justice40 principles. With consultation of the Interim Guidance, these issues will be reviewed in DEEP's Equitable Energy Efficiency (E3) Proceeding, for both residential and commercial customers. DEEP looks forward to coordinating with the EEB and the new DEI consultant to work towards a more equitable energy efficiency landscape. This work will also require a more granular understanding of parity among commercial and industrial customers, who are grouped into a single class in the 2022-2024 parity analysis. As further described in the Conditions of Approval, the Utilities are directed to perform a parity analysis that breaks down revenues and spending by C&I customer size.

DEEP agrees that equity requires more than parity, and that the increased funding should be provided for low-income programs. As noted above, the C&LM Plan is funded in part by RGGI auction proceeds. The use of RGGI auction proceeds for energy efficiency provides for a "virtuous cycle" of decarbonization as part of the State's participation in this regional cap-and-invest program, by investing proceeds from the sale of emissions allowances to fossil fueled power plants into energy efficiency measures that further reduce greenhouse gas emissions from the electricity sector while lowering participants' energy bills.

As further set forth below, as a result of this Determination, delivered fuel customers will be able to participate in the C&LM programs without the need to be funded by RGGI proceeds, which have historically been utilized to fund efficiency for delivered fuels. In considering allocation of RGGI funds within the C&LM Plan, it is appropriate to ensure those funds are directed in a manner that benefits those who have the highest energy burden and have been most impacted by emissions from fossil fuel generation. Since RGGI funds are not subject to the five-percent carryover threshold, and are carried over without a cap, the Utilities can engage in more flexible planning and budget management with the RGGI funding than the CAM funding. DEEP will direct the Utilities on how to prioritize the RGGI funding in a manner that complements the goals of this Determination.

 ⁴⁶ See EEB Technical Consultant presentation, Limited Income Program Budget Benchmarking, January 12, 2022, available at: <u>https://app.box.com/s/ht3zucrs01vm9izl0tlzow57p9afcakn/file/905320117622</u>
⁴⁷ See Id.

⁴⁸ See Young, Mallory, & McCarthy, The Path to Achieving Justice40, July 20, 2021, <u>https://www.whitehouse.gov/omb/briefing-room/2021/07/20/the-path-to-achievingjustice40/#:~:text=Justice40%20is%20a%20whole%2Dof,clean%20energy%20to%20disadvantaged%20communities</u>.

⁴⁹ *Id*.

⁵⁰ Id.

DEEP recognizes that it is important to sustain programs with consistent incentive levels so that customers, including businesses, homeowners, landlords, and renters, know the level of support they can expect to receive. This consistency also helps to maintain stability for businesses that provide energy efficiency services. To that end, the Utilities are directed to take full advantage of all available funding sources, including those described above. Specifically, the Utilities should redirect RGGI funding as needed to programs benefitting low-income and other underserved customers, including those who live in multi-unit dwellings. The Utilities are directed to immediately apply this guidance to address any budgetary issues and keep DEEP apprised of any pressure on budgets, particularly those related to low-income programs, so that reallocations or other solutions can be investigated.

f. Enabling Fuel-Neutral Efficiency Upgrades

Existing statutes support fuel-neutral approaches within the C&LM Plan. Public Act 18-50 establishes a requirement "to reduce energy consumption by not less than 1.6 million MMBtu, or the equivalent megawatts of electricity, as defined in subdivision (4) of section 22a-197 of the general statutes, annually each year for calendar years commencing on and after January 1, 2020, up to and including calendar year 2025."⁵¹ Accordingly, the proposed 2022-2024 C&LM Plan outlines the Utilities' annual fuel-neutral MMBtu savings goals in service of state legislative goals. Additionally, in 2018, Conn. Gen. Stat. § 16-245m(d)(3) was changed to provide that cost effectiveness analysis under the C&LM Plan must include "the value and payback period of program benefits for all energy savings," not just savings of electricity and natural gas. The statute does not limit calculation of energy savings based on fuel-type; rather, it is fuel neutral. Based on this language, C&LM incentives should enable the most cost-effective and efficient upgrades on a fuel-neutral basis. Prior to the issuance of this Determination, this statutory change had not been fully implemented.

Currently, oil- and propane-heated homes participating in the residential program offerings are funded by proceeds from the Regional Greenhouse Gas Initiative (RGGI), as available, rather than through the CAM funding provided for in Section 16-245m(d)(1). Given the statutory requirement set forth above, oil- and propane-heated homes should have no barrier to participating in the programs. Additionally, program structures, and especially savings baselines, may constrain customers' ability to select more efficient equipment when doing so would result in conversion from one fuel source to another. For example, from a pure energy-savings perspective, residential heat pumps have higher demonstrated efficiencies than efficient gas furnaces on an annual basis, even in cold climates like Connecticut. Yet, the current incentive structure may make a less-efficient gas furnace more attractive to a customer. Incentive offerings should be crafted to provide all customers with *affordable, reliable,* and *efficient* upgrade options that also align with state and federal policy objectives to reduce our state's carbon footprint. Aligned and appropriate incentives are especially important given the planned phase out of natural gas equipment incentives for residential customers during the 2022-2024 Plan term, as described in Section II.b., above.

Some stakeholders have raised these same concerns, noting that programmatic barriers may be preventing customers from adopting the most energy-efficient and cost-effective measures. They also requested additional guidance from DEEP on this topic.⁵² In their response to DEEP's November 3, 2021, Requests for Information, the Utilities reiterated the need for guidance, stating, "regulatory direction from DEEP is

⁵¹ See Public Act 18-50, An Act Concerning Connecticut's Energy Future, May 24, 2018, available at: https://www.cga.ct.gov/2018/act/pa/pdf/2018PA-00050-R00SB-00009-PA.pdf

⁵² See for instance, Acadia Center comments regarding the 2022-2024 C&LM Plan, submitted to the EEB August 23, 2021, available at: <u>https://app.box.com/s/43qlddtxd0cxg4z9z4iaon4k0fc1utbh/file/850323139512</u> and Connecticut Industrial Energy Consumers' comments submitted to DEEP, December 3, 2021, available at: <u>http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/c7169b9a0c2c9fea852587a700</u> 558f44?OpenDocument

needed regarding whether electric ratepayer funds can be used for fuel switching and fuel optimization offerings."⁵³

Recent guidance from DEEP has offered support for fuel neutrality. The 2018 CES supports electrification in the building and transportation sectors.⁵⁴ In 2018, DEEP's conditional approval of the 2019-2021 C&LM Plan authorized the Heat Pump Pilot program, while noting: "DEEP is not endorsing the use of conservation and load management dollars for fuel avoiding a full endorsement of fuel neutral incentives.⁵⁵ However, in recognition of increased focus on delivered fuel savings in C&I projects, DEEP's conditional approval of the 2021 Plan Update contained a provision allowing the Utilities to "calculate energy savings using a base building or baseline that reflects a fuel type that would have been chosen, absent incentives, regardless of whether it is a different fuel type than that chosen after incentives."⁵⁶ This savings calculation method has not been universally applied to C&LM programs. Meanwhile, neighboring jurisdictions, including New York and Massachusetts, have instituted policies that allow for a more widespread fuel-neutral approach to energy efficiency incentives.

As further described in the Conditions of Approval, DEEP is directing the Utilities to develop a fuelneutral approach to C&LM incentives that prioritizes energy savings irrespective of fuel type, supports customer choice and affordability, and aligns with state policy goals, including the Global Warming Solutions Act. This approach should consider appropriate, fuel-neutral baselines and accounting methods to allocate costs equitably among programs, in accordance with Public Act 18-50.⁵⁷ It should also account for changes to the cost-effectiveness test, as further described in Section III.g. and Attachment B of this Determination.

The state and region are taking steps to prepare the grid for any increased load associated with heat pumps. ISO-New England (ISO-NE), the regional system operator, has built state-specific heat pump deployment projections into its forecasts, and Connecticut participates in the working group to provide feedback while ISO-NE develops that forecast.⁵⁸ These projections are updated annually, which ensures that ISO-NE and the state are not caught off guard as adoption of heat pumps accelerates. Connecticut has also begun planning for increased electrification from deployment of heat pumps and electric vehicles in its Integrated Resources Planning (IRP) process. In the most recently released 2020 IRP, DEEP assumed heat pump deployment would triple by 2040 under its electrification modeling scenarios.⁵⁹

⁵³ This statement was in reference to additional information the Utilities would need to transition the Heat Pump Pilot to a standard program offering. *See* Utility response to RFI #BETP-013, November 16, 2021, *available at*: <u>http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/6cadc3d093b715be852587920</u> 0580264?OpenDocument

⁵⁴ See DEEP Comprehensive Energy Strategy, February 8, 2018, *available at*: <u>https://portal.ct.gov/-/media/DEEP/energy/CES/2018ComprehensiveEnergyStrategypdf.pdf</u>.

⁵⁵ See DEEP Conditional Approval of the 2019-2021 C&LM Plan, Attachment A: Schedule of Conditions of Approval, December 20, 2018, *available at*: <u>https://portal.ct.gov/-</u>

[/]media/DEEP/energy/ConserLoadMgmt/AttachmentAScheduleofComplianceConditionsofApprovalof20192021CL MPlanFinal121918pdf.pdf?la=en&hash=BF8647A8C1BE168FA7C93619D7171469

⁵⁶ See DEEP Determination: Approval With Conditions of the 2021 Plan Update to the 2019-2021 C&LM Plan, March 4, 2021, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/d80f7ae5059c5efc8525868e00 598e40/\$FILE/Determination Approval%20with%20Conditions%202021%20Plan%20Update%20%202020%20P MI%20Adjustment%20(002).pdf

⁵⁷ See Public Act 18-50, An Act Concerning Connecticut's Energy Future, May 24, 2018, available at: https://www.cga.ct.gov/2018/act/pa/pdf/2018PA-00050-R00SB-00009-PA.pdf

⁵⁸ See e.g., ISO-NE's Final 2021 Heating Electrification Forecast, *available at* <u>https://www.iso-ne.com/static-assets/documents/2021/04/final_2021_heat_elec_forecast.pdf</u>.

⁵⁹ Connecticut DEEP, 2020 Integrated Resources Plan- Appendix A1 Factor Inputs, October 7, 2021, available at: https://portal.ct.gov/-/media/DEEP/energy/IRP/2020-IRP/Appendix-A1--Factor-Inputs-and-Assumptions.pdf.

Additionally, PURA is conducting a series of dockets focused on modernizing the electric grid and increasing its resilience.⁶⁰ The programs adopted through these dockets will support increased electrification both directly and indirectly. One of these dockets focuses specifically on DEEP's 2020 Integrated Resources Plan (IRP) (which analyzed the effects of increased heat pump adoption on electric demand)⁶¹ and seeks to "leverage DEEP's work [in the IRP] to ensure that cost-effective and efficient approaches are in place to deliver on the recommended pathways to achieving a one hundred percent zero carbon electric grid by 2040."⁶² PURA also approved an energy storage program in Docket No.17-12-03RE03. Combining solar and/or storage with heat pump deployment will allow the load from heat pumps to be offset by these distributed energy resources, thereby providing resilience benefits, reducing participants' energy bills, and reducing harmful emissions. Accordingly, the state has mechanisms in place to ensure grid reliability as heat pump adoption increases.⁶³ Moreover, given the barriers to heat pump deployment identified herein, DEEP does not expect near-term uptake to significantly impact electric load and will continue to assess heat pump deployment through sensitivities in its electric system modeling of the biennial Integrated Resources Plan.⁶⁴

g. Continuing Review of Program Cost-Effectiveness Testing

Pursuant to Conn. Gen. Stat. 16-245m(d)(3), DEEP has reviewed the Utilities' cost-effectiveness screening for the programs in the C&LM Plan. The Plan includes detailed information on the development of the Utilities' programs' savings goals, as well as their individual benefit/cost ratio models, demonstrating that they are seeking to acquire all cost-effective energy efficiency resources.

Concurrent with development of the proposed 2022-2024 C&LM Plan, DEEP has been conducting a review of cost-effectiveness screening processes and opportunities to align these processes with state policy priorities, including the Global Warming Solutions Act. Drawing on past public processes, testimony received during the November 16, 2021, Technical Meeting, and subsequent public comments, DEEP has developed an updated approach to cost-effectiveness practices that is provided in Attachment B. A draft of this approach was issued with DEEP's Draft Determination on the 2022-2024 C&LM Plan on April 12, 2022.

DEEP modified this approach to cost-effectiveness testing based on feedback received through a public comment process that concluded on April 27, 2022.⁶⁵ Based on that feedback, the cost-effectiveness test update included as Attachment B makes two modifications to current cost-effectiveness testing practices: 1) it applies a Modified Utility Cost Test to all C&LM programs in order to capture all fuel savings, and 2) it incorporates the impact of avoided greenhouse gas emissions resulting from C&LM programs using the Avoided Energy Supply Cost Study (AESC) New England marginal abatement cost, derived from the

http://www.dpuc.state.ct.us/2nddockcurr.nsf/8e6fc37a54110e3e852576190052b64d/6bd751ea426c80198525879100 60dbd8/\$FILE/17-12-03RE10%20NOP.pdf.

⁶⁰ See PURA, Interim Decision, Docket No. 17-12-03, PURA Investigation into Distribution System Planning of the Electric Distribution Companies (Oct. 2, 2019).

 ⁶¹ See Connecticut 2020 Integrated Resources Plan, App. A1. Factor Inputs, at p. 19-22, available at https://portal.ct.gov/-/media/DEEP/energy/IRP/2020-IRP/Appendix-A1--Factor-Inputs-and-Assumptions.pdf.
⁶² See PURA, Notice of Proceeding, Docket No. 17-12-03RE08, PURA Investigation into Distribution System Planning of the Electric Distribution Companies – Building Blocks of Resource Adequacy and Clean Electric Supply, at 1 (Nov. 18, 2021), available at

⁶³ See also Conn. Gen. Stat. § 16a-3b, Implementation of the Integrated Resources Plan ("[PURA] shall oversee the implementation of the Integrated Resources Plan and the Procurement Plan.")

⁶⁴ See supra note 53

⁶⁵ See for instance, comments submitted by Acadia Center, Northeast Energy Efficiency Partnerships; Sierra Club Connecticut; CMC Energy Services, New England Smart Energy Services, and HE Energy Services; and Eversource Energy and Avangrid Networks, Inc., *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/\$EnergyView?OpenForm&Start=1&Count=30&Expand=7.1&Seq=3

electric sector, for non-embedded emissions (currently \$125/short ton).⁶⁶ While stakeholders did express support for the use of social cost of carbon in the cost-effectiveness test,⁶⁷ DEEP notes that the AESC New England marginal abatement cost for the electric sector and the AESC recommended value for social cost of carbon are comparable at \$125/short ton and \$128/short ton, respectively.⁶⁸ DEEP's rationale for this and other modifications to the cost-effectiveness test are further explained in Attachment B.

h. Aligning Incentives with Policy Goals

The Global Warming Solutions Act (GWSA) requires an economy-wide greenhouse gas (GHG) emissions reduction of 45 percent from 2001 to 2030.⁶⁹ Although the GWSA does not establish specific reduction targets for individual sectors of the economy, applying the 2030 economy-wide target to individual sectors illustrates the scale of the challenge of adequately decarbonizing residential and commercial thermal loads.

Between 2001 and 2018, *residential* thermal emissions from combustion of fossil fuels (principally heating oil, natural gas, and propane) fell 10.6 percent -- far less than the 26.4 percent reduction the GWSA's 2030 economy-wide target implies was needed during that period.⁷⁰ Bringing residential emissions in line with the target of 45 percent reductions by 2030 will require reducing emissions 3.6 times faster between 2018 and 2030 than they fell between 2001 and 2018. Meanwhile, between 2001 and 2018 *commercial* GHG emissions from combustion of fossil fuels were essentially unchanged. This means that the full 45 percent reduction the GWSA's 2030 economy-wide target implies for commercial buildings for the period 2001-2030 will need to be accomplished between 2018 and 2030. Moreover, in both of these sectors fossil fuel emissions will need to continue decreasing sharply between 2030 and 2050, when the GWSA requires an additional 35 percent emission reduction across the economy (against a 2001 baseline). Because thermal electrification is one of Connecticut's decarbonization strategies for residential and commercial buildings, DEEP recognizes it will need to factor the pace of decarbonization in the electricity sector (via the Renewable Portfolio Standard and state zero-carbon procurements) into the C&LM Plan.

Results from Connecticut's 2018 Greenhouse Gas Emissions Inventory demonstrate the importance of decarbonizing the buildings sector.⁷¹ In accordance with these findings and public feedback received during DEEP and the EEB's public processes regarding the 2022-2024 C&LM Plan, this Determination outlines strategies to phase out incentives for residential natural gas equipment during this term.⁷² Other

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/\$EnergyView?OpenForm&Start=1&Count=30&Expand=7.1&Seq=3 ⁶⁸ See Avoided Energy Supply Components in New England: 2021 Report, May 2021, available at:

https://www.synapse-energy.com/sites/default/files/AESC 2021_20-068.pdf

https://portal.ct.gov/DEEP/Climate-Change/CT-Greenhouse-Gas-Inventory-Reports

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/\$EnergyView?OpenForm&Start=1&Count=30&Expand=7.1&Seq=2

⁶⁶ See Avoided Energy Supply Components in New England: 2021 Report, May 2021, available at:

https://www.synapse-energy.com/sites/default/files/AESC 2021_20-068.pdf

⁶⁷ See for instance, comments submitted by Acadia Center *available at*:

⁶⁹ See An Act Concerning Connecticut Global Warming Solutions, PA 08-98, 2008, *available at:* <u>https://www.cga.ct.gov/2008/ACT/PA/2008PA-00098-R00HB-05600-PA.htm</u>.

⁷⁰ See DEEP, "2018 Connecticut Greenhouse Gas Emissions Inventory," 2021, available at: <u>https://portal.ct.gov/-/media/DEEP/climatechange/GHG Emissions Inventory 2018.pdf</u>.

⁷¹ See DEEP 2018 Greenhouse Gas Emissions Inventory, October 2021, available at:

⁷² For public comments regarding the cessation of natural gas incentives, *see*, for instance, public comments submitted to DEEP by the Conservation Law Foundation, Nicolette Doukas, Dwayne Escola, Save the Sound, Elizabeth Raisbeck, Tom Swan, December 3, 2021, *available* at:

and verbal comments before DEEP from, among others, Sierra Club Connecticut, Yale Center on Climate Change and Health, CT Green Building Council, Kimberly Stoner, The Nature Conservancy Connecticut, *available at*: https://ctdeep.zoom.us/rec/play/WCdG84uKS96n2x-

gjZuROpKeb3euh4VdU4s1Pfe3MXz9x9oQNaLgyTgToHI_RDzAMfLdesaiLbZWS8ww.lUvzeQ55sAxsCmcG?co

elements of this Determination, including modifications to the cost-effectiveness test and permitting fuelneutral incentives, will support and enable these strategies to align C&LM incentives with state policy goals.⁷³

Although electrification is a key strategy for reducing emissions resulting from the buildings sector, DEEP recognizes that the consumer economics of thermal electrification are sometimes challenging. In late 2021 and early 2022, the EEB Technical Consultants conducted a preliminary cost-effectiveness analysis of space- and water-heating electrification across a variety of displacement scenarios. For the residential sector, this analysis found that replacing oil or propane systems with a ductless or ducted air-source heat pump was almost always cost-effective from the Utilities' perspective. However, there was more variation in utility cost-effectiveness in scenarios where air-source heat pumps replaced a natural gas system. From the customers' perspective, these heat pumps had a positive annual cash flow and life cycle cost only in propane and some oil displacement scenarios under the conditions the consultants examined.⁷⁴ This Determination includes provisions -- including changes to cost-effectiveness testing practices and savings calculation baselines -- to more accurately account for all utility-system impacts of C&LM measures.⁷⁵ These changes will capture previously unaccounted benefits, allowing for increased incentives that are expected to make air-source heat pumps a more viable option for customers.

In the C&I sector, ductless and ducted air-source heat pumps and heat pump hot water heaters replacing oil and propane systems were found to be cost-effective from the Utilities' perspective, while gas system replacements were not. From the customers' perspective, the results were mixed: while gas displacements always had a negative first-year cash flow and lifecycle cost, oil and propane displacement cash flows and lifecycle costs often were positive. The C&I analysis also found that variable refrigerant flow air-source heat pump systems combined with a dedicated outdoor air system could be cost-effective from the Utilities' perspective, though the customer first year cash flow and life cycle costs were still negative.⁷⁶ Additionally, an analysis of ground source heat pumps demonstrated that they were generally cost effective from the Utilities' and customers' perspective.⁷⁷

To ensure investment in measures that optimize decarbonization and energy affordability, DEEP will work with the EEB, Utilities, and other stakeholders to ensure that the transition away from residential natural gas equipment incentives does not place undue burden on customers and makes alternatives, including heat pumps, a viable and affordable option. While changes to cost-effectiveness testing and a fuel-neutral approach to incentives may result in higher incentives for heat pumps, DEEP will also explore strategies to bring down the operating cost of heat pumps for customers, including pathways to advocate for aligned rate design, a strategy supported by stakeholder comments.⁷⁸ Progress toward these

ntinueMode=true (November 16) and <u>https://ctdeep.zoom.us/rec/play/kCpwKDLuHnAuZp2-1uApHI64bJczhRs6KAcuqkuQMdd6lTkEBGx_zwb74V7SmxsFmicNzh0VhkOV-_0f.M5DYF7R-</u>HI7Eaiy42continueMode=true (November 18)

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/\$EnergyView?OpenForm&Start=1&Count=30&Expand=7.1&Seq=2

HI7Eajy4?continueMode=true (November 18) ⁷³ For more on cost-effectiveness testing modifications and fuel-neutral incentives, see Sections III.f. and III.g. of this Determination.

 ⁷⁴ See EEB Technical Consultant presentation, Heating Electrification Analysis – Preliminary Residential Results, December 8, 2021, available at: https://app.box.com/s/kewx0nkos2c2xsvts8xitm0d60hipwsl/file/893583006058

⁷⁵ More information on these provisions can be found in Sections III.f. and III.g. of this Determination.

⁷⁶ See EEB Technical Consultant presentation, Heating Electrification Analysis – Preliminary C&I Results, February 8, 2022, *available at:* <u>https://app.box.com/s/cy013af0uvcsgyynz7hlxw19bbfb4wgq/file/916308638383</u>

⁷⁷ Ground source heat pumps were cost-effective under the Modified Utility Cost Test, which is the primary mechanism for cost-effectiveness testing of C&LM programs. The incremental cost of a ground source heat pump was not cost-effective under the Total Resource Cost Test, which is used as a secondary testing mechanism. Ground source heat pumps in commercial new construction demonstrated positive annual lifecycle costs but negative first-year cash flow. *See* Id.

⁷⁸ See for instance, comments submitted to DEEP by the CT Green Building Council, December 3, 2021, *available at*:

strategies is already underway, including through PURA's May 4, 2022, release of a Low-Income Discount Rate Straw Proposal in Docket No. 17-12-03RE11.⁷⁹

Customer understanding of heat pumps and inadequate workforce development present additional barriers to thermal decarbonization, as noted by stakeholders during public processes provided by DEEP and the EEB.⁸⁰ Positive customer experiences are an important driver of heat pump adoption – an outcome that requires, on one hand, a highly trained workforce that is familiar with the latest technologies, customer needs, and installation practices, and, on the other hand, thorough customer education regarding best operation and maintenance practices. In their response to DEEP's November 30, 2021, Requests for Information, the Utilities elaborated on their plans to launch a Heat Pump Installer Network (HPIN) in 2022 that will connect contractors with training resources on proper installation of heat pump technologies, customer outreach and education strategies, and operation and maintenance practices.⁸¹ Additional DEEP findings regarding the HPIN and customer outreach regarding heat pumps can be found in Sections III.i. and III.k of this Determination.

The 2022-2024 C&LM Plan presents a critical opportunity to address barriers to deeper decarbonization measures like heat pumps and lay the groundwork for enhanced deployment. In early 2022, DEEP began the process of developing a new Comprehensive Energy Strategy (CES) that will concentrate on, among other things, energy-related GHG emissions from residences and commercial buildings, affordable clean heating and cooling options, deployment targets for renewable thermal technologies, and policy mechanisms to meet the state's GHG emissions-reduction goals.⁸² While the more specific goal setting of the CES is underway, there are opportunities to enhance the C&LM Plan's decarbonization strategies and position the C&LM portfolio to achieve greater GHG reductions in accordance with the GWSA upon completion of the CES.

These opportunities (listed below) present an incremental approach to advancing decarbonization goals through the C&LM Plan. As a first step, the Utilities will evaluate the continued need to incentivize a sub-set of highly efficient natural gas measures that may no longer be cost-effective. Second, DEEP will work with the Utilities, EEB, and other stakeholders on a comprehensive plan to address barriers to heat pump adoption. Third, the Utilities are directed to develop a proposal for transitioning the Residential New Construction program to an all-electric offering by July 2023.

In addition to the fuel-neutral approach to incentives and modifications to cost-effectiveness testing described above, these three steps present near-term actions to remove barriers to and prepare the market for increased decarbonization measures. Additional guidance from the CES will provide further direction on feasible steps to achieve the goal, established in this Determination, of phasing out residential natural gas equipment incentives during this Plan term.

http://www.dpuc.state.ct.us/2nddockcurr.nsf/All/789310985B3927D385258838006E9B2C?OpenDocument ⁸⁰ See for instance, public comments submitted to DEEP by the Northeast Energy Efficiency Partnerships and Dandelion Energy, December 3, 2021, *available at*:

⁷⁹ See PURA Notice of Issuance of Low-Income Discount Rate Straw Proposal and Request for Associated Tariffs, May 4, 2022, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/\$EnergyView?OpenForm&Start=1&Count=30&Expand=7.1&Seq=2 and public comments submitted to the EEB by People's Action for Clean Energy, August 18, 2021, available at: https://app.box.com/s/43qlddtxd0cxg4z9z4iaon4k0fc1utbh/file/853820676721

⁸¹ See Utilities' response to DEEP Request for Information #BETP-29, December 9, 2021, available at: http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/8c9906744d48f122852587ab00 524a63?OpenDocument

⁸² See DEEP Notice of Proceeding and Scoping Meeting, 2022 Comprehensive Energy Strategy, January 6, 2022, *available at:* <u>https://portal.ct.gov/-/media/DEEP/energy/CES/2022CESnoticeofproceedingpdf.pdf</u>

i. Assessing the continued need, and potentially discontinuing incentives, for certain natural gas measures

Currently, the C&LM Plan provides incentives only for natural gas measures that qualify as efficient and does not provide incentives for delivered fuel measures. In response to DEEP's November 3, 2021, Requests for Information, the Utilities provided a list of incentive levels and year-to-date reporting for all Residential and Commercial & Industrial natural gas measures supported by the C&LM Plan.⁸³ These include incentives for highly efficient condensing equipment and comparatively less efficient non-condensing equipment.

As noted in the EEB's response to public input on the proposed 2022-2024 C&LM Plan, "a recent evaluation has found that the C&I boiler market is largely transformed, such that it is now industry standard practice in many cases for customers to purchase an efficient condensing boiler for new construction or at end of life for an existing boiler. As a result of this finding, the baseline has risen to a point where it may no longer be cost effective to offer incentives for many commercial boilers."⁸⁴ This study indicates the need for further examination of incentives for certain natural gas measures in the C&I sector.

Similarly, in the residential sector, the Utilities have committed to "investigating whether program supported natural gas-fired condensing equipment is replacing existing condensing equipment as a market baseline, and if so, whether there is continued need to [continue program support for this equipment]" during the 2022-2024 Plan term.⁸⁵

As further described in the Conditions of Approval, if an evaluation determines that such equipment should now be considered the baseline, incentives should be discontinued. Thus, DEEP is directing the Utilities to work with the Evaluation Administrator to complete this investigation by June 2022 and file their findings with DEEP for approval shortly thereafter.

Moreover, DEEP will work with the Utilities and the EEB to phase out residential incentives for other gas measures over the 2022-2024 Plan term.

ii. Developing a comprehensive strategic plan to address barriers to heat pump deployment.

Key barriers to deployment of heat pumps identified in a Yale University report in 2017 continue to hinder these renewable thermal technologies in 2022.⁸⁶ The Plan establishes decarbonization as a priority and highlights efforts to promote installation of more heat pumps. While the Plan outlines many of the elements necessary to achieve decarbonization, it does not

/media/DEEP/energy/ConserLoadMgmt/Final-2022-2024-Plan-to-EEB-1112021.pdf ⁸⁵ See Proposed 2022-2024 Conservation and Load Management Plan, November 1, 2021, available at: <u>http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/04d115cb68d338b7852587880</u> <u>07091c9?OpenDocument</u>

⁸³ See Utilities' response to DEEP Request for Information #BETP-1, November 12, 2021, available at: <u>http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/7091cd5a6937ac52852587900</u> 05a4d16?OpenDocument

⁸⁴ See EEB's response to August 1, 2021 comments from Sierra Club of Connecticut (Samantha Dynowski), proposed 2022-2024 C&LM Plan (Appendix B), *available at*: <u>https://portal.ct.gov/-</u>

⁸⁶ See Helle Gronli et al., "Feasibility of Renewable Thermal Technologies in Connecticut – Barriers and Drivers," 2017, *available at*: <u>https://cbey.yale.edu/research/feasibility-of-renewable-thermal-technologies-in-connecticut-barriers-and-drivers</u>.

place these elements in an explicit strategic framework and does not lay out specific benchmarks.

In its review of the proposed Plan and through the stakeholder engagement process, DEEP has identified three primary categories of barriers to heat pump deployment: 1) workforce development and quality assurance, 2) financial impediments, and 3) marketing, education, and customer support. DEEP will work with stakeholders and the Utilities to develop a strategic plan to address those barriers. The strategic plan will also consider strategies to co-deliver heat pumps with measures that will improve customer experiences, such as weatherization, integrated controls, and demand response. Specific items for consideration include:

Adequate workforce and quality assurance

- workforce development, training, and certification;
- potential additional support for the installer network;
- HES vendor training and compliance;
- post-installation quality certification;
- training for HVAC contractors on implementing thermal envelope improvements to improve in-home comfort levels and achieve cost savings in heat pump deployment;

Optimal incentives and measure delivery

- expected impacts of changes in cost-effectiveness testing on heat pump incentives;
- potential revision of heat pump installation incentives;
- potential need to revise equipment specifications for heat pump incentives in light the availability of the new national specifications (including the Energy Star v. 6 specification, the DOE Cold-Climate Heat Pump Challenge specification, revised NEEP cold-climate specification; and EXP07 Load-based Test Procedure);
- potential need to revise loan programs;
- opportunities for integration with rooftop PV;
- opportunities for integration with HES and weatherization;
- potential role of controls that integrate operation of heat pumps with operation of conventional fossil-fuel equipment;

Effective marketing, education, and consumer support

- market research and market segmentation;
- enhancement of heat pump emphasis and effective public messaging in Energize CT;
- potential employment of advertising and public-service announcements;
- heat pump installer locator tool;
- consumer education before and after installation;
- concierge service;

Co-delivery of heat pumps with other supportive measures:

- incentivizing the co-delivery of weatherization and heat pumps;
- integrating heat pump offerings with distributed generation, including through PURA's new statewide electric storage program and solar tariff;
- coupling heat pump installations with integrated controls and active demand response measures

To develop this strategic plan, DEEP will convene a Technical Meeting to explore barriers and best practices related to the topics above. This Technical Meeting will be held pursuant to this C&LM Plan approval proceeding as well as in conjunction with DEEP's ongoing CES development process. Following the Technical Meeting, DEEP will issue a framework for the

strategic plan. The Utilities will then be directed to propose a strategic plan for addressing barriers to heat pump adoption that reflects the discussion from the Technical Meeting. This plan should propose strategies that can be incorporated into annual C&LM Plan Updates.

iii. Transitioning to an all-electric new construction offering.

All-electric new construction presents an opportunity for savings on both up-front costs and utility bills compared to new mixed-fuel homes. Analysis of residential new construction markets found the net present cost of all-electric new construction to be lower than mixed-fuel new construction in Boston and New York City. In these markets, all-electric new construction is also projected to result in significant carbon emissions reductions, with individual homes reducing annual carbon emissions by 46 and 51 tons (for Boston and New York City, respectively) over a 15-year period.⁸⁷

Despite some of the highest electricity rates in the nation,⁸⁸ efficient, all-electric homes can still provide savings for Connecticut customers. An analysis of all-electric homes with heat pumps in Connecticut found similar or lower operating costs compared to homes that utilize heating oil or natural gas.⁸⁹ When paired with solar and/or storage installations, costs could be considerably lower. Additionally, advances in heat pump technology have made them a more viable option for keeping homes warm throughout the winter months.⁹⁰

Therefore, in the Conditions of Approval, the Utilities are directed to develop a proposal for transitioning the Residential New Construction program to an all-electric offering by the end of 2023.

The timeline for implementing these strategies will also be dependent on the adoption and contents of Connecticut's new Statewide Building Code. Governor Lamont's Executive Order No. 21-3 directs the State Building Inspector to present the most recent International Energy Conservation Code (IECC) for adoption and "develop a plan to incorporate the reduction of greenhouse gas emissions as a core consideration when adopting the State Building code given the health implications of continued greenhouse gas emissions."⁹¹ As the Codes and Standards Committee works to adopt the 2022 Connecticut State Building Code, DEEP will ensure that the provisions of this Determination align with the new code, including making adjustments to timelines as necessary.⁹²

Additionally, the Utilities are directed through the Conditions of Approval to propose updated incentives for C&LM measures, applying the tenets of Sections III.f.-III.h. In this proposal, DEEP has directed the Utilities to consider optimal incentive structures for building decarbonization, including an exploration of renewable thermal measures that are not currently incentivized through C&LM, and methods of co-

⁸⁷ See RMI, The New Economics of Electrifying Buildings: An Analysis of Seven Cities, 2020, available at: <u>https://rmi.org/insight/the-new-economics-of-electrifying-buildings?submitted=1983dhtw8</u>

⁸⁸ See State Electricity Profiles, U.S. Energy Information Administration, November 2, 2020, *available at*: <u>https://www.eia.gov/electricity/state/</u>

⁸⁹ See Acadia Center comments to the Energy Efficiency Board regarding the 2022-2024 Conservation and Load Management Plan, August 2021, *available*

at: https://app.box.com/s/43qlddtxd0cxg4z9z4iaon4k0fc1utbh/file/846152859499

⁹⁰ See Load-based and Climate-Specific Testing and Rating Procedures for Heat Pumps and Air Conditioners, Northwest Energy Efficiency Alliance, July 7, 2020, *available at*: <u>https://neea.org/img/documents/CSA-EXP07-Interim-Testing-Report.pdf</u>

⁹¹ See Governor Lamont Executive Order No. 21-3, December 16, 2021, *available at*: <u>https://portal.ct.gov/-</u>/media/Office-of-the-Governor/Executive-Orders/Lamont-Executive-Orders/Executive-Order-No-21-3.pdf

⁹² More information about the Connecticut State Building Code adoption process is available on the Department of Administrative Services website: <u>https://portal.ct.gov/DAS/Office-of-State-Building-Inspector/Building-and-Fire-Code-Adoption-Process</u>

delivery with supportive measures including weatherization, integrated controls, demand response, and distributed generation.

The steps outlined in Sections III.f.-III.h, of this Determination are near-term actions that, paired with forthcoming guidance from the CES, will build towards a phase-out of residential natural gas equipment during this Plan term. DEEP will continue to work with the Utilities, EEB, and other stakeholders to further explore decarbonization opportunities in the commercial and industrial space. For example, DEEP anticipates that the CES will establish a need for a broad strategy for industrial decarbonization and outline a series of steps to move in this direction, which would likely include opportunities for stakeholder engagement on, among other issues, alignment with corporate sustainability goals.⁹³

i. Customer Outreach and Engagement

DEEP recognizes the importance of community outreach and, in particular, engaging municipalities and other partners in promoting and delivering energy efficiency. The proposed 2022-2024 C&LM Plan includes continuation of the Community Partnership Initiative, the Utilities' strategy to work with community groups, nonprofit organizations, and municipalities to raise awareness and measurable participation in energy efficiency programs.⁹⁴ The Utilities selected the first round of sponsors in Fall 2021 to deliver local outreach strategies to advance goals such as increasing participation in Home Energy Solutions and Home Energy Solutions-Income Eligible, increasing the number of rebates submitted for heat pumps that displace electric-resistance heating, and increasing participation in the Small Business Energy Advantage program.⁹⁵ During the 2022-2024 Plan term, DEEP will be prioritizing community approaches to energy efficiency, and the EEB Diversity, Equity, and Inclusion Consultant will play an important role in consulting on the design and implementation of community outreach strategies.⁹⁶ As further described in the Conditions of Approval below, the Utilities are directed to use subsequent rounds of the Community Partnership Initiative to serve the Plan's three priorities: equity, decarbonization, and energy affordability. Future rounds of the Community Partnership Initiative will occur according to the Utilities' schedule. The Utilities will engage the expertise of the EEB's Diversity, Equity, and Inclusion Consultant (once hired) on future rounds of the Community Partnership Initiative and keep DEEP and the EEB apprised of developments.

Decarbonization of residential thermal demand requires significant time and attention by building owners and residents and warrants significant enhancements to the proposed concierge service. Residential thermal decarbonization often involves multiple steps, starting with the initial visit and audit of a building, addressing health and safety barriers to weatherization, and providing air sealing and installing insulation. Deeper measures often involve sophisticated, rapidly evolving technologies such as heat pumps with which many HVAC installers – and the vast majority of residents – have limited or outdated experience. These low- and zero-carbon technologies must be integrated into complex home energy systems that HES assessments touch upon in only limited fashion. As the Utilities attempt to make clean

⁹³ See comments from Connecticut Industrial Energy Consumers regarding DEEP's Draft Determination on the 2022-2024 C&LM Plan, April 27, 2022, available at:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/5416f73f42c7aeef8525883300 33d242?OpenDocument

⁹⁴ See Energize CT, Community Partnership Initiative, *available at*: Proposed 2022-2024 Conservation and Load Management Plan, November 1, 2021, available at:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/04d115cb68d338b7852587880 07091c9?OpenDocument

⁹⁵ See Energize CT, Community Partnership for Energy Efficiency Engagement Initiative: Round 1 Application for Interested Parties, August 4, 2021, *available at*: <u>https://energizect.com/sites/default/files/2021-</u>08/20210804%20REVISED%20Community%20Partnership%20Application_p1-8.pdf

⁹⁶ See Request for Proposals: Consultant on Diversity, Equity, and Inclusion to the CT Energy Efficiency Board, December 2, 2021: <u>https://energizect.com/sites/default/files/2021-12/DEI%20Consultant%20RFP.pdf</u>

heating technologies more visible, more accessible, better understood, and more effectively implemented in Connecticut homes, they are acknowledging a need for a concierge service that will help residents navigate this new terrain successfully.

During the 2022-2024 Plan term, the Utilities plan to offer heat pump consultation services that will support customer adoption of heat pumps statewide. As of March 1, 2022, the Utilities have contracted with an implementation vendor to provide a variety of services, including virtual no-cost heat pump consultations, community webinars, and educational resources for customers and contractors, and to manage the new Heat Pump Installer Network.⁹⁷ These consultation services will give customers access to a heat pump specialist that possesses the technical expertise required to assist customers throughout the process of selecting and installing a heat pump.⁹⁸

In addition to the heat pump consultation services that the Utilities plan to provide during this Plan term, the 2022-2024 C&LM Plan also includes a residential energy efficiency concierge service. The concierge service the Utilities proposed relies on a combination of e-mail communications and a series of videos that provide instruction on add-on measures such as insulation, heat pumps, efficient appliances, and clean energy and electrification technologies.⁹⁹ This is considerably less comprehensive than a program recently launched by Mass Save, which involves live, virtual, one-on-one sessions with subject-matter experts who are associated with a Mass Save vendor.¹⁰⁰ These experts, who have technical backgrounds and HVAC training, answer questions and direct customers to qualified contractors participating in Mass Save's Residential Heat Pump Installer Network. The Massachusetts utilities plan to conduct post-participation surveys.

Other initiatives now underway in the region provide additional examples of comprehensive concierge services. The I Heart My Home (IHMH) service offered by the nonprofit organization NHS New Haven¹⁰¹ allows home-energy specialists to meet one-on-one with customers and provide custom, indepth analyses of their homes' energy needs and upgrade options. They assist customers in navigating technical options and utility incentive programs, work with customers to understand contractor proposals, and, if needed, communicate with contractors on behalf of customers. IHMH provides these services as a disinterested third party. The City of Hartford is also developing an energy coaching program for residents.¹⁰²

The HeatSmart CT program,¹⁰³ run by the nonprofit People's Action for Clean Energy and carried out in some instances with the assistance of IHMH, demonstrates that hands-on programs that holistically address each home's specific energy needs can address the generally low level of public understanding of heat pumps and other efficient technologies. The program also equips the HVAC community with more knowledge of heat pump technologies.

⁹⁷ See Eversource Energy and Avangrid Networks, Inc. comments in response to DEEP's Draft Determination regarding the 2022-2024 C&LM Plan, April 27, 2022, *available at:*

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/a3a4beb1919a550d8525883e00 6e1a87?OpenDocument

⁹⁸ See Id.

⁹⁹ See Proposed 2022-2024 Conservation and Load Management Plan, November 1, 2022, available at: https://portal.ct.gov/-/media/DEEP/energy/ConserLoadMgmt/Final-2022-2024-Plan-to-EEB-1112021.pdf

¹⁰⁰ See Mass Save, Heating & Cooling Consultations, *available at*:<u>https://www.masssave.com/en/heatpump/hvac-specialist</u>

¹⁰¹ See Neighborhood Housing Services of New Haven, I Hear My Home CT, available at: <u>https://nhsofnewhaven.org/homeownership/i-heart-my-home-ct/</u>

¹⁰² See "Hartford's Office of Sustainability Launches Energy Coaching Service," January 10, 2022, available at: <u>https://www.hartfordct.gov/Government/Departments/Mayors-Office/Mayor-News/EnergyCoaching</u> ¹⁰³ See HeatSmart Connecticut, available at: https://heatsmartct.org/

Additionally, during the public process that informed DEEP's Determination on the 2022-2024 C&LM Plan, stakeholders noted that a concierge service would be an especially impactful resource for low-income communities.¹⁰⁴ Low-income customers face additional barriers to receiving weatherization services; however, weatherizing and retrofitting low-income households presents an outsize opportunity to advance the Plan's three priorities: equity, decarbonization, and affordability.¹⁰⁵

These initiatives signal the need for the Utilities to draw more fully upon growing experience and expertise within the region. As further described in the Conditions of Approval, DEEP is directing the Utilities to move forward with the concierge service described in the proposed 2022-2-24 C&LM Plan while simultaneously exploring an enhanced concierge service that includes the elements and best practices described above. During the development process, the Utilities are directed to solicit information and feedback from administrators of other successful energy concierge services, and to work with DEEP to identify any competitive IIJA programs through which funding may be available for an enhanced program.

j. Steps to Weatherize 80 Percent of Residential Units by 2030

Pursuant to Conn. Gen. Stat. § 16-245m(d)(1), in order to be approved by DEEP, a C&LM Plan must "include steps that would be needed to achieve the goal of weatherization of eighty per cent of the state's residential units by 2030." The residential portfolio of the 2022-2024 C&LM Plan includes strategies to increase performance for a variety of home types. The definition of "weatherization" is left at DEEP's discretion and has not yet been formally established to assess progress towards the statutory weatherization goal.

Over the past several years, DEEP and the EEB have conducted studies and public processes to inform this definition. On August 22, 2012, the EEB submitted to DEEP a recommended weatherization standard for single-family homes. On November 7, 2012 DEEP issued its Draft Determination to Establish Weatherization Standards for Single-Family Dwellings in Connecticut Pursuant to Section 33 of Public Act 11-80 (Draft Determination).¹⁰⁶ Following issuance of the Draft Determination, DEEP conducted a Technical Meeting on December 7, 2012, where stakeholders were provided the opportunity to comment on the Draft Determination. Following the Technical Meeting, ten stakeholders submitted written comments to DEEP.¹⁰⁷

These comments called on DEEP to wait for the completion of the Baseline Weatherization Study conducted by the EEB before issuing a final decision on the weatherization standard. Consequently, DEEP determined that it would re-engage stakeholders on the weatherization standard when the Baseline

¹⁰⁴ See for instance, comments from Kathy Fay and Acadia Center in response to DEEP's Draft Determination regarding the 2022-2024 C&LM Plan, April 27, 2022, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/\$EnergyView?OpenForm&Start=1&Count=30&Expand=7.1&Seq=2 ¹⁰⁵ See comments from Acadia Center in response to DEEP's Draft Determination regarding the 2022-2024 C&LM Plan, April 27, 2022, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/036f5cb6b6a05b3c8525883300 33182c?OpenDocument

¹⁰⁶ See DEEP Draft Determination to Establish Weatherization Standards for Single-Family Dwellings in Pursuant to Section 33 of Public Act 11-80, November 7, 2012, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/c4b6caf0059e937f85257aaf005 faea2/\$FILE/Draft%20Determination%20for%20Weatherization%20Standards%20for%20Singlefamily%20Dwellings.pdf

¹⁰⁷ See DEEP Request for Additional Comments: Weatherization Standard for Single-Family Dwellings, June 28, 2013, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/4bae084458559b7c85257b010 04b7b09/\$FILE/1-28-13%20Notice%20of%20Request%20for%20Additional%20Comments.pdf

Study was finalized. The Baseline Study submitted to the Energy Efficiency Board on June 3, 2014.¹⁰⁸ On August 13, 2015, DEEP conducted another Public Information Meeting to review its weatherization definition.¹⁰⁹ The previously proposed standard was not adopted for a variety of reasons, including the high technical cost of assessing homes using this standard at a large scale and its inability to account for multifamily properties, which comprise approximately 36 percent of Connecticut's housing stock.

Public comments received as part of DEEP's review of the Proposed C&LM Plan indicated a need for developing a standard of weatherization that would allow DEEP to assess progress towards its statutory goals.¹¹⁰ In response to these comments, and in furtherance of its statutory goals, DEEP included a draft weatherization standard as Attachment D to its Draft Determination issued on April 12, 2022 and requested feedback on the draft weatherization standard through the associated public comment process.¹¹¹ Stakeholders provided significant feedback on the draft standard.¹¹² Additionally, the Single-Family Weatherization Assessment and Update evaluation study that is currently being prepared for the Energy Efficiency Board will assess the draft standard for veracity and feasibility.¹¹³ As a result, DEEP will issue a final determination on the weatherization standard separately from this Determination at a later date. The draft standard that was issued with DEEP's Draft Determination on the 2022-2024 C&LM Plan is included with this Determination as Attachment D.

Additionally, during the 2022-2024 C&LM Plan term, DEEP is launching a Weatherization Barriers Remediation Program to increase the number of low-income households that are able to receive weatherization services. After a thorough stakeholder process, DEEP issued a Request for Proposals in November 2021 for a Program Operator(s) that will leverage funding from the Low-Income Heating Assistance Program (LIHEAP) and American Rescue Plan Act (ARPA) to remediate health and safety barriers, such as mold and asbestos, that currently prevent an estimated 23 percent of income-eligible homes from receiving weatherization services.¹¹⁴

k. Leveraging Demand Flexibility and Pay-for-Performance

¹⁰⁸ See Single-family Weatherization Baseline Assessment (R5) Final Report, June 3, 2014, *available at*: <u>https://energizect.com/sites/default/files/R5-Connecticut%20Weatherization%20Baseline%20Assessment-FINAL%2006-04-14.pdf</u>

¹⁰⁹ See DEEP Notice of Public Information Meeting and Opportunity for Comments, Definition of "Weatherization" For Single-Family Residential Units, August 4, 2015, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/67db1b8d9400b48a85257e970 0531584/\$FILE/Notice%20of%20Info%20Mtg%20081315%20%20Comments%20re%20Weatherization%20Defini tion%20%20FINAL.pdf

¹¹⁰ Public comments from DEEP's review of the 2022-2024 Conservation and Load Management Plan can be found here:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/\$EnergyView?OpenForm&Start=1&Count=30&Expand=7.1&Seq=2 ¹¹¹ See DEEP Draft Determination Attachment D – Draft Weatherization Standard, April 12, 2022, available at: http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/ea714e056ac86bef8525882200 71e982?OpenDocument and DEEP Notice of Public Meeting and Request for Written Comments, April 12, 2022, available at:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/41b827a51e5366bb852588220 0719b45?OpenDocument

¹¹² See for instance, comments submitted by People's Action for Clean Energy; CT Green Building Council; Aeroseal, LLC; Dandelion Energy; Michael Uhl; Kathy Fay; Save the Sound; Northeast Energy Efficiency Partnerships; Sierra Club Connecticut; CMC Energy Services, New England Smart Energy, and HE Energy Services; Connecticut Energy Efficiency Board; and Eversource Energy and Avangrid Networks, Inc., *available at*: <u>http://www.dpuc.state.ct.us/DEEPEnergy.nsf/\$EnergyView?OpenForm&Start=1&Count=30&Expand=7.1&Seq=7</u> ¹¹³ See Connecticut R2029 Single-Family Weatherization Assessment and Update Kickoff Presentation, March 17, 2022, *available at*: http://energizect.com/sites/default/files/2022-

^{05/}CT_R2029_SF_Wx_Assessment_KO_Slides_20220317_Out70x2a.pdf

¹¹⁴ See Request for Proposals for Statewide Weatherization Barrier Remediation Program Operator, November 10, 2021, available at: <u>https://portal.ct.gov/-/media/DEEP/energy/ConserLoadMgmt/FINAL-DEEP-WxBRP-RFP.pdf</u>

Active demand response offerings and performance-based incentives drive further innovation in the energy efficiency space by providing more granular customer usage insights and deeper energy savings. The proposed 2022-2024 C&LM Plan continues existing active demand response offerings and includes pilot programs for pay-for-performance incentive structures in the residential sector and natural gas demand response.

Avangrid's natural gas demand response pilot programs provide an opportunity to address natural gas reliability concerns, alleviate temporary physical pipeline constraints in low pressure areas, negate the need for additional pipeline capacity, and reduce emissions from gas use.¹¹⁵ Avangrid provided status updates and initial results from these pilot programs in their response to DEEP's November 2021 Requests for Information.¹¹⁶ While these programs are in the pilot phase, DEEP will be closely following program results and will explore the possibility of expanding the pilot offerings to additional customers or service territories.

The more-established portfolio of electric active demand response offerings in the proposed 2022-2024 C&LM Plan provide additional reliability, savings, and emissions reduction benefits. Given the focus on electrification described in the sections above, active demand response in the electric sector will be a critical strategy for meeting future demand. DEEP will work with the Utilities, EEB, and other stakeholders to further promote the co-delivery of active demand response and other energy efficiency measures throughout this Plan term.

In addition to the active demand response offerings, the Utilities are also launching a HES pay-forperformance pilot during the 2022-2024 Plan term that aims to determine the persistence of energy savings resulting from comprehensive, multi-measure projects and customer behavioral changes. Incentives through this pilot will be contingent open energy savings realized through customer energy bills.¹¹⁷ In the development of this pilot, the Utilities consulted with entities including BlocPower and Sealed. They also utilized the evaluation, measurement, and verification capabilities of Recurve and are exploring additional pay-for-performance opportunities that this data can provide.¹¹⁸ As further described in the Conditions of Approval, DEEP is directing the Utilities to move forward with the residential payfor-performance pilot as proposed. DEEP will continue to work with the Utilities to explore additional pay-for-performance opportunities.

DEEP recognizes that the deployment of some of these strategies is hindered by a dearth of advanced metering infrastructure (AMI), particularly in Eversource territory.¹¹⁹ When there is more widespread deployment of AMI in Connecticut, DEEP expects that the Utilities will develop comprehensive demand

¹¹⁵ For a summary of Avangrid gas demand response pilot programs and objectives, *See* Utility presentation at the November 18 C&LM Technical Meeting, 2022-2024 Energy Efficiency Plan Demand Management Programs, November 18, 2021, *available at*: <u>https://portal.ct.gov/-/media/DEEP/energy/ConserLoadMgmt/20211118-CLM-Technical-Meeting.pdf</u>

¹¹⁶ See Avangrid responses to DEEP Requests for Information #BETP18-24, November 18, 2021, available at: http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/6cadc3d093b715be852587920 0580264?OpenDocument

¹¹⁷ See Utilities' response to DEEP Condition of Approval #3 of the 2021 C&LM Plan Update, March 25, 2021, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/8525797c00471adb852586a30 069e413?OpenDocument

¹¹⁸ See Utilities' response to DEEP Request for Information #BETP-15, November 18, 2021, available at: <u>http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/6cadc3d093b715be852587920</u> 0580264?OpenDocument

¹¹⁹ See Utilities' response to DEEP Request for Information #BETP-04, November 12, 2021, available at: <u>http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/7091cd5a6937ac52852587900</u> 05a4d16?OpenDocument

management and performance-based programs that take full advantage of the usage insights that AMI provides.¹²⁰ PURA is currently considering utility proposals for AMI deployment in one of its grid modernization dockets. On February 28, 2022, DEEP filed comments in that docket noting federal funding opportunities available to the Utilities for AMI through IIJA and encouraging PURA to "consider how to fairly motivate the utilities to make good faith effort(s) to access and leverage external dollars that will reduce the burden on ratepayers."¹²¹

Given the prospect of increased AMI deployment statewide, the C&LM Plan must be prepared to fully leverage the capabilities of AMI in its active demand response and pay-for-performance programs. To develop a record of potential opportunities and stakeholder feedback, DEEP will hold a Technical Meeting, in conjunction with its ongoing CES development process, to explore new opportunities for active demand response offerings and pay-for-performance incentive structures. This Technical Meeting may include presentations from experts in fields such as demand flexibility and aggregation and virtual power plants. Following this Technical Meeting, DEEP will work with the Utilities and EEB to incorporate findings into forthcoming annual C&LM Plan Updates.

1. Developing Connecticut's Energy Efficiency Workforce

The energy efficiency industry, which includes high efficiency and traditional HVAC, renewable heating and cooling, efficient lighting, advanced materials, construction, and other sectors is the largest source of clean energy jobs in Connecticut, accounting for over 35,000 jobs in 2020.¹²² However, during the public input process, stakeholders noted a need for enhanced workforce training and educational opportunities, particularly in sectors, like HVAC, where the programs may see increased participation over the next Plan term.¹²³ This feedback has been further supported by multiple surveys of contractors, installers and manufacturers both before and during the COVID-19 pandemic.¹²⁴

Recognizing the importance of Connecticut's clean energy workforce, Governor Lamont's Executive Order No. 21-3 established the Connecticut Clean Economy Council (CCEC) to "advise on strategies and policies to strengthen our climate mitigation, clean energy resilience, and sustainability programs, thereby lowering emissions and advancing the state of economic and environmental justice for our residents."¹²⁵ The CCEC, currently in the formative stage of development, will be a critical resource and source of guidance for strategies to develop and sustain the energy efficiency workforce. Additionally, in 2021 the Office of Workforce Strategy announced its new CareerConneCT grant program to support innovative

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/\$EnergyView?OpenForm&Start=1&Count=30&Expand=7.1&Seq=2 ¹²⁴ See CBIA and Connecticut DEEP, 2017 Survey of Energy and Energy Efficiency Workforce Needs, 2017, pg 4, available at: https://portal.ct.gov/-/media/DEEP/energy/efficiency/CTEnergyWorkforceSurvey17pdf.pdf. and Energize Connecticut, Connecticut Clean Energy Industry Report, September 2021, pg 25, available at: https://www.ctgreenbank.com/wp-content/uploads/2022/01/2021-CT-Clean-Energy-Industry-Report.pdf. ¹²⁵ See Governor Lamont Executive Order No. 21-3, December 16, 2021, available at: https://portal.ct.gov/-/media/Office-of-the-Governor/Executive-Orders/Lamont-Executive-Orders/Executive-Order-No-21-3.pdf

¹²⁰ For examples of potential opportunities to leverage AMI, *See* Eversource Advance Metering Infrastructure Business Case and Implementation Plan, July 31, 2020 (refiled October 19, 2020), *available at*: <u>http://www.dpuc.state.ct.us/2nddockcurr.nsf/8e6fc37a54110e3e852576190052b64d/c28735f2972646b78525875200</u> 79905d?OpenDocument

¹²¹ See DEEP comments, Docket 17-12-03RE02, PURA Investigation into Distributed System Planning – AMI, February 28, 2022, available at:

http://www.dpuc.state.ct.us/2nddockcurr.nsf/8e6fc37a54110e3e852576190052b64d/e655412de4dcf619852587f700 62dbf8?OpenDocument

¹²² See Energize CT, Connecticut Clean Energy Industry Report, September 2021, available at: https://www.ctgreenbank.com/wp-content/uploads/2022/01/2021-CT-Clean-Energy-Industry-Report.pdf

¹²³ See for instance, written comments submitted to DEEP by the CT Green Building Council, Dandelion Energy, People's Action for Clean Energy (PACE), Northeast Energy Efficiency Partnerships (NEEP), Save the Sound, and Henry Auer, *available at*:

workforce development programs in high-demand industries such as energy efficiency.¹²⁶ This funding opportunity was shared with energy efficiency stakeholders at DEEP's November 18, 2021 Technical meeting.¹²⁷

The proposed 2022-2024 C&LM Plan outlines the Utilities' workforce development and educational offerings, including programs such as GreenSTEP and ee Smarts. These programs have had thousands of students participate since their inception but, as ratepayer funded education programs, they should be regularly reviewed to ensure they are tailored to, and reaching, all of Connecticut's students. In 2021, an evaluation of these programs conducted through the EEB's Evaluation Committee found that the programs would be improved with better-defined goals and metrics.¹²⁸ More specifically, the study identified that the eeSmarts program measured quantitative data like number of assemblies, number of students in attendance, and teachers at professional development workshops, but not how the content is absorbed and applied by students and teachers, i.e. whether their families are more likely to participate in Home Energy Solutions program or whether the program outcomes are influenced by geographic, demographic or socioeconomic indicators, To that end, the Conditions of Approval include directives for the Utilities to develop and regularly report on metrics to track progress in supporting and expanding educational programs that will contribute to a robust energy efficiency workforce in Connecticut.

Additionally, a highly skilled and trained workforce is necessary to expand customer choice and access to renewable thermal technologies like heat pumps. Positive customer experiences with these technologies will be essential for their continued deployment, and comprehensive installer training will support these outcomes. Course offerings should align with "best-in-class" trainings available elsewhere in the region and be made available to Connecticut contractors. In 2022, the Utilities plan to launch a Heat Pump Installer Network (HPIN) of independent, licensed, and insured HVAC contractors that demonstrate the requisite training for successful heat pump installation.¹²⁹ This approach was taken in Massachusetts, and provides heat pump installers with training resources, rebates and financing options; sales tools; listing on the Mass Save website; and other resources.¹³⁰ DEEP supports the development of the HPIN as a tool to aid in quality control and contractor education during the early stages of heat pump market development, and has directed the Utilities to provide additional information and regular reporting on the HPIN and associated heat pump technology trainings in the Conditions of Approval.

The Utilities and DEEP have also been working jointly to identify opportunities for federal funding to be used in advancing other workforce programs, trainings, and wrap around services. DEEP will continue to act as a partner and advisor with respect to the multiple, concurrent strategies and funding sources that now exist to further develop Connecticut's energy efficiency educational programming and workforce.

¹²⁶ See Office of Workforce Strategy, CareerConneCT, available at:

https://portal.ct.gov/careerconnect?language=en_US

¹²⁷ See Office of Workforce Strategy presentation on CareerConneCT (slide 72), November 18, 2021, available at: <u>https://portal.ct.gov/-/media/DEEP/energy/ConserLoadMgmt/20211118-CLM-Technical-Meeting.pdf</u> recording available at:

https://ctdeep.zoom.us/rec/play/ltR67izDoCJUBryEr3nMcHRMdxjziWuD38V0wh_DkV36dc74Ymf14a4DlNkH7a k8ZbAumtnrLLKciVdu.wWBY0dm24hAbl-6J?continueMode=true& x zm rtaid=fSaU-

jGkSa6eJMonBwzGSg.1637337535995.253bf4dffb9e6955e3ed0b3b70047c0c& x zm rhtaid=623

¹²⁸ See Illume, Energize Connecticut Evaluation of Educate the Workforce, Educate the Students, Educate the Public, and Customer Engagement Initiatives, presented to the EEB Evaluation Committee December 10, 2021, *available at:* <u>https://energizect.com/sites/default/files/2022-</u>02/X2022 Presentation%20Slides%20 10Dec2021 v3x2.pdf

¹²⁹ See Utilities' response to DEEP RFI #BETP-29, December 9, 2021, available at:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/8c9906744d48f122852587ab00 524a63?OpenDocument

¹³⁰ See Mass Save, Heat Pump Installer Network, available at: <u>https://www.masssave.com/learn/partners/heat-pump-installer-network</u>

DEEP's role on the CCEC will promote coordination between these various efforts to maximize the benefits and outcomes. In order to do this, it is essential that DEEP has the opportunity to review and approve the development of workforce programs offered through the C&LM programs, and that DEEP is provided with reporting on the results of such programs. This will allow DEEP to better assist the Utilities and other energy efficiency stakeholders in identifying opportunities to coordinate with these statewide workforce development initiatives. The Conditions of Approval will allow DEEP to track progress and help identify gaps that can be met with other resources.

m. Evaluation Data

DEEP's Approval with Conditions of the 2021 C&LM Plan Update noted significant delays in conducting evaluation studies due to contracting and data availability and directed the Utilities to develop a secondary metric to their Performance Management Incentive (PMI) to encourage improvement.¹³¹ Throughout 2021 the EEB Evaluation Administrator has worked with the Evaluation Committee and the Utilities to develop an assessment mechanism to inform this secondary metric.

However, testimony from the Evaluation Administrator provided at DEEP's November 16, 2021 Technical Meeting indicates that data provisioning for evaluation studies is an ongoing challenge with significant impacts on study timelines and costs.¹³² A report from the Evaluation Administrator found that Eversource and UI met due dates and provided adequate data about 69% and 46% of the time, respectively, for studies conducted in 2021. This report estimated that the cost of data delays and inadequacies exceeded \$300,000 in 2021, averaging about 5.5 percent of the cost of projects reviewed. According to the Evaluation Administrator, this cost is, "equivalent to the loss of 1-2 Evaluation projects per year, and provides a strong incentive for contractors to overcharge what they would bid for similar work in other states – and higher than the cost difference, because of the unknown level of risk by project."

The Utilities are statutorily mandated to "communicate with the Evaluation Administrator for purposes of data collection . . . and providing necessary factual information during the course of evaluations."¹³³ Calculation of the Utilities' performance incentives is based on program savings, and DEEP will hold the Utilities responsible for facilitating the production of data needed to calculate and verify those claimed savings. Any failure by the Utilities to provide timely and responsive data or information necessary for evaluations impairs the evaluation process and slows down the program savings verification process. Inefficiencies with data collection, along with any other problems identified in the evaluation process, will be reflected in the Utilities' evaluations.¹³⁴ Additionally, the Evaluation Administrator has identified other elements that would improve evaluation study data. DEEP, in the Conditions of Approval, directs the Utilities to take certain steps in the 2022-2024 C&LM Plan term to improve the timeliness and quality of data provided for evaluation studies.

n. Performance Management Incentives

¹³² A recording of the November 16, 2021 Technical Meeting is available at: <u>https://ctdeep.zoom.us/rec/play/WCdG84uKS96n2x-</u>

¹³¹ See DEEP Determination: Approval with Conditions of the 2021 Plan Update to the 2019-2021 Conservation and Load Management Plan, March 4, 2021, *available at*: <u>https://portal.ct.gov/DEEP/Energy/Conservation-and-Load-Management/Conservation-and-Load-Management</u>

gjZuROpKeb3euh4VdU4s1Pfe3MXz9x9oQNaLgyTgToHI_RDzAMfLdesaiLbZWS8ww.lUvzeQ55sAxsCmcG?co_ntinueMode=true

¹³³ Conn. Gen. Stat. § 16-245m(d)(4).

¹³⁴ See Id. ("All evaluations shall contain a description of any problems encountered in the process of evaluation, including, but not limited to, data collection issues").

In their review of the 2022-2024 C&LM Plan, the EEB directed the Utilities to consider an alternative Performance Management Incentive (PMI) structure.¹³⁵ These changes included:

- i. Including all fuel savings in primary metrics, consistent with program goals and the current Modified Utility Cost Test and Total Resource Cost Test used to assess program cost-effectiveness.
- ii. The addition of demand response metrics that target demand reduction efforts.
- iii. Revised sector-specific allocations for electric and gas based on average C&LM Plan benefits as assessed by the Modified Utility Cost Test.
- iv. Revised payout v. performance percentages and goals.

Items (i)-(iii) have been incorporated into the PMI structure included in the 2022-2024 C&LM Plan and DEEP finds that the inclusion of these items in the PMI structure is reasonable.

With respect to item (iv) above, the EEB Technical Consultants and Utilities proposed different variations to the payout v. performance structure. In July 2021, the Utilities and Consultants presented two proposed payout v. performance structures to the EEB. These proposals and the structure that was employed during the 2019-2021 C&LM Plan term are outlined in the table below.¹³⁶ On October 13, 2021, the EEB voted to approve a PMI structure that matched the Technical Consultants' proposal.¹³⁷ The proposed 2022-2024 C&LM Plan filed with DEEP on November 1, 2021 contains this PMI structure.

From 2019-2021, Eversource reached an average of 115.81 percent achievement on its metrics with an average 5.6 percent payout across metrics. UI reached an average of 104.03 percent achievement on its metrics with an average 4.05 percent payout across metrics. ¹³⁸

DEEP has developed an alternative PMI structure that brings Connecticut into closer alignment with regional practices while maintaining incentives for high performance.¹³⁹ This structure contains elements of the proposed structures described in the table below. Under this structure, the Utilities will need to achieve higher levels of performance to receive the same payout that they would have achieved through the current structure. Additionally, while the 2019-2021 structure has a maximum payout of 8 percent, this proposed structure has a maximum payout of 7 percent.

¹³⁷ A record of the vote on this matter and accompanying discussion by EEB members can be found in the minutes from the October 13, 2021 EEB meeting, *available at*: <u>https://energizect.com/sites/default/files/2021-12/Oct%202021%20EEB%20Minutes_f.pdf</u>

¹³⁸ See Eversource 2020 Annual Filings, March 1, 2021, available at:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/8525797c00471adb852586870 0526385?OpenDocument, and UI/CNG/SCH Annual Filings, February 26, 2021, available at:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/8525797c00471adb852586880 0725477?OpenDocument, and YGS 2019 Annual Filing, February 28, 2020, *available at:*

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/8525797c00471adb8525851c0 0502e09?OpenDocument, and Eversource 2019 Annual Filing, February 28, 2020, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/8525797c00471adb8525851c0 050043d?OpenDocument, and UI/CNG/SCG 2019 Annual Filings, March 3, 2020, *available at*:

http://www.dpuc.state.ct.us/DEEPEnergy.nsf/c6c6d525f7cdd1168525797d0047c5bf/8525797c00471adb852585200 049b8d6?OpenDocument

¹³⁵ See EEB Technical Consultant Presentation, Proposed Revisions to the 2022-2024 Performance Management Incentives (PMIs), June 9, 2021, *available at*:

https://app.box.com/s/iytqdw372f6bxxyllgrx9mjnn5g8fbc5/file/820206797419

¹³⁶ Proposals from both parties can be found in July 14, 2021 presentations to the EEB. Technical Consultant presentation *available at*: <u>https://app.box.com/s/qeafcjvp2n6k8072bn6oc3vbzk3enz57/file/833748999480</u>. Utilities' presentation *available at*: <u>https://app.box.com/s/qeafcjvp2n6k8072bn6oc3vbzk3enz57/file/83370169646</u>.

¹³⁹ For a comparison of regional PMI structures *see* EEB Technical Consultant Presentation on Performance Management Incentives (PMI), delivered November 16, 2021, *available at*: <u>https://portal.ct.gov/-/media/DEEP/energy/ConserLoadMgmt/20211116-CLM-Technical-Meeting.pdf</u>

Deufeuree	Payout					
Level	2019-2021 Structure	2022-2024 Structure ¹⁴⁰	Utility Proposal	DEEP Proposal		
75%	2.0%	2.50%	2.50%	2.50%		
80%	2.5%	3.00%	3%	3.00%		
85%	3.0%	3.50%	3.50%	3.50%		
90%	3.5%	4.00%	4%	4.00%		
95%	4.0%	4.50%	4.50%	4.50%		
100%	4.5%	5.00%	5.00%	5.00%		
105%	5.0%	5.50%	5.50%	5.29%		
110%	5.5%	6.00%	6%	5.57%		
115%	6.0%	6.50%	6.50%	5.86%		
120%	6.5%		6.75%	6.14%		
125%	7.0%		7.00%	6.43%		
130%	7.5%		7.25%	6.71%		
135%	8.0%		7.50%	7.00%		

o. Previous Orders and Conditions are rescinded, some are updated into new Conditions of Approval

All Compliance Orders and Conditions of Approval from previous DEEP Determinations on C&LM Plans are rescinded. DEEP has reviewed all previous Compliance Orders and Conditions of Approval and those still applicable are updated into Conditions of Approval for this Determination.

p. Areas of further refinement and Conditions of Approval

DEEP has determined that some further refinement of the 2022-2024 C&LM Plan is necessary to achieve stated objectives of the Plan as well as state policy priorities. Consequently, DEEP has included a Schedule of Compliance for Conditions of Approval, including required modifications and requirements to submit reports on certain topics, as detailed in Section IV, below.

IV. <u>Conditions of Approval</u>

DEEP's approval of the 2022-2024 C&LM Plan is subject to the 2022-2024 Schedule of Conditions of Approval included as Attachment A. This Schedule includes all Conditions of Approval with which the Utilities must comply during the 2022-2024 C&LM Plan term. Conditions of Approval from previous DEEP determinations that are still applicable have been carried over to the 2022-2024 Schedule of Conditions of Approval and in some cases. In some cases, these Conditions have been modified and assigned new numbers. As stated above, all prior Conditions of Approval that have not been included in the 2022-2024 Schedule are rescinded.

V. <u>Conclusion</u>

The 2022-2024 C&LM Plan has the potential to advance equity, decarbonization, and affordability through Connecticut's energy efficiency programs. This three-year term presents an opportunity to

¹⁴⁰ This is the PMI structure included in the proposed 2022-2024 C&LM Plan that was approved by the Energy Efficiency Board.

position the C&LM Plan to be responsive to existing and new state policies and plans, including the upcoming CES, ensuring that the energy efficiency and demand management programs can continue providing significant benefits to Connecticut businesses and residents.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-002C Page 1 of 23

NEW HAMPSHIRE ENERGY EFFICIENCY CALCULATION OF PERFORMANCE INCENTIVE BEGINNING IN 2020

Report Issued by the NH Performance Incentive Working Group

> Docket No. DE 17-136 July 31, 2019

Table of Contents

Ι.	Introduc	Introduction						
	Α.	Scope of the PI Working Group	2	2				
	В.	Executive Summary	2	2				
	С.	Minimum Thresholds & Requirements	2	4				
11.	Review o	of Existing Performance Incentive Framework						
	Α.	Current Threshold Requirements	(6				
	В.	Electric Programs	6	6				
	С.	Natural Gas Programs	(6				
.	Opportu	nities for Improving the Performance Incentive Model						
	Α.	Narrow Focus of Existing Framework	7	7				
	В.	Limited Emphasis on Demand Reduction	8	8				
	С.	Incentive Eligibility Threshold	8	8				
	D.	Sector Level Incentive Eligibility	9	9				
	Ε.	Benefit Cost Ratio Component	<u>c</u>	9				
IV.	Revised	PI Framework						
	Α.	Current Framework Formula		10				
	В.	Revised Framework Formula	2	11				
V.	Income I	Eligible Customers						
	Α.	Review by the Working Group		12				
	В.	Funding	2	13				
VI.	Issues fo	r Future Consideration						
	Α.	Energy Optimization/Electrification	, -	13				
	В.	Revised Cost Effectiveness Tests	, -	14				
	С.	Gas Demand	, -	14				
	D.	Income Eligible Participation	-	15				
Appe	ndices							
	Α.	2020 PI Savings Calculation Templates (Electric and Gas)		17				
	В.	Members/Participants of the PI Working Group		19				
	С.	Consultants Assisting the PI Working Group	2	20				
	D.	Glossary of Terms Used in the LBR Template	-	21				

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-002C Page 3 of 23

I. Introduction

A. Scope and Members of the PI Working Group

The scope of the Performance Incentive Working Group's ("PI Working Group" or "Working Group") activities is defined by New Hampshire Public Utilities Commission ("Commission" or "PUC") Order Nos. 26,095 and 26,207 in Docket DE 17-136, which approved the Settlement Agreements filed on December 8, 2017 and December 13, 2018, respectively. The Settlement Agreements direct the PI Working Group to undertake a review of potential PI methodologies that could further promote the achievement of New Hampshire's EERS goals, with the objective of implementing any changes to the performance incentive calculation beginning in the 2020 program year. The PI Working Group was tasked with considering metrics designed to encourage income eligible participation in energy efficiency programs and to encourage peak load reductions. Per the Settlement Agreement, the intent of the PI Working Group is to make its recommendations in time to incorporate proposed methodologies into the 2020 New Hampshire Statewide Energy Efficiency Plan Update. This Report represents the PI Working Group's fulfilment of that assignment.

During its extensive 16-month review of the issues surrounding the current, and alternative, PI methodologies, the Working Group reviewed and produced many documents, some of which are posted to a page on the <u>Commission website</u> http://www.puc.state.nh.us/EESE%20Board/EERS WorkingGroups.html. These documents are posted for informational purposes only and the PI Working Group members do not necessarily adopt or endorse the information and findings contained in these documents.

This Report is largely a consensus document produced by the Working Group members. However, while this Report was guided by and results from the Settlement Agreements filed December 8, 2017 and December 13, 2018, it is not intended as, and should not be construed as a Settlement Agreement. As such, Working Group members reserve the opportunity to take consistent or contrary positions when PI is at issue in future proceedings before the Commission. The Report is a public document and may be used in future Commission proceedings. The Working Group meetings and related discussions that lead to the Report were not conducted as privileged or confidential sessions.

This Working Group Report, along with any member/stakeholder comments, has been posted to the <u>Commission website</u> under the PI Working Group section.

The members of the PI Working Group devoted many hours to meetings, research, information responses and preparation of slide presentations and this Report is the product of a collaborative effort enriched by the creative ideas each member brought to the table. A full list of members is included in Appendix B.

B. Executive Summary

The PI Working Group met in order to review the current, and alternative, PI calculation methodologies and to recommend an appropriate PI framework to be implemented for the 2020 period. The Working Group considered including potential metrics to encourage electric system peak load reductions and to

increase participation by low income groups and households in energy efficiency programs. The discussions of the PI Working Group occurred over a sixteen-month period between January 2018 and July 2019, and the salient documents from these discussions are posted to the <u>Commission website</u>.

A significant portion of the Working Group's time was spent studying and revising minimum PI thresholds, calculation methodologies, and developing a more comprehensive and transparent framework for calculating PI that constitutes a good replacement for the existing methodology. The new proposed framework is based on the following:

• Categorizing and weighting five separate performance indicators (components), at the portfolio level, each involving minimum savings thresholds (as well as other minimum thresholds summarized below) that must be met in order for any PI to be earned for that component.

PI #	Component	Description	Incentive	Minimum	Maximum	Verification
	Title		Weight	Threshold	PI Level	
1	Lifetime	Actual/Planned	35%	75%	125%	Annual PI
	KWN Savings	Lifetime kwn				Filing
		Savings	1001	====(1050(W/PUC
2	Annual kWh	Actual/Planned	10%	75%	125%	Annual PI
	Savings	Annual kWh				Filing
		Savings				w/PUC
3	Summer	Actual/Planned	12%	65%	125%	Annual PI
	Peak	ISO-NE				Filing
	Demand	System-wide				w/PUC
	Savings	Summer Peak				
		Passive kW				
		Savings				
4	Winter Peak	Actual/Planned	8%	65%	125%	Annual PI
	Demand	ISO-NE				Filing
	Savings	System-wide				w/PUC
		Winter Peak				
		Passive kW				
		Savings				
5	Value	Actual/Planned	35%	75%	125%	Annual PI
		Net Benefits ¹				Filing
						w/PUC
Total			100%			

Performance Incentive Components (Electric)

¹ Total resource benefits (See Appendix D) less utility costs (not including PI).

PI #	Component	Description	Incentive	Minimum	Maximum PI	Verification
	Title		Weight	Threshold	Level	
1	Lifetime	Actual/Planned	45%	75%	125%	Annual PI
	MMBtu	Lifetime				Filing
	Savings	MMBtu				w/PUC
		Savings				
2	Annual	Actual/Planned	20%	75%	125%	Annual PI
	MMBtu	Annual MMBtu				Filing
	Savings	Savings				w/PUC
3	Value	Actual/Planned	35%	75%	125%	Annual PI
		Net Benefits ²				Filing
						w/PUC
Total			100%			

Performance Incentive Components (Gas)

• The source data for the PI value of each performance indicator is taken from the Benefit-Cost model spreadsheets utilized by the utilities in the preparation of their annual PI filings showing calculations of program cost effectiveness and present value of benefits. Note: The reporting requirement and the compilation of this data on an annual basis will not change – only the calculation of PI has changed.

C. Minimum Thresholds and Requirements

- Most of the existing minimum PI requirements/parameters remain unchanged as follows:
 - ✓ Maintain existing target PI equal to 5.5 percent of each company's program spending with a maximum PI equal to 6.875 percent of actual spending.
 - ✓ Maintain actual spending as the basis of the calculation of PI, rather than the budget.
 - ✓ Maintain a minimum portfolio-wide threshold benefit-cost ratio ("BCR") of 1.0 before PI can be earned, but – remove the BCR from calculation of PI.³
 - ✓ Maintain the cap on incentives that can be earned equal to 125 percent of design PI, equivalent to 6.875 percent of actual spending.
 - Maintain existing use of "adjusted gross savings" for annual and lifetime savings calculations, exclusive of market effects (free ridership and spillover) and inclusive of applicable realization rates achieved by the programs as indicated by third party evaluations and adopted by the Evaluation Measurement and Verification ("EM&V") Working Group.
 - ✓ Maintain the minimum portfolio-wide threshold of 55% of lifetime energy savings from electric measures in the electric programs. As is the case currently, if this threshold is not

² Id.

³ The minimum threshold for cost-effectiveness in this PI framework will be based on the current Total Resource Cost test. The Benefit-Cost and EM&V Working Group are currently evaluating the B/C test used by the New Hampshire energy efficiency programs. A final report is expected to be completed by September of 2019. The PI Working Group members did not address in depth as to whether future PI calculations will reflect any changes to the B/C screening test from that review.

met, then a lower coefficient (4.4 percent rather than 5.5 percent) is to be used in the calculation of PI, along with a corresponding cap of 5.5 percent.

- The following PI requirements/parameters were revised or discontinued:
 - ✓ The existing practice of calculating PI based on achievements at the sector level (i.e. Residential/Income Eligible and Commercial/Industrial sectors) will be replaced by a calculation based on achievement at the portfolio level as a whole (i.e. combination of both sectors).
 - ✓ The existing minimum threshold of 65 percent of planned lifetime savings, which must be met before any PI is earned for that component, will be increased to 75 percent for each of the lifetime and annual savings components as well as the net benefits component. For the new PI components associated with passive electric summer and winter peak demand, the minimum threshold will be 65 percent (see table above).

The Working Group supports the revised PI framework for the following reasons:

- It uses metrics that are <u>transparent</u> e.g., performance is incentivized within separate key metric areas that are clear and well-defined, and aligned with EERS goals.
- It is <u>administratively expedient</u> e.g., provides an easy to use one-page template based on the existing data compilation methods used by the utilities.
- It increases <u>focus</u> on targets and promotes various policy objectives by applying incentives to each performance component separately e.g., peak demand.
- It establishes minimum thresholds for <u>each performance indicator</u> to encourage performance on each of the targets.
- It preserves <u>effective elements</u> of the existing minimum PI requirements as outlined above e.g., baseline target and cap, BCR, actual savings, etc.
- It uses a <u>portfolio approach</u>, which provides the utilities with greater flexibility in terms of program implementation and innovation, and increasing low income participation through fuel-neutral measures.

II. Review of Existing Performance Incentive Framework

The current energy efficiency program administration performance incentive framework was initially proposed by the Energy Efficiency Working Group in its final report to the Commission on July 6, 1999,⁴ and approved by the Commission in November 2000.⁵ Aside from Commission modifications to the framework in September 2013,⁶ and again when it approved the Energy Efficiency Resource Standard in 2016,⁷ the framework developed nearly two decades ago remains the foundation of New Hampshire's energy efficiency program administration performance incentive framework today.

⁴ Docket No. DE 96-150. Energy Efficiency Working Group Final Report. (July 1999) Page 21. Available at: <u>https://www.puc.nh.gov/Electric/96-</u>

^{150%20%20}NH%20Energy%20Efficiency%20Working%20Group%20Final%20Report%20(1999).pdf

⁵ Order No. 23,574 at 19. See also, Order No. 23,982 at 13.

⁶ Order No. 25,569 at 7. The Commission added the tiered incentive described *infra* at note 7 as a means of balancing the Commission's recently approved fuel neutral programs.

⁷ Order No. 25,932 at 60. The modification was to the size the of the performance incentive

A. Current Threshold Requirements

To be eligible for a performance incentive for a specific sector (Residential/income-eligible programs, and Commercial/Industrial, inclusive of the Municipal program for electric programs), the gas or electric utility currently must achieve the following:

- 1. A BCR of greater than 1.0 in that sector for the electric utilities and gas utilities or not receive PI for the BCR portion.
- 2. Actual lifetime kWh savings at or above 65 percent of the planned savings in that sector for the electric utilities or no PI is earned for the kWh savings portion.
- 3. Actual lifetime MMBtu savings at or above 65 percent of the planned savings in that sector for the gas utilities or no PI is earned for the MMBtu savings portion.

B. Electric Programs

Once the above-mentioned threshold requirements have been satisfied, the current performance incentive for the electric energy efficiency programs is calculated on a sector specific basis, and based on the following factors:

- If actual electric lifetime savings (for both electric and non-electric measures) are greater than or equal to 55 percent of total lifetime energy savings, the multiplier for the savings component is 2.75 percent of sector spending; if it is less than 55 percent then the multiplier is 2.2 percent of sector spending⁸
- 2. The actual dollars spent (by the utility and by customers) to carry out programs;
- 3. The actual BCR compared to the planned BCR;
- 4. The actual lifetime electric energy (kWh) savings compared to the planned lifetime electric energy (kWh) savings;
- 5. The BCR component and the kWh savings ratio component are each capped at 3.4375 percent for each sector and each sector PI is capped at 6.875 percent; and
- 6. Actual spending amounts for the PI calculation may exceed the total budget by up to 5 percent.

The current performance incentive formula ties these factors together is as follows for each sector:

(1) (2) (3) (4) PI= [(2.75% or 2.2%) x Actual Spend] x [(BCR Actual/BCR Planned) + (lifetime kWh Actual/lifetime kWh Planned)]

C. Natural Gas Programs

The performance incentive framework for the natural gas programs is similar to the electric programs, except that it uses MMBtu savings from natural gas instead of lifetime kWh and the incentive percentage and total PI cap is not dependent on achieving a minimum portion of total energy savings from gas measures.

⁸ If at least 55 percent of the overall energy savings are in the form of electric energy, then the utility earns PI using the higher 5.5 percent (i.e. 2.75 percent for the savings component and 2.75 percent for the benefit-cost component). If less than 55percent of the overall savings are from electric energy, then the utility earns PI using the lower 4.4 percent multiplier (i.e. 2.2 percent for the savings component and 2.2 percent for the benefit-cost component). The 55% electric savings threshold also determines the overall performance incentive cap; if the 55% threshold is reached, the maximum PI is 6.875% of actual expenditures, otherwise it is 5.5% of actual expenditures. This is meant to focus the majority of the SBC-funded budget towards electric savings rather than gas and other fossil fuel savings.

The current performance incentive formula for the natural gas programs is as follows for each sector:

(1) (2) (3) PI= [2.75% x Actual Spend] x [(BCR Actual/BCR Planned) + (lifetime MMBtu Actual / lifetime MMBtu Planned)]

III. Opportunities for Improving the Performance Incentive Model

The PI Working Group stakeholders identified several aspects of the current model which could be improved to reflect the State of New Hampshire's priorities, and account for changes that have taken place in our energy systems in the two decades since the framework was originally adopted.

The opportunities for improvement were focused on the following aspects of the existing framework: (1) a narrow focus on lifetime savings and BCR; (2) a limited emphasis on the value of electric peak demand reduction; (3) a threshold for incentive eligibility that begins at 65 percent of lifetime savings goals; (4) a threshold for incentive eligibility at the sector level rather than portfolio level; and (5) a focus on the ratio of benefits to costs rather than on net benefits.

A. Narrow Focus on Lifetime Savings and BCR

The existing performance incentive framework's narrow focus on BCR and lifetime kWh savings excludes other performance metrics or outcomes stakeholders believe the utilities should target based on the policies of the State of New Hampshire and priorities of the Commission. The American Council for an Energy Efficient Economy (ACEEE) suggests, "Multifactor performance incentives that incorporate multiple metrics can also work to meet other policy objectives... like reducing peak demand (and system costs), creating savings for low-income customers, and others."⁹ Several jurisdictions, such as Vermont, utilize a framework based on several quantifiable performance indicators (QPIs).

While the working group acknowledged the importance of utility performance as it relates to lifetime energy savings, as well as maximizing the overall benefits and minimizing the overall costs of the programs, it also reached consensus that other performance indicators merited attention in the framework.¹⁰

⁹ American Council for an Energy Efficient Economy (ACEEE). Topic Brief: Snapshot of Energy Efficiency Performance Incentives for Electric Utilities. (December 2018) Page 3. Available at: https://aceee.org/sites/default/files/pims-121118.pdf

¹⁰ In addition to reviewing the Vermont QPI framework, the Working Group also reviewed Massachusetts' PI framework, which focuses on the gross and net dollar benefits delivered by energy efficiency programs. After including seven program metrics in its PI formula for several years, the Massachusetts Department of Public Utilities subsequently excluded these metrics stating "performance metrics should induce Program Administrators to undertake activities they would not otherwise undertake" Massachusetts DPU Order 13-67 (December 11, 2014), page 10. Available at https://fileservice.eea.comacloud.net/FileService.Api/file/FileRoom/9230369

B. Limited Emphasis on Peak Demand Reduction

The existing performance incentive framework accounts for the benefits associated with electric peak demand reduction indirectly within that framework's benefit cost component. This contrasts with several states in the region that have recently placed a greater emphasis on the value of demand reduction by including a specific incentive associated with the achievement of planned demand reduction goals.¹¹ The group also notes that the New Hampshire PUC asked the utilities to explore and pursue peak reduction in several recent dockets as a means to control increasing transmission costs.¹²

While the Working Group members acknowledge that the value of summer peak demand reduction is already indirectly accounted for in the current performance incentive framework's BCR component, the group reached consensus on including components for both a passive summer and passive winter peak demand reductions in the electric programs' PI framework. The group also reached consensus that future opportunities for adoption of a demand reduction metric for natural gas programs should be explored as part of the 2021 -2023 planning process.

C. Incentive Eligibility Threshold

Under the existing performance incentive framework, a utility begins earning an incentive on the savings component upon achieving 65 percent of its targeted lifetime savings goal. However, in several other New England states, including Massachusetts,¹³ Connecticut,¹⁴ and Rhode Island,¹⁵ the threshold for earning an incentive is 75 percent of the program targets. As a result, consensus emerged among the working group members that New Hampshire should raise its incentive eligibility thresholds to align better with neighboring jurisdictions. However, the Working Group members also agreed that given the uncertainty surrounding passive summer and winter peak demand reductions and their dependence upon the programs' measure mix, a 65 percent minimum threshold would be applied to those new demand-related components.

¹¹ National Grid. 2018-20 Energy Efficiency and System Reliability Procurement Plan. (August 2017). Page 63-65. Available at: <u>http://rieermc.wpengine.com/wp-content/uploads/2017/08/2018-2020-3-year-plan-puc-8-30-17.pdf</u>; Order Re: Compensation Set-Aside and Performance Targets for Efficiency Vermont. (November 2017) Page A-1. Available at: <u>https://drive.google.com/file/d/1oFLJ3yOdHyCv-3UmXQsXpf1MBUnTWS9m/view?usp=sharing</u>; Memorandum dated October 19, 2018, Program Administrator Guide to Updates to the September 14, 2019- 2021 Draft Plan. Page 7. Available at: <u>http://ma-eeac.org/wordpress/wp-content/uploads/Memo-from-PAs-to-EEAC-10-22-18.pdf</u>

¹². See, e.g., Order No. 26,042 at 5 (July 24, 2017) (stating that transmission costs are tied to peak loads and requiring Unitil to consider what measures could be taken to mitigate increases in transmission costs); DE 18-089, Eversource Energy, 2018 Transmission Cost Adjustment Mechanism, Hearing Transcript of July 12, 2018, at 19-20; DE 18-051, Liberty Utilities (Granite State Electric) Corp., Annual Retail Rate Filing, Hearing Transcript of May 9, 2018, at 46-52.

¹³ Massachusetts 2019-21 Energy Efficiency Plan. (October 2018) Page 160. Available at: <u>http://ma-eeac.org/wordpress/wp-content/uploads/Exh.-1-Final-Plan-10-31-18-With-Appendices-no-bulk.pdf</u>

¹⁴ Connecticut 2019-21 Conservation and Load Management Plan Update. (March 2019) Page 368. Available at: https://www.energizect.com/sites/default/files/FINAL%202019%202021%20Plan%20%283-1-19%29.pdf

¹⁵ Rhode Island 2019 Energy Efficiency Program Plan. (October 2018) Page 42. Available at: <u>http://www.ripuc.org/eventsactions/docket/4888-NGrid-EEPP2019(10-15-18).pdf</u>
NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-002C Page 10 of 23

D. Sector Level Incentive Eligibility

Under the existing performance incentive framework, each utility's targets and related performance incentives are calculated on a sector-specific basis. As a result, if a utility under-performs in one sector, it cannot make up for that underperformance by over-performing in the other sector. This sends a signal that is inconsistent with the EERS: rather than pursue a statewide efficiency target as the EERS mandates, the existing framework suggests that there are two targets, one for each sector, thus encouraging the utilities to pursue them independently.

According to the National Efficiency Screening Project's Database of State Efficiency Screening Practices, many states, including Arizona, California, District of Columbia, Illinois, Michigan, New Mexico, New York, Oklahoma, Ohio, Pennsylvania, Rhode Island, Vermont, Washington, and Wisconsin, assess the cost-effectiveness of their programs at the portfolio level.¹⁶

While there is some inherent logic to incenting performance on a sector specific basis, Working Group members agreed that doing so limits flexibility to implement new programs and might unnecessarily limit the savings or cost-effectiveness pursued in a sector. In such a case, the utility would be reluctant to pursue all-cost effective programs, especially those with a lower BCR, if the utility is unable to offset the savings uncertainty associated with new programs in one sector by investment in highly cost-effective programs in the other sector.

Rewarding a utility's performance at the sector level also has implications for how income eligible programs are delivered. The Commission has the authority to approve income-eligible programs such as Home Energy Assistance (HEA) program where the BCR is less than 1.0.¹⁷ However, for the purposes of the performance incentive eligibility, HEA falls within the residential sector and represents a significant portion of the sector's overall budget goals. This limits the utility's ability to utilize the flexibility provided by the Commission regarding HEA program cost-effectiveness because the PI earned will potentially be less if the sector level BCR is less. By moving the calculation of incentives to the portfolio level, this flexibility is maintained because more programs can be used to offset a lower BCR from the HEA programs.

E. Benefit Cost Ratio Component

The existing performance incentive framework focuses half of the incentive on actual versus planned BCR. This is a primary component of the current framework. In most jurisdictions however, the BCR is treated as a threshold that must be met at either the measure, program or portfolio level before implementation of that measure, program, or portfolio is approved by a Commission, rather than a metric against which a program administrator is rewarded. While there is some inherent logic in encouraging the utilities to maximize the cost effectiveness of the programs, there was consensus among Working Group members that the energy efficiency portfolio should be focused on other metrics so that the BCR should set a floor for portfolio performance at 1.0. Stated another way, using a minimum B/C threshold of 1.0 before PI can be earned ensures that the benefits exceed the costs.

¹⁶ National Efficiency Screening Project. Database of State Efficiency Screening Practices. Accessed June 21, 2019. Available at: <u>https://nationalefficiencyscreening.org/state-database-dsesp/</u>

¹⁷ See Docket No. 96-150, Order No. 23,574 dated 11/01/2000 at 4.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-002C Page 11 of 23

Neighboring jurisdictions, including Massachusetts and Vermont, have embraced this approach to set the BCR as a threshold requirement and focus on other metrics for the PI components.

IV. Revised Framework

A. Current Framework Formula

Assuming a utility meets the minimum threshold of 55 percent of electric program total energy savings (electricity, natural gas, oil, propane, kerosene and wood) coming from electricity, the performance incentive earned by each electric utility under the current framework is as follows:

PI = [2.75% x ACTUAL] x [(BCRACT / BCRPLN) + (kWhACT / kWhPLN)]

Where:

PI = Performance Incentive in dollars ACTUAL = Total dollars spent less the performance incentive BCRACT = Actual Benefit-to-Cost ratio achieved BCRPLN = Planned Benefit-to-Cost ratio kWhACT = Actual Lifetime Kilowatt-hour savings achieved kWhPLN = Planned Lifetime Kilowatt-hour savings

If the minimum threshold of 55 percent of electric program energy savings from electricity is not achieved, then the PI formula is modified so that the 2.75 percent multiplier is replaced by a 2.2 percent multiplier. Otherwise it remains the same. For each sector, the BCR must be 1.0 or greater or no incentive is earned for the cost-effectiveness performance component for that sector. Actual lifetime savings must be at least 65 percent of the planned lifetime savings or no incentive is earned for that sector. Performance incentive is calculated separately for the two sectors Residential/Income Eligible and Commercial/Industrial. Total PI is the sum of the two.

The natural gas programs have no equivalent minimum kWh to total energy threshold requirement. Otherwise the calculation is identical except that the unit used for lifetime savings is MMBtu rather than kWh.

PI is currently capped at the component level for each of the following:

- Residential sector BCR
- Residential sector lifetime savings
- C&I sector BCR
- C&I sector lifetime savings

Taken together, the maximum performance incentive a utility can earn is the sum of 6.875 percent of the spending in each sector, with each sector calculated separately.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-002C Page 12 of 23

B. Revised Framework Formula

Under the revised framework, several additional components have been added, including two components related to summer and winter peak electric system passive demand¹⁸ and an annual savings component and a net benefits component.

PI =

[(1.925% x ACTUAL) x (kWhl-act/kWhl-pln)] + [(0.55% x ACTUAL) x (kWha-act/kWha-pln)] + [(0.66% x ACTUAL) x (kW_{SUM-ACT}/kW_{SUM-PLN})] + [(0.44% x ACTUAL) x (kW_{WIN-ACT}/kW_{WIN-PLN})] + [(1.925% x ACTUAL) x (NET-BEN_{ACT}/NET-BEN_{PLN})]

Where:

PI = Performance Incentive in dollars ACTUAL = Total dollars spent (less PI) kWhL-ACT = Actual Lifetime kWh kWhL-PLN = Planned Lifetime kWh kWhA-ACT = Actual Annual kWh kWhA-PLN = Planned Annual kWh kWsuM-ACT = Actual passive summer peak kW kWsuM-PLN = Planned passive summer peak kW kWwiN-ACT = Actual passive winter peak kW kWwiN-PLN = Planned passive winter peak kW

NET-BEN_{ACT}= Actual net benefits (in NPV dollars) (i.e. total benefits less utility costs and 5)¹⁹

NEI's)19

NET-BENPLN= Planned net benefits (in NPV dollars)

Additional requirements are as follows:

- The utility's portfolio of programs must be cost-effective before any PI can be earned, meaning the BCR must be at least 1.0;
- If electric program portfolio does not meet a minimum threshold of 55 percent of total energy savings from electricity, the coefficient will be reduced to 80 percent of the design value, that is, the total incentive level decreases to a maximum of 4.4 percent (e.g., for lifetime electric savings the PI would change from a target of 1.925 percent to a maximum of 1.54 percent, etc.);
- Lifetime savings must be at least 75 percent of planned lifetime saving in order for any PI to be earned on the lifetime savings component;
- Annual savings must be at least 75 percent of planned annual saving in order for any PI to be earned on the annual savings component;
- Passive summer peak kW savings must be at least 65 percent of planned passive summer peak kW in order for any PI to be earned on the summer demand component;

¹⁸ These demand components are excluded from the calculation of performance incentive for the natural gas programs. See Section C. under "Issues for Future Consideration" below.

¹⁹ See Appendix D.

- Passive winter peak kW savings must be at least 65 percent of planned passive winter peak kW in order for any PI to be earned on the winter demand component;
- The portfolio Net Benefits must be at least 75 percent of the planned Net Benefits in order for any PI to be earned on the Net Benefits component ;
- Earned PI on each component is capped at 125 percent of that component's coefficient, that is, the maximum total PI is 6.875 percent;
- PI will be calculated on actual portfolio spending up to 105 percent of approved portfolio budget, excluding performance incentive, without prior Commission authorization. That is, the actual spending may exceed the planned budgets, including all sources of funding and excluding the performance incentive, by up to 5 percent. A utility may request approval from the Commission to spend in excess of 105 percent of proposed budget in a given year if it can demonstrate good reasons why the cap should be exceeded. PI is then calculated against actual program spending at the portfolio level, up to 105 percent of the revised, Commission-approved budget, or as otherwise ordered.²⁰

V. Income Eligible Customers

A. Review by the Working Group

The Commission specifically tasked the Working Group with investigating the participation of income eligible customers in energy efficiency programs. Throughout its discussions, the Working Group weighed whether proposed changes would result in any unintended consequences related to design or implementation of the Home Energy Assistance program (HEA), or negatively impact the interests of income eligible customers. The group carefully considered including a specific metric related to achievement of goals in those programs, including establishing minimum spending or participation requirements. Input and feedback from The Way Home, which represents the interests of low income customers, as well as by the Office of Consumer Advocate, which represents residential customers, was sought throughout the process.²¹

²⁰ This represents a departure from the methodology set out in Order No. 25,189, Docket No. DE 10-188 at 9, whereby the performance incentive will be calculated using actual expenditures 'up to a maximum of 5% of the total approved by the Commission for each utility's residential and C&I sectors, <u>including performance incentive</u>...'[emphasis added]. Upon review, it was the conclusion of the Working Group that continuing with including the performance incentive as an expense in calculating the cap under the new proposed framework (now based on the portfolio approach) would introduce a circular component into the calculation that would allow the utilities to earn a performance incentive on the performance incentive. Accordingly, in keeping with the Working Group's assignment to review and propose new and alternative methodologies, it was the consensus of the group to modify the calculation by removing the cost of the performance incentive in setting the 105 percent cap.
²¹ On July 24, 2018, the PI Working Group and the B/C Working Group convened a special meeting to review current low-income programs (primarily HEA) and obtain feedback from Community Action Agencies, the utilities, project managers, and low-income advocates on program effectiveness and potential improvements.
²¹ On July 24, 2018, the PI Working Group and the B/C Working Group convened a special meeting to review current low-income programs (primarily HEA) and obtain feedback from Community Action Agencies, the utilities, project managers, and low-income advocates on program effectiveness and potential improvements.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-002C Page 14 of 23

B. Funding

Ultimately, the group reached consensus that the current 17 percent budget earmark for spending on low-income energy efficiency programs was sufficient and should be maintained. The Working Group also agreed that the recently instituted mandate to carry over any budgeted but unspent funds from HEA programs would ensure that sufficient funds were dedicated to these programs. Similarly, concerns that cost-effectiveness requirements (involving a BCR of 1.0 or greater) might limit participation of income eligible homes, have been addressed by a move from a sector level approach to a portfolio level approach. By moving to a portfolio level framework, in contrast to the sector level framework with its budgetary requirements, the Working Group was comfortable that the income eligible programs would be served adequately without adding a specific PI metric or component. In addition, the Working Group concluded that the net benefit component would help incent fossil fuel savings, which make up the primary benefit of weatherization activities in the income eligible programs. As a result, the Working Group members agreed that the income eligible programs would receive adequate investment and prioritization without the inclusion of a specific PI metric related to that customer segment in program year 2020. Should the PI framework be adjusted during the planning process for the next three-year plan, the topic of a specific income eligible metric may be revisited.

VI. Issues for Future Consideration

Over the course of the Working Group meetings, members reviewed many presentations from external experts as well as from the utilities and the OCA, and engaged in thoughtful discussion covering various aspects of performance incentive design. As these discussions progressed, several emerging developments in the energy efficiency field were considered but set aside due to the need for additional study and in the interest of reaching group consensus for the 2020 Program Year. This does not preclude future adjustment to the PI Framework to accommodate the evolution of program design, the adoption of new cost-effectiveness testing, the incorporation of a gas demand component, or other methods of calculating savings. Some of the ideas that may merit future investigation are discussed below.

A. Energy Optimization/Electrification

Energy Optimization (EO) is a concept that is known by different names in different jurisdictions. EO is a strategy undertaken by the utilities to provide customers with fuel-neutral education and encourage them to minimize energy usage through various energy efficiency measures. In practice, this has typically (but not exclusively) meant fuel switching from less efficient to more efficient, cleaner sources of energy. Heat pump technology and combined heat and power (CHP) are examples of common technologies considered under energy optimization. EO is also referred to in some circles as strategic electrification.

Both the existing PI Framework and the revised PI Framework focus on electricity savings (for electric programs) and natural gas savings (for natural gas programs), with some consideration given to other fuels saved. The current and revised PI frameworks do not consider overall energy savings, when switching from one fuel to another. Throughout the region, interest and investment in more holistic approaches to energy efficiency is increasingly involving technologies and appliances that shift energy use from dirtier fossil fuels to cleaner and more efficient natural gas and electric power. Massachusetts,

Vermont, Connecticut, Maine, and Rhode Island have begun placing a greater emphasis on *energy* savings as opposed to strictly *electric* savings among energy efficiency program planners and implementers.

One of the stumbling blocks encountered by the Working Group in judging the merits of creating a viable PI metric in this area is that EO is an emergent concept in New Hampshire in terms of policy, program design, implementation, and evaluation. An additional impediment was the availability of state-specific data involving deployment and utilization of optimization technologies. Currently, the EM&V Working Group and the B/C Working Group are working with Navigant, a third party evaluation firm, to investigate how other jurisdictions are handling EO in their energy efficiency planning, cost-effectiveness testing, and reporting, and the policies that support implementation.²²

Depending on the outcome of the Navigant-led study, and the EERS priorities for the 2021-2023 term, the utilities and the stakeholders may want to adjust the PI framework in the future to incent overall energy reductions, rather than just those energy reductions that result from a decrease in the use of electricity or natural gas alone. If that is the case, there will need to be further discussion about how to convert energy savings resulting from the efficiency programs to a common unit of energy, and whether to do so at the customer site or the generating source. A study to investigate these issues is currently being scoped in Massachusetts, the results of which may help to inform future New Hampshire energy efficiency program design.

B. Revised Cost Effectiveness Tests

The EM&V Working Group and the B/C Working Group are working with Synapse, a third-party firm, to review policies related to New Hampshire's cost-effectiveness test for energy efficiency programs, in accordance with the framework established in the National Standard Practice Manual ("NSPM"). Synapse will prepare a report that summarizes the key elements of the NSPM and how the B/C Working Group can apply those elements to the energy efficiency cost-effectiveness test are expected to be implemented beginning in 2021.

As described above, Total Resource Cost test is the current benefit/cost test for program screening and is expected to be the, the basis for the PI for 2020. If the screening cost-effectiveness test changes with a start date of program year 2021, then the PI framework, including the components and requirements, will need to be revisited since the benefit/cost test and the PI calculation overlap.

C. Gas Demand

As coal, oil and nuclear decline as fuels for the generation of electricity in the northeast, natural gas, along with renewables and energy efficiency, have filled in the gap. This additional demand for natural gas to meet the demand for electricity generation has strained already congested gas pipeline capacity in our region. This strain has been particularly acute during the winter months when demand for natural gas for heating homes and businesses reaches a peak. Short-term natural gas supply shortfalls have led

²² The Commission is currently investigating grid modernization, including strategic electrification, in Docket IR 15-296.

to wholesale price instability that regional energy planners, the Independent System Operator of New England ("ISO-NE"), regulators and the natural gas distribution companies throughout the region are attempting to address. Similarly, at the distribution level, natural gas utilities (including in New Hampshire) are experiencing peak day demand growth that threatens to exceed the level of firm supply that can be accessed without major new infrastructure investments. Reducing end users' natural gas demand will free up more pipeline capacity.

Unlike electricity measures and end uses, for which hourly load-shapes have been developed by energy efficiency evaluators as well as ISO-NE, the Working Group was not aware of readily available studies or related data sources for peak gas demand. Nor did the group find evaluation studies that show the peak gas demand reduction related to specific energy efficiency measures. There is currently no mechanism to put a dollar value on the demand reduction value of natural gas conserving activities during peak periods. This relationship is further complicated by the way in which natural gas is procured for the purpose of generating electricity (short term, spot market) versus the way it is procured by end-using customers who purchase from a natural gas local distribution company to heat their homes and businesses (long-term contracts, regulated rates).

While the Working Group members were in broad agreement that natural gas efficiency programs help ameliorate the winter gas supply issues, the gas utilities said that they do not track peak demand savings in New Hampshire. Without such information, the Working Group could not establish a meaningful goal or determine whether or not the natural gas programs have achieved it. Consequently, the Working Group agreed that the natural gas utilities would stay abreast of various studies in the region that are investigating the issue of natural gas peak demand in order to consider development and inclusion of a peak demand reduction metric for the next three-year plan period.²³

D. Income Eligible Participation

As noted above, the Working Group examined the feasibility of additional PI metrics to incentivize increased participation by low-income households in energy efficiency programs, including adoption of specific participation and savings targets. After considerable discussion and review, including outreach to other stakeholders outside the working group process, consensus was reached that maintaining adequate levels of investment and funding continues to be the most effective means of serving this community, at least through 2020. However, this is an evolving issue in many other jurisdictions, and

²³ One potential example of a peak day proxy strategy was recently identified by gas program administrators in Connecticut. As a condition of approval of the Connecticut 2019-2021 Statewide Energy Efficiency Plan, the Connecticut Department of Energy and Environmental Protection required the Connecticut Program administrators to "provide a quantification and discussion of the effects of conservation, load management, and energy efficiency investments, both electric and gas, on winter peak demand and as applicable, winter fuel reliability." In response to this condition, the program administrators provided a compliance filing describing the gas peak day savings by end use and measure-type groupings. *See* Connecticut Department of Energy and Environmental Protection. Attachment A: Schedule of Compliance Conditions of Approval. (December 2018) Available at: https://app.box.com/s/zv7bcoe283tjvppnt853ojmwfa89zahg/file/392424970636. *Also see* Connecticut Energy Efficiency Program Administrators. 2019-2021 Plan Compliance Item #7 – July 1 filing. Available at: https://app.box.com/s/u0kn24qi4f7baxypfionf5oeiam8lq2i/file/488657645351

the development and adoption of potential income eligible metrics merits further study and should be a consideration during the planning process for the next three-year plan.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-002C Page 18 of 23

Appendix

Appendix A: 2020 PI calculation templates

Proposed PI Calculation for Electric Utilities

			Portfoli	o Planned	Versus Actual	Performance	- 20	20				
					Design	Actual				125% of		
Portfolio	Planned	Threshold	Actual	% of Plan	Coefficient	Coefficient	- 1	Planned PI	F	Planned Pl	Actual PI	Source
1 Lifetime kWh Savings	169,249,199	126,936,899			1.925%		\$	1,204,667	\$	1,505,834		Planned and Actual from Cost Eff Tab
2 Annual kWh Savings	140,178,883	105,134,162			0.550%		\$	344,191	\$	430,238		Planned and Actual from Cost Eff Tab
3 Summer Peak Demand kW	16,769	10,900			0.660%		\$	413,029	\$	516,286		Planned and Actual from Cost Eff Tab
4 Winter Peak Demand kW	19,383	12,599			0.440%		\$	275,352	\$	344,191		Planned and Actual from Cost Eff Tab
5 Total Resource Benefits	\$ 206,636,229											Planned and Actual from Benefits Tab
6 Total Utility Costs ¹	\$ 62,580,111											Planned and Actual from Cost Eff Tab
7 Net Benefits	\$ 144,056,118	###########			1.925%		\$	1,204,667	\$	1,505,834		Line 5 minus line 6
8 Total					5.500%		\$	3,441,906	\$	4,302,383		

		Total Resource	Cost Test	
		Planned	Actual	Source
9	Total Benefits (incl. NEIs)	\$ 227,299,852		Planned and Actual from Cost Eff Tab
10	Performance Incentive	\$ 3,441,906		from row 6 above
11	Participant Costs	\$ 52,022,201		Planned and Actual from Cost Eff Tab
12	Total Utility Costs	\$ 62,580,111		from row 4 above
13	Portfolio TRC BCR	1.93		row 9 divided by rows 10+11+12

For illustrative purposes only. All dollar values are expressed in 2020 dollars. The numbers reflect the cumulative budget, savings, benefits, and costs of all the utilities combined based on the original 2020 Plan. Each utility will file its own utility-specific version of the table as part of the 2020 Plan Update.

¹ Note that in order to avoid a circular reference in the calculation of performance incentive, "Total Utility Costs" does not include the value of PI.

		Portfolio Planned Versus Actual Performance - 2020													
							Actual								
						Design	Coefficie		125% of						
	Portfolio	Planned	Threshold	Actual	% of Plan	Coefficient	nt	Planned Pl	Planned Pl	Actual PI	Source				
1	Lifetime MMBtu Savings	2,306,693	1,730,020			2.475%	5	\$ 226,656	\$ 283,320		Planned and Actual from Cost Eff Tab				
2	Annual MMBtu Savings	163,616	122,712			1.100%	5	\$ 100,736	\$ 125,920		Planned and Actual from Cost Eff Tab				
3	Total Resource Benefits	\$21,622,091									Planned and Actual from Benefits Tab				
1	Total Utility Costs	\$ 9,157,813									Planned and Actual from Cost Eff Tab				
5	Net Benefits	\$ 12,464,278	\$ 9,348,208			1.925%	5	\$ 176,288	\$ 220,360		Line 5 minus line 6				
5	Total					5.500%	6	\$ 503,680	\$ 629,600						

Proposed PI Calculation for Gas Utilities

			Tota	al Resource Cost Test
		Planned	Actual	Source
7	Total Benefits (incl. NEIs	\$23,784,300		Planned and Actual from Cost Eff Tab
8	Performance Incentive	\$ 503,680		from row 8 above
9	Participant Costs	\$ 5,999,410		Planned and Actual from Cost Eff Tab
10	Total Utility Costs	\$ 9,157,813		from row 6 above
11	Portfolio TRC BCR	1.52		row 9 divided by rows 10+11+12

For illustrative purposes only. All dollar values are expressed in 2020 dollars. The numbers reflect the cumulative budget, savings, benefits, and costs of all the utilities combined based on the original 2020 Plan. Each utility will file its own utility-specific version of the table as part of the 2020 Plan Update.

¹ Note that in order to avoid a circular reference in the calculation of performance incentive, "Total Utility Costs" does not include the value of PI.

Appendix B: The members/participants of the PI Working Group:

- Jay Dudley, PUC
- Jim Cunningham, PUC
- Paul Dexter, PUC
- Elizabeth Nixon, PUC
- Leszek Stachow, PUC
- Brian Buckley, Office of Consumer Advocate
- Donald Kreis, Office of Consumer Advocate
- Rebecca Ohler, New Hampshire Department of Environmental Services (NH DES)
- Joe Fontaine, NH DES
- Christopher Skoglund, NH DES
- Kate Peters, Eversource
- Miles Ingram, Eversource
- Marc Lemenager, Eversource
- Christopher Plecs, Eversource
- Erica Menard, Eversource
- Tom Fuller, Eversource
- Christopher Goulding, Eversource²⁴
- Matthew Fossum, Eversource
- Cindy Carroll, Unitil
- Mary Downes, Unitil
- Eric Stanley, Liberty
- Heather Tebbetts, Liberty
- Trish Walker, Liberty
- Mike Sheehan, Liberty
- Carol Woods, NH Electric Coop
- Melissa Birchard, Conservation Law Foundation
- Raymond Burke, NH Legal Assistance/The Way Home
- Ellen Hawes, Acadia Center
- Amy Boyd, Acadia Center
- Scott Albert, GDS Associates
- Madeleine Mineau, Clean Energy NH
- Brianna Brand, Clean Energy NH

²⁴ Christopher Goulding is now employed by Unitil.

Appendix C: Consultants who assisted and contributed to the work of the PI Working Group:

- Denise Rouleau, Northeast Energy Efficiency Partnerships (NEEP)
- Emily Levin, Vermont Energy Investment Corporation (VEIC)
- David Farnsworth and Jessica Shipley, Regulatory Assistance Project (RAP)
- Philip Mosenthal, Optimal Energy
- Martin Kushler, American Council for an Energy Efficient Economy (ACEEE)
- Lisa Skumatz, Skumatz Economic Research Associates (SERA)
- Ralph Prahl, SERA
- Robert Wirtshafter, SERA

Appendix D: Glossary of Terms

Actual: The amount of savings, spending, net benefits or BCR the programs achieved, as reported in each utility's annual report and associated Benefit Cost models.

Adjusted gross savings: The amount of savings resulting from energy efficiency measures, adjusted to reflect realization rates and other impact factors quantified in third party evaluations, exclusive of free-ridership and spillover.

Annual savings: The reduction in electricity use (kWh) or fossil fuel use (therms or MMBtus) over a oneyear period resulting from energy efficiency programs.

Benefit-Cost Ratio ("BCR"): As calculated by the NH Utilities' Benefit/Cost test, currently the Total Resource Cost ("TRC") test, the BCR is the ratio of total benefits and total costs. Total benefits are the net present value of avoided energy and non-energy impacts resulting from program measures. Total costs are the net present value of utility costs, including performance incentive, plus out-of-pocket incremental costs that customers pay for energy efficiency measures, relative to a standard efficiency measure.

Demand savings: Demand savings is the reduction in electricity demand (kW). Demand savings can result from active resources, which are activated when dispatched (i.e., demand response), or passive resources (e.g., installation of more efficient equipment) and not in response to a dispatch instruction. For purposes of the PI calculation, the peak demand savings are coincident with ISO-NE system peak demand periods.

Independent System Operator of New England ("ISO-NE") peak demand savings: The savings resulting from passive peak demand reduction occurring during the "on-peak" hours defined by ISO-NE. Specifically, summer peak demand reductions are the average reduction in demand during summer peak hours (non-holiday weekdays, 1:00 p.m. to 5:00 p.m., during June, July, and August) and winter peak demand reductions are the average reductions in demand during winter peak hours (non-holiday weekdays, 5:00 p.m. to 7:00 p.m., during December and January).

Lifetime savings: The reduction in electricity use (kWh) or fossil fuel use (therms or MMBtus) over the lifetime of installed energy efficiency measures, based on the life of a measure as determined through evaluation.

Net Benefits: Net Benefits are the Net Present Value of Total Resource Benefits less Total Utility Costs (not including Performance Incentive). Neither the value of customer costs nor non-energy impacts is considered in determining Net Benefits for purposes of calculating the performance incentive.

Planned: The amount of savings, spending, net benefits or BCR the programs are expected to achieve, based on the utilities' Three-Year Plan and typically updated each year in Annual Update filings and associated Benefit Cost models.

Portfolio: The total set of energy efficiency programs offered by a utility, including those activities that do not directly save energy (e.g., education, EM&V, marketing, lending programs, etc.) across all sectors.

Sector: A group of customers with similar characteristics, usage patterns and billing rates. Residential, and Commercial and Industrial (C&I) are the two primary sectors in the NH Saves programs.

Total Resource Benefits: Avoided costs due to program impacts on electric capacity, electric energy, Demand Reduction Induced Price Effects (DRIPE), gas benefits, other fuels, and water resources.

Utility costs: All expenditures by the program administrator to design, plan, administer, deliver, monitor, and evaluate efficiency programs, including performance incentive.

Date Request Received: November 01, 2022 Data Request No. RR 1-006 Date of Response: December 16, 2022 Page 1 of 3

Request from: New Hampshire Public Utilities Commission

Request:

<u>Reporting on Spending by Category:</u> Reporting requirement 9.iii. required the Utilities to report on their spending by category. To ensure the Commission has the most accurate and clear data, the Commission requests that the Joint Utilities submit one response to each of the following questions:

- 1. Please define how the Joint Utilities use the term "rebate", and what expenses are included in this category.
- 2. Please fill out the table in the attached excel (Tab: Request VI.2 Contractors) in order to compile the individual 2021 Expenses by Contractors reports filed pursuant to reporting requirement 9.iii into a single program-wide dataset while further breaking out the cost of the contracted services from the rebate portion of the total and providing this information by program.
- 3. Please fill out the table in the attached excel (Tab: Request VI.3 Program Expenses) in order to combine the individual 2021 Expenses/Revenues by Category reports filed pursuant to reporting requirement 9.vi into a single program-wide dataset while further breaking out the cost of the contracted services from the rebate portion of the total. Please provide planned and actual data for each completed program year since 2016.
- 4. The Commission understands that rebates flow through contractors and consultants to New Hampshire ratepayers. This request is intended to get a clearer picture of Energy Efficiency Program expenditures on services provided by contractors and consultants to understand to what degree program expenditures are flowing out of the New Hampshire economy. Please calculate the 2021 Energy Efficiency Program expenditures for contractors and consultant services broken out by the New Hampshire and Non-New Hampshire business addresses.

Date Request Received: November 01, 2022 Data Request No. RR 1-006 Date of Response: December 16, 2022 Page 2 of 3

Response:

 Please refer to Bates 23 of the NHSaves 2022-2023 Plan, which was filed on March 1, 2022, to Docket No. DE 20-092 at Tab 7 and approved on April 29, 2022 by Order No. 26,621. "Customer Rebates and Services" is a cost category that includes:

"Costs associated with incentives that reduce the cost of equipment as well as costs for services to speed adoption. This includes direct rebate dollars paid to distinct participants, as well as indirect incentives for equipment discounts. It also includes services such as technical audits, employee and contractor labor to install measures, expenses, materials, and supplies."

2.

11/30/2022 Response:

The Utilities were granted an extension for the response to this question.

12/16/2022 Response:

Please see Excel Attachment RR 1-006B for the summation of the utility information provided to the Commission on August 1, 2022, in this docket, with data split out by program(s) served. "Customer Rebates and Services" are tracked as a single cost category, as detailed in the approved 2022-2023 Plan and in response to RR 1-006 part 1. Therefore, the Utilities cannot accurately segregate the subcomponents of this singular category and aggregate the data across the Utilities to provide a single program-wide dataset. Although the "Total Amount" has not changed from the August 1, 2022, reporting requirement, Eversource identified an additional \$531,925.26 in "customer rebates and services" that were not previously identified as such.

The Utilities would like to direct the Commission's attention to the additional written context provided for this filing requirement within the August 1, 2022, filing and also note that the results presented are representative of the programs supported by contractor and vendor efforts solely in 2021, which may differ from other program years, as the same vendors may support different programs or other vendors not utilized in 2021 may be involved.

3. Please see Excel Attachment RR 1-006A for the summation of the utility information provided to the Commission on August 1, 2022, in this docket. There is also written context explaining this data that was provided along with this filing requirement in the August 1, 2022, filing. The Utilities incorporate that written narrative into this response by reference.

Date Request Received: November 01, 2022 Data Request No. RR 1-006

Date of Response: December 16, 2022 Page 3 of 3

4. This information was provided in response to reporting requirement 3.1, filed on August 1, 2022, in this docket, where the Utilities provided a summary of the expenditures, based on business billing addresses, broken out by state or country.

The Utilities' response to Reporting Requirement 3.1 also provided additional context, and the Utilities reiterate that context here:

The Utilities note that the data provided in and of itself does not fully reflect the impact of the programs at large. While the utilities have not collected this information, nor would it be administratively feasible to collect this information, it is important to consider the following when reviewing the table:

- The business address of a given contractor or vendor is not necessarily reflective of the location of the individual(s) who work on NHSaves programs. For example, there are vendors with corporate addresses outside of New Hampshire whose employees work within New Hampshire.
- Contractor and vendor payments are not reflective of the full economic impact of the projects performed. The column labeled 'Rebate Portion of Total Amount' reflects the amounts that flowed through the contractors to the customers in the form of cash rebates and/or the cost of procuring and installing energy efficiency measures on customer premises, which necessarily take place within New Hampshire and benefit utility customers.
- Additional Federal dollars were invested into the income-eligible project[s] completed.

Date Request Received: November 01, 2022 Data Request No. RR 1-007 Date of Response: December 16, 2022 Page 1 of 5

Request from: New Hampshire Public Utilities Commission

Request:

Reporting on Low-Income Program Offerings: The Legislature requires that a minimum of 20% of the program expenditures are spent on low-income ratepayers. In order for the Commission to understand the rationale of meeting this requirement through separate programming, as opposed to fully including low- income ratepayers within the general programming, the Commission requests that the Joint Utilities submit one response to each of the following questions:

- 1. Please report on the history of when and why programming for low-income individuals was separated from the general program offering.
- 2. Please report on how much was expended for low-income ratepayers during the program year 2021 and please demonstrate that the statutory requirement of 20% was fully met.
- 3. Please calculate the benefit-to-cost ratio for the Home Energy Assistance Program (HEA) in 2021 and how it compares to the other programs. Have the Joint Utilities explored other more cost-effective options to serve low- income ratepayers?
- 4. Please calculate how much has been spent on the HEA program since 2016. Of this amount, how much was spent on rebates, and how much on contractor and consultant services? Please break the contractors and consultant services expenditures out by New Hampshire and Non-New Hampshire business addresses.
- 5. Please provide data on the aggregate program year 2021 cost of the SBC paid by ratepayers who also qualify for the HEAProgram.

Response:

1. Income-eligible programs provided to low-income customers pre-date the establishment, in 2001, of the CORE/NHSaves energy efficiency programs in New Hampshire, and they have remained separate and distinct ever since. While dedicated weatherization services are offered at no cost to those who are determined to be income eligible by the Community

Date Request Received: November 01, 2022 Data Request No. RR 1-007

Date of Response: December 16, 2022 Page 2 of 5

Action Agencies that implement the programs, these customers are not precluded from participating in any other NHSaves residential energy efficiency program or offering.

The specific focus on economically vulnerable customers is common practice nationally, and reflects the unique needs of this population, which experiences disproportionate energy burden, or percentage of household income spent on heat and electricity, is not typically in a position to contribute to the cost of weatherization or other energy-related upgrades to their homes, and contains a disproportionate number of renters (https://www.aceee.org/topic/low-income, https://www.aceee.org/sites/default/files/pdfs/ACEEE-01 Energy Burden - National.pdf). Yet, low-income customers pay for energy efficiency services through their rates. Therefore, many states require utilities to provide low-income energy efficiency programs, and many states have alternative cost-effectiveness rules for low-income programs such as different rebate amounts to overcome first cost barriers, funding for health and safety repairs and differences in savings (due to different baselines, housing conditions, and rates of replacement).

There is extensive literature on both the needs of and benefits to low-income populations resulting from the energy efficiency programs that target them. A few such resources are linked below. Some of this literature has been commissioned by the Federal Department of Energy to assess the effectiveness of the Weatherization Assistance Program ("WAP"), which provides funds to State Energy Offices (e.g., NH Department of Energy) to serve eligible households. To extend the impact of limited WAP funding, the NH DOE and Community Action Agencies collaborate with the Utilities and the NHSaves programs to reach more households each year than either funding source could achieve independently. https://www.aceee.org/toolkit/2021/04/supporting-low-income-energy-efficiency-guide-utility-regulators

https://www.aceee.org/sites/default/files/publications/researchreports/u1713.pdf https://weatherization.ornl.gov/wpcontent/uploads/pdf/WAPRetroEvalFinalReports/ORNL_TM-2014_338.pdf http://www.threecubed.org/

2.

11/30/2022 Response:

The Utilities were granted an extension for the response to this question.

12/16/2022 Response:

Please refer to the tab labeled "1. Att E1 Cost Eff" in the Benefit Cost Models provided as part of reporting requirement 2.2a on August 1, 2022, in this docket and the 2021 Performance Incentive filings filed between May 31 and June 2, 2022, in Docket No. DE 17-136.

Date Request Received: November 01, 2022 Data Request No. RR 1-007

Date of Response: December 16, 2022 Page 3 of 5

Each year, the NH PUC performs a fiscal audit of each utility's energy efficiency programs. During that audit, the revenues and expenditures of each program are scrutinized and an assessment of compliance with statutory requirements is undertaken, including the requirements pertaining to income-eligible program allocations and expenditures. Should spending on the income-eligible programs fail to meet the 20% threshold established by HB 549, each utility that experienced a shortfall shall increase the budget for income-eligible programs the subsequent year to continue to maintain compliance with the threshold.

Please see Attachment RR 1-007B for a summary of each electric utility's 2021 income-eligible budget as filed on August 13, 2021, in Docket No. DE 17-136, actual SBC revenues, calculation of 20% of the SBC revenues, and actual income-eligible program expenditures in program year 2021. Please note that the gas utilities' energy efficiency rates in effect for most of 2021 were approved by the Commission in late 2020 and were determined in anticipation of the 2021-2023 Plan also being approved. Because the plan was not approved, the gas utilities collected more revenue than was needed for the 2021 budget given the continuation of the budget at 2020 levels. Thus, the budget, revenues, and expenditures for 2021 are displayed.

 Each NH utility includes the B/C ratio for each program in their annual performance incentive filings, filed annually on or around June 1 of the following each program year. Additionally, the reporting requirements stemming from Order No. 26,621, filed on August 1, 2022, included this information in the B/C Models for each utility under the Excel worksheet labeled "Cost Eff".

As noted in the response to Question RR 1-007 part 1, income eligible ratepayers are not precluded from accessing other residential energy efficiency programs, such as the Energy Star Products program, however the Utilities do not track the income levels of customers participating in those programs. The Utilities have also provided services beyond weatherization to low-income customers and the single and multi-family properties where they live, including refrigerator replacements, lighting projects for both common and in-unit spaces, heating system replacements, and mailed energy efficiency kits including low-cost items such as light-bulbs, smart strips and educational materials.

The Utilities continuously review program offerings with partners and vendors to improve outreach approaches, consider additional cost-effective measures, and to adjust vendor reimbursement in light of market changes related to labor costs and supply chain issues.

Date Request Received: November 01, 2022 Data Request No. RR 1-007

Date of Response: December 16, 2022 Page 4 of 5

4.

11/30/2022 Response:

The Utilities were granted an extension for the response to this question.

12/16/2022 Response:

Information regarding the expenses for the HEA Program, as well as all other programs, are provided in each utility's annual Performance Incentive filings, which are filed on or around June 1 following each program year in Dockets DE 14-216 (2016 and 2017) and DE 17-136 (2018-2021). For ease of review, the expenses since 2016 have been summarized in Attachment RR 1-007C. The NH Utilities track 'rebates and services' in a single line item per the definition provided in each Plan¹ and approved by the Commission. The Utilities do not distinguish between "rebates" and "contractor and consultant services" in their reporting or bookkeeping and therefore cannot provide the breakdowns requested.

For administrative efficiency, rebate payments are typically provided by the utility directly to the lead contractor undertaking the work at the customer's address. In the case of the HEA program, those payments are typically provided to the responsible Community Action Agency, but in some cases may be provided to an independent weatherization, lighting or appliance contractor. Functionally, this is no different than the utility providing payment (i.e., a "rebate") to the customer, who would in turn be responsible for paying the contractor / vendor.

While there are indirect out-of-state impacts resulting from the funds expended by the NHSaves programs (i.e., materials related to weatherization are generally manufactured somewhere other than New Hampshire), 100% of the direct benefit related to the rebates and services funded by the NHSaves programs are realized by New Hampshire customers. For contractor and vendor costs by state, based on each vendor's address on file, please refer to the Utilities' response to reporting requirement 3.1, filed on August 1, 2022, in this docket and Attachment RR 1-006B in response to RR 1-006 part 2.

¹ The definition of 'rebates and services' is as follows and can be found in Table 1-13 on Bates page 23 of the March 1, 2022, filing of the "2022-2023 NH Statewide Energy Efficiency Plan." "Costs associated with incentives that reduce the cost of equipment as well as costs for services to speed adoption. This includes direct rebate dollars paid to distinct participants, as well as indirect incentives for equipment discounts. It also includes services such as technical audits, employee and contractor labor to install measures, expenses, materials, and supplies."

Date Request Received: November 01, 2022 Data Request No. RR 1-007

Date of Response: December 16, 2022 Page 5 of 5

5. The Utilities do not maintain a comprehensive list of customers who qualify for the HEA Program. Customers are eligible for services under the HEA Program if their household income is verified by a Community Action Agency as less than 60% of area median income, which varies by household size as shown in the table below (retrieved 11/10/2022 from https://www.energy.nh.gov/consumers/help-energy-and-utility-bills/assistance-programs-eligibility). Customers are under no obligation to sign up for services provided by the Community Action Agencies such as heating assistance, electric energy assistance and weatherization assistance, and doing so involves income verification, which presents a barrier to many customers.

While the Utilities offer discounted electricity and natural gas rates for customers who have been income qualified by their local Community Action Agency, enrollees represent only a fraction of customers who are eligible for assistance. Referencing data from the American Community Survey, the number of eligible households is greater than 100,000 of the nearly 550,000 households in the state, which would be a minimum of 18% of total households statewide. This is inferred based on the count of households earning less than \$35,000 annually, an income level somewhat lower than the \$38,969 threshold for eligibility for a single-person household (Please see Attachment RR 1-007A for the ACS output for New Hampshire for totals by household income).

Even if a precise number of households eligible for the HEA Program could be determined, the amount of electricity used by those households and therefore the proportion of SBC funds contributed by those households to the NHSaves programs could not be determined. As a result, the question cannot be answered.

60% New Hampshire State Median Income Guidelines Income Eligibility Based on Family Size and Gross Annual Income For a family of 1, a maximum gross annual income of \$38,969 For a family of 2, a maximum gross annual income of \$50,959 For a family of 3, a maximum gross annual income of \$62,950 For a family of 4, a maximum gross annual income of \$74,941

- For a family of 5, a maximum gross annual income of \$86,931
- For a family of 6, a maximum gross annual income of \$98,922
- For a family of 7, a maximum gross annual income of \$101,170
- For a family of 8, a maximum gross annual income of \$103,418

2021 HEA Budgets and Expenses

	Eversource		NHEC		Liberty Electric		Liberty Gas		Unitil Electric		Unitil Gas	
HEA Budget	\$	8,110,149	\$	838,773	\$	1,201,849	\$	1,482,845	\$	1,628,055	\$	413,000
Actual SBC /LDAC EE Revenues Collected	\$	41,008,456	\$	4,079,423	\$	4,762,864	\$	9,467,273	\$	6,243,595	\$	3,001,278
20% of Actual EE Revenues Collected	\$	8,201,691	\$	815,885	\$	952,573	\$	1,893,455	\$	1,248,719	\$	600,256
Actual HEA Expenditures	\$	\$ 9,033,638		687,222	\$	1,128,321	\$	1,649,014	\$	1,730,657	\$	401,744

Notes:

1. The approved gas rates for program year 2021 were higher than the effective budget for 2021. In the absence of a Commission Order,

the 2021 budget reverted to the approved 2020 budget. Revenues collected were higher than what the gas companies were authorized to spend.

2. The budget for income eligible gas programs in 2020 (and by extension 2021) was equal to 17% of the budget, not 20%.

NHSaves Energy Efficiency Programs NHPUC Docket No. IR 22-042 11-01-2022 IR Requests Attachment RR 1-007C Page 1 of 1

	2016		201	17	201	18	201	19	202	20	2021			
	Rebates	Total												
Eversource	\$2,358,313	\$2,630,799	\$2,637,387	\$2,930,672	\$4,496,482	\$4,934,976	\$6,643,780	\$7,295,396	\$7,124,091	\$7,908,121	\$8,508,193	\$9,033,638		
NHEC	\$218,191	\$292,376	\$212,248	\$280,148	\$248,970	\$348,316	\$423,481	\$529,830	\$605,446	\$731,532	\$580,981	\$687,222		
Liberty Electric	\$261,644	\$314,958	\$266,151	\$352,987	\$262,918	\$319,646	\$533,885	\$670,522	\$730,624	\$921,576	\$982,737	\$1,128,321		
Liberty Gas	\$914,958	\$1,003,642	\$673,059	\$826,371	\$838,637	\$984,077	\$1,481,848	\$1,683,152	\$1,465,382	\$1,676,017	\$1,479,298	\$1,649,014		
Unitil Electric	\$241,021	\$369,329	\$248,165	\$388,231	\$451,291	\$662,392	\$642,145	\$1,004,405	\$867,053	\$1,170,180	\$1,423,361	\$1,730,657		
Unitil Gas	\$147,055	\$232,459	\$139,654	\$221,118	\$252,483	\$350,001	\$287,733	\$380,642	\$295,418	\$424,240	\$275,309	\$401,744		

2016-2021 HEA Rebates/Services and Total Expenditures

HEA Rebates/Services Business Addresses: New Hampshire vs. Other States

	2016		2017		20:	18	20:	19	202	20	2021	
	NH	Other States	NH	Other States	NH Other States		NH	Other States	NH	Other States	NH	Other States
Eversource	\$2,358,313	\$0	\$2,637,387	\$0	\$4,496,482	\$0	\$6,643,780	\$0	\$7,120,817	\$3,275	\$8,508,193	\$0
NHEC	\$218,191	\$0	\$212,248	\$0	\$248,970	\$0	\$423,481	\$0	\$512,186	\$93,260	\$580,981	\$0
Liberty Electric	\$236,210	\$25,434	\$254,017	\$12,134	\$244,534	\$18,384	\$444,374	\$89,511	\$658,123	\$72,501	\$837,007	\$145,731
Liberty Gas	\$912,256	\$2,702	\$669,459	\$3,599	\$836,126	\$2,511	\$1,435,419	\$46,429	\$1,451,612	\$13,770	\$1,430,044	\$49,255
Unitil Electric	\$224,670	\$16,351	\$226,804	\$21,361	\$404,733	\$46,558	\$501,457	\$140,689	\$857,758	\$9,295	\$1,413,424	\$9,937
Unitil Gas	\$147,055	\$0	\$139,654	\$0	\$252,483	\$0	\$287,733	\$0	\$295,418	\$0	\$275,309	\$0

Date Request Received: November 04, 2022 Data Request No. RR 2-001 Date of Response: December 16, 2022 Page 1 of 2

Request from: New Hampshire Public Utilities Commission

Request:

Topic I. Granite State Test (GST), Total Resource Cost Test (TRCT), Discount rates

- a) The Commission is generally interested in understanding approaches to assessing the costeffectiveness of energy efficiency programs based on statutory or other implementation frameworks. Please provide information on approaches to assessing cost-effectiveness by or basedon:
 - i. Other state utility commissions;
 - ii. Federal regulatory entities;
 - iii. Economic theory that addresses social welfare impacts; and
 - iv. Statutory definitions of "cost-effective" from other jurisdictions.
- b) The Commission is aware that in some cost-benefit models, economists use two different discount rates to analyze the cost-effectiveness of investments; a social discount rate, sometimes referred to as the rate of social time preference, and a capital discount rate, sometimes referred as the marginal cost of capital. A social discount rate is used to depreciate the flow of future benefits of an investment into present terms over the lifetime of the investment. The capital discount rate is used to depreciate the present value of an investment over the lifetime of the investment. Both the GST and the TRCT appear to use a singular discount rate to depreciate the costs of Energy efficiency investments and the estimated benefits of energy savings.
 - i. Do the participants view the discount rate used in the GST and TRCT as a capital discount rate or a social discount rate?
 - ii. Is there a specific rational for using a single discount rate to represent the marginal cost of capital and society's preference for time?
 - iii. Between utility ratepayers, utility companies, and potential investors, whose discount rate is most appropriate to use given the sources of funding?

We encourage participants to provide any information/documents/research that pertains to questions a) and b) above.

Date Request Received: November 04, 2022 Data Request No. RR 2-001 Date of Response: December 16, 2022 Page 2 of 2

Response:

(A) The Utilities do not readily possess any of the information requested except for Massachusetts and Connecticut, which was provided in response to the questions asked on November 1st in this docket. HB 549 effectively defined "cost-effectiveness" for the NHSaves programs by enshrining the GST, which was already fully defined, including the utilization of a low discount rate. The GST was first approved by the Commission for effect in 2020 as the primary cost-effectiveness test.

(B) Please refer to the response to Part (a) of this question, as well as the filing requirements provided on August 1, 2022, in this docket and the response to RR 1-001 part 8.

Date Request Received: November 04, 2022 Data Request No. RR 2-002 Date of Response: December 16, 2022 Page 1 of 2

Request from: New Hampshire Public Utilities Commission

Request:

Topic II. Performance Incentive

- a. Through HB 549 of the 2022 session, the legislature mandated that the existing performance incentive (PI) remain in place until changed by an order or operation of law, and future PI structure is not prescribed.
 - i. What was the original purpose of the PI?
 - ii. Does the current PI structure achieve the purpose provided in response to question II-a-i above?
 - iii. Has the purpose of PI changed? Please address this purpose with respect to changes in Lost Base Revenue and Decoupling mechanisms.
 - iv. Is there a simpler and/or more effective way to achieve this purpose through the PI structure?
 - v. Are the current PI thresholds and rates effective at maximizing energy savings?
 - vi. Are the current PI thresholds too high or too low?

Response:

Please refer to the Performance Incentive Working Group report for the current PI mechanism, which revised the older mechanism. The PI mechanism that resulted from the Working Group's efforts was developed during a period where the Utilities had begun decoupling, and the Report explains the metrics and rationale behind the final framework. The proposed changes to the PI mechanism were approved by the Commission for effect beginning in the 2020 Program Year. The Utilities continue to deliver a suite of programs consistent with those delivered in 2020.

Among other changes described in detail in the Report, as well as in the Commission's Order approving the adoption of the new calculation, the resulting PI formula includes more components than the previous version, was changed from being calculated independently for each sector (i.e., residential/income-eligible and commercial/industrial) to being calculated at the portfolio level, and revised the calculation of net benefits to include utility costs only, exclusive of customer costs.

Date Request Received: November 04, 2022 Data Request No. RR 2-002 Date of Response: December 16, 2022 Page 2 of 2

As referenced in footnote 9 of the report, the American Council for an Energy-Efficient Economy published a 'Topic Brief' on Energy Efficiency Performance Incentives for Electric Utilities in December 2018, which provides a brief description of the role of PI in ensuring successful energy efficiency programs. That Brief is provided here as Attachment RR 2-002A and notes that performance incentive mechanisms "are critical to advancing utility-sector efficiency programs and performance."

Date Request Received: November 04, 2022 Data Request No. RR 2-003 Date of Response: December 16, 2022 Page 1 of 2

Request from: New Hampshire Public Utilities Commission

Request:

Topic V. Market Barriers

"Market Barriers," as used in the Energy Efficiency Program, is not a statutorily defined term; as such, the Commission solicits the submission of information responsive to the following:

- a) The 1999 Energy Efficiency Working Group was unable to reach a consensus on what constituted a Market Barrier¹, and 23 years later, neither the Commission nor the General Court appears to have adopted a formal definition. Consistent with RSA 374-F:3, X, please submit definitions of "Market Barrier" based on the following:
 - i. Economic theory;
 - ii. The practices of other state utility commissions; and
 - iii. Any applicable practices of federal regulatory entities.

Response:

Please refer to the response provided to RR 1-005 part 1.

Two sources that define market barriers as they relate to energy efficiency programming, which reflect the concept of market barriers as articulated within the NHSaves plan filings:

In the 1992 Lawrence Berkeley Lab technical report, "Market barriers to energy efficiency", the authors define market barriers as "the factors behind the so-called 'efficiency gap' – the differential between the level of energy-efficiency actually achieved and the level judged to be cost-effective at prevailing prices" (PDF, 3). The authors list the following examples of market barriers:

- Consumers do not have complete information about energy-efficient options (this is an example of the market failure called imperfect information).

- Consumers discount the value of future savings by too much when they consider investing in energy efficient options.

Date Request Received: November 04, 2022 Data Request No. RR 2-003 Date of Response: December 16, 2022 Page 2 of 2

- In some cases, different groups receive the costs and benefits from investing in energy efficiency. For example, in the case of rental housing, landlords bear the cost of investment while tenants receive the benefits (this is an example of the market failure called split incentives).

- The additional transaction costs from purchasing energy-efficient technologies may outweigh the benefits individual consumers can get when they consider whether to invest in energy efficiency.

The authors of the 2012 ACEEE Summer Study paper, "Guiding the Invisible Hand: Policies to Address Market Barriers to Energy Efficiency", found that researchers had most commonly defined market barriers either as equivalent to conventional market failures from economic theory, or as anything that works against investment in energy efficiency. The authors of this paper define market barriers as "any reason for economically inefficient levels of investment in energy efficiency" (PDF, 2). The authors note that "most of these are the market failures recognized by conventional economics; however, we also include systematic reasons for low investment in energy efficiency that are not market failures" such as government regulations, transaction costs, and cognitive and behavioral factors (PDF, 2-3).

Regarding other state regulatory definitions of market barriers, one example is the Iowa Administrative Code (IAC), and the chapter on utility energy efficiency planning which defines market barrier as "a real or perceived impediment to the adoption of energy efficient technologies or energy efficient behavior by consumers" in its <u>definitions list</u>.

Sources:

Howarth & Andersson (1992). "Market Barriers to Energy Efficiency". Retrieved from <u>https://www.osti.gov/servlets/purl/7001460</u> Ungar, Sobin, Humphrey, Simchak, Gonzalez & Wahl (2012) "Guiding the Invisible Hand: Policies to Address Market Barriers to Energy Efficiency." Retrieved from <u>https://www.aceee.org/files/proceedings/2012/data/papers/0193-000214.pdf</u> Iowa Administrative Code. 199-35.2 (476) Definitions. (Chapter 35, Page 2) Retrieved from <u>https://www.legis.iowa.gov/docs/iac/rule/10-06-2010.199.35.2.pdf</u>

Date Request Received: November 04, 2022 Data Request No. RR 2-004 Date of Response: December 16, 2022 Page 1 of 1

Request from: New Hampshire Public Utilities Commission

Request:

Topic VII. Reporting on Low-Income Programming

Given the legislative mandate that 20% of total program expenditures be made for the benefit of low-income ratepayers, the Commission solicits information responsive to the following question:

a) Have the participants considered other programmatic designs that are less reliant on the Home Energy Assistance program, and which might provide a more direct financial benefit to low-income ratepayers and/or have greater overall cost-effectiveness?

Response:

The Utilities, in concert with Community Action Agencies and other stakeholders, continually assess the HEA program in an effort to improve services to customers while ensuring cost-effectiveness. For example, the Utilities increased the HEA general maximum rebate level to \$15,000 per home in the 2022-2023 Plan in order to ensure adequate funding to address all energy efficiency upgrades identified during the audit rather than capping services at a lower dollar amount or providing support to the same customer over multiple program years. Best practices for income-eligible programs are detailed in a recently published report from the American Council for an Energy Efficient Economy ("ACEEE"): https://www.aceee.org/research-report/u2205. Many of these best practices are already incorporated in the NHSaves program and the NH Utilities continually look to other jurisdictions to inform what enhancements might be applicable and appropriate for New Hampshire.

Date Request Received: September 12, 2022 Data Request No. RR 1-001 Date of Response: December 16, 2022 Page 1 of 1

Request from: New Hampshire Public Utilities Commission

Request:

Reference reporting requirement i.2 from Order No. 26,621. The Commission notes that the year for the nominal discount rate that has been applied in the Granite State Test (GST) and Total Resource Cost Test (TRCT), per **Lookups** tab in each Benefit Cost submission, varies from the year noted in the separate but associated discount rate tabs that were provided to support those submissions.

The Joint Utilities are requested to clarify the year of the discount rate that was applied into the GST and TRCT tests. For any variation from the latest Prime Rate available at the time of the study, the Joint Utilities are requested to provide an explanation of why more recent discount rates were not leveraged in each submission.

Response:

The Benefit Cost Models are set up to plan three distinct program years, each with its own budget and goals. The years referenced in the "Lookups" tab of the BC model correspond to the three years of the current 2021-2023 triennium. For both the 2021 Plan and 2021 Plan Year Report filed in response to Order No. 26,621, the Prime Rate used was based on '2021'.

The practice of establishing the real discount rate is based on the Prime Rate adjusted annually "on or around June 1" for the year preceding each Plan. For the 2021 Plan, the rate was updated as of June 2020. Those assumptions included in the model for planning purposes, including but not limited to the Prime Rate, the discount rate and avoided energy costs, are then retained for reporting purposes to accurately assess performance compared to the plan.